DocuSign Envelope ID: C6090006-247A-471C-86A1-B1FF4A9F9929

# BID FOR THE ACQUISITION OF AN ISOLATED PRODUCTION UNIT

# EXHIBIT 3.4

Draft of the Sale and Purchase Agreement

# SHARE PURCHASE AGREEMENT VIA ISOLATED PRODUCTION UNIT AND OTHER COVENANTS<sup>1</sup>

# ENTERED INTO, ON ONE SIDE

# OI MÓVEL S.A. - UNDER JUDICIAL REORGANIZATION

# **TELEMAR NORTE LESTE S.A. – UNDER JUDICIAL REORGANIZATION**

# AND, ON THE OTHER SIDE

# [BUYER]

# AND, AS INTERVENING CONSENTING PARTIES

# CALITEIA RJ INFRAESTRUTURA E REDES DE TELECOMUNICAÇÕES S.A.

# AND

# **OI S.A. – UNDER JUDICIAL REORGANIZATION**

# [DATE]

# **Table of Contents**

#### Page

Clause 1	Definitions	7
1.1	Definitions	7
1.2	Other Definitions	15
1.3	Interpretation	16
Clause 2	Transaction	16
2.1	Sale and Purchase	16
2.2	Transfer of Shares	17
Clause 3	Purchase Price and Payment	17

<sup>1</sup> <u>Note for the Draft</u>: Excerpts in brackets to be completed by the Parties with the information applicable at the time of the conclusion of the Agreement.

3.1	Purchase Price	17
3.2	Pre-Closing Adjustment	17
3.3	Payment of the Preliminary Purchase Price	17
3.4	Automatic Discharge	17
3.5	Transaction Taxes	18
3.6	Price Adjustment	18
3.7	Payment of the Price Adjustment	19
Clause 4	<b>Conditions Precedent for the Closing</b>	19
4.1	Conditions Precedent of the Parties	19
4.2	Conditions Precedent of Buyer	19
4.3	Conditions Precedent of Sellers	20
4.4	Notice to CADE	21
4.5	Deadline	22
Clause 5	Closing	22
5.1	Closing	22
5.2	Obligations at Closing	22
5.3	Delivery of Documents	23
Clause 6	<b>Representations and Warranties of Sellers</b>	23
6.1	Representations and Warranties of Sellers	23
Clause 7	<b>Representations and Warranties of Buyer</b>	30
7.1	Representations and Warranties of Buyer	30
Clause 8	Acts between Signing and Closing	31
8.1	Ordinary Course of Business	31
Clause 9	Indemnification	33
9.1	Sellers' Obligation to Indemnify	33
9.2	Buyer's Obligation to Indemnify	34
9.3	Indemnification Procedures	35
9.4	Third-Party Demand	35
9.5	Compliance with Procedures	36
9.6	Payment of the Indemnification	36
9.7	Limitations on Indemnification Obligation	37
9.8	Effective Amount to be Indemnified	37
9.9	Best Efforts	38
Clause 10	Infrastructure Items and Lease Agreements	38
10.1	Full Regularization	38
10.2	Irreversible Situation	38
10.3	Regularization Costs	39
10.5	Overload Irregularities	39
10.6	Sellers' Obligations — Payment Maintenance	39
10.6.1	Revenue Received After Closing	39
10.8	Replacement of Guarantees	40
10.9	New Infrastructure Items	40
10.10	Insurance Hiring	40
Clause 11	Termination Termination Events	<b>40</b>
11.1	Termination Events	40
11.2	Fine CADE Extension	41
11.3	CADE Extension	41

11.4	Termination Effects	41
11.5	Exercise Limitation	41
Clause 12	Notices	41
12.1	Notices	41
12.2	Reception Convention	41
12.3	Change	42
Clause 13	Conflict Resolution	42
13.1	Arbitration	42
Clause 14	Miscellaneous	43
14.1	Amendments	43
14.2	Assignment	43
14.3	Financing	43
14.4	Exhibit Update	43
14.5	Sharing Agreement	44
14.6	Binding Act	44
14.7	Comprehensive Consensus and Documents	44
14.8	Severability	44
14.9	Subsequent Commitment	44
14.9	Press and Public Releases	44
14.11	Unfavorable Interpretation	44
14.12	Confidentiality and Non-Disclosure	44
14.13	Fiduciary Assignment	45
14.14	Liabilities	45
14.15	Intervening Consenting Sellers	45
14.16	Guarantee	45
14.17	Waiver, Termination and Discharge	46
14.18	Forbearance	46
14.19	Liability for Costs	46
14.20	Corporate Reorganization	46
14.21	Specific Performance; Enforceable Instrument	46
14.22	Governing Law	46

# List of Exhibits

Exhibit	Description
Exhibit A	Existing Sharing Agreements
Exhibit B	Infrastructure Items
Exhibit 4.2.1(viii)	Existing Sharing Agreements Subject to Prior Consent
Exhibit 4.2.1(ix)	Lease Agreements Identified
Exhibit 4.4.6(i)	Final Purchase Price Adjustment - Substantial Competitive Restrictions
Exhibit 5.2.1(v)	Closing Instrument
Exhibit 5.2.1(vi)	Oi Sharing Agreement
Exhibit 5.2.1(viii)(A)	Properties of Affiliates of Sellers
Exhibit 5.2.1(viii)(B)	Form and Commercial Conditions of the New Lease Agreements for the Sellers
	Affiliates Real Properties
Exhibit 5.2.1(ix)	Draft of the Temporary Power-of-Attorney for the New Managers
Exhibit 6.1.12	Financial Statements

Exhibit 6.1.16	Relevant Company Agreements
Exhibit 6.1.19(i)	Company's Powers-of-Attorney
Exhibit 6.1.19(ii)	Company's Checking Accounts and Authorized Persons
Exhibit 6.1.20	Litigation

# SHARE PURCHASE AGREEMENT VIA ISOLATED PRODUCTION UNIT AND OTHER COVENANTS

On one side:

I. **OI MÓVEL S.A.** — **UNDER JUDICIAL REORGANIZATION**, a corporation with its head offices in Brasília, Distrito Federal, no Setor Comercial Norte, Quadra 03, Bloco A, Edifício Estação Telefônica, Térreo, Parte 2, ZIP CODE 70.713-900, registered with the National Register of Legal Entities of the Ministry of Economy ("CNPJ/ME") under No. 05.423.963/0001-11, herein represented pursuant to its Bylaws (hereinafter referred to as "<u>Oi</u> <u>Móvel</u>");

And, on the other side:

II. **TELEMAR NORTE LESTE S.A.** — **UNDER JUDICIAL REORGANIZATION**, a corporation with its head office located in Rio de Janeiro, State of Rio de Janeiro, at Rua do Lavradio, n° 71, 2° andar, Centro, ZIP CODE 20.230-070, registered with the CNPJ/ME under No. 33.000.118/0001-79 and in the Commercial Registry of the State of Rio de Janeiro under NIRE 33.30015258-0, herein represented pursuant to its Bylaws (hereinafter referred to as "<u>TMAR</u>"); and, together with Oi Móvel, "<u>Sellers</u>".

And, on the other side:

III. [**BUYER**], with its head offices at [•], registered with the CNPJ/ME under No. [•] (hereinafter referred to as "<u>Buyer</u>");

With Sellers and Buyer hereinafter jointly referred to as "<u>Parties</u>" and individually and indistinctly as "<u>Party</u>".

And, as Intervening Consenting Party:

IV. CALITEIA RJ INFRAESTRUTURA E REDES DE TELECOMUNICAÇÕES S.A., a corporation incorporated on January 13, 2020, with its head offices in Rio de Janeiro, State of Rio de Janeiro, at Rua do Lavradio, n° 71, sl. 201/801, Centro, ZIP CODE 20.230-070, registered with the CNPJ/ME under No. 35.978.982/0001-75 and with the Commercial Registry of the State of Rio de Janeiro under NIRE 33.300.333.215, herein represented pursuant to its Bylaws (hereinafter referred to as "<u>Company</u>");

And also as an intervening, consenting and guarantor of the obligations of Sellers:

V. **OI S.A. - UNDER JUDICIAL REORGANIZATION**, a publicly-held company, with its head offices in Rio de Janeiro, State of Rio de Janeiro, at Rua do Lavradio n.° 71, Centro, ZIP CODE 20.230-070, registered with the CNPJ/ME under No. 76.535.764/0001-43, herein represented pursuant of its Bylaws (hereinafter referred to as "<u>Oi</u>" and, together with Company, the "<u>Intervening Consenting Sellers</u>").

# WHEREAS:

**A)** On June 20, 2016, Oi, together with other companies in its economic group, filed for judicial reorganization before the 7<sup>th</sup> Lower Commercial Court of the Judicial District of the Capital City of the State of Rio de Janeiro, under No. 0203711-65.2016.8.19.0001 ("Judicial <u>Reorganization</u>");

**B**) Within the scope of the Judicial Reorganization, the Judicial Reorganization Plan expressly sets forth the possibility of disposal, by Sellers, of movable or immovable property that is part of its permanent assets, by means of constituting an UPI under articles 60, 66, 140, item II, 141 and 142, item II, of the LFR;

C) On June 15, 2020 and on [•], 2020, Oi and Sellers, together with other companies in its economic group, filed before the 7<sup>th</sup> Lower Commercial Court of the Judicial District of the Capital City of the State of Rio de Janeiro ("Judicial Reorganization Court") a request for Amendment to its Judicial Reorganization Plan ("<u>Amendment to the PRJ</u>"), which provided for (i) the creation of an Isolated Production Unit (UPI) formed by the shares issued by Company, to which all the Infrastructure Items were poured and to which the Licenses for the Site, Lease Agreements and Existing Sharing Agreements relating to such Infrastructure Items were or will be assigned and transferred, as the case may be, subject to the provisions of this Agreement; and (ii) the sale of one hundred percent (100%) of the Shares issued by Company in a competitive proceeding;

**D**) The Amendment to the PRJ was approved at the Creditors' General Meeting held on [•], 2020 and ratified by the Judicial Reorganization Court on [•], 2020;

**E**) As provided for in the Judicial Reorganization Plan and Amendment to the PRJ, as preparatory acts for the sale of Company and the Infrastructure Items through UPI, Sellers and their Affiliates approved and implemented the Corporate Reorganization;

**F**) Upon completion of the Corporate Reorganization, Company started to hold all assets, liabilities, contracts, Licenses, rights and obligations necessary to conduct the business of development, maintenance, lease and assignment for use of the Infrastructure Items (" $\underline{Business}$ ");

**G**) The sale of Company (and, therefore, of the Infrastructure Items) through UPI was subject to auction of closed bids, under the terms of [Clause 5.3.8.2] of the Amendment to the PRJ and articles 60, 66, 140, item II, 141 and 142, item II, of LFR ("<u>Auction</u>"), according to the public notice published on [*date*] ("<u>Public Notice</u>");

**H**) Buyer's bid, submitted on the Auction date, was declared the winner in the Auction and ratified by the Judicial Reorganization Court on [*date*];

**I**) Buyer wishes to acquire from Sellers all the shares issued by Company representing the entire share capital of Company ("<u>Shares</u>"), free of any liens and successions from Sellers and any of their Affiliates or Related Parties, pursuant to articles 60, 66, 140, item II, 141 and 142, item II, of the LFR and subject to the terms and conditions agreed upon in this instrument; and

**J**) The Infrastructure Items are the subject matter of Infrastructure Sharing and/or Assignment for Use Agreements, or other Contracts that formalize the use of the space and installed capacity of these Infrastructure Items by Sellers, their Affiliates and/or Third Parties, as the case may be, as indicated in Exhibit A to this Agreement ("Existing Sharing Agreements").

Now, therefore, the Parties hereby **DECIDE** to enter into this Share Purchase Agreement via UPI and Other Covenants ("<u>Agreement</u>"), pursuant to the following clauses and conditions.

# Clause 1. Definitions

1.1. **<u>Definitions</u>**. For purpose of this Agreement, the terms below, whenever in capital initials, shall have the meanings attributed thereto in this Clause 1.1:

"<u>Affiliate</u>" means, in relation to any Person, any Person who directly or indirectly Controls, is Controlled by, or is under common Control with that Person, provided that Persons in the portfolio of funds managed by [*name*] will not be considered Affiliates of Buyer.

"<u>Price Adjustment</u>" means the amount equivalent to the Final Purchase Price minus the Preliminary Purchase Price.

["<u>CADE Approval</u>" means whichever happens first among (a) the certification of the end of the term for appeal of fifteen (15) days counted from the date of publication in the Official Gazette of Brazil of the decision for approval of the Transaction by the General Superintendence Office of CADE, without any appeal or measure having been filed before CADE or by any Third Party with standing for such in the Transaction analysis proceedings and/or without any *certiorari* by any member of the CADE administrative tribunal; or (b) the certification of the final and unappealable decision by the CADE Administrative Tribunal to approve the Transaction, in case the case records are submitted to its analysis, in any of the events set forth in Law No. 12,529/11 and the CADE regulations.]<sup>2</sup>

"Leased Area" means each area of property or coverage of property leased, subleased, assigned, occupied or used in any capacity by Company for the implementation, maintenance, operation, lease, sharing or assignment for use of an Infrastructure Item

 $<sup>^{2}</sup>$  <u>Note for the Draft</u>: Provisions related to antitrust approvals to be excluded if not applicable at the time of executing this Agreement.

(including extensions of land or other immovable property in which the Infrastructure Items are located).

"Independent Auditor" means any one amongst (a) Deloitte, (b) KPMG, (c) E&Y, (d) PWC or (e) BDO.

"<u>Governmental Authority</u>" means any governmental agency, independent agency, division, department, court or other federal, state or municipal government body of the direct or indirect public administration, as well as any and all courts, judicial authorities and/or arbitration tribunal.

"**Brazil**" means the Federative Republic of Brazil.

"<u>CADE</u>" means the Administrative Council for Economic Defense.

"<u>Cash</u>" means the consolidated position of cash deposits and immediately available funds, deposited in checking accounts, savings accounts, investment funds, all financial investments and funds deposited with financial institutions and equivalents with availability within thirty (30) days. Cash must exclude any cash retained and not freely available for being subject to any restrictions, limitations relating to use by Law, Contractual Instrument or otherwise, including guarantees for loans or subject to judicial deposits.

"<u>Working Capital</u>" means, without any duplicity, (a) the balance of the current asset accounts – namely: (i) accounts receivable, (ii) inventories, and (iii) other operating accounts for current assets; minus (b) the balance of current liability accounts, namely: (i) suppliers, (ii) salaries and charges, (iii) taxes payable, and (iv) any other operating accounts for current liabilities, in each case calculated in accordance with the Brazilian GAAP. For avoidance of doubt, the definition of Working Capital shall not consider the amounts included in the calculation of Cash or Debt.

"Brazilian Civil Code" means Brazilian Federal Law No. 10,406, dated January 10, 2002, as amended.

"Brazilian Code of Civil Procedure" means Brazilian Federal Law No. 13,105, dated March 15, 2015, as amended.

"<u>Knowledge</u>" means, with respect to any Person, (i) the actual knowledge of that Person and/or any manager (including non-statutory director) of such Person, or (ii) the knowledge that is required from any of the Persons referred to in item "(i)" above due to Law.

"<u>Oi Sharing Agreement</u>" means the agreement to be entered into between Company, in the capacity of contractor, Oi Móvel, in the capacity of Principal and Oi, in the capacity of intervening guarantor, to regulate the use of the Infrastructure Items as of Closing, pursuant to <u>Exhibit 5.2.1(vi)</u>.

"Lease Agreements" means the contracts (written or verbal) related to the lease or occupation of the respective Leased Areas and spaces in which the Infrastructure Items are

installed, such as specific lease agreements, loan for use (*comodato*), assignment for use of space, among others. The Lease Agreements that are formalized in writing and whose copies signed by the respective signatory parties have been duly identified by Sellers until the date hereof are described and specified in Exhibit 4.2.1(ix) (such Lease Agreements jointly referred to as "Identified Lease Agreements").

"Relevant Agreements" means, altogether: (i) the Existing Sharing Agreements, (ii) the Lease Agreements, and (iii) the types of Contracts listed below: (a) service agreements or employment contracts that cannot be terminated unilaterally without penalty; (b) collective labor or collective bargaining agreements entered into with unions; (c) agreements establishing investment or CAPEX commitments; (d) agreements related to purchase of assets, right to use, isolated production units or line of business; (e) real property purchase agreements; (f) shareholders' agreements or members' agreements or similar instruments; (g) investment contracts in, or purchase and sale of, shares, quotas, equity interests or any other security or operating businesses or material assets; (h) business sale agreements or goodwill purchase agreements; (i) contracts of any nature that establish or agree to any obligations to pay and/or right to receive, in the next twelve (12) months from the date hereof, amounts in aggregate value equal to or greater than five hundred thousand reais (BRL 500,000.00); (j) contracts constituting, formalizing, granting or in which there is an assumption of Debt and/or constituting any type of Lien; (k) contracts limiting or prohibiting the non-occurrence of Debt, distribution of earnings, dividends and/or retention of profits; (1) judicial or extrajudicial agreements to avoid or end Demands; (m) licensing, assignment, sharing, permission or transfer agreements for intellectual property or other intangible assets; (n) call option or put option contracts that grant any Person the right of first refusal, peremption or similar rights; (o) Profit Sharing agreements, or association, operation, partnership (commercial, operating or other) and/or joint-venture exploration agreements; (p) contracts providing for or agreeing upon the payment of royalties; (q) contracts entered into with any Governmental Authority; (r) agreements of any kind entered into between, on one side, Company and, on the other side, Sellers, their Affiliates and/or their Related Parties, including, but not limited to, lease agreements and cost-sharing agreements; (s) contracts for, or containing (i) obligations of secrecy or confidentiality, (ii) obligations of exclusivity and/or which limit or prohibit the performance or expansion of activities, operations or business (territorial or any other type) and/or (iii) pact or obligation for non-competition or non-compete or that prevent the hiring of managers, workers, employees or other Persons (non-solicit agreements); and/or (iv) obligation to dispose of material assets to conduct and transact the Business; (v) restrictions on the sale of Shares, Infrastructure Items and/or Business and that therefore require the consent of counterparties in such Contract Instruments is necessary for this purpose; and/or (t) any Contractual Instrument, pre-contract or promise to constitute or execute, directly or indirectly, any of the contracts described in items "(a)" through "(s)", even if effectively or with suspended or conditioned execution. The Material Contracts entered into, assigned or that will be assigned to Company as a result of the Corporate Reorganization and/or involving, directly or indirectly, the Business or Assets of Company, are listed in Exhibit 6.1.16.

"<u>Control</u>" means, in relation to a Person, the power to, directly or indirectly, individually or jointly with other Persons, manage and order the direction of such Person's management and policies, by means of ownership of the majority of its voting capital, by operation of a

Contractual Instrument, or by any other means. Terms derived from Control, such as "Parent", "Controlling", "Controller" and other related words will have meanings similar to Control.

"Transaction Costs" means, insofar as they occur before the Closing or simultaneously to the Closing, or the respective generating event is prior or simultaneous to the Closing, and provided that they have not been paid by Company before the Closing, without duplication, the following costs incurred or borne by Company: (i) all fees and expenses incurred or subject to reimbursement in connection with the preparation, negotiation or execution of this Agreement or the consummation of the transactions contemplated by this Agreement, or in connection with or in anticipation of any alternative transactions considered in relation to Company and/or the Business, including any costs related to the Auction, fees, expenses, commissions and costs payable or reimbursable by Company to financial advisers, brokers, banks, consultants, accountants, court fees, legal advisers or other consultants, professionals or service providers; (ii) any contracted payments due related to the change of control, transaction bonus or similar payments or indemnification that are or become due to any employee, independent service provider, consultant, executive or manager of Company and/or its Affiliates (if applicable) in connection with the transactions contemplated by this Agreement; (iii) any costs, fees or expenses related to the Corporate Reorganization (including the registration of the Corporate Reorganization documents with JUCESP or any other Governmental Authority); and (iv) any costs or expenses incurred by Company in conducting the Auction or any costs or expenses related to the procedures contemplated by the Judicial Reorganization Plan or the Amendment to the PRJ; provided that, however, such transaction costs shall not include: (a) any Debt; or (b) any amounts included in the calculation of the Working Capital.

"<u>Ordinary Course</u>" in relation to Company, means conducting its activities (a) in a consistent manner in nature, scope and magnitude with past practices, and (b) in compliance, in all its relevant aspects, with the applicable law.

"<u>Sellers' Fundamental Representations and Warranties</u>" mean the representations made by Sellers in Clauses 6.1.1 (Corporate Good Standing), 6.1.3 (Ownership), 6.1.4 (Power, Corporate Jurisdiction and Approvals), 6.1.5 (Sellers' Bylaws and Judicial Reorganization Plan), 6.1.6 (Outstanding Securities and Issuance of Shares), 6.1.10 (Solvency), 6.1.26 (Pre-Operating Company), 6.1.27 (Brokerage Commission) and 6.1.28 (Good Practices).

"<u>Demand</u>" means any action, proceeding, deficiency notice, order, judicial or extrajudicial notification, notice of violation or non-compliance, claim, notice, plea, complaint, investigation, execution, judicial or administrative proceeding, arbitration proceeding or any investigation of any nature.

"<u>Business Day</u>" means any day (i) that is not a Saturday or Sunday, or (ii) when commercial banks are open for business at regular business hours in the municipality of Rio de Janeiro, State of Rio de Janeiro, in the municipality of São Paulo, State of São Paulo, in Brazil, and in New York, State of New York, in the United States of America.

"Drop-Down of Indoor Sites" means the increase in the Company's share capital decided on [date], which was subscribed by Oi Móvel and paid in by Oi Móvel through the contribution, assignment and transfer of the Indoor Sites to Company's capital, so that Company became the legitimate holder and owner of said Indoor Sites.

"Drop-Down of Outdoor Sites" means the increase in the Company's share capital decided on [date], which was subscribed by TMAR and paid in by TMAR through the contribution, assignment and transfer of the Outdoor Sites to Company's capital, so that Company became the legitimate holder and owner of said Outdoor Sites.

"Material Adverse Effect" means any circumstance, condition, event, change or occurrence ("Event") that, individually or in conjunction with one or more Events, adversely and materially affects (i) the Business, financial or operating conditions, assets and/or the results of the Company's operations; or (ii) the ability of the Company and the Parties to comply with the obligations assumed herein and to consummate the legal transactions provided for herein, within the terms set forth in this instrument. Without limiting the foregoing, any Events that, individually or in conjunction with one or more Events, result in a Loss (and, notwithstanding the definition of Loss and exclusively for the purposes of this definition of Material Adverse Effect, it will include loss of profits and loss of revenues), contingent or effective, negative financial impacts, imposition of payments or disbursements in an amount equal to or greater than one hundred million reais (BRL 100,000,000.00), will necessarily be considered and interpreted as a "Material Adverse Effect" for the purposes of this Agreement. However, no Event will be considered a "Material Adverse Effect" for the purposes of this Agreement when arising from or attributable to any of the factors listed below, either alone or in conjunction with each other: (a) general changes in economic or political conditions that generally affect the sectors or market in which Company and Buyer operate; (b) acts of war (declared or not), sabotage or terrorism, military actions or military escalation after the date hereof, (c) foreseeable developments, on the date of execution of this Agreement, of pandemics and public health crises, or (d) any changes in the applicable laws or accounting or tax rules, which may occur after the date hereof; provided, however, that any Event arising from or attributable to the events "(a)" through "(d)" above will not be disregarded for the purposes of determining a Material Adverse Effect only if and to the extent that such an Event has a disproportionate adverse impact on the business or operations of Company, when compared to other Persons operating in the same sectors and markets in which these Persons operate.

"Debt" means, in relation to a Person: (i) all long and short term debts, past due and unpaid and due, including with financial institutions and suppliers, regardless of whether they are secured or not; (ii) any overdue amounts owed to Employees and service providers that have not been paid by the originally agreed term; (iii) all overdue amounts, in installments or not, due and not paid to federal, state or municipal tax and contribution collection agencies; (iv) any overdue accounts payable; (v) any and all operating and/or financial leasing transactions contracted and in force; (vi) any dividends, interest on the stockholders' equity, other pecuniary advantages and/or any other form of profit sharing, in cash or in kind declared and unpaid; (vii) any Transaction Costs; and (viii) any and all amounts related to interest incurred up to the date of assessment of the Debt, as well as any fines incurred and not yet incorporated into the principal amount; in any case, calculated under Brazilian GAAP. "Operator Equipment" means, all together, the generators, telecommunications, radio and other equipment present in the structures of any Infrastructure Item and which are the property of Sellers and/or their Affiliates, as well as their antennas and/or connection cables for such equipment that are apt to provide Personal Mobile Service and/or Specialized Mobile Service. Operator Equipment include, but not limited to, dedicated equipment, power cables for Sellers' and their Subsidiaries' shelter/container, fiber optic cables, radio frequency and transmission antennas and their cables, exclusive and specific grounding system for these equipment and associated grounding devices with coaxial cables, shelter / container / Outdoor Base Radio Station of Sellers and their Affiliates, all electronics and batteries associated with the shelter / container / Outdoor Base Radio Station of Sellers and their Affiliates, specific internal and external grounding of the shelter / container / Outdoor Base Radio Station of Sellers and/or their Affiliates (including the physical grounding material and the physical connection between the Operator Equipment and the general grounding system of the site), generators, diesel tanks, active and passive security system, telephone lines, COW (Cell on Wheels) base radio station, and any substations and power transformers dedicated solely to Sellers and/or their Affiliates, all energy contracts related to the Operator Equipment, as well as the operating rooms and masonry shelters, but always excluding their respective foundations.

"<u>Site Equipment</u>" means the physical assets (with the exception of the property, shelters, containers or other types of equipment), located in a given Infrastructure Item, which are utilized for the use, operation, exploration or management of the Infrastructure Item in question.

"<u>Base Radio Station</u>" means the equipment of the Personal Mobile Service or Specialized Mobile Service network located in an Infrastructure Item that is responsible for communicating with the terminal equipment of the users of these services, through the transmission and reception of signals in the frequency bands authorized for the Personal Mobile Service or Specialized Mobile Service.

"<u>Closing</u>" means the closing of the Transaction, by means of the assignment and transfer of the Shares to Buyer and practice of the other acts provided for in Clause 5.2.

"**Brazilian GAAP**" means the accounting principles generally accepted in Brazil, based on the Brazilian Corporations Law, the regulations of the Federal Accounting Council - CFC, the pronouncements of the Accounting Pronouncement Committee - CPC, and the regulations of the Brazilian Securities Commission - CVM.

"<u>Lien</u>" means any encumbrances, claims, pledges, liens, security interests, options, right of first refusal, rights of first offer, debts, charges or restrictions, shareholders' agreements or of any other type (including as related to the use, voting, transfer, receipt of earnings or exercise of any of the ownership rights).

"<u>Dommo Merger</u>" means the merger of **Dommo Empreendimentos Imobiliários Ltda.**, a limited liability company with head offices in the City of São Paulo, State of São Paulo, at Avenida das Nações Unidas, n.º 12.901, 27° andar, Brooklin Paulista, ZIP CODE 04578-

910, and registered with the CNPJ/ME under No. 04.034.792/0001-76 ("Dommo") into TMAR, carried out on July 1, 2020, whereby TMAR succeeded Dommo in all the right and obligations and Dommo, in turn, was extinguished by operation of law, under the terms and as provided for in the Brazilian Corporations Law and the Brazilian Civil Code.

"<u>Sensitive Competition-Related Information</u>" means any and all data and information that may in any way influence the Parties' business decision-making process and business strategies, and influence their ability to compete independently in the market, including, but not limited to: (i) business plans, budgets, undisclosed/published accounting information, information regarding production capacity, costs, terms of contracts with suppliers, among others; and (ii) any industrial, trade or any other type of secrets related to the activities carried out, as well as information on assets, products, production levels, pricing and discount strategies, technology, market studies, customer lists, commercial and/or marketing strategies, future expansions or business, logistics, employee compensation, non-public information about brands and patents (R&D), among other.

"<u>Minimum Revenue Index</u>" means a set of Infrastructure Items that, in aggregate, represent eighty percent (80%) of the Annual Pre-Closing Revenue.

"<u>Contracts</u>" means any agreement, contract, order, commitment or other instrument entered into by any Person.

"<u>IPCA/IBGE</u>" means the Broad Consumer Price Index — IPCA, disclosed by the Brazilian Institute of Geography and Statistics — IBGE, or another index that may replace it.

"Infrastructure Items" means, specifically and altogether, the telecommunication infrastructure sites (including, therefore, Outdoor Sites and Indoor Sites) specified in Exhibit  $\underline{B}$ , which are comprised of the set including a Tower and/or Roof Top. For the avoidance of doubt, the definition of Infrastructure Items expressly excludes any and all Operator Equipment, but includes Site Equipment.

"Law" means any law, code, ordinance, rule, resolution, normative instruction, regulation, treaty, convention and/or any other determination, order, writ, injunction, decision, judgment, award and/or decree issued by any competent Governmental Authority regarding any Party and/or their respective assets.

"**Brazilian Corporations Law**" means Brazilian Federal Law No. 6,404, dated December 15, 1976, as amended.

"<u>Brazilian Anti-Corruption Laws</u>" means, as amended from time to time, (i) all Brazilian Laws on corruption, bribery, fraud, conflict of public interests, misconduct, violation of public bidding and procurement, money laundering, political or electoral donations, or management of businesses without commitment to ethics, including, among others, Decree-Law No. 2,848/1940 (Criminal Code), Brazilian Federal Law No. 8,429/1992 (Law on Misconduct in Office), Brazilian Federal Law No. 9,504/1997 (Electoral Law), Brazilian Federal Law No. 12,813/2013 (Conflict of Interest Law), Brazilian Federal Law No. 9,613/1998 (Money

Laundering Act) and Brazilian Federal Law No. 12,846/2013 (Anti-Corruption Law), as amended by Federal Decree No. 8,420/2015 (Anti-Corruption Decree); (ii) the Foreign Corrupt Practices Act of 1977 of the United States of America; and (iii) the list of Specially Designated Nationals and Blocked Persons kept and frequently updated by OFAC, as well as any act imposed or applied by OFAC.

"<u>LFR</u>" means Law No. 11,101, dated February 9, 2005, as amended.

"<u>License</u>" means any and all authorization, permit, consent, approval, order, resolution, license, concession, permission, exemption, declaration, grant, certification, decree, registration and/or annotation issued, granted or conferred by any Governmental Authority, including, but not limited to, environmental, regulatory and urban licenses.

"<u>Site License</u>" means any and all applicable License, as necessary and/or required by Law for the construction, implementation, installation, lease, use, exploration, operation and management of the Infrastructure Items and/or for occupation of the Leased Areas.

"<u>Verification Moment</u>" means 11:59 pm (Brasília standard time) of the day before the Closing Date.

"<u>OFAC</u>" means the Office of Foreign Assets Control of the U.S. Department of the Treasury.

"Indemnified Party" means Buyer Indemnified Party or Sellers Indemnified Party, as the case may be.

"**<u>Related Party</u>**" means, with regard to any Person, any of its Affiliates, as well as (i) in case of a legal entity, or other organization with or without legal personality, (a) its managers, including non-statutory directors, and their respective spouses and/or relatives up to the  $3^{rd}$ degree (and respective spouses thereof); and (b) managers, including non-statutory directors, and their respective spouses and/or relatives up to the  $3^{rd}$  degree (and respective spouses thereof) of any Affiliate of the Person in question, (ii) in case of an individual, the spouse and/or relatives up to the  $3^{rd}$  degree (and respective spouses thereof) of such Person, or (iii) any legal entity in which any of the Persons mentioned above holds, individually or jointly, directly or indirectly, more than ten percent (10%) of its share capital (or equivalent equity interest thereon).

"<u>Person</u>" means any individual, legal entity, businessperson, general partnership or company (including, but not limited to, corporations, limited liability companies or other types of companies), foundation, investment fund, association, partnership, consortium, trust, fiduciary entity or any other entity or organization, with or without legal personality, or any Governmental Authority.

"Loss" means any and all losses, damages, liabilities (including obligations related to Taxes), shortcomings, obligations, losses, penalties, encumbrances, claims, actions, proceedings, decisions (judicial, administrative, arbitration or similar), judgments, awards, agreements and, provided that they are reasonable and documented, costs, expenses and disbursements (including, but not limited to, court deposits, attorneys', accountants' and experts' fees, costs

of loss of suit and administrative fees/costs, including those incurred in the conduct of any proceeding, action or Demand) of any type or nature, as well as adjustment for inflation, default and/or compensatory interest, fines and any other additions and/or penalties applicable, whether by operation of law or Contractual Instrument. The following are expressly excluded from the definition of "Loss": moral damages, indirect damages, image damages, loss of profits and losses of revenue.

"Judicial Reorganization Plan" means the judicial reorganization plan of Oi and its direct and indirect subsidiaries TMAR, Oi Móvel, Portugal Telecom International Finance BV — Under Judicial Reorganization, and Oi Brasil Holdings Cooperatief UA — Under Judicial Reorganization — jointly with companies COPART 4 Participações S.A. — Under Judicial Reorganization and COPART 5 Participações S.A. — Under Judicial Reorganization, which were subsequently merged, respectively, into Oi and TMAR — approved at a creditors' general meeting held on December 19 and 20, 2017 and ratified by the Judicial Reorganization Court on January 8, 2018, as amended pursuant to the Amendment to the PRJ approved at the Creditors' General Meeting held on [*date*].

"<u>Annual Revenue Price Disclosed</u>" means the amount of  $[\bullet]$  reais (BRL  $[\bullet]$ )<sup>3</sup>, corresponding to the sum of the net revenue accumulated in April, May and June 2020 from the Outdoor Sites (which corresponds to the amount of  $[\bullet]$  reais (BRL  $[\bullet]$ ))<sup>4</sup>) and the Indoor Sites (which corresponds to the amount of  $[\bullet]$  reais (BRL  $[\bullet]$ ))<sup>5</sup>, <u>multiplied</u> by four (4), to establish the revenue in annual terms, as disclosed by Sellers in the scope of the Auction and contained in the Public Notice. The Annual Revenue Price Disclosed will be subject to the adjustments provided for in Clause 3 below.

"<u>Annual Indoor Revenue Price</u>" means (i) the sum of the accumulated net revenue of the Indoor Sites in the last three (3) full months immediately prior to the month in which the respective calculation will occur, as adjusted to exclude the effects of revenues arising from expired or extinct Contracts (either by decision of either party or due to expiration thereof), (ii) <u>multiplied</u> by four (4), to establish such revenue in annual terms; and (iii) <u>multiplied</u> by  $[\bullet] ([\bullet])^6$ .

"<u>Annual Outdoor Revenue Price</u>" means (i) the sum of the accumulated net revenue of the Outdoor Sites in the last three (3) full months immediately prior to the month in which the respective calculation will occur, as adjusted to exclude the effects of revenues arising from expired or extinct Contracts (either by decision of either party or due to expiration thereof), (ii) <u>multiplied</u> by four (4), to establish such revenue in annual terms; and (iii) <u>multiplied</u> by  $[\bullet] ([\bullet])^7$ .

<sup>&</sup>lt;sup>3</sup> <u>Note for the Draft:</u> To be filled in with Annual Revenue Disclosed in the Auction/Public Notice.

<sup>&</sup>lt;sup>4</sup> <u>Note for the Draft:</u> To be filled in with the amounts disclosed by Sellers in the Public Notice.

<sup>&</sup>lt;sup>5</sup> Note for the Draft: To be filled in with the amounts disclosed by Sellers in the Public Notice.

<sup>&</sup>lt;sup>6</sup> <u>Note for the Draft:</u> To be filled in with the multiple of the Indoor Sites offered in the winning bid of the Auction.

<sup>&</sup>lt;sup>7</sup> <u>Note for the Draft:</u> To be filled in with the multiple of the Outdoor Sites offered in the winning bid of the Auction.

"<u>Annual Pre-Closing Revenue Price</u>" means the Annual Revenue to be estimated by Sellers with the calculation base date being the Closing Date, in the manner and subject to the provisions in Clause 3 below, provided that, for the purposes of calculating the Annual Pre-Closing Revenue, Sellers will hire an Independent Auditor to calculate the Annual Outdoor Revenue and Annual Indoor Revenue of at least one (1) of the three (3) months used as the basis for the calculation of said Annual Pre-Closing Revenue and that Sellers will deliver supporting documentation, including proof of payments, to Buyer as provided for in this Agreement.

"**Full Regularization**" means, in relation to a Lease Agreement, (a) the formalization, by means of a written Contractual Instrument, of the occupation and/or lease by Company of the Leased Area(s) pertinent to the respective Lease Agreement; (b) the assignment and transfer of the Lease Agreement to Company (as lessee, assignee and/or grantee), under the same commercial conditions currently in force (including term and other terms and conditions), except for annual remuneration, which may vary up to five percent (5%) compared to the value in effect on the date hereof (any amounts of the annual remuneration mentioned above that exceed the percentage of five percent (5%) referred to above, "Regularization Cost Increase"); and (c) the obtainment of consent by the counterparty to said Lease Agreements, where applicable according to their terms, for the implementation of the Corporate Reorganization and the Closing and performance of the acts provided for in item "(b)" above. For clarification purposes, the value in effect, on the date hereof, of the annual remuneration of the Lease Agreements related to the Lease Areas, is broken down, individually for each Infrastructure Item, in the corresponding column "Current Lease Value" in Exhibit  $B^8$ .

"<u>Corporate Reorganization</u>" means, altogether, the following operations and legal transactions: (a) Dommo Merger; (b) Drop-Down of Outdoor Sites; and (c) Drop-Down of Indoor Sites.

["<u>Substantial Competitive Restrictions</u>" means, for the purposes of this Agreement, the following measures: (a) the divestment, sale, disposal of, burdening, cancellation of contracts or obligation to license any transaction, business or asset of Buyer and/or its Affiliates; (b) the sale or disposal of any portion of Company's operations, businesses or assets; or (c) disbursement of any amounts that cause any substantial adverse change, compared to the situation of Company or Buyer and/or their Affiliates on the date of execution of this Agreement, with respect to the business, assets, condition (financial, equity or other), operating prospects or results of the Company or Buyer and/or their Affiliates, as the case may be, which exclusively with respect to item "(b)" above represent any restrictions on operations, businesses or assets of Company that account for more than twenty percent (20%) of the Annual Revenue Disclosed generated by all Infrastructure Items.]<sup>9</sup>

<sup>&</sup>lt;sup>8</sup> <u>Note for the Draft:</u> To be filled in with the information on the base date of executing the agreement. Information supporting the base values will be submitted for purposes of validating such values prior to execution of the agreement.

<sup>&</sup>lt;sup>9</sup> <u>Note for the Draft:</u> Provisions related to antitrust approvals to be excluded if not applicable at the time of executing this Agreement.

"**Roof Tops**" means, in infrastructure, the set of structures and spaces allocated to bear the radiofrequency and transmission antennas, with safety and within the acceptable limits for angular deformation (bending plus torsion) and their respective accessories: support and fixation base for the vertical structures, lighting (including light barrier, photocell and wiring controls), pedalboards (including fall arrest safety cable), vertical or horizontal cable bed, grounding system (including lightening rod, wires and ground connections and the specific grounding mesh for the structure), beacons, access stairways and platforms, fixation/support structure for the equipment, switchboards, power piping and electrical cables. Roof Tops are usually assembled on: (i) slabs of building roofs; (ii) roofs; (iii) water tanks; (iv) facades (antennas only); and (v) any combination of the items above. Additionally, the characterization of Roof Tops considers that they have a fully functional Base Radio Station owned by Sellers, its respective Affiliates and/or Third Parties.

"<u>Corporate Reorganization Oi Companies</u>" mean, altogether, the following companies: (i) TMAR; and (ii) Oi Móvel.

"Indoor Sites" means, altogether, the Infrastructure Items identified in Exhibit B in the "Indoor Sites" tab.

"<u>Outdoor Sites</u>" means, altogether, the Infrastructure Items identified in Exhibit B in the "Outdoor Sites" tab.

"DI Rate" means the annual average rate (considering a year of 252 business days) related to transactions with Interbank Deposit Certificates – CDI, with a term equivalent to one (1) Business Day (over), assessed and disclosed on a daily basis by B3 S.A. – Brasil, Bolsa, Balcão, by rounding the daily factor in the eighth decimal place. If, for any reason, the DI Rate is extinguished, replaced or not disclosed, the interest rate that officially replaces it, or, in its absence, any other that best reflects the average variation of the raising costs in the national interbank market shall be applied.

"TED" means electronic transfer of funds immediately available.

"Third Party" means any Person other than the Parties.

"**Towers**" means every structural set capable of supporting the installation of transmission and radiofrequency antennas with safety and within the acceptable limits for angular deformation - bending plus torsion, including the tower structure, the foundation of the tower structure, the tower lighting (including light barrier, photocell and wiring controls, cables), the tower work platform, all antenna supports and tower equipment, tower rest platforms, stairways to the tower (including the fall arrest safety cable, body guard, bracings, vertical and horizontal stretching, the general grounding system of the tower (including lightning rods, wires and ground connectors for the tower and the land grounding mesh), site grounding system (including the global grounding system for the site regarding fences, walls, doors, recipients, gates and energy inputs), power input panel where the meters are located, concrete foundations and/or metal shelters for power input, power infrastructure from the utility company's distribution network, the power input pattern, including ducts, posts and power piping and fiber optic, junction boxes as well as the materials related to the site perimeter (such as walls, fences, gates, etc.), metal skids for the Base Radio Station, concrete base for the Base Radio Station, metallic "eco box" (structure in metal profiles and checkered-plate floor and variable sizes) for the Base Radio Station, site lighting system, industrial outlet for a generator (steck), excluding any equipment that are installed in or attached to the Tower. Additionally, the characterization of Towers considers that they have a fully functional Base Radio Station owned by Sellers, its Affiliates and/or Third Parties.

"<u>Taxes</u>" mean all taxes, contributions, charges, rates, fees, duties, social contributions or other governmental charges of any nature, including, without limitation, all income taxes, withheld at the source or not, on capital gains, share capital, transfer, sale, use, occupation, ownership, consumption, franchise, severance, paid leave, payroll, federal, state, municipal and local taxes withheld at source and other taxes, as well as accretions, fines and interest in relation to any said amounts.

"<u>UPI</u>" means an isolated production unit, pursuant to article 60 of the LFR and the Judicial Reorganization Plan.

"<u>Annual Revenue Amount</u>" means the sum of the Annual Outdoor Revenue Price and the Annual Indoor Revenue Price, in both cases calculated as described in this Agreement, on the same date and with respect to the same calculation period.

1.2. <u>Other Definitions</u>. Without prejudice and in addition to the terms defined in Clause 1.1 above, the following terms are defined in the wording of the Agreement:

<u>"Shares"</u>	7
"Amendment to the PRJ"	6
"Closing AGE"	23
"Arbitration"	42
"Company Assets"	26
"Regularization Cost Increase"	13
"Final Cash"	18
"Pre-Closing Cash"	17
"Chamber"	42
"Final Working Capital"	18
"Pre-Closing Working Capital"	17
"Closing Certificate"	18
"Pre-Closing Certificate"	17
"CNPJ/ME"	6
"Company"	6
"Buyer"	6
"Exhibit Update Communication"	44
"Buyer Conditions Precedent"	19
"Closing Conditions Precedent"	20
"Parties Conditions Precedent"	19
<u>"Sellers Conditions Precedent"</u>	20
<u>"Agreement"</u>	7
"Existing Sharing Agreements"	7

"Lease Agreements Identified"	8
"Agreements Pending Regularization"	38
"Relevant Company Agreements"	28
"Regularization Costs"	39
"Adjustment Payment Date"	19
"Closing Date"	22
"Closing Deadline"	22
"Defense"	35
"Third Party Demand"	35
"Financial Statements"	26
"Dispute"	42
"Dommo"	11
"Public Notice"	7
"Employees"	29
"Final Debt"	18
"Pre-Closing Debt"	17
"Event"	10
"Sellers Affiliates Real Properties"	23
"Confidential Information"	45
"Intervening Consenting Sellers"	6
"Overload Irregularities"	39
"Sites Irregularities"	25
"Judicial Reorganization Court"	6
"Auction"	7
<u>"Maximum Limit"</u>	37
<u>"Business"</u>	7
"Notice of Third Party Demand"	35
"Notice of Indemnification"	35
"Notice of Procedure"	35
"Notice of Irreversible Situation"	39
"Notice of Sellers Disagreement"	18
<u>"Oi Móvel"</u>	6
<u>Oi</u>	6
"Transaction"	17
<u>"Term Installment"</u>	17
<u>"Cash Installment"</u>	17
"Purchase Price Installments"	17
"Buyer Indemnified Party"	33
"Sellers Indemnified Party"	34
"Indemnifying Party"	35
<u>"Party"</u>	6
"Buyer Indemnified Parties"	33
"Sellers Indemnified Parties"	34
"Parties"	6
"Global De Minimis Amount"	37
"Individual De Minimis Amount"	37
"Amicable Negotiation Period"	18

"Base Price"	17
"Final Purchase Price"	17
"Preliminary Purchase Price"	17
"Final Annual Revenue"	18
"Revenue Received"	39
"Judicial Reorganization"	6
"Regulation"	42
"Final Adjustment Report"	18
"Positive Balance"	19
"Irreversible Situation"	38
"TMAR"	6
"Tribunal"	42
"Sellers"	6

1.3. <u>Interpretation</u>. The titles and headings contained in this Agreement are included for convenience purposes only and shall not affect in any way whatsoever the meaning or interpretation of this Agreement. In case of conflict or ambiguity among any provision included herein and any provision included in any Exhibit, the provision included herein shall prevail. Except if the context states otherwise, (i) references to sub-clauses, clauses and Exhibits are references to sub-clauses, clauses and Exhibits of this Agreement and the words "herein", "hereof", "hereby" and the like, whenever used in this Agreement, refer to this Agreement as a whole and not to any provision of this Agreement in particular; (ii) references to one gender include all genders and the use of singular includes the plural, and vice-versa; (iii) whenever the word "including", "inclusive" or "include" is used in this Agreement, it shall be deemed to be followed by the expression "but not limited to"; (iv) except if expressly set forth otherwise herein, all references to any Parties include their successors, beneficiaries, representatives and permitted assignees; (v) references to any contracts, agreements or other instruments include all changes, substitutions and consolidations and respective complements thereto, unless expressly provided otherwise, and (vi) if any action is to be taken by either Party, under the terms of this Agreement, on a date other than a Business Day, such action shall be performed on the first subsequent Business Day, and any and all terms set forth herein must be counted pursuant to article 132 of the Brazilian Civil Code.

#### Clause 2. Transaction

2.1. <u>Sale and Purchase</u>. Subject to verification (or waiver, as applicable) of the Closing Conditions Precedent and subject to the other terms and conditions in this Agreement, Sellers irrevocably and irreversibly undertake to sell and transfer, on the Closing Date and under the terms of the Articles 60, 66, 140, item II, 141, item II and 142, item II, of the LFR, all Shares to Buyer, free and clear of any liens, and Buyer, in turn, irrevocably and irreversibly undertakes to acquire the Shares from Sellers as set forth in this Agreement ("Transaction").

2.2. <u>Transfer of Shares</u>. The Shares will be transferred by Sellers to Buyer, on the Closing Date, free and clear of any liens, so that Buyer becomes directly or indirectly the sole holder of all the Shares, without prejudice to the provisions of Clause 5.2 of this Agreement.

# Clause 3. Purchase Price and Payment

# 3.1. Purchase Price.

3.1.1. The "<u>Preliminary Purchase Price</u>" will be equivalent to (i)  $[\bullet]$  reais (BRL  $[\bullet]$ )<sup>10</sup> ("<u>Base Price</u>"), (ii) <u>minus</u> the amount, if any, of the Pre-Closing Working Capital that is less than zero; (iii) <u>plus</u> the amount, if any, of the Pre-Closing Working Capital that is greater than zero; (iv) <u>minus</u> the amount, if any, of the Pre-Closing Debt that is greater than zero; (v) <u>plus</u> the amount, if any, of the Pre-Closing Debt that is greater than zero; (v) <u>plus</u> the amount, if any, of the Pre-Closing Cash that is greater than zero; (vi) <u>minus</u> the amount, if any, of the Pre-Closing Cash that is less than the Annual Revenue Price Disclosed; (vii) <u>plus</u> the amount, if any, of the Pre-Closing Annual Revenue Price that is greater than the Annual Revenue Price Disclosed.

3.1.2. The "<u>Final Purchase Price</u>" will be equivalent to (i) the Base Price, (ii) <u>minus</u> the amount, if any, of the Final Working Capital that is less than zero; (iii) <u>plus</u> the amount, if any, of the Final Working Capital that is greater than zero; (iv) <u>minus</u> the amount, if any, of the Final Debt that is greater than zero; (v) <u>plus</u> the amount, if any, of the Final Cash that is greater than zero; and (vi) <u>minus</u> the amount, if any, of the Final Annual Revenue Price Disclosed; (vii) <u>plus</u> the amount, if any, of the Final Annual Revenue Price that is greater than the Annual Revenue Price Disclosed.

3.2. <u>Pre-Closing Adjustment</u>. Within fifteen (15) days before the Closing Date, Sellers must prepare and submit to Buyer their calculations ("Pre-Closing Certificate") of their best estimate: (i) for the Company's Working Capital for the Verification Moment ("Pre-Closing Working Capital"); (ii) for the Company's effective Debt at the Verification Moment ("Pre-Closing Debt"); (iii) for the Company's Cash at the Verification Moment ("<u>Pre-Closing Cash</u>"); (iv) for the Pre-Closing Annual Revenue Price (together with a copy of the interim balance sheets - one of which must be audited by an Independent Auditor, as defined in the Pre-Closing Annual Revenue Price provided for in Clause 1.1 above - and supporting documentation); and (v) based on Pre-Closing Working Capital, Pre-Closing Debt, Pre-Closing Cash and Pre-Closing Annual Revenue, the calculation of the Preliminary Purchase Price, in all cases, made in accordance with Brazilian GAAP.

3.3. <u>Payment of the Preliminary Purchase Price</u>. Payment of the Preliminary Purchase Price must be made by Buyer to Sellers (always proportionally to their respective interest in the Company's share capital, to be indicated by Sellers in the Pre-Closing Certificate), net of any withholding at source or deduction of any Tax or banking fees, in Brazilian currency and in immediately available funds, by means of available wire transfer — TED to Sellers' bank accounts to be indicated by Sellers in the Pre-Closing Certificate, simultaneously to the transfer to Buyer of the title to said Shares, as follows: (i) an initial installment of the Preliminary Purchase Price ("<u>Cash Installment</u>") will be paid to Sellers on the Closing Date; and (ii) an installment corresponding to twenty percent (20%) of the Preliminary Purchase

<sup>&</sup>lt;sup>10</sup> <u>Note for the Draft:</u> Corresponds to the amount of the Annual Revenue Disclosed multiplied by the multiples offered in the winning bid of the Auction.

Price ("<u>Term Installment</u>" and, together with the Cash Installment, the "<u>Purchase Price</u> <u>Installments</u>") will be retained and will serve as a guarantee for payment of the Price Adjustment. The Term Installment will be retained and/or paid (in whole or in part) by Buyer to Sellers or reverted (in whole or in part) to Buyer under the terms of Clause 3.6.4.

3.4. <u>Discharge</u>. Proof of payment via TED in the amount of any of the Purchase Price or Price Adjustment Installments to the bank accounts owned by Sellers as indicated on the Pre-Closing Certificate or to the bank account owned by Buyer to be indicated by in writing (as the case may be) will serve, for all legal purposes and effects, as proof of discharge from the obligation to pay the Purchase Price or Price Adjustment Installments (as the case may be), as well as the fullest, complete, general, absolute, irrevocable and irreversible discharge granted by Sellers to Buyer or by Buyer to Sellers (as the case may be), so that the Parties can no longer complain, at any time and under any title, in relation to the payment of such amounts.

3.5. <u>Transaction Taxes</u>. Each Party will be fully liable for the payment of any and all Taxes that may be levied (i.e., the obligation of which is imposed on the Party in question under the terms of the Law) as a result of this Transaction.

3.6. <u>Price Adjustment</u>. Within seventy-five (75) days from the Closing Date, Buyer must prepare and send to Sellers its calculations ("<u>Closing Certificate</u>") for (i) the Annual Revenue with base date of calculation for the Closing Date ("<u>Final Annual Revenue</u>"), (ii) the Company's effective Working Capital at the Verification Time ("<u>Final Working Capital</u>"), (iii) the Company's effective Debt at the Verification Time ("<u>Final Debt</u>"), (iv) the Company's effective at the Verification Time ("<u>Final Cash</u>"), and (v) based on Final Annual Revenue, the Final Working Capital, Final Debt and Final Cash, the calculation of the Final Purchase Price and the Price Adjustment; in all cases, calculated in accordance with the Brazilian GAAP. If Buyer fails to deliver a Closing Certificate to Sellers within seventy-five (75) days from the Closing Date, the Price Adjustment will be equivalent to zero and Buyer shall pay the Term Installment to Sellers as provided for in Clause 3.6.4.

3.6.1. If Sellers disagree with the amounts included in the Closing Certificate, any of Sellers must submit to Buyer a notice to that effect within thirty (30) days from receipt of such Closing Certificate, in order to justify their disagreement in details by specifying each item that makes up the Final Annual Revenue, Final Working Capital, Final Debt, Final Cash, Final Purchase Price and Price Adjustment, and enclosing relevant supporting documentation ("Notice of Sellers Disagreement"). The absence of timely submission by any of Sellers of said Notice of Sellers Disagreement will be considered as tacit approval by Sellers of the amounts for the Final Annual Revenue, Final Working Capital, Final Debt, Final Cash, Final Purchase Price and Price Adjustment notified by Buyer on the Closing Certificate. In the event that Sellers send a Notice of Sellers Disagreements amicably within fifteen (15) days from receipt of such Notice of Sellers Disagreement ("Amicable Negotiation Period").

3.6.2. If, at the end of the Amicable Negotiation Period, the Parties cannot reach an agreement, then the Parties shall designate by mutual agreement, within fifteen (15) days, one of the Independent Auditors (excluding those who, on the contracting date, are the

independent auditor of either Party) to review only the controversial items of the Closing Certificate and determine the amount of Final Annual Revenue, Final Working Capital, Final Debt, Final Cash, Final Purchase Price and Price Adjustment. Upon the request of the Independent Auditor, the Parties shall provide all working papers and all other information and materials in their possession related to the items on which there is disagreement. The Parties will instruct the Independent Auditor so that the amount of the Final Annual Revenue, Final Working Capital, Final Debt, Final Cash, Final Purchase Price and Price Adjustment is not, under any circumstances, lower than the lowest amount or higher than the highest amount informed in the Closing Certificate or in the Notice of Sellers Disagreement. Within thirty (30) days after being hired, the Independent Auditor must send the Parties a written report with its conclusions on the controversial items and the amounts for the Final Annual Revenue, Final Working Capital, Final Debt, Final Cash, Price of Final Purchase and Price Adjustment, together with the calculation spreadsheets and corresponding justification ("Final Adjustment Report"). The Final Adjustment Report will be considered final and binding on the Parties, unless fraud or manifest errors occur, such as basic algebra errors, when either Party may report the error to the Independent Auditor in writing, and with a copy to the other Party, so that the Independent Auditor can correct such an error.

3.6.3. If the Price Adjustment determined by the Independent Auditor is closer to the amount informed in the Closing Certificate, the costs related to the hiring of the Independent Auditor shall be borne entirely and exclusively by Sellers. If the Price Adjustment determined by the Independent Auditor is closer to the amount informed in the Notice of Sellers Disagreement, the costs related to the hiring of the Independent Auditor shall be borne entirely and exclusively by Buyer.

3.6.4. If the Final Purchase Price, as indicated in the Closing Certificate, the Final Adjustment Report or the one agreed between the Parties during the Amicable Negotiation Period, as the case may be, is equivalent to an amount:

(i) <u>less</u> than the amount of the Cash Installment, then (a) Buyer will permanently retain the Term Installment for its own benefit, and nothing else will be due by Buyer to Sellers as payment for the Shares, and (b) Sellers will pay Buyer an amount corresponding to the difference between the amount of the Cash Installment and the Final Purchase Price adjusted by the DI Rate from the Closing Date until the date of the actual payment thereof;

(ii) <u>equal</u> to the amount of the Cash Installment, then Buyer will permanently retain the Cash Installment for its own benefit, and nothing else will be due by Buyer to Sellers as payment for the Shares;

(iii) <u>greater</u> than the amount of the Cash Installment, but <u>less</u> than the Preliminary Purchase Price (such an amount referred to as the "<u>Positive Balance</u>"), then (a) Buyer will pay to Sellers a portion of the Term Installment corresponding to the Term Installment less the Positive Balance (adjusted by the DI Rate from the Closing Date to the date of the actual payment thereof), and (b) Buyer will permanently retain the balance of such Term Installment; (iv) <u>greater</u> than the Preliminary Purchase Price, then Buyer will pay to Sellers the full amount of the Term Installment, as well as any amount that exceeds the Term Installment (adjusted, in both cases, by the DI Rate from the Closing Date until the date of the actual payment thereof).

3.7. <u>Payment of the Price Adjustment</u>. The Price Adjustment amount due by one Party to the other pursuant to Clause 3.6.4 shall be paid by Buyer or by Sellers, as the case may be, by the fifth (5<sup>th</sup>) Business Day counted from the final determination of the Price Adjustment amount under the terms stipulated above ("Adjustment Payment Date"), through TED to the (i) checking account held by Sellers indicated in the Pre-Closing Certificate, if Sellers are entitled to receive such amounts; or (ii) bank account held by Buyer to be indicated in writing in due course, if Buyer is entitled to receive such amounts.

3.7.1. If the Price Adjustment is to be paid by Buyer to Sellers, and Buyer fails to fully pay the amounts due within the period indicated above, Sellers may, at any time and at their sole discretion, withhold the payment flow to which Company is entitled based on the Oi Sharing Agreement and offset the defaulted amounts, by sending a simple notice to Buyer and Company, accordingly.

3.7.2. Failure to pay the Price Adjustment within the periods indicated in Clause 3.7 above will subject the defaulting Party to bear the payment of late-payment interest of one percent (1%) per month on the outstanding balance, calculated *pro rata temporis*, without prejudice to the provisions in article 389 of the Brazilian Civil Code.

# Clause 4. Conditions Precedent for the Closing

4.1. <u>Parties Conditions Precedent</u>. Without prejudice to the provisions of Clauses 4.2 and 4.3 below, as applicable, the obligation of the Parties to consummate the Closing, as detailed in Clause 5.2, is subject to the satisfaction of the following conditions precedent, up to or on the Closing Date ("<u>Parties Conditions Precedent</u>"):

4.1.1. List of Parties Conditions Precedent:

(i) all requirements and formalities set forth in the LFR and in the Amendment to the PRJ that are required for the Closing and to consummate the Transaction must have been fulfilled;

(ii) absence of any Law prohibiting, suspending, altering or limiting, in any way, the consummation of the operations contemplated by this Agreement and/or its Exhibits;

(iii) [the Parties must have obtained CADE Approval, pursuant to Clause 4.4 below]<sup>11</sup>;

(iv) the legal period for the filing of any appeals must have elapsed or, if any appeals have been filed, a judicial decision granting a suspensive effect to such appeals filed against (a)

<sup>&</sup>lt;sup>11</sup> <u>Note for the Draft</u>: Provisions related to antitrust approvals to be excluded if not applicable at the time of executing this Agreement.

the judicial decision for ratification of the Amendment to the PRJ shall not be in force; and/or (b) the judicial decision for ratification of the winning bid of the Auction, under the terms of the Amendment to the PRJ and the LFR;

(v) no judicial decision shall be in force that prohibits, limits or determines the postponement of the consummation of the operations contemplated by this Agreement and/or its Exhibits.

4.2. <u>Buyer Conditions Precedent</u>. Without prejudice to the provisions of Clauses 4.1 and 4.3, as applicable, Buyer's obligation to consummate the Closing, as detailed in Clause 5.2, is subject to the satisfaction (or waiver, at Buyer's sole discretion) of each of the following conditions precedent, up to or on the Closing Date ("<u>Conditions Precedent for Buyer</u>"):

4.2.1. List of Buyer Conditions Precedent:

(i) all representations and warranties made by Buyer in this Agreement must remain true, complete and exact, both on the date of execution of this Agreement and on the Closing Date;

(ii) on the Closing Date, also, Sellers and Consenting Intervening Sellers shall not be in breach of the obligations undertaken by them under this Agreement, including those stipulated in Clause 8.1 hereof;

(iii) any event that constitutes a Material Adverse Effect has not occurred from the date of execution of this Agreement to the Closing Date;

(iv) the estimated Working Capital and Debts amounts shall each be in the range between BRL 0.00 and twenty five million reais (BRL 25,000,000.00) at the Verification Time;

(v) Sellers and/or their Affiliates have not made any new changes to the Judicial Reorganization Plan or Amendment to the PRJ with respect to the Company and its sale as UPI, Business or Infrastructure Items;

(vi) the contracts, agreements and corporate acts formalizing the Corporate Reorganization must have been registered with the Federal Revenue Service of Brazil and duly filed with the applicable commercial registries;

(vii) Sellers and Company must have sent a notice to the Governmental Authorities that issue or grant the Site Licenses indicated in the "Licensing" columns of <u>Exhibit B;</u>

(viii) Sellers and Company must have obtained consents from the counterparties to the Existing Sharing Agreements for which it is eventually required, under the terms of the agreements in question, the prior consent from the respective counterparties for the implementation of the Corporate Reorganization, the Closing and the assignment and transfer of the Existing Sharing Agreements in question for the Company, as listed in Exhibit A; and

(ix) Sellers and Company must have (a) carried out the Full Regularization of as many Lease Agreements Identified, as listed in Exhibit 4.2.1(ix), as necessary to achieve the Minimum Revenue index or more; and (b) submitted supporting documentation to Buyer.

4.2.2. The Parties acknowledge that Buyer Conditions Precedent have been stipulated for Buyer's sole and exclusive benefit. Thus, the Parties agree that Buyer may, at its sole discretion, waive compliance with said Buyer Conditions Precedent that may not have been satisfied by the Closing Date, by sending written notice to the other Parties. If Buyer waives any of Buyer Conditions Precedent and provided that the other Closing Conditions Precedent have been implemented, Buyer, Company and Sellers will be required to proceed with the Closing and implement the obligations contained in this Agreement.

4.2.3. Sellers shall notify Buyer of the fulfillment of Buyer Conditions Precedent within two (2) Business Days from the date of fulfillment thereof.

4.3. <u>Sellers Conditions Precedent</u>. Without prejudice to the provisions of Clause 4.1 and 4.2, as applicable, the obligation of Sellers to consummate the Closing, as described in detail in Clause 5.2, is subject to the satisfaction (or waiver, at the sole discretion of Sellers) of each of the following conditions precedent, up to or on the Closing Date ("<u>Sellers Conditions Precedent</u>" and, together with the Parties Conditions Precedent and Buyer Conditions Precedent, the "<u>Closing Conditions Precedent</u>"):

4.3.1. List of Sellers Conditions Precedent:

(i) all representations and warranties provided by Buyer in this Agreement contained in Clause 7 must remain true, complete and exact, both on the date of execution of this Agreement and on the Closing Date; and

(ii) on the Closing Date, also, Buyer must have complied with the obligations undertaken by it in this Agreement.

4.3.2. The Parties acknowledge that Sellers Conditions Precedent have been stipulated for Sellers' sole and exclusive benefit. Thus, the Parties agree that Sellers may, at their sole discretion, waive compliance with said Sellers Conditions Precedent that may not have been satisfied by the Closing Date, by sending written notice to the other Parties. If Sellers waive any of Sellers Conditions Precedent and provided that the other Closing Conditions Precedent have been implemented, Buyer, Company and Sellers will be required to proceed with the Closing and implement the obligations contained in this Agreement.

4.3.3. Buyer shall notify Sellers of the fulfillment of Sellers Conditions Precedent within two (2) Business Days from the date of fulfillment thereof.

4.4. <u>[Notice to CADE</u>. The Parties shall mutually cooperate and use their best efforts for the Transaction to be approved by CADE. The Parties agree that Buyer shall actively and diligently lead the preparation, submission and follow-up on the respective request for approval, and Sellers and Company, to the extent applicable thereto, shall fully cooperate with Buyer, its lawyers and consultants, regarding the information and data to be submitted

to CADE by supplying, in a manner that is timely and compatible with the fulfillment of the obligations agreed herein, all information and data that must be submitted to CADE, pursuant to Resolution No. 2/2012 of CADE, or that the Parties, by mutual agreement, consider useful to obtain CADE Approval, also subject to the following:

4.4.1 The costs related to the Transaction notification fee to CADE shall be fully borne by Buyer and each Party shall bear the costs related to its own lawyers for monitoring the analysis process with CADE on its behalf.

4.4.2. Any penalty that may be imposed in such a procedure shall be paid by the Party that caused that penalty.

4.4.3. Upon (i) collaboration of Sellers and Company in providing the necessary documents and information, as well as (ii) review and approval by Sellers of the applicable documentation, Buyer undertakes to use its best efforts to present the first draft of the formal request to obtain CADE Approval within ten (10) Business Days from the date of execution of this Agreement.

4.4.4. Each Party shall inform the other Party of any and all communications received from CADE in the context of obtaining CADE Approval, within 48 hours of receipt thereof.

4.4.5. (i) To the extent required or permitted by CADE, both Parties will participate in any meeting or contact with CADE; (ii) the Parties shall cooperate with each other and align in advance with respect to any meetings or contacts with CADE, for the purposes of item "(i)" of this Clause 4.4.5; (iii) the Parties must send to the other Party a copy of any and all documents received or sent to CADE in relation to the legal transactions covered by this Agreement, by omitting Sensitive Competition-Related Information of the sending Party; and (iv) any written statements before CADE in relation to the legal transactions covered by this Agreement shall be submitted jointly, subject to review and approval by both Parties.

4.4.6. If an objection to the Transaction is filed by CADE's General Superintendence, which suggests its disapproval or approval with restrictions by the CADE Court, then the Parties, led by Buyer, must file their answer to such objection within the legal term, and must also negotiate in good faith and diligently, aiming to identify the minimum level of restrictions required up to the limit provided for in item "(i)" below, and propose sufficient restrictions to consensually eliminate the concerns identified in the approval process. The implications of the restrictions required by CADE or, even negotiated with CADE, will depend on the impact that any assets to be divested will have on the consolidated turnover of Buyer's group, subject to the following parameters:

(i) in the event that the restrictions imposed by CADE do not qualify as Substantial Competitive Restrictions, Buyer will be required to comply with the restrictions imposed for obtaining CADE Approval and proceed with the Closing of the Transaction, provided that the other Closing Conditions Precedent provided for in this Clause 4 are verified and/or waived, as applicable. In this case, if the restrictions fall totally or partially on Infrastructure Items, the Final Purchase Price shall be adjusted in accordance with Exhibit 4.4.6(i); and

(ii) in the event of the imposition of Substantial Competitive Restrictions, Buyer may terminate this Agreement unilaterally, without incurring any fine, provided that Buyer will not be required to litigate before CADE so that such Substantial Competitive Restrictions are reconsidered.]<sup>12</sup>

4.4.7. Notwithstanding anything to the contrary contained in this Agreement, under no circumstances will Buyer be required to litigate against any Governmental Authority (including CADE) or sell or otherwise dispose of its assets in order to obtain CADE Approval.

4.5. <u>Deadline</u>. The Parties shall use their best efforts to ensure that the Closing Conditions Precedent are met so that Closing occurs within one hundred and eighty (180) days from the date hereof ("Closing Deadline"), subject to, where applicable, the provisions of Clause 11.2.

# Clause 5. Closing

5.1. <u>Closing</u>. The Closing will occur on the twelfth (12<sup>th</sup>) Business Day counted from the receipt, by either Party, of a written notice sent by the other Party, informing about, as the case may be, (i) the fulfillment of all Closing Conditions Precedent (except for those Closing Conditions Precedent to be fulfilled on the Closing Date) and forwarding copies of the necessary instruments to prove compliance with the Closing Conditions Precedent; or (ii) the waiver of the Closing Conditions Precedent not met until then, as applicable; or any other date that may be agreed between Buyer and Sellers (the date on which the Closing actually occurs, hereinafter referred to as "Closing Date").

5.1.1. In any event, Closing shall occur on the Closing Date, at 10:00 am, at Rua Humberto de Campos, n° 425, in the City and State of Rio de Janeiro, except if other place and time are mutually agreed in writing by the Parties.

5.2. <u>Obligations at Closing</u>. On the Closing Date, the Parties undertake to perform the following acts, each act (i) being conditioned to the occurrence of each of the other acts, and (ii) will be considered to have occurred simultaneously to each of the other acts, regardless of the order in which they are presented in this Clause 5.2, so that none of the acts described herein will be considered performed until all other acts have been performed:

5.2.1. List of obligations at Closing:

(i) Sellers will deliver to Buyer waiver instruments signed by all the Company's managers, including the granting of the broadest, most general, absolute and unrestricted discharge to the Company, in relation to any rights, compensations and/or other payments to which they could be entitled, due to their conditions as managers of the Company;

<sup>&</sup>lt;sup>12</sup> <u>Note for the Draft</u>: Provisions related to antitrust approvals to be excluded if not applicable at the time of executing this Agreement.

(ii) Sellers and Buyer shall execute the transfer instruments recording the transfer to Buyer of ownership over the Shares in the Company's Registered Share Transfer Register and the Company's managers shall record said transfers in the Company's Registered Share Register, free and clear of any Liens;

(iii) Sellers must present the instruments of revocation of all powers-of-attorney granted by Company and/or that in any way are related to the Infrastructure Items, including those identified in Exhibit 6.1.19, accompanied, whenever possible, by a copy of the grantee's acknowledgment instruments;

(iv) Buyer must make the full payment of the Cash Installment on behalf of Sellers, pursuant to Clause 3;

(v) the Parties will enter into and deliver to each other the Transaction closing instrument, in the form of Exhibit 5.2.1(v);

(vi) Sellers and/or their Affiliates will enter into the Oi Sharing Agreement, substantially in the form of Exhibit 5.2.1(vi);

(vii) Sellers and/or their Affiliates and Company will terminate any infrastructure sharing and/or assignment agreements that may have been entered into and in force on the Closing Date, and grant each other the broadest, general, absolute and unrestricted discharge with respect to the obligations arising from such agreements;

(viii) Sellers and/or their Affiliates will: (a) enter into new Lease Agreements with Company related to the properties listed in <u>Exhibit 5.2.1(vii)(A)</u>, which are owned or held by, and are leased by Company to the Affiliates of Sellers ("Sellers Affiliates Real Properties"), according to the form and commercial conditions provided for in <u>Exhibit 5.2.1(viii)(B)</u>; and (b) terminate the Lease Agreements for Sellers Affiliates Properties infrastructure sharing and/or assignment agreements that may have been entered into and in force on the Closing Date, and grant each other the broadest, general, absolute and unrestricted discharge with respect to the obligations arising from such agreements; and

(ix) Buyer will hold a Special Shareholders' Meeting of the Company ("<u>Closing AGE</u>"), in order to (a) recognize the resignation of the current managers of Company with discharge for the respective management acts, except for any acts performed with intent or fraud, with the granting of a temporary power-of-attorney to the new managers of the Company to be appointed by Buyer in accordance with the draft set out in <u>Exhibit 5.2.1(ix)</u>; and (b) approve the nomination and appointment of the new managers, to be nominated and appointed by Buyer.

5.2.2. Within five (5) Business Days from the Closing Date, Buyer will file the Closing AGE minutes with the competent Commercial Registry, as well as provide registration updates to the Federal Revenue Service of Brazil. In case the Commercial Registry or any Governmental Authority presents any requirements for said filing and/or update, the Parties shall practice any and all acts required for immediate fulfillment of such requirements. Within five (5) Business Days counted from the date of actual filing of the Closing AGE minutes before the

competent Commercial Registry, Buyer shall provide Sellers a certified copy of the respective filing certificate. Within five (5) Business Days counted from the date of completion of the updates referred to above, Buyer shall provide Sellers a certified copy of the pertinent documents proving the completion of said registration updates.

5.2.3. Buyer hereby acknowledges that Company's managers may, as of the Closing Date, freely resign their respective positions and cause their respective resignation instruments to be annotated with the competent Commercial Registry, at any time, with the managers or Sellers not being liable for any loss that Company and/or Buyer may suffer due to the lack of appointment of new members to the Company's management as of the Closing Date.

5.3. <u>Delivery of Documents</u>. Within fifteen (15) days from the Closing Date, Sellers must deliver to Buyer, in physical or digital media, the following documents: (a) the permits, licenses, documents and authorizations that are necessary for the operation of the Infrastructure Items that exist and are in the possession of Sellers; (b) the Station Operating Permit issued by the National Telecommunications Agency - ANATEL - and the Radiometric Report; (c) the reports and other technical documentation that exist and are in the possession of Sellers related to the Infrastructure Items, including, but not limited to, projects, sketches, plans, designs, structural and foundation reports, Technical Responsibility Notes, etc., including those that exist in digital channels; and (d) all Relevant Company Agreements.

# Clause 6. Representations and Warranties of Sellers

6.1. <u>Sellers Representations and Warranties</u>. Sellers hereby, jointly and severally, provide Buyer with the representations and warranties below, which Sellers state to be true, accurate, legitimate, complete, exact and correct on the date hereof and which shall remain so until the Closing Date (except for those representations and warranties with reference to a specific date, which are true, accurate, legitimate, complete, exact and correct on the date to which they refer):

6.1.1. <u>Corporate Good Standing</u>. Sellers and Company are regularly incorporated, validly existing legal entities in good standing pursuant to the Brazilian Laws, and were duly registered before all competent agencies, Commercial Registries and Governmental Authorities, except for those whose absence will not or may not put this Transaction at risk. Sellers and Intervening Consenting Sellers have the corporate authority, power and jurisdiction to own, lease and/or otherwise hold their respective assets and property, as well as to conduct and operate their business as currently conducted.

6.1.2. <u>Accounting books and corporate records</u>. Company's accounting books, records and documents are in order and in compliance with the applicable legal standards, and have been kept, since their constitution, under the applicable legal and accounting standards, and substantially reflect, under all relevant aspects, the events and transactions that must be recorded therein, always in accordance with the Brazilian GAAP.

6.1.3. <u>Ownership</u>. Sellers are the legitimate owners of the Shares and have legal, good, fair and commercial title and ownership over them, being the sole and exclusive owners and beneficiaries of the Shares, which are free and clear of any and all Liens. No Contractual Instrument (except for this Agreement), commitment or obligation was entered into or undertaken by Sellers with any Person (including Third Parties and/or any Related Party), for the sale, assignment, donation, encumbrance, creation of Lien, put or call option, transfer and/or disposition, directly or indirectly, of the Business and/or any of the Shares, in any form. Sellers have the right and the powers required by the applicable Law to freely dispose of the Shares and Business, as provided herein. There is no Demand (or, to the knowledge of Sellers, threatened Demand) involving or whose object is the Shares and/or the Business and/or which challenges the ownership of the Shares.

(i) there is no shareholders' agreement or similar instrument binding the Shares, filed or not at the Company's headquarters; and

(ii) upon acquisition of the Shares by Buyer, Buyer will have a valid and effective title over all the Shares and over the Company, which will not be subject to any Liens, encumbrances and successions of Sellers and any of their Affiliates or Related Parties, pursuant to articles 60, 66, 140, item II, 141 and 142, item II, of the LFR. Company will be, after the Corporate Reorganization, the sole holder and owner of the Infrastructure Items, which are free and clear of any Liens.

# 6.1.4. Power, Corporate Jurisdiction and Approvals.

(i) Sellers and Intervening Consenting Sellers have full power, authority and corporate jurisdiction to enter into this Agreement and any other documents to be signed by them as a result of and/or in relation to the Transaction, as well as to fulfill their respective obligations and responsibilities undertaken in this Agreement, in the Exhibits hereto and any other documents to be signed by them as a result of and/or in relation to the Transaction. The competent corporate bodies of Sellers and Intervening Consenting Sellers approved, as applicable, the execution of this Agreement, the consummation of the Transaction and performance of all obligations undertaken herein;

(ii) except for CADE Approval, if necessary, the execution of this Agreement and any other documents resulting from the Transaction and the consummation of the Transaction does not: (a) depend on the approval or authorization of any Governmental Authority and/or Third Party; (b) violate any Law; and/or (c) results in the creation of a Lien on the Shares and/or on the Infrastructure Items and/or the Leased Areas, neither it constitutes a breach of any obligations of Company and/or Sellers, nor will it result in the loss or termination of any right by Company; and

(iii) this Agreement constitutes a legal, valid and binding obligation on Sellers and Intervening Consenting Sellers, pursuant thereto. There is no Demand (or, to the knowledge of Sellers, threatened Demand), which, if decided negatively, prevents or impair the ability of Sellers and/or and Intervening Consenting Sellers to comply with their obligations under this Agreement and/or proceed with the Closing.

6.1.3. <u>Sellers' Bylaws and Judicial Reorganization Plan</u>. The execution and formalization of this Agreement by Sellers and by Intervening Consenting Consenting Sellers, as well as compliance by Sellers and by Intervening Consenting Sellers with their obligations undertaken herein shall not conflict or result in non-compliance with the respective bylaws and/or the Judicial Reorganization Plan of Sellers and/or Intervening Consenting Sellers, as amended from time to time (including any Amendment to the PRJ), or any corporate document and/or any applicable Law or regulation. Buyer will not succeed any obligations or debts of Sellers or any of their Affiliates or Related Parties, pursuant to articles 60, 66, 140, item II, 141 and 142, item II, of the LFR.

6.1.6. <u>Outstanding Securities and Issuance of Shares</u>. Except for the Shares, there are no (i) convertible and/or negotiable securities and/or any other shares representing Company's share capital issued by Company; (ii) guarantees, Liens, options and/or rights related to the Shares; and/or (iii) any interest of Third Parties and/or any Related Party in the Company. All Shares are validly issued and fully paid in by Sellers. Company does not hold any direct or indirect equity interest in any Person, nor does it undertake to constitute or participate in the share capital of any Person or to subscribe or acquire any securities convertible into shares of any Person. Company has no subsidiaries and has never maintained any subsidiary throughout its existence.

6.1.7. <u>Relations with Third Parties</u>. The execution and formalization of this Agreement and compliance by Sellers and by Intervening Consenting Sellers with their obligations arising herefrom shall not conflict or result in violation of any agreement, covenant and/or Contractual Instrument to which Sellers or Intervening Consenting Sellers are subject or bound to, except for the cases in which the prior consent of Third Parties is required for the assignment of Existing Sharing Agreements and Lease Agreements.

6.1.8. <u>Tax Matters</u>. Oi Companies Corporate Reorganization (exclusively in relation to Business) and Company have complied and comply in a timely manner, to the extent they become due and/or, as applicable, have timely and in good faith opposed to, by the appropriate means, their obligations related to any Taxes that they are required to pay by virtue of the Law and/or Contractual Instrument and, in case of any delays, the related fines, interest and accretions were collected, as set forth in the Law, except to the extent such non-compliance, lack of opposition or lack of collection, individually or in aggregate, may cause a Material Adverse Effect. Oi Companies Corporate Reorganization (exclusively in relation to Business) and Company have not received any notice from any Governmental Authority stating their intention to carry out a tax due diligence in relation to the Company and/or its Business.

6.1.9. <u>Infrastructure Items</u>. Exhibit B contains a correct, true and complete list of all Infrastructure Items and an indication of their respective Leased Areas, including: (i) address of each of the Infrastructure Items; and (ii) the respective properties in which such Infrastructure Items are located, as well as the Lease Agreements related to each of the Infrastructure Items. Sellers further represent that they acknowledge that Buyer's decision to enter into this Agreement was made considering the Infrastructure Items specifically described in Exhibit B, which may not be subject to any increase or decrease in the number of said Infrastructure Items and/or any material changes, modifications or exclusions.

Company is the lawful holder and owner of the Infrastructure Items, which are free and clear of any and all Liens, and are in lawful possession of the Leased Areas. The Infrastructure Items listed in Exhibit B are all the Infrastructure Items necessary for Company to be able to conduct its Business, in the way that such business had been conducted. Except as specifically indicated in Exhibit B to this Agreement individually for each Infrastructure Item (together, the "Site Irregularities"), the Infrastructure Items and their respective Lease Agreements do not have, on the date hereof, any type of irregularity, including (a) documentary, technical or legal, (b) structural or engineering (c) referring to the possession of the Leased Areas, to licensing; (d) that requires some type of structural reinforcement or engineering work to meet the contracted capacities in their agreements. Except as specifically indicated in Exhibit B to this Agreement individually for each Infrastructure Item, (y) the Infrastructure Items were and are being operated in accordance and in compliance with all Laws, Agreements and Licenses for the Site; and (z) the Infrastructure Items have the applicable Site Licenses and their operation was and is being carried out in accordance and in compliance with such Site Licenses. Except for the Site Irregularities, the Infrastructure Items are in operating condition (except for wear and tear due to normal use).

(i) The Site Licenses informed in the "Licensing" columns of <u>Exhibit B</u> have all the supporting documentation necessary to prove the existence of said Site Licenses.

(ii) <u>Exhibit B</u> shows, in relation to each Infrastructure Item, the (i) current maximum capacity, (ii) current installed capacity, and (iii) contracted capacity to be installed.

(iii) <u>Exhibit A</u> indicates a complete and correct list of all Existing Sharing Agreements.

(iv) Company holds all necessary rights to access the Infrastructure Items and there is no threatened or existing Demand that would have the ability to limit or extinguish such rights.

(v) The Infrastructure Items are not reversible assets or related to concession, being certain that once the concessions related to the fixed telephony service granted by the competent Governmental Authorities to Sellers or their Affiliates, the Infrastructure Items (as well as all rights and privileges related thereto) will not return to the granting power.

6.1.10. <u>Solvency</u>. Sellers and Intervening Consenting Sellers are solvent under the applicable Law and able to pay their debts as they become due. Except for the Judicial Reorganization Plan, there is no procedure pertaining to any transaction or settlement with creditors, or any procedure of bankruptcy, judicial or extrajudicial reorganization, much less other insolvency procedure involving or threatened against Sellers and Intervening Consenting Sellers and/or Business and there has been no event that, under the terms of the applicable law, justified the initiation of such procedures. The fulfillment of the obligations provided for or arising from this Agreement by Sellers and by Intervening Consenting Sellers shall not constitute fraud to creditors and/or fraud to the execution under the terms of the applicable Law.

6.1.11. <u>No Violations</u>. Company and Oi Companies Corporate Reorganization (exclusively with respect to the Business):

(i) are not in default, nor have any event occurred that, including due to notice or expiration, or both, may be characterized as evidenced default and/or otherwise may limit the capacity of Company and/or Oi Companies Corporate Reorganization (exclusively with respect to the Business) to exercise its rights and/or enforce compliance with obligations owed thereto, in the form and conditions existing until the date of execution hereof, subject to the cases where prior consent of Third Parties is needed to assign for the Corporate Reorganization, the Closing e for the assignment and transfer of the Existing Sharing Agreements, with the exception of the Site Irregularities or except to the extent that it is not reasonably expected that the any defaults on the part of Company and/or Oi Companies Corporate Reorganization (exclusively with respect to the Business) may cause a Material Adverse Effect;

(ii) with the exception of the Site Irregularities, they did not violate in any material aspect any applicable Law and/or Contractual Instrument to which Company and/or Oi Companies Corporate Reorganization (exclusively in relation to the Business) or their respective assets are subject; and

(iii) except for the Site Irregularities, Oi Companies Corporate Reorganization (exclusively in relation to the Businesses) have and Company will have, on the Closing Date, all the licenses, permits and approvals necessary for the conduct of the Business, and were not prevented from obtaining and/or renewing any License (including Site Licenses) necessary to secure ownership of their assets or the development of their respective activities, except to the extent that it is not reasonably expected that any violation and/or restriction on the renewal of such Licenses may cause a Material Adverse Effect.

6.1.12. <u>Financial Statements</u>. <u>Exhibit 6.1.12</u> contains copies of the complete set of (i) [unaudited] financial statements of Company for the period ended on [•]; and (ii) financial statements of Company for the period ended on [•] ("<u>Financial Statements</u>"). The Financial Statements are true, complete and correct, they were prepared in accordance with the applicable law and with the Brazilian GAAP, and were prepared based on the Company's accounting books and they adequately demonstrate (i) the Company's financial position; (ii) the Company's assets and liabilities; (iii) the Company's business and financial rights and obligations; and (iv) the Company's profits and losses.

(i) The Company's accounts receivable, as provided for in the Financial Statements, constitute all accounts receivable of the Company, derive from the regular provision of services, within the Ordinary Course, and they are valid, binding and enforceable against their respective debtors, pursuant thereto.

(ii) Company is the lawful holder, proprietor and owner, or, as the case may be, lessee, tenant or grantee of all assets, properties and rights contained in the Financial Statements, including the Infrastructure Items and Leased Areas ("<u>Company Assets</u>"). Company Assets were accounted for in accordance with the Brazilian GAAP. Oi Companies Corporate Reorganization, exclusively in relation to the Business, were, up to the date of the conclusion of the Corporate Reorganization, the lawful holders, proprietors and owners, or, as the case may be, tenants or lessees of all assets, rights and properties related to the Business, including the Infrastructure Items and Leased Areas, which were transferred to Company as a result of the Corporate Reorganization.

(iii) <u>No Undisclosed Liabilities</u>. There is no debt, liability and/or obligation, effective or contingent, of Company and/or in relation to the Business, which is not reflected and duly disclosed in the Financial Statements, in accordance with the Brazilian GAAP and IFRS 16.

(iv) <u>No Substantial Changes</u>. Since [•]<sup>13</sup>, Company (or, exclusively in relation to Company Shares, Assets and/or Business, any of Sellers) have not: (a) experienced any event or circumstance that would be reasonably expected to cause a Material Adverse Effect; (b) contracted capital investment outside the Ordinary Course of business in an amount that exceeds [•] reais (BRL [•]); (c) had any substantial cancellation or withdrawal in any Demand, litigation or right over amounts, or any sale, transfer, assignment, constitution of Lien, distribution or other disposals of any Asset in an amount that exceeds [•] reais (BRL (e); (d) disposed of any right in relation to the use of any intellectual property right; (e) performed any material change in any accounting policy or maintenance of accounting books or practices; (f) undertaken, except in the Ordinary Course of its business, any obligation or liability, including, but not limited to, any liability for breach or termination of any Contractual Instrument in an amount that exceeds [•] reais (BRL [•]); (g) made any sale, encumbrance, transfer or other disposal of any Company Assets; (h) performed any remission, pardon or any type of unilateral extinction of any rights or credits held by Company against its members, employees, service providers and/or any other Persons; (i) performed any free donation, assignment and/or transfer of any assets, rights and/or any other kind of Company Asset to any of its members, employees, service providers and/or any Persons; (j) hired any manager, worker and/or employee; (k) made a substantial change to the Company's corporate documents; (1) issued, delivered or sold any shares or securities issued by Company to any Person (including the Shares); (m) redeemed or repurchased shares issued by Company; (n) made an increase or decrease in the Company's capital or reclassification of its shares; (o) amended, modified or terminated any Contractual Instrument; and/or (p) entered into any agreement, commitment or Contractual Instrument undertaking to implement or carry out any of the business or acts described in items "(a)" through "(p)" above.

6.1.13. <u>Assets</u>. The Company is the owner or holder of rights to use all Company Assets necessary to conduct the Business, as such Business has been conducted by Company or had been conducted by Oi Companies Corporate Reorganization prior to the conclusion of the Corporate Reorganization. Except for the Site Irregularities, (i) Sellers have no Knowledge of a fact or circumstance that can reasonably be expected to interfere with the right of ownership or use by Company of Company Assets; and (ii) Company Assets are in operating condition for use (except for wear and tear from normal use) and there are no latent defects related to such Company Assets except to the extent that it is not reasonably expected that such defects could cause a Material Adverse Effect. None of Company Assets used or employed in conducting the Business has been assigned, pledged, encumbered, recorded, sold, donated or, through any other business or legal act, assigned, transferred or promised to another Person (and Company, Oi Companies Corporate Reorganization and/or Sellers have not entered into any agreement or Contractual Instrument in this regard).

<sup>&</sup>lt;sup>13</sup> <u>Note for the Draft:</u> This date must be the first among the dates of the Financial Statements.

(i) Company Assets constitute all the assets necessary to conduct the Business, as such business has been conducted by Company and had been conducted by Oi Companies Corporate Reorganization prior to the conclusion of the Corporate Reorganization and such Company Assets were effectively transferred to Company at the time of the Corporate Reorganization.

6.1.14. Real Property. Exhibit B, in its "Geolocation", "Address" and "Lease Agreements" columns, contains complete and correct lists of all Leased Areas and all Lease Agreements. The Leased Areas where the Infrastructure Items are located (a) are under Company's tame and peaceful possession; and (b) the occupation of such Leased Areas is exercised by Company (b.1) in accordance with the applicable Law; and (b.2) except as indicated in the "Lease Agreements" column of Exhibit B, in accordance with the respective Lease Agreements; and (c) they are regularly leased, subleased or assigned by the respective owners of such real properties, as indicated in the "Lease Agreements" column of Exhibit B; except, in each case, insofar as it is not reasonably expected that any disputes involving such properties may cause a Material Adverse Effect. All Lease Agreements represent a valid and effective obligation, exercisable in accordance with their terms. The Lease Agreements are sufficient for Company to conduct the Business, as it has been conducted. Company is not the owner and has never owned any real property. None of the Infrastructure Items goes beyond their respective Leased Areas according to their Lease Agreements, and none of the Leased Areas is located in an environmentally protected area (including, but not limited to, UCs and/or APPs) and/or listed (or, when located in such areas, the Licenses applicable for such installation and occupation have been obtained). Sellers Affiliates Real Properties are the lawful property of Sellers' Affiliates (or such Sellers' Affiliates have a tame, peaceful and undisputed possession of such real properties), being able to freely lease them to Company as from the Closing Date, according to Clause 5.2.1(viii) above.

6.1.15. <u>Environmental</u>. Except for the Site Irregularities, Company complies with all applicable Environmental Laws, except to the extent that such non-compliance does not generate a Material Adverse Effect. No notice of breach, notice of liability or request for information has been received with respect to the Infrastructure Items and no litigation is pending or, to the knowledge of Company and/or Sellers, threatened by any person with respect to the Infrastructure Items and no penalty or fine has been applied and remains due against Company (or related to the Infrastructure Items) with respect to any Environmental Law that may give rise to a Material Adverse Effect.

6.1.16. <u>Agreements</u>. Except as provided for in <u>Exhibit 6.1.16</u>, there is no other Material Contract entered into, assigned or that will be assigned to Company as a result of the Corporate Reorganization and/or involving, directly or indirectly, the Business or Company Assets (together, "Relevant Company Agreements"). Except for the Relevant Company Agreements that require prior consent/authorization, there is no fact that: (i) may cause the termination, even partially, or the early maturity of any Relevant Company Agreement or other Contractual Instruments entered into by Company (including the Existing Sharing Agreements and Lease Agreements); (ii) may compromise the receipt, by Company and/or by Oi Companies Corporate Reorganization (prior to the completion of the Corporate Reorganization), of any outstanding amounts under the terms of any Relevant Company

Agreements or other Contractual Instruments entered into by Company and/or by Oi Companies Corporate Reorganization (exclusively with respect to the Business) (including the Existing Sharing Agreements and Lease Agreements); and/or (iii) may prevent or impair the performance and/or the renewal of any Relevant Company Agreement or other Contractual Instruments entered into by Company and/or by Oi Companies Corporate Reorganization (exclusively with respect to the Business) (including the Existing Sharing Agreements and Lease Agreements); and/or by Oi Companies Corporate Reorganization (exclusively with respect to the Business) (including the Existing Sharing Agreements and Lease Agreements), except to the extent that it is not reasonably expected that the possible existence of any facts that could result in any of the consequences indicated above will have a Material Adverse Effect. No Relevant Company Agreement or Contractual Instrument entered into by Seller or by Dommo generates or has the potential to generate Liens on the Infrastructure Items.

(i) The Relevant Company Agreements and other Contractual Instruments entered into by Company (including the Existing Sharing Agreements and Lease Agreements) constitute a valid and binding legal obligation of its signatories, enforceable thereunder;

(ii) The Relevant Company Agreements are all the Contractual Instruments essential for the conduction of the Business by Company, in the way that such Business had been conducted by Oi Companies Corporate Reorganization prior to the Corporate Reorganization; Except to the extent that it is not reasonably expected that any default may cause a Material Adverse Effect, Company is in compliance with all Relevant Company Agreements and other Contractual Instruments entered into by Company and, to the Knowledge of Sellers, the counterparties to such Relevant Company Agreements and Contractual Instruments are also in compliance with all relevant aspects of such Relevant Company Agreements and Contractual Instruments, and neither Company, Oi Companies Corporate Reorganization, nor Sellers, sent and/or received any notice of termination or default of such Relevant Company Agreements or Contractual Instruments.

(iii) No Relevant Company Agreement or Contractual Instrument entered into by Company (including the Existing Sharing Agreements and Lease Agreements) establishes any obligation for Company to carry out CAPEX/OPEX investments and/or expansions in an amount greater than five hundred thousand reais (BRL 500,000.00) and/or the counterparty's right to participate in Company's revenues and/or other receivables under such Relevant Company Agreements or Contractual Instruments.

(iv) Company and/or Oi Companies Corporate Reorganization (exclusively with respect to the Business) are not part of the Relevant Company Agreements or Contractual Instruments containing an exclusivity clause, non-competition provisions or that limit the performance and expansion of Company and/or contain a limitation or prohibition on the disposal of assets, including Company Shares and Assets.

6.1.17. <u>Transactions with Related Parties</u>. Except for the Lease Agreements of Sellers Affiliates Real Properties, there are no contracts, agreements, guarantees provided and/or understandings between, on one side, Company and, on the other side, any Related Party of Company and/or Sellers.

6.1.18. <u>Obligations of Third Parties</u>. The Company has not assumed, guaranteed, endorsed or otherwise remained or is directly or subsidiarily liable for any obligation or debt of any Third Party.

6.1.19. <u>Powers-of-Attorney and Checking Accounts</u>. Except for the instruments listed in <u>Exhibit 6.1.19(i)</u>, Company granted no powers-of-attorney that are in force. <u>Exhibit 6.1.19(ii)<sup>14</sup></u> contains the list of all the bank accounts of Company, as well as of the Persons authorized to operate them.

6.1.20. <u>Litigation</u>. Except according to <u>Exhibit 6.1.20</u>, there are no Demands of any nature filed or initiated, to the Knowledge of Sellers, threatened against Company and/or in relation to the Infrastructure Items and/or the Business, or any investigation that involves or affects Company (or any Relevant Company Agreements and/or Business.

6.1.21. Protests. There are no protests of negotiable instruments against Company.

6.1.22. <u>Labor Matters</u>. Company does not have, never had and will not have, on the Closing Date, any employee, trainee, representative, collaborator, self-employed, temporary or outsourced worker or agent ("<u>Employees</u>"). On the Closing Date, there will be no amount due and not paid by Company to any Employee. To the Knowledge of Sellers, Company took the necessary steps to prevent the characterization of an employment relationship between Company and its service providers (and employees of such service providers).

6.1.23. <u>Insurance</u>. The Infrastructure Items are currently included in the insurance program of Sellers and their Affiliates, by means of Operating Risks (RO) and General Civil Liability (RCG) policies. The RO policy insures the fixed assets of Company, Sellers and their Affiliates, including the Infrastructure Items, against property damages that such fixed assets may suffer, including, without limitation, fire, electrical damage, windstorm, flooding and other damages arising from natural events. The RCG policy seeks to insure the amounts for which Company, Sellers and their Affiliates may be civilly liable, or in a settlement validated by the insurer, which are related to personal and property damages and losses directly caused to third parties by the Infrastructure Items and other assets, such as antennas.

6.1.24. <u>Antitrust Matters</u>. Company and Oi Companies Corporate Reorganization (exclusively with respect to the Business) have complied and comply with all antitrust laws and have not taken any action that could, in any way, be considered a practice of dumping and/or cartel.

6.1.25. <u>Corporate Transactions</u>. The Corporate Reorganization and any and all corporate transactions (including) including transformation, consolidation, spin-off, transfer of control, capital increase or reduction and merger, which have involved and/or do involve Company, since its incorporation, have been carried out in strictly compliance with the applicable law and presented economic, financial and accounting backing, in all its relevant aspects. Sellers or their Affiliates paid or collected any and all Taxes related to or resulting from the Corporate Reorganization.

<sup>&</sup>lt;sup>14</sup> <u>Note for the Draft</u>: information on bank accounts to be filled in before signing.

6.1.26. <u>Pre-Operating Company</u>. Prior to the conclusion of the Corporate Reorganization, Company had never operated or carried out any activity since the date of its incorporation, nor has it incurred, undertaken or promised to incur or undertake any obligation towards any Person (including Governmental Authorities). Prior to the conclusion of the Corporate Reorganization, Company had never owned any relevant or immovable assets, nor had it ever hired Employees or service providers.

6.1.27. <u>Brokerage Fee</u>. Sellers and Intervening Consenting Sellers have no obligation or liability for the payment of any fees or commission to any broker or agent in relation to the Transaction, for which Buyer or Company may be held liable.

6.1.28. Good Practices. Sellers and Intervening Consenting Sellers, including through any Employee, manager, agent or representative, (i) have not violated or violate any Anti-Corruption Laws; (ii) have not offered, paid, committed to pay or promised to pay or authorized the payment of any money or other valuables, contributions, expense reimbursement, nor have they given any gifts, benefits or delivered any type of property to any Person who is an officer, agent, servant or representative of any Governmental Authority or any political party, any candidate for public office or political party offices, or any other Person, knowing or having reason to believe that any or all of the money or valuable offered, given or promised would (a) facilitate or seek to facilitate, obtain favorable treatment in business, (b) remunerate favorable treatment in business, (c) facilitate or seek to facilitate obtaining special concessions or would serve as payment for concessions already obtained, in favor of or in relation to Sellers and Intervening Consenting Sellers, or (d) represent the violation of Anti-Corruption Laws. Company has adequate practices to avoid noncompliance with Anti-Corruption Laws. There is no investigation, allegation or request for information related to the violation of Anti-Corruption Laws by Sellers, Intervening Consenting Sellers and their Affiliates, or any of their Employees, officers, managers, agents, consultants or representatives.

#### Clause 7. Representations and Warranties of Buyer

7.1. <u>Buyer Representations and Warranties</u>. Buyer hereby provides Sellers with the following representations and warranties, which Buyer represents to be true, accurate, correct, legitimate, complete, exact, correct and unqualified on the date hereof, and will remain so until the Closing Date:

7.1.1. <u>Corporate Good Standing</u>. Buyer is a regularly incorporated, validly existing legal entity and in good standing pursuant to the Brazilian Laws, and was duly registered with the Commercial Registry and Governmental Authorities that require such registrations pursuant to the Law.

7.1.2. <u>Corporate Power and Jurisdiction</u>. Buyer has total corporate power, authority and jurisdiction to execute this Agreement and any other documents to be signed thereby due to and/or in relation to the Transaction, and also to comply with its obligations and responsibilities assumed in this Agreement and its respective Exhibits. The competent

corporate bodies of Buyer approved the execution hereof, the consummation of the Transaction and compliance with all obligations and responsibilities undertaken herein, and this approval constitutes any and all corporate acts required to comply with the obligations undertaken herein.

7.1.3. <u>Buyer's Bylaws</u>. The execution of this Agreement by Buyer and compliance by Buyer with its obligations undertaken herein shall not conflict or result in non-compliance with its bylaws, any other corporate document and/or any applicable Law or regulations.

7.1.4. <u>Relations with Third Parties</u>. The execution and formalization of this Agreement and compliance by Buyer with its obligations arising herefrom shall not conflict or result in violation of any contract, agreement or instrument to which Buyer is subject and/or bound, in that such conflict or violation may reasonably adversely affect the capacity of Buyer to comply with its obligations arising herefrom.

7.1.5. <u>Financial Capacity</u>. Buyer will have financial resources of its own or obtained by means of financing with third parties, in immediately available funds and in a sum sufficient to comply with all its obligations undertaken herein. Buyer assures that all amounts to be paid to Sellers do not directly or indirectly arise, derive or constitute profits from illegal activities under the Anti-Corruption Laws of the United States of America or Brazil.

7.1.6. <u>Litigation</u>. There is no Demand in progress or, to the Knowledge of Buyer, threatened, that affects Buyer's financial capacity under Clause 7.1.5 above or otherwise affects Buyer's ability to comply with all obligations undertaken in this Agreement, consummate the transactions and perform all acts provided for herein.

7.1.7. <u>Brokerage Fee</u>. Buyer has no obligation or liability for the payment of any fees or commission to any broker, prospector or agent in relation to the Transaction, for which Sellers or the Company (the latter, until the Closing Date) may be held liable.

7.1.8. <u>State of Infrastructure Items</u>. Buyer is aware that the interest in Company is being acquired with the Infrastructure Items as are on the Closing Date, pursuant to <u>Exhibit B</u>, whereby, except as expressly provided for in this Agreement, Buyer exempts Sellers and their Affiliates from any responsibility for such conditions, provided that, after the Closing Date, the transaction of the Company's Business, as well as the management of Infrastructure Items by Company, will be the responsibility of Buyer, as Company's Controller.

7.1.9. <u>Buyer's Capacity on the Closing</u>. On the Closing Date, Buyer shall have all the necessary structure, including systems and workforce, to conduct the regular operations of Company, especially the Infrastructure Items.

7.1.10. <u>Negotiation with Sellers</u>. Buyer acknowledges and agrees that, as permitted by law, except as provided in this Agreement:

(i) Neither Sellers, Company or any of their respective directors, officers, shareholders, members, Employees, Affiliates, controllers, agents, advisors, representatives or any other Related Party makes or has made any representation or provided any express or implicit

warranty regarding the accuracy or integrity of any projections or other statements on Company's future expectations, or even on any information on Company, its assets or other matters, in any case, except those included in this Agreement or in its exhibits;

(ii) Sellers have provided no representation or warranty on any rights of Third-Party beneficiaries or other rights that Buyer could claim under any studies, reports, tests or analyses prepared by any Third Parties for Company, Sellers or any of their Affiliates, even if made available for analysis by Buyer or its representatives; and

(iii) None of the documents, information or other material supplied at any moment or in any format by Sellers, by Company or by any of their respective Affiliates or representatives constitute legal advice, and Buyer hereby waives any right to claim to have received any legal advice from Sellers, Company, any of their respective Affiliates or their respective representatives or lawyers.

7.1.11. <u>Contact with Clients and Suppliers</u>. Neither Buyer or any of its Employees, agents, representatives or Affiliates has directly or indirectly contacted, without prior written consent of Company and Sellers, any officer, director, Employee, member, franchisee, supplier, distributor, client or any other Person in commercial relations with the Company or Sellers or any of their Affiliates before the Closing Date to discuss matters related to Company, except their managers, Employees, consultants and advisors as needed to assess and price the Transaction.

7.1.12. <u>Solvency</u>. Buyer is solvent under the applicable Law and able to pay its debts as they become due. To the Knowledge of Buyer, there is no procedure pertaining to any transaction or settlement with creditors, or any procedure of bankruptcy, judicial or extrajudicial reorganization, much less other insolvency procedure, involving or imminent against Buyer, and no event has occurred that, under the applicable Law, could justify the initiation of such procedures.

71.13. <u>No Violations</u>. Buyer is not in default, nor has any event occurred that, including due to notice or expiration, or both, may be understood as evidenced default and/or otherwise may limit the capacity of Buyer to exercise its rights and/or enforce compliance with any obligations owed thereto, in the form and conditions existing until the date of execution hereof.

7.1.14. <u>Sophisticated Investor</u>. Buyer has knowledge and experience in financial and business matters, in the telecommunications infrastructure industry, notably in the segment of towers and Roof Tops for the installation of antennas for transmission and radio frequency, being, therefore, able to evaluate Company and the Infrastructure Items, as well as the advantages and risks of an investment in the Shares and in the operation of such Infrastructure Items.

7.1.15. <u>Due Diligence</u>. Buyer acknowledges that it conducted due diligence of the Company and the Infrastructure Items based on information and documents provided by Sellers. Buyer acknowledges that, except for the representations and warranties provided in Clause 6.1 above, neither Sellers, Company nor any other Person has made any representation or

warranty, express or implicit, as to the accuracy or completeness of any information about Company, Infrastructure Items, Leased Areas, Site Licenses or other issues not included in this Agreement. Nothing in this Clause 7.1.15 will restrict or otherwise affect Buyer's right to be indemnified for any Losses under this Agreement.

# Clause 8. Acts between Signing and Closing

8.1. Ordinary Course of Business. From the date of execution (signing) of this Agreement to the Closing Date, Company will conduct its activities on the Ordinary Course, and Sellers undertake to make Company conduct its activities on the Ordinary Course, in a substantially equal and consistent manner with which Business had been conducted until the date of execution of this Agreement. Sellers will not perform and will not allow Company to perform any act or activity outside of its Ordinary and regular Course, and Sellers must perform all acts reasonably necessary to protect the assets, properties, rights and agreements, in order to preserve its business until the Closing Date. Without prejudice to the foregoing, Sellers may perform all acts necessary for the ordinary conduct of the business or prospects of Company, including in relation to the Infrastructure Items, or any of its operations, subject to the obligations not to do, as agreed herein (including Clauses 8.1.1 and 8.1.2 below).

8.1.1. Sellers and Company hereby undertake to, from the date hereof to the Closing Date, on an irrevocable and irreversible manner, not assign, dispose of, donate, transfer, lease and/or create any Liens, directly or indirectly, free of charge or for consideration, (i) part or all of the Company's Shares; and/or (b) the rights conferred by the Shares (political and/or equity), including preemptive rights in the acquisition of Shares and in the subscription of new shares or other securities issued by Company, as well as not to create and/or contract any obligation that may reduce the value of the Shares, and/or (c) any Infrastructure Items, other Company Assets and/or any Site Licenses related to the Infrastructure Items.

8.1.2. Sellers undertake, from the date hereof to the Closing Date, not to perform, and to prevent Company from performing, any of the acts described below (as well as not to authorize, agree, engage in negotiations or commit to perform any of the acts listed below), unless previously approved in writing by Buyer:

(i) amendment to Company's Bylaws;

(ii) the issuance or creation, by Company, of equity interests, options or any right to acquire or subscribe for shares or any securities issued by Company (including securities convertible into, or exchangeable for, shares issued by Company or that grant to its holder the right to acquire or subscribe for shares or any securities issued by Company);

(iii) any legal transaction that has the effect of changing the number of Shares;

(iv) any split, reverse split, redemption, amortization, reclassification or repurchase of shares issued by Company;

(v) the transformation, consolidation, spin-off, and merger transactions, as well as any other form of corporate reorganization or transaction involving Company and/or its Assets;

(vi) the acquisition, constitution of Liens and/or the disposal, by Company, of any equity interests in any companies, partially or totally, free of charge or for consideration, directly or indirectly, as well as Company's equity interest in any association, partnership or joint venture;

(vii) the waiver, renunciation or pardon by Company, of any rights, options, faculties or privileges, except those that together amount to less than one million reais (BRL 1,000,000.00);

(viii) any loans or capital contributions to any Person by Company;

(ix) the entering into agreements or proposition for entering into agreements within the scope of Claims involving Company, Company Assets or the Business;

(x) the assumption of any new obligation outside the Ordinary Course, individually or jointly, active or passive, in an amount exceeding one million reais (BRL 1,000,000.00), including the assumption of any liability for non-compliance with or termination of any legal transaction;

(xi) the contracting of any Debt or the granting of any type of security interest or personal guarantee whether for own or Third Party obligations;

(xii) any remission, pardon or any type of unilateral extinction of any credits held by Company against any Persons;

(xiii) the sale, donation or, in any way, the disposition of any assets, rights and/or any other Company Assets, including the Infrastructure Items, as well as the constitution of Liens on any Company Assets, including the Infrastructure Items and the Site Licenses applicable to such Infrastructure Items;

(xiv) any changes in the Company's financial, tax, accounting, social security, insurance, labor and environmental policies, methods or practices;

(xv) entering into, modifying or terminating any Relevant Agreement, except for the purposes of Full Regularization and subject to the terms of this Agreement;

(xvi) the disbursement of any amounts relating to Cash in the period from the Verification Moment to the Closing Date;

(xvii) any legal business with Related Parties, except for the Lease Agreements for the purposes of Full Regularization and subject to the terms of this Agreement;

(xviii) the hiring of any Employees by Company;

(xix) the making or commitment to make any capital investments (CAPEX) in the Company whose payment or consideration occurs at any time after the Closing Date;

(xx) entering into, amending or terminating, at the initiative of Company, any agreements that involve, individually or jointly, amounts in excess of one million reais (BRL 1,000,000.00);

(xxi) introduction of any substantial change in commercial and/or administrative practices of the business; and

(xxii) filing for Company's bankruptcy or judicial or extrajudicial reorganization, as well as the dissolution, termination or extinction of Company.

8.1.3. Buyer, through consultants, accountants and other authorized representatives, may, upon reasonable notice and provided that it is strictly necessary for the assessment of the Closing conditions, from the date of CADE Approval to the Closing Date, have access to the Company's facilities, books, agreements and records. Access to documents and information of Company under the terms of this Clause 8.1.3 will not affect or modify any representation or warranty provided by Sellers hereunder.

#### Clause 9. Indemnification

9.1. <u>Sellers' Obligation to Indemnify</u>. Sellers hereby jointly and severally undertake to indemnify, defend and hold Buyer and its Related Parties, including, after the Closing, Company (individually "<u>Buyer Indemnified Party</u>" and collectively "<u>Buyer Indemnified Parties</u>") harmless and to undertake liability for any Loss that may be actually incurred or borne by Buyer Indemnified Parties, as a consequence or as a result of any of the issues listed below (except, in any case, for the Regularization Costs, which will not be indemnified under the terms of this Agreement and will be subject to the provisions of Clause 10.3 below):

9.1.1. Hypotheses of Indemnification by Sellers:

(i) non-compliance, inaccuracy, omission, falsehood or error in any representation and/or warranty provided by Sellers in Clause 6 hereof;

(ii) any and all acts, facts or omissions relating to the Infrastructure Items, the Leased Areas, the Site Licenses applicable to such Infrastructure Items and the Relevant Company Agreements, which occurred only and exclusively until the Closing Date (although its manifestation or knowledge occurs in a supervening moment), which result in the disassembly, assembly, reallocation and/or construction of any new infrastructure item (but expressly subject to the Regularization Costs, under the terms of Clause 10.3 below), in which case Sellers will be liable for all Losses associated with such disassembly, assembly, relocation and/or construction (including costs and expenses associated with the termination or expiration of the applicable Lease Agreement);

(iii) any breach, omission or default, partial or total, of any obligation of Sellers (or of Company, until the Closing Date) that results in the breach of the obligations contained herein; provided that such breach and any Loss resulting therefrom are not remedied within thirty (30) days from receipt of written notice from Buyer of such breach;

(iv) [except with respect to a Third Party Demand arising from Site Irregularities (as provided for in Clause 9.1.1(v) below), any and all acts facts or omissions related to Infrastructure Items, Leased Areas, Site Licenses applicable to such Infrastructure Items, to Relevant Company Agreements, to the Business or Company, of any nature, including labor, social security, civil, insurance, environmental, tax or any other nature, in each case, whose generating event has occurred totally or partially until or on the Closing Date, even if the effects thereof only materialize in the future];

(v) Demands from Third Parties based on Site Irregularities (including Overload Irregularities), whose generating event occurred at any time before the Closing Date and within twelve (12) months from the Closing Date, even if the effects thereof only materialize in the future;

(vi) eviction or any other defect that affects the validity and/or effectiveness of the transactions through which Buyer acquires the Shares or that prevents the ownership, possession and free use, enjoyment and disposition of the Shares by Buyer; and

(vii) acts, facts or omissions liable to Sellers and their Affiliates and Related Parties, of any nature, occurring prior or after the Closing Date, which are directly charged from any of Buyer Indemnified Parties due to succession or under claims that such Persons are part of the same economic group, including, but not limited to, those related to civil, environmental, labor and tax liabilities.

9.1.2. Waiver, in whole or in part, by Buyer as to compliance with one or more Closing Conditions Precedent will not exempt Sellers from the obligation to indemnify Buyer Indemnified Party for any Losses incurred due to the waived conditions precedent.

9.1.3. Even if Buyer will approve, or cause to be approved, the accounts and financial statements of any subsequent fiscal year of Company, such approval will not mean that Company has waived the right to demand the redress provided for in this Clause 9.1, with Sellers remaining bound as provided for in this Agreement.

9.1.4. The indemnification obligations provided for in this Clause 9.1 shall apply even if any Buyer Indemnified Party has or becomes aware, at any time, of any non-compliance, violation, omission or inaccuracy of any of the representations or warranties provided herein, as well as of any contingencies, liabilities, debts or other obligations, subject, however, to the provisions of Clause 9.1.1(iv) above with respect to the Exhibits to this Agreement.

9.2. <u>Buyer's Obligation to Indemnify</u>. Buyer hereby undertakes to indemnify, defend and hold Sellers and its Related Parties (individually "<u>Sellers Indemnified Party</u>" and collectively "<u>Sellers Indemnified Parties</u>") harmless and to undertake liability for any Loss that may be

actually incurred or borne by Sellers Indemnified Parties, as a consequence or as a result of any of the issues listed below:

9.2.1. Hypotheses of Indemnification by Buyer:

(i) non-compliance, inaccuracy, omission, falsehood or error in any representation or warranty provided by Buyer under Clause 7 of this Agreement;

(ii) any breach, omission or default, partial or total, of any obligation of Buyer that results in a breach of the obligations contained herein; provided that such breach and any Loss resulting therefrom are not remedied within thirty (30) days from receipt of a written notice from any of the Sellers of such breach; and

(iii) acts, facts or omissions liable to Company or related thereto, of any nature, occurring prior or after the Closing Date, which are directly charged from any of Sellers Indemnified Parties due to succession or under claims that such Persons are part of the same economic group.

9.2.2. Buyer shall have no liability or obligation to indemnify Sellers for any Losses whose generating event is prior to the Closing Date, especially with regard to administrative fines and penalties imposed by any Governmental Authorities, nor will Buyer succeed Sellers in any obligation and/or liability pertinent to Sellers.

9.3. <u>Indemnification Procedure</u>. Subject to the limits provided for in this Agreement (including, but not limited to, those provided for in Clause 9.7 below), if there is a Demand that may result in the indemnification obligation by Sellers or Buyer, pursuant to Clauses 9.1 and 9.2 above, respectively, the Parties and Company will be subject to the following procedures:

9.3.1. In the case of a direct Demand from one of the Parties, the Indemnified Party may, at any time after the Closing Date, send a notice to the Party from whom it seeks indemnification under the terms of this Agreement ("<u>Indemnifying Party</u>"), stating in detail the respective Loss, together with a copy of all available documentation related to the Loss in question ("<u>Notice of Indemnification</u>").

9.3.2. In the case of a Claim filed by a Third Party (including Governmental Authorities) that gives or may give rise to a recoverable Loss under the terms of this Agreement ("<u>Third Party</u> <u>Claim</u>") against the Indemnified Party (and, for the purposes of this Clause 9.3.2, Company will be considered as Buyer Indemnified Party), the Indemnified Party must send a notice to the Indemnifying Party (i) before the completion of one third of the period available for the submission of the defense or any other applicable, judicial or administrative measure ("<u>Defense</u>"), or (ii) if the period for Defense is less than five (5) Business Days, before half of the period available for Defense (in both cases "<u>Notice of Third Party Demand</u>"), together with copy of all available documentation related to a Third Party Demand.

9.3.3. The Indemnifying Party shall, within five (5) Business Days after receiving the Notice of Indemnification, send a notice ("Notice of Procedure") informing the Indemnified Party if:

(i) it agrees with its liability for the payment of the Loss in question (already materialized or not), with its silence being valid as agreement; or

(ii) it does not agree with its liability for the payment of the Loss in question (already materialized or not), in which case, it must provide all the arguments, documents and information to support its understanding.

9.3.4. Whenever the Indemnifying Party sends a Notice of Procedure informing that it does not agree with its liable for the payment of the Loss in question (already materialized or not), the Parties must meet within the next five (5) Business Days to try to reach, in good faith, an agreement on the Demand related to such Loss (already materialized or not). Failure of the Parties to reach an amicable agreement on the liability for the payment of such Loss shall be resolved in Arbitration.

9.4. <u>Third Party Demand</u>. If the Demand in question is a Third Party Demand against the Indemnified Party, the following mechanisms must be observed if the Indemnifying Party agrees with its liability:

9.4.1. The Indemnifying Party shall, in the first half of the legal term to submit due defense or challenge, (i) make the payment or authorize the reimbursement of the amount in question; or (iii) delegate to the Indemnified Party the submission of Defense to a Third-Party Demand, in which case the Indemnified Party shall defend against the Third-Party Demand in a diligent manner, with the silence of the Indemnifying Party being understood as its choice for the provisions in subitem "(b)" of this Clause.

9.4.2. If the Indemnifying Party chooses to present a Defense, the Indemnifying Party shall appoint and hire a lawyer responsible for conducting such Defense, giving priority to the lawyers already appointed for such cases, with the Indemnified Party undertaking to grant to the appointed lawyer the necessary powers to conduct the Defense, as well as to provide all documents and information required for preparing the Defense. The Parties shall cause the appointed lawyers to keep the Parties constantly informed about the progress of the Third-Party Demand, by providing copies of all procedural papers that may be reasonable requested.

9.4.3. If the Indemnifying Party assumes the Defense of the Third Party Demand, the Indemnified Party will have the right to participate in the respective Defense and hire a lawyer, at its own expense, regardless of the lawyer hired by the Indemnifying Party, it being understood that the Indemnifying Party shall control such Defense and that, in this case, each Party shall bear the costs of its own lawyer. Regardless of whom conducts the Defense, the Parties shall cooperate with the Defense, and such cooperation shall include, to the extent requested, on reasonable grounds, the provision of records and information relevant to the Defense against the Third Party Demand, as well as, where applicable, the granting of power-of-attorneys necessary for the conduct of the Defense.

9.4.4. The Party that assumes the defense against a Third Party Demand shall consult the other Party in advance as to enter into any agreement, commitment or release from such Third Party Demand and obtain the agreement of the other Party to do so.

9.4.5. Notwithstanding the provisions in this Clause 9.4, et. seq., the Parties agree that the Indemnified Party may choose to conduct the Defense (without this fact removing the Indemnifying Party's obligation to indemnify under this Agreement) in the following cases: (a) the Third Party Demand involves imputations of a criminal nature or related to Anti-Corruption Laws to the Indemnified Party, (b) in Third Party Demands in which, in terms of interlocutory relief or provisional remedy, account freezing and/or imposition of Liens on assets or properties of the Indemnified Party, and (c) Third Party Demands whose nature of the object represents risk or possibility of setting an adverse precedent against the Indemnified Party (as, for example, in repeated matters). In the cases above, the Indemnified Party shall (x) send a Notice of Third Party Demand to the Indemnifying Party informing that it will assume and conduct the Defense of the Third Party Demand, (y) keep the Indemnifying Party duly informed about the progress of the respective Third Party Demand and provide supplementary information and copies of any documents reasonably requested by the Indemnifying Party, and (z) refrain from entering into agreements without the prior consent of the Indemnifying Party, which will not be unreasonably denied. The Parties also agree that, in the event that it is possible to determine that the largest portion of the obligation object of a Third Party Demand is not the liability of the Indemnifying Party, but rather of the Indemnified Party, regardless of the period referred to in the Third Party Demand or the date of submission of a Notice of Third Party Demand, then the Indemnified Party may also conduct the Defense of such Third Party Demand under the terms set forth in this Clause 9.4.5.

9.4.5.1. If the Indemnified Party assumes the Defense of the Third Party Demand pursuant to Clause 9.4.5 above, the Indemnifying Party shall have the right to participate in the respective Defense and hire a lawyer, at its own expense, regardless of the lawyer hired by the Indemnified Party, being understood that the Indemnified Party shall control such Defense and that, in this case, the Indemnifying Party shall bear the costs of its own lawyer. Regardless of whom conducts the Defense, the Parties shall cooperate with the Defense, and such cooperation shall include, to the extent requested, on reasonable grounds, the provision of records and information relevant to the Defense against the Third Party Demand, as well as, where applicable, the granting of power-of-attorneys necessary for the conduct of the Defense.

9.4.6. For the avoidance of doubt, regardless of whom conducts the Defense, the Indemnifying Party will be liable for all costs and expenses associated with the Defense of any Third Party Demand not challenged by Notice of Procedure (except for fees and other expenses resulting from the hiring of lawyers by the Indemnified Party under Clause 9.4.3 above), including, but not limited to, any and all costs, expenses, fees, administrative or court fees and judicial and administrative deposits required or necessary to allow the Defense to be filed and properly conducted.

9.1.7. In the case of Demands already pending before the Closing Date, their Defense will remain under the responsibility of the Party that has been leading them until the Closing, including with regard to its costs, by adopting the strategy and using the specialized lawyer chosen and indicated by such Party. However, it is agreed between the Parties that Sellers may not enter into agreements involving the assumption of commitments or obligations by any Buyer Indemnified Party, without Buyer's prior consent, which will not be unreasonably denied. For the purposes of the foregoing, any of Sellers shall notify Buyer of the terms of any agreement Sellers intend to enter into, and Buyer will have a period of five (5) Business Days to respond whether or not Buyer agrees with the proposed agreement.

9.5. <u>Compliance with Procedures</u>. Any failure by the Indemnified Party in complying with the procedures and commitments undertaken herein (especially those included in this Clause 9) will not exempt the Indemnifying Party from its obligation to reimburse or indemnify the Indemnified Party for the Loss in question, except to the extent such Loss could be settled, mitigated, reduced or prevented in case the Indemnified Party had complied with the provisions herein.

9.6. <u>Payment of the Indemnification</u>. The payment of any indemnification due by the Indemnifying Party to the Indemnified Party, under the terms of this Clause 9, must be made within fifteen (15) Business Days from the date on which a Loss object of a Notice of Indemnification or Notice of Third Party Demand and not challenged by a Notice of Procedure, is actually disbursed or supported by the Indemnified Party.

9.6.1. Failure to pay the indemnification within the period indicated in Clause 9.6 above will subject the Indemnifying Party to the payment of late-payment interest of one percent (1%) per month on the outstanding balance, calculated *pro rata temporis*, without prejudice to the provisions of article 389 of the Brazilian Civil Code. In the event the Indemnifying Party fails to comply, fully and in a timely manner, with the provisions of Clause 9.6 above, the Indemnified Party may exercise all its rights, file all claims and rely on all remedies provided for in this Agreement and the applicable Laws until full satisfaction of the Indemnifying Party's obligations to indemnify, pursuant to Clause 9.

9.7. <u>Limitations on Indemnification Obligation</u>. The Parties hereby acknowledge and agree that the obligation (a) of Sellers, provided for in Clause 9.1 above, and/or (b) of Buyer, provided for in Clause 9.2 above, will be subject to the following conditions:

9.7.1. <u>Maximum Limit</u>. Except as provided in Clause 9.7.5, the aggregate amount of all indemnified Losses by the Indemnifying Party may not exceed the amount equivalent to fifty percent (50%) of the Final Purchase Price ("<u>Maximum Limit</u>").

9.7.2. <u>Time Limit</u>. The indemnification obligation provided for in Clauses 9.1 and 9.2 above will remain valid and effective with respect to any Losses notified by the respective Indemnified Parties during the statutory preemptive or limitation period of each of the different types of obligations provided for in this Agreement, as of the date hereof, and/or until the res judicata of the respective administrative, judicial decision, or final arbitration decision or final settlement that is reached, whichever is longer.

9.7.3. <u>Individual De Minimis Amount</u>. The Indemnifying Party shall not be required to indemnify for any individual Losses at an amount less than fifty thousand reais (BRL 50,000.00) (the "<u>Individual De Minimis Amount</u>"); provided that (a) Losses having the same taxable event (or causes of action of the same nature) will be aggregated for purposes of reaching such Individual Floor Individual De Minimis Amount; and (b) Losses below the Individual De Minimis Amount will not be considered for the purposes of calculating the Global De Minimis Amount.

9.7.4. <u>Global De Minimis Amount</u>. The Indemnifying Party shall not be required to indemnify until the global amount of the Losses under its responsibility is greater than fifty thousand reais (BRL 500,000.00) ("<u>Global De Minimis Amount</u>"). Once the Losses reach such a Global De Minimis Amount, Sellers or Buyer, as the case may be, will be liable for the entire amount of the Losses, including those incurred before the Global De Minimis Amount was reached.

9.7.5. <u>Exceptions</u>. The Maximum Limit, the Individual De Minimis Amount and the Global De Minimis Amount do not apply to Losses arising or resulting from: (a) the breach of any of Seller's Fundamental Representations and Warranties or Buyer's representations provided for in Clauses 7.1.1 (Corporate Good Standing) and 7.1.2 (Corporate Power and Jurisdiction); and/or (b) any acts of proven willful misconduct, fraud or bad faith; and/or (c) the indemnity assumptions provided for in Clauses 9.1.1(iii), 9.1.1(vi), 9.1.1(vii), 9.2.1(ii), 9.2.1(iii) and/or 10.2; and/or (d) the provisions in Clause 10.2.3 below.

9.7.6. Sellers will not be required to indemnify for the Regularization Costs, according to Clause 10.3.

9.7.7. Sellers will not be required to indemnify for any Losses already covered by the mechanism in Clause 10.2.2 for recomposition of revenues due to an Irreversible Situation.

9.7.8. Accumulated Losses on the Global De Minimis Amount will be adjusted by the positive variation of the IPCA/IBGE index from the date they would have become due had it not been for the existence of the Global De Minimis Amount.

9.8. <u>Effective Amount to be Indemnified</u>. For purposes of the indemnification obligations (a) of Sellers, as provided for in Clause 9.1 above, and/or (b) of Buyer, as provided for in Clause 9.2 above, for the calculation of the effective amount of reimbursement for any Loss, the following principles shall be applied:

9.8.1. If the respective Indemnified Party has the right to claim the Loss of any Third Parties due to insurance or the sum shall be reduced in order to fully reflect any indemnification (including due to insurance) that the Indemnified Party and/or the Company in fact received due to the fact, condition or circumstance originating the indemnified Loss;

9.8.2. The amount of the Loss shall be added by the amount corresponding to any Taxes levied on the indemnity (less the reduction in the amount of Taxes payable resulting from such Loss, all calculated under the accrual basis), and the costs and expenses incurred due to the receipt of these indemnities and/or reimbursement, by making the due gross-up, if

applicable, so that the Indemnified Party receives, the total net amount of the Losses whose indemnification is due to the Indemnified Party hereunder; and

9.8.3. Exclusively with respect to Sellers' obligation to indemnify, if there are any court deposits within the scope of a Third Party Demand in which the Loss is discussed (except for the court deposits made after the Closing Date that have not been supported by Sellers), the amount of the Loss to be considered will be net of the balances of such deposits.

9.9. <u>Best Efforts</u>. The Parties and the Intervening Consenting Sellers agree to use their best commercial efforts (and also to cause their respective Indemnified Parties to do so) to, in good faith and considering the market practices, avoid constituting any Loss and, in case it is constituted, to mitigate the effects thereof. Nothing in this Clause 9.9 (a) will restrict, limit or otherwise affect Buyer's right to be indemnified for any Losses under this Agreement (nor should it be construed as a restriction or limitation of such Buyer's right); and/or (b) will impose any type of obligation of Buyer and/or Company after the Closing Date to perform any acts, adopt rules and/or observe specific standards of conduct, nor does it constitute any type of limitation or restriction on the way in which Buyer will conduct and direct its activities and those of Company (including management) of the Infrastructure Items after the Closing Date.

# Clause 10. Infrastructure Items and Lease Agreements

10.1. Full Regularization.

10.1.1. Sellers will use their best efforts to, by the Closing Date, (i) complete the Full Regularization of all Lease Agreements, including those Lease Agreements that are not listed in Exhibit 4.2.1(ix), and (ii) send a written notice to Buyer providing a copy of the existing Lease Agreements that have not been included in Exhibit 4.2.1(ix) on the date of execution of this Agreement.

10.1.2. Any Lease Agreements that are made available to Buyer pursuant to Clause 10.1.1 will then be considered as Lease Agreements Identified for the purposes of Buyer Conditions Precedent provided for in Clause 4.2.1(ix) and, in the case of Full Regularization until the Closing hereunder, will be computed for the purposes of reaching the Minimum Revenue index, and Exhibit 4.2.1(ix) will be considered automatically amended to include such Lease Agreements.

10.1.3. Lease Agreements that are not listed in <u>Exhibit 4.2.1(ix)</u> on the date of execution of this Agreement and the Lease Agreements Identified for which there was no Full Regularization until the Closing are jointly referred to as "<u>Agreements Pending Regularization</u>". Sellers agree that (a) they must pay or promptly reimburse Buyer or Company for fifty percent (50%) of any and all costs related to the Full Regularization of the Agreements Pending Regularization incurred by Buyer and/or Company after the Closing Date; and (b) if Company loses the right to use any of the Infrastructure Items due to reintegration and/or loss of ownership of the Leased Areas related to the Agreements Pending

Regularization, such situation will constitute an Irreversible Situation for the purposes of this Agreement.

10.1.4. In the event that the counterparties in the Lease Agreements require financial compensation (including an increase in the amount of annual remuneration/rent) for making the Full Regularization in an amount greater than five percent (5%) of the annual remuneration/rent due under such Lease Agreements, Sellers may, at their sole discretion, (a) fully comply with the financial conditions imposed by the respective counterparty by making payment in full and in cash of the amounts of said financial considerations to the counterparty, or request Company to apply the provisions of the Oi Sharing Agreement; or (b) not to make the Full Regularization of such Lease Agreement, in which case the Leased Area and the Infrastructure Item located therein will be considered to be in an Irreversible Situation.

10.2. <u>Irreversible Situation</u>. The Parties acknowledge and agree that, if due to any acts, facts, omissions, events or circumstances occurring up to or on the Closing Date (including, but not limited to, the Site Irregularities), Company (i) loses the right to use any of the Infrastructure Items listed in <u>Exhibit B</u> (including chances of losing possession of any of the Leased Areas in which one or more Infrastructure Items are currently installed); and/or (ii) prove a technical (including engineering defects), legal and/or documentary impossibility for the licensing, operation, occupation or maintenance of an Infrastructure Item, and/or (iii) are required, by an unappealable decision in a Demand by a Governmental Authority and/or Third Party, to terminate the operation of an Infrastructure Item, such Infrastructure Item or Leased Area will be considered in an "Irreversible Situation".

10.2.1. Upon identification of an Irreversible Situation, Buyer and/or Company may, at its sole discretion, notify the parties to the Oi Sharing Agreement, pursuant to Clause [•] thereof, about the occurrence of an Irreversible Situation ("<u>Notice of Irreversible Situation</u>").

10.2.2. Upon submission of a Notice of Irreversible Situation, the provisions of Clause [•] of the Oi Sharing Agreement will apply (without the application of any of the limitations provided for in Clauses 9.7.1 to 9.7.4), so that the amount corresponding to the amount of revenue generated by the specific Infrastructure Item object of the Irreversible Situation is divided by the number of Infrastructure Items remaining, plus the amount of the monthly Base Consideration (as defined in the Oi Sharing Agreement) and other applicable considerations provided for in the Oi Sharing Agreement paid by Sellers or their Affiliates in the Oi Sharing Agreement.

10.2.3. Sellers acknowledge and agree that, within the scope of an Irreversible Situation, Company may also incur additional losses to the loss of revenue generated by the specific Infrastructure Item object of the Irreversible Situation, which, with the exception of the Regularization Costs (which will be treated in accordance with Clause 10.3), will be indemnified under the terms, limits and conditions of Clause 9.1 of this Agreement (including the Losses associated with the disassembly, assembly, relocation, construction and costs and expenses associated with the termination or expiration of the Lease Agreement), pursuant to Clause 9.1.1(ii)).

10.3. The Parties agree and covenant that if the Oi Sharing Agreement is no longer in force for any reason (except in the event that Buyer and/or any of its Affiliates that enter into the Oi Sharing Agreement causes such termination), but the obligations provided for in Clauses 10.1 and/or 10.2 are still in effect and taking effect, then Clauses 8.1.4 and 8.5 of the Oi Sharing Agreement and the other provisions of the Oi Sharing Agreement that deal with the Irreversible Situation or Regularization Cost Increase will be automatically incorporated into this Agreement by reference, *mutatis mutandis*, as if they were transcribed here.

10.4. <u>Regularization Costs</u>. Buyer acknowledges and agrees that the Infrastructure Items are subject to Site Irregularities, as specifically indicated in Exhibit B to this Agreement, provided that, as indicated in <u>Exhibit B</u>, <u>no indemnification will be paid</u> under this Agreement for any costs and expenses incurred by the Buyer and/or Company to remedy, mitigate, alter or regularize the Site Irregularities (e.g., expenses with lawyers' fees, costs, fees of Governmental Authorities, etc.) and costs and expenses related to the necessary repairs, interventions and engineering works identified in <u>Exhibit B</u> ("<u>Regularization Costs</u>").

10.4.1. Buyer undertakes to use its best efforts to remedy the Site Irregularities within twelve (12) months from the Closing Date.

10.5. <u>Overload Irregularities</u>. The Parties acknowledge that the Site Irregularities include irregularities related to overload, as described in <u>Exhibit B</u> ("<u>Overload Irregularities</u>"). If (i) an Irreversible Situation occurs related to such Overload Irregularities within twelve (12) months from the Closing Date, the procedure provided for in Clause 10.2 above will apply; or (ii) an Irreversible Situation occurs related to such Overload Irregularities after twelve (12) months from the Closing Date, Buyer will not be entitled to apply the procedure provided for in Clause 10.2.

10.6. <u>Sellers' Obligations — Payment Maintenance</u>. Without prejudice to the obligation to indemnify under Clause 9.1.1(ii), Sellers further undertake, on their behalf and on behalf of their Related Parties, not to interrupt the payments due under the Oi Sharing Agreement, related to the Infrastructure Items that have their activities temporarily suspended (i) due to a Third Party Demand (including Governmental Authority) arising from Site Irregularities, or (ii) for the regularization or remediation of a Site Irregularity, except for regularization or remediation of Overload Irregularities carried out after twelve (12) months from the Closing Date, in which case the provisions of this Clause 10.6 will not apply while the temporary suspension of activities continues.

10.6.1. Sellers' obligation undertaken in Clause 10.6 above on behalf of their Related Parties constitutes a third party fact promise, under the terms and for the purposes of article 439, *et seq.*, of the Brazilian Civil Code.

10.7. <u>Revenue Received After Closing</u>. If any of Sellers and/or their respective Related Parties receive any amount, money and/or revenue relative to or involving, directly or indirectly, any Infrastructure Item after the Closing Date ("<u>Revenue Received</u>"), then any of Sellers that receives such amounts undertakes, in an irrevocable and irreversible manner, to (i) promptly transfer and return such Revenue Received to Company, which must be done in

no later than ten (10) Business Days after the date of receipt of the Revenue Received in question, under penalty of Sellers to bear adjustment for inflation based on the positive variation of the IPCA/IBGE index from the date scheduled for payment until the date of actual payment thereof, and be subject to the payment of a non-compensatory fine of five percent (5%) on the amount due, as well as the payment of late-payment interest of one percent (1%) per month on the outstanding balance, calculated *pro rata temporis*, without prejudice to the provisions of article 389 of the Brazilian Civil Code; and (ii) notify the Person who made the payment of the Revenue Received in question, within ten (10) Business Days counted from the receipt of such Revenue Received, that all future payments must be made in favor of Company.

10.8. <u>Replacement of Guarantees</u>. Buyer undertakes to use its best efforts to replace the guarantees presented by Sellers with respect to Contractual Instruments (including Lease Agreements) entered into by Company and/or related to the Business, as such guarantees may be subject to notice of any of Sellers to Buyer requesting the replacement of guarantees provided by Sellers for the obligations undertaken by Company or related to the Business, by attaching supporting documentation to that effect. Buyer must provide Sellers with proof of release of the respective guarantees within one hundred and twenty days (120) days from the date of receipt of each notice sent by any of Sellers in the manner above in this sense. In the impossibility of replacing any guarantee, the Parties undertake to discuss the best solution for both Parties, and it is hereby established that Sellers undertake not to revoke the guarantees they have provided, and Buyer shall, after the expiration of the above period, start to compensate Sellers in the amount corresponding to two integers and five tenths per cent (2.5%) per year of the value of the guaranteed and not replaced lease.

10.9. <u>New Infrastructure Items</u>. Buyer shall use its best efforts to provide Sellers and/or its Affiliates with access to new infrastructure items that may be acquired by Buyer, by Company and/or their Affiliates from Sellers and/or their Affiliates, if it is of interest of these, under reasonable commercial conditions and within market standards.

10.10. <u>Insurance Hiring</u>. As of the Closing Date, Buyer shall provide, on its own account and at its own expense, the insurance covers Buyer deems necessary and convenient for the Infrastructure Items and the Company's operations.

# Clause 11. Termination

11.1. <u>Termination Events</u>. This Agreement may be terminated prior to Closing in the events described in detail below:

(i) by mutual and express agreement between Buyer and Sellers;

(ii) by any of the Parties, if the Closing does not occur by the Closing Deadline (extendable under Clause 11.2), or at a later date that may be agreed in writing between the Parties;

(iii) by the Party that does not cause the delay, if the Closing does not occur within fifteen (15) days from (a) the 12<sup>th</sup> Business Day from the receipt, by either Party, of the notice referred to in Clause 5.1; or (b) the waiver of Closing Conditions Precedent, as applicable; or any other date that may be agreed between Buyer and Sellers;

(iv) [by Buyer, in the event provided for in Clause 4.4.6(ii);]<sup>15</sup>

(v) by Buyer, in the event of a Material Adverse Effect or, in the case provided for in Clause 14.4, if Sellers present an update to Exhibits that represents a Material Adverse Effect;

(vi) by either Party, should the other Party and/or Company become insolvent or bankrupt, be voluntarily or compulsorily dissolved or liquidated, or require any type of bankruptcy reorganization or judicial or extrajudicial reorganization with creditors, except for the Judicial Reorganization in progress;

(vii) by any of the Parties, if the result of the Auction, in which Buyer was the winner, is declared null in some way by a final and unappealable judicial decision and the sale of the Shares under the Judicial Reorganization has to be redone through the organization of a new auction;

(viii) by either Party, if there is any Law or order from any Governmental Authority that makes the transactions provided for in this Agreement illegal or that permanently restricts the closing of the transactions contemplated by this Agreement; and (ix) by either Party, in the event of a breach of or non-compliance with any obligation, representation or warranty of the other Party provided for in this Agreement that has not been remedied within thirty (30) days from the date of receipt of a notice of said breach or non-compliance.

11.2. <u>Fine</u>. If this Agreement is terminated in the event provided for in Clause 11.1(iii), the Party causing the termination will be required to pay the other Party a non-compensatory fine in the amount of thirty-seven million and five hundred thousand reais (BRL 37,500,000.00), provided that, if the Party that caused the termination is Buyer, the fine will be due to Sellers in proportion to their respective equity interests in Company's share capital, as indicated by Sellers in the Pre-Closing Certificate. The fine provided for in this Clause may be offset against any amounts that may be owed by one Party to another, including as a result of other Contractual Instruments.

11.3. [CADE Extension. Sellers may, at their sole discretion, regardless of justification, extend the Deadline for an additional period of one hundred and eighty (180) days, to the extent necessary to obtain CADE Approval.]<sup>16</sup>

11.4. <u>Termination Effects</u>. In the event of termination of this Agreement due to the events stipulated above, this instrument will automatically cease to have effect before the Parties,

<sup>&</sup>lt;sup>15</sup> <u>Note for the Draft:</u> Provisions related to antitrust approvals to be excluded if not applicable at the time of executing this Agreement.

<sup>&</sup>lt;sup>16</sup> <u>Note for the Draft:</u> Provisions related to antitrust approvals to be excluded if not applicable at the time of executing this Agreement.

except for the provisions of Clause 12 (Notices), Clause 13 (Conflict Resolution) and Clause 14.11 (Confidentiality and Non-Disclosure), which will remain valid and effective and will therefore survive the termination of this Agreement, provided that the termination of this Agreement will not relieve the Parties from their liability for breach of this Agreement or affect the Parties' right to claim specific performance of the obligations provided for herein.

11.5. <u>Exercise Limitation</u>. Notwithstanding any other provision included in this Agreement, the right to terminate this instrument under this Clause 11 may not be exercised by the Party: (i) that has provenly failed to comply with its obligations set forth herein or arising herefrom; or (ii) whose non-compliance with its obligations or covenants hereunder has been the cause or resulted in the lack of fulfillment of any Closing Condition Precedent, except if such non-compliance has arisen from any evidenced event of Force Majeure or Act of God, as set forth in article 393 of the Brazilian Civil Code.

### Clause 12. Notices

12.1. <u>Notices</u> All communications, notices and/or notifications relating to this Agreement shall be made in writing and delivered: (i) by registered mail; or (ii) through a Register of Titles and Documents, or by judicial means; or (iii) by letter to be delivered by hand or telegram, return receipt requested, or by electronic mail (e-mail) with confirmation of receipt to the Party and to the addresses indicated below, or to another address that such Party may indicate by written notice as provided for in this Clause.

If to Sellers and/or Intervening Consenting Sellers:
[•]

If to Buyer: [•]

If to Company:

a) Until Closing:

b) From Closing:

12.2. <u>Reception Convention</u>. Notices delivered under this Clause will be deemed given: (i) at the time they are delivered, if delivered in person; (ii) at the time they are delivered, if sent by mail or courier service; and, (iii) on the date shown on the confirmation of receipt of the transmission, if sent by e-mail, unless such date is not a Business Day, in which case it will be deemed received on the Business Day immediately following.

12.3. <u>Change</u>. The Parties may change their respective addresses, as indicated above, by giving notice to all other Parties to this Agreement.

#### Clause 13. Conflict Resolution

13.1. <u>Arbitration</u>. Any disputes arising from this Agreement or otherwise related to it, including as to its existence, validity, effectiveness, interpretation of its terms, conditions, execution or termination ("<u>Dispute</u>") shall be resolved by arbitration as provided for in this Clause 13 ("<u>Arbitration</u>").

13.1.1. The Parties agree that, before initiating Arbitration to resolve any Dispute, they shall attempt to reach an agreement for an amicable settlement of such Dispute, within a period not exceeding fifteen (15) Business Days after receipt by a Party of notification of the existence of such Dispute, as sent by the other Party. The Parties agree that their obligation to resolve any Disputes amicably is an ancillary obligation that does not preclude the immediate initiation of Arbitration at any time, at the sole and discretionary criteria of either Party.

13.1.2. At the end of this period or if, at the discretion of either Party, it is impossible to reach an amicable solution, the Interested Party may submit the Dispute to Arbitration before the International Chamber of Commerce (<u>CCI</u>) Arbitration Chamber ("<u>Chamber</u>"), in accordance with its arbitration regulation ("<u>Regulation</u>") in force on the date of the request for the establishment of Arbitration, with the exception of the changes provided for herein.

13.1.3. The Arbitration will be conducted by three (3) arbitrators ("<u>Tribunal</u>"), one nominated by the requesting party and the other nominated by the requested party, in accordance with the Regulation. If there is more than one requesting party and/or more than one requested party, then the requesting parties and/or requested parties must together indicate their respective arbitrator. In the absence of agreement between the requesting parties or requested parties to indicate their respective co-arbitrator, all arbitrators must be appointed by the Chamber. The two arbitrators thus appointed shall nominate, by mutual agreement, the third arbitrator, who will act as president of the Arbitral Tribunal, within the term provided for in the Regulations. If any of the three arbitrators is not appointed within this period, then the Chamber shall appoint them, as provided for in the Regulation. Any and all controversies regarding the appointment of arbitrators by the parties, as well as the choice of the third arbitrator, will be settled by the Chamber. The Parties, by mutual agreement, remove the application of the provisions in the Regulation that limit the choice of a sole arbitrator, co-arbitrator or president of the arbitral tribunal to the list of arbitrators of the Chamber.

13.1.4. The Arbitration will be based in the City of São Paulo, State of São Paulo, Brazil, where the arbitration award will be issued, and it will be conducted in the Portuguese language. The Arbitral Tribunal shall judge the merits of the Dispute in accordance with Brazilian law and shall not judge by equity.

13.1.5. Any order, decision, determination or award handed down by the Arbitral Tribunal will be final and binding on the parties and their successors, who expressly waive any appeal. The arbitral award may be enforced before any judicial authority with jurisdiction over the parties and/or their assets.

13.1.6. Each party will bear the costs and expenses that it causes in the course of the Arbitration and the Parties will share equally the costs and expenses whose cause cannot be attributed to one of them. The arbitral award will assign to the losing party, or to both parties to the extent that their claims are not upheld, the final liability for the cost of the proceeding, including the reimbursement of attorneys' and other advisers' fees of reasonable value. The arbitration award shall not impose the payment of attorneys' fees due to loss of suit.

13.1.7. The parties choose the jurisdiction of the Judicial District of São Paulo, State of São Paulo, Brazil, and waive any other, however privileged, for the exclusive purposes of obtaining urgent relief to protect or safeguard their rights prior to the establishment of the Arbitral Tribunal, without this being considered a waiver of Arbitration. Any measure granted by the Judiciary Branch shall be promptly notified by the party that requested such measure to the Chamber. The Arbitral Tribunal, once constituted, may review, maintain or revoke the measures granted by the Judiciary Branch.

13.1.8. The parties agree that all aspects related to Arbitration, including its own existence, must be kept confidential. All its elements (including, but not limited to, the Parties' allegations, evidence, reports and other statements of third parties and any other documents presented or exchanged in the course of the arbitration proceeding) will only be disclosed to the Arbitral Tribunal, to the parties, their lawyers, Chamber officials, and any person needed to develop the Arbitration, unless disclosure is required to comply with any obligations imposed by applicable law, or by any Governmental Authority.

13.1.9. Any and all disputes related to the confidentiality obligation will be settled in a final and binding manner by the Arbitral Tribunal, which may take any measure to safeguard the confidentiality of the arbitration procedure, or any other matter related to Arbitration.

13.1.10. If two or more disputes arise in relation to this Agreement, or in any way related to it, the resolution thereof may occur through a single arbitration proceeding, pursuant to the Regulation. Before the constitution of the Arbitral Tribunal, the Chamber will be responsible for consolidating the referred disputes in a single arbitral procedure, in accordance with the Regulation. After the constitution of the Arbitral Tribunal, in order to facilitate the resolution of related disputes, the Arbitral Tribunal may, at the request of one of the parties, consolidate the arbitration procedure with any other pending arbitration procedure that involves the resolution of disputes arising from this Agreement, or otherwise related to it. The Arbitral Tribunal shall consolidate the procedures, provided that (i) they involve the same parties; (ii) there are factual and/or legal issues that are common between them; and (iii) the consolidation in these circumstances does not result in losses arising from unjustified delays in resolving such disputes. The jurisdiction to determine the consolidation of procedures and conduct the consolidated procedure will be the first arbitral tribunal constituted. The consolidation decision will be final and binding on all parties involved in the disputes and arbitration procedures object of the consolidation order.

13.1.11. The Intervening Consenting Sellers are expressly bound by this arbitration clause for all legal purposes.

#### Clause 14. Miscellaneous

14.1. <u>Amendments</u>. Any changes or amendment to this Agreement will only be considered valid if agreed upon in writing, in an instrument signed by all Parties.

14.2. <u>Assignment</u>. It is hereby established that the Parties may only assign and transfer, in whole or in part, the rights and obligations arising from this Agreement to Brazilian Affiliated Company(ies), with the prior written consent of the other Party, and subject to the following: (i) prior to such assignment, assignee shall assume, by means of a written instrument, signed and formalized by the other Parties, the fulfillment and compliance with all obligations, covenants and agreements established in this Agreement on behalf of assignor, and (ii) no assignment will exempt assignor from its obligations, covenants and agreements established in this Agreement or partial assignment of its contractual position, assignor will remain jointly and severally with the respective assignee(s) liable for full and timely compliance with such obligations, covenants and agreements, except for the assignment and/or succession of the contractual position resulting from spin-off, merger, combination or other form of corporate reorganization.

14.3. <u>Financing</u>. In view of Buyer's intention to obtain financing to make the transactions contemplated by this Agreement feasible, Sellers undertake, subject to the competitive restrictions that may apply, to use their best efforts to cooperate with Buyer with respect to the preparatory measures of Buyer for obtaining such financing, including: (i) provide Buyer with reasonably requested information regarding the Company, the Business and the Infrastructure Items, including for the purposes of (*i.a*) preparing the disclosure materials required to be presented to possible sources of financing, and (*i.b*) evaluating Company's Assets, with the purpose of constituting a guarantee on them (except that any Liens will be constituted only and with effect after Closing, and at Buyer's account and risk); (ii) provide Buyer with information related to "*Know your Customer*" procedures, and practices anti-corruption, against money laundering and terrorism; (iii) request the executives of Sellers and/or Company to participate in meetings with Buyer and possible sources of financing, at times and places to be defined by common agreement between the Parties).

14.4. Exhibit Update. The information contained in the Exhibits to the representations and warranties set forth in Clause 6.1 reflects the situation of Company, of Company's Assets, of the Infrastructure Items, Licenses of the Infrastructure Items, Leased Areas, Relevant Company Agreements, the Business and other information provided therein on the base dates indicated therein. The Parties agree that Sellers may update in good faith the information included in such exhibits (except for the list of Infrastructure Items contained in Exhibit B and for the Site Irregularities described therein and for the Overload Irregularities described in Exhibit B), provided that such updates may only refer to (i) acts, facts or omissions occurring after the date hereof or, exclusively with respect to representations and warranties referring to a specific date or period, after the date or period to which they refer; and (ii) the representations and warranties other than Seller's Fundamental Representations and Warranties. If Sellers wish to make any updates to such attachments during the period from the date hereof and the Closing Date, then Sellers shall send to Buyer a notice to that effect, which must contain: (a) the Exhibit subject matter of such update; and (b) the nature of the

event that gave rise to the need to update such exhibit ("Exhibit Update Communication"). If such updating of Exhibits to the representations and warranties (1) represents a Material Adverse Effect, Buyer shall have the right to terminate this Agreement, without incurring any penalty or right of indemnification on the part of Sellers, under the terms of Clause 11.1 below; or (2) does not represent a Material Adverse Effect, Buyer will have the obligation to accept such modification and/or inclusion, which will integrate the representations and warranties of Sellers for the purpose of fulfilling the condition established in Clause 4.2.1(i), provided, however, that any Losses incurred by Buyer Indemnified Parties related to such modifications and/or inclusions will be indemnified by Sellers under the terms of Clause 9.1.1.

14.4.1. Any update to Exhibit 4.2.1(ix) by Sellers will also be subject to the provisions of Clause 10.1 and its sub-clauses above.

14.5. <u>Oi Sharing Agreement</u>. Sellers, on their own behalf and on behalf of their Affiliates, hereby acknowledge and agree that, in the event of sale of business or any other form of disposal of all or a substantial part of Sellers' assets or business (including by virtue of the creation of an isolated production unit gathering the assets), the Oi Sharing Agreement shall be mandatorily assigned to the new holder of such assets, unless otherwise agreed by Buyer.

14.6. <u>Binding Act</u>. This Agreement binds the Parties and their authorized successors and assigns.

14.7. <u>Comprehensive Consensus and Documents</u>. This Agreement and all its Exhibits constitute the sole and entire agreement between Sellers and Buyer with respect to the purpose hereof, and will prevail over and replace any and all other documents and/or related, verbal or written agreements previously entered into by the Parties.

14.8. <u>Severability</u>. If any provision of this Agreement is found to be void, voidable, invalid or inoperative, the remaining provisions will not be affected and will therefore remain in full force and effect, as if the void, voidable, invalid or inoperative provision were not contained in this Agreement. In this case, the Parties will use their best efforts to replace the respective provision with a valid one and, to the extent possible, with the same economic effect.

14.9. <u>Subsequent Commitment</u>. In accordance with the terms and conditions hereof, each Party will perform and cause all other acts to be performed, will sign and formalize all other instruments, certificates and other documents that the other Party reasonably requests in order to comply with the purpose and achieve the objectives of this Agreement and the consummation of the Transaction.

14.10. <u>Press and Public Releases</u>. None of the Parties may issue or cause the issuance of any press release or public communication (including a "Material Fact") related to this Agreement without the prior knowledge and written approval of the other Party; provided, however, that no provision herein shall prohibit any of the Parties to issue or cause the issuance of any press release or public communication, to the extent such disclosure is required by Law or regulations of any applicable national stock market, in which case the

Party responsible for such disclosure shall grant the other Party reasonable time to comment in writing such communication before its issuance and/or disclosure.

14.10.1. For the avoidance of doubt, nothing in this Clause 14.10 shall be construed as a limitation on the Parties from making information about this Agreement available to their Affiliates, investors and representatives.

14.11. <u>Unfavorable Interpretation</u>. This Agreement is the result of joint analysis, discussion and negotiation by the Parties. Thus, in the event of any uncertainties or ambiguities that may arise from its wording, this Agreement will be interpreted as if drawn up jointly by the Parties, and no presumption of burden will arise favoring or disadvantaging any Party nor any provision herein will be interpreted in favor or against any of the Parties because of their effective role in drafting this Agreement.

14.12. Confidentiality and Non-Disclosure. The Parties hereby agree to keep confidential, for a period of two (2) years, counting from the date of execution of this Agreement, all technical, legal, operating and commercial information (except for the existence of this Agreement and its content, which may be registered by the Parties with the competent Registers of Titles and Documents) ("Confidential Information") that are directly or indirectly related to this Agreement and the Transaction covered herein, as well as all information obtained under the terms of the necessary negotiations and investigations for the execution of this Agreement, except, in the case of Buyer, when such availability is made to its Affiliates, Related Parties, creditors, investors and advisors, as well as to the banks with which Buyer eventually raises the funds necessary to pay the Preliminary Purchase Price or Final Purchase Price, in which case such Persons shall be required in advance to maintain the confidentiality of the information, and Buyer shall be liable for maintaining the confidentiality of such information. The delivery of any Confidential Information to any Third Parties by either Party must be preceded by the express approval of the other Parties, as the case may be. Sellers undertake not to disclose to any Third Party and/or Related Party any of the information to which they may have access, or which may be disclosed to them as a result of their position as shareholders of Company and the conduct of business (including manuals, know-how, procedures, etc.) of Company and any information already known by Sellers.

14.12.1. Confidential Information will exclude any information disclosed by the disclosing Party to the receiving Party that: (i) was already known by the receiving Party at the time of its disclosure by the disclosing Party, and has not been obtained through violation of any confidentiality clause or commitment; (ii) was in the public domain prior to their receipt by the receiving Party; (iii) became part of the public domain after receipt thereof by the receiving Party, for reasons not attributable to the receiving Party's actions or omissions; (iv) has been, after the disclosure, acquired in good faith, without any restriction of confidentiality, from a Third Party that is not bound by any non-disclosure agreement with the disclosing Party; or (vi) has been disclosed within the scope of the Judicial Reorganization Proceedings of Sellers and Oi.

14.12.2. The receiving Party may disclose the Confidential Information that is required by a Governmental Authority, including antitrust authorities, strictly to the extent that it is required, provided that such Party informs, if possible, previously and immediately such fact to the disclosing Party of the need for such disclosure, so that the latter can take appropriate measures with respect thereto.

14.12.3. Notwithstanding anything to the contrary in this Agreement, Confidential Information relating to this instrument may be disclosed (i) to Affiliates of the Parties and/or to investors, co-investors, auditors, financiers and consultants, creditors, potential buyers and managers of the Parties, (ii) if its disclosure is required by Law or regulatory rule applicable to any of the Parties or Intervening Consenting Sellers, or (ii) if its disclosure is required from the receiving Party by a Governmental Authority or any self-regulatory entity with jurisdiction over the Party or its Affiliates (including, for the avoidance of doubt, any requests for information within the scope of regulatory due diligence, inspections or verifications carried out in the ordinary course, within the scope of the Party's activities and business), limited, in any case, to what is strictly necessary, provided the receiving Party informs the disclosing Party in advance or immediately, if possible, of the need for such disclosure, so that the disclosing Party can take the applicable measures with the applicable Governmental Authority (or self-regulatory entity) to prevent such disclosure.

14.13. <u>Fiduciary Assignment</u>. The Parties agree that all receivables arising from the Oi Sharing Agreement are subject to fiduciary assignment to the banks financing the Transaction and/or Buyer.

14.14. <u>Liabilities</u>. Sellers acknowledge that they are and will continue to be solely liable for their contingencies or obligations, including the payment of any taxes due by Sellers or their Affiliates, whether or not they are included in the Judicial Reorganization Plan.

14.15. <u>Intervening Consenting Sellers</u>. Intervening Consenting Sellers represent that they are fully aware of all the terms and conditions of this Agreement and have no restrictions on such terms and conditions, the execution of this Agreement and/or the exercise of the rights and the fulfillment of the obligations agreed herein.

14.15.1. Company and Oi hereby undertake, for all legal purposes and effects, to (i) sign any and all documents, agreements, instruments, and corporate books as necessary to comply with the terms and conditions of this Agreement and the implementation of the Transaction; and (ii) always act, with regard to the terms, conditions, obligations and rights provided herein, in such a way that does not prevent, limit or restrict the respective fulfillment, implementation and/or compliance, as the case may be.

14.16. <u>Guarantee</u>: Oi enters into this Agreement in the capacity of guarantor, joint and several debtor and principal payer of all obligations undertaken by Sellers and by Company (in the latter case, until the Closing Date) under this Agreement, and grants a personal guarantee in the form of a suretyship in favor of Buyer.

14.16.1. Oi's obligations undertaken herein will not be affected by any acts or omissions that could exempt Oi from its obligations or affect them, including, but not limited

due to: (a) any extension of term or agreement between the Parties; (b) any novation or nonexercise of any right of Buyer; and (c) any limitation or disability, including arising from a request for extrajudicial reorganization, a request for judicial reorganization, intervention or bankruptcy.

14.16.2. Oi expressly waives, in an irrevocable and irreversible manner, the benefits of order, division, rights and exemption powers of any nature provided for in articles 333, sole paragraph, 365, 366, 368, 821, 824, 827, 834, 835, 837, 838 and 839, all of the Brazilian Civil Code, and articles 130 and 794 of the Brazilian Code of Civil Procedure.

14.16.3. Oi hereby expressly waives the right to subrogate to Buyer's credit rights, if it will honor, in whole or in part, Company's obligation under this Agreement.

14.16.4. This suretyship is provided by Oi on an irrevocable and irreversible basis, and it will enter into force on the date of execution of this Agreement, remaining valid in all its terms until full payment of all the obligations secured by it hereunder and in accordance with Article 818 of the Brazilian Civil Code.

14.16.5. Oi hereby acknowledges that the suretyship is provided for a term of two hundred and forty (240) months, which may be extended for another two hundred and forty (240) months in the event of beginning of foreclosure of the guarantees and, therefore, article 835 of the Brazilian Civil Code shall not apply.

14.17. <u>Waiver, Termination and Discharge</u>. No waiver, termination or discharge of this Agreement or any of its terms or provisions will bind any of the Parties, unless confirmed in writing. The tolerance of either Party with respect to the possible or continued failure by the other Party to comply with any obligation under this Agreement may not, under any circumstances, be construed as a waiver or novation to such rights and will not affect the Party's right to exercise them in the future.

14.18. <u>Tolerance</u>. A Party's delay in exercising a right provided for in this Agreement shall not be construed as a waiver of such right or novation of the terms established herein, unless expressly provided otherwise in this Agreement.

14.19. <u>Liability for Costs</u>. Unless expressly provided otherwise in this Agreement, each Party will bear its own costs and expenses incurred in negotiating, preparing and concluding this Agreement (including the respective fees of financial advisers, lawyers, auditors and/or any other consultants) and will be liable for the payment of their respective Taxes, and Sellers cannot, under any circumstances, transfer the burden of said costs, expenses and Taxes to Company. If any Transaction Cost or part thereof remains unpaid as of Closing, or is triggered, invoiced and/or payable after the Closing and has not, for any reason, been reflected in the Debt for the purposes of Working Capital and Debt Adjustment, Sellers must pay such Transaction Costs directly (through their own resources) or reimburse Company for these amounts within ten (10) Business Days from the date Buyer notifies Sellers of such payment.

14.20. <u>Corporate Reorganization</u>. No provision herein shall be interpreted or understood as an express or implicit limitation to the execution of the Corporate Reorganization or other transformation, combination, spin-off, merger, or share merger transactions, or any other manner of corporate reorganization or transaction involving any of Sellers, Oi and/or their respective assets, provided that such corporate reorganization or transaction does not affect, directly or indirectly, Company, the Infrastructure Items, and Company Assets.

14.21. <u>Specific Performance; Enforceable Instrument</u>. Each Party acknowledges that the remedies provided by law for non-compliance or threat of non-compliance with this Agreement may be inappropriate and, in acknowledging this fact, the Parties will have the right, without the need to provide security and in addition to any other remedies that may be available, to seek relief in the form of specific performance of their mandatory injunctions and/or obligations to pay, temporary restrictive measures, temporary or permanent remedy that may be available under the terms of articles 294 to 311, 497, *et. seq.*, of Brazilian Code of Civil Procedure. Specific performance will not, under any circumstances, be limited and/or impaired by any provision of this Agreement. Each Party further agrees that this Agreement, which is duly signed by two (2) witnesses, is an extrajudicial enforceable instrument for the purposes of Article 784, III, of the Brazilian Code of Civil Procedure.

14.22. <u>Governing Law</u>. This Agreement shall be governed by and interpreted in accordance with the laws of Brazil.

IN WITNESS WHEREOF, the Parties executed this Agreement in four (4) counterparts, in the presence of two (2) witnesses.

Rio de Janeiro, [•] [•], 20[•].

(*Remainder of this page intentionally left blank. Signature page to follow*)

# [SIGNATURES]

# SHARE PURCHASE AGREEMENT VIA ISOLATED PRODUCTION UNIT AND OTHER COVENANTS

# EXHIBIT A

# Existing Sharing Agreements

1. Infrastructure Sharing Agreement No. CO-UNA-009/2008, entered into on 07/31/2008, by and between, on one side, TNL PCS S.A., and, on the other side, Nextel Telecomunicações Ltda., as amended and modified from time to time;

2. Infrastructure Sharing Agreement No. CO/TELEMAR/UNA-IE/024-2015, entered into on 05/18/2005, by and between, on one side, Telemar Norte Leste S.A., and, on the other side, Nextel Telecomunicações Ltda., as amended and modified from time to time;

3. Infrastructure Sharing Agreement No. DNO.060/INFRASTRUCTURE/2006, entered into on 11/01/2006, by and between, on one side, OI S.A. — Under Judicial Reorganization, and, on the other side, Nextel Telecomunicações Ltda., as amended and modified from time to time;

4. Infrastructure Sharing Agreement No. DNO/117/INFRA/2003, entered into on 10/31/2003, by and between, on one side, OI S.A. — Under Judicial Reorganization and OI Móvel S.A. — Under Judicial Reorganization, and, on the other side, AMERICEL S.A. and BCP S.A., as amended and modified from time to time;

5. Infrastructure Sharing Agreement entered into on 01/31/2008, by and between, on one side, BCP S.A., and, on the other side, Amazônia Celular S.A., as amended and modified from time to time;

6. Infrastructure Sharing Agreement No. CO/TELEMAR/UNA-IE/0001-2005, entered into on 05/20/2005, by and between, on one side, Telemar Norte Leste S.A., and, on the other side, ATL Telecom S.A., AMERICEL S.A., TELET S.A., STEMAR Telecomunicações Ltda., TESS S.A., and BSE S.A., as amended and modified from time to time;

7. Letter of Intent of Mutual Infrastructure Assignment Agreement, entered into on 09/08/2003, by and between, on one side, TNL PCS S.A., and, on the other side, ATL — Algar Telecom S.A., STEMAR Telecomunicações Ltda., and BSE S.A., as amended and modified from time to time;

8. Mutual Infrastructure Assignment Agreement No. DNO.129 — INFRA/2004, entered into on 09/30/2004, by and between, on one side, OI S.A. — Under Judicial Reorganization, and OI Móvel S.A. — Under Judicial Reorganization, and, on the other side, Vivo S.A., as amended and modified from time to time;

9. Infrastructure Sharing Agreement No. CO/TELEMAR/UNA-IE/025-2006, entered into on 05/02/2006, by and between, on one side, Telemar Norte Leste S/A, and TNL PCS S.A., and, on the other side, Vivo S.A., as amended and modified from time to time;

10. Infrastructure Sharing Agreement No. CO/TELEMAR/UNA-IE — 0004-2004, entered into on 07/01/2004, by and between, on one side, Telemar Norte Leste S.A., and TNL PCS S.A., and, on the other side, Vivo S.A., as amended and modified from time to time;

11. Infrastructure Sharing Agreement entered into on 01/31/2007, by and between, on one side, Amazônia Celular S.A., and, on the other side, Vivo S/A., as amended and modified from time to time;

12. Infrastructure Sharing Agreement entered into on 06/16/2004, by and between, on one side, Telemar Norte Leste S.A., and, on the other side, Telefônica Empresas S.A., as amended and modified from time to time;

13. Mutual Infrastructure Assignment Agreement entered into on 12/01/2001, by and between, on one side, TNL PCS S.A., Telemar Norte Leste S.A., and Tele Norte Leste

Participações S.A., and, on the other side, Telemig Celular S.A., Amazônia Celular S.A. — Amazonas, Amazônia Celular S.A. — Pará, Amazônia Celular S.A. — Amapá, Amazônia Celular S.A. — Maranhão, Amazônia Celular S.A. — Roraima, as amended and modified from time to time;

14. Infrastructure Sharing Agreement No. CT/OI/RO — 181-2009, entered into on 11/25/2009, by and between, on one side, Tim Celular S.A., and TIM Nordeste S.A., and, on the other side TNL PCS S.A., and OI Móvel S.A. — Under Judicial Reorganization, as amended and modified from time to time;

15. Infrastructure Sharing Agreement No. CT/OI/RO — 185-009, entered into on 11/25/2009, by and between, on one side, Tim Celular S.A., and, on the other side, OI S.A. — Under Judicial Reorganization, as amended and modified from time to time;

16. Infrastructure Sharing Agreement No. CT/OI/RO — 183-2009, entered into on 11/25/2009, by and between, on one side, Tim Celular S.A., and TIM Nordeste S.A., and, on the other side, Telemar Norte Leste S.A., as amended and modified from time to time;

17. Infrastructure Sharing Agreement No. CO/TELEMAR/UNA-IE/016-2004, entered into on 03/28/2005, by and between, on one side, Empresa Brasileira de Telecomunicações S.A. — EMBRATEL, Vésper S.A., and Vésper São Paulo S.A., and, on the other side, Telemar Norte Leste S.A., and TNL PCS S.A., as amended and modified from time to time;

18. Infrastructure Sharing Agreement No. CO/TELEMAR/UNA-IE-0010-2006, entered into on 04/13/2006, by and between, on one side, Primesys Soluções Empresariais S.A., and, on the other side, Telemar Norte Leste S.A., and TNL PCS S.A., as amended and modified from time to time;

19. Mutual Infrastructure Assignment or Sharing Agreement No. DNO/138/INFRASTRUCTURE/2004, entered into on 06/01/2004, by and between, on one side, Empresa Brasileira de Telecomunicações - EMBRATEL, Vésper S.A., and Vésper São Paulo S.A., and, on the other side, OI S.A. - Under Judicial Reorganization, and OI Móvel S.A. - Under Judicial Reorganization, as amended and modified from time to time;

20. Infrastructure Sharing Agreement No. CO/OI/RO-024-2010 and CONT/CNO/0003/2010, entered into on 03/03/2010, by and between, on one side, OI S.A. - Under Judicial Reorganization, and, on the other side, CTBC Multimídia Data NET S.A., as amended and modified from time to time;

21. Colocation Agreement (Infrastructure Sharing) No. CO/TELEMAR/UNA-IE/0005-2004, entered into on 11/01/2004, by and between, on one side, Telemar Norte Leste S.A., and, on the other side, Companhia de Telecomunicações do Brasil Central, as amended and modified from time to time;

22. Colocation Agreement (Infrastructure Sharing Agreement) No. CO/TELEMAR/UNA-IE/0009-2004, entered into on 11/01/2004, by and between, on one side, Pegasus Telecom

S.A., and, on the other side, CTBC Celular Participações S.A., as amended and modified from time to time; and

23. Infrastructure Sharing Agreement No. CO/OI/DA - 219-2015, entered into on 10/26/2015, by and between, on one side, OI Móvel S.A. - Under Judicial Reorganization, and, on the other side, Algar Multimídia S.A., Algar Telecom S.A., and Algar Celular S.A., as amended and modified from time to time.

# SHARE PURCHASE AGREEMENT VIA ISOLATED PRODUCTION UNIT AND OTHER COVENANTS

EXHIBIT B

Infrastructure Items

# Keynels (KEYERED D) Terer Kans (NEVERSD D) Ein 10 Latinde (derinat) Langlad	Address ade (desime) Ped Uni Dity Address	Elserbor Elle Type - Indexe Berelson Type Height (m) WEA Evaluated (m*) WEA Contexted (m*) Benelsere Capacity (m*)	Verbad Brushers Maker Cascing/Jackaring Cascing/
1 000 002 0000 0000 0000 0000 -10 2 1078 00 00000 0000 00000 -10000 -10 1 1078 000000 00000 00000 -100000 -10	11/102 El Six Contactor Surface Section 2011 El USAN 21 andress     11/1028 K2 Bala Interview Annual Augusta de Long, 1029 - Francisco Langella de     11/1028 El Six Contactor Section 2011 El Six Contactor Section 2012 - Francisco Contactor Section	Odar         Genuese Ration         A3	Sci Davidsto         St.         Sci         Sci <t< th=""></t<>
4         22461         MO [INVECTOR: WEDERAL         EPVET201	14.4840 P Ean Parks An KOOP NTECH SAVED, 1999 / 11     14.4850 P Ean Parks An KOOP NTECH SAVED, 1999 / 11     15.1510     15 Finished An Statement Saved International Saved Internat	Index         Draphy         A,6         N.6         A,0         A,0           Index         Separt         A,0         R,0         R,0         R,0           Index         Separt         A,0         R,0         R,0         R,0	401 3940400 OK OK OK N01 407 Lawad Dependitioning 401 3950400 OK OK OK N02 407 Lawad Dependitioning 401 3950400 OK OK OK N02 407 Lawad WolfLawing
1         Science         Point (Decision)         Distribution         Distribution         -24           4         User / Doint         16x1 (Decision)         Decision31 (Decision)         -24         -44           5         User / Doint         16x1 (Decision)         Decision31 (Decision)         -21         -44           5         User / Doint         16x1 (Decision)         Decision31 (Decision)         -21         -44           5         User / Doint         Decision31 (Decision)         Decision31 (Decision)         -21         -24           11         User / Doint         Decision31 (Decision)         Decision31 (Decision)         -24           13         User / Doint         Decision31 (Decision)         -24         -24           14         User / Doint         Decision31 (Decision)         -24         -24	n vator or Garolina faci Prejinini Edva Sir Linearatine Dearbine 6.42420 DF Garolina faci Siria United Structure (and the Maria (Containe) 8.4310 H Konter Plant Structure (Containe) 8.4310 H Konter Para Graphs (Salad No. 6) 1.4445 (Maria Containe) 1.4445 (Maria Containe) 1.4455 (Maria Cont	page         statistic         3.4         3.3         3.4         3.4           bilar         disprit         3.4         3.3         3.4         3.4	No. constraint         vit.         O.         O.         O.         D.         Longet Listing           V01 202000         O.         O.         O.         S.         NO.         VO.         Listing         Dimensional Listing           V01 202000         O.         O.         O.         O.         S.         NO.         VO.         Listing         Dimensional Listing           V01 202000         O.         O.         O.         O.         Dimensional Listing         Dimensional Listing           V01 202000         O.         O.         O.         Dimensional Listing         Dimensional Listing         Dimensional Listing           V01 202000         O.         D.         D.         Dimensional Listing         Dimensional Listing
11         24/04         01/02/01 (02/04)         01/02/01 (02/04)         04/02/01         04/04/01	LEVIL 24 Evalue CEE 2 Level 11 Evaluate Descript     LEVIL 24 Evaluate CEE 2 Level 11 Evaluate Descript     LEVIL 24 Evaluate CEE 24 Eval	Desire         All         KL         All         KL           Delar         Desprey         All         KL         All         KL           Delar         Desprey         All         KL         All         KL         KL           Delar         Desprey         All         KL         KL         KL         KL	
14         4435         VHX (PAAC2611, 902493)         DPAC (Bit)         -2.22200         40           15         21798         DPAC (Bit)         2.64310(0021)         -300200         41           14         21928         BPAC (Bit)         0.64310(0021)         -300200         41           14         21928         BPAC (Bit)         0.64310(0021)         -3246300         43	CAULE OF Annuelian An Annuelian Annuelian Constantian Constan	Index Degring A/A R/A N/A N/A N/A Defer Path Belden A/A N/A A/A N/A Index Degring A/A N/A A/A N/A Index Degring A/A N/A A/A N/A	AVI 2015/0010         Cit         Cit         Cit         Linemat         Depresent Linemary           AVI 2015/0010         Cit
	1.2010     2     1.2020     10	Odar         Dustion         A1         A3         A3         A4         M0           Outer         Dustion         A1         A3         A3         M0         M0           Infar         Dustion         A3         A3         A3         M0         M0           Infar         Dustion         A3         A3         A3         M0         M0           Infar         Dustion         A3         A3         M0         M0         M0	All Biddelic         OC         All         Open and Lineary         Description           All Biddelic         OC         OC         OC         Diff         Lineary         Description           All Biddelic         OC         OC         Diff         Diff         Lineary         Description           All Biddelic         OC         OC         Diff         Diff         Lineary         Diff           All Biddelic         OC         OC         Diff         Diff         Lineary         Diff           All Biddelic         OC         OC         Diff         Diff         Lineary         Diff
20         4450         54.00 (Participation account)         Participation         -21.45.00         -44.           20         24.427         MM (FR 20.061)         40.00000         -22.44.00         -22.44.00         -22.44.00         -22.44.00         -22.44.00         -22.44.00         -22.44.00         -22.44.00         -22.44.00         -22.44.00         -22.44.00         -23.44.00         -23.44.00         -23.44.00         -43.0	21.2020 DP Ear Park     31.2020 DP Ear Park     41.2020 E1 Ear De Longing     41.2020 E1 Ea	John         Damina         S.S.         S.S.         S.G.         D.S.           John         Damina         S.S.         S.S.         S.G.         S.S.         S.S. <t< th=""><th>AVT INFORMD         OX         OX         OX         NOT         Listmail         Document Listmain           VD1 INFORMD         OX         OX         OX         SOT         VD1         Listmail         Document Listmain           VD1 INFORMD         OX         OX         OX         SOT         VD1         Listmail         Document Listmain           VD1 INFORMD         OX         OX         SOT         VD1         Listmail         Document Listmain</th></t<>	AVT INFORMD         OX         OX         OX         NOT         Listmail         Document Listmain           VD1 INFORMD         OX         OX         OX         SOT         VD1         Listmail         Document Listmain           VD1 INFORMD         OX         OX         OX         SOT         VD1         Listmail         Document Listmain           VD1 INFORMD         OX         OX         SOT         VD1         Listmail         Document Listmain
25 (11) 753 (799)203 (2004) 799(20) 21100 - 4 25 (11) 753 (799)203 (2004) 799(20) - 111720 - 4 25 (11) 753 (2004)2004 (2004) 791(20) - 111720 - 4 26 (11) 753 (2004)2004 (2004) 791(20) - 111720 - 4 26 (11) 753 (2004)2004 (2004) 791(20) - 111720 - 4 27 (11) 753 (2004)2004 (2004) 791(20) - 111720 - 4 28 (11) 753 (2004)2004 (2004) 791(20) - 111720 - 4 29 (11) 753 (2004)2004 (2004) 791(20) - 111720 - 4 20 (11) 753 (2004)2004 (2004) 791(20) - 111720 - 4 20 (11) 753 (2004)2004 (2004) 791(20) - 111720 - 4 20 (11) 753 (2004)2004 (2004) 791(2004) - 111720 - 4 20 (11) 753 (2004)2004 (2004) 791(2004) - 111720 - 4 20 (11) 753 (2004)2004 (2004) 791(2004) - 111720 - 4 20 (11) 753 (2004)2004 (2004) 791(2004) - 111720 - 4 20 (11) 753 (2004)2004 (2004) 791(2004) - 111720 - 4 20 (11) 753 (2004)2004 (2004) - 111720 - 4 20 (11) 753 (2004) - 111720 - 111720 - 4	1990 P for Pair     1990     1990     1990     1990     1990     1990     1990     1990     1990     1990     199	Inter         Desting         A5         A5         A4         A5           Inter         Desting         A5         A5         A4         A5           Inter         Desting         A5         B3         A4         A5           Inter         Desting         A5         B3         A4         A5           Inter         Desting         A5         B3         A4         B5	M2 (2000)         M2         M2         M2         M3         Main Management           M2 (2000)         M2         <
28 21451 CONDUCT BOARD 29 21458 CALLED BOARD CONTRACT BOARD	H 2015s 20 General day fan Verde Gal 2001101     Karlow Star Verde Gal 2001101     Karlow Star Gal Star Star Star Star Star Star Star Star	Default         Au         Au         Au         Au           Default         Au         Au         Au         Au         Au	427 (2016)     427      427     42     4     42     42     42     42     42     4
11         1704         2418         (ACC)2418         04882 <th0< th=""><th>41.2183 El Bio Canacine Perge Ferenze, qu'e Canacine Enviropet     41.21832 EP Canacines     41.2183</th><th>Index         Paths Relating         A,K         N,K         A,U         A,U           Index         Observe         5.6         9.6         5.6</th><th>401 3940400 OK OK OK N01 407 Lawad Dependitionaria 401 3959400 OK OK OK N02 407 Liawad Dependitionaria 401 3959400 OK OK OK N02 407 Liawad Copenditionaria</th></th0<>	41.2183 El Bio Canacine Perge Ferenze, qu'e Canacine Enviropet     41.21832 EP Canacines     41.2183	Index         Paths Relating         A,K         N,K         A,U         A,U           Index         Observe         5.6         9.6         5.6	401 3940400 OK OK OK N01 407 Lawad Dependitionaria 401 3959400 OK OK OK N02 407 Liawad Dependitionaria 401 3959400 OK OK OK N02 407 Liawad Copenditionaria
16         2200         Php ("Resona works)         Providing         -211680         -41           16         2200         Php ("Resona works)         Photosital         -211680         -41           16         1000         Kot (us resonal scowna)         Bacrosona         -21000         -41           16         1000         Kot (us resonal scowna)         Bacrosona         -21000         -41           16         1000         Kot (us resonal scowna)         Bacrosona         -21000         -41           16         1000         Kot (us resonal scowna)         Bacrosona         -21000         -41	1. 2011 Pl. Leading. ECODA CLUB CARCUS 51     1. 2011 Pl. Leading. ECODA CLUB CARCUS 54     1. 2011 Pl. Leading. ECODA CLUB CARCUS 54     1. 2012 Pl. Leading. ECODA CLUB CARCUS 54     1. 2012 Pl. Leading. ECODA CLUB CARCUS 54     1. 2012 Pl.	Index         Description         A.S.         A.S.         A.G.         M.G.	ADD/04/10         DX         DX <thdx< th="">         DX         DX         &lt;</thdx<>
III         CHI         CHI (DWCHE)? WEXHLI         CHI (DWCHE)?	LE GELLE DE Las Pauls de Contraction Menuelles 1000 - Contractives Menuelles 1000 - Contractives Menuelles 1000 - Menuelles Menuelles 1000 - Menuelles 100	John         Distance         \$10         \$11         \$10         \$10           John         Company Existing         \$11         \$12         \$10         \$10         \$10           John         Company Existing         \$14         \$12         \$10         \$10         \$10           John         Description         \$16         \$12         \$10         \$10         \$10	V2 (0)(0)(0)         O/C         O/C <t< th=""></t<>
Fit         DARN         TPL2 [DERTERS WORKEL]         DEREMITION         -22.6810         -13.6810           42         FIL40         OPT (SPREGR)         SPREGR)         52.6810         -32.6810         -43.6910           42         FIL40         OPT (SPREGR)         SPREGR)         SPREGR)         -23.6810         -43.6910           42         FIL40         RCL (SPREGR)         SPREGR)         SPREGR)         -33.6910         -44.691	L.Se40 El Ein Cleannin de dynam Eanna, 2020 - Viele Paper, Chiande Hall     Eanyani de Paper, P	Johan         Own MA         All         NA         NA           Johan         Company Bullion         5.6         5.6         5.6         5.6           Johan         Company Bullion         5.6         5.6         5.6         5.6         5.6	407 304040     0X     0X     0X     0X     107     14mmat     10queent     1aming     407 304040     0X     0X     0X     0X     107     14mmat     10queent     1aming     104     104040     10
44 2420 4417571624 200742 004710101 - 21000 - 41 45 2420 702 7021418 7021 0042418 - 21000 - 41 46 11721 006 1000000 200740 00421 - 21000 - 41 46 11721 006 10000000 200740 004000 - 21000 - 41 46 11721 006 10000000 200740 - 21000 - 41 47 1172 - 21000 - 41	11.0020 El Bio de Lentre de Securito Characta 114 - Defense Pales (PAR)     11.0020 El Bio Restava de Lentre Marcin (Dalo 1 Garrante Mel Desarta     11.0112 El Conse de Garran de Carlos Marcin (Dalo 1 Garrante Mel Desarta     11.0112 El Conse de Garran de Carlos de Marcin (Dalo 1 Carlos de Securito Alexando Alex	Index         Adds         Add         Add         Add           Index         Descript         Add         Add         Add         Add	ACT DISPAND         DC         DC         DC         DC         Lineard         Counterd Linearies (Lineard           ACT DISPAND         DC
10         110         00110000000         0000000         011000000         -17           45         4720         90010000000         9000000         9000000         -211000         -47           55         7120         90010000000         9000000         9000000         -211000         -47	27.620 D Enrola Enrola Manage Period II - Enrola Dessina     46.620 D Enrola Enro	Mar         Databar         Mar         Mar         Mar         Mar           Mar         Databar         Mar         Mar         Mar         Mar         Mar           Mar         Databar         Mar         Mar         Mar         Mar         Mar           Mar         Databar         Mar         Mar         Mar         Mar         Mar	V2 (0)5040         OC
11         444         D-1 SPECIAL SOFIL         PRACES         PRACES         -1           12         476         MAX SPECIAL SOFIL         PRACES         -21,100         -1           13         444         CEL IPCTURE SOFIL         PRACES         -21,100         -4           13         444         CEL IPCTURE SOFIL         PRESAM         -21,100         -4	1.4440 P Trans do in line learn 100 - France Disention     1.42400 P Campion Reducts Com Nation 1, 11, 1- Calcula Diseptem     10, 12710 P Calculate do: Inst Disent Hashan, 201 - Cardina Calculate Diseptem     10, 12710 P Calculate do: Inst Disent Hashan, 201 - Cardina Calculate Disenting	Inter         Despiny         Ab         Ab         Ab         Ab           Inter         Despiny         Ab         Rb         Ab         Ab         Ab           Inter         Despiny         Ab         Rb         Ab         Ab         Ab	AVI INFORM         OK         OK         NOT         AVI Learned         Descreet         Learned           VD INFORMO         OK         OK         OK         NOT         AVI         Learned         Descreet         Learned           VD INFORMO         OK         OK         OK         NOT         AVI         Learned         Descreet         Learned         Descreet         Learned         Descreet         Learned         Descreet         Learned         Descreet         Learned         Descreet         Descreet         Learned         Descreet         Learned         Descreet         Learned         Descreet         Descreet         Learned         Learned         Descreet         Learned         Learned </th
312     3123     3123     312     31     312     31	H. 1128 PT Colda PL Coldary Colda	Index         Degring         A,B         N,B         A,B         B,D           Index         Degring         A,B         N,B         A,B         B,D         B,D           Index         Degring         A,B         N,B         B,D         B,D         B,D           Index         Degring         A,B         N,B         A,U         B,D           Index         Degring         A,B         N,B         A,U         B,D	All 305040         OK         OK         OK         NOT         627         Linnad         Depend Lensing           All 305040         OK         OK         OK         NOT         627         Linnad         Depend Lensing           All 305040         OK         OK         OK         NOT         627         Linnad         Depend Lensing           All 305040         OK         OK         OK         NOT         627         Linnad         Depend Lensing           All 305040         OK         OK         OK         NOT         Linnad         Depend Lensing           All 305040         OK         OK         OK         NOT         Linnad         Depend Lensing           All 305040         OK         OK         OK         NOT         Linnad         Depend Lensing
16         4480         DB         D	Lotto C Polator Line Lance Lanc	Mater         Mater         A.1         A.2	Registration         Sol         Sol <t< th=""></t<>
14.         11.1         7.         7.671 (1975,247)         995,3471         7.212 (1975)         -1.7           14.         11.0         -         7.671 (1975,247)         9.957,447         7.971 (1975)         7.014 (1975)	17 5010 07 Dar Serla, Pauli da, Paula da, 201 Index 100 Daria 17 5010 11 El El For Interim Interimentation Paula Paula Daria 17 5010 11 El El Contentino Interimentatione Paula Paula Daria 18 5010 07 Interimentatione Paula Paula Daria	Inter         James         M1         M3         M4         M4           Inter         Bapping         A.1         A.3         A.4         M4         M4           Inter         Bapping         A.1         A.3         A.4         M4         M4           Inter         Bapping         A.1         B.3         A.4         M4         M4	N2 00000         X<
Inc.         Mail (Ministry)         Mail (Ministry)         Mail (Ministry)         -4233700         -44           24         1798         THE (Ministry)         Ministry)         -2134700         -44           24         1798         THE (Ministry)         Ministry)         -2144800         -44           24         1141         TER (Ministry)         Ministry)         -2144800         -44           27         404         1141         TER (Ministry)         Ministry)         -2144800         -44           27         404         TER (Ministry)         Ministry)         -2144800         -44           26         1041         TER (Ministry)         Ministry)         -2144800         -44           26         1041         TER (Ministry)         Ministry)         -2144800         -44		Joint         July         Al         Al         Al         Mit           Joint         Salay         Sa         Sa         Sa         Mit           Joint         Dargery         Sa         Sa         Sa         Mit           Joint         Dargery         Sa         Sa         Sa         Mit           Joint         Dargery         Sa         Sa         Sa         Mit	No.         M.         M.         M.         M.         Linead         Depend Lensing           M21 050401         G
M         MILE         VICTOR         MARKET         Partner         MARKET	1.0042 D Ear Fail (y Co Court Joine, 50) (Refer Fair 2000), Surin in northin) 41.0042 D Ear Fail (y Court Joine, 100), Start (y Court Joine, 100) 10.0055 D Start (y Court Joine, 100) 10.0055 D Start (y Court Joine, 100)	Indus Conserv Adding Add Add Add Add Add Add Add Add Add Ad	ACT INVANU         OC
11         184         VMD 19 A0098         67 A0198         -21 A009         -41           12         184         VMD 19 A0098         62 A0198         -31 A009         -40           12         1849         A1 A0098         62 A0198         -31 A019         -40           12         1849         A1 A0198         100 A0198         -40 A0198         -41 A019         -40           12         1849         A1 A0198         -40 A0198         -40 A0198         -41 A019         -41           12         1849         A1 A0199         -40 A0198         -40 A0198         -41 A019         -41           12         1849         -40 A0198         -40 A0198         -40 A0198         -41 A019         -41           13         1849         -40 A0198         -40 A0198         -40 A0198         -41 A0198         -41           14         -40 A0198         -40 A0198         -40 A0198         -41 A0198         -41	Al BEOR DF Local Active Testing Testing Control (Control (Contro) (Con	Inter         Descent         AV         AV         AV         AV           Inter         Descent         AV         AV         AV         AV         AV           Inter         Descent         AV         AV         AV         AV         AV         AV           Inter         Descent         AV         AV         AV         AV         AV         AV           Inter         Descent         AV         AV         AV         AV         AV         AV	NT 305900 00 00 00 00 00 00 00 00 00 00 00 00
17         144         CDD (PCDD)         PACKA         PACKAC         145000         -11           17         144         CDD (PCDD)         PACKAC         PACKAC         -114500         -41           18         1000         COD (PCDD)         PACKAC         PACKAC         -114500         -44           17         44.44         LML (PACKAC)         PACKAC         -114500         -44	Altrice - Second Barbary Internet for Product States Function Altrice - An Event States Function Altrice - An Event States Function Altrice - An Annual States - Annual States Altrice - Annual -	Inter- inter         Logistical         A.3         A.3         A.4         A.4           Inter- inter         Description         A.5         A.5         A.4         A.4         A.4           Inter- inter         Description         A.5         A.5         A.4         A.4         A.4	Act action         Act         M <t< th=""></t<>
TE         4474         LDB (INCOME)         (INCOME)         (INCOME)         44418         31           75         3454         644         (INCOME)         (INCOME)         44418         31           76         3454         644         (INCOME)         (INCOME)         43410         32           76         3454         644         (INCOME)         64574300         43140         34           77         3454         644         (INCOME)         64574300         43140         34           78         3444         644         (INCOME)         64574300         43140         34           78         3444         644         (INCOME)         64574300         34140         34440           79         3444         644         14444         34140         34440	11 2010 Bit Butch School Constant School (2012) 11 23 2020 CF Fundament Annual Constant School (2012) 12 23 2020 CF Fundament Annual Constant School (2012) 2020 (2012) 2020 2021 CF Fundament Annual Constant School (2012) 2020 (2012) 2020	Index         Description         A.S.         B.S.         A.G.         B.G.           Index         Description         A.S.         B.S.         A.G.         B.G.         B.G.           Index         Description         A.S.         B.S.         A.G.         B.G.         B.G.           Index         Description         A.S.         B.S.         B.G.         B.G.         B.G.	ADI (AMARD)         GC         GC         GC         BO         BO/         Linear         Dependent linearies           ADI (AMARD)         GC         GC         GC         BO/         E-DI         Linearie         Dependent linearies           ADI (AMARD)         GC         GC         GC         BO/         E-DI         Linearies         Dependent linearies           ADI (AMARD)         GC         GC         GC         BO/         E-DI         Linearies         Dependent linearies           ADI (AMARD)         GC         GC         GC         BO/         E-DI         Linearies         Dependent linearies
Fit         Amm         Add (IP/CCORT)         IP/CCORT         -2-1/12/12         -44           Fit         Amm         Add (IP/CCORT)         IP/CCORT         -2-1/12/12         -44           Fit         IP/CCORT         IP/CCORT         -2-1/12/12         -44           Fit         IP/CCORT         IP/CCORT         -2-1/12/12         -44           Fit         IP/CCORT         IP/CCORT         IP/CCORT         -44           Fit	Income or Onten Eastern Franciscus (1) - Concer Rear Dispite Income File Dentes Eastern Franciscus (1) - Concer Rear Dispite 14.0114 Pl File	contr         control (1)         A         A         A         A           Infar         Dispersion         A.3         A.3         A.4         A           Infar         Dispersion         A.4         A.4         A         A           Infar         Alation for the state         A.4         A         A         A           Infar         Alation for the state         A.4         A         A         A           Infar         Alation for the state         A.4         A         A         A	No STATUDE UN. 06 06 06 007 007 Lamas Departed Lamang Not STATUDE UN. 06 06 06 007 007 Lamas Departed Lamang NOT STATUDE UN. 06 07 007 007 Lamas Departed Lamang NOT STATUDE UN. 07 07 07 007 007 007 007 007 007 NOT STATUDE UN. 07 07 07 007 007 007 007 007 007 007 0
BI         BIVE         INC (INCOMENT WEARING         BALEWARD         251 (200)         41           BI         4779 (7128)         FILL (INCOMENT WEARING         BALEWARD         251 (200)         41           BI         4779 (7128)         FILL (INCOMENT WEARING         BALEWARD         421 (200)         41           BI         255 (112)         FILL (INCOMENT WEARING         BALEWARD         421 (200)         41	ISTUTE IN A Second Particle Pa	Idar         Bupper         \$1         <	CT INVERT         Dis         No.         N
18         L114         Value (UPPERSIT)         DOMAIN         DPMENSIT         -23.1200         -41           48         22278         22091 (ALMODISM, UPDERAL)         ALMODISM         -3.5180         -45           50         50.21         20.8         20.9         Value (COURS)         -3.5180         -3.5           51         52.3         20.8         20.0         20.9         Value (COURS)         -3.5180         -3.5	II. 2010 07 Providencia functiona functio	Johan         Dominin         A.1         K.3         A.0         K.3           Johan         Dominin         A.0         RA         RA<	M01040000         OX         DX         NO         NOT         Listenal         Doutral Listenal           M01040000         OX         OX         OX         SO         SOT         Listenal         SOT Listenal         SO
%         #6778         229         \$1,2,6,6,7,7,8         \$1,2,6,7,7,7         \$2,2,4,7	13.2020 KJ Ku Carlon Back Structure Back Star Star Structure Back Star Star Star Star Star Star Star Star	Index         Company Adding         A.B.         N.B.         A.B.         N.B.           Index         Density         A.B.         N.B.         A.B.         N.B.         N.B.           Index         Density         A.B.         N.B.         S.B.         N.B.         N.B.           Index         Density         A.B.         N.B.         S.B.         N.B.         N.B.	K27 305040     OK
B         Link         Data (Section)         Description         2110/05         -44           46         Link         Matti (Section)         Section)         Section)         -44           47         Matti (Section)         Matti (Section)         Section)         -44	2010 D Geten E Literativ Review Romanne Domine     2010 D Denni     2	Later Concerning All RA All MA RA Inter Concerning All RA RA All RA Inter Anti-Index All RA RA All RA	Virginities         Oct.         Oct.         Virginities         Oct.         Virginities         Oct.         Oct.         Virginities         Oct.
56         L216         VADO DPM/DBML WOMPALL         DPM/DBML         -2211600         41           26         6784         VMD1 MPM/DBML WOMPALL         DPM/DDML         -211600         -41           261         6784         VMD1 MPM/DBML WOMPALL         DPM/DDML         -211600         -41           263         17266         GBML RESERVAND         SERVERING         -214600         -41	11.42160 DP Providents Products Dealer Transmiss     do: Manual Confect, 2420 - Productionation     44.9110 DP for Parks     Find Samou Transmiss Dealer (Database Of Prior Easterne)     44.9110 BP for Original Statement Parks     State Transmission, 244 - State Conference     State Transmission, 244 - State Conference	John         Dominis         S.S.         S.S.         S.S.         M.J.           John         Dominis         S.S.         S.S.         S.S.         S.S.         M.J.           John         Dominis         S.S.         S.S.         S.S.         S.S.         M.J.           John         Dominis         S.S.         S.S.         S.S.         M.J.         M.J.           John         Dominis         S.S.         S.S.         S.S.         M.J.         M.J.	AVT INFORMD         OX         OX         OX         NOT         Listmail         Desamel Listman           VD1 INFORMD         OX         OX         OX         VD1         VD1         Listmail         Desamel Listman           VD1 INFORMD         OX         OX         OX         VD1         VD1         Listmail         Desamel Listman           VD1 INFORMD         OX         OX         VD1         VD1         Listmail         Desamel Listman
Link         DOI:10.000014         DOI:00.00014         DOI:000144         DOI:000144 <thdoi:000144< th="">         DOI:0</thdoi:000144<>	12450 P Barbar Inter In Col. Present Prever Inter 100 - Barbar Domina 1.2010 E Barbar Domina 1.2010	Inform         Desting         A3         A3         B4         B4           Inform         Convertion Conference         A3         N3         A4         B4           Inform         Desting         A3         N3         A4         B4           Inform         Desting         A3         N3         A4         B4           Inform         Desting         A3         N3         A4         B4	ADD/04/10         DX         DX         DX         DX         DATE           ADD/04/10         DX
201         2448         dia Schedulty Total         50 (Schedulty Total)         50 (Schedulty Total)         -44           201         1174         Qui (Schedulty Total)         10 (Schedulty Total)         -44         -44           201         1174         Qui (Schedulty Total)         10 (Schedulty Total)         -44         -44           201         1174         Qui (Schedulty Total)         10 (Schedulty Total)         -44         -44           201         1001         900 (Schedulty Total)         90 (Schedulty Total)         -44	41 U121 IC Parameters     12 Parameters	Defair         Defair         All         NA         All         MA           Defair         Despiry         All         All         All         MA         MA           Defair         Despiry         All         All         All         MA         MA           Defair         Despiry         All         All         MA         MA         MA	V2 (0)(0)(0)         O/C         O/C <t< th=""></t<>
208         21701         8400 (MCLIMBR 20048)         MCLIMBR 2-0.44300         -49           209         18448         COLATICR-LID98 (MODREL)         EACPLAD98         -3.73331         -38           213         18338         VOL9 (MOLATE2) (SARE)         MOLATE2         -1646050         -43	11-2005 PE Contrate Generative Concernance Automatica Automat Automatica Automatica Auto	Johan         Domping         A/A         NA         NA         NA           Johan         Domping         A/A         NA         A/A         A/A         A/A           Johan         Domping         A/A         A/A         A/A         A/A         A/A	401 3940400 OK OK OK M01 407 Lawad Daward Lawaig 401 3950400 OK OK OK 501 407 Lawad M01 award 401 3950400 OK OK OK 507 407 Lawad M01 award Lawaig
111         121         024         0294711         0294712         0294712         -211620         -41           113         1209         557 (527-52)         557 (527-52)         557 (527-52)         -41 <th>10.0004 D Sar Parks A. Security Perks 199     10.0004 D Sar Parks     10.</th> <th>Index         Description         A.S.         A.S.         A.G.         M.G.           Index         Description         A.S.         A.S.         A.G.         M.G.</th> <th>ADD 20100         DX         <thdx< th="">         DX         DX         &lt;</thdx<></th>	10.0004 D Sar Parks A. Security Perks 199     10.0004 D Sar Parks     10.	Index         Description         A.S.         A.S.         A.G.         M.G.	ADD 20100         DX         DX <thdx< th="">         DX         DX         &lt;</thdx<>
11         34.0         80.0         0000001         0000001         01000001         01000001         01000001         0100000001         0100000001         0100000001         0100000001         0100000001         0100000001         0100000001         0100000001         0100000001         0100000001         0100000001         0100000001         0100000001         0100000001         0100000001         0100000001         0100000001         0100000001         01000000001         01000000001         0100000000000000000000000000000000000	41 (21) (2) Canadran Jonetin Loudine Kalinchet in Cheries 1047 Dessein Terresorver     41 (21) (2) Canadra Canadr	Defair         Defair         All         NA         MA         MA           Defair         Despiry         A.1         A.4         A.4 <td< th=""><th>427 (2016)     427      427     42     4     42     42     42     42     42     4</th></td<>	427 (2016)     427      427     42     4     42     42     42     42     42     4
141         156         TPR (\$155,514,400,916)         455,800,524         -42,155         -43,155	11.2020 El Esc De lancies Bar Banks de Sie Fonniene, 28 - Donnies Bankaren Hollmanien artiste Janateres     20.2030 ES Via Velag Bankrag de La Sec Dirit Uter Velage Velage Velage     40.2032 ES Via Velag Anne Velage Velage Velage Velage Velage     40.2032 ES Via Velage Anne Velage Velage Velage Velage     40.2032 ES Via Velage Velage Velage Velage Velage Velage     40.2032 ES Via Velage Velage Velage Velage Velage Velage Velage     40.2032 ES Via Velage     40.2032 ES Via Velage Vel	Infar         Duration         5.5         5.5         5.6         5.6         5.6           Infar         Duration         5.6         5.8         5.6         5.6         5.6           Infar         Duration         5.8         5.8         5.4         5.6         5.6           Infar         Duration         5.8         5.3         5.4         5.6         5.6           Infar         Duration         5.8         5.3         5.4         5.6         5.6	AVI 305000         OK         OK         OK         Diff         Linnal         Descriptioning           AVI 305000         OK         OK         OK         Diff         Linnal         Descriptioning           AVI 305000         OK         OK         OK         Diff         Linnal         Descriptioning           AVI 305000         OK         OK         Diff         Linnal         Descriptioning           AVI 305000         OK         OK         Diff         Diff         Linnal         Descriptioning
D2         Bit III         VPL (VPL CONT NOTION)         PPL CONT         -41           D2         Bit III         FE CONTINUE         PPL CONTINUE         -41         -41           D2         Bit III         FE CONTINUE         PPL CONTINUE         -41         -41         -41           D2         Bit III         FE CONTINUE         PPL CONTINUE         PPL CONTINUE         -41         -41           D2         Bit III         FE CONTINUE         PPL CONTINUE         PPL CONTINUE         -41         -41           D3         EXPECTION         PPL CONTINUE         PPL CONTINUE         -41         -41         -41	10.2127 P Gr Paris     10.2127 P Gr Paris     10.2027 P Grant Par	Odar         Dustion         A1         A3         A3         A4         M0           Outer         Dustion         A1         A3         A3         A4         M0           Infar         Dustion         A1         A3         A3         A4         M0           Infar         Dustion         A3         A3         A4         M0           Infar         Dustion         A3         A3         A4         M0	All Biddelic         OC         OC         OC         All         Linnal         Description           All Biddelic         OC         OC         OC         Diff         Diff         Linnal         Description           All Biddelic         OC         OC         OC         Diff         Linnal         Description           All Biddelic         OC         OC         Diff         Linnal         Description           All Biddelic         OC         OC         Diff         Diff         Linnal         Description           All Biddelic         OC         OC         Diff         Diff         Linnal         Description
121         H78         DHI SERVICI SUBJELI         DPECIDI         -26.2740         460           121         64.0         Addit HPMARKE SUDJELI         DPMARKET         -26.2740         460           121         64.0         Addit HPMARKE SUDJELI         DPMARKET         -26.12420         460           127         5122         FOR         SUDJELIS         DPMARKET         -25.15420         -440	22.2135 (2) Personalamin Jac Libbris Genera 601 - Donaine Personalismin 44.3056 (2) Far Paulo A. Carlinge, mainte est e fine Calls of the Contro Lana 44.1116 (2) Far Paulo (2) For Paul	Johan         Distance         Dist	927 19450400 OK OK OK MOT VOT Liamad Dawared Liamina 927 19450400 OK OK OK 920 VOT Liamad Dawared Liamina 928 19450400 OK OK OK 057 VOT Liamad Dawared Liamina
Ldi         Ldi         Part (Party Dire 2004)         Part (Party Dire 2004)         Party Dire 2004         -40           Ldi         Ldi         Vel (Party Dire 2004)         Party Dire 2004         -40         -40           Ldi         Ldi         Vel (Party Dire 2004)         Party Dire 2004         -40         -40           Ldi         Ldi         Vel (Party Dire 2004)         Party Dire 2004         -40         -40           Ldi         Ldi         Ldi         Ref Control (Dire 2004)         Barcologie (Dire 2004)         -40           Ldi         Ldi         Ldi (Dire 2004)         Ldi (Dire 2004)         -40         -40           Ldi         Ldi (Dire 2004)         Ldi (Dire 2004)         Ldi (Dire 2004)         -40         -40           Ldi         Ldi (Dire 2004)         Ldi (Dire 2004)         Ldi (Dire 2004)         -40         -40           Ldi         Ldi (Dire 2004)         Ldi (Dire 2004)         Ldi (Dire 2004)         -40         -40           Ldi         Ldi (Dire 2004)         Ldi (Dire 2004)         Ldi (Dire 2004)         -40         -40           Ldi (Dire 2004)         Ldi (Dire 2004)         Ldi (Dire 2004)         -40         -40         -40	14.4445     19     20	Index         Description         A.S.         A.S.         A.G.         M.G.           Index         Description         A.G.         N.S.         M.G.         M.G.           Index         Description         A.G.         N.S.         M.G.         M.G.           Index         Description         A.G.         N.S.         M.G.         M.G.           Index         Description         A.G.         N.G.         M.G.         M.G.	AD 2015010         DX         DX         DX         DATE         Lineard         Descent (instance           AD 2015010         DX
112         114         201 (PLC14) #20144         (PLC14)         -11200         -11           131         647         1127 (PAPCHE) #20144         (Public)         -11200         -41           132         647         1127 (PAPCHE) #20144         (Public)         -11200         -41	11 (617)     19     5 (as loss Des Carross     10 (11)     19     5 (as loss Des Carross     10 (11)     10 (	Defair         Defair         All         NA         NA         NA           Defair         Despiry         A.1         NA         NA         NA         NA           Defair         Despiry         A.1         NA         NA         NA         NA           Defair         Despiry         A.1         NA         NA         NA         NA	42 (2004)     42 (2004)     42 (2004)     42 (2004)     42 (2004)     42 (2004)     42 (2004)     42 (2004)     42 (2004)     42 (2004)     42 (2004)     42 (2004)     44 (2004)
LH         LH         PHD (MAXYADD WARWAR)         DERADATION         -4.1           LH         BADIE         MOT MADDED         DERADATION         -4.1           LH         BADIE         MOT MADDED         DERADATION         -3.1           LH         BADIE         CONT MADDED         DERADATION         -3.1	H.1.2010         Bit Dis laurin         Bas Neuroph de Uis Vienies, D.1.: Ourpring de Dans           41.2010         Bit         Bit Dis Laurin         In Display de Dans           41.2010         Bit         Bit Internet         In Display de Dans           41.2010         Bit         Bit Internet         Averdale Instantion Carlos Ling, 2011.           41.2010         Bit         Bit Internet         Averdale Instantion Carlos Ling, 2011.           41.2010         Bit         Bit Internet         Averdale Instantion Carlos Ling, 2011.           41.2010         Bit         Bit Internet         Averdale Instantion Carlos Ling, 2011.           41.2010         Bit         Bit Internet         Averdale Instantion Carlos Ling, 2011.	Indur         Draphing         A,b         NA         NA         NA           Indur         Draphing         A,b         SA         SA         SA         SA	401 304040 OK OK OK 601 M07 Liamad Dependitionary 401 304040 OK 0K 0K 400 M07 Liamad Dependitionary 401 304040 OK 0K 0K 400 M07 Liamad Not Depend
1.0         1.00         P410 (Party New Science)         Party (Party New Science)         -40           1.01         1.01         CL1 (Party New Science)         Party (Party New Science)         -40           1.01         0.01         CL1 (Party New Science)         Party (Party New Science)         -40           1.01         0.01         CL1 (Party New Science)         Party (Party New Science)         -40           1.01         0.01         CL1 (Party New Science)         Party (Party New Science)         -40           1.01         0.01         Party (Party New Science)         Party (Party New Science)         -40           1.01         CL1 (Party New Science)         Party (Party New Science)         -40         -40           1.01         CL1 (Party New Science)         Party (Party New Science)         -40         -40         -40           1.01         CL1 (Party New Science)         Party (Party New Science)         Party (Party New Science)         -40           1.01         CL1 (Party New Science)         Party (Party New Science)         -40         -40           1.01         CL1 (Party New Science)         Party (Party New Science)         Party (Party New Science)         -40           1.01         CL1 (Party New Science)         Party (Party New Science)         -40 <td>41.92718 INS Enclosures Annual A</td> <td>Description         Description         <thdescription< th=""> <thdescription< th=""></thdescription<></thdescription<></td> <td>ADJ 2012400         DX         DX         DX         DATE         Lineard         Descent Strategy           ADJ 2012400         DX         DX</td>	41.92718 INS Enclosures Annual A	Description         Description <thdescription< th=""> <thdescription< th=""></thdescription<></thdescription<>	ADJ 2012400         DX         DX         DX         DATE         Lineard         Descent Strategy           ADJ 2012400         DX
121         6487 / 2480         1501 (Participation)         Strategy (Participation)         2140000         -1           121         1440         doc) (Distance and and an analysis)         Strategy (Participation)         214000         -4           124         1440         Maximum (Participation)         Strategy (Participation)         -4           124         1440         Maximum (Participation)         Strategy (Participation)         -4	Bill diele         Prof.         San Paris         An Discourses, 1031 - Simulation Interaction           28 diele         C4         Frankeise         An Discourses, 1031 - Simulation Interaction           28 diele         C4         Frankeise         An Discourses, 1031 - Simulation Interaction           28 diele         C4         Frankeise         An Discourses, 1031 - Simulation Interaction           28 diele         C4         Frankeise         An Discourses, 1031 - Simulation Interaction           28 diele         C4         Frankeise         An Discourses, 1031 - Simulation Interaction           28 diele         C4         Frankeise         An Discourses, 1031 - Simulation Interaction           28 diele         C4         Frankeise         An Discourses, 1031 - Simulation           28 diele         C4         Frankeise         An Discourses, 1031 - Simulation           28 diele         C4         Frankeise         An Discourses, 1031 - Simulation           28 diese         C4         Frankeise         An Discourses, 1031 - Simulation           28 diese         C4         Frankeise         An Discourses, 1031 - Simulation           28 diese         C4         Frankeise         An Discourses, 1031 - Simulation	Data         Data         Diff         Diff         Diff           Latar         Data         A1         A1         A1         A1           Latar         Data         A1         A1         A1         A1	V2 (00000)         OC
Lot         Lot         Description         Description         -46           Lot         248         248         Description         -46         -46           Lot         248         248         Description         -46         -46           Lot         248         Description         Description         -46         -46           Lot         Description         Description         Description         -46         -46         -46           Lot         Description         Description         Description         -46         -46         -46         -46           Lot         Desc	Distance         Description	John         Damins         Ma         Ma         Ma           John         Damins         Ma         Ma         Ma         Ma           John         Damins         Ma         Ma         Ma         Ma           John         Damins         Ma         Ma         Ma         Ma	AVE INFORM         OK         OK         MO         HOT         Lowered         Descend         Descend <thdescend< th="">         Descend         Descend</thdescend<>
101         102         241         102         103         102         103 <td><ul> <li>Control or Fair Park FF (2017) District Information</li> <li>Control of Control of</li></ul></td> <td>Differ         Director         All         All         NA         MA         MA           Infer         Director         All         All         All         MA         MA</td> <td>No. 000000         Vit.         OC         OC         OC         Literal         Literal         Dimension Literation           NO. 000000         OC         OC         OC         OC         OC         OC         Dimension Literation           NO. 0000000         OC         OC         OC         OC         OC         Dimension Literation           NO. 0000000         OC         OC         OC         OC         Dimension Literation           NO. 00000000         OC         OC         OC         Dimension Literation         Dimension Literation           NO. 000000000000000000000000000000000000</td>	<ul> <li>Control or Fair Park FF (2017) District Information</li> <li>Control of Control of</li></ul>	Differ         Director         All         All         NA         MA         MA           Infer         Director         All         All         All         MA	No. 000000         Vit.         OC         OC         OC         Literal         Literal         Dimension Literation           NO. 000000         OC         OC         OC         OC         OC         OC         Dimension Literation           NO. 0000000         OC         OC         OC         OC         OC         Dimension Literation           NO. 0000000         OC         OC         OC         OC         Dimension Literation           NO. 00000000         OC         OC         OC         Dimension Literation         Dimension Literation           NO. 000000000000000000000000000000000000
III         All (Parting 2004)         Parting         Status         <	14.73188 DF Eas Factor Arc Gaussian Danielli, 1814 - Despite glanders Ball     2014 DF Gaussian Ball     2012 DF Gaussian Despite and	Johan         Dought         A.S.         A.S.         N.S.	ACI INVAND         GK         GK         ACI         ACI         Linear         Linear <thlinear< th="">         Linear         Linear</thlinear<>
Link         Yukh Myuki (200 proprint)         Proprint)         -2114000         -44           101         dista         CPU (Proprint)         Proprint)         -2114000         -44           101         dista         CPU (Proprint)         Proprint)         -2114000         -44           101         dista         CPU (Proprint)         Proprint)         -2114000         -44         -44           101         dista         CPU (Proprint)         Proprint)         -444         -44         -44	na manan ar Ban Pada da Arizandona, 1931 - Dominin Leith Andrea 14.4845 UP Ban Pada da Arizandona, 1931 - Dominin Leith Arizandona, 14.4845 UP Ban Pada da Arizando da Arizand	page         statistic         3.4         3.3         3.4         3.4           balar         Daniety         3.4         3.3         3.4         3.4	No. constraint         vit.         OS         OS         OS         OS         Diff         Leaned         Descriptions           V01 2012040         OS         OS         OS         OS         OS         Diff         Leaned         Descriptions         Diff         Leaned         Dispatch         Diff         Leaned         Dispatch         Dispatch         Dispatch         Diff         Leaned         Dispatch         Dispatc
III         IIII         IIIII         IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII	21 USD DP Ear Parls     5. Consulty: Cream, 12 - Disputs Field Endorser Found     5. 2010 DP Ear Parls     5. 2010 DP Ear Parls     5. Consulty: Consult Vision Engine Media Endorser     5. 2010 DP Ear Parls     5. Consults: Consult Vision Engine Media Endorse     5. Consults: Consult Vision Engine Media Endorse	Inter- Inter- Inter- Inter-         No.         No.         No.         No.           Inter- Inter-         Despiny         A.0         A.0         No.         No.         No.           Inter-         Despiny         A.0         A.0         A.0         No.         No.	ADI INFORME         OC
HO         GAD         CONTINUE WOOPEN         DPWIADIM         -0141200         -48           HO         HA         HO         1990 D014         DPMIADIM         -2144100         -48           HO         DED         1990 D014         DPMIADIM         DPMIADIM         -2144100         -48           HO         DPMIA         DPMIADIM         DPMIADIM         DPMIADIM         -2144100         -48           HO         DPMIADIM         DPMIADIM         DPMIADIM         DPMIADIM         -2144100         -48           HO         DPMIADIM         DPMIADIM         DPMIADIM         DPMIADIM         -2144100         -48           HO         DPMIADIM         DPMIADIM         DPMIADIM         DPMIADIM         -2144100         -48	14.5533 07 San Paulu B. Geningun Agrain, 14. Straphop Steel Straph 24.5553 07 Kan Januarin Da Camar Pranc Sanad Schwiss, 201 Control Medical 24.5556 PR. Contra Control Control Schwiss Research Medical Controls Schwiss Research Schwiss, 14.55	Inter         Despite         A/L         B/L         B/L           Inder         Despite         A/L         B/L         B/L           Inder         Despite         A/L         B/L         B/L           Inder         Despite         A/L         B/L         B/L	MD (and bit)         DX         DX         DX         E01         E01         Example         Deprese Example           MD (and bit)         DX         DX </th
HA         HOM         RADA WILL ADARTS         HAL CANTS         -14228         -23           264         1998         AADI SACCESSI BODYAL         MACCANTS         -24,0780         -33           264         1998         CANT WALL         BODYAL         BODYAL         -24,0780         -43           264         1997         CONT WALL         BODYAL         PATABOL         -34           264         1997         THE MARCH BODYAL         PATABOL         -34           264         1997         THE MARCH BODYAL         PATABOL         -34	General and Note: Journal Learning Constraint Stream and States and Stat	Desire         Static         NA	No. constraint         vit.         OK         OK         MO         Internet         Learned         Descend Linearia           M21 2019040         OK         OK         OK         MO         Horizonta         Depared Linearia           M21 2019040         OK         OK         OK         MO         Horizonta         Depared Lineary           M21 2019040         OK         OK         OK         MO         Depared Lineary
Set         Col: (specific)         Specific)         Specific) <t< th=""><th>24 00740 DP Ear Parls De Heardparls 104 - Descent Paris Heardparls 24 0011 DP Earthan De Heardparls 104 - Descent Paris Heardparls 24 0011 DP Earthan E. Street &amp; Russ. 105 - Descent Paris Heardparls United Descent Paristal</th><th>Index         Descript         A.0         B.3         B.4         B.4           Index         Descript         A.0         B.3         B.4         B.4           Index         Descript         A.0         B.3         B.4         B.4           Index         Descript         A.0         B.3         B.4         B.4</th><th>ADI BASING         Col         Col         ADI         ADI         ADI         ADIMAC         Constant         Constant</th></t<>	24 00740 DP Ear Parls De Heardparls 104 - Descent Paris Heardparls 24 0011 DP Earthan De Heardparls 104 - Descent Paris Heardparls 24 0011 DP Earthan E. Street & Russ. 105 - Descent Paris Heardparls United Descent Paristal	Index         Descript         A.0         B.3         B.4         B.4	ADI BASING         Col         Col         ADI         ADI         ADI         ADIMAC         Constant
173         1870         -         1874         -         1874         - <t< th=""><th>11.51156 and Inter-Prevent Location in Contern, MOL - Double Pair Second 31.5124 84 Solvator Bill CONE, Specifical APRE 185700 21.6500 97 Particular Processing Conternation     21.51251</th><th>Inform         Journey         5.01         5.31         M1         221           Inform         Density         5.01         5.31         5.01         201           Inform         Density         5.01         5.31         5.01         3.01           Inform         Density         5.31         5.31         5.01         5.01</th><th>NE (2004)         X</th></t<>	11.51156 and Inter-Prevent Location in Contern, MOL - Double Pair Second 31.5124 84 Solvator Bill CONE, Specifical APRE 185700 21.6500 97 Particular Processing Conternation     21.51251	Inform         Journey         5.01         5.31         M1         221           Inform         Density         5.01         5.31         5.01         201           Inform         Density         5.01         5.31         5.01         3.01           Inform         Density         5.31         5.31         5.01         5.01	NE (2004)         X
10         come         201 (PAMMG ROUPA)         PAMMG         -211200         -11           15         5202         PR (PAMMG ROUPA)         DEGISSO         -211200         -11           16         700         PR (PAMMG ROUPA)         DEGISSO         -211200         -11           17         4070         ESE (PAMMG ROUPA)         DEGISSO         -211200         -11           17         4070         ESE (PAMMG ROUPA)         DEGISSO         -211200         -41           17         4070         ESE (PAMMG ROUPA)         DEGISSO         -211200         -41           17         4070         ESE (PAMMG ROUPA)         DEGISSO         -411400         -41	Constant or Constant Sea Config. 2011 Constant Para Lippice     Constant Sea Config. 2011 Constant Para Lippice     Constant Para Constant Para Lippice Para Lippice     Constant Para Lippice	prime         prime         prime         All         All         MA         MA           prime         prim         prim         prim	No. District         Vit         OC         OC         Oc         District         Lansed         Command Longituding           VIT 305240         OC         OC         OC         OC         OC         OC         District         Distrind         District         District
D1         3434         5479 (204534) (20304)         45945347         457450         443           24         2014         255 (204534) (20304)         2004340         2004340         41           24         2411         326 (2045354) (203044)         2004340         2411/21         41	LA TALLE ED Auto Volue E. Ta Realizer Designing Fairs Volue     La Ta Realizer Designing Fairs Volue     La Taulor E Example E Example To Taulore     La Taulor E Example To Taulore     La Taulor E Example To Taulore	Industry         Damping         A.S.         R.S.         R.S.         R.S.           Industry         Damping         A.S.         R.S.         R.S.         R.S.         R.S.           Industry         Damping         A.S.         R.S.         R.S.         R.S.         R.S.	ACI ANSAND         GX         GX         ACI         Linear         Linear         Depend Lineary           ACI ANSAND         GX         GX         GX         GX         GX         Linear         Depend Lineary           ACI ANSAND         GX         GX         GX         GX         Linear         Depend Lineary           ACI ANSAND         GX         GX         GX         Linear         Depend Lineary           ACI ANSAND         GX         GX         GX         Linear         Depend Lineary
H2         H20         Control (Control (Contro) (Control (Contro) (Control (Contro) (Control (Co	14 2010 HA General Article Art	Inter         James         MI         All         MI         Till           Inter         Bayers         All         All         MI         Till         Till           Inter         Bayers         All         All         MI         Till         Till           Inter         Bayers         All         Bill         MI         Till         Till	NJ 89990         X         S         S         S         Lift         Listed         Serrer21 series           NJ 89990         X         K         K         S         S         S         Lift         Listed         Serrer21 series           NJ 89990         X         K         K         S         S         Lift         Listed         Serrer21 series           NJ 89990         X         K         K         S         Lift         Listed         Serrer21 series           NJ 89990         X         K         S         Lift         Listed         Serrer21 series
Active         Memory (Schwarzsky, Schwarzky,	20111 NO for same 20111 NO for same 20111 Di Car Jana 20111 Di Car Jana	NA         NA         NA         NA           Infanc         Despite         X.0         N.3         X.0         NA	
List         July (Relation, ECONE)         Addition         41044         -9           101         507         COT (SWERK)         ECONE         2004204         -31040         -4           101         507         COT (SWERK)         ECONE         2004204         -31040         -4           101         1041         Add (SWERK)         ECONE         2004204         -31040         -4           101         1041         Add (SWERK)         ECONE         2004204         -31040         -4	Latin Longitudi Mangalan III     Longitudi Approximite Magalan III     Longitudi     Longitudi II     Longitudi     Longi	Industry         Damping         A.S.         R.S.         R.S.         R.S.           Industry         Damping         A.S.         R.S.         R.S.         R.S.         R.S.           Industry         Damping         A.S.         R.S.         R.S.         R.S.         R.S.	ACI ANSAND         GX         GX         GX         GX         GX         Linear         Linear         Depend Lineary           ACI ANDAND         GX         GX         GX         GX         GX         Linear         Depend Lineary           ACI ANDAND         GX         GX         GX         GX         Linear         Depend Lineary           ACI ANDAND         GX         GX         GX         DEP         Linear         Depend Lineary           ACI ANDAND         GX         GX         GX         DEP         Linear         Depend Lineary
HO         2007         2007         Part Inscients         Provide         24,000         -8           201         2007         2007         Provide         2007	10 2010 P1 Contra Biol 2020 Discussion P1 Forestation P1 Education P2 P2 Contra Discuss Discus P1 Forestation P2 P2 Contra P2 P2 Contra Discus P2 P2 Contra P2 P2 Contra P2 P2 Contra P2	Inter         Descent         MI         R3         MI         R4           Inter         Descent         MI         R3         MI         R4           Inter         Descent         MI         R4         MI         R4           Inter         Descent         MI         R4         MI         R4           Inter         Descent         MI         R4         MI         R4	NY 89990         X         K         X         V/Z         L/T         L/trail         Semantic server           NY 89990         X         X         X         V/Z         L/T         L/trail         Neurophic server           NY 89990         X         X         X         V/Z         L/T         L/trail         Neurophic server           NY 89990         X         X         X         X/Z         L/T         L/trail         Neurophic server           NY 89990         X         X         X         X/Z         L/T         L/T         L/trail         Neurophic server
26         160         100         000         000         000         010	Alterna Statement Constantial Constantiana Constanti	Inter         Autor         Aut         Aut         Aut         Aut           Inter         Descent         Aut         Aut         Aut         Aut         Aut           Inter         Descent         Aut         Aut         Aut         Aut         Aut           Inter         Descent         Aut         Aut         Aut         Aut         Aut	All         Mail
DH         DHAL         TELEVISION         DMPL         SELECT	2 2020 11 Vera Annual Inna Gebrard de Managerie, 1121 - Complex Vera 24 2715 24 San Park & Salaria San Park 2020 11 Vera Annual San Park	Industry Description Add Add Add Add Add Add Add Add Add Ad	ACT APARTO         OC
Avv         tam.         Cold (PROCRE)         EPOCRE         COLD (Cold	A Average of Branches Average Strengther Str	contr         control         All         NA         NA         NA           Inform         Dispersion         A.M.         NA         NA         NA         NA           Inform         Dispersion         A.M.         NA         NA         NA         NA           Inform         Experimental         A.M.         NA         NA         NA         NA           Inform         Experimental         A.M.         NA         NA         NA         NA	No Partena V. 66 06 02 107 Longer Longer No Partena V. 66 06 02 107 Longer Longer No Partena V. 66 06 02 107 Longer Longer No Partena V. 66 06 06 107 Longer Longer Longer
201         2014         7100         2014/2011	M. 60174         D         Sectors         N. 6018         Sectors         Control           M. 60174         D*         Benefits         Max 220, CO 1, Li. 0         Sectors           M. 60105         D*         Sectors         ds. Charlin Extension, 1107         Sectors           M. 60105         D*         Sectors         ds. Charlin Extension, 1107         Sectors           M. 60106         D*         Sectors         ds. Charlin Extension, 1107         Sectors	Infar         Despity         \$1	CT 19/04/D         Dir.         Dir.         Dir.         Dir.         Literatil         Director Literative           X21 39/04/D         Dir.         Dir.         Dir.         Director Literative         Director Literative           X21 39/04/D         Dir.         Dir.         Director Literative         Director Literative         Director Literative           X21 39/04/D         Dir.         Director Literative         Director Literative         Director Literative           X21 39/04/D         Dir.         Director Literative         Director Literative         Director Literative
201         201 <th>13 2020 DF Loss a figure Density     25 202</th> <th>Inter         Low         All         Kit         Kit           Inter         Despiny         A.0         A.0         Kit         Kit           Inter         Despiny         A.0         A.0         Kit         Kit           Inter         Despiny         A.0         A.0         Kit         Kit</th> <th>ADI INFORME         Disc         <thdisc< th=""> <thdisc< th=""> <thdisc< th=""></thdisc<></thdisc<></thdisc<></th>	13 2020 DF Loss a figure Density     25 202	Inter         Low         All         Kit         Kit           Inter         Despiny         A.0         A.0         Kit         Kit           Inter         Despiny         A.0         A.0         Kit         Kit           Inter         Despiny         A.0         A.0         Kit         Kit	ADI INFORME         Disc         Disc <thdisc< th=""> <thdisc< th=""> <thdisc< th=""></thdisc<></thdisc<></thdisc<>
131         17963         AMM (MCTEMER, NCOPEA)         BMMCTEMER         -20           131         PH11         SAT AMAZONI         BMMCTEMER         -20           131         PH11         SAT AMAZONI         BMMCTEMER         -20           133         PH11         SAT AMAZONI         BMMCTEMER         -20           134         GE11         CE11 (PARRONILL SCATMA)         SAMMONTE         -20           135         GE11         CE11 (PARRONILL SCATMA)         SAMMONTE         -20	13.1186 El Ra De Foreire Base Con Harroll, (b. 15 debut de Balado 11.2187) HA Ran Lots Parla 14.2187 El Ran Lots De Carl Parla Con La Contra de Carl d	Index         Auto Bolley         AU         RA         RA         RA           Index         Auto Bolley         AU         RA         BA         RA           Index         Auto Bolley         AU         RA         BA         RA           Index         Toxet         AU         RA         BA         RA	NO 845640         O.K.         DS         DS         DST         Listed         Deprese Listening           Add 345640         DG         DS         DS         DST         Listening         Deprese Listening           Add 3456400         DG         DS         DS         DST         Listening         Deprese Listening           Add 3456400         DS         DS         DS         DS         DST         Listening         Deprese Listening
	n server av fan Parja Ballelle Mei Uarrienta Arrente Lander av de Arrente de Lander av Server 20.2005 Mei Uarrienta Arrente Lander Kenne de Arre, 1311 - Uarrienta Centre Desprey 27.2005 Mei Ballelle de Carrene de Lander av Server av Server av Server 28.2005 Mei Ballelle de Carrene de Arrente av Server av	come         come         A/A         A/A         B/A         B/A </th <th>No. 092000         Vit         Vit         Longet         Descriptions           Vit MV2000         O         O         O         Vit         Vit         Longet         Descriptions           Vit MV2000         O         O         O         O         Vit         Vit         Vit         Longet         Descriptions           Vit MV2000         O         O         O         O         Vit         Vit         Vit         Longet         Descriptions         Descriptions</th>	No. 092000         Vit         Vit         Longet         Descriptions           Vit MV2000         O         O         O         Vit         Vit         Longet         Descriptions           Vit MV2000         O         O         O         O         Vit         Vit         Vit         Longet         Descriptions           Vit MV2000         O         O         O         O         Vit         Vit         Vit         Longet         Descriptions
211         dise         Constitution statute         distance         222220         -10           221         7021         Ford Declarate statute         Statute         Statute         -102220         -0           221         7021         Ford Declarate statute         Statute         Statute         -102330         -0           221         523         Ford Declarate statute         Statute         -0         -0         -0	21 JUNE 81 Star Dar Leaster The Yames, Trees day Manadata - Val Boad Donaster M 4764 C4 Particles & Star Parts 42 JUNE 42 Star Parts 42 Star Parts	Inform         Description         A.0.         B.3.         B.0.         B.0. <thb.0.< th="">         B.0.         B.0.</thb.0.<>	ADI INFORMO         OC
101 Hell PHA (Discours worker Discours -1140010 -17	PAGE 27 Ends RATHER REPORT AND 1-204 COLD APRAN	Inter Addr. All All All All All	AN INVESTIGATION OF THE OWNER

|  |   |   | Geala   
  | ative.   |  |  
   | Address   
   |  | Moulture  |  
  |  | Сарастр   |  |  |  |  | General   
   | Adaptation   |  |  | anding<br>Bistori of the   
  |
--	---	---
--
--|---
--|---|---
--|---|--|--|--|--|---
--|--|--
---|
| 1 \$3564   | CDDS (CM30038322)   | NUT DE LO   | Latitude (decimal)<br>-22.93127778  
  | Longitude (decimal)<br>-63.37381667  | Fed. Del   | Ro De Janezo   
   | Address<br>Estrada do Guerenguê, %  
   | Site Type - Outdear<br>Greetfeid   | Structure Type PDST   | Height (m)   
  | 118A Seatsfield (41*)<br>12A4  | VEA Contracted (m <sup>4</sup> )  | Structure<br>Capacity (m*)<br>N/D                        | Vertical Moultane Maker  | NOx  | Closing/Driverzation   | Notes Notes   
   | Adaptations - Nite   | Adaptations -<br>Structure<br>POST<br>OVERIDADED   | Kieles of Kie<br>Licenses<br>(General)<br>FilliD   | NEA'S<br>Environmental<br>Diseases<br>NOT INFORMED   
  |
| 2 531174<br>3 531686   | FAD4 [CH(80060537]<br>CTGE [CH(8008627]   | SP8F03737<br>SPCA04938  | -23.46780556<br>-23.87822   
  | -66.70027778<br>-66.41487  | 59<br>59   | Sao Paulo<br>Cubatas   
   | entarda do babão, nº 416 - Preclesia do O<br>Rea jose gorçalves torres s.n cubatão(sp   
   | Greefed<br>Greefed   | SELF-SUPPORT.   | 43<br>50   
  | 3.54   | NOT INFORMED<br>NOT INFORMED  | ND<br>ND   | NOT INFORMED<br>NOT INFORMED   | NOK  | ox<br>ox   | NCK   
   | NIEDS EXTENDED JANETORIAL SERVICES<br>NIEDS EXTENDED JANETORIAL SERVICES   | REPLACE REST<br>REPLACE REST<br>RATIONN AND<br>USER NEW REP  | NOT SPORMED  | NOT DAFORMED<br>NOT  
  |
| 4 \$31712  | PR09 (CH10009900)   | R3L3R3002   | -22.036339  
  | -43.184782   | ю  | No De Janeso   
   | Rua Ábaru Chinese, 41 – Laratgenas  
   | Routhap  | MAST  | 5  
  | 2.603  | NOT INFORMED  | ND   | NOT INFORMED   | NOK  | NDK  | NDK   
   | NAST LADDER DAMAGED +<br>MERDS JANTONIAL SERVICES  | REAK OF<br>STRUCTURE<br>COLLAPSE DUE TO<br>PREAT FASTERIAL<br>PELAR BREAKING<br>OFF  | NOT INFORMED   | NOT INFORMED   
  |
| 5 532110<br>6 532542   | CVE1 (CM300001067)<br>PPL3 (CM30032063)   | SPUCV368  | -23.5025  
  | -46.66111111   | 10<br>10   | Sao Paulo<br>Sao Paulo   
   | RUA PROPEVEIMA VIERA, 21 - CARA VIERE<br>Av. Maerin, 81 - Planita Paulata, 550 Paula - 59   
   | Creatived<br>Routhap   | SELF-SUPPORT.   | 42<br>(2,2+2,3+2,2)  
  | 2.7<br>2.411   | NOT INFORMED  | ND<br>ND   | IVERST<br>NOT INFORMED   | OK<br>NOK  | OK<br>NOK  | OK<br>NOK   
   | MOT<br>EQUIPMENT CAEDATT OXIDATED +<br>NEEDS JAVETOKIAL SERVICES   | NOT<br>NEWFORCEMENT<br>OF THE ANCHORU  | NOT SPORMED  | NOT INFORMED   
  |
| 2 512572<br>8 512678<br>8 512729   | 2002 (CM 00020011)<br>FACO (CM 0002007081)<br>CAR1 (CM 0010206)   | SPWPCIH2<br>SPWPCIH2<br>SPWCRISH  | -23.5711<br>-23.48425<br>-21.47355556   
  | -46.47742<br>-46.77328889<br>-66.66231667  | 10<br>10<br>10<br>10   | 530 724/0<br>530 724/0<br>530 724/0  
   | HOR MANDEL DA NAVA SĂ 175<br>MIR BANDA Nº 521 - PRECENSIA DO Ô<br>Bas Pedro Rodrigues Brja, 111 - Vili Marier   
   | Granted<br>Granted<br>Granted  | SELF-SUPPORT.<br>SELF-SUPPORT.<br>SELF-SUPPORT.   | 40<br>40<br>60.5   
  | 2,832<br>3.67<br>2.74  | ACT IN COME<br>ACT IN COME<br>NOT INFORMED  | N0<br>N0<br>N0   | MOT INFORMED<br>MOT INFORMED<br>MOT INFORMED   | 08<br>1928<br>08   | 0X<br>0X<br>0X   | 88  
   | NOT<br>NEEDE EXTENDED AMPORTAL SERVICES<br>NOT   | MASES<br>MOT<br>NOT<br>NOT   | NOT DECAMID<br>NOT DECAMID<br>NOT DECAMID  | NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED   
  |
| 10 532850  | CBG6 [CM30656787]   | sewerce:  | -23.56755556  
  | -16.5175   | 59   | Sao Paulo  
   | RUA ANTONIO RUSI N 118 - OMĆANA BILINZIMO   
   | Greenheid  | SELF-SUPPORT.   | 40   
  | 4.854  | NOT INFORMED  | ND   | NOT INFORMED   | œ  | ox   | ак  
   | NOT  | ROPLAS STRUCTOR<br>ROPLAS LACON<br>REPLACE THE REST<br>RUTTORS, CON-<br>SECT THE LED-FROM  | NOT INFORMED   | NOT INFORMED   
  |
| 11 512993<br>12 513034   | PTA1 (CM30010027)<br>ACS1 [CM30000014]  | 5PW712996<br>5PWCR1695  | -21.4M15667<br>-23.6803   
  | -55.5277778  | g  | 530 F240<br>530 F240   
   | Rus Dr Ando Torroll STV – Jaeden Silo 2004<br>Rus HERREDO SANECLI 222 – MO PRADO  
   | Granting<br>Granting   | SELF-SUPPORT.   | 40<br>40   
  | 1447<br>3.64   | NOT INFORMED  | NO<br>NO   | MICCOM.<br>NOT INFORMED  | OK<br>NOK  | ox<br>OX   | OK<br>OK  
   | NEEDS EXTENDED JANFORDAL SERVICES  | NINE LIGHTES.  | NOT DECAMID  | NOT INFORMED<br>NOT INFORMED   
  |
| 13 511279  | 0E72 (CH30032609)   | SPWST3117   | -23.423728  
  | -46.698729   | 57   | Sao Paulo  
   | Av. Rope Peters Jime, 1999 - Chican Sarta Artists (2013 Sul), Ma<br>Paulo - SP  
   | Routhap  | MAST  | 10   
  | 2.068  | NOT INFORMED  | NO   | NOT INFORMED   | œ  | œ  | OK  
   | 801  | OVERLOADED.<br>REDFORCE<br>AND ORDE TO<br>ALL CONDETT<br>BASES.<br>LIPELINES OF<br>THE MASES<br>OVERAFIES  | LICENSARUE   | NOT  
  |
|  |   |   |   
  |  |  |  
   |   
   |  |   | | | | |
  |  |   |  |  |  |  |   
   |  | LIPELINES OF<br>THE MAST'S<br>OCEDATED.  |  |  
  |
| 14 511992<br>15 511541<br>16 511567  | 0400 (CM10010923)<br>0400 (CM100208551<br>0300 (CM100225517)  | SPIREMUTER<br>SPIRATIONE<br>SPIRATIONE  | -21.48441667<br>-21.4941<br>-21.7554  
  | -46.47972222<br>-46.5874<br>-46.705  | 97<br>197<br>197   | Sao Paulo<br>Sao Paulo<br>Sao Paulo  
   | RUA FIDRAVANTE LOPIS GARCIA ISIO<br>Rua Dicitar de Pedeson, 184, Vila Hadrivia<br>Rua DR. Antônio Santest de Canadio, 555 - 31, 330 Bertando - 550 Paulo - 57   
   | Greethed<br>Greethed<br>Greethed   | SELF-SUPPORT.<br>SELF-SUPPORT.<br>SELF-SUPPORT.   | 43<br>42<br>42   
  | 20.01<br>3.02<br>5.58  | NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED  | ND<br>12<br>ND   | NOT INFORMED<br>SECCOMM<br>MOT INFORMED  | NOK<br>OK<br>OK  | ox<br>ox   | OK<br>OK  
   | NEEDS EXTENDED JAATTORIAL SERVICES<br>NOT  | STATE OF<br>CREATION<br>NOT  | FILED<br>NOT RECOMMEND<br>NOT RECOMMEND  | NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED   
  |
| 17 57605<br>18 579525<br>19 57806<br>20 579717   | 5715 (CM40020809)<br>3004 (SRC8FLA232)<br>EFSR (CM40022582)<br>TNLD (SRC8FLA2198)   | \$200F040114<br>\$200F040114<br>\$200F040114  | -6.471125692<br>-2.721815<br>-2.721483<br>-1.7221483  
  | -25.54274509<br>-38.49388<br>-38.48396<br>-28.5381584  | a<br>a   | Teor<br>Portaleza<br>Portaleza<br>Portaleza  
   | Rus BRO VERS 4/0 - 360<br>Rus VCerlle Lette, 620 - Marvins.<br>Edifico San Lucio - Rus Prof. São Reeva, 420 Papicu<br>Rus PENTO MARRAN,752  
   | Routhing<br>Routhing<br>Routhing<br>Routhing   | NAAT<br>NAAT<br>NAAT<br>NAAT  | 30<br>(3+1+1,5)<br>2.2<br>4  
  | 9.50<br>35.1586<br>2.2568<br>NOT INFORMED  | NOT IMPORTED<br>NOT IMPORTED<br>NOT IMPORTED<br>NOT IMPORTED  | 15<br>NO<br>NO<br>NO                                     | MOT INFORMED<br>MOT INFORMED<br>MOT INFORMED<br>MOT INFORMED   | NX<br>NX<br>NX   | ox<br>ox   | NDK<br>NDK<br>OK<br>NDK   
   | NOT<br>PLATFORM/SIGD CODATED<br>MAST LADDER = L3DHTNING KCD DAMAGED<br>MAST LADDER DAMAGED   | OCULATED<br>MASTS<br>MOT   | HURD   | NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED   
  |
| 21 578784<br>22 581502<br>23 646902<br>24 647729<br>25 647941  | PCNC (198291.6032)<br>TPS (044023546)<br>PC0 (04603546)<br>MGRG (04604473)<br>PSDC (04604473)   | S2CEPLA0202<br>CEMILD-LLS<br>MODHOR 708<br>MOLINARI72<br>MOLINARI72   | -3.742146<br>-4.31239<br>-18.31668<br>-18.88414<br>-18.88414<br>-18.11226   
  | -28.511967<br>-39.00223<br>-43.61795<br>-47.82787<br>-44.5554  | 01<br>20<br>20<br>20<br>20   | Rotalica<br>Molengu<br>Kantana Do Racho<br>Liberaba<br>Augusto De Lima   
   | AVANCE ANTOREO BALES, 1117.<br>Radova CE-DES (Inclus Mulangi-Analosa), 4/4 - Mario da CETEL<br>Bua da Engenha, 6/47<br>Rod. Br-DES, 501 195<br>R. Cendida Lazari, Murri, 324  
   | Realities<br>Greatifies<br>Greatifies<br>Greatifies<br>Greatifies  | MART<br>MILF-SUPPORT.<br>SELF-SUPPORT.<br>SELF-SUPPORT.   | 4<br>40<br>35<br>30<br>40  
  | M01 INFORMED<br>14.71<br>26.67<br>6.73<br>1.75   | NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED  | NO<br>NO<br>2547<br>25<br>NO                             | MOT INFORMED<br>MOT INFORMED<br>MOT INFORMED<br>Belowis<br>MOT INFORMED  | NDK<br>OK<br>OK<br>NDK   | 8<br>8<br>8<br>8<br>8  | NX<br>XX<br>XX<br>XX  
   | WARNING LIGHTING DOES NOT WORK<br>NOT<br>PLATFORMUSE DOLEATED<br>NEEDS EXTENDED JANTORIAL SERVICES   | MOT<br>MOT<br>MOT<br>MOT   | FILIED<br>LICENSARLE<br>LICENSARLE<br>LICENSARLE<br>LICENSARLE   | NOT INFORMED<br>NOT INFORMED<br>NOT<br>NOT   
  |
| 26 648307<br>27 648391<br>28 649623<br>29 652212   | 504P (CH10054725)<br>2548 (CH1005508)<br>667P (CH10056680)<br>546P (CH10056680)   | PRP000516<br>PRP1,90254<br>PRCTA0567<br>PRCTA0546   | -25.1275278<br>-26.477325<br>-25.4219167<br>-25.3980556   
  | -50.1540556<br>-52.00383<br>-49.2545<br>-49.254557   | 14 HL HL   | Porta Golda<br>Patrial<br>Curtiba  
   | RUA THADMATINGO DE ATRVIDO - RI<br>RUA SERASTÃO CORRA NOURA, S/N<br>RUA URALDINO DO ANNAA, 37, ESTÁRIO COUTO PEREIRA<br>AVENDA PADAM ITÁS   
   | Rodhae<br>Greatheid<br>Rodhae  | MAST<br>SELF-SLIPPORT.<br>PRAST<br>PRAST  | 6<br>43<br>(4,2+4,2+6,2)<br>(2+2×3)  
  | 2.47<br>11.92<br>(0.72+0.73+0.73)<br>(0.68+0.68+0.68)  | NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED  | 50<br>50<br>50<br>50                                     | MOT INFORMED<br>MOT INFORMED<br>MOT INFORMED<br>MOT INFORMED   | NOK<br>OK<br>NOK   | 8<br>8   | 8 9 98  
   | EQUIPMENT CAENET CODATIO<br>NOT<br>EQUIPMENT CAENET CODATIO. PROBLEMS<br>INTER CAELE ARRANGEMENT.<br>ECUIPMENT CAENET CODATIO  | NOT<br>NOT<br>NOT  | NOT BEORMED<br>LICENSAILE<br>FILID<br>FILID  | NOT INFORMED<br>NOT INFORMED   
  |
| 20 652254<br>21 654565<br>22 655624<br>23 655624   | 1143 (CM10011882)<br>0148 (CM10010511)<br>PR04 (CM10045120)<br>641P (CM10018770)  | BRAPTICI<br>S2BABROCCO<br>MORCASON<br>PRETADAS  | -12.85525<br>-12.97904<br>-21.217%<br>-25.4243056   
  | - 38.482907<br>- 38.483489<br>- 43.79795<br>- 43.7776.389  | M N0   | Rabaka<br>Salvador<br>Rabakan<br>Curtika   
   | AV JUTAHI MADAWARS, 600<br>RUA BERGINDO SPECTOR, 94<br>Morto morte Maria - 6/07<br>RUA TREAT DE MADA 191  
   | Greenfield<br>Greenfield<br>Greenfield<br>Southait   | SELF-SUPPORT  | 21<br>30<br>61.5<br>6  
  | 4.88<br>2.9<br>162.57<br>2.19  | NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED  | 20 2 20  | MOT INFORMED<br>MOT INFORMED<br>ART WORKER   | ox<br>ox<br>NX   | 8<br>8<br>8  | 99.9  
   | NOT<br>NOT<br>MANAGERIE AND  | NOT<br>TOYATR BASE<br>SHOULS PROBLEM<br>NOT  | LICENSARLE<br>LICENSARLE<br>LICENSARLE<br>FILED  | NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED   
  |
| 24 657545<br>25 658575<br>26 65864   | 559 (CH10018602)<br>0067 (CH10012666)<br>6787 (CH10012566)  | PRECADESS<br>BREADEDS<br>PRECADE/R  | -21.29575<br>-14.763407<br>-25.4078611  
  | -29.063629   | PR. MA   | Rampa<br>Ibeus<br>Curtika  
   | AX: DOWN SOPHIA RASICLARF, 466<br>AVENDA PROCLAMACAO, 3/N<br>RUN COMINDADOR AMRUDO, 720. HOTEL BER  
   | Routhap<br>Greatheid<br>Routhap  | MAST<br>SELF-SLIPPORT.<br>MAST  | 6<br>43<br>(6+6)   
  | (1.07_107+107)<br>8.52<br>2.55   | NOT INFORMED<br>NOT INFORMED  | 50<br>8.52<br>50   | MOT INFORMED<br>BELMEREX LTDA  | NDK<br>NDK   | ox<br>ox   | 8 8   
   | EQUIPMENT CARDIET CODATED<br>NEEDS SETENDED SANTONIAS SERVICES.<br>NO GROUND CARLIES AT TOWER INSE.<br>SUBSTITUTE WARATING LOSATING<br>EQUIPMENT CARDIET OXIDATED  | NOT<br>NOT   | FILED<br>LICENSABLE<br>FILED   | NOT INFORMED   
  |
| 17 6593%<br>18 659280  | 2528 (CH10053228)<br>685P (CH10053228)<br>0008 (CH100532589)  | PROMODIA<br>PROPHOSAS<br>\$2844980061   | -24.2113899<br>-24.037<br>-12.9883889   
  | -48,2820278<br>-50,9252778<br>-52,369  | PR 10  | Campo Mourao   
   | RUR SÃO FAILO. 99<br>Rua Laurado Burges, 1398 - Centro, Campo Mourán - PR<br>Avendon COM JOAO VI, 466   
   | Reaching<br>Reaching   | MUP-SUPPORT   | 44<br>(4+4)  
  | 13.87<br>(0,810 + 0,426)<br>9.41   | NOT INFORMED  | NO<br>1187<br>NO<br>8.60                                 | MOT INFORMED<br>NOT INFORMED   | ok<br>Nok  | ox<br>ox   | 8 8   
   | NOT  | STRUCTURE<br>OVERLOADED  | FLED   | NOT INFORMED   
  |
| 40 660433<br>41 660404<br>42 660515  | 722# (CM10053632)<br>704# (CM10053632)<br>651# (CM10053306)   | PHP020722<br>PHP020736<br>PHCTAD650   | -25.53894<br>-25.53797<br>-25.3838889   
  | -38.4810556<br>-54.57558<br>-54.58844<br>-49.2320278   | PR   | Faz Do Igaico<br>Faz Do Igaico<br>Curtiba  
   | Rua Rui Badoco, 785<br>Rua Rui Badoco, 475<br>Avianza Patena 3025   
   | Greatfaid<br>Routtap<br>Routtap<br>Routtap   | MAST<br>MAST<br>MAST  | (4+8)<br>(5+8)<br>(5+8+6+8)  
  | (0,72+1,47)<br>(0,77+1.52)<br>(1,86+0,78+0,74+0,74)  | NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED  | ND<br>ND   | NOT INFORMED<br>MOT INFORMED<br>NOT INFORMED   | NDK<br>NDK<br>NDK  | ox<br>ox<br>ox   | NDK<br>OK<br>NDK  
   | QDCA OXIDATED. SUBSTITUTE THE REST<br>PLATFORMS OF MARTS 1 AND 2<br>EQUIPMENT CARDART OXIDATED<br>EQUIPMENT CARDART CODATED  | NOT<br>NOT   | LICENSED<br>LICENSED   | 188<br>188<br>188  
  |
| 43 661251<br>64 662256   | 0006 (CH10069436)<br>4249 (CH10060426)  | PR.DADIG4   | -12.54975   
  | -51.1853   | PR.  | Landera  
   | RIA ABIJO FARAS. 455<br>Ar. Hypenópolis, 75 - Izedro Hypenopolis  
   | Graathid<br>Rodtap   | MUF-SUFFORT.  | 32<br>(6+6+6)  
  | 2.48<br>2,43 (0,833 + 0,853 + 0,853)   | NOT INFORMED  | N0<br>N0   | MOT INFORMED   | OK<br>NOK  | ox<br>ox   | ax<br>ax  
   | NOT<br>EQUIPMENT CABINET COZDATED  | STRUCTURE<br>OVERLOADED  | PILID  | NOT INFORMED   
  |
| 45 662571<br>46 662983<br>47 828570  | 5738 (CH50059853)<br>690P (CH50059737)<br>0202  | R8PX00571<br>PRCTA0690<br>\$2864/90099  | -28.57009<br>-25.4344194<br>-12.990628  
  | -53.43772<br>-69.2626511<br>-39.512699   | KS<br>PK<br>M  | 3alvador   
   | RUA JULIO DE CASTELHOS, S/N<br>Rua Mardino Torres, 926<br>RUIS DA PECINORIA: É  
   | Gravitied<br>Routiap<br>Routian  | SELF-SLIPPORT.<br>MAST<br>MAST  | 36<br>(4+t)<br>3.5   
  | 18.93<br>(0,8 + 1,47)<br>3.21  | NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED  | 18.93<br>NO  | BANSTAL<br>NOT INFORMED<br>NOT INFORMED  | NOK<br>NOK<br>NOK  | NDK<br>OK<br>OK  | NDK<br>NDK<br>NDK   
   | REPART CLOSENG<br>REPART CLOSENG<br>RECEMENT CARENT CREATED.<br>REST PLATFORM OF MAST 1 CREATED.<br>RECOMMENT CARENT COSCARED  | * NOT<br>NOT<br>NOT  | LICENSARLE<br>FILED<br>LICENSARLE  | NOT INFORMED<br>NOT INFORMED   
  |
| 48 879478<br>49 4634370<br>50 7140610<br>51 7140610  | 0138<br>7887<br>0FX (ALPHIG285_TORKS)<br>2019 [FEBOUDDIS_TORKS]   | S28APCK0008<br>PRELF0788<br>ACPY10285<br>PERV02019  | -12.999556<br>-26.1769944<br>-9.622775<br>-8.129981   
  | - 38.510111<br>- 53.3003811<br>- 37.307347<br>- 38.50500<br>- 39.50500   | 1 1 4 10   | Saleador<br>Saleado Filho<br>Practine<br>Recifie   
   | RUA PROFESSOR ADSITUES NOVE, 12, FEDERAÇÃO<br>BUR DAS EDICOS, UN<br>AVADOS SERCEPS, AN<br>EDI VIDEO EN ESTA ESTA ESTA ESTA ESTA ESTA<br>EDI VIDEO EN ENTRE ESTA ESTA ESTA ESTA ESTA ESTA<br>EDI VIDEO EN ENTRE ESTA ESTA ESTA ESTA ESTA ESTA  
   | Konflap<br>Graatfad<br>Graatfad<br>Roffie<br>Roffie  | NAST<br>SALF-SUPPORT.<br>SALF-SUPPORT.<br>NAST  | (7+7+7)<br>50<br>60<br>(5+5+5)<br>(4+5)  
  | 3.34<br>2.15<br>2.4<br>MOT DECOMID   | NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED  | N0<br>5<br>24<br>50                                      | NOT INFORMED<br>CAW<br>NOT INFORMED<br>NOT INFORMED  | NOK<br>OK<br>OK  |  | A N X   
   | NUMBERTITUTE LADDER.<br>NOT<br>NOT   | NOT<br>MOT<br>NOT<br>NOT   | ICONSALS<br>FUED<br>ICONSALS<br>FUED   | NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED   
  |
| 52 7140701<br>54 7140780<br>55 7140041   | CES (PEESP3020 TONE)<br>2244 (PEES2244 TONE)<br>PAA4 (EPFACHES_TONE)<br>PAA1 (EPFACHES_TONE)  | PERSP3020<br>PERCEIDIN<br>SPFACENSE<br>SPFACENSE  | -8.036946464<br>-8.085229<br>-20.53422<br>-20.529   
  | -43.892325<br>-36.872466<br>-47.42859<br>-47.392   | 11<br>12<br>12<br>12   | Recte<br>Recte<br>Franca   
   | AV COV. ADMINIST MADAVARIS 265- HERBORISO<br>RAN CARD DA ALFAREDES, SUN - SECURIDA ALFARENZA<br>Rua Voluntário Plano Mazine, 2013 - Hegenópolis<br>Av. Precidente Varigue 666 - Cidade Nova, Franca - SP  
   | Routhap<br>Routhap<br>Routhap  | MAST<br>MAST<br>MAST  | (4+4)<br>2<br>(1,02+1,02+1,02)<br>(5+5=5)  
  | NOT INFORMED<br>NOT INFORMED<br>(0,98 + 0,48 + 0,48)<br>2,034 (1,504+0,260+0,260)  | NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED  | N0<br>N0<br>N0   | NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED   | ok<br>NOK<br>OK  | ox<br>ox<br>ox   | NDK<br>NDK<br>NDK   
   | NOT<br>NOT<br>WANNENG LIGHTING DOES NOT WORK<br>NOT  | NOT<br>NOT<br>STRUCTORE<br>OVERLOADED  | LICENSARIA<br>LICENSARIA<br>LICENSARIA   | NOT INFORMED<br>NOT INFORMED<br>NOT<br>NOT   
  |
| 56 7141226   | VHLI (SPWCS1728_TONS)   | SPIECS1728  | -23.636   
  | -46.614  | se   | Sao Paulo  
   | Nas disclore Dar LDA, 433 - Vila Notaes   
   | Realitap   | MAST  | 50   
  | 2.569  | NOT INFORMED  | NO   | NOT INFORMED   | OK.  | œ  | ак  
   | NOT  | STRUCTURE<br>CUSRLOADED<br>STRUCTURE   | NOT DECEMBED   | NOT INFORMED   
  |
| 57 7141253<br>58 7141285   | CEEL (SPWAA35%_TORIS)<br>VIIIS (SPGCO436_TORIS)   | SPEAA3575<br>SPOCOIS36  | -23.538<br>-23.549855   
  | -46.484<br>-46.760417  | 59<br>59   | Sao Paulo<br>Osasco  
   | Ras da Padosena, 83 - Jandan Nardeche, 586 Paulo - 59<br>Ras Rosa D'Éspelo Paupo, 256 - Via Yana  
   | Routtap<br>Routtap   | MAST  |  
  | 2.819  | NOT INFORMED  | ND<br>ND   | NOT INFORMED   | OK<br>NOK  | ox<br>ox   | NDK<br>NDK  
   | NOT<br>LIGHTNENG ROD DAMAGED +<br>CRILING DIFETRIZEN   | OVERLOADED<br>STRUCTURE<br>OVERLOADED  | NOT DEFORMED   | NOT INFORMED   
  |
| 59 7141345<br>50 7141350<br>61 7141375   | 3081 (SP580-K98, T0898)<br>8LH1 (SP980-K91, T0898)<br>CT05 (SP580-K07, T0898)   | SPEROISH<br>SPANEORD<br>SPEROIS07   | -23.765333<br>-23.54997<br>-23.71126  
  | -46.555151<br>-46.58456<br>-46.58588   | 59<br>59<br>59   | Sao Bernardo Do<br>Calingo<br>Sao Revio<br>Sao Bernardo Do<br>Calingo  
   | Rua Ruy Barbota, 200<br>Rua Kodo Bartista de Lacieron, 261<br>Rua Munacijul, 200  
   | Routhap<br>Routhap<br>Routhap  | MAST<br>MAST<br>MAST  | (5+5+5)<br>(4+4+1)<br>(4+4+6)  
  | (1,04 + 1,58 + 0,83)<br>NOT BUIGHMED<br>(1,04 + 1,47 + 0,83)   | NOT INFORMED<br>NOT INFORMED  | N0<br>N0<br>N0   | NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED   | 0K<br>OK   | OK<br>OK   | NDK<br>NDK<br>DK  
   | NOT<br>NOT<br>NOT  | NDT<br>MDT<br>NDT  | RUID<br>LICENSARUE<br>LICENSARUE   | TOM<br>TOM<br>TOM  
  |
| 62 7141378   | POSI (SPWBP2582, TORR)<br>PTT (SPWBP2585, TORR)   | SPWW2582  | -23.533   
  | -45.555  | 57   | Sao Paulo  
   | Au. Francisco Patranzoo, 229 - Água Branca, São Paulo - SP<br>Bua Josef Wyos, 319 - Farque Industriol Tantos Solan  
   | Routhap  | MAST  | (5+5+5)<br>(6+6+6+2)   
  | 2.17<br>7,873<br>(1,655+0,909+4,526+1,786)   | NOT INFORMED  | ND<br>ND   | NOT INFORMED   | OK<br>OK   | ox<br>ox   | NDK<br>NDK  
   | NOT  | STRUCTURE<br>OVERLOADED<br>STRUCTURE<br>OVERLOADED   | LICENSED   | vas<br>vas   
  |
| 64 7140462   | 9011 (3P9892585, T0998)<br>5805 (3P9806532, T0998)  | SPSBO463D   | -23.517<br>-23.736927   
  | -46.501711   | 59   | Sao Paulo<br>Sao Bernardo Do<br>Campo  
   | Rua Netra Accèncio, 128 - Fernazigaile - Mis Bernado do Campo   
   | Routtap  | MAST  | (6+6+5+2)<br>(6+12)  
  | (0,655+0,909+4,524+1,786)<br>(0,565 + 4,229)   | NOT INFORMED  | ND<br>ND   | NOT INFORMED   | OK<br>NOK  | ox<br>ox   | NOK   
   | NOT<br>CIGLING INFILTRATION  | STRUCTURE<br>OVERLOADED  | NOT DECRMID  | NOT INFORMED   
  |
| 65 7142666<br>66 7142673   | 5ADS (5PSCH637_TORR)<br>8VA3 (5PW8V2581_TORR)   | 5P5CM637<br>5PWW/589  | -23.418725<br>-23.56629<br>-23.63066  
  | -46.574405<br>-46.45031  | 59<br>59   | Sau Cartani Di Sul<br>Sau Paulo  
   | Rua Acchieto Feorari, 118 - Starto Antilino - Sila Caetana do Sul<br>Antesiza Paulistra, 810  
   | Routiap  | MAST  | (4+4+2)<br>(4+4+4)   
  | 0.862<br>NOT DIFORMED  | NOT INFORMED<br>NOT INFORMED  | ND<br>ND   | NOT INFORMED   | OK<br>OK   | ox<br>ox   | NOX<br>OX   
   | NOT  | STRUCTURE<br>OVERLOADED  | NOT INFORMED   | NOT INFORMED<br>VES  
  |
| 67 7141532<br>68 7141620   | VPA1 (SPWCR341_TORS)  | SPWCROHL<br>SPHERELSE   | -23.6366  
  | -66.66083<br>-66.96.7025   | se<br>se   | Sao Paulo<br>Magi Miran  
   | Rua Miceniec, 274 - Vila Paulata<br>Rua Caronel Verdincio Perveia Alvec Adorno, 89 - Saúde  
   | Routhap<br>Routhap   | MAST  | (6+6+6)<br>(6+6+6+2)   
  | 3.683<br>7.158   | NOT INFORMED  | ND<br>ND   | NOT INFORMED   | NOK<br>OK  | ox<br>ox   | NDK<br>NDK  
   | PLATFORM/SHOL DODATED  | OVERLOADED<br>STRUCTURE<br>OVERLOADED  | FILED  | NOT  
  |
| 59 7141725<br>70 7141794   | VAL (\$P\$0541_T088)<br>VAL (\$P\$18553_T088)<br>CT02 (\$P\$105460_T088)  | SPECIAL<br>SPECIAL<br>SPECIAL   | -21.58731<br>-22.75586<br>-23.59397   
  | -47.80729<br>-47.41877<br>-45.89044  | 2  | Riberas Preto<br>Santa Barbara<br>Direite  
   | Rua Bevarizino de Campos, 1236<br>Rua TREZE DE MADO, 749 - CENTRO   
   | Routhap<br>Routhap   | MAST<br>MAST<br>MAST  | (2+5+5)<br>(4+4+4)   
  | (0,47 + 3,06 + 1,91)<br>4,853 (0,839+0,826+1,382)  | NOT INFORMED  | ND<br>ND   | Tel Telscanuscações<br>NOT INFORMED<br>PORTO SEGURD  | 0K   | OK<br>OK   | NK NK   
   | NOT NOT  | STRUCTURE<br>OVERLOADED  | PLIED  | NOT<br>NOT INFORMED  
  |
| 71 7141806   | CT02 (SPE2CH40_T098)<br>POES (SPP085272_T098)   | \$P\$3C5460<br>\$P\$085272  | -23.19007   
  | -45.89044  | 59<br>59   | Campos<br>Pisas Grande   
   | Nua Euclides Meagan, 26 - Centra<br>Ani, Presedente Costa e Sicke, 13 - Preja Grande - Sp   
   | Routtap<br>Routtap   | MAST  | 1.5<br>(6+6+6)   
  | 1.49<br>(2,375 + 1,824 + 1,731)  | NOT INFORMED  | ND<br>ND   | NOT INFORMED   | NOK<br>OK  | OK<br>OK   | OK<br>NOK   
   | REPAIR SUPPORT   | NOT<br>STRUCTURE<br>OVERLOADED   | PILIED   | VES<br>NOT INFORMED  
  |
| 73 7141838   | vieta (sipsiscietz_tona)  | \$P\$305472   | -23.179228  
  | -06.889161   | 57   | Sau Jose Dos<br>Cempos   
   | Avenda 35a 2016, 626 - Iarden Bela Veza   
   | Realities  | MAST  | (4+4+6+4)  
  | 6.965  | NOT INFORMED  | ND   | NOT INFORMED   | NOK  | ox   | NOK   
   | LADDER DANKED +<br>CELLING IMPERMENELIZATION   | STRUCTURE<br>CUERLOADED.<br>REPLACEMENT<br>OF THE<br>EXPANSIVE<br>ANOHORS BY<br>CHEMICAL<br>ANOHORS.   | LICENSED   | ves  
  |
| 74 7142843   | (B+2001 (SPST55379, TORRE)  | SP9735329   | -23.959   
  | -46.346  | 59   | Santas   
   | Nuo Saturnos de Brita, 256 - Marapó - Santos  
   | Routhap  | MAST  | (3+3=6)  
  | 1.579  | NOT INFORMED  | ND   | NOT INFORMED   | OK.  | ox   | NOK   
   | 207  | ANCHORS BI   | LICENSARLE   | NOT  
  |
| A Marci  |   |   |   
  |  |  |  
   |   
   |  |   |  
  |  |   |  |  |  |  |   
   | NOT  | REPPORCE<br>AND HORS TO<br>ALL CONCRETE<br>BASES.  | L. HUNKLI  | 1671   
  |
| 75 7140905   | ves3 (sestsizes_toos)   | SP97352%  | -23.949   
  | -46.33   | 57   | Santas   
   | An. Ana Costa, 121 - Vila Matan   
   | Realtap  | MAST  | {3+3+3+3}  
  | 3.272  | NOT INFORMED  | N/D  | NOT INFORMED   | NOK  | OK.  | NOK   
   | LIGHTWING KOD GAMINGED   | LIFELINES OF<br>THE POSTS  | FULED  | NOT INFORMED   
  |
| 76 77.4787.1   |   |   |   
  |  |  |  
   |   
   |  |   | | | | |
  |  |   |  |  |  |  |   
   |  | AND ANCHORS<br>CODDATED.   |  |  
  |
| 77 7141970   | OBD (SPSTSSHD3_TONG)<br>SME2 (SPSLMSHD1_TONG)   | SPST35423<br>SPSLMieds  | -23.935   
  | -46.311  | 59<br>59   | Santas<br>Surare   
   | Rua Verquero Stedel, 304 - Aparocita<br>Rua Sulo Antôno Stared, 28 - 31. Bust Retro - Sustani   
   | Routop   | MAST  | (3+3×3)<br>B   
  | 1249   | NOT INFORMED  | ND<br>ND   | NOT INFORMED   | ox<br>ox   | ox<br>ox   | NDK<br>NDK  
   | NOT NOT  | AND ANCHORS<br>CODATED.<br>STRUCTURE<br>OVERLOADED<br>STRUCTURE<br>OVERLOADED  | FILED NOT DEFORMED   | NOT INFORMED   
  |
| 77 7142970<br>78 7142007<br>79 714208<br>80 714208<br>81 714208  |   |   | -22.84806<br>-12.9948889<br>-12.83982<br>-14.83889<br>-11.4881111   
  |  |  |  
   | Rea 3050 Antibio Source, 18 - 53. Biole Retro - Surané<br>Rio Roberts Collit, 19<br>Aurela Poro de Garciana, 19 A. Miller Mani,<br>Rea 4 Spars, 519 - Collon Mani,<br>Rea 4 Spars, 519 - Collon J.  
   |  |   |  
  |  |   |  | NOT INFORMED<br>NOT INFORMED<br>SECCODEL<br>NOT INFORMED   |  |  | NOX<br>NOX<br>OX<br>OX<br>NOX   
   |  | AND ANCHORS<br>CODUCTIVE<br>CODUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>STRUCTIVE<br>CONFLICACED<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>STRUCTIVE<br>ST   |  | NOT<br>NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED  
  |
| 77 7140800<br>78 7142807<br>78 7142807<br>78 714288<br>80 714288<br>81 714288<br>83 7142887<br>83 7142887  | SME2 [SPSUMSe01_TORM]   |   |   
  | -47.18647  |  |  
   | Rus Joho Antánio Soarres, 18 - 32. Bon Retro - Surrané  
   |  | MART<br>MART<br>SBU-SUPPORT.<br>SBU-SUPPORT.<br>SBU-SUPPORT.<br>MART  |  
  |  |   |  | NOT INFORMED   | OK<br>NOK<br>OK<br>NOK<br>NOK  | 05<br>05<br>07<br>07<br>07<br>05<br>05<br>05<br>05<br>05<br>05           | NCK<br>NCK<br>OK<br>NCK<br>NCK  
   |  | AND ANCHORS         COLUMBIC           COLUMBIC         STELECTURE           OVARIGACED         STELECTURE           STELECTURE         OVARIGACED           S         NOT           S         NOT      S         NOT  |  | NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED<br>NOT  
  |
| 77         7143933           78         7142812           78         7142812           80         714281           81         714281           82         714281           82         714281           81         714281           82         714281           81         714281           81         714281   | SHE2 [SHIM GO1_TORE]<br>ODG [AND/MOS_TORE]<br>O23 (AND/MOS_TORE]<br>ORC [INGCOL23, TORE]<br>ORC [INGCOL23, TORE]<br>OTE INACLOSE TORE]<br>PHO [SPARKING_TORE]   | 578204601<br>52842040018<br>52845001223<br>52845001223<br>8840001223<br>8840012430<br>579270388   | -22.84806<br>-12.9948889<br>-12.8992<br>-14.25889<br>-14.85889<br>-12.689278  
  | -47.28647<br>-28.475889<br>-28.45588<br>-29.46588<br>-27.4111889<br>-46.679722   | 57<br>85<br>85<br>87   | Surary<br>Sahadar<br>Sahadar<br>Gangogi<br>Ma Real<br>Sao Paulo  
   | No Job Artislos Scares, 18 - 31. Bos Retro - Surazá<br>BOR HUMPS (2011) 194<br>Antonio 1942 de Davisor A La Acone Pasa<br>Roa da Spara, 1974 - Caron<br>Roa da Spara, 1974 - Caron<br>Roa Corresta, Bravardo La Jo<br>Avenia Nove da Julio, 5024 - Judio Paulita - 588 Paulo  
   | Rostrap<br>Rostrap<br>Great/Add<br>Great/Add<br>Great/Add<br>Rostrap   | MAST<br>MAST<br>SELF-SUPPORT.<br>SELF-SUPPORT.  | 8<br>33<br>45<br>72<br>10  
  | 3.78<br>(1,31+6,85+6,85)<br>2.78<br>16.16<br>2.694   | NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED  | ND<br>ND<br>34<br>9416<br>ND                             | NOT INFORMED<br>SECTOMA<br>MOT INFORMED<br>NOT INFORMED<br>NOT INFORMED  | OK<br>NOK<br>OK<br>NOK   | OK<br>NGK<br>DK<br>DK<br>NGK<br>OK                                       | K K K K   
   | NOT<br>INSICS EXTRADED INSITTORIAL BERVICES<br>NOT<br>SECO EXTRADED INSITTORIAL SERVICES<br>VARANNAL LISEN VARIA +<br>LISEN VARIA KOD DAVAGED +<br>NGD DESARTED  | Mo MCHORE<br>CORRESPONDED<br>THE COMPLEMENT<br>OWNER AND THE<br>OWNER  | NOT INFORMED<br>FILIED<br>LECENSARIES<br>LECENSARIES<br>LECENSARIES  | NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED  |
| 17 7143973<br>78 7143973<br>79 7143827<br>79 7143828<br>80 7143782<br>81 714383<br>84 714383<br>84 714388<br>85 7143868<br>85 7143869  | SPE2 (SPEAKGOL, TORIS)<br>0056 (MADWEES, TORIS)<br>0258 (MADWEES, TORIS)<br>0258 (MADWEES, TORIS)<br>0258 (MADWEES, TORIS)<br>PRES (SPAPENIES, TORIS)<br>VGC (SPATALISE, TORIS)   | SPALMAGI<br>SINAZZWOJA<br>NIMOCOJI<br>SINACCOJI<br>SANCIAN<br>SPUPANIS<br>SPUPANIS<br>SPUTALIS  | - 22.84806<br>- 12.946889<br>- 12.84932<br>- 14.12889<br>- 13.881113<br>- 23.585278<br>- 23.54583   
  | -47.18647<br>-38.455889<br>-38.4638<br>-17.451889<br>-46.4595722<br>-46.4595722<br>-46.5659  | 57<br>M<br>M<br>M<br>S7<br>S7  | Surders<br>Statudor<br>Balador<br>Balador<br>Sa Real<br>Sao Paulo<br>Sao Paulo   
   | Rea John Andreas Ealers, EH 32, man Anton - Harmed<br>Rear Harman Markani, EH.<br>Mannah Anton and Anton Anton Anton<br>Anton & Sagara, M.Y Corris<br>Man Canada, Markani, Markani, S Markani, S<br>Anton, M Sadan Fander, S Sala<br>Anton Sana, S Sadan Fander, S Sala<br>Rad, Canagan, 734 Vila Green, Cardon - Sal   
   | Roaft sp<br>Roaft sp<br>Greatified<br>Greatified<br>Greatified<br>Roaft sp<br>Roaft sp   | MAGE<br>MAGE<br>BALF-AUFFORT,<br>BALF-AUFFORT,<br>BALF-AUFFORT,<br>MAGE   | 8<br>55<br>45<br>73<br>53<br>53<br>53<br>(3+3=2)   
  | 3.78<br>(3,18+6,88+6,88)<br>2.78<br>36 16<br>2.694<br>2.899  | NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED  | ND<br>ND<br>24<br>24<br>ND<br>ND                         | NOT INFORMED<br>INCLUSIONAL<br>INCLUSIONAL<br>INCLUSIONAL<br>INCLUSIONAL<br>INCLUSIONAL<br>INCLUSIONAL<br>INCLUSIONAL  | OK<br>NOK<br>OK<br>NOK   | OK<br>NOK<br>OK<br>NOK<br>OK   | OK<br>OK<br>NGK<br>NGK  
   | NOT<br>REFORMANT ANTICAL SERVICES<br>NOT<br>NOT<br>NOT<br>NOT<br>NOT<br>NOT<br>NOT<br>NOT  | AND ANCHORS<br>CORRATE.<br>CORRATE.<br>CONSTLUENCE<br>OVERLANDED<br>OVERLANDED<br>OVERLANDED<br>STRUCTURE<br>STRUCTURE<br>OVERLANDED<br>STRUCTURE<br>STRUCTURE<br>OVERLANDED<br>OVERLANDED<br>OVERLANDED<br>OVERLANDED<br>OVERLANDED<br>OVERLANDED<br>OVERLANDED<br>OVERLANDED   | NOT DEGRMED<br>FILED<br>FILED<br>LICENSIARLE<br>LICENSIARLE<br>LICENSIARLE   | NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED<br>NOT<br>NOT<br>YES  
  |
| 17         74.0903           77         74.0903           78         74.0903           78         74.0903           81         74.0903           82         74.0903           84         74.0403           84         74.0403           85         74.0403           86         74.0403           86         74.0403   | 5.962 [IPE0/5601_T0000]<br>0396 [MAXIFORM T0000]<br>0396 [MAXIFORM T0000]<br>0496 [MAXIFORM T0000]<br>0496 [MAXIFORM T0000]<br>0496 [IPE0/2010.01000]<br>VGC1 [IPE0/2010.01000]<br>VGC1 [IPE0/2010.01000]   | SPILMIG1<br>SIMADHODIA<br>SIMADHODIA<br>SIMADHODIA<br>SPILMODIAL<br>SPILMODIAL<br>SPILMODIAL<br>SPILMODIAL<br>SPILMODIAL<br>SPILMODIAL  | - 22.84006<br>- 12.940008<br>- 12.97912<br>- 13.72079<br>- 13.4853131<br>- 23.56082<br>- 23.56082<br>- 23.50881   
  | -47.38647<br>-38.452839<br>-38.45127<br>-38.6638<br>-37.451389<br>-46.679722<br>-46.5649<br>-46.67499  | 17<br>M<br>M<br>M<br>17<br>17<br>17  | Suraw<br>Salvador<br>Salvador<br>Salvador<br>Salv Paulo<br>Salv Paulo<br>Salv Paulo  
   | Ra Iski ostim Isano, H Ia Rostenio - Jamai<br>Ra Iski ostim - Karana (H. 1997)<br>- Ra Isano Isano, H. 1997<br>- Ra Isano Isano, H. 1997<br>- Ra Isano, H.  
  | Rodtap<br>Bestap<br>Greating<br>Greating<br>Rodtap<br>Rodtap<br>Rodtap   | МАЛТ<br>МАЛТ<br>МЕР-МИРОНТ,<br>МЕР-МИРОСТ,<br>МЕР-МИРОСТ,<br>МАЛТ<br>МАЛТ<br>МАЛТ   | 8<br>10<br>40<br>73<br>10<br>(3+3+3)<br>(3+3+4)   | 3.78<br>(3.31+6.87+6,89)<br>6.33<br>2.494<br>2.494<br>2.494<br>3.909<br>3.961                         
  | ACT INFORMED<br>ACT INFORMED<br>ACT INFORMED<br>ACT INFORMED<br>ACT INFORMED<br>ACT INFORMED  | N/D<br>N0<br>24<br>314<br>N0<br>N0<br>N0                 | NOT INFORMED<br>INC. INFORMED<br>INC. INFORMED<br>INC. INFORMED<br>INC. INFORMED<br>INC. INFORMED<br>INC. INFORMED   | OK<br>NOK<br>OK<br>NOK<br>NOK<br>OK                                    | 0X<br>10X<br>05<br>102<br>102<br>0X<br>0X                                | OK<br>OK<br>NEX<br>NEX<br>NEX<br>NEX  | NOT<br>ANEXA EXPLICITLY AN ADMITTER<br>BOD<br>ADMITTER ADMITTER ADMITTER<br>ADMITTER ADMITTER<br>ADMITTER ADMITTER<br>ADMITTER ADMITTER<br>ADMITTER
ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>ADMITTER<br>AD | AND ANCHOUSE<br>COMMISSION OF A<br>COMMISSION OF A<br>C   | NOT DEGRMED<br>FILED<br>FILED<br>LICENSIARLE<br>LICENSIARLE<br>LICENSIARLE   | NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED<br>NOT<br>YES  |
| 1         10000           17         10000           18         10000           11         10000           12         10000           13         10000           14         10000           15         10000           16         10000           17         10000           18         10000           16         10000           16         100000           17         100000           18         100000           18         100000           18         100000           18         100000           18         100000           18         100000           18         100000           18         100000           18         100000           18         100000           18         100000           19         100000           10         100000           11         100000           12         100000           13         100000           14         100000   | 5442 [07104161,1090]<br>005 [0400945 1099]<br>015 [0400945 1099]<br>015 [0400411 1099]<br>012 [0400411 1099]<br>945 [0400411 1099]<br>945 [0400411 1099]<br>945 [0400410,1099]<br>945 [0400410,1099]  | SPECHICOL<br>SUBJUCTOLIZ<br>SUBJUCTOLIZ<br>SUBJUCTOLIZ<br>SUBJUCTOLIZ<br>SUBJUCTOLIZ<br>SUBJUCTOLIZ<br>SUBJUCTOLIZ<br>SPECHICOLIZ<br>SPECIFICIALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SPECIFICALIZ<br>SP  | - 22.44056<br>- 12.940289<br>- 12.24022<br>- 14.12499<br>- 14.12499<br>- 14.12499<br>- 23.540273<br>- 23.54081<br>- 23.54081<br>- 23.532491<br>- 23.53292   
  | -6.3867<br>-38.475889<br>-39.4458<br>-17.43189<br>-46.47772<br>-66.47772<br>-66.4749   | 57<br>M<br>M<br>M<br>57<br>57<br>57<br>57  | Surbay<br>Shihadar<br>Shihadar<br>Surbay<br>Sur Paulo<br>Suo Paulo<br>Suo Paulo<br>Suo Paulo   
   | An adda and the strength of the distribution strength<br>interaction of the strength of the stre  
   | Rudtap<br>Bartop<br>Geschel<br>Geschel<br>Rudtap<br>Rudtap<br>Rudtap   | 94527<br>942 - 567407<br>943 - 567407<br>943 - 567407<br>9437 - 567407<br>9457<br>9457<br>9457  | 8<br>10<br>40<br>70<br>10<br>(2+2+24)<br>(2+2+24+4)<br>(4+6+6)   
  | 3.78<br>(1.33.46,846,80)<br>3.78<br>3.78<br>2.494<br>3.809<br>3.861<br>3.861   | NOT INFORMO<br>ROT INFORMO<br>SCI INFORMO<br>SCI INFORMO<br>NOT INFORMO<br>NOT INFORMO<br>NOT INFORMO<br>NOT INFORMO  | ND<br>ND<br>34<br>NO<br>NO<br>NO                         | NOT INFORMO<br>NOT INFORMATION<br>NOT INFORMATION<br>NOT INFORMATION<br>NOT INFORMATION<br>NOT INFORMATION<br>NOT INFORMATION  | OK<br>NOK<br>OK<br>NOK<br>NOK<br>OK                                    | 05<br>102<br>102<br>102<br>102<br>102<br>102<br>102<br>102<br>102<br>102 | OK<br>OK<br>NEX<br>NEX<br>NEX<br>NEX  | NOT<br>SASES STREED INSTITUCE, LINE SEE<br>WE<br>SERVICE UNIT OF STREED INSTITUCE<br>SERVICE UNIT OF STREED IN SERVICE<br>SEC OLIVER IN SERVICE IN SERVICE IN<br>USE OF STREED IN SERVICE IN SERVICE<br>INSTITUCE IN SERVICE IN SERVICE IN<br>NOT   
  | ADD 005-005. ADD 0   | NOT DEGRMED<br>FILED<br>FILED<br>LICENSIARLE<br>LICENSIARLE<br>LICENSIARLE   | NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED<br>NOT<br>YES  |
| 17 71-000<br>17 71-000<br>17 71-000<br>10 100<br>10 10 | 100 [headint'ional<br>and headint'ional<br>and headint'ional<br>and headint'ional<br>and headint'ional<br>and headint'ional<br>and headint'ional<br>and headint'ional<br>and headint'ional<br>and headint'ional   | SPRINGEL<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATER  | -22.4406 -1.2.4006 -1.2.400809 -1.2.1012 -1.1012 -1.1022 -21.4023 -21.4023 -21.4023 -21.4024 -21.432 -21.43 -21.43 -2   |
-0.1867<br>-0.45000<br>-0.45000<br>-0.45000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000   | 37<br>M<br>M<br>M<br>37<br>37<br>37<br>37<br>37<br>37  | Surters           Stratular           S  
   | All and handling latery, at - 0, personants Annual<br>Annual Annual Annual Annual Annual Annual<br>Annual Annual Annual Annual Annual<br>Annual Annual Annual Annual<br>Annual Annual Annual Annual<br>Annual Annual<br>Annual Annual<br>Annual Annual<br>Annual Annual<br>Annual Annual<br>Annual
Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annua  | Rutho<br>Rotho<br>Desided<br>Parallel<br>Rutho<br>Rutho<br>Rutho<br>Rutho<br>Rutho   | MART<br>MART<br>1947-4 SUPPORT<br>1947-4 SUPPORT<br>1947-4 SUPPORT<br>MART<br>MART<br>MART<br>MART  | 8<br>00<br>00<br>00<br>00<br>00<br>00<br>00<br>00<br>(1+2+20)<br>(1+2+20)<br>(1+2+20)<br>(1+2+20)  
  | 2.17<br>1.9.16 and, and,<br>2.09<br>2.409<br>2.409<br>2.409<br>2.409<br>2.409  | NOT IMORED<br>NOT IMORED<br>NOT IMORED<br>NOT IMORED<br>NOT IMORED<br>NOT IMORED<br>NOT IMORED<br>NOT IMORED<br>NOT IMORED<br>NOT IMORED  | N/D<br>N/D<br>2014<br>2014<br>N/D<br>N/D<br>N/D<br>N/D   | ACT INFORMS<br>ACT INFORMS<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFOR | OK<br>NOK<br>NOK<br>NOK  | 06<br>102<br>06<br>06<br>06<br>06<br>06<br>06<br>06<br>06<br>06<br>06    | DK<br>DK<br>NDX<br>NDX<br>NDX<br>NDX<br>NDX   
   |  | ADJOINT CONTRACTOR CON   | NOT DEGRMED<br>FILED<br>FILED<br>LICENSIARLE<br>LICENSIARLE<br>LICENSIARLE   | NOT INFORMED<br>NOT EVENTS<br>NOT EVENTS<br>NOT EVENTS<br>NOT<br>NOT<br>NOT<br>NOT<br>NOT<br>NOT INFORMED   |
| 1         7.1.54367           77.         7.44377           78.         7.44377           78.         7.44377           78.         7.44377           78.         7.44387           88.         7.44481           88.         7.44481           88.         7.44481           88.         7.44481           88.         7.44481           88.         7.44481           88.         7.44485  | sina (unionistici, totea)<br>one (unionistici, totea)<br>one (unionistici, totea)<br>one (unionistici, totea)<br>one (unionistici, totea)<br>voici (unionistici, totea)<br>voici (unionistici, totea)<br>voici (unionistici, totea)<br>voici (unionistici, totea)<br>voici (unionistici, totea)   | SPRANKALI<br>SUBATING<br>SUBATING<br>SUBATING<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK<br>SPRANK   |
-22.8406<br>-23.949807<br>-3.12807<br>-3.12807<br>-3.12807<br>-3.14207<br>-23.5407<br>-23.5407<br>-23.5407<br>-23.5407<br>-23.5201<br>-23.5201<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23.5207<br>-23. | -6.3867<br>-86.45887<br>-36.45887<br>-37.45887<br>-37.4588<br>-46.47972<br>-6.3668<br>-6.3498<br>-6.3498   | 37<br>M<br>M<br>M<br>37<br>37<br>37<br>37<br>37<br>37  | Surgery           Matchalar           Oncopy           Sa Paulo   
   
  | <ul> <li>Bar San Hann, Y. J., Barvelen, Mark</li> <li>Anna Mark, S. J., Barvelen, J. B., Barken, J. Barken,</li></ul>  | Nurlap<br>Rechar<br>Describer<br>Describer<br>Nurlap<br>Ruftap<br>Ruftap<br>Ruftap   | MART<br>MART<br>1947-4 SUPPORT<br>1947-4 SUPPORT<br>1947-4 SUPPORT<br>MART<br>MART<br>MART<br>MART  | 8<br>23<br>40<br>22<br>23<br>23<br>(2+3+3)<br>(2+3+3+4)<br>(2+3+3+4+)<br>(2+6+4)<br>(2+6)  
  | 2.79<br>(1.21 (4.24 (2.24)))))))))))))))))))))))))))))))))))   | NOT INVOLVED<br>NOT INVOLVED<br>NOT INVOLVED<br>NOT INVOLVED<br>NOT INVOLVED<br>NOT INVOLVED<br>NOT INVOLVED  | N0<br>X0<br>24<br>41<br>X0<br>N0<br>N0<br>N0<br>N0       | NOT INFORME<br>INFORMATION AND AN ADDRESS<br>INFORMATION AND AN ADDRESS<br>INFORMATION AND ADDRESS<br>INFORMATION ADDRESS<br>INFORMAT               | OK<br>NOK<br>NOK<br>NOK  | 06<br>102<br>06<br>06<br>06<br>06<br>06<br>06<br>06<br>06<br>06<br>06    | DK<br>DK<br>NDX<br>NDX<br>NDX<br>NDX<br>NDX   |  
   | A Discrete A Construction of the Construction  | NOT DEGRMED<br>FILED<br>FILED<br>LICENSIARLE<br>LICENSIARLE<br>LICENSIARLE   | NOT INFORMED<br>NOT EVENTS<br>NOT EVENTS<br>NOT EVENTS<br>NOT<br>NOT<br>NOT<br>NOT<br>NOT<br>NOT INFORMED   |
| Image: constraint of the second sec  | 100 [headint'ional<br>and headint'ional<br>and headint'ional<br>and headint'ional<br>and headint'ional<br>and headint'ional<br>and headint'ional<br>and headint'ional<br>and headint'ional<br>and headint'ional   | SPRINGEL<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATERIA<br>SUMATER  | -22.4406 -1.2.4006 -1.2.400809 -1.2.1012 -1.1012 -1.1022 -21.4023 -21.4023 -21.4023 -21.4024 -21.432 -21.43 -21.43 -2   |
-0.1867<br>-0.45000<br>-0.45000<br>-0.45000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000<br>-0.5000   | 37<br>M<br>M<br>M<br>37<br>37<br>37<br>37<br>37<br>37  | Surters           Stratular           S  
   | All and handling latery, at - 0, personants Annual<br>Annual Annual Annual Annual Annual Annual<br>Annual Annual Annual Annual Annual<br>Annual Annual Annual Annual<br>Annual Annual Annual Annual<br>Annual Annual<br>Annual Annual<br>Annual Annual<br>Annual Annual<br>Annual Annual<br>Annual
Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annual<br>Annua  | Rutho<br>Rotho<br>Desided<br>Parallel<br>Rutho<br>Rutho<br>Rutho<br>Rutho<br>Rutho   | MART<br>MART<br>1947-4 SUPPORT<br>1947-4 SUPPORT<br>1947-4 SUPPORT<br>MART<br>MART<br>MART<br>MART  | 8<br>00<br>00<br>00<br>00<br>00<br>00<br>00<br>00<br>(1+2+20)<br>(1+2+20)<br>(1+2+20)<br>(1+2+20)  
  | 2.17<br>1.9.16 and, and,<br>2.09<br>2.409<br>2.409<br>2.409<br>2.409<br>2.409  | NOT IMORED<br>NOT IMORED<br>NOT IMORED<br>NOT IMORED<br>NOT IMORED<br>NOT IMORED<br>NOT IMORED<br>NOT IMORED<br>NOT IMORED<br>NOT IMORED  | N/D<br>N/D<br>2014<br>2014<br>N/D<br>N/D<br>N/D<br>N/D   | ACT INFORMS<br>ACT INFORMS<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFORMATION<br>INFOR | OK<br>NOK<br>NOK<br>NOK  | 06<br>102<br>06<br>06<br>06<br>06<br>06<br>06<br>06<br>06<br>06<br>06    | NDX<br>NDX<br>NDX<br>NDX<br>NDX<br>NDX<br>NDX<br>NDX<br>NDX   
   |  | A Discrete Section of Control of  | NOT DEGRMED<br>FILED<br>FILED<br>LICENSIARLE<br>LICENSIARLE<br>LICENSIARLE   | NOT INFORMED<br>NOT EVENTS<br>NOT EVENTS<br>NOT EVENTS<br>NOT<br>NOT<br>NOT<br>NOT<br>NOT<br>NOT INFORMED   | | | | | | | | | | | | | | | | | | |
| 10         75.04007           70         74.04007           70         74.04007           70         74.04007           82         74.04007           82         74.04007           82         74.04007           82         74.0400           85         74.0400           85         74.0400           86         74.0400           87         74.0400           88         74.0400           89         74.0400           89         74.0400           89         74.0400           80         74.0400  | Ant Towarding Looki<br>Ant Towardi<br>Ant Towarding Looki<br>Ant Towarding Looki<br>Ant Towarding | 878245622<br>8.98247523<br>4.98247523<br>4.9825023<br>8.9997038<br>8.9997038<br>8.9977353<br>8.9977354<br>8.9977363<br>8.9977364<br>8.9972364<br>8.9972364<br>8.9972364   | - 22.46%<br>- 12.66%<br>- 22.66%<br>- 12.22%<br>- 12.22%<br>- 22.66%<br>- 23.52%<br>- 23.65%<br>- 23.6 | -0.3847<br>34 47000<br>-0.47000<br>-0.4500<br>-0.4500<br>-6.500<br>-6.500<br>-6.500<br>-6.500<br>-6.500<br>-6.500  | u<br>M<br>M<br>M<br>U<br>U<br>U<br>U<br>U<br>U<br>U<br>U<br>U<br>U<br>U<br>U<br>U<br>U<br>U<br>U | Surfars         Surfars           Standar         Standar  | Ale la hol targe target, a - 0, per series to series<br>menta a series a series a series a series a series a series<br>de la series de la serie de la series de la series de la series<br>de la series de la serie de la series de la series de la series<br>de la series de la series de la series de la series de la series<br>de la series de la series de la series de la series de la series<br>de la series de la series de la series de la series de la series<br>de la series de la series de la series de la series de la series<br>de la series de la series de la series de la series de<br>de la series de la series de la series de la series de<br>de la series de la series de la series de la series de<br>de la series de la series de la series de la series de<br>de la series de la series de la series de la series de<br>de la series de la series de la series de la series de<br>de la series de la series de la series de la series de<br>de la series de la series de la series de la series de<br>de la series de la series de la series de la series de<br>de la series de la series de la series de la series de<br>de la series de la series de la series de la series de<br>de la series de la series de la series de la series de<br>de la series de la series de la series de la series de<br>de la series de la series de la series de la series de<br>de la series de la series de la series de la series de<br>de la series de la series de la series de la series de<br>de la series de la series de la series de la series de la series de<br>de la series de la series de<br>de la series de la series de<br>de la series de la  | ระสาม<br>- คะสาม<br>- คะสาม<br>- คะสาม<br>- คะสาม<br>- ระสาม<br>- ระสาม<br>- ระสาม<br>- ระสาม  | 9027<br>9027<br>902 - 2019<br>902 - 2019<br>902 - 2019<br>9027<br>9027<br>9027<br>9027<br>9027<br>9027<br>9027<br>9027  | 8 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1   | 1.78<br>1.6.10 2014 (0)<br>2.00<br>2.409<br>2.409<br>2.409<br>2.409<br>2.409<br>2.409<br>2.409<br>2.409<br>2.409   | ACT INFORME<br>ACT INFORME<br>ACT INFORME<br>ACT INFORME<br>ACT INFORME<br>ACT INFORME<br>ACT INFORME<br>ACT INFORME<br>ACT INFORME<br>ACT INFORME  | ND<br>ND<br>ND<br>ND<br>ND<br>ND<br>ND<br>ND<br>ND       | NOT INFORME<br>NOT INFORMED<br>INFORMATION<br>NOT INFORME<br>NOT INFORME<br>NOT INFORME<br>NOT INFORME<br>NOT INFORME  | 0x<br>10x<br>0x<br>0x<br>10x<br>10x<br>10x<br>0x                       |  | NDX<br>NDX<br>NDX<br>NDX<br>NDX<br>NDX<br>NDX<br>NDX<br>NDX   | UN UN THE AND A DECEMBENT OF A DECEMBENTAL DECEMBENT OF A DECEMBENT OF A DECEMBENTAL DECEMBENT OF A DECEMBENTAL DECEMBENT OF A DECEMBENTAL DECEMBEN   | ADD/DOCIDE         ADD/DOCIDE           CONTRACTOR         CONTRACTOR           CONTRACTOR         CONTRACTOR <t< td=""><td>NOT SPECIAL CONTRACTOR CONTRACTON</td><td>NOT INFORME<br/>STATE PROVINE<br/>STATE PROVINE<br/>STATE PROVINE<br/>STATE PROVINE<br/>STATE PROVINE<br/>NOT<br/>VIS<br/>NOT<br/>NOT<br/>NOT<br/>NOT<br/>NOT<br/>NOT<br/>NOT<br/>NOT</td></t<> | NOT SPECIAL CONTRACTOR CONTRACTON   | NOT INFORME<br>STATE PROVINE<br>STATE PROVINE<br>STATE PROVINE<br>STATE PROVINE<br>STATE PROVINE<br>NOT<br>VIS<br>NOT<br>NOT<br>NOT<br>NOT<br>NOT<br>NOT<br>NOT<br>NOT  |
| 1         7         FL6007           7         FL6007         7         FL6007           10         FL6007         1         FL6007           21         FL6007         2         FL6007           21         FL6007         2         FL6008           25         FL6008         7         FL6008           26         FL6008         2         FL6008           27         FL6008         2         FL6008           28         FL6008         3         FL6008           29         FL6008         3         FL6008   | And December 2004<br>and December   | 87824641<br>578254641<br>5782547813<br>5805547813<br>5805547813<br>58055458<br>587875488<br>587875488<br>587875286<br>587875286<br>5878752861<br>5878752861<br>5878752861<br>5878752861   | - 22.5000<br>- 22.5000<br>- 41.000<br>- 41.000<br>- 41.000<br>- 41.000<br>- 23.5000<br>- 23.50000<br>- 2                                     | -0.360<br>2.4.260<br>3.4.2673<br>3.4.2673<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.0700<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.0   |  | Surters           Surders  | Mar bin house hand, in , in parameters have a<br>metal metal and the second secon  | Nutlog  | 9457<br>9629<br>9640 (1979)<br>962 (1979)<br>9637<br>9637<br>9637<br>9637<br>9637<br>9637<br>9637   | 8   | 1.0<br>(L264.8)<br>3.0<br>3.0<br>2.40<br>2.40<br>2.40<br>2.40<br>2.40<br>2.40<br>2.40<br>2.  | NOT INDERS           ST INDERS  | ND<br>ND<br>ND<br>ND<br>ND<br>ND<br>ND<br>ND<br>ND<br>ND | Kor sindered  | οκ<br>1968<br>30<br>30<br>30<br>30<br>30<br>30<br>30<br>30<br>30<br>30 |  | OK         OK           OK         OK           MEX         MEX   | And<br>Sector 1 and a sector and a sector of<br>the sector and a sector of the sector of the sector<br>and a sector of the sector of the sector of the sector<br>and a sector of the sector of the sector of the sector<br>and a sector of the sector of the sector of the sector<br>and a sector of the sector of t   | OVERLOADED.<br>REIDFORCEMENT<br>OF THE ARCHOR<br>FOR THE<br>CONDETE MASE<br>REPLACE<br>LIFEEMES OF<br>THE MASE<br>STREATURE<br>OVERLOADED.   | NOT SPECIFIC<br>THAT<br>THAT<br>I THAT<br>I THAT | NOT INFORME<br>STATE PROVINE<br>STATE PROVINE<br>STATE PROVINE<br>STATE PROVINE<br>STATE PROVINE<br>NOT<br>VIS<br>NOT<br>NOT<br>NOT<br>NOT<br>NOT<br>NOT<br>NOT<br>NOT  |
| 1         71         Fielders           78         Fielders         74           81         Fielders         74           82         Fielders         74           83         Fielders         74           84         Fielders         74           85         Fielders         74           86         Fielders         74           87         Fielders         74           88         Fielders         74           89         Fielders         74           80         Fielders         74           81         Fielders         74           82         Fielders         74           83         Fielders         74           84         Fielders         74           85         Fielders         74           86         Fielders         74           87         Fielders         74           88         Fielders         74      <  | And December 2004<br>and December   | 87824641<br>578254641<br>5782547813<br>5805547813<br>5805547813<br>58055458<br>587875488<br>587875488<br>587875286<br>587875286<br>5878752861<br>5878752861<br>5878752861<br>5878752861   | - 22.5000<br>- 22.5000<br>- 41.000<br>- 41.000<br>- 41.000<br>- 41.000<br>- 23.5000<br>- 23.50000<br>- 2                                     | -0.360<br>2.4.260<br>3.4.2673<br>3.4.2673<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.0700<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.2.070<br>4.0   |  | Surters           Surders  | Mar bin house hand, in , in parameters have a<br>metal metal and the second secon  | Nutlog  | 9457<br>9629<br>9640 (1979)<br>962 (1979)<br>9637<br>9637<br>9637<br>9637<br>9637<br>9637<br>9637   | 8   | 1.0<br>(L264.8)<br>3.0<br>3.0<br>2.40<br>2.40<br>2.40<br>2.40<br>2.40<br>2.40<br>2.40<br>2.  | NOT INDERS           ST INDERS  | ND<br>ND<br>ND<br>ND<br>ND<br>ND<br>ND<br>ND<br>ND<br>ND | Kor sindered  | οκ<br>1968<br>30<br>30<br>30<br>30<br>30<br>30<br>30<br>30<br>30<br>30 |  | OK         OK           OK         OK           MEX         MEX   | And<br>Sector 1 and a sector and a sector of<br>the sector and a sector of the sector of the sector<br>and a sector of the sector of the sector of the sector<br>and a sector of the sector of the sector of the sector<br>and a sector of the sector of the sector of the sector<br>and a sector of the sector of t   | ADDITION         ADDITION           ADDITION         ADDITION<   | NOT SPECIFIC<br>THAT<br>THAT<br>I THAT<br>I THAT | NOT INFORME<br>STATE PROVINE<br>STATE PROVINE<br>STATE PROVINE<br>STATE PROVINE<br>STATE PROVINE<br>NOT<br>VIS<br>NOT<br>NOT<br>NOT<br>NOT<br>NOT<br>NOT<br>NOT<br>NOT  |
| 90 7143476   |   | 878246451<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01/21<br>2002/01  | - 21 ANN<br>- 21 A | 4. 2009<br>3. 2009<br>3. 2009<br>3. 2009<br>3. 2009<br>4. 2009<br>4. 2009<br>4. 2009<br>4. 2009<br>4. 2009<br>4. 2009<br>4. 2009<br>4. 2009  
   |  | Burdent           The Address           The Address           Base Provide   
  | Mar balan bandar, da e de barbar harris<br><b>Barbar Barbar Harris</b><br><b>Barbar H</b> | Nutrice           Autority           Autority           Autority           Autority           Autority           Autory   |  
  | (researd<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(researd)<br>(resear  | 2.40<br>2.40<br>2.40<br>2.40<br>2.40<br>2.40<br>2.40<br>2.40<br>2.40<br>2.40<br>2.40<br>2.40<br>2.40<br>2.40<br>2.40   | EV INCOME     EVICE      EVICE     EVICE     
EVICE      EVICEV | 40<br>   | Ye (1998)  | 66<br>97<br>96<br>96<br>96<br>96<br>96<br>96<br>96<br>96<br>96<br>96   |  | 05         02           0.0         0.0           0.0         0.0           0.0         0.0           0.0         0.0           0.0         0.0           0.0         0.0           0.0         0.0           0.0         0.0           0.0         0.0           0.0         0.0           0.0         0.0   | Miles and a second seco   | OVERLOADED.<br>REIDFORCEMENT<br>OF THE ARCHOR<br>FOR THE<br>CONDETE MASE<br>REPLACE<br>LIFEEMES OF<br>THE MASE<br>STREATURE<br>OVERLOADED.   | NOT SPEAKED  | NOT INFORMED<br>NOT INFORMATION<br>NOT INFORMATION<br>NOT INFORMATION<br>NOT INFORMATION<br>NOT INFORMED<br>NOT INFORMED<br>NOT INFORMED   
  |
90 7343475 91 7343477 92 7342534		87824663 87824663 8782467 878247 878258 87858 877858 8778		40.000 40.000		Burdent           Hardent           Hardent <td< td=""><td>Mar balan bandar, 4 - 0 Barton, Mari Marine Marine Marine, 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1</td><td>Nutrice           Nutrice           Nutrice</td><td>No.07           No.07           No.07</td><td>(15400) (16400) (16400) (16400) (1640) (1640) (1640) (1640) (1640) (1640) (1640) (1640) (1640) (1640) (1640) (1640) (1640) (1640) (1640) (1640</td><td>2 - 25 - 25 - 25 - 25 - 25 - 25 - 25 -</td><td>April 2004/00     April 2</td><td></td><td>Vir servines     Vir servines</td><td>00 00 00 00 00 00 00 00 00 00</td><td></td><td>ID         ID           ID         &lt;</td><td>١٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠</td><td>OVERLANDIG NERVORCHMEN OF THE ARCHOR FOR THE CONCRETE BASES REFACE UNERDATE SEQUENTS OVERLANDIG OVERLANDIG OVERLANDIG THE ARCHORT FOR ARCHORT REFEACE CONCRETE BASES REFLACE</td><td>LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL</td><td>ACT APOLINE, ATT APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL</td></td<>	Mar balan bandar, 4 - 0 Barton, Mari Marine Marine Marine, 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1	Nutrice	No.07	(15400) (16400) (16400) (16400) (1640) (1640) (1640) (1640) (1640) (1640) (1640) (1640) (1640) (1640) (1640) (1640) (1640) (1640) (1640) (1640	2 - 25 - 25 - 25 - 25 - 25 - 25 - 25 -	April 2004/00     April 2		Vir servines	00 00 00 00 00 00 00 00 00 00		ID         ID           ID         <	١٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠	OVERLANDIG NERVORCHMEN OF THE ARCHOR FOR THE CONCRETE BASES REFACE UNERDATE SEQUENTS OVERLANDIG OVERLANDIG OVERLANDIG THE ARCHORT FOR ARCHORT REFEACE CONCRETE BASES REFLACE	LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL	ACT APOLINE, ATT APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL
90 7143476		878246451 2002/01/21 2002/01	- 21 ANN - 21 A	4. 2009 3. 2009 3. 2009 3. 2009 3. 2009 4. 2009 4. 2009 4. 2009 4. 2009 4. 2009 4. 2009 4. 2009 4. 2009		Burdent           Hardent           Hardent <td< td=""><td>Mar balan bandar, da e de barbar harris <b>Barbar Barbar Harris</b> <b>Barbar H</b></td><td>Nutrice           Autority           Autority           Autority           Autority           Autority           Autory           Autory           Autory           Autory           Autory           Autory           Autory           Autory           Autory           Autory</td><td></td><td>(researd (researd) (resear</td><td>2.40 2.40 2.40 2.40 2.40 2.40 2.40 2.40 2.40 2.40 2.40 2.40 2.40 2.40 2.40</td><td>EV INCOME     EVICE      EVICE     EVICE      EVICE</td><td>40 </td><td>Ye (1998)     Ye (1998)</td><td>66 97 96 96 96 96 96 96 96 96 96 96</td><td></td><td>05         02           0.0         0.0           0.0         0.0           0.0         0.0           0.0         0.0           0.0         0.0           0.0         0.0           0.0         0.0           0.0         0.0           0.0         0.0           0.0         0.0           0.0         0.0</td><td>Miles and a second seco</td><td>OVIRIGANES REIN-ORCHMIST OF THE ADDRESS CREATER AND REPACE LIFELINES OF THE MARTS REPACE LIFELINES OF THE MARTS REPACE STRUCTURE OF THE ADDRESS OF THE ADDRE</td><td>NOT SPEAKED</td><td>NOT INFORMED NOT INFORMATION NOT INFORMATION NOT INFORMATION NOT INFORMATION NOT INFORMED NOT INFORMED NOT INFORMED</td></td<>	Mar balan bandar, da e de barbar harris <b>Barbar Barbar Harris</b> <b>Barbar H</b>	Nutrice           Autority           Autority           Autority           Autority           Autority           Autory		(researd (researd) (resear	2.40 2.40 2.40 2.40 2.40 2.40 2.40 2.40 2.40 2.40 2.40 2.40 2.40 2.40 2.40	EV INCOME     EVICE      EVICE     EVICE      EVICE	40 	Ye (1998)	66 97 96 96 96 96 96 96 96 96 96 96		05         02           0.0         0.0           0.0         0.0           0.0         0.0           0.0         0.0           0.0         0.0           0.0         0.0           0.0         0.0           0.0         0.0           0.0         0.0           0.0         0.0           0.0         0.0	Miles and a second seco	OVIRIGANES REIN-ORCHMIST OF THE ADDRESS CREATER AND REPACE LIFELINES OF THE MARTS REPACE LIFELINES OF THE MARTS REPACE STRUCTURE OF THE ADDRESS OF THE ADDRE	NOT SPEAKED	NOT INFORMED NOT INFORMATION NOT INFORMATION NOT INFORMATION NOT INFORMATION NOT INFORMED NOT INFORMED NOT INFORMED
90 7343475 91 7343477 92 7342534		87824663 87824663 8782467 878247 878258 87858 877858 8778		40.000 40.000		Burdent           Hardent           Hardent <td< td=""><td>Mar balan bandar, 4 - 0 Barton, Mari Marine Marine Marine, 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1</td><td>Nutrice           Nutrice           Nutrice</td><td>No.07           No.07           No.07</td><td>(15400) (1640) (16400)</td><td>2 - 25 - 25 - 25 - 25 - 25 - 25 - 25 -</td><td>April 1995/991     April 19</td><td></td><td>Vir servines     Vir servines</td><td>00 00 00 00 00 00 00 00 00 00</td><td></td><td>ID         ID           ID         &lt;</td><td>١٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠</td><td>OVIRIGANES REIN-ORCHMIST OF THE ADDRESS CREATER AND REPACE LIFELINES OF THE MARTS REPACE LIFELINES OF THE MARTS REPACE STRUCTURE OF THE ADDRESS OF THE ADDRE</td><td>LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL</td><td>ACT APOLINE, ATT APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL</td></td<>	Mar balan bandar, 4 - 0 Barton, Mari Marine Marine Marine, 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1	Nutrice	No.07	(15400) (1640) (16400)	2 - 25 - 25 - 25 - 25 - 25 - 25 - 25 -	April 1995/991     April 19		Vir servines	00 00 00 00 00 00 00 00 00 00		ID         ID           ID         <	١٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠٠	OVIRIGANES REIN-ORCHMIST OF THE ADDRESS CREATER AND REPACE LIFELINES OF THE MARTS REPACE LIFELINES OF THE MARTS REPACE STRUCTURE OF THE ADDRESS OF THE ADDRE	LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL LITINGAL	ACT APOLINE, ATT APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL APOLICAL
10         73.63676           11         73.63677           12         73.63567           13         73.63564           14         73.63564           14         73.63556		могран мос мос мос мос мос мо мос мо мос мо мо мо мо мо мо мо мо мо м мо м м м м м м м м м м м м м																			
  | 4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000  |  | 1          
   | An in both the sample, is i - is in terms to have in the same in the same in the same interval of the same interva  
   | Notice  |   | (1440)<br>(1440)<br>(1440)<br>(1440)<br>(1440)<br>(1440)<br>(1440)<br>(1440)<br>(1440)<br>(1440)<br>(1440)<br>(1440)<br>(1440)<br>(1440)<br>(1440)<br>(1440)<br>(1440)<br>(1440)<br>(1440)  | 2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00  
  |   |  |  |  |  | IN         IN           00         00           02         02           02         02           02         02           02         02           02         02           02         02           02         02           02         02           02         02           02         02           02         02           02         02           02         02           02         02           02         02           02         02           02         02   | Mail and a second  |
OVERAGESE<br>STATUSES<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE  | NT 200400<br>NT 2004000<br>NT 2004000<br>NT 2004000<br>NT 20040000<br>NT 20040000<br>NT 2004000<br>NT 2004000<br>NT 2004000000   |   |
| 10         71.63676           91         73.63677           92         73.63656           93         73.63556  |   | Anarona<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>Makana<br>M  |   
  | 4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847<br>4.4.4847 4.4.4847<br>4.4.4847<br>4.4.4847 4.4.4847<br>4.4.4847<br>4.4.4847 4.4.4847<br>4.4.4847 4.4.4847 4.4.4847 4.4.4847 4.4.4847 4.4.4847 4.4.48   |  | 1           
  | An in house bank, by . 0, is house, hour<br>in the second second second second second second<br>An in the second second second second second second<br>An in the second second second second second second second<br>An in the second second second second second second second<br>An in the second  
  | Notice  |   | (1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-real)<br>(1-   |
2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00<br>2.00 |   |  |  |  |  | IN         IN           ASI         ASI   | Mail and a second  | Original Sector  | ытаюне<br>на<br>на<br>на<br>на<br>на<br>на<br>на<br>на<br>на<br>на  
  |   |
| 10         73.63676           11         73.63677           12         73.63567           13         73.63564           14         73.63564           14         73.63556  |   | аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>алан<br>алан<br>алан<br>алан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>ал<br>алан<br>ал<br>ал<br>ал<br>ал<br>ал<br>ал<br>ал<br>ал<br>ал<br>ал<br>ал<br>ал<br>ал  |   
  | 4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000   |  |  
   | Mar bin hand an species, or a discovery of a second and a   
  | Notice  | Not           Not           Market           Market           Note  | (19990)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(1990)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)       | 3.03 3.03 3.00 3.00 3.00 3.00 3.00 3.00  
   |   |  |  |  |  | IN         IN           925         60           925         60           925         60           925         805           805         805           805         805           805         805           805         805  |  |
OVERAGESE<br>STATUSES<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE<br>PORTE  | L'I SIANE<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2   |   |
| 10         73.63676           11         73.63677           12         73.63567           13         73.63564           14         73.63564           14         73.63556  |   | а. Бания<br>Бания<br>Бания<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима   |   
  | 4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000   |  | 1000           2000  
   | Mon below bench, or - 0 bences. More  
   | Notice  |   | 8  0  0  0  0  0  0  0  0  0  0  0  0  0  
   |  |   |  |  |  |  |   |  
   | OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPE  | ыт акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана  |   
   |
| 10         73.63676           11         73.63677           12         73.63564           13         73.63564           14         73.63564           14         73.63556  |   | аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>алан<br>алан<br>алан<br>алан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>аланан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>алан<br>ал<br>алан<br>ал<br>ал<br>ал<br>ал<br>ал<br>ал<br>ал<br>ал<br>ал<br>ал<br>ал<br>ал<br>ал  |   
  | 4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000   |  |  
   | Mar bin hand an species, or a discovery of a second and a   
  | Notice  | Not           Not           Mark           Mark           Not   | (19990)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(1990)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)<br>(19900)       | 3.03 3.03 3.00 3.00 3.00 3.00 3.00 3.00  
   |   |  |  |  |  | IN         IN           925         60           925         60           925         60           925         805           805         805           805         805           805         805           805         805  |  | Original Sectors of Control of Co              
  | L'I SIANE<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2005<br>2   |   |
| 80         75.63679           91         75.63679           92         75.63536           93         75.63536           94         75.63536           95         75.63536           94         75.63536           95         75.63536           94         75.63536           94         75.63536  |   | Полнования   |  |
4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.0000<br>4.0000<br>4.0000<br>4.0000<br>4.0000<br>4.0000<br>4.0000<br>4.0000<br>4.0000     |  | 1998            
   | Montherman (Montherman (Montherman))<br>Montherman (Montherman)<br>Montherman (Montherman)<br>Montherm  | Notice
  | Not           Weil           Weil           Weil           Weil           Notified   | 8 (101000) (  |
3.93<br>3.93<br>3.94<br>3.94<br>3.94<br>3.94<br>3.94<br>3.94<br>3.94<br>3.94<br>3.94<br>3.94<br>3.94<br>3.94<br>3.94<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95 |   |  |  |  |  | Imp         Imp           Imp |  |  
   |  |   |
| 10         73.63676           11         73.63677           12         73.63564           13         73.63564           14         73.63564           14         73.63556  |   | а. Бания<br>Бания<br>Бания<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима<br>Алима   |   
  | 4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000   |  | 1000           2000  
   | Mon below bench, or - 0 bences. More  
   | Notice  |   | 8  0  0  0  0  0  0  0  0  0  0  0  0  0  
   |  |   |  |  |  |  |   |  
   | OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPENANCE<br>OPE  | ыт акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана<br>акана  |   
   |
| 80         75.63679           91         75.63679           92         75.63536           93         75.63536           94         75.63536           95         75.63536           94         75.63536           95         75.63536           94         75.63536           94         75.63536  |   | Полнования   |  |
4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.0000<br>4.0000<br>4.0000<br>4.0000<br>4.0000<br>4.0000<br>4.0000<br>4.0000<br>4.0000     |  | 1998            
   | Mon below bench, or , i o bence. Mon i<br><b>Bench and Bench and Bench</b>  
   | Notice   | Not           Weil           Weil           Weil           Weil           Notified   | 8 (101000) (  |
2.93<br>2.93<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94 |   |  |  |  |  | Imp         Imp           Imp |  |  
   |  |   |
| 80         75.63679           91         75.63679           92         75.63536           93         75.63536           94         75.63536           95         75.63536           94         75.63536           95         75.63536           94         75.63536           94         75.63536  |   | Полнования   |  |
4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.0000<br>4.0000<br>4.0000<br>4.0000<br>4.0000<br>4.0000<br>4.0000<br>4.0000<br>4.0000     |  | 1998            
   | Mon below bench, or , i o bence. Mon i<br><b>Bench and Bench and Bench</b>  
   | Notice   | Not           Weil           Weil           Weil           Weil           Notified   | 8 (101000) (  |
3.93<br>3.93<br>3.94<br>3.94<br>3.94<br>3.94<br>3.94<br>3.94<br>3.94<br>3.94<br>3.94<br>3.94<br>3.94<br>3.94<br>3.94<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95<br>3.95 |   |  |  |  |  | Imp         Imp           Imp |  |  
   |  |   |
| Image: Constraint of the second of   |   | плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосо   |  |  
   |  | 1998            
   | An and an   
   | Notice                  |   | 8  1  1  1  1  1  1  1  1  1  1  1  1  1  |
3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00 |   |  |  |  |  |   |   
  |  |  |   |
| 40         71.63534           81         71.63534           82         71.63534           82         71.63534           94         71.63534           95         71.63534           94         71.63534           95         71.63534           94         71.63534           95         71.63534           96         71.64534           97         71.64534           98         71.64534  |   | Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>Пологар<br>По   |   
  | 4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.000<br>4.0000<br>4.0000<br>4.0000<br>4.0000<br>4.0000<br>4.0000<br>4.0000<br>4.0000<br>4 |  | 1000            
   | Month Marken Barry, N Ho Honson, Mark<br>Marken Marken, M Honson, M<br>Marken Mark, M. & Marken, M<br>Marken Mark, M. & Marken, M<br>Marken, M Marken, M<br>Marken, M Marken, M<br>Marken, M Marken, M<br>Marken, M<br>Marke  
  | Notice  | Notified         Notified   | 8  0  0  0  0  0  0  0  0  0  0  0  0  0  |
2.93<br>2.93<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94<br>2.94 |   |  |  |  |  |   |   
  |  | 1         201           1 <t< td=""><td></td></t<>   |   |
| Image: Second   |   | <ul> <li>Аллогод</li> <li>Аллогод</li></ul>  |  |  |  | 1998          
1998           1998           1998           1998           1998           1998           1998           1998           1998           1998           1998           1998           1998           1998  
   | An all hand and and a set of a  | Notice           Notice | - Mod           - Mod </td <td>*</td> <td>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00</td> <td></td> <td></td> <td></td> <td></td>
<td></td> <td></td> <td></td> <td>And and a set of the set of th</td> <td>1         2000000000000000000000000000000000000</td> <td></td> | *   | 3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00 |   |  |   
  |  |  |   |  | And and a set of the set of th  | 1         2000000000000000000000000000000000000  |  
  |
| Image: Control of the contro  |   | плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосов<br>плосо   |  |   
  |  | 1998             
  | An and an  
  | Notice                  |   | 8  1  1  1  1  1  1  1  1  1  1  1  1  1  |
3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00 |   |  |  |  |  |   |   
  |  |  |   |
| Image: Constraint of the second sec  |   | <ul> <li>Аллогод</li> <li>Аллогод</li></ul>  |  |  |  | 1998           1998        
  1998           1998           1998           1998           1998           1998           1998           1998           1998           1998           1998           1998           1998           1998           1998   
  | An all hand and and a set of a  | Notice           Notice | - Mod           - Mod </td <td>*</td> <td>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00<br/>3.00</td> <td></td> <td></td> <td></td>
<td></td> <td></td> <td></td> <td></td> <td>And and a set of the set of th</td> <td>1         2000000000000000000000000000000000000</td> <td></td> | *   | 3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00<br>3.00 |   |  |   
  |  |  |   |  | And and a set of the set of th  | 1         2000000000000000000000000000000000000  |  
  |
| Image: Constraint of the second sec  |   | плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>пл   |  |  |  | Image           Image </td <td>Mu bi hola anat, a 1, a to tota. Mu i<br/>I anatomic anatomic and a second and a second and a second<br/>A second and a second and a second and a second and a second<br/>A second and a second and</td> <td></td> <td></td> <td>(reed)     (reed)     (reed)</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>And and a set of the set of th</td> <td></td> <td></td>   | Mu bi hola anat, a 1, a to tota. Mu i<br>I anatomic anatomic and a second and a second and a second<br>A second and a second and a second and a second and a second<br>A second and a second and   |  |   | (reed)   |  |   |  |  |  |  |   |  | And and a set of the set of th  |  |   |
| Image: Second   |   | плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>плониза<br>пл   |  |  |  | Image           Image </td <td>Mu bi hola anat, a 1, a to tota. Mu i<br/>I anatomic anatomic and a second and a second and a second<br/>A second and a second and a second and a second and a second<br/>A second and a second and</td> <td></td> <td></td> <td>(reed)     (reed)     (reed)</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>And and a set of the set of th</td> <td></td> <td></td>   | Mu bi hola anat, a 1, a to tota. Mu i<br>I anatomic anatomic and a second and a second and a second<br>A second and a second and a second and a second and a second<br>A second and a second and   |  |   | (reed)   |  |   |  |  |  |  |   |  | And and a set of the set of th  |  |   |

8 Xeycode	Tanar Name	Nile 10	Geolo	ation	ted De		Address	Nia Tura - October	Structure	South Real	1984 Statistical (sol)	Capacity	Stratlare		Cleaning/Janifaring	Ousing/Urbanization	General Minuclare	Adaptations	Adaptations -	Lice Mature of Nile	nting Mislasi a' Une Mila's
	PG62 (SPURM SHIR, TORIS)	SPW9922818		-46.656297		Sau Paulo	NUR, SALTO, 79 - MORINA	Routhap	MAST	(3+0+0+0)	2.89	NOT INFORMED	Capacity (se <sup>4</sup> )	NOT INFORMED	NX	ox	NDK		ESTRUCTURE OVERLOADED REPLACE NED- FORCEMENT OF AMOUNTER FOR	(Ganeral)	En drummental Licenses
10 716245	PGU [SPERK BING TOOL]	1PANOLISE	-22.575465	-64.656.797	5	Sao Paulo	RUA SALITO, 79 - MCENAL	Kono	HALF	(3+6+6+6)	1.8	NOT INCOME.	40	NOT INFORMED	NDK	a	NOK	PLATFORMING GIGAND	CONORITE BASES AND SEVERAL CRIZATED CONFORMATE	FELED	183
126 7142647	MEAL (SPRIMMING_TONS)	5P6/903901	-23.611	-46.668	59	Sao Paulo	Au. dos Interés, III - Indianópolis, São Paulo - SP	Reality	маат	(12+12+6)	8.714	NOT INFORMED	ND	NOT INFORMED	NOK	œ	NOK	PLATFORM/INED DISEMISED. MEEDS JANETORINE.	CURRECADED REPLACE RED- FORCEMENT OF AND/ORS FOR AND/SEVERAL OKEN/TED	PELED	wa
																			COMPONENTS. ESTRUCTURE OVERLOADED REPLACE RED- FORCEMENT OF		
127 7142649	HEAL (SPERMINE)	SPANNOICI	-23.60555	-66.66825	57	Sao Paulo	Alameda Inaperi, 2003 - Moeria - SP	Realtap	MAST	(3+3+3)	2.629	NOT INFORMED	ND	NOT INFORMED	ox	OK.	NOK	GENERAL SANTORING.	AND SEVERAL CONCRETE BASIES AND SEVERAL CORPORENTS.	LICENSED	48
101 7142650	ным (кримнікон_токкі)	5767053564	-23.59734	-16.66923	59	Sao Paulo	Rub Porfessor Matters Kantana, 100 - VS Nova Conceção	Reality	MAST	(6-49)	2,125 (0,829 + 1,30)	NOT INFORMED	ND	NOT INFORMED	ox	ox	лак	GENERAL MALTORING.	CVERLORDED REPLACE REP- FORCEMENT OF ANCHORS FOR CONCRETE BASES AND SEVERAL COSTATED	LICENSED	18.8
																		FORMAT CARDIT OVERTID +	COMPOSENTS. ESTRUCTURE OVERLOADED REPLACE RED- FORCEMENT OF		
129 7142669	VOIJ (SPAGIJISE_TOPO)	SPACECOM	-23.596901	-46.678679	57	Sao Paulo	Rus Fandesas, 479 - Taan Ris	Realtap	MAST	(6+6+6)	1.764	ACT INFORMED	ND	NOT INFORMED	NOK	OK.	NDK	EQUIPMENT CARENET CODEXTED + MANNENG LIGHTING DOES NOT WORK + LADDER DAMAGED	ANCHORS FOR CONUNATE BASES AND SEVERAL CREDATED COMPONENTS.	FILED	YES
110 7142725	VHL (SPARKIS27_TORS)	SPWM82827	-23.611674	-66.704826	59	Sao Paulo	Avenda Barla de Marte Mar, 75 - Via Maraniz	Realisp	MART	(4+4+6+2)	3.081	NOT INFORMED	ND	NOT INFORMED	ox	ox	NOK	GENERAL MACTORING.	OVERLOADED REPLACE RED- FORCEMENT OF ANDORSFOR CONCRETE BASES AND SEVERAL	LICENSARIA	NOT
																			COMPONENTS. ESTRUCTURE OVERLOADED REPLACE NEW-		
111 7142710	STAL (SPWSNOST_TORS)	SP45394257	-22.4956	-66.6255	57	Sao Paulo	Nue volucións de Pilon, 2708 - Santana, São Paulo - 59 Res Roberto Matodato de Paulas I VA - Chican Santa Ambero - 50	Rodtap	MAST	(2+2+2+6)	2.81	NOT INFORMED	ND ND	NOT INFORMED	NOK	œ	NOK	LIGHTNENG ROD DAWAGED	ANCHORS FOR CONCRETE BASES AND SEVERAL CREDATED COMPONENTS	LICENSED	188
113 7142781	Vies (spwsioks), tokke)	SPWSUH103	-23.625	-46.623	57	Sao Paulo	Nus Betwu Relaco 160 - Vila da Saúda, São Paulo - SP	Realize	MAST	(8+6+6)	3.061	NOT INFORMED	ND	NOT INFORMED	NOK	œ	NOK	EQUEPMENT CARENET ORDATED	STRUCTURE OVERLOADED REPLACE RED- FORCEMENT OF ANCHORS FOR CONCRETE BASES	LICENSARLE	NOT
114 7142812	SUE3 (SPWSIDEL17_TORM)	SP#SU4117	-23.612922	-46.628404	59	Sao Paulo	Av. Jakaquan, 970 – Saide – São Paulo	Reality	MAST	3	(2,72 + 1,66)	NOT INFORMED	ND	NOT INFORMED	NOK	œ	NOK	EENTIPEID STRUCTURAL PROBEMAS OF TH BALDING, OSDATID COMPONENTE, NEEDS AMENDANG.	MO SEVERAL CORDATED COMPONENTS.	FLED	188
115 7146288	777F [DFSTH0777_TORM]	DFSTM0.777	-16.02138889	-48.00944444	DF 32	Basia	QAENA 210, CONE. 1, CABA 17 - SANTA MANSA	Rodtap	MAST	5	1.451	NOT INFORMED	ΝD	NOT INFORMED	OK.	OK.	NOK	MEDIS ANTIONING. COLDATED COMPONENTS.	NDT EXTRUCTURE OVERLOADED REPLACE RED- FORCEMENT OF	NOT DECEMIN	NOT INFORMED
116 714400	OW (SPWCKIEJZ_TORNI)	SP80-0332	-23.47294	-46.66687	59	Sao Paulo	Av. Panda Pitto, 207 - Vi. Roca Cachoesina - Sila Pado	Realizy	MAST	(6+6+6)	(2,12+0,69+0,67)	ACT INFORMED	ND	NOT INFORMED	NOK	OK.	NOK	RATIONALING COLLATED.	ANDHORS FOR CONCRETE BASIES AND SEVERAL CONFORMENTS.	FLIED	48
117 7144426	STCI (SPWC03064_TOP98)	SPWC03064	-23.547	-46.658	59	Sau Paulo	Av. Angélica, 1683 - Santa Cecillo, São Paulo - SP	Reality	MAST	$\{\mathbf{i}_i, 2\mathbf{\hat{i}} + \mathbf{\hat{i}}_i, \mathbf{f}\}$	$\{1,423+0,455\}$	NOT INFORMED	ND	NOT INFORMED	NOK	OK.	мж	BQUEPHINT CARDIET COLDATED. MATCHINE.	ESTRUCTURE OVERLOADED REFLACE RED- FORCEMENT OF ANCHORS FOR CONORITE BASES AND SEVERAL ORDATED	LICENSED	ves
118 7144582	CEM (SPWSLISS2_TOPPS)	8PW503522	-23.60954	-96.45231	52	Sao Paulo	Avenda Indaolpole, 1182 - (Chile Kim Libaolé) - Pisvato Pauleta, Ma Pinda	Realtap	MAST	(3+3+3)	2.254	NOT INFORMED	ND	NOT INFORMED	ox	ox	NOK	GINERAL SANTORING.	CREATED COMPOSITIO COMPOSITION REID/FORCEMENT OF THE ANCHORS FOR THE CONCRETE BASES	NOT DECKMID	NOT INFORMED
119 7144619 120 7144642 121 7145243	5KVT (AMSKV3K26_TORM) 9521 [PERCE3073_TORM] 458 (SCPREORSE_TORM)	AMIRGHAN PERCENT/I SCPNIDLE	-2.843191667 -8.056138889 -27.58991	-58.21368333 -34.8725 -48.51513	AN PE SC	Silves Recte Floranopolis	Rub Roub Toldy, 256 Rub Brum Golymbi Record Any 200 An. Fradry Berynnicza, 587	Greatfield Roaftap Roaftap	SELF-SLIPPORT. MAST MAST	56.27 4 (2,5+2,5)	9.58 NOT INFORMED (0,86+0,9)	NOT INFORMED NOT INFORMED NOT INFORMED	N0 N0 N0	NOT INFORMED CLIMIK INSIN-WER NOT INFORMED	NDK OK OK	NDK OK OK	NDK OK OK	MARKING LEARTING IN PROCESS. MARKING LEARTING IN PVC - NIPLACE BY POP LEARTING IND PACE MERLICE IN PARTING AT THE MULLERAL MARKING IN MATTING AT THE MULLERAL AND NO NIET RATIONS AT THE MULLERAL	NDT NDT NDT	LICENSARIA LICENSARIA LICENSARIA	NDT YES NDT
122 7145269 1231 7145281 124 7145310 125 7145411	500 (SCPA50500 TORNE) 545 (SCPA50545, TORNE) 545 (SCPCP4564 TORNE) 7259 (PRCARCY3, TORNE) 305CPA50405, TORNE	SCPASSOS SCPASSOS SCPASSOS SCPASSOS PREAMORES	-27.5913 -26.46633 -26.96797 -21.29978 -27.592944	-48.5527 -53.59884 -50.54 -51.228725 -48.55444	SC SC FR	Plotanceck Niccela Ro Do Carpo Carbo	<ol> <li>Katewa Mana, Mei Castan Rus Bata Catana, (N° Par Tasberg, Cur, Casta Rus 2018 CARCER Hurrato, An Av. Ru Banco, 309</li> </ol>	Realized Greatfield Greatfield Realized	NAST BELF-SUPPORT. BELF-SUPPORT. SELF-SUPPORT.	18+4+6 50 50 50 50	( 822 + 0.22 + 1.84 ) 1.33 2.86 2.72 1.43	ACT INFORMED ACT INFORMED ACT INFORMED ACT INFORMED	50 50 2.72 50	MOT INFORMED NOT INFORMED C.A.W MOT INFORMED MOT INFORMED	OK OK OK NK	a a a a a	ak ak ak ak	AD ABLE PLATIONES AT THE MATE NOT SOT SOT SOT LIGHTER OR DO COLLETED +	MOT MOT MOT STRUCTURE WITH SENSER.	LICENSARUE LICENSARUE PLIED LICENSARUE	NOT NOT NOT IMPORTED NOT
126 7145688 127 7145688 128 7145687	276 (SCW3M0274 TOWS) 276 (SCW3M0274 TOWS) 211 (SCM50111 TOWS) 5(7/550340_TOWS)	SCHMODS SCHMODN SCHMODN SCHMODE	-27.802884 -26.817145 -27.61289811 -27.61175	-48.554444 -48.855321 -48.59412861 -48.5904567	50 50 50	WEREAUX Photocook	Ar. No Bacco, 300 Ecolo Grad. (N. B. al Calories, 117 – 87 – 2007 2006) R. al Calories, 117 – 87 – 2007 2006 Raia Meteori Picale, pr. 42	Rootap Greatfaid Rootap Rootap	POST POST PAAT	4 52 (6+6+6)	2.3 2.45 (0,38+0,38+0,38)	NOT INFORMED NOT INFORMED NOT INFORMED	ND ND ND	CAW MOT INFORMED NOT INFORMED	NDK NDK NDK	ox ox		LACOR DAVAGED	NOT NOT	LICENSARCE	NOT NOT NOT
130 7146168 131 7146296	221P (PRECZOV23 TORRE) Selip (PRETACIES_TORRE) osiap (PRETACIES_TORRE)	PRETADÁS	-25.52985 -25.4520555 -25.4520555	-54.2008 -49.2525 -49.2554225	n n n	Fail Do Souncu Curtiba	Rua Cado Nontrocenso, 202 Rua Adol Nontrocenso, 202 Rua X00201 Cualli, 417 Rua X0466, 43	Greatfuld Roaftap Roaftap	MART	25 (2+0)	2.88 2.38 (0,67 + 1,57)	NOT INFORMED NOT INFORMED	ND ND	NOT DEPOSITED NOT DEPOSITED INVASIL TELECOM	OK NOK NOK	OK OK	OK OK NDK	PLATFORMS CORDATED. NO 1D PLATE LIGHTNING ROD CRIZATED. NO 1D PLATE. LIGHTNING ROD CRIZATED. NO 1D PLATE. SKOUND ARCHINE AND GUY WREE AS	MOT NOT	UCIMINO PEUD	103 NOT INFORMED
133 7146357 134 7146423	734P (PRCTA0734_TORM) ARD (CEFLA0391_TORM)	PRETAD734 CIFLAD195	-25.4290278 -3.73354	-49.2824722	ря. С1	Curtiba Fortaixaa	Rus Vict. do Ro Bracca, 803 Avenda Victoriagian Soares, er Si	Realtap Realtap	MAST MAST	6 1.66	1.42	NOT INFORMED	ND ND	NOT INFORMED	NOK	ox ox	NDK NDK	WELL AS REET PLATFORM SISTEM COLDATED. LL ROD, GROUND ANCHORS, REET PLATFORM AND ACCESS LADORS ORIGATED. NO WARRING LIGHTISM AND RO TO TO PLATE. REET PLATFORM CRIZATED.	NDT NDT	FILID	NOT INFORMED
135 7146677 136 7147118 137 7147385	845. (FEBA3037 TORS) Th. (BA9304536, TORS) OE2 (BA251208, TORS)	PERVOJSTI BAPKDIŠTIS BACIŠLIJON	-8.123294 -12.29805 -12.73811	-24.88726 -38.06745 -38.46071	PS BA	Rechte Pedra Alexandre Candelas	BLE ADLS AZUGO 33 BELGAR PAIA BERURAS, SJM BAR ADLS AR BAR VIZA	Realities Greatified Greatified	SALT SELF-SUPPORT. SELF-SUPPORT.	18+4+61 30 18-1	507 DF08MED 28.8 9.26	NOT INFORMED NOT INFORMED NOT INFORMED	505 5/0 3	NOT INFORMED NOT INFORMED	NDK NDK OK	OK NOK OK	OK OK	LE ROD CARACIES NO WARNES LISURISMS NELOS EXTENDED JANETORIAL SERVICES + REMARCIONING NOT	NOT NOT STRUCTURE OVERLOADED	HLED LICENSARLE LICENSARLE	NOT INFORMED
138 7148702 139 7148941	VPH (MOSALINE_TORM) VER (30M00VS5325_TORM)	MOBALLINER XEPHERNSSI25	-22.12111	-41.05713	MG MG	Sao Lourenco Governador Valadares	Alameda das Orquistas, 385 Au. 36, 1157	Greenheid Reaftap	POINT MASE	8.2	0.79 (3,62 + 1,05 + 0,53 +1,81 +0,53 +1,81)	NOT INFORMED	ND ND	NOT INFORMED	OK NOK	ox ox	NDK DK	NO 12 PLATE, NO HART PLATFORM, NO WARKS LIGHTING, NO LIGHTING NGC, THERE IS NO FIRES AND NO LIATE. DOOR VANDALIZED. RAELDING SHOTING STRUCTURE, PROBLING, M LIGHTING RED, NO MATTING AND NO RES IN ATTINUA Y THE STRUCTURE.	NDT NDT	LICENSARLE	N2T N2T
140 7148988 141 7149040 142 7150705 143 7150705 143 7150984 144 7150904	4528 (RSPAR0651_TORR) 4882 (BACK14923 TORR) 0988 (REGT0314_TORR) 8288 (REGT0314_TORR) 8288 (REGT0314_TORR) 8388 (REGT0314_TORR)	5285PA80651 8ACX54593 838070354 30995170857 30503940629	-28.8783% -12.43817 -23.025 -22.700112 -28.82041	-51.189987 -22.45.875 -61.555 -51.792438 -51.27487	NA NO PR	Poto Nega Mara de Sal Nac Ro De Jacelo Salta Dacio Salta Dacio	NUA PROF. 30ÃO DE SCUSA FEBRIRO, 859 ME TA: - COPPLIED COSTA DO MARTE BODAD REGESTILARY MAN, 3867 - BANG GARCEDO MIL APERIDO ANYA FLUOR, 821 Rea VIDEO 1044, 627, CARDO	Kooftap Rooftap Greatfad Greatfad	MAST MAST MAST SELF-SUPFORT. SELF-SUPFORT.	1.77 5 10 41 30	2.87 5.42 1.088 1.25 5.08	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED	N0 N0 125 N0	NDT INFORMED CLEMAR NDT INFORMED NDT INFORMED NDT INFORMED	NOK NOK OK NOK	OK OK NXK NXK	OK NOX NOX NOX	LIGHTING ROD GREATED + WARANG LISHTING DOES NOT WORK WARANG LISHTING DOES NOT WORK MARKE EXTERNED INSTITUTE, SERVICE MEDIC EXTERNED INSTITUTE, SERVICE	NOT MOT NOT MOT	LICENSED LICENSER NOT DECEMBLE LICENSER LICENSER	VES ADT INFORMED ADT INFORMED NOT NOT
145 640232 145 648290 147 878929	51W (CH30052761) PBB (CH3004N72) TML-1	PRCPP2528 MI2FA5020 52MA5051543	-23.1832222 -21.76208 -2.532556	-52.6573256 -43.30952 -65.200329	PE NG	Comelio Procepio Juli De Forb São Luis	AV, NOSKA SIM-ERA DO ROCIO, MA Av. Baño do No Banco - 2288 Rai Grande, 6450	Kudhap Kudhap Kudhap	MAST MAST MAST	(5+6+5) (5+1,60+2) (3+3+3)	6.37 (2,6+0,67+0,74) (4,29+4,29+4,29)	NOT INFORMED	N/D N/D N/D	NDT INFORMED NDT INFORMED NDT INFORMED	NOK OK	OK OK	OK NDK NDK	ICOMPENT CARDET CODATES CAMPAS MATTORING, NO ED PATE DO ARET PATFORIE, AO PATTING AT THE BUILDING, NO LODITION ROD. COMPENING WITH COMPANIANT SUBME. MATTING CODATED, CAREAL MATTORING.	NDT NDT NDT	PEAD PEAD	NOT INFORMED
148 578305 149 657208 150 6463377 151 578327	OBE (CHE022540) 658P (CHE028527) THE (CHE024604) THE (CHE0231715)	PRCTADUR PRCTADUR MODIFIES	-1.722778 -25.4015555 -22.40487 -2.542556	-28.4986.511 -88.2787222 -45.42258 -46.238944	MG MG	Fortabus Curtilia Rajulia Nao Luis	ADDRESS LITE. 200 ANNUES RELVA SANCEN, UNIS R. Tayloi - Marie da Pé, Ros 3 R. Senador Vitonic Press, p² 1958 - Revu Arsena	Rodino Rodino Rodino Rodino	MAST MAST MAST	4 (3,5+4,5) (4+1,80+1,53) (5+5+5+5+5=5)	107 00/08/00 (1,57+0,79) (2,09+0,15+1) 4.89	NOT IMPORTED NOT IMPORTED NOT IMPORTED	N0 N0 N0	MOT INFORMED MOT INFORMED MOT INFORMED IMF CONSTRUÇÕES	OK NOK OK NOK	ox ox ox	OK NOK NOK	SOT LIGHTSNE ROD OKDATIO NO REST PLATFORM, SO PRITTING AT THE STRUCTURE. COMPONENTS ORDARIO. PLATFORM, SOCIOCIDENEL, IMATES NEID BITTER AND TRANS. SANTOSTAL.	MOT MOT MOT	HLED LICENSED LICENSARLE	NOT INFORMED VISS NOT NOT INFORMED
152 500408 153 7142284 154 511084 155 878726	2263 11N (849641174, 70898) 7962 79621	ALMCODINI BAPGULLIN ALCORDINI MORAVEDIR	-9.672061 -16.4511667 -9.162367 -18.82959	- 25. 745829 - 29.06/20278 - 27.358192 - 43.80725	AL BA AL 1900	Macelo Posto Seguro Ouro Bolnos Brio Hordonite	AV ANNE CHATEAUEDADD, 2018 HUA SÃO PECHO, 185 HUA CONDEL LUCINA, SIN LUCZ G, DE CARVALHO AV. DARA PEDRAS, 455	Rooftap Rooftap Greatfaid Rooftap	MAST MAST SELF-SUPPORT. MAST	(3+2+2) 16 40 (3+2.50+2.50)	0.75 3.31 5.15 ( 0.71 + 0.52 +0.49 )	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED	N0 N0 N0	MOT INFORMED MOT INFORMED EPHETAL MOT INFORMED	NDK OK OK	0X 0X 0X	OK OK NOK	NO WARNENG LESHTEN, NO LESHTENG ROD MEDO ANTOPING GENERAL JANTORING, NO LO PLATE, GENERAL MATTORING, NO LESHTENG ROD AND NO REST PLATION	NDT NDT NDT NDT	LICENSING LICENSING LICENSING	NOT IMOUNED 123 NOT IMOUNED NOT
156 658976 157 7146622 158 7145782	741 (2450557114) 0074 (58C8FL80074) 5239 (98C7A0523_10898)	RELANDING SICEPLADOM PRETADS23	-25.54889 -3.724462 -25.432987	- 56.82975 - 38.528741 - 48.299796	RS CS PR DF	tingustara Fortakca Curtiba	BR 472, XM Sandi Rua Banador Fungan, 649 Rua Pacific Ancietta, 2128	Greenfield Routhap Routhap	SELF-SUPPORT. MART MART	60 7.5 (6+2+6+2)	4.32 1.9736 4.53	NOT INFORMED NOT INFORMED NOT INFORMED	4.32 N/D N/D	LTDA NOT INFORMED NOT INFORMED	OK NDK NDK	OK OK OK	OK NDK OK	GENERAL SANTOKING. LACOR DAMAGED. NO VARIANS LENHTING. LIGHTE ROD COZDATED + SANTHDI LIGHTE DOES MOT SOCK + CLARK UP CARLES	NDT NDT NDT STRUCTURE	FELED FELED LICENSED	NOT INFORMED NOT INFORMED YES
159 7144325 140 7147346 341 7147345		DFEAM0754 DFSAM0754 DFSAM0753 S2REPA60850	-15.829 -15.87916667 -15.88166667 -33.105	-47.875 -48.10944444 -48.00951111 -51.2878	DP DP	Boola Boola Boola Pote Near	SCH techo 2 kcs 22 QK 411CONL 4, CHAA 34, SAMANAKAA QK607, CONL 8 (CANERATO 8) are chake atty	Roaftap Roaftap Roaftap Roaftap	MAST GUNID TUBULAR MOUNT MAST	(4+6+2,3) 17 17	4,290 (1,732+1,280+1,80) 1.45 1.451	NOT INFORMED NOT INFORMED NOT INFORMED	ND ND ND	Teleconnicações 3/A Metálicas Engenharia LTDA NOT INFORMED NOT INFORMED	OK OK NOK	ox ox ox	OK OK	NOT GENERAL INVETORING, COMPONENTIE WITH CONDUCTION SIGNS, GENERAL, CONFORMENTIES WITH CONDUCTION SIGNS,	NOT NOT	NOT DECEMIN NOT DECEMIN NOT DECEMIN	NOT INFORMED NOT INFORMED NOT INFORMED YES
142 7144450 141 532489 144 532890 145 533333	850 (X5MAD850_T0993) 2001 (XM3003M51) C3A1 (XM3003034) ITBB (XM3003724)	SPATURES SPATURES SPATURES SPATURES	-33,105 -23,4490555 -23,42997 -23,5061	-46.0906067	NA SP SP	Sao Paulo Sao Paulo Sao Paulo	RU, GUARAN, 4137 RUR HELMRED COFF, ANDA - VAR HANNIÅRERA RUA RED JURERAL IN 157 - PERUS. Av Pol Celestina Biumoud, 813 - Landa	Realitap Greatfield Greatfield Greatfield	POST SELF-SUPPORT. POST	6 43 60 40.6	2.887 12 4.6 2.85	NOT INFORMED NOT INFORMED NOT INFORMED	N0 N0 N0	NOT SPOKNED SECCOME NOT INFORMED SECCOME.	NOK OK OK	OK OK OK	NDK DK DK	PLATFORM/IKID OXIDATED NORE TORER BRACES DAMAGED. DIDATION SPOTE. GENERAL SANTORISE. ADVINCED COLDATION AT STREE FORT	MDT MDT MDT MDT	FILED NOT INFORMED FILED NOT INFORMED	NOT INFORMED YES
166 7141354 167 7141374	PEL2 (SPRC03717_T0996) VLML (SPSRO405,T0998)	SP8003717 SP5804605	-22.55527 -23.666259	-46.569175	59 59	Sao Paulo Sao Bernardo Do Campo	Rua Mirak Genak, 288 - Ngjenópols Rua Marechal Badogilo, 25 - Vila Moscolne	Roathap Roathap	MART	(3+3+3) (4+4+4+4+4+4+4)	2.61 4.719	NOT INFORMED	N0 N0	Telenort Eigeshara De Telecomicações 8/A Telenort Eigeshara De Telecomicações 8/A	NDK OK	ox ox	NDK NDK	EQUIPHENT CARDINET CELEMENTE + LIGHTNENS ROD DAMAGED GRARAE JANITORING, COMPONENTS WITH CODDRIDON BIONS.	STRUCTURE OVERLOADED STRUCTURE OVERLOADED	LICENSED	VIS NOT INFORMED
168 7142393 169 7142395	PEDS (SPECHEDE, TOPES) 3POR (SPECHEDE, TOPES) TTES (SPECTADE, TOPES)	SPWIRIEE SPWIRIEZ SPWIRIEZ	-23.51361 -23.58111 -23.50055556	-46.76555	59 59 59	Sao Paulo Sao Paulo	Rua Cannelhenn Annaral, 206 - Yaguari, 35a Paula - 59 Rua Hungus, 664 - Yardres, 35a Paula - 59 Run Gan, Gaulis, 72 - TATUARI, 53G Multo-597 - 02080-033	Roathap Roathap Roathap	MAST	(3+3+3) (4+4+6) (4+4+6+6+6)	3.423	NOT INFORMED	ND ND	Telenort Eigenhala De Teleconincações 8/A Telenort Eigenhala De Teleconincações 8/A	ox ox	ox ox	NDK NDK	COMPONENTS WITH CRODUTENESION COMPONENTS WITH CREDUTENESION COMPONENTS WITH CREDUTENES	CVERLOADED STRUCTURE CVERLOADED STRUCTURE	PILIED LICENSED	NOT INFORMED VES
175 7142401 171 7142498 172 7142597	TTE3 (SPWTAL26_TOPE) REAL (SPWERC2_TOPE) SCAI (SPWSCROOL_TOPE)	SPWTA4136 SPWRA052 SPWSC034	-23.535556 -23.537794 -23.53388	-66.55956667 -65.636368 -66.66055	59 59 59	Sao Paulo Sao Paulo Sao Paulo	RAN GBN: GBLÁS, 72 – TATUMVÉ, SÁO PALLO-87 – GONO-GOO Ruis Westington Law, 282 – Centro Hendrica – Sán Paulo Pinga Júlo Pentine, 20 - Canpos Elénos, Sán Paulo - SP	Roaftap Roaftap Roaftap	MAST MAST MAST	(4+6+6+6) (4+6+6) (2+3+2)	5.014 2.377 3.4 N	NOT INFORMED NOT INFORMED NOT INFORMED	N0 N0 N0	NOT INFORMED NOT INFORMED Teleform Expendiana De Teleformica (De 13/A	OK OK NOK	ox ox ox	NOK NOK	WITH CREDITION REDAY. GENERAL BACTORING, COMPOSING WITH CREDITION REDAY. EQUIPM, CREDITION REDAY.	OVERLOADED STRUCTURE OVERLOADED NOT	LICENSED LICENSED	vas vas
173 7142672 174 7142778 175 7143807	NERT (SPURMISOT_TORIS)	SPANNOST	-23.634	-46.662	57 57	Sao Paulo	Alameda dos Tupesquese, 1225 - Piletalta Piluteta Red Folduar Couteño, 1990 - Piluteta - Sila Piluto - SP	Realize	MAST	(5+5+5) (3+3+2)	18%	NOT INFORMED	ND ND	NOT INFORMED	ox ox	ox ox	NDK OK	COMPONENTS WITH CREARSON SECRE	STRUCTURE OVERLOADED. REPLACE THE LIFELINES OF THE MASTS MOT	LICENSARLE	NOT
175 7143827 175 7144205 177 7144283	VLB (SPWHHZSH_TOHS) CEE (SPWHRSH_TOHS) P31 (SPWLRDH_TOHS)	SPWYM2564 SPWVM3099 SPWL80089	-23.6016 -23.575 -23.57664	-45.66842 -42.653 -46.632723	59 59 59	Sao Paulo Sao Paulo	Nai Kharbu, 763 Av. Brgadero Luis Atlâns, 2019 - Jacim Paulita Rai Topázio, 420 - Vili Mastria	Realtap Realtap Realtap	MAST MAST MAST	(2+2+2+2) (6+2) (4+4+1+2)	4.26 1.259 2.023	NOT INFORMED NOT INFORMED	ND ND ND	MOT DEPORTED	OK NOK OK	ox ox	OK NDK NDK	METHI GALENTON SEARS. GINERAL JANITORING, COMPONENTS METHI GALENTON SEARS. GINERAL JANITORING, COMPONENTS METHICADE WAS AND	NOT STRUCTURE OVERLOADED STRUCTURE OVERLOADED	LICENSED	vas vas vas
178 521363 179 521528	TNL (CM20006487) 8000 (CM20006220)	PARPENDIN	-2.55438889	-42.44694444	PA	Spoura Do Para	Rus Sargerto Simplica, 272 Rus Sargerto Atelina	Greathed	SELF-SLIPPORT.	16.79	26.92	NOT INFORMED	ND ND	NOT INFORMED	OK NDK	OK NDK	OK NOK	WITH COORDITION SOME. "MANNING LIGHTING INCREATIVE, NO ID PLATE, COMPOSITING WITH CORDITION SCORE. NEEDS EXTENDED INATIONAL SERVICES. WANNING LIGHTING INCREATED PLATE, COMPOSITING WITH CORDITION	NOT	LICENSARLE PELED	NOT INFORMED
180 7144021 181 660251	0PMA [PABLPULIS_TOPPS] 110P [CH10010348]	PARLAUISE PRETADUID	-1.43722222 -25.432481	-48.44527778 -49.362761	РА РК	Brien Curtilia	Ro. Tanànés Biolog, 632 - Manantana Rol. IV 277 Cia - Orleans / Rol. di Carli - Calade Industrial De Cartilla	Greenfield Reaftag	POST MAST	27.86 (2+2)	6.81 (1,64+1,91)	NOT INFORMED NOT INFORMED	6 N/D	Belinevia NOT INFORMED	OK NDK	ox ox	ак	SIZAN CONFORMETS WITH OXIDITIEN SEINS BOLEPHENT CARENET COLDATED. WANTERE LIGHTER SUCHARTER.	STRUCTURE OVERLOADED NOT	PELED	NOT INFORMED
182 660380 181 7146242 184 7146243 185 533008	1439 [CM80033167] 4221 [PRCTA0422 TORRE] 4221 [PRCTA0423 TORRE] 3P05 [CM80033276]	PRCTA0143 PRCTA0122 PRCTA0123 SPURA5117	-25.40077867 -25.4019749 -25.40197418 -22.54296	-05.25854283 -48.27251823 -05.29525188 -07.38998	н. н. 12	Curtiba Curtiba Curtiba Limena	Rub da Palo, 202 Rub Dr. Malor Dr. Malor, 450 Rub Dr. Manoel Landes, B. M. Caylor Roco Rub Dr. José Canadia Penesa, 4/0	Routhap Routhap Routhap Greatfield	MAST MAST MAST SELF-SUPPORT.	5 (3+3+3) (3+3+3) 40	2.38 (5.69+0.52+0.69) (5.73+0.73) 2.09	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED	N0 N0 N0	NOT INFORMED MOT INFORMED NOT INFORMED NORE SERVICE Trendentics	NDK NDK NDK OK	OK OK OK	OK OK NOK	EXCEMENT CARDINE CODATED + MARTIN LIBERTING DOES HOT WORK + GAMPE RAL LADDER DAVINED BOLEPHAT CARDINE CODATED BOLEPHAT CARDINE CODATED BOLEPHAT CARDINE CODATED BOLEPHAT CARDINE CODATED	NDT MDT MDT NDT	PELIED PELIED LICENSIED LICENSIABLE	NOT INFORMED NOT INFORMED YES NOT INFORMED
185 533295 187 7143271 188 7143271 189 714327 189 7143357 190 7143527	306 (043033938) COOI (596C03711,70908) COOI (596C03725,70908) Helis (596C03721,70908) CTOS (5950464,70908)	SPACESIS SPACOFIS SPACOFIS SPACOFIS	-21.54134722 -21.552 -22.552 -22.5413 -22.1928	-66.75434722	9 9 9 9	Sao Paulo Sao Paulo Sao Paulo Sao Paulo Jundar	Rub Cardindolle, 910 - Jaguari Rub di Carolingky, 2451 Sab Bill Control, 450 An. Angkina , Nr 1189 - Constructio Rub And Station	Rodtop Rodtop Rodtop Rodtop Rodtop	HAST HAST HAST HAST	(3+3+3) (3+3+3) (3+2+3) (3+5+3) (3+5+3) (4+4+6+6+6+6+6+6)	(2,88+0,67+0,67) (1,28+1,38+1,18) (2,58+1,12+0,781 (1,48+0,55+0,55) (0,72+0,75+1,81+0,21+0,75+ (0,72+0,75+1,81+0,21+0,75+	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED	N0 N0 N0 N0	NUT SERVICE Television Loss MOT DE ORRED MOT DE ORRED MOT DE ORRED MOT DE ORRED	OK OK NXK OK NXK	OK DK DK OK	NDK DK NDK NDK	NOT NOT EQUIPMENT CARENT CREATED. EQUIPMENT CARENT CONDUCTO. LADORE DEPOSIT. NO FLATE INVEST WALL	NDT SDT NDT NDT	LICENSABLE LICENSED LICENSED LICENSABLE	TCK 887 887 887 887 887 704
140 7140808 141 7140808 142 533003 143 7141382	COL (PROVINCION) COL (PROVINCION) BVID (CHIODIDIZTI) CHI (PROCIDIZTICAR)	SPSAD454 SPST35174 SPWRV2596 SPWCV2637	-23.9542 -23.55802 -23.5638	-46.224 -46.64291 -46.62271	9 9 9 9	Santos San Paulo San Paulo	Rua Audurtio Paulitos 37 RUA LOURIMOO GRANATO, 224 ANIMERA LACENDA FRANCO, 210, CEP: 1536-000	Rodhap Rodhap Rodhap	MAST MAST MAST	(4+4+4+2) (4+4+4+2)	0,50+2,41) 10.56+1.02+0.32+0.761 NOT 20FORMED NOT 20FORMED	NOT INFORMED NOT INFORMED NOT INFORMED	ND ND ND	Poto Securi MOT INFORMED NOT INFORMED	NCK NCK	ox ox	NDK NDK NDK	NAMES OF A CONTRACT OF A CONTR	MOT MOT MOT	HLED LICENSED LICENSED	NOT NOT NES VES
384 7141323 385 7140411	AMA2 (SPMAR2SS7_TOPR) BR21 (SPMB23612_TOPR) CAC1 (SPMB23612_TOPR)	SPANGISS7 SPW833512	-21.55969 -21.5425667	-66.57475 -66.60605556	59 59	Sao Paulo Sao Paulo	RUN TEOPILO DINA, 212 RUN XOSE PORTERO, 223 RUNNES CAMPACIAN DE LA CONTRACTA DI ANTONIO	Roathap Roathap	MAST MAST	(4+4+4) (4+4)	NOT DECEMID	NOT INFORMED	N0 N0	NOT INFORMED	NOK NOK	ox ox	ok ok	LADOR DAMAGED, OTHER COMPONENTS WITH COLDARIZON SEGNS, COMCRETE FLOOD BHOVES TAKING ON ANAL EQUIPMENT CARENET CHIDANIST CHIZANIST CHIDATED, OHINK COMPONENTS CRIZANIST IND WARKING LICHTING.	NDT NDT	LICENSED	783 783
149 7143770 197 7148679 198 7141871 199 7141871 200 7141873	CALL SPECTRE TORS OWN (CPTSON, TORS) CPAL (SPITSON, TORS) EAL (SPITSON TORS) EAL (SPITSON TORS) CALL (SPITSON TORS)	SPATT22H RPT500M SPSTS53B7 SPSTS52B SPSTS52B7	-21.5948 -22.3994 -21.9994 -21.957 -21.957	-64,4530 -61,1580 -66,3126 -64,2580 -64,2580 -64,2580	22 22 22 22 22 22 22 22 22 22 22 22 22	Sat Paulo Petropolis Santas Santas	ANNUA CHARANIA SUPECI CAMANILA DEL - ROSERA EL de USA SI ADADO, 2018 A ROSERA Ra Ala Cola, 355 Ra Bache Provingenda, 171 Ra Bache Provingenda, 171	Rooffield Greatfield Rooffield Rooffield Rooffield	POST POST MAST NAST	4 8.4 (3+3+3) (5+6+6+6) (3+4+6+0) (3+1+6+0)	NOT DECEMBLO 0.79 2.03 3.72 2.24 	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED	50 50 50 50	ACT INFORMED ACT INFORMED Puto Segue Engentaria Puto Segue Engentaria Puto Segue Engentaria	OK NDK OK NDK	ox NOS OX	NDK NDK NDK NDK	MEDI EXTINUED IMPORTAL EXAMPLE IMPORTAL COMPONENTS WITH OPTIMIZES STORE WITH STRELLOWTING COSES MOT WORK COMPONENTS SHORE COSPORTS	MOT MOT MOT MOT	LICENSED LICENSABLE FILED FILED FILED FILED	NOT NOT NOT INFORMED NOT INFORMED
201 7140908 202 7140936 203 7140938 204 7140938 205 7140409 205 7140409	CTO2 (SPST35398_TORS) ENC1 (SPST35406_TORS) CGA2 (SPST35408_TORS) PA2 (SPA194811_TORS) 2911 (SPA39345_TORS)	SPSTS5200 SPSTS5426 SPSTS5428 SPA0P2820 SPA0P2845	-23.82542 -23.86483 -23.86483 -23.86483 -23.56588	-46.33177 -46.3299 -46.34093 -46.560963 -46.655097	57 57 57 57 57	Santas Santas Santas Santas Santas Santas Santas	Piaça dos Andriadas, 92,193 Rua DX. Latos das Andriadas, 93,00 Rua Nocionada de Piana, 200 Rua Stock Rosta Latos, 1887 Rua Piangtano, 225 - Jacobe Piantza	Roaftap Roaftap Roaftap Roaftap Roaftap	MAST MAST MAST ROOT MAST	(3+2) (3+3+2+2) (3+2+3) (8+8) (6+4)	2.16 2.4 2.15 1.36	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED	N0 N0 N0 N0	Puto Seguro Ergenharo Puto Seguro Ergenharo Puto Seguro Ergenharo NOT DEPORTAD NOT DEPORTAD	NOK NOK OK NOK	OK OK OK OK	NDK NDK OK NDK	SHOW CODATION LADDER DEMICIE + CELLINE INFUTATION COMPONENTS WITH CREATION ANNU. NOT CEMPONENTS UNIVERSITY ANNU CREATION CEMPONENTS LIGHTING DOES NOT WORK COMPONENT	NDT NDT NDT NDT NDT	PELED PELED LICENNED LICENNED	NOT INFORMED NOT INFORMED ACT INFORMED WES VES
201 7142443 202 7142443 202 7147858 208 659821 209 645960 200 647967	CE3 (SPWPZ2957 10898)	SPREPZIES S265PA60870 ESCAND115 MICOPA601 ROPA5022	-21 515285 -23 515285 -33 148231 -15 5345 -25 39347222 -21 754453	-45.692718 -51.34334 -45.42734 -45.42734 -43.4225 -43.5536	17 13 15 15 15 15 15 15 15 15 15 15 15 15 15	Sal Paulo Roto Negre Cristina Ouro Parto Nego Parto	Next Program, Karl - March Product Sila Banda ACISSIO D (2) 31. UNRIDARE VICENA, 3246 ANIMENA CHILLIAN VINANA, 504 ANIMENA CHILLIAN VINANA, 504 MAL CO CHILLIAN 2, 248 ANIMENA DO SEE BRANCE J.1451	Kontop Kontop Kontop Kontop Kontop	MAST MAST MAST	(4+4) (3+2+3) 4 (6,7+6,7) 20 (4+4)	1.98 1.93 NOT DEFORMED 0.48 1.492 2.24	ACT IMODANED NOT IMODANED NOT IMPORATE NOT IMPORATE NOT IMPORATE	20 20 20 20 20 20	NOT DECEMED Instances Encodera NOT DECEMED NOT DECEMED NOT DECEMED	NX NX NX OX	888	2 X X X X X X X X X X X X X X X X X X X	CONFIGURE & WITH COORDENIESS. BOT GENERAL CREATED NO ID PLATE. LADGER DAMIGED ND LIGHTNING ROD. NO LIGHTNING ROD. ND READERS LIGHTNING 2017	MDT NOT NOT NOT	LICENSIED LICENSIALS PELED FELED FELED	NOT NOT NOT NOT INFORMADO NOT INFORMED
212 647947 211 7142492 212 7142513 213 7142567 214 7142792	CED (CHIDOMASS) 30L1 (IPHOQ3807_TORM) CRID (IPHWT603 TORM) CRID (IPHWT603 TORM) 35A2 (IPHWHC823_TORM)	9029 A5022 SPW3Q0827 SPW87 6003 SPW5C6035 SPW982823	-21.54481 -21.647357 -21.585325 -21.592184 -21.592184	-41.3110 -46.611571 -65.7100118 -45.6009 -45.6009	90 97 97 97 97 97 97 97 97 97 97 97 97 97	Sao Paulo Sao Paulo Sao Paulo Sao Paulo	Antiona do Seo Hanara, 1441 Rua da Gurunanou, 1315 - Sucien Constal Rua Calculato Funness, 481 Seo Banaratas, 1317 Au, Lueu de Paula Machado, 1139 Inicion Evenett	Realize Realize Realize Realize Realize	MAST MAST MAST MAST MAST	(4+0) (5+5+5) (3+4+4) (3+2+2) (5+6+5)	2.24 (0,81 + 0,93 + 0,90) (0,25 + 0,41 + 0,25) (1,22 + 1,22 + 0,82) (1,64 + 1,39 + 0,68)	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED	80 N0 80 N0	ADT BEOMAD Zopore Engelharia MOT SECOMED MOT DECOMED Tel Telconancações	NK NK NK NK	ox ox ox	NX NX NX NX	WAMPEND LIGHTING SCER NOT WORK + LIGTHNENG ROD DAVISIO. PLATION (SKO DAVISIO). CONFIGNITIS US TO OCCUTATE AMERICA LIGHTING ROD DAVISIO. LIGHTHNENG ROD DAMAGED.	MDT MDT MDT MDT	ADD BEGRMED	ACT INFORMED 103 103 103 103 103 103 103 103
215 7142746 216 7142748 217 7142785	DAL (SPARLINE, TORS) DAL (SPARLINE, TORS) DAL (SPARLINE TORS)	SPACIO248 SPACIO255 SPACIO255	-23.577905 -23.582762 -23.6229056	-46.612134 -46.607048 -46.60904722	519 519	Sao Paulo Sao Paulo	Na Vesconsto District, 196, Ipranga Na Costa Aguar, 991 - Ipranga Ar. Educationa. 235	Rodtap Rodtap Rodtap	MAST MAST MAST	(0+0+0) B (0+0+0)	(0,96 + 0,75 + 0,75 ) 2.7 (0.83 + 0.83 + 1.33)	NOT INFORMED NOT INFORMED NOT INFORMED	ND ND	NOT INFORMED NOT INFORMED	ox NOK OX	ox ox	OK NOK	COMPONENT AT A COLLECCIONOLS. NACIONAL COMPONENTS WITH CALENTRY SEAS. MANAGED LIGHTING DOES NOT HODE: A LIGHTING AND AND HODE CALOO COMPONENTS WITH CALENTRY SEAS. 2010	NDT NDT NDT	LICENSED LICENSED	183 183
217 7140785 218 714449 219 7144581 220 7144581 221 7144581 221 7144620 222 7144681	CID (SPARCING) TORIS CID (SPARCING) TORIS (MO2 (SPARCING) TORIS CTO (SPARCING) TORIS CTO (SPARCING) TORIS COS (SPARCING) TORIS	594261153 595753213 595243100 5942901000 594392067	-21,21879 -21,86965 -21,56812858 -21,609865 -21,56194444	-32.43917 -46.33925 -66.31464501 -46.660513	2222222	Santas Santas Sacano Sac Paulo	Rai Warcesho Ku Destadoka 1985 - 1985 - Nikata Maria Anatola Presidera Wala Desta Raina Bali Berjana Wala Ka José Panya Bali Berjana (Constant, 1983) Marenda Gal Anatonia (PPP), Mantu Alamada Taridi, 2011 - Indon Madaa	Realise Realise Realise Realise Realise Realise	MAST MAST MAST MAST MAST	(3,20+4,22+3,20) (4+4+6) (4+4+6) (4+4+6) (4+4+6) (4+4+6) (4+4+6)	(1,04 + 1,04 + 0,05) (2,76 + 0,76 + 1,32) (2,76 + 1,40 + 2,64 + 0,81 + 2,01) (2,57 + 0,68 + 1,02) (2,67 + 0,67 + 0,84)	NOT INCOME NOT INFORMED NOT INFORMED NOT INFORMED	ND ND ND ND ND	ADT INFORMULT The Treasment (dec Parto Tecuro Tronham Halanti ADT INFORMED ADT SPORMED	OK NOK OK OK	а а а	90X 90X 90X 90X 90X	COMPONENTS WITH CREMINER SEAS.	MDT NDT NDT	FILE LICENSARLE NOT INFORMED LICENSED	133 807 19/00/800 193 193
221 581225 224 552968 225 7142113 226 7142180	884F (CH4002644) 1294 (CH60016938) 1352 (BASQ1229, TOWS) 5466 (BAPPACER, TOPO)	PTEAQUITZ MALEALIZHA BACEQUITZH BAPPACIZHI	-6.587028 -14.328056 -11.5297778 -14.94391667		PI BA BA	Tangar Do Plau Ubarata Jaguaguan Prepa	MC. Prescipal, 4/10 PERA DO CACASA SIN MAIA SE DE MONTRARIO, 5/10 PEACE, SOLAQUIR ZOSA PERBERO, 5/10 PEACE, SOLAQUIR ZOSA PERBERO, 5/10 PEAR PERBERO, 5/10 (CALSA	Greatfield Greatfield Greatfield Greatfield	SELF-SUPPORT. SELF-SUPPORT. SELF-SUPPORT. SELF-SUPPORT.	40 50 50	2.36 MOT SPORMED MOT SPORMED 28.03	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED	ND ND 18.07 28.03	EMITAL DICLETRIAL METALURIZA LTCA. BRASSAT HARLD S.A. MOT DA DRIED DAN HERBIT SAT \$/A	OK NOK OK		ak Max ak	CENERAL ENTERIES AND SUBSTITUTION OF THE RANNING LOW-THAT NEEDS ESTEMBED DARIOGENE ENVIRON ENTERIES DATED AND TO SHARE UNREAS LOW-TO SHARE CENERAL INSTITUTION	NDT NDT NDT NDT NDT	LICENSARLE LICENSARLE LICENSARLE LICENSARLE	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED
227 7142382 228 7142382 229 7142382 229 7142329 230 878865 231 7143828 232 714609	NAP (BAPOULLY2_TORNS) 1112 (BARCK1312_TORNS) 22 (BRCMA022_TORNS) 0108-2 000 (PAEX02214_TORNS) 0FE (PAEX02214_TORNS)	BAPGU1173 BASUX1112 REFERENCE2 S2MASLS1512 PARCO22M PARL90115	-16.4294444 -10.943083 -26.35401 -2.522222 -1.20117 -1.209385	- 29.56722222 - 40.407972 - 50.92508 - 46.323511 - 48.22611 - 48.29138 - 48.29138	8A 83 9A 7A 7A	Poto Seguo Saulo Canto São Luis Brian Brian	Anterna (and constraint) Anterna (and constraint) Anterna (and constraint) Anterna (and constraint) Anterna (anterna (anterna (anterna)) Anterna (anterna (anterna (anterna)) Anterna (anterna (anterna)) Anterna (anterna (anterna)) Anterna (anterna (anterna)) Anterna (anterna)	Greenfield Greenfield Rooft so Rooft sp Greenfield Greenfield	NAT SUP SUPPORT	30 40 (1+2+3) 40 40	NOT INFORMED 2.08 2.09 NOT INFORMED NOT INFORMED NOT INFORMED	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED	N0 2.08 50 50 50 50 50	NOT INFORMED SECCOM. MOT INFORMED MOTA EXCENANTA BELMERTZ BELMERTZ	OK DK NDK DK OK	0X 0X 0X 0X	OK DK NOK DK	CAMBREL DANTONING NOT ECOSPHENT CAMPAT CRIDATED PLATFORM/SKID CRIDATED NOT	MDT MDT MDT MDT MDT MDT	LICENSED LICENSERIE LICENSERIE FILED FILED FILED	YES ACT INFORMED ACT INFORMED ACT INFORMED ACT INFORMED ACT INFORMED
233 7144034 234 7144035 235 7147345	2284 (PAR.HOLDE TORR) PHRA (PAR.H2540_TORR) OPD (PARA2287_TORR)	PARLACI 15 PARLACI 18 PARLACI 10 PARLACI 10 PARLACI 10	-1.2544 -1.4413889 -1.226821 -7.7678	-48.38391 -48.4643256 -48.39312 -38.86440	PA PA PA	Brien Brien Santa Badaan Do Fala	TRAV. LOMAS VALINTINAS, 40 TRAVEISIA ONKO, 2009 REDOUTA ALCENTO MEIAN FELHO, 5(N - CIMTRO AVENDA FUNCIO REDO CONTINO, M*41, MARIAN	Greatfaid Greatfaid Greatfaid Roofbo	HU-SUPOR. SU-SUPOR. SU-SUPOR. SUP-SUPOR.	42 42 43	NOT INFORMED NOT INFORMED NOT INFORMED	ACT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED	80 80 80	BLARKS BLARKS MIDNING INDRAVORI DO BRASE LIDA NOT INFORMED NOT INFORMED	OK OK NOK	OK OK NDK OK	OK OK NOK	NOT NOT MIED'S EXTENDED JANTORIAL SERVICES MANANG LIGHTING DOES NOT WORK	MDT NDT NDT	HUED LICENSED LICENSERLE	NOT INFORMED YES NOT INFORMED

2 Keycode	Tauar Natas	Sile ID	Geslei Latitude (decimal)	Longitude (decimal)	Fed. Do	a ony	Address Address	Nile Type - Ouldcar	Structure Structure Type	Helght (m.)	(es) belaised (#*)	Capacity VEA Contracted (m*)	Neulare Ceasily (m*)	Vertical Structure Maker	Cleaning/Janifaring	Closing/Urbanization	General Mruclare Status	Adaptations Adaptations - Nite	Adaptations - Nicolare	Lices Matus of Sile Licenses	naing Nucleos of the Bits Statemental
237 579409 238 57962 239 7143884	T396. [CP400023757] TPGP (CP400023757] 4571 (P4CT04571_T0998)	PECROVISIA PERSONALISIS PECTORS71	-7.123997 -7.50113 -7.120231	-34.825 -34.8362 -34.8365	72. 73. 74.	30a0 Peccas 30a0 Peccas 30a0 Peccas	RUA JOSE RAMALINO LETTE, 1233 AVENDA SAPE. 1880 AVENDA BARRIDO, 123	Rooftap Rooftap Rooftap	MAST MAST MAST	5 22 4	NOT DECKMED NOT DECKMED NOT DECKMED	NOT INFORMED NOT INFORMED NOT INFORMED	N0 N0 N0	NOT INFORMED NOT INFORMED NOT INFORMED	NDK DK	OK OK NDK	NDK DK DK	WARRENS LIGHTING DOSS NOT WORK + LADDER DIAMAGE NOT WARANG LIGHTING DOES NOT WORK	MOT MOT MOT	FILID FILID FILID	NOT INFORMED NOT INFORMED NOT INFORMED
243 579525 241 7542674 242 7543674	3036 [CH40023837] 3044 [PERV03044_TORM] 3087 [PERV03087 TORM]	PERVGION	- 8.108136 - 8.1244444 - 8.127947	- 34.895775 - 34.897222 - 34.959309	n n	tecte tecte	BUA 3058 ADERGAL CHANES, 264 Bua Yangeri Avenda Bua Yangeri, 4000 Kdf Vária. Bua Andrico ROMORASI CORLEG. 422	Routtap Routtap Routtap	MAST MAST	(6+6+6) (5+6+6) (5+6+5)	NOT DECKMED NOT DECKMED NOT DECKMED	NOT INFORMED NOT INFORMED NOT INFORMED	N0 N0	NOT INFORMED NOT INFORMED NOT INFORMED	OK NOK NOK	OK OK	NDK NDK	ABIOS SATTORIAS, NO WAAVANG USATTAN COMPONENTS WITH CHEATTON SEARS. SLE ROD DAMAGED NO WAAPANE USDATTANE COMPONENTS WITH CHEATTON SEARS. LIGHTNENS FCD DAMAGED	NOT NOT	FILED FILED FILED	NOT INFORMED NOT INFORMED
244 7140715 244 7140715	3006 (PERV12004_TORRE) 3006 (PERV12004_TORRE) TAL6 (PERTA1002_TORRE)	PERCENT	-8.051383 -8.051383	-34.802306		Recte	AUX DELE DE LE LINNEL, 228 AV DOV. ADAMENON MADALHÈS DES-ESPENDEND REA MANDES DE HERVE. 352 - SANCO ANTONO - RECIPE - PE	Rollap	MAST	(2+2+3) (2+2+3+2,5) (3+2+5)	NOT INFORMED	NOT INFORMED	NO NO	SUMINS	NOK NOK	a a	ax ax	EQUIPH, CARDINE OCDATED, NO WRANING ISSETING, MANUAL WITCHEAAD DAMAGED OTHER COMPONENTS SHOW OXIDATION. LIGHTNING ROD DAMAGED	NOT	FILID	NOT INFORMED
248 532756 248 532756 249 533725 250 533865	CALE (PESEFERE TONE) PER (CM80032331) END4 (CM80082451 3C08 (CM80082451 3C08 (CM8008272)	PISSP1369 R3EL0417 R3FCR0387 R3FCR0387	-7.012652 -22.876825 -22.876825 -22.60638889 -22.9234763	-42.98392 -42.464399 -43.95388889 -43.95388889 -43.95388889	2 2 2 2	Rocores Ro De Janeio Ro De Janeio Ro De Janeio	Rus Statutes (201 Rus Statutes (201 State / Centry Rus CONCRET. TAMARINO, 1888 TCORENES States Ferenz, 22 Rus Marqués De Sacaregagué, 1606 4	Greatfield Greatfield Greatfield Greatfield	SELF-SUPPORT. POST SELF-SUPPORT. POST	42 42 32 32	2.25 5.17 5.34 3.54	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED	200	NA NA NAS SERVICE MENORS TENDON BLASS	8888	0 0 0 0 0	9999	RANNELLENTERI DANKED GAMINEL MITTALI DANKED GAMINEL MITTALI MEDITEXTENDES JOHTORIJE BERVICES NOT	MOT MOT MOT NOT	LICENSARCE NOT DECRMED NOT DECRMED NOT DECRMED	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED
251 7147233 252 532784	311 (BCX88031_TORM) 3A41 (CM30033821)	SCHOREDII SPWIGIRZO SPWIGIRZO	-28.562275 -23.46926 -23.56262	-49.182278 -46.72859 -46.39481	SC SP	Jeragua Do Sul Sao Paulo Sao Paulo	Rua dieñtardz, 4/n - Ro da Lui Rua fielastino Lavargenis 252	Greenfield Greenfield Greenfield	SELF-SLIPPORT. SELF-SLIPPORT. SELF-SLIPPORT.	30 50 50	2.6 2.32 NOT DIFORMID	NOT INFORMED NOT INFORMED NOT INFORMED	2 ND ND	CAW Briefal	OK OK	OK OK	OK OK	MALTORING, COMPONENTS WITH GROWTICN SIGNS GREENL SAMPTORISE, TO WERE SHOP CREATION SHOTE.	OVERLOADED NOT	LICENSAILE NOT DECRMID	NOT INFORMED
253 7142557 254 7141273	CEP (SPWC03714_TONE)	SPIIC02714	-23.54972	-46.65982	50	Sao Paulo	RAA GBEDICIN KU Averala Angélica, 2000 – Consolução – São Paulo	Roaftap	MAST	[4+6+6+2]	2.8	NOT INFORMED	ND	NOT INFORMED	NOK	OK OK	NOK	BRACES MESSING. LIGHTNENG RCD DAMAGED + CELING INFILTRATON MENSIONT-CUT REST RATION LIGHTNENG RCD CODATED + LADDER	STRUCTURE OVERLOADED	LICENSED	VIS
255 7142780	PHE2 (SPAPE2874_TORM)	5PWP0331	-23.566777	-46.60182	89	Sao Paulo Sao Paulo	Rua Janquin Antanes, 549 - Poberos - Sila Paulo Rua Cancelheso Radinguel Alver, 393 - Vila Plastana	Roding	MAST	(6+6) (6+2×2)	NOT DECEMID	NOT INFORMED	ND ND	NOT INFORMED	NOK	ox ox	NDK NDK	DAMAGED, MAST BASS SHOW CARCKS. GUT SCREWS SHOW CKENTION. NOT	NDT STRUCTURE OVERLOADED. RED/FORCEMENT	LICENSED PLUED	VES NOT
256 7142549					5										OK.				OF THE ANDIORS FOR THE CONCRETE BASES STRUCTURE OVERLOADED.		
257 7142581	VCEL (SPALHIBBE, TORIS)	SPULPCIER BUYCOD785	-23.49081	-16.68186	57	Sao Paulo Mata De Sau Juao	Rua Antônio Riberia de Manas, 264 - Vila Carlione, 58o Paulo - SP Rudova BR209	Routop	MAST	(4+4+8) (5+5+5)	(0,778 + 1,155 + 0,778) (0,79+0,47+0,79)	NOT INFORMED	ND ND	Telenort Ingestaria De Telecomicações 8/A	NDK DK	ox ox	NDK NDK	CELING INFLUENCES	ISON CREMENT OF THE ANDHORS FOR THE CONDRETS BASES	LICENSED	vis
259 7142377	CPE (BAICH1146_TONS)	BACPEI146	-12.54906	- 28, 996-98	BA BA	Rata De Sau Juao Ilha De Cuntandatuba	Ruduvia BROM Rua de Corrandatulos, s/h	Routtap	MAST	(3,5+3,5+3,5+2,5)	(0,79+0,47+0,79) 2.39	NOT INFORMED	NO NO	NOT INFORMED	NOK	ox ox	NDK	WITHOUT COMMIC CALLS, MATTS SHOW COLDATION SEENS, GENERAL SMITTHENS, NEEDS SMITTHENS, NO 1D PLATE, MATTS SHOW COLDATION SEENS	NDT	LICENSARLE	NOT INFORMED
260 7143934 261 7144002	CHU (FAALU2311_TOHKE) CHMA (FABLH2000_TOHKE)	PARLAGED	-1.38777	-48.38159 -48.47451	ΡΛ ΡΛ	Aranindeua Brien	Rua Cuberto Lendo, 4/11 Rua Cuberto, Nº23 (Riq. Ro. 20120 CR200)	Greenfield	SELF-SUPPORT.	50 30	6.23 12,13 (11,09+1,44+2,47)	NOT INFORMED	ND 6	Akatel Telecommissiones Bilmines LTDA	ox ox	OK OK	ак	NOT NOT	NOT STRUCTURE OVERLOADED	LICENSARUE FILED	NOT NOT INFORMED
262 101000	OMC [CM40026427]	PEPOLS	-8.397003	-25.063306	75	Балиса	Rus Pril Vicente Saludor, 202 Rus Pril Vicente Saludor, 202 Rus Mites Bichara Solewa. 66	Greenfield	SELF-SUPPORT.	30	9.47	NOT INFORMED	-	Tencon	OK OK	OK	OK OK	MATE SHOW CODATION SIGNS ROUTMINT CASING AND PLATFORM SKID	STRUCTURE OVERLOADED	LICENSARLE	NOT INFORMED
265 7140737 265 7140737	TORK (PECRO-6517_TORM) 3306 (PERO3306_TORM) 3225 (PEPOX3225_TORM)	PERCENSIO PERCENSIO PERCENSION	-7.1446M -8.175417 -8.0112M	-34.813 -36.817167 -36.820806	78 75	Juan Pessaa Jabontan Dos Guarantepes Recte	Rub Mirtes Bichara Notwera, 66 Anwenda Bernando Viesa de Mello, 2600 Bichoda das Ubasis 624	Routtap Routtap Routtap	MAST MAST MAST	4 (R+8+8+8+8+8) (11+11+11)	1.31 (1,07+1,07+1,07+1,13+1,13+ 1,13) (0.88+0.88+0.89)	NOT INFORMED	ND ND	SERVING NOT INFORMED NOT INFORMED	NOK OK	OK OK	NOK NOK	LIGHTING INOPERATIVE. COMPONENTS SHOW CODMISON SEGNE. COMPONENTS SHOW CODMISON SEGNE COMPONENTS SHOW CODMISON SEGNE	NDT NDT NDT	PLUID LICENSARLE LICENSARLE	NOT INFORMED
267 7147089 268 7147080 269 879926 270 576778	000 (P48593039_70988) 000 (P485,3038 70998) 3424 6005 (CM40000364)	PERSP2229 PERS 2018 PEOLEDHOA REPTLECOS	-8.04175 -8.0683611 -7.976369 -5.790002	-34.8928 -36.892722 -36.834582 -35.202139	n n n	Recife Recife Olinda Nortal	REA QUAMENTA E GETO NYEAH - BEF, NECEANJ E REA CRF, 2016 DA USZ 26 Avenda 3046 Augusta Manna 2258 - BEF Catade de Dinda Avenda Prezente De NORME, 757	Rootap Rootap Rootap	MART MART	7 5 (2+2+2) 27	2.9 1.45 (0,88+1,2+1,2) 801 IMORMED	NOT INFORMED NOT INFORMED NOT INFORMED	N0 N0 N0	NOT DECEMED NOT DECEMED NOT DECEMED	8 8 8	8 8 8	NOX NOX NOX	CONFORMERS SHOW CODMEDY SERVE NAMES ATTHOUT BASE FARTICIES WARENG LIGHTING DOES NOT WORK + LIGHTINGS COD DAMAGED COMPONENTS SHOW CREDATION SOONS	MOT MOT MOT	LICENSARUE LICENSARUE LICENSARUE	NOT INFORMED NOT INFORMED NOT INFORMED
271 7140797 272 7141356 273 7142479	3128 (FEF5M3128_TORRE) COD7 (SFWC03728_TORRE) CT01 (SFWR9400H_TORRE)	PSP5M3128 SPWC03718 SPWR04004	-8.036028 -23.5465 -23.56646667	-36.813778 -66.6481 -66.83949464	11 52 52	Recife Sao Paulo Sao Paulo	Averala 17 de Agosta 500 Rua Armol Guegel, 518 - Consolação Rua Concest Xavier de Tuinda, 75 - Napúbrica	Routtap Routtap Routtap	TUBULAR MOUNT HART HART	(12+12+12) (5+5=5) (5+6=6)	(5,9+0,9+0,9) (5,76+0,97+0,76) (1,25+0,62+0,62)	NOT INFORMED NOT INFORMED NOT INFORMED	ND ND	NOT INFORMED NOT INFORMED NOT INFORMED	NDK NDK DK	OK OK	NDK NDK NDK	PLATFORM/SKID CREATERS NO WARNENG LEXHTING LADDER DAMISMID + PLATFORM DAMISMID + PLATFORM DAGED CREATERS COMPONENTS SHOW CREDATION SEGNS	NDT NDT NDT	FELED LICENSED	NOT INFORMED YES
274 7142471 275 7142695 275 7142750	CTO2 ISPURA4005 TORE) BEE (SPWEE775 TORE) BME (SPWEE752 TORE)	5P8/56005 5P9/36279 5P9/362753 5P8/722347	-21 53869722 -23 592321 -23 5957 -23 5957	-46.67580817 -46.675807 -46.60534 -46.675025	2 2 2	Sao Paulo Sao Paulo Sao Paulo	Ay, Pantra Nuan, 202 Rub Musim Amuda, 327 - Zam Blo Rub Casta Aguna; 2446 - Ijatoga	Roditao Roditap Roditap	MAST MAST	(4+4+4) (2+4+4) (4+4+4) (5+5+5)	(0.88+0.16+0.68+0.88) 2.83 (0.70+0,48+0,95) (1.51+0,72+0,93)	NOT INFORMED NOT INFORMED NOT INFORMED	ND ND ND	NOT INFORMED NOT INFORMED NOT INFORMED	8 8 8	ox ox	NDK NDK NDK	NOT COMPONENTS SHOW COLDMIDW SOLIVE COMPONENTS SHOW COLDMIDW SOLIVE NO WARKING LIGHTERS LADDER AND LIGHTERS RCD DAMAGED.	MDT MDT MDT MDT	LICENSED LICENSED LICENSED	195 195 195 NOT
278 7543055	CEA2 (SPGA4N3_TORR)	SPEAMINE	-23.99758	-66.26031	v v	Gange	Rua Mária Roma, 201	Roaltap	MART	(3+3+3)	(0,98+0,76+0,76)	NOT INFORMED	NO	Naka Samana Networks	NOK	OK	NDK	CORPORTS SHOW COLDATION SIGNS LIGHTISHE ROD DAMAGED. COMPONENTS SHOW CREDATION SIGNS. NO VANESNO LIGHTISE. EQUIPMENT CARDINE AND OTHER	NDT	LICENSARLE	NOT
280 7141856 281 7141856	UNIX [276333009_10898] COC1 [599685277_10998] 3917 [3999993444 *******	SPG365009 SPP585277 SP96292848	-24.00209047 -24.02057 -23.55875278	-66.26845833 -66.47790 -66.69931111	29 29 29	Guanga Pisas Gaande Sao Paulo	Rub José Avelnu de Oberna, 200 Rub Afunos Chaves, 43 Av. Paulitik. 2006 - Belo Vitik	Rootap Rootap Rootap	MAST MAST MAST	(4+6+6) (3+3+2+2) (2+2)	(1,38+1,02+1,02) NOT DAVORMED (2,68+1,52)	NOT INFORMED	N0 N0	NOT DAFORMED	NOK OK NOK	OK OK	NDK NDK NDK	CONCREMENTS OF DATES. NEEDS SANTORING NO PLATE. COMPONENTS SINCE OF DATES. NO 2D FLATE. EQUIPMENT CARENT COMMEN- COMPONENTS NEEDS	NDT NDT NDT	FLED FLEP	NOT INFORMED YES
282 7142458 282 7142463 283 7142769	2012 [27:437269_10893] 2012 [27:437269_10893] 2012 [27:437269_10893]	5PW2P2848 5PW2P2854 5PW002560	-23.55875278 -23.55857 -23.5946	-46.69911111 -46.67254 -66.5872	y y y	Sao Paulo Sao Paulo Sao Paulo	Av. Pauleta, 2006 - Bela Veta Avenda Bradi, SjN - Jackin Areleca (Fraça Noota Seehora da Basil) Rua Verag I, Via Schependinca	Routap Routap Routap	MAST MAST MAST	(3+2) (3+2+2) (3+2+2)	(0,68+1,52) 2.43 (0,94 + 0,63+0,81)	NOT INFORMED NOT INFORMED NOT INFORMED	ND ND ND	NOT INFORMED Tel Teleconunicações NOT INFORMED	NDK DK DK	OK OK	NDK DK NDK	NO 20 PLATE. NO 20 PLATE. NO 20 PLATE. PARTE SHOW COLDATION MISSION NO 20 PLATE. NATE SHOW COLDATION MISSION	NDT NDT NDT	FEED NOT DEFORMED	NOT DIFORMED NOT
284 7544475 285 7545723	2728 (SPA373068_T0998) C012 (SPMA5229 T0998)	SPW2P2068 SPW2P2068	-23.56922	-46.65788	10 10 10	Sao Parlo Pracuada	Noi Veneg a, via atarpenanta Noi Bristan, Sili Noi Pisa da Catedral Dominesta de Pisala, 170 - Centra	Roattap	MAST	(4+4+4) (1+2+2+2)	(0,00 + 0,0140,01) (0,68 + 0,89 + 1,22) (0,21+1,99+0,43+1,30)	NOT INFORMED	ND ND	NOT INFORMED NOT INFORMED	8 8	ox ox	NDK NDK	NO 10 MILLION COLLATION BIONE NO 20 PLATE. PRATE SHOW CRATE. LIGHTING KOD DAMAGED. NO 20 PLATE. PRATE SHOW CRATE.	NDT NDT	LICENSED	YES
288 533291 287 532340 288 7542924	0026 [CM30039923] 0032 [CM3003064] CT02 [SPCA84648_T0986]	SPACLING SPACALING SPACALING SPECIMENT	-21.42115 -21.42115 -21.4314 -22.90113	-45.75023 -66.7901 -67.05827	10 10 10 10	Sao Paulo Sao Paulo Campitas	Rad Frig, a de Consente Consentantes de Falan, 2.0 - Consen Rad Cavela, 5 - Sachter Consentat RZ, Felgop Sturks, 5,N - Anbaguens Rad General Oxion, 1233 - Centra	Greatfield Greatfield Routtap	POST SELP-SUPPORT. MAST	40 50 (2+2+5+5)	4.95 20.95 2.3	NOT INFORMED NOT INFORMED NOT INFORMED	10 15 NO	Tolenout fivered trigentaria de tinfa- estatution NOT INFORMED	8 8 8	OK OK	OK OK NOK	COMPONENTS SHOW CRITERIA	NDT NDT NDT	LICENSABLE NOT DEFORMED	NOT DECKMED NOT DECKMED
289 7141035 290 7141038 291 7141038 291 7141421 292 7140986	CTOS (EPCAS4E71_TORIE) INEL (EPCAS4E75_TORIE) CTOL (EPICAS4E75_TORIE) CTOL (EPICAS4E7_TORIE) CTOL (EPICAS4E7_TORIE) INCAL (EPICAS4E7_TORIE)	SPCAMERI SPCAMERI SPTAMERI SPTAMERI SPTAMERI	-22.80455 -22.84 -23.006 -22.9014389 -21.006	-47.05312 -47.05226 -45.55006 -47.45195278 -86.6006	97 97 97 97	Campitas Campitas Tasta Bascata Proces	Rea Xini do Mesqueta, 36 Rea Variani do Ruí. 555 - Reacera con Anesta Bolen Found Rea Quertos Bicaleso, 3,03 Rea Cantón Resconcela Film. 177 Egy Nobe Fuerton VIP	Rodiao Rodiao Rodiao Rodiao	POLIS POLIS POLIS POLIS POLIS POLIS	(2,3+2,3+3,2) (3+6+0) (5+6+0+2) (4+6+0+2) (4+6+0+1) (4+6+0+1)	(2,85+1,66+0,53) (1,37+0,86+0,86) MOT DIFORMED (0,75+0,50+0,76) (1,75+0,70+0,76)	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED	ND ND ND ND 22.99	Naka Secrets Networks TH Telconung actes NOT INFORMED Parts Security Torentary MIT Networks	28.88	0X 0X 0X 0X 0X 0X	10X 10X 10X 10X	COMPORENTS SHOW ORIGINATION STORE COMPORENTS SHOW ORIGINATION STORE COMPORENTS SHOW ORIGINATION STORE COMPORENTS SHOW ORIGINATION STORE COMPORENTS SHOW ORIGINATION STORE	MOT MOT MOT MOT MOT	LICENSED LICENSABLE NOT DECEMBLE LICENSABLE NOT AMOUNT	NOT INFORMED NOT INFORMED NOT
294 644859 295 660382 296 651280	00% (015050566) 0548 (01505566) 1519 (015055612)	SPRECISION SCREUDOOS PRETADOIA PRETADOIA	-25.58983287 -26.98771144 -25.59061389 -25.4333356	-48.6014218 -48.63645491 -48.27754722 -48.277025	57 5C PR PR	Bainearia Caniboria Curitiba Curitiba	Ruk 3056 Castland, XF9 Ari, Alvin Bauer, XF9 Ban Pict. Spice G. Adr. Collision, 201 Run XF2 do Navembro, 340	Roattap Roattap Roattap Roattap	POST MAST MAST MAST	22 (1,5+1,5+1,5) (2+2+2) 4	11.32 (0,47+0,47+0,47) (0,75+0,75+0,75) 2.36	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED	12.96 N/O N/O	MOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED	NDK NDK NDK	OK OK OK	OK OK NOK	CELING DESTRICTION + OI RUE DESTRICTION + EQUIPMENT CARINET ORDERTED EQUIPMENT CARINET ORDERTED. NO SITE (CARINET CARDATED.	MOT MOT MOT MOT	HUD FUED	NOT INFORMED NOT INFORMED
297 879569 298 650365	06/2P 31-0P (CM60056473)	RECEIDAD PREDAD214	-28.16858 -23.31271389	-51 18822 -51.1759	83 191	Canada de Sur Landera	Rub Standou, 2002 Rub Statu, Souda Naves, 45	Rodhap	MAST	4	/1.48 + 0.74) 1.82	NOT INFORMED	NO NO	NA NOT INFORMED	NDK NDK	ox X	NDK NDK	IQUERANT CARDAT AND CONDUCTS CALENTS BQUERMENT CARDAT OF DATED, GRARAL JANTTORINE, REPLACE REST PLATFORM IQUERMENT CARDAT CARDATED + WANADAY UNDERSTAND INVESTIGATION + WANADAY	NOT	PELID	NOT
200 658738	138P (CM8036389) 142P (CM80383738)	PRETADO26	-25.4263889	-49.2561111	PR. PR.	Curtilia	Ran Satis da Abril, 571 Av. 3850 Gashiwita, 623	Routtap	MAST	5 [3+2+0+2]	2.38	NOT INFORMED	ND ND	NOT DEPORTED	NOK	OK OK	ак	JANTORIAL COMPONENTS DEDATED NO SITE DEMITPICATION EQUIPMENT CADATE AND OTHER SMALLER COMPONENT CADATE AND OTHER SMALLER EQUIPMENT CADATE AND	NDT NDT	FLID FLID	NOT INFORMED
301 659790 302 660390	0578 (CH50053073) 1239 (CH50053472)	PRETADDIO PRETADDIO	-25.40752844 -25.434267	-49.29173333	PR.	Curtaba	Largo Alfredo Pascol, 1441 Altarreda Dr. Carlos de Carvalho, 271 - Centra	Roaftap Roaftap	MAST	(2+2+2) 3	2.6	NOT INFORMED	ND ND	NOT DEFORMED	NDK NDK	OK OK	OK NOK	CORPORATE CREATED CRASHEL INFORMATIVE WARNENG LIGHTING INCOMPARITYE RQUEPMENT CARENT CREATED. NO ID PLATE. COMPONENTS INFORMATION SIGNE.	NDT NDT	FILIED FILIED	NOT INFORMED
303 7144833 304 7147523 305 879316	052C (HECKL0052_TOHME) 4627F (HEPGL0062_TOHME) 2037F-3	RECELUDE2 PRECADER	-28.168377 -25.07564622 -25.4293567	- 51.203406 - 50.176363 - 49.2633332	15 14.	Canada Do Suil Polita Grossa Curitta	Au. 33ku de Castilhon, 4054 Ruis Vanhägen, 366 Ruis Parcellas Decidios. 950	Kushap Kushap Kushao	MAST MAST PRAT	6 (5+6+6) (2+8)	1.67 2.25 5	NOT INFORMED NOT INFORMED NOT INFORMED	N0 N0	NOT INFORMED NOT INFORMED NOT INFORMED	NDK NDK	OK OK	NDK DK DK	LIGHTNENG ROD DAMAGED, NO IDENTIFICATION WARKING LIGHTING DESCRIPTION OF THE COMPONENTS SHOW COLDATION SIGNAL LIGHTNENS ROD CO2DATED LIGHTNENS ROD CO2DATED	NOT NOT	LICENSABLE NOT DECKNED	NOT NOT INFORMED
306 879542 307 532516	054P PPA1 (CM30008364)	RICEL0054 SPRCIDE/H	-28.154564 -23.50513	-51.175622	83 59 59	Canada Do Sul Sao Paulo	Rua Gen. Arcy da Racha Robrega, SOI- Jackin América R. Haria Angélica Poinci, - Via Franci	Routtap Greatfield	NAST SELF-SUPPORT.	6	1.67 5.42	NOT INFORMED	N/0 9.58	NOT DECRMED NOT DECRMED NETWORKER TELECOM	NDK DK	OK OK	NDK DK DK	EQUIPMENT CARDNET CODMED, SHOWING INADEQUATE POWER CONDUCT. BEST FLATIONER CREATED. MATCHINE SANTORNEL SCHE COMPONENTS	NOT NOT	LICENSARUE LICENSARUE	NOT NOT
308 533252 309 533269	084 (CH3800897) 035 (CH38018576)	SPACP3017 SPACP3018	-23.49238 -23.50641	-46.3926	59 59	Sao Paulo Sao Paulo	R, Manuel Martins de Mela, 5/N- Vila Itann Avenda Tamé Das Lago, 5/N - Itann Pauleta	Greenfield Greenfield	SELP-SUPPORT.	50 60	2.29	NOT INFORMED	12 NO	NOT INFORMED NOT INFORMED	ок	OK OK	OK OK	SHOW BEGINNEN CHEMTEN, WARRING LICHTING WITHOUT BRACK. SANTORING, LICHTING RCD CALLE BROKER COMPONENTS, SHOW BEGINNENG CHEMTION	NDT NDT	NOT DECRMID	NOT INFORMED
102 650363 311 660363 312 531542	2109 (CM30055471) 0158 (CM30052779) 2584 (CM30009140)	PRIDAD218 PRASSIDLS SPW1H0830	-21 311102M -23.3735 -23.46289	-51.12669753 -50.843 -66.45926	н. 92	Accel Sao Paulo	R. Atlanta. 80- Hedr Rua Mancel Ribel, 150 Rua Tetuansa, 146 - Jardan melena	Rolf to Greatfield Greatfield	NAST SELF-SLIPPORT. SELF-SLIPPORT.	(248) 30 46	2.14 8.59 1.92	ACT INFORMED NOT INFORMED ACT INFORMED	N0 N0 N0	ADT INFORMED	NDK DK	OK NOK OK	8 8 8	NECKENET GEDATED ANTORAL SERVICES. NO WANING LIGHTING CONFORMES WOW OWNERTING SEANS. NOT	MOT NOT NOT	FILIED LICENSABLE FILIED	NOT NOT YES
313 533328 314 7144621	0006 (CH30032719) 308 (SPW/R0676_T0R68)	SPURSID00	-23.47285 -23.60001	-46.3944	57 57	Guaralhos Sao Peulo	Rua Candelinia, NN - Jackim Jacy Rua Judei Renet, estilido Cicero P. or Tairedo, portilo D. Mourillo	Greatfield Reaftap	SELF-SUPPORT.	40	1.69 NOT DECRMID	NOT INFORMED	12	NET WORKER TELECOM	NDK OK	NDK OK	ox ox	NEEDS EXTENDED JANFORDAL SERVICES + REPAIR FENCE + NERRING LIGHTING DAMAGE NO LIGHTING ROO, NO WERNING LIGHTING	NDT NDT	LICENSARLE	NOT INFORMED
315 880077 215 7148853 317 532881 318 532560	ESS 13 (ESPAROLI MART) NOFA [CH30009734] SCRA (CH30000382]	RIPASORIA RIPASO11 RICPEZZIS RISCTOLIS	-33.118054 -33.065389 -22.85206 -22.899465	- 51, 253744 - 51, 255222 - 43, 174784 - 43, 200305	83 83 80 80	Porto Alegra Porto Alegra No De Jaceso Ro De Jaceso	Av. West-ensus Bocchar, 5807 Rus Connert Austrice Bronner, 834 - 686m Av. Pinceelas Balari, 9* 320 Rus Cardono Stanisho, 18	Rodia Rodia Rodia Rodia	MAST MAST MAST	6 (6+6) (5+2×3) 6	5 (2,70 + 0,04 + 0,58) 3.13	NOT IMORMO NOT IMORMO NOT IMORMO	ND ND ND ND	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED	8 8 8 8 8 8 8	0X 0X 0X	OK NDK NDK	NOT GUARDAG, COSDATED NOT COMPONENTS SHOW CORDATION.	MDT MDT MDT MDT	HILED LICENSED NOT DECEMED	YES YES NOT INFORMED NOT INFORMED
319 531670	3608 (CM10008807)	824PM0027	-22.841119	-43.312	RJ	No De Jaceso	Rua Enge Lafante Stockier 961, Via da Panha Setar Cartosi - Piaça da República. CEP. 75503-900	Routtap	NAST SELF-SUPPORT.	(6+6)	5,13+2,17	NOT INFORMED	ND ND	NOT INFORMED	NOK OK	OK OK	NDK	ACCESS SHOWS MATCHAE DAKES WARE WARENG LOWTING DOES NOT WORE THES DAMAGED NO ID RATE TOWER STRUCTURES SHOW HERY CORPOSION. LACCER STRUCTURES	NDT STRUCTURE OVERLOADED	NOT DECEMBE	NOT INFORMED
221 648465 222 648467	CHEL (CHEODET363) BHD (CHEODET353)	MORLZ2004	-15.85048 -25.968181	-41.96383	80 80	Brio Horsenter Brio Horsenter	Avenda Antino Carlos, 8200 - Parguita Mal Bul Minor Carlos, 8200 - Parguita Mal Bul Minor Carlos, 9200 - Parguita Mal	Rolling Rolling Rolling	MAST MAST	1 (16-40)	1.85	NOT INFORMED	NO NO	NOT INFORMED	28.8	0K	ax ax	DEDATED, LEDITATED FROM CALLE BROKEN WRATENG LEDITENG FROM RATENE. NOT NOT	NOT NOT	FILED	ACT INFORMED
124 648284 125 754868 126 7558749	CELL (CHIROMMAG) CTCE (IRMODALION_TONN) OND [IRROTOLINT TONN)	MI2TO(308 MI2VL308 3980703347	-18.81892 -20.1419 -22.898906	-43.92906 -44.8893 -43.592738	NG NG	Belo Horganite Demopolis Ro De Janego	RUA CAREDON, 424 - SALA DE (PRAÇA SETE DE SETENBRO) Rua Rio de Zanero, 434 Ria montertecto, 23	Routtap Routtap Routtap	MAST MAST MAST	(2,5+2,5+2,5+2,5+2,5+2 ,5) (6+6) 6	(1,231+1,231+1,231) 2.8 0.11	NOT INFORMED NOT INFORMED NOT INFORMED	N0 N0	NOT INFORMED NOT INFORMED NOT INFORMED	OK OK	0K 0K	OK OK	NOT MENTEMANCE OF THE LIGHTNENS ROD SUBTRATATION CABLES NOT	NDT NDT NDT	LICENSED LICENSABLE NOT DECAMED	YES NOT INFORMED
327 531326 328 531259	NEU (CHI30008851) HC04 (CHI30008665)	NAL RECPONE	-22.7575837 -22.930766	-43.0665363	RJ RJ	Nova Spiacu Ro De Janeso	AVENIDA NUO PECANNA, MIO AVENIDA PAULO DE PROMI DA 729	Roattap	MAST	4	2.51	NOT INFORMED	ND ND	NOT INFORMED SIM	OK OK	OK	OK NOK	ND MATTING, NO WARNING LIGHTING. COMPONENTS SHOW OXIDATION AND MAINTENANCE INCLUSION. PREVENT IN RELITING INCRUMING DUE TO DISTRICT IN	NDT NDT	NOT DECRMED	NOT INFORMED
129 531613 130 532663 131 7147639 132 532360	BPGE (CM30008501) O38D (CM30008311) 425P (SC0845425 T0898) BT9H (CM30010377)	838730728 828870343 600880425 83873066	-22.997497 -22.968802 -16.6598889 -22.987641	-43.238571 -43.238633 -49.2995833 -49.2995833	888	Ko De Jaceso Ko De Jaceso Genera Ko De Jaceso	AVANDA GAS AMERICAS, 4646, CIP-226311402 Rua Pacheco Leño, 1221 - Fundos - Sardon Batlenco, Bo de Janeso - 82, 22460 803 ANOLIAR, BOLTPEE - RUA 44 RUA MANDERAR DE FINIDALA. 2021	Rodiap Rodiap Rodiap Rodiao	MAST MAST MAST MASTERP-809	9 10 12.5 (747)	2.73 2.23 3.14 (1.03 + 1.86)	NOT IMPORTED NOT IMPORTED NOT IMPORTED NOT IMPORTED	20 20 20 20 20 20 20 20 20 20 20 20 20 2	MOT INFORMED MOT INFORMED MOT INFORMED ZOPONE	8 8 8	OK OK	a a a	90001900 L30HT145 D040400 N07 N07 N07	MOT MOT MOT	LICENSABLE NOT DECRMED LICENSABLE NOT DECRMED	NOT NOT INFORMED NOT NOT INFORMED
223 533025 234 7146525 235 7146529 235 7146529 235 7146598 237 533888	CREF [CH10009411] CREF [SUB1135/5] TORMS] CREW [KUMURADOLA_TORMS] CREW [KUMURADOLA_TORMS] CREW [CH1000082891	ROOM 1002 REF 12675 REF0000046 DFR0000118 REF0000114	-22.814308 -23.001226 -22.950579 -25.95189 -22.555189	-41.982536 -42.09578 -44.072399 -47.215306 -45.182517	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	Cabo Fris Ro De Taneso Mangaratika Costalina Mase	ISTRADA DE CARO PRO E BUZERS, SYN ASTRED DA AMERICA. 7777 BODOVIA NO SAMECE, 7 RODVIA NO SAMECE, 0 RODVIA NO SAMECE, 0 RODVIA NO ASTREDERIK MISSING, 1075 3. RODVIA RAMA CONSULT S.	Greathed Roaft ap Roaft ap Greathed Greathed	NAT SELF-SUPPORT. NAT SELF-SUPPORT. SELF-SUPPORT.	40 21 12 40 40	5.32 2.37 (0,50 + 0,50) 1.65 2.92	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED	11.41 NO NO 12 NO	GETEX SERVIÇOS LITA MOT INFORMED MOT INFORMED MITALICAS EXTRUTUROS MOT INFORMED	8888	OK OK OK	9988	NOT 34411092405 NOT 5007 340704040, NO 94441940 LEDHTEK	NOT NOT NOT NOT	LICENSARIE LICENSARIE LICENSARIE LICENSARIE MOT DECEMBO	NOT INFORMED NOT NOT NOT NOT INFORMED
338 \$31624 339 \$31770	RTDR (CH3000E503) CTLA (CH3000E822)	KBT30412 KXCT00960	- 22.99681 - 22.90602	-43.3462 -43.177934	RJ RJ	Ro De Jaceso Ro De Jaceso	An. Die Andrices, 3434 - Bloco S. Candonino Mára Henrape Eimonen Rus Sato Jose, 90	Routtap Routtap	MAST	5 (5+6+6)	2.45	NOT INFORMED	ND ND	NETWORKER TELECON IND E CON REPRES LTDA NOT INFORMED	NDK DK	OK OK	NDK DK	MATTING AND OTHER COMPONENTS DREATED, TOWER BASE BLOOK ABED MATTEANCE. SOME PARTS OF THE TOWER STRUCTURES SHOW CODUCTION SEGNE.	NDT NDT	NOT DEPORTED	NOT INFORMED NOT
341 643318 341 7146949	2289 [CH3006435] 2289 [CH3006344] 36606463452 TORE PPP [CH3009665]	GENALDES GENERALES GENERALES KINAMODIA	-16.475694 -16.475694 -12.85132	-49,259136 -49,265222 -49,2665699 -43,285486	8 8 8	General Aparecida De General General Reo De Janeso	AV. THE CHARMAGE Q 22, 0 REA CAMBANGER Q 22, 0 REA ON ESCI. C REA 22-A. CO.SA. LT SL. NY SCI. SETCH CHARMAL PRODUCT AD MERCADING FRANKVLL	Rostap Rostap Rostap	POST MAST MAST	10 6 4 2	11.47 4.81 2.57 (8,72 + 1,04 + 0,40)	ACT INFORMED ACT INFORMED ACT INFORMED ACT INFORMED	ND ND ND	NOT INFORMED NOT INFORMED MOT INFORMEDTEL TELECOMUNICAÇÕES	OK OK NOK	0K 0K	OK NOX	NCT LISENTNENS ROD ADEQUATION NCT LISENTNENS ROD COMMERC. TOTALE RASE OF COMMERCE.	MDT MDT MDT NDT	NOT DEVOLUTION	NOT INFORMED NOT
141 7143147 141 7143147 145 7143124 246 829524	PPP [CRISCORCAS] OHT [KINE12112 TORRE] HOT [ROPOD23 TORRE] RDIA-1	KH73011 KH73011 KH70023 KX700218	-22.998397 -22.998397 -22.925248 -22.908728	-43,280486 -43,36342 -43,325317 -43,536954	10 50 50 70	Ro De Janeo Ro De Janeo Ro De Janeo	REDING X/J HERLENNED FRANKULL Ar. dat Andreas, 2122 Rea Hawingar Parts, 450 Anatola Contas Halan, 196	Roaltap Roaltap Roaltap Roaltap	MAST	1	(8,72 + 1,04 + 0,40) 0.45 11.721 2.54	NOT INFORMED NOT INFORMED NOT INFORMED	ND ND ND	LIDA MOT INFORMED MOT INFORMED NOT INFORMED	NDK DK DK DK	ox ox ox	NDK DK NDK	NOT A CONTRACT OF AN AND AND AND AND AND AND AND AND AND	MOT MOT MOT	NOT DECAMED NOT DECAMED NOT DECAMED	ACT INFORMED ACT INFORMED ACT INFORMED
347 532399 348 531862	DQ12 (CH30008584) 8120 (CH30008334)	R30Qx2829 R387334033	-22.294744 -22.987281	-63.335603	KU KU	Duque De Castas Rio De Janeeo	RIA PORTO ALEGRE, SIS Ria L, 4 - Bara da Tauca, Ro de Janezo - RO (ATREA DA OLERANDAREA. RO (IRRA)	Rootsep Greenfield	MAST POST	(5+6) 30	(1,56+2,80) NOT INFORMED	NOT INFORMED	NO NO	PROMON TELECOM NOT INFORMED	NDK OK	OK OK	OK OK	NEEDS JAATTORING, NO WARNING LIGHTING LI, KOO JAND WARNING LIGHTING INCHERATION	NDT NDT	NOT DECRMID	NOT INFORMED
349 532459 350 532794 351 647499 352 648287	FSF1 (CH3000630) CHE (CH3000095) CCEE (CH30047809) IBUI (CH5047869)	K31QHE550 K04KCB532 HIGEFCH302 HIGEFCH323	-22.8938 -22.914911 -18.94934 -19.962233	-42.4685 -42.9326 -42.9331 -43.95657	КЈ КЈ 1902	Saquarena Eaborar Brio Hortonte Brio Hortonte	Rua Sagatindo Bovo, 9 HAI GEMAR DOS SANTOS DUARTE, SVI - QUEL STE2 Bui CRUSING, STE CON BUI RUA RECOND., 72	Greathed Greathed Roditio Greathed	POST POST MAST POST	40 30 20 30	1.07 7 (1.805 + 0.951) 6.96	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED	6 7 ND ND	LTDA CLEMAR ENGINEER MOT ENFORMED	OK NDK OK OK	OK OK OK	OK OK OK	NOT ANNOS EXTINDED SALTORIAL SERVICES ACT SALTORIAN AND REPLACEMENT OF CODUCTED SCHWS	NDT NDT NDT	raad LICENSED PLUED PLUED	NOT INFORMED YES NOT INFORMED NOT
151 7142957 354 521576 355 7141440	SCL (HOPLAHISI TORR) CMN (CHOODEJ31) BMAS (SPWBSSE71_TORR)	NOPLMEISI PROBAZIO SPWEISSTI	-28.82711 -5.2052 -23.54322	-61.6262 -69.613 -66.61921	900 9A 59	Bala Maraba Maraba Sao Paulo	Ял. 1806. 279 РОСНА 22, ОО. 12, 1071 В., 1, N. – NOVA МАКАВА Изак DD GARDMETRO, N41, CEP1 2004 - 000	Analise Gesether Routise	NAST SELF-SUPPORT. NAST	80 (6+6+6)	2.548 NOT DVORMED NOT DVORMED	NOT INFORMED NOT INFORMED	N0 N0	NOT INFORMED NOT INFORMED NOT INFORMED	88	0K 0K	OK NOX	AGO AGO NO DERITIFICATION, NO WAARING LIDHTING REST FLATFORM, TOWER BASES AND FASTENING SCHEME BHOW CREEKTON.	MOT MOT MOT	ISCENSARIA LICENSARIA	NOT INFORMED NOT
155 7141443 157 521825 158 642011	etA: [SPWEUS35_TORE] CTNL [CROSSE340] ONC [CROSSE340]	SPWRJD525 PRMR2311	-23.57289 -5.34899 -23.214249 -11.214249	-46.20561 -48.13479 -40.24406	57 PA 13	Sao Paulo Maraba Vill Willia	RUA ARMANCO FARMANCA, 310 RUA CINCO DE AMES, 3410 RUA ALTICO EDEADES, 200 RUA ALTICO EDEADES, 200	Realitap Greatifield Greatifield	HAST SELF-SLIPPORT. SELF-SLIPPORT.	(8+8+8) 40 2	NOT INFORMED R.S158 NOT INFORMED	NOT INFORMED NOT INFORMED NOT INFORMED	ND ND	NOT INFORMED NOT INFORMED NOT INFORMED	8 8 8	OK OK	NDK OK	ANALY SHOW CRACK. NO? SALENDERS, NO SEANNING LISHTING, CONCRETE BASE SHOWS CALCO, MANTORING MANTORING	NDT NDT NDT	LICENSED LICENSARLE FILED	NOT IN CASE
359 648825 360 647828 361 651384	0827 (CH30047580) BANE (CH50047435) PEPES (CH50046763)	MOLINESH MOLINESH	-21.54003 -18.34664 -17.61712	-47.93671 -46.09007	DIM DIM DIM	too De Fois Cheata Baolanda De Mead	RUK MAREINAL ERECERE 444 Proja Rui Barkosa, 200 ; ( KLYRA SHCPPENS ) FAZINEK GLIRA DA BARRA, 3/N	Rooftap Rooftap Rooftap	MAAT	(2+6) (5+6+5) 2.3	NOT DECEMED (1,45 + 2,023 + 1,183) NOT DECEMED	NOT INFORMED NOT INFORMED NOT INFORMED	N0 N0	NOT INFORMED	OK NOK OK	OK OK	NDK DK	LIGHTING AD WAREHAU LIGHTING LIGHTING ROD DIMAGE. WAREHOUSINFING ROD DIMAGE. NO DENTIFICATION, NO WAREHOUSING WAREHOUSING ROUT	NDT NDT NDT	FLIED FLIED LICENSED	NOT INFORMED NOT INFORMED YES
362 647757 363 659009 364 656975	CDDA (CM10046840) 0851 (CM10042723) 2749 (CM100186183)	NO2FA5116 NGCR06387 GOVP55274	-21.766517 -17.187222 -16.056813	-41.347744 -41.695389 -47.883383	MG 00	Julii De Fora Carali Valpanasa De Guras	AUTINEA BANKO DO KEO BIONCO RUA JOSE VICIANTE COMPRA, JEO QUADRA 5, 0, CEM JORTZOS	Routtap Graatfield Routtap	MAST POST MAST	(2+6) 42 6	4.164 NOT DIFORMED 3.17	NOT INFORMED NOT INFORMED NOT INFORMED	ND ND ND	NOT INFORMED MOT INFORMED PROMON TILLECOM	NDK DK DK	OK OK	OK DK NDK	LIGHTNING ROD AND LADER DAMAGED ROT STRUCTURE REQUERE URGENT MAININGROUP DE HEAVY CREEKTON, DAMAGE DUE TO HEAVY CREEKTON,	NDT NDT NDT	PELED LICENSABLE LICENSABLE	NOT INFORMED NOT NOT
365 658852 366 7149948	MOR (CHIOCEES) CLR (CLRCSSC, YORK)	MORPHONE RELEVANCE TO	-18.07736	-45.20436	MG KJ	Ro Paranaba Ro De Janeso	REELVA BR 254, 0 Rab Elas Pecenal, Nº 658	Greenfield Routtap	SELP-SUPPORT.	50 3	2.59 2.35	NOT INFORMED	n NO	NETWONSER TELECOM IND. COM REPRESENTAÇÕES LITOR NOT INFORMED	OK NOK	OK OK	OK OK	COMPONENTS SHOW CREDATOR SERVE. CLEAR UP CARLIES + LADDER OCERATED. NO NUMBER LINESTER	NDT NDT	LICENSARIE	NOT
367 658278 368 7146590 359 647506 270 661823	225P (CM50098053) 24 (DF85A00M T09881 T854 (CM50047836)	DFBLACO 14 PERIADO 14 PERIELADOS	-15.76348 -15.76531 -19.879664 -18.879663	-47.83425 -47.875705 -43.9682555 -43.95488	09 07 90	Baala Baala Bro Korgosta	Saan Quadro 3 Lote 870 - N S/N SET SAM OD 3 BL A - O SET AUTARDISAS MORTE. FLAND FEOTO Rodova Br-336 , 2007 - Br Shopping	Kostiap Bootian Kostiap Kostiap	GUNED HART MART MART	20 4 3	2.15 2.58 1.57 2.57	NOT INFORMED NOT INFORMED NOT INFORMED	N0 N0 N0	NOT INFORMED NOT INFORMED NOT INFORMED	OK OK NDK NDK	0X 0X 0X	OK NOK OK	PRIVENTIVE MAINTENANCE AT THE TOBER SCHWEIDUE TO CREDATION ACC INC. CARDINT CODATION INCOMENDIAL UNITED INCOMENT CARDINE TO PRIVATE INCOMENT CARDINE TO PRIVATE	NDT MDT NDT NDT	NOT DECEMBE NOT DECEMBE FLUE	NOT INFORMED NOT INFORMED NOT
270 661822 271 532341 272 659486	OEE2 (CHEOHELES) BED3 (CHEODELES) MOR (CHEODELES)	MIFLORIZS RIBERIDAN MIZRIDABEZ	-15.89643 -22.75329 -20.31091667	-41.431385 -43.72011111	NG KJ MG	Beford Ross Ealerto	Rud Jacué, IBEZ Avienzda Retifico da Poppiensia, 595 Estrada prencipal, 5/N	Roattap Roattap Greenfield	MAST MAST SELF-SLIPPORT.	6 (5+5+6+5+6+5+6+5+6+ +5+5+5 40	1.57 R.58 0.85	NOT INFORMED NOT INFORMED NOT INFORMED	N0 N0	AM TELECON Naka Servers Networks NOT INFORMED	NDK NDK NDK	OK OK NDK	OK OK NOK	WAMMING LIGHTING DOER NOT WORK NEEDS MACTORING, MADITEMANCE AT WARRING LIGHTINE AND LIGHTINES ROD. NEEDS STREAMED MINITORIA, SERVICES + REPLACEMENT OF FENCE	NDT NDT NDT	FLID LICENSABLE	NOT INFORMED NOT INFORMED NOT
373 512987 224 7145458 275 7142435	OFFL (CH3000603) 563 (SCS000503 T0986) 382 (SPW0P3841_T0986)	829870526 863000560 894092840	-22.84684 -27.558884 -23.5815	-43.05836 -48.635475 -66.6754	КІ 50 57	Nteru	Rua Áda Nures, Nº 54 RIA AURIDO DA SEUR. 223 Avecida Sila Gaber, 238 - Suedo Puedra	Greatfield Greatfield Roaftap	SELF-SLIPPORT. SELF-SLIPPORT. REAT	30 18 (2+3+2×3+4)	6.38 6.11 2.44	NOT INFORMED	6 6.11 NO	NOT INFORMED SECCOM. NOT INFORMED MOT INFORMED TELEMONT EXpension	8 88	OK OK	ax ax	NOT NOT	STRUCTURE OVERLOADED MOT	WITHOUT LIGERLATION LICENSARLE FILIED	N/A NOT NOT
27% 7142460 277 7142541 278 714264 279 714264 279 7142637 280 7142635	2P14 (SPATPLES TONS) CES2 (SPATCALER TONS) 205 (SPATECALER TONS) 202 (SPATCALER TONS) CP11 (SPACE277) TONS) CP11 (SPACE277) TONS)	5PW2P3850 SPU5C6338 SPW302777 SPW205115 SPP065273	-21.5736 -21.5379647 -25.66311110 -21.1877 -24.21368	-86.660612 -66.65056111 -66.657264 -67.6119 -66.62288	2222	Sto Sco Sto Paulo Sto Paulo Sto Paulo Sto Paulo Sto Paulo Romas Paulo Paulo Calade	Alternolo Casamina Lano Na Mangara Lan, Su Balanaka Cagla, Su Ra Alao Cagla, Si Ra Bala Saladari, So - Camanana Na Bala Saladari, So - Camanana Na Pandanta Walana	Roof too Roof too Roof too Roof too Roof too	MAST MAST MAST MAST MAST	(2+0+2+2+4) (8+6+6) (8+6+6) (8+6+6) (8+6+6) (8+6+6)	2.25 2,62 (0,10 + 0,82 + 0,82) 2.52 (0.84 + 0.84 + 0.86) 20.70 2.02	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED	50 50 50 50 50	MDT DAYOBARD TELEMONT Expendence MDT DAYOBARD NOT DAYOBARD Parto Segure Expendence NOT DAYOBARD	NX NX NX X X X	8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	10X 10X 10X 10X 10X	LIGHTIME ROD OKEATHO LIGHTIME ROD OKEATHO LIGHTIME ROD OKEATHO 800 800	NOT NOT NOT NOT	LICENSED LICENSED LICENSED FILED FILED	VIS VIS ACT INFORMED VIS ACT INFORMED
381 7143875 382 7143907 383 7142762	2801 (SPSTESSE) TORS) CTOL (SPSTESSE) TORS) CSAL (SPALTOTZ_TORS)	SPSTSS2H SPSTSS2H SPUST6272	-21 86798 -23 81713 -21 632317	-46.32359 -46.32359 -46.703938	9 9 9	Santas Santas Sao Paulo	Pieça Don Iális José Suares, 42 - Centro Rua Rodrigues Pais, 183	Routtap	MAST MAST MAST	(2+2+2+2,2+2,3) (2+2+2+2,2+2,3) (5+5+5)	2.24 3.03 2,87 (0,42 + 0,68 + 1,09 + 0.68)	NOT INFORMED NOT INFORMED NOT INFORMED	ND ND ND	Tel Teleconuncações NOT INFORMID	NOK NOK	0X 0X 0X	NDX NDX NDX	NOT WANNENS LIGHTING DOES NOT WORK + LIGHTINGS ROD DAMAGED. LIGHTINGS ROD COIDATED	MDT MDT MDT	FLED FLED FLED	NOT INFORMED NOT
384 7142808 385 7142034 386 7142493	3851 (SP4PA3455 TOPS) CTO4 (SPA34575 TOPS) 2QM (SP43Q3638_TOPS)	SPMPICHS SPCASHER SPM2Q2808	-23.55236111 -22.85396 -23.65476	-46.60051004 -47.06009 -46.6003	2	Sao Paulo Campitus Sao Paulo	Rua Dana, 780 Rua José de Aleccas, 675 - Centra Rua Porceleca, 193 - Vila do Encontro - São Paulo	Roaftap Roaftap Roaftap	MAST MAST MAST	(8+6+6) (2+3+3+5+5) (8+6+6)	2,67 (1,03 + 0,82 + 0,82) 3.08 3,113 (0,782 + 0,782 + 1,548)	NOT INFORMED	N0 N0	NOT INFORMED	NX NX	OK OK	NX NX	NGT DEDIK DAWARD LADIK DAWARD	SOT STRUCTURE OVERLOADED	LICENSED LICENSARLE NOT DIFORMED	NOT INFORMED
387 649466 388 7547750	DEB (CHEDINGS)	PEAKONE	-25.67638889 -8.25444	-09.54325556 -34.90888	78. 75	Contenda Recife	EST. COL. 3D. ANA M. CORDERO, S/N - ND Rua Real da Turre, 282 - Madalena, RecEr - PE	Greatfield Roaftap	SELF-SUPPORT.	76 (5+5+5)	18.49	NOT INFORMED	ND ND	NOT INFORMED Telenant Engestiana De Teleconscações 8/A	NOK	NDK DK	OK NDK	MEDS BITENDED JAATONIK, SERVICES + KENDE FENCE, STRUCTURE SHOWS CRACK NO MANDRE LIGHTING, NO LIGHTING ROD CELENG PAPETWICK, MEDS JANTONIKI,	NDT STRUCTURE OVERLOADED	LICENSAILE FLIED	NOT INFORMED
389 7142670 390 7142586	VOLI (SPWEITER_TORE)	SPW082258 SPW020889	-23.5884 -23.52361	-66.6881	59 59	Sao Paulo Sao Paulo	AVENEDA HERÁCIO LAFER, 702 Rus Albao, 229 - Laga, São Paula - SP	Roattap Roattap	MAST	(1+2+2) (1+2+2+2)	3,891 (0,231 + 1,220 + 2,440) 3,082 (1,837 + 0,133 + 0,777 + 1,155)	NOT INFORMED	ND ND	Telenort Ingestata De Telenort Ingestata De Telenort Ingestata De Telenort Ingestata De	OK OK	OK OK	OK OK	NEEDS JAAPTORING NEEDS JAAPTORING	NOT STRUCTURE OVERLOADED	LICENSED	YES NOT
381 7145472 382 512891 381 511107	INVA2 (SPWINV2588_TORNS) PRES (CM30010035) INVA (CM30010728)	SPWEV2588 SPWP52570 SPWP52088	-23.56081 -23.403568 -23.403783	-46.65187 -46.99115 -46.994729	v v	Sao Paulo Sao Paulo Sao Paulo	ALAMIDA ND CLAND, INLA VEITA - 157 SÃO - 59 R. ANDRO 2018 ANALYEL, NY, VIA FANIN An. DECIMINA 512 - INDROME	Roattap Greatfaid Roattap	MAST POST MAST	(4+4+4) 43 (3.40+5.40)	NOT INFORMED 8.53 2.45	NOT INFORMED NOT INFORMED NOT INFORMED	N0 13	NOT INFORMED SECCEDARL NOT INFORMED	ox ox	0K 0K 0K	OK DK	MICRISTITY OF A FEW ANTI-CORRORING REVAINS: LACORE WITHOUT GUMERALL NO VANNESSE LISSHERG. NEEDS UNFORMATION CONFORMENTS SAME AND ADDRESS.	NOT NOT NOT	LICENSABLE NOT DECEMBER	NOT NOT DE CIENCI
240 51107 254 881185 245 7140750	0404 (CH 80030728) 7%LQ-1 2011 (PEMADID11 TORRE)	MARLEISZO PERADODI	-21.501701 -2.538122222 -8.05805	-46.556729 -46.31566922 -36.95475	57 NA 75	Sao Paulo Sao Luis Noclo	Ar. Bounnas, 617 - Extendents Ruis do Egos, 281 - Centro Ruis Derificatio de Sucia PEO 71 - EFF SOLAR DO SOLIPEINA.	Roditao Roditao Roditao	MAST MAST BAST SELF-SLPPORT.	(14041404140) 5 (14040)	249 19.13 244	NOT INFORMED	N0 N0	MOT INFORMED MOT INFORMED MOT INFORMED MOT INFORMED	ox ox	ox ox	ax ax	MEEDS EXTERNED ANTONIA SERVICES. SEVERAL COPPONENTS CRIZATED. NO WARRAND LISHTING. NO WARRAND LISHTING. NGC MAITONIAL REALTY.	MDT NDT MDT NDT	LICENSARUE LICENSARUE FILIED	NOT INFORMED
286 7143829 287 7142446 288 7142854	2240 [FAIC02240_TORM] BTAK [SPWBL2528_TORM] CAV2 [SPPGB5275_TORM]	PAICC0240 SPWBUS28 SPPGE5275	-1.34999 -23.57368 -24.01983	-48.45756 -46.75277 -46.46942	PA 5P 5P	Brien Sao Paulo Sao Paulo	BETRADA DO TAPANA, SJAL HY I L'1225 - TAPANA Antineda Carsingle, 175 Rea Balanando Camilineu, 100	Greenfield Routtap Routtap	MAST MAST	43 (4+4+4) (6+3+3)	2.1522 NOT INFORMED NOT INFORMED	NOT INFORMED NOT INFORMED NOT INFORMED	N0 N0	NOT INFORMED NOT INFORMED PORTO SEGURD ENGENHARIA	OK OK NDK	OK OK	OK OK NDK	KOD AND WANK LEDNT, RIMULE SWITCH CHO NO ID PLATE. SOM COMPONENTS SHOW GRIDATION SDAN. LADDER DAMAGED. NO WARKENE LEDHTING	NDT NDT NDT	LICENSED LICENSABLE PLIED	YES NOT YES
399 7142462 400 7142720	COM (SPREP3852_TORR) BOSS (SPRER22M TORR)	SPW3P3853 SPW3R284	-23.56227	-46.66273 -46.09205	59 19	Sao Paulo Sao Paulo	AL ITU, 2017 VILA CARAGUNTA CONDOMÍNDO ROPÍCIO JOSÉ HALHEM AN. EXEMPLOS CARAGUNTAS BOS	Rooftap Rooftap	MAST	(4+6+6+2) (6+6+6)	(0,897705+0,633408+0,63340 8+0,226095) NOT DEFORMED	NOT INFORMED	NO NO	NOT INFORMED	OK NOK	OK OK	NDK NDK	NO IDENTIFICATION, MEDIS MANTENANCE AT THE FASTINDES OF SEVERAL TOWERS. MATTENS OF DESTINATION TOWERS, MOLDERS, REST FLATFORMS,	NDT MDT	LICENSED	VES VES NOT INFORMED
401 7144064 402 7143964	OPEN [CEFLAG230_TOPRE] ODE [PAAR122221_TOPRE]	CIFLA0230 PAA0.0221	-3.766629	-38.4696403 -48.35395	C8 PA	Fortaix33 Brien	RUA VELENALDO AGLERA, BEL JAANNA 2200 RUA DESTRITO INEUKITRIAL DE ANANIMERUA, SJN	Roattap Greenteid	MAST SELF-SUPPORT.	(4+5+5) 60	(0,73104+0,73108+0,73104) 6.909138	NOT INFORMED	ND ND	NOT INFORMED	ок	OK OK	NOK OK	PRITIVE, BASES AND PRAVACE TANCHORS NEED MAINTENANCE. MED MAINTENANCE. MEDS MAINORS, REST PLATFORM ENDIE NEEDS REPARE, REST PLATFORM ENDIE	NDT NDT	PILID	NOT INFORMED
403 7140821	THE (PESTAIDOL_TOPPE)	PERTAIDO	-8.06436	- 34, 879722	**	Recte	RUR IMP D FIDRO II., 325 Avenual Guaranges 154	Routtap	MAST	s	NOT DIFORMED	NOT INFORMED	ND	NOT INFORMED	ox	OK	OK .	NEEDS JANEFORDED, NO WAARNAD LIGHTIME BASES SHOW SIGNE OF WEAK CLEAR UP CARLES. COMPONENTS HOW OF CARLES. NO WAARNED LIGHTMEN MATTER	NOT	RLID	NOT INFORMED
404 7140823 405 7149558 409 7142435 407 580400	3022 (PESTA3022_TORM) COV (PEMO3382_TORM) BUSS 15PSCP3842 TORM) OMC (CH40025613)	PERTA3022 PEAF01382 SPW2P3840 PEEPNE360	-8.06311 -8.500037 -23.56307 -7.362638889	-36.879228 -41.411361 -46.6535 -41.3525389	5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	Recte Missio Sao Paulo Gerenano	Avenida Guaranges 154 Mill 2016 VICIATE DI AMAGO, 3 AMERICA PRAETA ERPERTA AMARSI 274 AV BRASE, Nº 230	Rooftap Greenfield Rooftap Greenfield	MAST SELF-SUPPORT. MAST SELF-SUPPORT.	4 43 (5+6+6) 50	NOT DECRIPTION 0.8281 0.8284 C 20120 C 20120 0.8209 C 20120 C 20120 0.001 DIVIDIO C 20120	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED	N0 N0 N0 N0	NOT INFORMED NOT INFORMED NOT INFORMED CLEMAR ENGENHAURA LTDA	8 8 8	OK DK OK	OK NOK NOK	OTHER COMPOSENTS COSDERTED. NOT NOT COMPAGES AT THE LADDER BASE PLATE AND LIFELTRE AND LIFELTRE	NDT NDT NDT NDT	PELED LICENSARIE LICENSARIE LICENSARIE	NOT INFORMED NOT INFORMED NOT INFORMED
408 520993	2134 (CM2000419)	PARADISH	-1.6678	-48.49011	PA	Beien	AV. DR MCRAES 121 NAZARÉ	Routhap	MART	(4+4+2)	(0,7286+0,7286+0,7286)	NOT INFORMED	ND	NOT INFORMED	NOK	OK.	NDK	CLEAR UP CARLES + LADDER COLDATED. MAINTENANCE MICESEARY AT THE REARCH. ANCHORE OF THE MARTY FAITUREDS AND OF THE WARKING LIDHTING	NOT	LICENSED	ves
409 7540675	3058 (PERVG3058_TORNE)	PERVG2258	-8.143272	-34.90816	п	Recte	Rua Major Armandu do Bouza Melo, 430 - 807 Jacunã	Routtap	MAST	(7+7+7)	NOT DECKMED	NOT INFORMED	ND	NOT INFORMED	NOK	OK	NDK	CODATION SSONE. NO WARRAGE LIGHTING	MDT	FILED	NOT INFORMED

8 Keptada Tauat Nama	Nie ID	Geide	tooligade (declara)	Ped. test City	Address	Nils Turns - Outblack	Minucleare Minucleare	100 March (100 )	THE BALL BALL AND	Capacit VIRA Controlled (m1)	Nireclare	Variation Resolution Maker	Cleaning/Janits rive	Closing/Tribanitation	General Minuthere	Adaptations	Adaptations -	Lices Status of Site	alog Status of the Sile's
430 \$76455 TPOD [CH40020543] 433 \$7909 GPG [CH4002343]	Representations PERCENTER	-5.998411111 -7.23285	-35.11871667 -35.31536 -46.63114	NY Panaran PE Sao Jaas Du	RUA CONTINUE, D. COMUNICIA. RUA DOSÉ LIANS, ES - CONTRO - 36/0 20/00 DO CANDRO	Greatfeid Greatfeid	SELF-SUPPORT.	48	1.560811	NOT INFORMED	Capacity (set) N/D N/D	NOT DAFORMED NOT DAFORMED	ox NOK	05 1935	NDK DK	NEEDS JANTICOING, MAXIMO LISHTING NIEDS MUNTRANCE, COMPONENT SHOW CRITERIZON CONTONION INCOM- MUNTRANCE DIE TO GROUND INTERNET	NOT NOT	(Deseral) LICENSED LICENSARLE	Not Decide
412 52477 5742 [CHORDEDIA] 413 579500 5FVC [CHARDEDIG] 414 7342827 8843 [SPWIR4128 TOP93]	CIPLATE?	-1.71289	-86.6004	CI Fortalez	ALL, FRANCISCO AAVES N. 28 SAAFAAA A. VICANIA OF CADIN, 6013 AV DO ESTADO, Nº 3854, CINTRO	Routhup Routhup	MART MART	43 (2+6) (6+6+6)	5.10 5.10	NOT INFORMED	NO NO	NOT INFORMED	NOK	a a	ak ak	WANNEY LIDHTING ABIES CONSERVATION. RETTING LID CEIDATED. REEDS JANTONING RASTS MEED MAINTENANCE OK REPLACEMENT DIE TO CODMITION. MO 3D PLATE.	NOT	NOT DPORMED	NOT INFORMED
414 7143427 B0A1 [SPANA411] TORM 413 532792 ECB (OHIDORETI) 414 518079 P141 (OHIDORETI) 417 641510 ONV (OHIDORETO)	RATINGTING RECEIPTING RATINGTING RATINGTING	-22.5267 -22.8267 -22.8267 -14.23836	-45.827768 -43.3852 -43.38555 -45.73563	SP Star Paul	AV DO BITADO, NY 2864, CINTRO S ANNOLA CRUMON DAMAG. 492 BIA MARKA. D Rul. MR. 30	Rodfup Rodfup Greatfaid Greatfaid	MART POST SELF-SUPPORT.	(4+4+4) 21 40 50	1.89 2.9 5.94	NOT INFORMED	10	ADT INFORMED GETEX SERVICES LTDA 2000AB Beitrers Efficientatura	ok OK OK	ак ак	ak ak ok	CONVERTIE BARE SHOWS CANONS. CONVERTIE BARE SHOWS CANONS. LEISTINGS REPARED REACHING AND ACT MEDIS INITIORING AND ID PLATE.	NOT NOT STRUCTURE OVERLOADED	LICENSING INCENSING LICENSARIE	NOT INFORMED
123     1245255     CT01 (\$P0004514 10898)     429     7245885     CT00 (\$P9905225_T0898)     420     511146     UPO [CM30008145]	SPOCO4504 SPEPCIA25 REMRC2521	-21.5277 -21.18116 -22.8571	-45.78 -47.50506 -42.7824	SP Diateo SP Konza P RJ Ro De Jan	Buil Do. Marano Jarahy Marcondes Perso, 215 - Centro Buil Bole de Molendor, 540 Ruis 42 Quintina Se LOTE 28 PMAR Das Ladoas	Routhap Routhap Greenfield	PORT MAST SELF-SLIPPORT.	10 (1,3+1,3) 12.5	4.02 NOT DECRMED 5.69	NOT INFORMED NOT INFORMED	80 80 80	HURATE NOT INFORMED Porto Seguio Engelstarte LTDA	OK NDK NDK	8	NCK NCK	AGO EQUIPMENT CAEDART OXIDATED REPLACE OXIDATED SUITCHEOMORE LADORE DAMAGE, RECOMMENDED	NOT NOT NOT	NOT DEVEMINE NOT DEVEMINE	NOT INFORMED NOT INFORMED NOT INFORMED
421         881872         6699           422         856355         768 (2%5058639)           423         714131         0010 (59968294_70998)	PRETADOSIS RELACEINE SPECIFICE	-25.4355 -29.4925 -23.4025	-49.25166 -51.5325 -66.6884	PR Curtiba RS Salvador Do SP Salv Paul	Au. Alonso Carango, MP - CHED RH - CURER - PR d LENA COMPEDA - CENTRO - IALUACIÓR DO SAL R. GLEMADOR, SPE	Routhap Greatfield Routhap	MAST SELF-SUPPORT. MAST	(3+3×3) 40 (4+4+4)	1.4M 1.25 NOT DECOMED	NOT INFORMED	NO NO	NOT DEVOLVED	NOK OK	a 05	NDK DK	RECONCRETING THE TOWER RARE, NELDS SANTORNO, REST FLATFORM FARTENESS SHOWS CREATION WANNESS LIGHTING INCREMENTAL TOWER MEDIC CLEANING, NCC	NOT NOT	LICENSED	VES NOT
424 7347883 3079 (PEAP2076_TOP83) 425 7342657 8656 (SPW6L4085_T0993)	PEAJP2079 SPWSLADES	-8.02555	-34.81222 -46.53823	PE Recte	Rua Gacár Petro, 283 - Casa Aravela, Necile - PE Avenda Rapapenta, Scili - Rapopenta - Xila Pisuta	Routhap Routhap	MASIT	(2+2) (8+8)	3.821 (2,49 + 0,68)	NOT INFORMED	NO NO	Teleconuncações 3/A NOT INFORMID	OK NOK	ax ax	NDK NDK	NOT STRUCTURAL PROBLEMS OF THE BUDLEMS DENTIFIED. NEVERAL COMPONENTS SHOW CODDITION SIGNI.	STRUCTURE OVERLOADED NOT	FLED NOT DIFORMED	NOT INFORMED
436 7142711 0020 (SPWSN059_T0993) 427 576000 6004 (CM40020218)	SPUSNOSIO RNPLUKOOI	-23.49944	-05.50822	SP Sao Paul	Rub Valdenar Matting, 136 Rub Filmettal 27	Routtap Greatfield	PORT	(4+4+4) 30	4.589	NOT INFORMED	NO 43	NOT INFORMED	NDK OK	OK NDK	NDK NDK	LIGHTNING ROD DRIVING, STRUCTURE NEEDS MAINTENANCE AND REDVORCEMENT. MEDS INVECTING, WARNED LIGHTING AND LIGHTNING ROD DO NOT WORK. NO 1825 FULTION.	STRUCTURE OVERLOADED STRUCTURE OVERLOADED	LICENSED	VES
428 7144038 BPR (PABLH2142_TOR98)	PARMINGS	-1.44583	-46.69462	PA Brien	Av. General Magalides, 5(N, Reduta	Greatfield	SELF-SUPPORT.	6	NOT INFORMED	NOT INFORMED	NO	NOT INFORMED	ox	ax	ax	COMPONENTS INFORM CREATION ROME, NIEDIS INFORMS, WINTCAL STRUCTURE INFORMATION AND AND AND AND AND AND STRUCTURE BASE SHOWS WORK, MATTING SHORE CORRESPOND LISEMINONS RED DECREMENTIVE.	NOT	LICENSED	vas
N/1         N/2011         Under Unde	SPWRACCS MARLESSE2 SPWRA25H	-23.55362 -23.55362 -23.55956	-46.64087 -46.27495623 -46.65527	SP Sao Paul PA Sao Lun SP Sao Paul	AL FRANKLIK AN AND THE POPULATION OF THE AND AND A THE AND AND A THE AND AND A THE AND	Reaftap Greatfail Reaftap	MAST POST MAST	(4+4+4+2+3) 11 (3+2+2+3)	(5,288+0,288+0+0,113+0) 1.02 (5,94980+0,949404+0,94940 4+0,949404)	NOT INFORMED NOT INFORMED	NO NO	NOT INFORMED Puto Seguro ergentaria NOT INFORMEDADT	S N N	ax Nax ax	NDK NDK DK	WARMENE LISH TING DOES NOT WORK + LIGHTING ROD DAMARGED. PROVESSON OF CLOSENS REQUESED WARMENE LISHTING AND HORE NOT DE LISHTING ROD WORK	NOT NOT NOT	PELID LICENSARIA LICENSARIA	NOT INFORMED NOT INFORMED NOT INFORMED YES
433         7343534         CBD1 [SPWCR3644_T098]           434         7342672         BB2 [SPW04006_T098]           435         7346796         ADD [SPW04007_T098]	SPWC83644 SPW884006 SPW180177	-23.52477 -23.54261 -23.57214	-66.66318 -66.6457 -66.6392	57 Sao Paul 57 Sao Paul 57 Sao Paul	Ruis Banonesa da Bela Veca, 851 (Henel Exe) - Via Congorhad Ruis BINTO PREITAN 186 Ruis APENENI, 519 - ACLIMAÇÃO	Roathap Roathap Roathap	MAST MAST MAST	(3+3+3) (5+5+6) (5+5+5)	NOT DIFORMED (0,231+0,52311+0,344) (0,158448+0,158448+0,1584 8)	NOT INFORMED NOT INFORMED NOT INFORMED	N0 N0 N0	INFORMED NOT INFORMED NOT INFORMED	OK NOK OK	OK OK NOK	OK OK	BUILDERG STRUCTURE AND CONCRETE BASE SHOULD CRACKE WIRKARDS LIGHTING DOES NOT WORK + LIGHTING RED DAMANGED. WIRANDIG LIGHTING DOES NOT WORK	NOT NOT NOT	PELIED LICENSED LICENSED	985 985 985
436         520570         3146 (CH20005513)           437         530842         VC2 (CH2000864)           438         531201         RAC (CH2000841)	ANNALSILAS RESERVICES REPROZES	-3.0806667 -22.817335 -22.865082	-60.0513611 -63.032304 -63.3569325	AN Natawa RD Sas Sarc RD Ro De Jao	Av. 1, 4(h - Rus Barle de Abril (Pré-mener Dum Basca)         Avanza Abril (Pré-mener Dum Basca)           Barla menejae Fenerati, 155 - Barca Balera         Avanza Abril (Pré-mene)           Avanza Abril (Pré-mene)         105 - Barca Balera           Avanza Abril (Pré-mene)         105 - Barca Balera	Greatfield Rouffield Rouffield	SALP-SLIPPORT. HAST HAST	50 (1,80+7.00) (7,5+7,5+3+3+3+2+1,30 )	18.1768 (2.697 + 2.796) 6,927 (2,938 + 0,652 + 1,338 + 0,652 + 0,654 + 0,658 + 0.452 + 0.0591	NOT INFORMED NOT INFORMED	N0 N0 N0	NOT DEPORTED NOT DEPORTED NOT DEPORTED	8 8 8	а а	OK NOK NOK	NO 30 FLATS. WARNERS LIGHTING REEDS NUMTERAACE NOT	NOT NOT NOT	LICENSARIA NOT DECEMBED	NOT INFORMED NOT INFORMED NOT INFORMED
	RODALD141 ROCOLDDDA ROMITERIZ	-22.818278 -22.836081 -22.690624	-41.24778 -41.38123 -41.87654	RU Ro De Jan RU Ro De Jan RU Page	AVMENA VEHT DE JAMERS, JUN - LIAR DO GUVENBACH - RO DE     AMERIO     MARINO     MA	Greatfield Routtap Greatfield	SELF-SUPPORT. MAST SELF-SUPPORT.	30 (4+6+6+2+2+2+2+2+2+2+2+2 1 28.5	4.2 6.512 8.4	NOT INFORMED NOT INFORMED	ND ND	BLMIRX NOT DFORMED	OK OK NOK	ox ox	OK NOK NOK	NOT NASTS SETHOLIT REST PLATFORMS AND SHOW COULD TOOL SCONE. NEEDS SCHEDOLING SHOULDS. STELL TONSK SHOULD SHOTS OF STELL TONSK SHOULD SHOTS OF	NDT NDT NDT	FLIED NOT DIFORMED	NOT INFORMED NOT INFORMED
643         531829         SBC1 (CM30008837)           643         531866         PT12 (CM3000883)           644         531869         SBCB (CM30008781)           645         53289         CMGB (CM3003234)	RJTRSOBRI RJTTCD114 KIMGRO207 SPERSORIS	-22.115275 -22.8004 -23.007368 -23.472935	-43.21056111 -43.2015 -64.010418 -46.525311	RD Tres Ro RD Ro Dr So RD Margard SF Guardia	RLA MAÑO DO HEO BRANCO, 201 AV. MORTED PRACE E BRAN Ilion Califia - MANARATIRA Risk Califia - MANARATIRA Ria Califiantina, 57	Rooftap Rooftap Greenfield Rooftap	TUBLEAR HOURT MAST SELF-SUPPORT. MAST	16.2 (1+5=6) 35 (1+2+0+3)	18.41 MOT DECAMED MOT DECAMED 2.493	ACT INFORMED ACT INFORMED ACT INFORMED ACT INFORMED	N0 N0 N0	NOT INFORMED NOT INFORMED NOT INFORMED	8 8 8	а а а	ak NDK ak	NOT NOT NEEDS INSTITUTIONNE COMPONENTS SHOW CONDATION. IQUIPMENT CARDING CONDATION	NOT NOT NOT	LICENSABLE NOT DECEMBED NOT DECEMBED LICENSABLE	NOT INFORMED NOT INFORMED NOT INFORMED TOR
446         532629         C804 [CR3008785]           647         532783         3082 [CR3008785]           648         53329         RCN [CR3008466]           649         53366         VMC5 [CR3000646]	SPRCS1725 SPW353824 R25L0558 SPWC2323	-23.54291 -23.54291 -22.88662 -23.51975	-66.4281 -46.74282 -48.74282 -46.4824	SP         Sau Paul           SP         Sau Paul           RD         Ru Do Sau           SP         Sau Paul           RD         Ru Do Sau           SP         Sau Paul	RUN KETELU UTURAMA, 324, CURRENO Run POINER, 324, CURRENO Run Alberto Dero, 1627 - Vien Mellar Sundi Rolatojo - Ro de Envero AV. FLAMBORO, 1365 - VIEA ROVA CURIÇĂ	Greathid Greathid Greathid Greathid	SELF-SUPPORT. SELF-SUPPORT. SELF-SUPPORT. SELF-SUPPORT.	40 40 30 40	1.62 1.62 4.44 1.40425	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED	N0 N0 N0	NOT INFORMED SECCEDANS NOT INFORMED	ox ox ox	8 8	OK OK OK	COMPONENTS INFOR COLDATION. NEIGH IANTORNA. COMPONENTS INFOR COLDATION. MECH IANTORNA. MECH IANTORNA. COMPONENTS INFOR COLDATION. MANUAL WHITCHBOAD	NDT NDT NDT NDT	LICENSARIE LICENSARIE NOT DIFORMED	NOT NOT NOT DEFORMED NOT
	RNPOCS758 R08KAS755 RNRUCS727	-5.4298223 -6.775556 -6.27225	-36.453072 -36.7233333 -36.32903	NY Paca Ban NY Santa Do S NY Junda	AUARDEA MANDEL REDREGERS, 555     AUARDEA REDREGERS, 555     AUARDEA REDREGERS, 555     GRUPO ANELA, 0	Greathid Greathid Greathid	POST SELF-SUPPORT. SELF-SUPPORT.	25 40 40	2.03575204 0.58344 2.295	NOT INFORMED NOT INFORMED NOT INFORMED	NO NO	NOT INFORMED NOT INFORMED	OK NOK NOK	OK NOK	NDK NDK DK	VINDALIZED. NO WAANINE LISHTING. COMPONENTIS COM OKEMIÇÃO. COMPONENTIS COM OKEMIÇÃO. MARIE SHOW CRACKE. NEEDE EXTENDED JANTORIJA, SERVICES + NEMIGI CLORING, EOLEMENT	NDT NDT NDT	LICENSARUE LICENSARUE	NOT INFORMED
	PEDPOZIS MASH-G179 PER-MINO	-8.020886 -4.041364 -6.951608	- 34.953323 - 64.470392 - 25.54775	PE Recte MA Sala Mateur Marantia PE Alapanto	Ruk Manoel de Mederos 9 Ruk Binu Lindo Ruk Binu Lindo	Greatfield Greatfield Greatfield	SELF-SUPPORT. SELF-SUPPORT. SELF-SUPPORT.	30 42 50	1.0937 5.6 MOT DPORMED	NOT INFORMED NOT INFORMED NOT INFORMED	NO NO	NOT DECKNED NOT DECKNED CLEMMR	NDK DK	OK OK	NOK OK	CARINET SHOWS CORPORED. MILLION EXTENDED CAMPTORIAL BRAVICES. MANULE SUITOHOMOND DAMAGED. SEM BALIZAMENTO, TORIE SEM PAGE RED. NOT	NOT NOT	LICENSARIA LICENSARIA LICENSARIA	NOT INFORMED NOT INFORMED NOT INFORMED
	PERSONALS	-22.3558889 -21.42333333	-61.6969969	PR Receip	Rua Talio Alexe Margara, Jan Rua Talio Alexe Martine, 3176 - Zona J	Greathid	POST	30	NOT DIVORMED	NOT INFORMED	NO NO	INFORMED	NDK NDK	ax ax	NOK OK	REMAR OF POWER HED NECESSARY EQUIPMENT CARDING OXIDATED ECUIPMENT CARDING OXIDATED +	NOT STRUCTURE OVERLOADED STRUCTURE	UCINIMALE PILIED	NOT
458 65327 5139 (CHIDDESSA) 559 85335 426 (CHIDDESSA) 550 85704 449 (CHIDDESSA) 561 85704 449 (CHIDDESSA) 561 85881 3378 (CHIDDESSA)	PRHEADEL1 MOLENDES BEROOME1 PRNOPL137	-22.413889 -20.3855 -27.59111 -24.6500811	- \$1.9481111 -42.7546 -48.69972 -48.9972	PR         Planuga           MO         Objective           SC         State Sol           PR         Advanuga	Rub Hamatá, 754 - Kennedy - Manugá Receptor HID Jak, NY 127 Rub NOXI MARINDER DERPERSIN, 149 Rub ALCERES BATTETA DAS, 205	Rodiap Granifad Rodias Granifad	MAST MUT-SUFFORT. MAST GUVED	5 20 4 40	2.156 1.21 MOT DEFORMED 1.49	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED	N0 N0 N0	NOT INFORMED SECCOMA NOT INFORMED NOT INFORMED	NDK DK NDK	8 8 8	NOK OK OK	EQUIPHENT CARENET CREATED + GUARANA, ORDATED EQUIPHENT CARENET CREATED NAICE EXTENDED JANTORA, SERVICES, COMPONENTS CREATED, MICE STATUTION, MICH.	NOT NOT	PELID LICENSIABLE LICENSIABLE	NOT NOT NOT
662 660226 4348 (241001001) 663 660425 428P (241001001)	PRE20101	-22.0343667	-52.4961111	PR Annua PR Pat Do Spa		Greatfield Roaftap	SELF-SLIPPORT.	100	NOT DIFORMED	NOT INFORMED	ND ND	NOT DAFORMED	OK OK	ax ax	ak ak	MEDE ANTORYG AND MANYEMACLINEKALSHENT OF CEDENTED COMPONENTS NOT	NOT STRUCTURE OVERLOADED	LICENSARIJE	NDT YES
664         7140689         0049 (PERVT3012_T0998)           665         7140813         6039 (PERV4009_T0998)           666         7140815         6037 (PERV4001_T0998)	PERV73332 R0880/609 R0430/6017	-8.06275 -5.801259 -5.809052	- 34.884442 - 35.211722 - 35.2115	PE Recfe NN Natal	AVENDA COMPE DA ROA VERTA, 312 - CENTRO Rua Noel Ruca Rúis, - Taros Vermelho Rua N. Ino, de Lourdet, 1378 - RJ. Manhatan	Greathaid Routhap Greathaid	SELF-SLIPPORT. MAST SELF-SLIPPORT.	44 2.7 32	2.94 2.475 2.08	NOT INFORMED NOT INFORMED NOT INFORMED	N0 N0 N0	NOT INFORMED NOT INFORMED NOT INFORMED	NDK NDK OK	NOK OK OK	NDK NDK	REMARE SUSTENTATION CELENE WARNES LESITENE DOES BOT WORK + LIGHTING ROO DAVISED NO REST RATIONE AND NO LEVITENE ROO, CELENE SHORE RAD NO LEVITENE	NOT NOT	FILID LICENSARIA LICENSARIA	NOT INFORMED NOT INFORMED
467 7540970 CB2 (SPC654642,1008) 468 7540991 VE1 (SPC654657,1098)	SPCASHED	-22.89805	-47.0518	SP Cangeta SP Cangeta	RUM BRANNITO LIANS, 2000 Rua Deutur Sales de Oliversa, 463 - 168 Mulazziel - Campinas - 5P	Rothap	MAST	(2+2)	(1,724+1,22)	NOT INFORMED	NO NO	NOT INFORMED	NDK	ax ax	NOK	THE TURNS ORIGATED AND EXPOSED WANNES LISHTING DOES NOT WORK + LIGHTING ROD DAVAGED WANNENS LIGHTING DOES NOT WORK + LIGHTING ROD DAVAGED + LADOR DAVAGED + CELLING DOES NOT WORK + SIGHTING ROD DAVAGED + LADOR	NOT STRUCTURE OVERLOADED	LICENSED	vas vas
469 714099 NCS1 [8FCX54882_T099] 470 7141173 SPG [SPWC13663_T099]	SPCASHED SPWCL3663	-22.80805	-47.05666	SP Campita SP Sau Paul	Rua Deutor Querea, 767 - Centro, Campenas - SP Rua Odenne, 520	Roaftap	MART	(3+3+3+3) (\$+6+6+1)	2.605	NOT INFORMED	ND ND	NOT DIFORMED	ox ox	OK OK	NDK NDK	COMPONENTS INFORM OCDATION MONE MEDIC IMPTORING NO MATCHINANCI (MELACIMINT OF CREDITED COMPONENTS	STRUCTURE OVERLOADED STRUCTURE OVERLOADED	LICENSARLE	NOT
471 7343351 IIFAI (SPWIP2579,70898)	\$PM#25.9	-23.53233	-66.6621	57 Sau 794	Largo Péricies, 0, Bava Funda	Reaftap	MAST	3+2+3	(1,293090371+0,739902+0,7 9902)	NOT INFORMED	NO	NOT INFORMED	NOK	ax	NOK	EQUIPMENT CALINET OXIDATED. NO 10 FLATS: BASIS SHOW CALOCK. NO GUARDIAL AT SCHE FARTS. COMPONENTS SHOW OXIDATION SERIES.	NOT	LICENSED	ves
472 7141372 3941 [SPSBO601_T099]     473 7141375 VEE [SPSBO601_T099]	SPEROHOD SPEROHOD	-23.68277778	-66.55111111	SP Sao Bernard Campo SP Sao Bernard Campo	20 Nua Bioga, 266 - Vila Lucitaria, São Bernarda do Campo - SP	Roditap Roditap	TUBULAR HOURT	15.2 (2+0+2+2+2)	2.39	NOT INFORMED	ND ND	NDT INFORMED Telencet Engesthata De Telecomicações 3/A	OK NDK	OK OK	NDK NDK	METALLEC TUBLIAR ANDUNT, MHAT REDUCES THE STRUCTURE STABLITE. EQUIPMENT CARENET OSEIATED + CEUING RPATHATEN	NOT STRUCTURE CVERLORDED	LICENSARIA LICENSARIA	NOT NOT
424 7242482 PED (\$P\$804632,7098)     425 7242465 OCZ1 [\$P\$024634,7098]     426 7242471 TP55 [\$P\$024634,7098]	SPSRO4513 SPSCM634 SPWW2587	-23.7229 -23.6297 -23.56664	-86.56934 -46.56236 -46.66423	SP Sao Berard Careo SP Sao Cartani SP Sao Paul	00         RV191224 ADLO 7 199100, 1224 - ASSUELO - SMO BERNADO OD CANYO -           Sud         Rua Rafael Carria Sangoo, 927 - Santa Paulo - S.C.S           Rua Rafael Carria Sangoo, 1566 - Birla Veta - S.S. Paulo	Rodisp Rodisp Rodisp	MAST MAST MAST	3 5 5	NOT DEVORING	NOT INFORMED NOT INFORMED	ND ND	NOT INFORMED NOT INFORMED NOT INFORMED	OK NOK OK	ox ox	NCK	NOT LISHMENG ROD DAMAGED + NEECS JANETORIAS ODDATION REENA AT THE MART, REET FLATFORM AND AND-HOLENS	NOT STRUCTURE OVERLOADED STRUCTURE	LICENSIABLE NOT DIFORMED LICENSED	NOT NOT INFORMED YES
	SPST35275 SPST35280	-23.98316 -23.96582	-46.20774 -46.332308	SP Santas SP Santas	Ruis Hard to Halo, Lono - and Unita - And Falan Ruis Maria Miana, 872 - Santae- São Paulo An. Dona Ana Costa, 376 - Gonzaga - Santae	Realtap	MAST	(3+3×3) (4+4)	2.493 NOT DECEMBED	NOT INFORMED	ND ND	Telenost Espenharia De Telecomocioles 815 NOT DEFORMED	ok NOK	ox ox	NDK NDK	NOT LADDIK DAWARD	NOT STRUCTURE CVSRLORDED	HLID LICENSARLE	NOT INFORMED NOT
479 7141878 Mill (875755385,70898)	5P5T35295	-23.89431	-46.32687	SP Santas	RUA JULIO DE PRESQUITA, 128 - VELA PARTANI- BANTOR- SP	Realtap	HAST	(6+6+6+1+1)	NOT DEGRMED	NOT INFORMED	ND	NOT DIFORMED	NDK	ок	NOK	EQUIPMENT CARDIET CEDATED = LIGHTNENG ISDO DAMAGED, GENERAL JANTORENS, REDUCKES AND-ONN BY RECONCERTING ALL CONVENTS BASIS (TOWERS AND GAVE) MART GUINS AND ANCHORS CEDEATED, MICH LIT WITH WITH OTHER CEDEATED.	STRUCTURE OVERLOADED	RLID	NOT INFORMED
483 7242929 MAC2 (SPSTSS409, TORM) 483 7242482 WAS (SPATA637, TORM)	SPWTA6132	-23.8607 -23.55296	-46.31687 -46.56225	SP Santas SP Sau Paul	РМСА РАЛИОНА, 8 Кол Арьсания, 1563 - Табарй - 1550 Рали	Kostiap Kostiap	MAST MAST	(3+3+3+4) (5+4+4)	NOT DVORMED	NOT INFORMED NOT INFORMED	ND ND	NOT DEVOLUTION NOT DEFORMED	NDK OK	0K 0K	NCK NCK	LADDER DAMAGED + NEEDS JANTORINE. POWER HOLSIS SHOWS HUMBOLTY, NOT CERLING DPET MITDLE. NEED'S JANTORINE.	NOT NOT	PELIED LICENSED	NOT INFORMED YES
482 7543425 3PA3 (SPW3P3834_T0998) 483 7543438 SP03 (SPW3P3844_T0998)	SPW0P3EM	-23.572782	-46.657295	SP Sau Paul	Alameda Joaquen Bugleto de 1960, 1652 - 30. Paulota - 58o Paulo Alameda Joá, 1167 - Jorden Paulota - 59	Roditap Roditap	MAST	(3+3+3+3) (6+5+5)	3.13 NOT DVIORMED	NOT INFORMED	ND ND	NOT DIFORMED	NDK	OK OK	NOK NOK	ACCESS LADDER SIGNN AND GEDATED. OM MEASURENE DEVICE DESPLAY BROKEN. REPAIR OF POWER FEED NECESSARY. GEVIS CREDITED.	OVERLOADED STRUCTURE OVERLOADED	LICENSED	ves
684         71.42442         0833 [SPW3P3848_T0998]           685         71.42445         PDS2 [SPWP23959_T0998]	5PW2P2848 5PWP22959	-23.5772 -23.534548	-66.6725	SP Sau Paul SP Sau Paul	Rus Maetou Illas Labo, 1960 - Janém Paulota - São Paulo Rus Murte Aleges, 670 - Perdaes - São Paulo	Routhap	MAST	(4+8) (4+8+8+8)	NOT DECEMBER 2.83	NOT INFORMED	ND ND	BHEAKON NOT INFORMED	OK NDK	OK OK	NDK NDK	CALCARLON IFOR AT THE WARTS, BAST PLATCON TRAVELOUSS AND DEPLATES CENERAL SWATCHING. RATION/SKID CREATED	STRUCTURE OVERLOADED STRUCTURE OVERLOADED	NOT DEVORMED	NOT INFORMED
485 7142475 RBA2 (SPARS-609_T099)     487 7142595 SCAT (SPARS-604_T099)     488 7142692 PR02 (SPARS-641_T099)	SPWSH000 SPWSOI043 SPWVH4217	-23.5406 -23.5291 -23.5895	-46.6605 -46.6514	57 Sao Paul 57 Sao Paul 57 Sao Paul	Lago da Anacche, Mi - Papidata - SP Rua Di Gabrel Das Bantos, 769 - Banta Cacita Rua Manzalis Da Ademica, 181	Roattap Roattap Roattap	MAST MAST MAST	(4+4+4) (3+3+3+2) (6+6+6)	NOT DIFORMED 2.454 NOT DIFORMED	NOT INFORMED NOT INFORMED	ND ND	NOT DEPORTED NOT DEPORTED NOT DEPORTED	OK OK	OK OK	NDK NDK	REST PATTORN AND INCO. 507 CEILING SHOWE CRACKL POWER HOUSE SHOWE INFILTRATION, TURING EXPOSED AND DOIDATED. MAST TURE AND REST PLATTORN SHOW OCCUSTOR.	NOT NOT STRUCTURE OVERLOADED	LICENSIED LICENSED FILED	NOT YES NOT INFORMED
489         7142651         MERS [SPURMO305_7000]           480         7142675         VUN1 [SPURMO32_7000]	SPANNOSS	-23.59882	-46.66286	SP Sau Paul SP Sau Paul	Rus Gavota, 22 - Moesta Rus Gavota, 465 - Indiavápola - Xão Paula	Rodtop Rodtop	MAST	(8+8+8) (8+8+8)	NOT DECRMED	NOT INFORMED	NO NO	NOT INFORMED	OK NOK	OK OK	NCK NCK	NOT LIGHTNING ROD DAMAGED, SKID, MARTS, SCHENS AND MATTING SHOW ORDATION AND CORPORED/D.	STRUCTURE OVERLOADED STRUCTURE OVERLOADED	LICENSED	vas vas
esz 7342690 vCD9 (SPWBU773_T0998) esz 7342656 TTE7 (SPWTA4133_T0998)	SPW182773 SPWTA4133	-23.61792 -23.50451	-46.0925	SP Sau Paul	Av. Sandardala, 246 - Monantin Rua amuso malatir, 1898, Vala Gomes Cardon	Reaftap	MAST	(2+6+6) (0+6+6)	2.509 (0,5856+0,5856+1,027989340 )	NOT INFORMED	ND ND	NOT INFORMED	OK OK	OK OK	NDK NDK	GLISS GUIDATED MANT FARTHING FRANKLISS, MART BASE AND REST PLATFORM SHOW CORRELECS. CABLE CLAMPS AND RATES ALSO SHOW	STRUCTURE OVERLOADED	NOT DIFORMED	NOT INFORMED
	CIMODER SPITISZEG	-6.3312M -23.84957	- 38.40525 - 46.32523	Cl Betallo 37 Santas	DESTRETO SATIO FAMAS, O Avenda Vischington Luiz, 41 - Via Matoi - Sarcoc - 59 Sul Rus Nameda São Cartano. 1728 - Santa Mario - S.C.S	Greatfield Roaftap	SELF-SUPPORT.	23 (4+4+5+5+5)	NOT DECEMED	NOT INFORMED	NO NO	TEL TELECOMUNECAÇÕES	ax ax	ax ax	OK NOK	GENERAL INVITCENTS GENERAL INVITCENTS GENERAL INVITCENTS ANCHOR AND MENALIZ SKID PROPILIES INVIDENT AND MENALIZ SKID PROPILIES INVIDENT AND DETAILS	NOT STRUCTURE OVERLOADED	LICENSARLE FILED	NOT INFORMED
	SPUREIDITZ BOSCENSIO382 PRETADESI	-21.538 -27.59367 -25.4	-46.5432 -48.5499 -49.2625	SP Sau Paul SC Plotanopo PR Curitão	REAL PROPERTIES AND CONTRACT AND A DESCRIPTION OF A DESCR	Rodiap Rodiap Rodiap	MAST MAST MAST	(6+6+6) 5 (6+1,5)	1.018 1.34 2.39	NOT INFORMED	NO NO NO	Telencet Expension De Telecommungdec S/A NOT DAPORARD NOT DAPORARD	NDK NDK NDK	ox ox	ox ox	EQUIPMENT CARENET CODATED + NEEDE JANITORING EQUIPMENT CARENET ORIZATED EQUIPMENT CARENET ORIZATED + CREAND INFUTRATION	STRUCTURE OVERLOADED	LICENSED LICENSED FLIED	VES NOT NOT INFORMED
699 7346263 6509 (PRCTADE6, TORIS) 500 7346268 6639 (PRCTADE6, TORIS)	РИСТАДНІЙ РИСТАДНЫІ	-25.41861	-48.34111	PR Cutto	Av-NAGA Senhora da Luo, 1800 - Jardin Sucia - DK Avenda da República, 6464 - Gualia - Curtika - PK	Roathap	MAST	(1,4+2+2+2)	2.262	NOT INFORMED	NO NO	NOT DIFORMED	NOK	OK OK	OK NOK	CLEANS INFLICTNICK BULLING INFLICTNIC DOES NOT HOK WARDENS LIGHTING DOES NOT HOK NECOMMINEED RECOVERITING THE GAY BASES AND TREATING THE CREATION SPOTS	NOT STRUCTURE OVERLOADED	PLIED	NOT INFORMED
501 7146379 5058 (PREDADED, TORRE) 502 7146549 44 (356/2802044_TORRE)	PRIDADIOS	-23.318355	-52.347825 -67.8493333	PR Landson AC Ro Banc	ANTHERA SANTOS DUMONT, SOO-BOA VISTA Rua Fábria Maia, 200 - Jados de Alah - Ro Bianco	Reaftap Reaftap	MAST MAST	(4+4) (3+1,30)	2.383	NOT INFORMED	ND ND	NOT DEVOKINGD	NDK	ox ox	NDK NDK	CELEVE INFECTION AND AND POLIFICATE	NOT STRUCTURE OVERLOADED	HLID LICENSARLE	NOT
503         7146946         1503 (SPC3V2221_TORM)           504         7146956         UB11 (SMCMODEL TORM)           505         7147000         68 (RICKLOBELTORM)	SPCING221 REACHDAIL RECEIDER	-23.5508 -22.04089 -28.38667	-56.3456 -52.09253 -52.17056	SP Carapical R3 R0 Date R8 Carabi De	FUR REVOCATA DE MELLO. \$1	Rostiap Rostiap Rostiap	MAST MAST MAST	10 6 6	1.353 MOT DIFORMED 2.02	NOT INFORMED NOT INFORMED NOT INFORMED	N0 N0 N0	NOT DECKNED NOT DECKNED NOT DECKNED	NDK NDK DK	а а	NDK NDK	LISHTING KOD DAMAGED + CELLING SHOWS HEREDITY EQUIPMENT CALINET CALINET GRANKAL SANTTONING, LISHTING ROD AND THE CLAMP NUTS OF THE REST FLATORIK	STRUCTURE OVERLOADED SCT STRUCTURE OVERLOADED	PLIED LICENSARLE	NOT INFORMED NOT NOT
506 7147290 CMB (MARAKAR_TONG) 507 7147585 255 (SCRUDIN_TONG)	PRIMIPASE SCIECULOSS	-4.245823 -24.999625	- 55. WESSS - 48. 625056	PA Eatuba	Nus Santo Automo - 0º 1689 AV. AV. 89355, 2010	Greatfield	SELF-SUFFORT.	40 (3+3)	NOT DIFORMED	NOT INFORMED	NO NO	NOT DAFORMED	NOK	N2K DK	OK NOK	LIVELINE BHOW CORMANN. NEEDS EXTERENTS DANTORIAL BERVICES. LISEPTING ROD GROUND CALLE BROKEN. TOWER SHOWS CORRORSION SHOPS. WANNENS LICEPTING.	NOT NOT	LICENSARLE	NOT
508 714769 74 [D'BADNI_TOPS] 509 714511 TLO [PEDDII] 7099] 501 53902 PEN [CH200012] 511 52904 TTL (CH2000722]	DPESADING PROSEINS PAPURSON PAPURSON	-15.804178 -8.012988880 -1.54015 -8.0409844	-47.85355 -34.87559444 -48.72761 -46.72761 -46.92755	CP Basila PS Recte PA Excave PA Partners	SHT SAFE QO 4 LT 1 JOINS CAVOO ADMINENT AVANDER PROFESSOR ADDARDE BUZIND, 5/15 - CINTRO BUZINNE FR, NOTI - DEFENSI MADRACHE (MARCA AUXIMUTE) MADA DE FUNDADA CANADA AUXIMUTE)	Ruchup Deveched Deveched Deveched	POST POST MUT-SUPPORT	4	1.4 6.78 MOT SWORMED MOT SWORMED	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED	N0 N0 N0 N0	SETEN ENGINERARIA NOT INFORMO NOT INFORMED	1 8 88 8	88.88	ok ok ok	GEVIS COCEANID	STRUCTURE OVERLOADED NOT SOT	NOT DECRMED	NOT INFORMED NOT INFORMED NOT INFORMED NOT
512         531409         THE2 (CH3000008)           513         531307         THL (CH3009896)           504         53218         acca (CH3000869)           504         53218         acca (CH3000869)	KINGCOGHE KINGCOGHE KINGCOGHE	-23.45894 -22.609 -22.90774	-46.58756 -46.649 -48.110038	3P Sab Paul RJ Recently RJ Notecia	<ol> <li>Xold Carlwyn Anstea, M. Gaapen Raa Douber Curita Ferenza, 70</li> <li>Roa Cel, Morres Centa, 266, Kanai Roa Cel, Morres Centa, 266, Kanai</li> </ol>	Reality Reality Reality	MART MART MART	43 (6+6+0+3+3+3+3+3+2+4+4 +6+6+6) (6+9)	2.83 10.87 2.35	NOT INFORMED NOT INFORMED NOT INFORMED	N0 N0	BORCH NOT INFORMED NOT INFORMED	OK NOK NOK	а а	NX NX	AMAGENE LESHING DARAGED WARNENG LESHING DOES NOT WORK. GLYS CREMIND. LADDIR DAMAGE. WART CHAOS IN THE BLOCK.	NOT STRUCTURE OVERLOADED	NOT DECEMBED	NOT APOINSD NOT N/A
513         512/00         X513         (CH10032780)           514         512/02         PCMA         (CH1003178)           517         512/02         PCMA         (CH1003178)           518         512/02         PCMA         (CH1003178)           518         512/02         PCMA         (CH1003178)           519         512/02         PCMA         (CH1003178)           519         512/02         PCMA         (CH1003178)           519         513/073         2/02         (CH100306/70)	SPECIALIST SPECIALIST SPECIALIST SPECIALIST RTRL0612	-21 58151644 -23 83797 -22 28166657 -23 49754 -22 416826	-65.78051111 -65.78271 -61.7905667 -65.56948 -62.982675	5P 530 Paul 5P 530 Paul 80 Vebnos 5P 530 Paul 81 Terrecapo	PEA XXVO SEALINESSE 278 AV NEO PEONES SIN - SERVICES DO CRUMPO - VALHEA B. DE K.	Greating Greating Routing Greating Routing	SELF-SLIPPORT. POST MAST SELF-SLIPPORT. MAST	40 40 21 41	10.97 5.83 6.95 6.48 2.11	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED	10 13 50 50 50	ALERIAN MOT ENFORMED MOT ENFORMED MOT ENFORMED	a a a a a	R R R R R	ak ak ak ak	COMPOSITION OF A COMPOSITICA OF A COMPOS	MOT MOT MOT MOT	NOT DECEMENT NOT DECEMENT NOT DECEMENT	NOT INFORMED NOT INFORMED NOT INFORMED
520         581445         EVCD [CM40027188]           521         seleto         CAC (CM5004474)           522         s5505         CD8 (CM504575)           523         659012         CD9 (CM504375)	CECEMOJIJ MOPPOSISE MORPORES	-155213889 -1855716 -1838570 -1838570	-60.65647222 -55.51665 -54.29717 -61.16867	CB Cavida NO Palos Dr M NO Brone NO Brone	Rua Tablello Acquito, 3/01- Cardou III Rua BOMAZIA ACQUITA GL'ADOU DI RUA BOMAZIA ACQUITA GL'ADOU DI LA BOMAZIA ACQUITA GL'ADOU DI LA ANTEANS	Great/faid Great/faid Rol/Tao Great/faid	SELF-SUPPORT MLF-SUPPORT SELF-SUPPORT	30 70 (1+d+t+d+t+d+t+d) 30	5.71 11.12 MOT INFORMED NOT INFORMED	NOT IMORNED NOT IMORNED NOT IMORNED	50 50 50 50	NOT INFORMED MOT INFORMED NOT INFORMED BELMERIC INFRAISTRUTURA	8 88 8	8888	OK NG K	WARANG LIGHTING DIDRIANTYS. LIGHTINGS ECO CARMED MEDIS DIPROVEMENT AT LIGHTING ROD AND REST RATION (2012) ROD TANTON	MOT MOT MOT	FILID LICENSIARU LICENSIARU LICENSIARU	NOT BROKE TOM NOT NOT
524 660855 2278 [CH00511144]     525 862287 1518 [CH0051144]     526 66262 UKP0 [H00516489]     527 7140705 BEMS [FECK_LD49 COMB]     528 7140705 BEMS TORMI	PRHEASU27 RSPARESS RSCAMPER PECQLISE SPCAMPES	-23.42243 -28.83275 -28.84411 -8.085 -22.89244	-52.87545 -51.18649822 -52.5669 -34.96697 -47.06221	PR Norman R3 Poto Ale R3 Stat Cabe PS Recfe SP Campita	Rub Neir Albert Martin, 2500 Mun Sharines Sho Archelo Canacani, 100 Rub Alaborato Fallowati, 100 Rub Ala de Neir Veza, 0 Av. Reda de Tajavo, 1511 - Tudro Ganabaio, Campinas - 19	Routhap Routhap Routhap Routhap Routhap	NAST TUBUAR MOUNT MAST MAST MAST	(1,5+1,5+1) 21.5 2 11 (1+2+1+2]	2.2 NOT SWORMED 2.41 3.24 NOT SWORMED	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED	N0 N0 N0 N0	NOT DAPORAD NOT DAPORAD NOT DAPORAD NOT DAPORAD	ы к 19 к 19 к	8 8 8 8	NDK DK NDK DK	COMPONENTS SHOW OCCUPYON NOT OCCUPYONENTS SHOW OCCUPYON NOT WARENO LICENTING. SOME COMPONENTS SHOW OCCUPYONENTS SOME COMPONENTS SHOW OCCUPYONENTS	NOT NOT NOT NOT	FLIED LICENSED LICENSED FLIED LICENSED	NOT 195 NOT NOT INFORMED 193
528 7142033 35A1 (JPCA54865, T0991) 529 7142059 75A1 (SPCA54865, T0991) 539 7142051 75A1 (SPC8500 T0991) 531 7142051 75A1 (SPC8504, T0991) 531 7142051 V801 (SPC85, 557 T0991)	SPEASABAS SPEASAGO SPEASAGA	-22.89294 -22.991553 -24.18441 -23.49851	-47.06333 -46.235457 -46.30722 -46.60727	3P Cangela 3P Galaxia 3P Zachael 3P Zachael	Av. Bioli de Taguno, 1511 - Judiel Gianobios, Canpinas - 19 Rea 166 de Taguno, 1512 - 2010 Ganobio, Canpest Rea 166 de Tagunos, 193 - Canico - Canico Rea 2020 Manusa, 193 - Canico - Tadolen ESTREAN DO SABÃO, 40	Routhap Routhap Routhap	MAST MAST MAST MAST	(2+2+2+2+1) (6+6)	4.553	NOT INFORMED	N0 N0 N0	Telecomicações da MOTIMORMO Porta Segue Experitaria NOTI DA ORMO	ox ox	a a a	NDK NDK	ALMS CALENDARD	NOT SCE STRUCTURE CVSRLOADED STRUCTURE	PLID PLID PLID	VES ADT INFORMED VES NOT
532 7343655 TAM (SPIPOSIS1_TOPR)	SPRP05311	-21.1764044	-47.8226667	SP Riberas Pr	NUM AMÉRICO BRASILLINER, 413	Roattap	MART	(8+6+2) 4	4.179	NOT INFORMED	ND	NOT INFORMED	NDK	ox	NDK NDK	VALUES INFERENCES OF CONTROL NEEDS INFERENCES BEGINNENG COSCILION AT MART 7. SITE WITHOUT WANNES LIGHTING. WANNES LIGHTING COSE NOT WORK, SITE WITHOUTS INFERENCES	STRUCTURE	LICENSARLE	NOT
533         7145844         BOQS (SPSTSSSRD_TOPO)           534         7145859         VLSS (SPSSSSSD_TOPO)           537         7145859         BODS (SPSTSSSRD_TOPO)	5PST35380 5PPG85280 5PST55400	-21.071389 -24.01515 -23.0605	-46.337222 -46.42055 -46.5286	3P Santas 3P Pisa Gai	Rad Vocette Carvalho,21 - Roperdio Rad Nocetoput, 49 - Posta Grando - São Poulo Astronomo Luc. 47 551	Routhap Routhap Routhap	MAST MAST MAST	5 (5+6+6) (5+5)	1.463	NOT IMPORTED	N0 N0	NOT DAFORMED	NDK OK NDK	ox ox ox	NDK NDK	MANTE S AND 2, ITS SCREWS AND ANCHORS SHOW OXIDATION. GUNS COEDATION.	NOT STRUCTURE OVERLOADED	FLID	NOT INFORMED
521         7142659         BOD2 (SPSTSSHOP TONE)           535         7142565         BFUN (SPNSC423A_TONE)           537         7142652         HEAR (SPNHH295, TONE)	SPWSC0234	-23.520903	-6.454235	SP Santos	Na Cruzers, 732, Bara Funda	Routtap	MAST	(8+6) (8+6+6)	12.77+0.451	NOT INFORMED	ND	NOT INFORMED	ы	a	ak Nok	LISHMAN AND DAMAGED RECOMMENDED PRAVINITYS PARTHANCO AT COLATION HICKS OF THIS SUPPORTING STRUCTURE AND INVERT PRESLATION / WRITING LISHTING CARDINT.	MDT NDT	PLIED PLIED LICENSED	101 AFORMO 113
537 7142652 98846 [#9409505, TORE     538 7143000 8FC [MSC094518_TORE]     539 7143184 MS20562 TORE     540 7143675 NCG [RMCT0638_TORE]     541 7143815 CAR4 [SFC83259 TORE]	NIZCON529 NIZCON529 NIZCON529 NIZON562 RINCTO128 SPCA32569	-23.59277 -18.63226 -18.515%389 -22.896329 -23.52205556	-45.65 -43.967% -50.33152778 -43.12265 -47.1051111	SP Sao Paul NG Confine RG United by 5 RJ Nitierio SP Camping	K. Ola, Navyla Aparelia Sulfa, 460 - Via Camartina - Munia - 87     KODOVIA ND 600,04 06 - ARECPOSITO THERMACIONE     Mark CODE: M 444     Na	Rodiap Rodiap Granted Rodiap Rodiap	MART MART MILP-SUPPORT. MART TUBLIAR MOUNT	(5+5+5) (5,5+5,5+5,5+5,5+5) 80 (5+6) 5	3.221 MOT DIFORMED 11.12 (+1,5+2,45) 11.85	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED	N0 N0 N0 N0	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED INSTALCOR	OK NOK NOK OK	8 8 8 8	NOK NOK OK OK	NAMANA LIGHTING CORE NOT WORK + LIGHTING KOD DAMAGED + DANTORNO AGT WAANANG LIGHTING DORE NOT WORK. MAATE WITHOUT EXT PLATICIPS.	NOT NOT NOT NOT	LICENSIARLE LICENSIARLE WITHOUT LISORUACIO LICENSIARLE	NOT INFORMED NOT NOT N/A
641         774.815         2544         00002249         70000           642         774.815         81%1         1919100001         10000           643         774.9127         SGL3         (895012559_70900)           644         774.9127         SGL3         (895012559_70900)           644         774.9127         SGL3         (895012559_70900)	SPECIALISM SPECIALISM SPECIALISM	-21 528000 -8.03875 -22.00886 -23.47626	-47.100111 -34.90425 -47.90558 -46.55392	NP Concess Pl Recte NP Sau Cells SP Sau Cells	RAM BARGO DE JOLINA, 197 - CHAUSTON R. ALARS, 202 - SUB BARK Rua Dava Mara Incetta, 241, 550 Canto - SP Av. São Paulo, 352 - Citade Bristi - Guandibo - SP	Rodiap Rodiap Rodiap	POST MART MART	58 (6+6+6) (6+6+6)	11.00 MOT DECEMED 3.689 2.827	NOT INCOMED NOT INFORMED NOT INFORMED	N0 N0 N0	NOT DECKNED	OK NOK NOK	8 8 8	OK NOK NOK	LASHTNENS RID NOT COMPLIANT PLATFORM/WED CREMTED LADDER DAMAGED. GLATE CXEDATED.	STRUCTURE OVERLOADED STRUCTURE OVERLOADED	ICONSARIA LICONSARIA	NOT INFORMED NOT NOT
545 7144604 OCH [SPWCT3015_T0898] 545 7144622 SPH (SPWT3019_T0898] 547 7144688 COD9 [SPUR23072_T0998]	SPUCT3015 SPW973029 SPW82072	-23.56721386 -23.47668 -23.55	-66.42094683 -66.71428 -66.668M	SP Sau Paul SP Sau Paul SP Sau Paul	Av. 555 Paulo, 352 - Odaže Brait - Gancibos - 5P Roz Tajojapela, 100 Roz Gri, Gorçabest dz Supers 1-68, Portuba Roz Augusta, 65	Routhap Greatifield Routhap	NAST SELF-SUPPORT. NAST	(4+6) 40.7 (3+3×3)	2.406 3.12 2.09	NOT INFORMED NOT INFORMED	N0 N0	Telenant Engenhana De Telecomocacies 8/A Briefal NOT DE/ORMED	а а	OK OK	OK OK	NOT GENERAL INSTORMAL WARNED LIGHTORIAL COMPOSED SHOT COMMUNICATION	NOT NOT	NOT DEFORMED LECENSABLE LECENSED	NOT INFORMED NOT YES
Bet         746888         CODE (PREATURE (CONSIGNE)           568         746518         OPCE (CONSIGNE)         70988           561         746455         STAP (OPELAGETS FORM)         500           551         746455         STAP (OPELAGETS FORM)         500	COAMOUS SCARESS DEBAGES PERCESS	-16.40711111 -77.1285 -15.795 -6.814861	-48.94197222 -52.2436 -67.92222 -41.345522	00 Avapole 8C Avapole 9P Boolds 91 Sala Sala 91 Canabra	VERA VP-3D - CONTROL ADDRESS PROJECTION C. DR. MARPELE (SAMA) - Audresses Lingen UNIT ANN ADDRESS TONS IN T Reduction Control ADDRESS PROJECTIONS IN Reduction Control ADDRESS Projection (SAMA) Ann Marcel Status, de Fástera, D	Greathid Greathid Boatto Roatto Greathid	SELF-SUPPORT. MILF-SUPPORT. TUBLIAR MODAP SELF-SUPPORT.	10 10 10 15 14	18.79 16.79 1.01 1.35 6.71	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED	40 NO 200 NO	NOT INFORMED CAR SETTIN ENGENIARIA SECCOMIL	8 8 8 8	888	ak ak Nak	MEDE BARTORING MEDE EXTINCES BARTORING BOT REALING ESTIMATION AND LEFTING SOCIATION	NOT NOT NOT NOT	LICENSALS LICENSALS NOT DECKNED LICENSED	NOT NOT NOT BEORED YES
552 7147044 R0CH91200_T0998 553 7147026 CMC (P3.800436_T0998)	RDCH961200 P52801436	-18.56833556 -4.589551	-63.62225 -62.862702	RD Campo Nov Rendom		Greatfeid Greatfeid	SELF-SUPPORT.	41	NOT 20/08/MID	NOT INFORMED	ND ND	CAM NOT INFORMED	ok OK	ox ox	OK OK	CONTRACT INTERVAL FOLINEATION BLOCKS NIED CARLES BROOKS. LIGHTYENG ROD CARLES BROOKS. NO WANNES LIGHTYS. WANNES LIGHTYS NOT CONFUSANT - NEED AND CONTRACTOR LIGHTS AND CONFUSANT.	NOT NOT	LICENSARUE	NOT INFORMED
44 714371 807 04751183 70881 553 714779 202 91850002 10881 554 7147773 3274 (95543274 7088)	PERODAL PERODAL PERODAL	-23.33647 -8.199664 -8.099861	-45.87244 -34.905833 -34.875458	N BCAN FE BACK FE BACK	<ol> <li>Can. 365 Socie de Macedio. 345 Ruis Son Structure Mechanol. 355 Ruis Voite e Qualità de Agolito, 189</li> </ol>	Reaftain Reaftain Reaftain	MAST MAST MAST	and a second	2.82 MOT DECAMED 1.424026	NOT INFORMED NOT INFORMED NOT INFORMED	20 20 20	NOT INFORMED MOT INFORMED CLEMER ENGINEERICA	NEX NEX OK	8	905 905 06	INCOMPARY CARDING CODATION TO CREAT OF DEDITION CREATE AND COMPARY LIGHTERS FOO OF SOME MAKES NOT COMPANY	MOT MOT NOT	DOMAND DOMAND FLID	NOT INFORMED NOT INFORMED

No.         No.        No.        No.        No.		Geslication		cation	Address		Minuture			Capacilly							Rilagitations			eating		
D         D        D        D        D        D	# Reycode	Tauna Natio	Nile ID	Latitude (decieval)	Longitude (decimal)	Ped Unit	City	Address	Site Type - Outdoor	Sinclure Type	Height (m)	UEA Installed (m*)	VEA Contracted (m <sup>1</sup> )	Structure Capacity (m*)	Vertical Structure Maker		Chooling/Urbanization	Deneral Minaibare Mafas	Adaptations - Site	Adaptations - Bineitare	Status of Site Licenses (General)	Status of the Statis Environmental Licenses
N         N	557 7147859 558 7150004			-33.152537 -22.33693	-52.138538 -43.554939	NS NO	Vassaults			MAST SBLF-SUPPORT.	9	NOT SPOKNED	NOT INFORMED NOT INFORMED	80 N	NOT INFORMED Porto Seguio Engenharia LTDA	OK NDK	OK NOK	NOK OK	NOT MEDIS EXTENDED JANTORIAL SERVICES	MOT MOT		NOT INFORMED
N         N        N         N         N <td>560 5210%</td> <td>CINCA (CM20006728)</td> <td>PAPUPODA</td> <td>-5.064247222</td> <td>-50 0004 No11</td> <td></td> <td>Santana</td> <td>RUA FRANCISCO REPORT 28 - SANTANA RUA FRANCISCO REPORT 28 - SANTANA</td> <td>Greenhed Greenhed</td> <td>SELF-SUFFCRT</td> <td>ACT INFORMED</td> <td>0.0185</td> <td>ACT INFORMED</td> <td>11.06</td> <td>MOT INFORMED</td> <td>ă.</td> <td>a a</td> <td>10X</td> <td>Not Name and States of Contract of Contracts</td> <td>MOT</td> <td>LICENSIARLE NOT RECENCE</td> <td>NOT INFORMED</td>	560 5210%	CINCA (CM20006728)	PAPUPODA	-5.064247222	-50 0004 No11		Santana	RUA FRANCISCO REPORT 28 - SANTANA RUA FRANCISCO REPORT 28 - SANTANA	Greenhed Greenhed	SELF-SUFFCRT	ACT INFORMED	0.0185	ACT INFORMED	11.06	MOT INFORMED	ă.	a a	10X	Not Name and States of Contract of Contracts	MOT	LICENSIARLE NOT RECENCE	NOT INFORMED
N         N        N         N         N         N																				STRUCTURE		
N         N        N         N         N         N	563 531604	CONQ [CHORDING]	ROCESSE 181	-22.945329	-41.362971	ю	No De Janezo	Eldr. Mal. Miguel Salazar Mendec de Monaes, 1409	Rottap	маят	5	2.007	NOT INFORMED	NO	NOT INFORMED	NOK	œ	NOK	EQUIPMENT CARDINET OCEDATED. DESCRIPTION SPOTS AT THE LOCONUTS OF THE PASTERIOUS SCREWS OF THE BRACINGS AND THE STRUCTURE SUPPORT CREDER.	NOT	NOT DECRMID	NOT INFORMED
N         N	564 532022	0512 (CM30006252)	KIPHODOSE	-22.879088	-42.445684	10	Ro De Jaceno	Rub Childle, 245, Flidde Misuel	Bollan	MART	(8+4+6)	(3.77+0.60+0.67)	NOT INFORMED	ND	NOT INFORMED	œ	×	NOK		NOT	NOT DECEMIN	NOT INFORMED
N         N	565 532520					ĸ	NORG	Rua São Pedro, nº 128, Centro	RealDap			NOT DEGRMED		NO		œ	08	OK.	NOT	MOT	ATTHOUT LEGESL	ŅΛ
N         N        N         N         N         N						57													COMPONENT'S SHOW ORDATION SIGNS.			
N         N	568 647220	038 [045004853]	NOPPOS625	-18.59074722	-66.50220556	200	Patios De Meas		Greekfald	POST	40	2.76	NOT INFORMED	NO	NETWORK TELECOM	0K	OK	OK.	NEEDS JANTORING.	MOT	LICENSABLE	NOT
	569 648887	315P (CM\$0056457)	PRIDADZIS	-23.305657	-51.178373	PR	Landsha	Rub Tälderbeil, 501	Routtap	TUBLEAR MOUNT	15.33	5.72	NOT INFORMED	12.96	NOT INFORMED	OK	OK	0K	SWEITORING.	NDT	FILED	NDT
No.         No.        No.        No.         No.	570 650088	021P (CM80056447)	MTCBM0021	-15.592404	-56.0915	MT	Custa	AV HERT RUBING DE MENDONCA - BAU	Routhap	aunio	10	(1,65 + 0,43)	NOT INFORMED	ND	NOT INFORMED	œ	ox	мж		NDT	LICENSARLE	NOT
N         N        N         N         N         N	\$71 650620	13%. [CMi0056167]	83PL12139	-31.754321	-52.315M	15	Pelitas	Av. Doringos de Almeida, 1982- Areal	Greenheid	SELF-SUPPORT.	40	4.51	NOT INFORMED		Breck	ак	OK	ак	RECORDER TREATMENT, REPLACE WARDING	MDT	LICENSAILE	NOT
Barry         Barry <t< td=""><td>572 651360 573 6544M</td><td>189P (CHE0055915)</td><td>DFERACCES ESC TROUCH</td><td>-15.811593</td><td>-47.903387</td><td>DP</td><td>Bisela Cachoess Dr</td><td></td><td>Routhap</td><td>MAST</td><td>5</td><td>2.29</td><td>ACT INFORMED</td><td>500 ND</td><td>NOT INFORMED ALT ALGAR TELECON LESTE</td><td>NOK</td><td>ox a</td><td>OK NOV</td><td>EQUIPMENT CARLINET OXIDATED</td><td>MOT</td><td>NOT DECRMID</td><td>NOT INFORMED</td></t<>	572 651360 573 6544M	189P (CHE0055915)	DFERACCES ESC TROUCH	-15.811593	-47.903387	DP	Bisela Cachoess Dr		Routhap	MAST	5	2.29	ACT INFORMED	500 ND	NOT INFORMED ALT ALGAR TELECON LESTE	NOK	ox a	OK NOV	EQUIPMENT CARLINET OXIDATED	MOT	NOT DECRMID	NOT INFORMED
D         D	576 561061 575 5610	BAR (CHSOMESH)	19/7A3002	-20 29791		13	Esperant Vitora		Scottage Boottage	7000 MLG7	1	ANT DECEMED	NOT INCOME	10	S/A Tel Telsconumacións	105		10K	IQ CARANT DODATA	- 101 MOT	FILIED	NOT INFORMED
No.         No.        No.         No.         No. <td>575 880080</td> <td>BS7 (KSPAEGES7_TOPPE)</td> <td>KSPAIDES7</td> <td>- 32.08097</td> <td>-51 22861</td> <td>85</td> <td>Poto Nega</td> <td>AMINEA CREATING DO MA, 2121</td> <td>Realisp</td> <td>MAST</td> <td>2</td> <td>NOT DECKMED</td> <td>NOT INFORMED</td> <td>80</td> <td>NOT INFORMED</td> <td>NOK</td> <td><u>88</u></td> <td>NOK</td> <td>BQ: CARDWIT COUDATED + LL ROD INOPERATIVE: SARETITUTE MATTING.</td> <td>MOT</td> <td>LICENSARIE</td> <td>NOT</td>	575 880080	BS7 (KSPAEGES7_TOPPE)	KSPAIDES7	- 32.08097	-51 22861	85	Poto Nega	AMINEA CREATING DO MA, 2121	Realisp	MAST	2	NOT DECKMED	NOT INFORMED	80	NOT INFORMED	NOK	<u>88</u>	NOK	BQ: CARDWIT COUDATED + LL ROD INOPERATIVE: SARETITUTE MATTING.	MOT	LICENSARIE	NOT
N         N        N        N        N        N        N      <	\$77 BB1342	POR	MTCBAD164	-15.623667	-56.087222	MT			Realitap	MAST	5	2.43	NOT INFORMED	ND	NOT INFORMED	NOK	OK	NDK	GROUND NEEDS REPAIR. SCREWS AND REST PLATFORMS CODDATED.	MDT		NOT
N         N        N        N        N        N     <	579 7140682	3222 (968/03222_108/8)	PERV03222	-8.120139	-34.900611	-			Routop	MAST	1	NOT INFORMED	NOT INFORMED	ND	SEIMENS	NDK	OK	NDK	GLIMERAS CODASD. WANING LIGHTING NEEDS MAINTENANCE.	NOT		NOT INFORMED
No.         No.        No.        No.         No.							Salicatas Des Guarantes Talicatas Des															
N         N        N         N         N         N	581 7140761 582 7141352			-23.52915	-65.68023	P1 32	Sao Paulo	Ruo Turbissi, 1863, Perdaec	Routhap	MAST	5 (3+3+2+2)	(0,79+0,18+0,79+1,29)	NOT INFORMED NOT INFORMED	NO NO	NOT INFORMED	OK.	OK OK	NDK NDK	NOT	MOT	RLED	
No.         Normal wild								HIS INCOMES PRAYED CONSIGNED AT MARKING - BARE MARKING TH				NOT DUFORMED										
No.         Normal Mark												4277261							BASE OF THE MAST TURES, LIFELINES AND	STRUCTURE		
No.         Result of all state         Res						-													RESPECTIVE TURNELIDULES AND PARTIAINES CLAMPS SHOW CORPATION.			
No.         No.        No.         No.         No. <td></td> <td>_</td> <td>SKED PROFILES SHOW COEDATION SISNS. MASTE SHOW HEEH REEK OF COLLAPSE</td> <td></td> <td></td> <td></td>																		_	SKED PROFILES SHOW COEDATION SISNS. MASTE SHOW HEEH REEK OF COLLAPSE			
Norme         Norma         Norme         Norme <t< td=""><td>587 7142775</td><td>GALY (KOGALELAZ TORRE)</td><td>59490340 505502142</td><td>-22.830507</td><td>-46.42333</td><td>39</td><td>Sao Paulo Ro De Janeiro</td><td>Ar. Charrie do Sul, 1030 - Canade An. Wite de Janeno, 3/39</td><td>Realize Greenfield</td><td>RAST</td><td>(4+4+4+4)</td><td>1.986</td><td>NOT INFORMED</td><td>NO NO</td><td>NOT INFORMED</td><td>ox ox</td><td>ok ok</td><td>OK OK</td><td>THE ROOF SHADE.</td><td>OVERLOADED</td><td>NOT DECRMUD</td><td>NOT INFORMED</td></t<>	587 7142775	GALY (KOGALELAZ TORRE)	59490340 505502142	-22.830507	-46.42333	39	Sao Paulo Ro De Janeiro	Ar. Charrie do Sul, 1030 - Canade An. Wite de Janeno, 3/39	Realize Greenfield	RAST	(4+4+4+4)	1.986	NOT INFORMED	NO NO	NOT INFORMED	ox ox	ok ok	OK OK	THE ROOF SHADE.	OVERLOADED	NOT DECRMUD	NOT INFORMED
District	589 754642	011 (CEPLA0171_TORM)	CIFLATIN	-3.72978	- 38.52563	а	Portaikaa	Rak Seta Madureta, 1879	Routtap	MAST	(2+2+2)	(9,578+6,594+7,733)	NOT INFORMED	ND	NOT INFORMED	ак	OK.	NOK	MANTS AND TOWERS OCCURED. BLAMS SUPPORTING THE TOWERS SHOW CRACKS AND RESC OF COLLAPSE. LIGHTNEDS ROD AND WERENES LIGHTING CAMAGED.	NDT	FILID	NOT INFORMED
N         N	590 7144175 591 7144252	OTTO PROVADINE TOWARD	000000000	-16.1946939 -15.81229	-47.454194 -47.959275	DP	Bischa Bischa	(FE-DF) SIA SAPS AS D	Greatfield Roofboo	8457	60 5	2.04 (1.22 + 0.69)	NOT INFORMED NOT INFORMED	50 50		0K OK	ox ox	OK OK	NO WARMING LIGHTING COMPONENTS SHOW OXIDATEWISION	MOT MOT	NOT DECRMID	NOT INFORMED
N         N	582 7144622 583 7144631	BLBF (AMPEGRAN_TORRS)					Precidentia Precidentia Providentia	Rua Teffi, 22 - Absan			75.6	2.17		NO NO	NOT INFORMED	NDK	NOK	NOK	NEEDS EXTENDED JANFORIAL SERVICES. TRACTINE OF ORDATED COMPONENTS.	MDT MDT	LICENSED	VES VES
B         C	594 7149955 585 7145167		GOGNADORE GOGNADIAZ	-16.67138	-49.35694	80	Guara		Realize	MAST	(4+2)	1.291	ACT INFORMED	ND ND		OK NOK	ox ox	OK NOK	STRUCTURE PEOLONS OF THE	MOT MOT	LICENSARUE	NOT NOT
N         N         Norma         Norma </td <td>595 7145982 587 7147095</td> <td>855580200 TORR 23 (85940022_TORR)</td> <td>R05080205 R07A80023</td> <td>-20.3437</td> <td>-54.0809 -51.38789</td> <td>53 83</td> <td>Riss Colored Poils Negre</td> <td>BR 290. SANTA MARGARDA DO SUL ESTRADA DA SERVINA SUN Av. Fersando Fersar, 2005</td> <td>Greenfaid Realitiep</td> <td>NUP-SUPPORT. PONT</td> <td>40 (4+6+6)</td> <td>5.21 NOT INFORMED</td> <td>NOT INFORMED NOT INFORMED</td> <td>ND ND</td> <td>CAM NOT INFORMED</td> <td>OK OK</td> <td>ox ox</td> <td>ax ax</td> <td>NO1 NO1</td> <td>MOT MOT</td> <td>FILIED LICENSABLE</td> <td>NOT INFORMED NOT</td>	595 7145982 587 7147095	855580200 TORR 23 (85940022_TORR)	R05080205 R07A80023	-20.3437	-54.0809 -51.38789	53 83	Riss Colored Poils Negre	BR 290. SANTA MARGARDA DO SUL ESTRADA DA SERVINA SUN Av. Fersando Fersar, 2005	Greenfaid Realitiep	NUP-SUPPORT. PONT	40 (4+6+6)	5.21 NOT INFORMED	NOT INFORMED NOT INFORMED	ND ND	CAM NOT INFORMED	OK OK	ox ox	ax ax	NO1 NO1	MOT MOT	FILIED LICENSABLE	NOT INFORMED NOT
N         N	588 7147472 589 7147533	729 [DF85A1729_T0998]	599C11882 DF85A0729	-23.32361 -15.29626	-46.956.38	57	To cave Brasila	Ria Giberta Manaza, 296 - Vila Fornaca, Jacanei - SP ESPLANEZA DOS MONESTERIOS, BLOCO R, MALA S7 - MENESTERIO DAS	Routtag	MAGE	5	2.493	NOT INFORMED NOT INFORMED	ND ND		NOK	ox X	NDK NDK	LADDER DAMAGED BQ. CARINET CKEDATED + LADDER DAMAGED	MOT	NOT INFORMED	NOT INFORMED
No.         Normanne	600 7149041	B37 (REPARTS TOPPE)	RSPA80837	-30.0669778	-51.2337083	88	Poto Nega	AVENIDA JOÃO PESSOA, 2050 - SANTANA	Routtap		6	3.26	NOT INFORMED	N/D	NOT INFORMED	OK.	OK	NDK	SETS SUPPORTING PRISONRY SHOWS CRACKS	MDT	LICENSED	188
No. 0         No. 0 <th< td=""><td>602 54762538 602 647625</td><td>CAEL (SPACEINS) TORREL PFAA (CHS0047886)</td><td>SPACE0551 MOFAM6020</td><td>-23.62529 -21.41225</td><td>-45.62456</td><td>57 MG</td><td>Sao Paulo Faria</td><td>Rub Barbo da Triunfo. 276 RUA SANTA IDVIDUES, S/N</td><td>Realize</td><td>MAST SELF-SUPPORT.</td><td>40</td><td>(1.01+1.19+1.01+1.97) NOT INFORMED</td><td>NOT INFORMED NOT INFORMED</td><td>ND ND</td><td>NOT RECEMBE NETWORKER TELECOM</td><td>0K</td><td>ox X</td><td>MOK OK</td><td>STRUCTURE SHOWS SPOTS OF DETERDRATION.</td><td>MOT</td><td>LICENSIE</td><td>VES NOT</td></th<>	602 54762538 602 647625	CAEL (SPACEINS) TORREL PFAA (CHS0047886)	SPACE0551 MOFAM6020	-23.62529 -21.41225	-45.62456	57 MG	Sao Paulo Faria	Rub Barbo da Triunfo. 276 RUA SANTA IDVIDUES, S/N	Realize	MAST SELF-SUPPORT.	40	(1.01+1.19+1.01+1.97) NOT INFORMED	NOT INFORMED NOT INFORMED	ND ND	NOT RECEMBE NETWORKER TELECOM	0K	ox X	MOK OK	STRUCTURE SHOWS SPOTS OF DETERDRATION.	MOT	LICENSIE	VES NOT
No.         No. <td>603 655692</td> <td>TNL1 [CM50045372]</td> <td>M050N6130</td> <td>-21.53665</td> <td>-41.30046</td> <td>MG</td> <td>Goana</td> <td>NAA JOAO DE SOUZA KBEDRO, O</td> <td>Greenfield</td> <td>SELF-SUPPORT.</td> <td>32</td> <td>NOT SPORMED</td> <td>NOT INFORMED</td> <td>ND</td> <td>PROMONTIS</td> <td>OK</td> <td>OK.</td> <td>OK</td> <td>MEEDS JAATTORING AND DESTRUIATION OF REST RATFORM</td> <td>NOT</td> <td>LICENSED</td> <td>185</td>	603 655692	TNL1 [CM50045372]	M050N6130	-21.53665	-41.30046	MG	Goana	NAA JOAO DE SOUZA KBEDRO, O	Greenfield	SELF-SUPPORT.	32	NOT SPORMED	NOT INFORMED	ND	PROMONTIS	OK	OK.	OK	MEEDS JAATTORING AND DESTRUIATION OF REST RATFORM	NOT	LICENSED	185
No.         No. <td>101 11109 101 147259 105 885951</td> <td>RCST [CRIGOEARD] 0001 (CRIGOEDRAN) PCCR</td> <td>MOEV92222 MOEV922222 MOEV922222</td> <td>-18.86708 -16.88909 -18.578</td> <td>-42.36448 -40.92729 -47.875</td> <td>P03 P03 P03</td> <td>Execution Concernation Cardination Reco</td> <td>NEA VIRGENO CAMERO, 463 REA SANCE, 198 REA SANCE, 198</td> <td>Greatined Greatined Greatined</td> <td>MUT SUFFOR</td> <td>75 42 60</td> <td>NOT DEPORTED</td> <td>ACT INFORMED</td> <td></td> <td>NET ACCOUNT AL</td> <td>0K 10X</td> <td>ok ok NOK</td> <td>OK OK NOK</td> <td>NEEDS SALTORING NEEDS SALTORING NEEDS EXTINDED SALTORIAL SERVICES</td> <td>MOT MOT</td> <td>LICENSIARUE LICENSIARUE</td> <td>925 927</td>	101 11109 101 147259 105 885951	RCST [CRIGOEARD] 0001 (CRIGOEDRAN) PCCR	MOEV92222 MOEV922222 MOEV922222	-18.86708 -16.88909 -18.578	-42.36448 -40.92729 -47.875	P03 P03 P03	Execution Concernation Cardination Reco	NEA VIRGENO CAMERO, 463 REA SANCE, 198 REA SANCE, 198	Greatined Greatined Greatined	MUT SUFFOR	75 42 60	NOT DEPORTED	ACT INFORMED		NET ACCOUNT AL	0K 10X	ok ok NOK	OK OK NOK	NEEDS SALTORING NEEDS SALTORING NEEDS EXTINDED SALTORIAL SERVICES	MOT MOT	LICENSIARUE LICENSIARUE	925 927
No.         Approx         Approx        Approx        Approx	607 647631	MOLE (CM 80047631)	MQLIKK5723	-18.82138	-47.82456	MG	theata		Greatheid	SELF-SUPPORT.	90	33.07	NOT INFORMED	N/D	BRASESAT HARALD S.A.	0K	OK .	ак	STRUCTURE AND PLATFORM MEED PAINTING	MOT	NOT DECRMID	NOT INFORMED
10         10         0.000	609 7146773 600 658632	27 (REPAROET TORR) PORE [CHOOL3702]	KUPABOOT7 MGGLR6667	-32.085365 -18.8433	-51, 177371 -46, 79333	#3 790	Poto Near Guirarana	SZZ, ENDOLE, 114 RUR TUPDAMBAS, 5151	Great/Seld Great/Seld		50 32	2.81 NOT INFORMED	NOT INFORMED NOT INFORMED	22	NOT INFORMED NETWORKER TELECOM	ă	a a	OK OK	LIGHTNING KOD NEEDS MAINTENANCE	101 101	NOT RECEMBED	101200040
D         D	612 658009	ICE (CM50043308)	RIPABODIS MICAA6385	-35.10923 -19.0726	-51 26187 -61.8649	83 993	Poto Alegra Capitalo Andede	KAN INTERNA DO CLUBE DOS JANGACEIROS AMIREDA LEVINDO DEKS, S/N	Greatfaid Greatfaid	SELF-SUPPORT.	24 30	NOT INFORMED	NOT INFORMED NOT INFORMED	N0 N0	BNL BECCEDINL Brieful Industria e conducto	0X 0X	ox ox	ax ax	NOT NOT	MOT	LICENSED	115
D         D	623 646844 604 7142923	ONC INCOMPAGE TORRES	MODIFIED	-20.3109	-41.71045	M3 M3	Ouro Mirco	AMINDA MR. XMOUN DANI TOLEDO, SIN	Greetfeld	POST	60 42	NOT DECAMED	NOT 11/06/450	ND	de Produzos Rectalógicos LTDA MOT INFORMED	NOK	ox NOX	705	MANNING LIGHTING NOT COMPLIANT NEEDS EXTENDED SAVETORIAL SERVICE	N MOT	FILED	9/21
No.         1000000000000000000000000000000000000	615 7146529 605 658pee	47 (REPARONT_TORRE) GED (CHEDHIDHE)	KSPA80047 MOCREA1W	- 30, 189	-51.162	#3	Poto Negra Coronel Math	INTING CHIPEU DO SOL, 681 BUN DOSA SERVICIA APARCEA, 5/17	Realitap	MAST	6	NOT DECRMID	NOT INFORMED	ND ND	NOT INFORMED	ox ox	0K	NOK OK	NO WARNENS LIGHTING. REST PLATFORM CKCAMIED.	MOT	LICENSED LICENSED	VES NOT
1         1						PRG												OK	MIED'S EXTERCED JANTORIAL SERVICES. LIGHTNEYS ROD NOT COMPLIANT.			
District												ACT INFORMED							NEEDS EXTENDED WATCHES NOT WORK.			
No.         No. <td>609 7147062 620 7144840</td> <td>D368 (ROPVO3D36_TORRE)</td> <td>NCT1L0640 RCPV00036</td> <td>-28.8339 -8.76386</td> <td>-61.88481</td> <td>3C 80</td> <td>Tinke Do Sul Porta tiviho</td> <td>RUR CARLOS SAVE, S/N RZ. GOV. JORGE TEDERRA, 629</td> <td>Greetfeld Greetfeld</td> <td>SELF-SUPPORT.</td> <td>40</td> <td>NOT INFORMED</td> <td>NOT INFORMED</td> <td>N/D N/D</td> <td>METALÚMERCA LITRA BRASEL SAT</td> <td>NDK OK</td> <td>ox ox</td> <td>OK OK</td> <td>NEEDS MANY EARCH. LIGHTNEVG ROD GROUNDONG BROKEN</td> <td>MOT</td> <td>NOT DECRMID</td> <td>NOT DEMONITION</td>	609 7147062 620 7144840	D368 (ROPVO3D36_TORRE)	NCT1L0640 RCPV00036	-28.8339 -8.76386	-61.88481	3C 80	Tinke Do Sul Porta tiviho	RUR CARLOS SAVE, S/N RZ. GOV. JORGE TEDERRA, 629	Greetfeld Greetfeld	SELF-SUPPORT.	40	NOT INFORMED	NOT INFORMED	N/D N/D	METALÚMERCA LITRA BRASEL SAT	NDK OK	ox ox	OK OK	NEEDS MANY EARCH. LIGHTNEVG ROD GROUNDONG BROKEN	MOT	NOT DECRMID	NOT DEMONITION
No.         No. <td>121 656314 122 646345 123 647651</td> <td>0001 (CM1004(806) 313 (CM10044032) 0024 (CM100470%)</td> <td>MONTPENSIO MONTACIONI MONTPENSIONI</td> <td>-18.88675 -20.0058 -19.6509</td> <td>-45.02067 -43.9687 -42.8684</td> <td>993 993</td> <td>Neva Lina Antonia Das</td> <td>PRACA XXXX PRAVA CARECOLD SUM RODOVA XXXXED SUBTOOL (\$8.040, 10.545, 5/N RUA BLO HORZONE, SIN</td> <td>Greatfaid Reaftap Greatfaid</td> <td>POST NAST MUT-SUPPORT.</td> <td>20 6 40</td> <td>NOT INFORMED</td> <td>NOT INFORMED NOT INFORMED</td> <td>22</td> <td>NOT INFORMED NOT INFORMED NOT INFORMED</td> <td>NX NX</td> <td>OK OK NOK</td> <td>NCK NCK</td> <td>NOT LIGHTN, ROD DAMAGED = LADDER DAMAGED MEEDS EXTENDED JANTORIAL SERVICES</td> <td>MOT MOT</td> <td>LICENSARUE LICENSARUE</td> <td>N21 N21 113</td>	121 656314 122 646345 123 647651	0001 (CM1004(806) 313 (CM10044032) 0024 (CM100470%)	MONTPENSIO MONTACIONI MONTPENSIONI	-18.88675 -20.0058 -19.6509	-45.02067 -43.9687 -42.8684	993 993	Neva Lina Antonia Das	PRACA XXXX PRAVA CARECOLD SUM RODOVA XXXXED SUBTOOL (\$8.040, 10.545, 5/N RUA BLO HORZONE, SIN	Greatfaid Reaftap Greatfaid	POST NAST MUT-SUPPORT.	20 6 40	NOT INFORMED	NOT INFORMED NOT INFORMED	22	NOT INFORMED NOT INFORMED NOT INFORMED	NX NX	OK OK NOK	NCK NCK	NOT LIGHTN, ROD DAMAGED = LADDER DAMAGED MEEDS EXTENDED JANTORIAL SERVICES	MOT MOT	LICENSARUE LICENSARUE	N21 N21 113
Distribution         Distribution<	624 649809	HIRED (CM 60048028)		-16.54986	-42.509455	MG CPM	306eeopods	MORD DAS ANTENAS, S/Nº	Greenfaeld		60	NOT INFORMED	NOT INFORMED	N/D	NOT INFORMED	OK	OK .	OK.	WAANING LIGHTING NOT COMPLIANT - COMMAND BOARD NEEDS MAINTENANCE	MOT	NOT DECRMID	NOT INFORMED
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	424 878575 427 579422	CMC 2000 (CM400221133)	MIPOLA188 CENEOL79	-11.0000 -22.248317 -7.2047	-41.82221 -44.834313 -29.3291	900 900 CE	Pauso Alta Pauso Alta Juazerro Do Rorte	RUA AVCRED FRMA, 4000 RUA 20ÃO LIMO DA SEJVA, S/N AVV. PADRE CIERDA 441, ROPICID 01	Greated Greathed	SUP-SUPPORT.	42 42	ACT INFORMED NOT INFORMED	ADT INFORMED NOT INFORMED NOT INFORMED	20	MOT INFORMED METWORKER TELECOM BELMERER LTDA	ax ax		OK OK	NOT NOT	MOT MOT	NOT SPORMED	NOT INFORMED NOT INFORMED
D         A         A         A         Control         Control <thcontrol< th="">         Control         Co</thcontrol<>	428 646872 829 648120	LINCA (CHICOHIONE) ST20 (CHICOHIPEH)	MCENP6404 MCENP6404	-18.62155 -18.17572	-42.32775 -41.86441	993 993		ALTO DO HORIO DA ISTAÇÃO TELIMAR. AN, SÃO CREITONÃO, SIN	Greeched Greeched	SELF-SUPPORT.	30 40	NOT INFORMED	NOT IMPORTED NOT IMPORTED	50 50	SECCOVA EPETAL	OX OX	ox ox	OK OK	NO1 NO1	MOT MOT	LICENSED	10.5
International         Internat	631 661189	C005 (CM10046455) TORL (CM50048994)	AGETERAT MOLIEUSION	-15.3406 -17.255	-62.163	MG		RUA EXPERIMENTAL SUN RUA PEDRO GANGA, Nº 47	Greekfueld	SELF-SUPPORT.	32 40	NOT INFORMED	ACT INFORMED ACT INFORMED	N/D N/D	INCEPARE LTON NETWORKER TELECOM	OK .	OK OK	OK OK	WARENG LIGHTING NOT COMPLEXIT - COMMAND BOARD MEDIS MAINTINANCE	MOT	LICENSABLE	NOT
	632 7145534	30507160572_10998	305CFN60572	-27.43006	-48.47656	sc	Plotanopolis	RDD. TERTULIAND INITO SAVER, 2352	Realtap	PAST	4	NOT INFORMED	NOT INFORMED	ND	NOT INFORMED	NDK	OK.	NDK	NEEDS EXTENSED MARTORIAL SERVICES. NO REST PLATFORM, LEGHTADAG ROD NOT	MOT	NOT INFORMED	NOT INFORMED
Mile         Optimized         System         Lists         Lists         Mile         Mile         State         System	631 7143168 634 7148981	CMB ISSMETTING TORRES 1117 (SCOPRILLIT_TORRES	\$59923760 \$C0P#1117	-21.0421 -26.9324	-40.831 -48.8535	13 5C	Макузарні Сакран				25 (10+10+10+10+7+7)	1.1		80 80		OK NDK	OK NDK	NOK OK	NO GUARDRAL	MOT MOT		NOT INFORMED NOT INFORMED
	625 655458 629 657754	003 (CM8046306) 003 (CM8043760)	MOMPLAN33 MOPTEMAG	-18.6581 -17.8044	-41.3974 -41.7983	201 PRG	Mendes Panentel Pote	RUA DA TORRE, S/N AN. DOUTOR (BTULE) VARIAS, 321	Greenfand Greenfand	POST	33		NOT INFORMED	N0 N0	IEMITAL IND. COM. LTDA. NOT INFORMED	NOK	NOK	OK OK	NIEDS EXTENDED MARTORIAL SERVICES + KENNEL CLORES NOT	MOT	NOT DECEMBED	NOT INFORMED
	637 7145368	053L (SCCC00052_87)-SCCC00052	\$CCC00052	-27.0342	-52.62662	sc	Chapeco	AVENZA FERNANDO MACHEZO, 3839	Realtap	MAST	(3+3)		NOT INFORMED	ND	NOT INFORMED	NOK	0K	NOK	WARNING LIGHTING DOES NOT WORK	COCEANED MASTE	NOT DECRMID	NOT INFORMED

				Address	
#	Site ID	State	Municipality	Address	Lease Agreements Identified
3	ALMCO2363	AL	Maceió	AV ASSIS CHATEAUBRIAND, 2998	YES
4	ALOUB2295	AL	Ouro Branco	RUA CORONEL LUCENA, S/N LUIZ G. DE CARVALHO	YES
6	AMMNS3146	AM	Manaus	Av. J, s/n - Rua Sete de Abril (Pró-menor Dom Bosco)	YES
8	APSQA2524	AP	Santana	AV. FRANCISCO DE MELO PARENTE, 147, REMEDIOS II	YES
12	S2BABRR0061	BA	Salvador	AVENIDA DOM JOAO VI, 466	YES
14	S2BAASG0093	BA	Salvador	RUA DA MOURARIA, 6	YES
15	S2BAPOK0008	BA	Salvador	RUA PROFESSOR ARISTIDES NOVIS, 12, FEDERAÇÃO	YES
16	S2BAIGW0058	BA	Salvador	RUA RUBENS GUELLI, 134	YES
22	BACXS4693	BA	Mata De São Joao	KM 76 - COMPLEXO COSTA DO SAUIPE	YES
23	BAPGU1174	BA	Porto Seguro	RUA SÃO PEDRO, 180	YES
24	BAUIA1294	BA	Ubaitaba	RUA DO CACAU S/N	YES
29	BAMJO0786	BA	Mata De São Joao	Rodovia BA099	YES
30	BAICM1146	BA	Ilha De Comandatuba	Rua de Comandatuba, s/n	YES
31	BABRM1286	BA	Brumado	Rod. BR 30	YES
32	S2CEFLA0232	CE	Fortaleza	Rua Vicente Leite, 630 - Meireles.	YES
33	S2CEFLA0114	CE	Fortaleza	Edifício San Lucas - Rua Prof. Silo Ribeiro, 420 Papicu	YES
34	S2CEFLA0198	CE	Fortaleza	RUA PINTO MADEIRA,757	YES
35	S2CEFLA0202	CE	Fortaleza	AVENIDA ANTONIO SALES, 1317,	YES
37	CEFLA0191	CE	Fortaleza	Avenida Washington Soares, n° 55	YES
38	S2CEFLA0188	CE	Fortaleza	AVENIDA LUIS, 300	YES
39	S2CEFLA0074	CE	Fortaleza	Rua Senador Pompeu, 649	YES
40	CEFLA0230	CE	Fortaleza	RUA VILEBALDO AGUIAR, ED. JARANA 2200	YES

41	CEFLA0187	CE	Fortaleza	Av. Vicente de Castro, 6043	YES
42	CEARJ0547	CE	Barbalha	DISTRITO SITIO FARIAS, 0	YES
47	DFBSA0236	DF	Brasília	SCES trecho 2 lote 32	YES
48	DFSAM0754	DF	Brasília	QR 411CONJ. 4, CASA 14, SAMAMBÁIA	YES
49	DFSAM0753	DF	Brasília	QS604, CONJ. B (CANDIDADTO B)	YES
50	DFBSA0225	DF	Brasília	Saan Quadra 3 Lote 870 - N S/N	YES
52	DFBSA0741	DF	Brasília	SET SAFS QD 4 LT 1 ZONA CIVICO ADMINIST	YES
54	DFBSA0189	DF	Brasília	SQS 307 BLOCO J - N° S/N	YES
55	DFROD0128	DF	Brasília	Rodovia Brasília - Unaí, BR251, Km 14 - n° S/N	YES
57	DFBSA0312	DF	Brasília	SHIS QI 11, BLOCO O. LAGO SUL	YES
59	ESCNA3115	ES	Colatina	AVENIDA GETULIO VARGAS, 500	YES
61	ESCIM3808	ES	Cachoeiro De Itapemirim	RUA JOSE FARIAS DE JESUS, 8	YES
62	ESVTA3002	ES	Vitoria	AV. NOSSA SENHORA DA PENHA,1495	YES
64	GOIUB0360	GO	Itumbiara	Setor Central - Praça da República. CEP. 75503-900	YES
65	GOGNA0425	GO	Goiânia	ARAGUAIA SHOPPING - RUA 44	YES
66	DFROD0119	GO	Cristalina	Rodovia Br 040, Brasília-Belo Horizonte, Km: 35 - N S/N	YES
67	GOGNA0393	GO	Goiânia	AV. 136. QUADRA 239. LOTE 9/10.	YES
68	GOACG0278	GO	Aparecida De Goiânia	RUA CAPIBARIBE Q 22, 0	YES
69	GOGNA0492	GO	Goiânia	RUA 04 ESQ. C RUA 02-A, QD.5A, LT.01, N°1.502. SETOR CENTRAL	YES
72	GOGNA0088	GO	Goiânia	Rua 17A Esquina Com 13A, 899 — ST. Aeroporto - GO	YES
73	GOGNA0462	GO	Goiânia	RUA BENEDITO ZUPELLI, 254-320	YES
74	S2MASLS1543	MA	São Luís	Rua Grande, n°490	YES
75	MASLS1557	MA	São Luís	Av. Senador Vitorino Freire, nº 1958 — Bairro Areinha	YES
76	S2MASLS1512	MA	São Luís	AVENIDA CASTELO BRANCO, 500	YES
77	MASLS1520	MA	São Luís	Rua do Egito, 283 - Centro	YES

79	MASMH1979	MA	São Mateus Do Maranhão	RUA BENU LAGO	YES
81	MGURA4872	MG	Uberaba	Rod. BR-050, Km 195	YES
82	MGASL5930	MG	Augusto De Lima	R Geraldo Lopes Alvim, 314	YES
85	3GMGGVS5325	MG	Governador Valadares	Av. JK, 1157	YES
86	MGJFA5020	MG	Juiz De Fora	Av. Barão do Rio Branco - 2288	YES
87	MGIJA5865	MG	Itajubá	Rd. Itajubá - Maria da Fé, Km 3	YES
88	MGSAV4348	MG	Belo Horizonte	Av. João Pinheiro, 495	YES
90	MGJFA5022	MG	Juiz De Fora	AVENIDA DO RIO BRANCO, 1441	YES
91	MGSLZ2004	MG	Belo Horizonte	Avenida Antônio Carlos, 8100 - Pampulha Mall	YES
92	MGBEL2018	MG	Belo Horizonte	Rua Afonso Costa Reis, 65	YES
93	MGCTO4130	MG	Belo Horizonte	Rua Espirito Santo - 1059	YES
94	MGCTO4308	MG	Belo Horizonte	RUA CARIJOS, 424 - SALA 08 (PRAÇA SETE DE SETEMBRO)	YES
96	MGSTO4302	MG	Belo Horizonte	Rua Carangola, 670 com Rua Cristina	YES
97	MGEPL4323	MG	Belo Horizonte	RUA BEGONIA, 72	YES
98	MGFUN4153	MG	Belo Horizonte	Av. Brasil, 709	YES
99	MGJFA5046	MG	Juiz De Fora	RUA MARECHAL DEODORO, 444	YES
100	MGURA5544	MG	Uberaba	Praça Rui Barbosa, 300, (ELVIRA SHOPPING)	YES
102	MGJFA5116	MG	Juiz De Fora	AVENIDA BARÃO DO RIO BRANCO	YES
105	MGBEL4055	MG	Belo Horizonte	Rodovia Br-356, 3049 - Bh Shopping	YES
106	MGFLO4125	MG	Belo Horizonte	Rua Jacuí, 1882	YES
107	MGIRO4867	MG	Itabirito	ESTRADA PRINCIPAL, S/N	YES
108	MGMRN4603	MG	Mariana	RUA JOSE MESQUITA, 167	YES
111	MGBET6472	MG	Betim	RUA ROMUALDA AUGUSTA DE MELO, 337	YES
116	MGSOE5014	MG	São Joao Del Rei	RUA GABRIEL PASSOS, 321	YES
118	MGGON5120	MG	Goiana	RUA JOAO DE SOUZA RIBEIRO, 0	YES

120	MCEVC2211	MC	Fronteira Dos		VEC
120	MGFVS2311	MG	Vales	RUA BAHIA, 198	YES
122	MGURA5723	MG	Uberaba	AVENIDA FILOMENA CARTAFINA, KM 17	YES
124	MGCAA6386	MG	Capitão Andrade	AVENIDA LEVINDO DIAS, S/N	YES
125	MGMEF6434	MG	Monte Formoso	MORRO DO CEMITERIO	YES
130	MGNLA2030	MG	Nova Lima	RODOVIA JUSCELINO KUBITSCHEK / BR -040, KM 545, S/N	YES
136	MGINH5259	MG	Itanhomi	AV. SÃO CRISTOVÃO, S/N	YES
141	MTCBA0021	MT	Cuiabá	AV HIST RUBENS DE MENDONCA - BAU	YES
142	MTCBA0164	MT	Cuiabá	AV. BEIRA RIO, 3100 JARDIM EUROPA	YES
144	PABLM2019	PA	Belém	Rua Santo Antônio	YES
153	PAPGN2432	PA	Paragominas	Avenida Júlia Kubitschek, s/n Quadra 19 Lote 06	YES
160	PAIAB2438	PA	Itaituba	Rua Santo Antonio - nº 1489	YES
161	PAVCB2387	PA	Barcarena	RODOVIA PA, KM21 - DISTRITO MURUCUPI (FABRICA ALUNORTE)	YES
162	PAPUP2026	PA	Parauapebas	MINA DE FERROSOS CARAJÁS N4 (PCE)	YES
164	PBMAN4510	PB	Cabedelo	AVENIDA FLAVIO RIBEIRO COUTINHO, N°143, MANAIRA	YES
165	PBCBO4516	PB	Joao Pessoa	RUA JOSE RAMALHO LEITE, 1233	YES
166	PBMAN4559	PB	Joao Pessoa	AVENIDA SAPE, 1800	YES
167	PBCTO4571	PB	Joao Pessoa	AVENIDA BARROSO, 433	YES
172	PEESP3020	PE	Recife	AV GOV. AGAMENON MAGALHÃES 2656- ESPINHEIRO	YES
173	PERCE3244	PE	Recife	RUA CAIS DA ALFÂNDEGA, S/N - SHOPPING ALFÂNDEGA	YES
175	PEBVG3037	PE	Recife	RUA RAUL AZEDO, 53	YES
176	PEBVG3036	PE	Recife	RUA JOSE ADERVAL CHAVES, 264	YES
177	PEBVG3044	PE	Recife	Boa Viagem Avenida Boa Viagem, 4000 Edf Vânia.	YES
178	PEBVG3087	PE	Recife	RUA ANISIO RODRIGUES COELHO, 423	YES

179	PEBVT3006	PE	Recife	RUA SETE DE SETEMBRO, 238	YES
180	PEGRX3016	PE	Recife	AV GOV. AGAMENON MAGALHÃES 2656- ESPINHEIRO	YES
181	PESTA3002	PE	Recife	RUA MARQUES DE HERVAL,150 - SANTO ANTONIO - RECIFE - PE	YES
182	PEIPJ4151	PE	Ipojuca	Rua Frei Vicente Salvador, 203	YES
184	PEPOX3225	PE	Recife	Estrada das Ubaias 634	YES
185	PEESP3039	PE	Recife	RUA QUARENTA E OITO Nº434 - EDF. NICOLAU II	YES
186	PEIDL3038	PE	Recife	RUA CAP. JOSÉ DA LUZ,25	YES
187	PEOLD3424	PE	Olinda	Avenida José Augusto Moreira 2258 - Edf Cidade de Olinda	YES
188	PEPNM3128	PE	Recife	Avenida 17 de Agosto 500	YES
189	PEMAD3054	PE	Recife	Rua Real da Torre, 292 - Madalena, Recife - PE	YES
190	PEMAD3011	PE	Recife	Rua Demócrito de Souza Filho 71 - EDF SOLAR DO SICUPEMA	YES
191	PESTA3001	PE	Recife	RUA IMP D PEDRO II, 325	YES
192	PESTA3022	PE	Recife	Avenida Guararapes 154	YES
194	PEBVG3058	PE	Recife	Rua Major Armando de Souza Melo, 430 - Edf Jacunã	YES
195	PEAJP3079	PE	Recife	Rua Oscár Pinto, 293 - Casa Amarela, Recife - PE	YES
196	PEDIM3236	PE	Recife	Rua Manoel de Medeiros	YES
197	PEBVT3032	PE	Recife	AVENIDA CONDE DA BOA VISTA, 331 - CENTRO	YES
199	PECQL3149	PE	Recife	Rua Alto de Bela Vista, 0	YES
200	PETMR3031	PE	Recife	R. Jundiá, 242 - Edf Sisal	YES
201	PEBVG3082	PE	Recife	RUA SGT SILVINO MACEDO, 155	YES
203	PEBVG3222	PE	Recife	Rua Fernando Simões Barbosa 316	YES
204	PEJBO3310	PE	Jaboatão Dos Guararapes	Avenida Bernardo Vieira de Melo, 4804	YES
205	PEJBO3302	PE	Jaboatão Dos Guararapes	Avenida Beira Mar, 520	YES

				RUA UBALDINO DO AMARAL, 37, ESTÁDIO	
215	PRCTA0667	PR	Curitiba	COUTO PEREIRA	YES
216	PRCTA0546	PR	Curitiba	AVENIDA PARANÁ 1758	YES
217	PRCTA0661	PR	Curitiba	RUA TREZE DE MAIO, 981	YES
218	PRMGA0556	PR	Maringá	AV. DONA SOPHIA RASGULAEFF,466	YES
219	PRCTA0678	PR	Curitiba	RUA COMENDADOR ARAUJO, 730, HOTEL IBIS	YES
221	PRCPM0685	PR	Campo Mourão	Rua Laurindo Borges, 1398 - Centro, Campo Mourão - PR	YES
222	PRFOZ0722	PR	Foz Do Iguaçu	Rua Rui Barbosa, 786	YES
224	PRCTA0651	PR	Curitiba	AVENIDA PARANA 3825	YES
225	PRLDA0424	PR	Londrina	Av Higienópolis, 70 - Jardim Higienópolis	YES
226	PRCTA0690	PR	Curitiba	Rua Mariano Torres, 916	YES
229	PRFOZ0721	PR	Foz Do Iguaçu	Rua Carlos Souttomayor, 393	YES
230	PRCTA0545	PR	Curitiba	RUA JOQUEI CLUBE, 437	YES
231	PRCTA0053	PR	Curitiba	Rua Belém, 43	YES
232	PRCTA0734	PR	Curitiba	Rua Visc. do Rio Branco, 803	YES
234	PRCPP3519	PR	Cornélio Procópio	AV. NOSSA SENHORA DO ROCIO,966	YES
235	PRCTA0658	PR	Curitiba	AVENIDA SILVA JARDIM, 1856	YES
236	PRCTA0523	PR	Curitiba	RUA PADRE ANCHIETA, 2128	YES
237	PRCTA0110	PR	Curitiba	Rod. Br 277 Cta - Orleans / Rod. do Café - Cidade	YES
237	PRCTAULIO	ΓK	Cultuba	Industrial De Curitiba	165
238	PRCTA0143	PR	Curitiba	Rua da Paz, 202	YES
239	PRCTA0422	PR	Curitiba	Alameda Dr. Muricy, 650	YES
240	PRCTA0423	PR	Curitiba	Rua Dr. Manoel Linhares de Lacerda, 84 - Capão Raso	YES
242	PRCTA0151	PR	Curitiba	Rua XV de Novembro, 360	YES
243	PRLDA0214	PR	Londrina	Rua Sem. Souza Naves, 45	YES
244	PRCTA0126	PR	Curitiba	Rua Sete de Abril, 571	YES
245	PRCTA0142	PR	Curitiba	Av. João Gualberto, 623	YES
248	PRPGO0467	PR	Ponta Grossa	Rua Varnhagem, 366	YES
249	PRCTA0103	PR	Curitiba	Rua Marechal Deodoro, 950	YES

250	PRLDA0219	PR	Londrina	R. Astorga, 80- Hedy	YES
253	PRCTA0469	PR	Curitiba	Av Afonso Camargo, 849 - Cristo Rei - Curitiba - PR	YES
254	PRMGA0497	PR	Maringá	Rua Néo Alves Martins, 3176 - Zona 3	YES
258	PRFOZ0428	PR	Foz Do Iguaçu	AVENIDA FELIPE WANDSCHEER 1276	YES
260	PRCTA0454	PR	Curitiba	Av. Nossa Senhora da Luz, 1800 - Jardim Social - Curitiba - PR	YES
261	PRCTA0463	PR	Curitiba	Avenida da República, 6464 - Guaíra - Curitiba - PR	YES
262	PRLDA0505	PR	Londrina	AVENIDA SANTOS DUMONT, 500-BOA VISTA	YES
277	RJRCP0091	RJ	Rio De Janeiro	AVENIDA PAULO DE FRONTIN, 739	YES
284	RJMGE0314	RJ	Magé	RUA RAFAEL COZOLINO, QUADRA L, LOTE 3.	YES
286	RJCTO0960	RJ	Rio De Janeiro	Rua São Jose, 90	YES
288	RJBTJ0411	RJ	Rio De Janeiro	Av. das Américas, 3120	YES
295	RJLEB0130	RJ	Rio De Janeiro	Rua Dias Ferreira, N° 658	YES
298	RJTNQ2900	RJ	Rio De Janeiro	AVENIDA GEREMARIO DANTAS, 402	YES
303	RJGAL0141	RJ	Rio De Janeiro	AVENIDA VINTE DE JANEIRO, S/N° - ILHA DO GOVERNADOR - RIO DE JANEIRO	YES
315	3GRJVAS3332	RJ	Vassouras	ESTRADA BARÃO DE IPIABAS, S/N° - VASSOURAS	YES
325	RNMLU6004	RN	Natal	Rua Florestal 27	YES
330	RNLNA6017	RN	Natal	Rua N. Sra. de Lourdes 1376 - Ed. Manhatam	YES
332	ROPVO3036	RO	Porto Velho	AV. GOV. JORGE TEIXEIRA, 629	YES
334	S2RSPAE0651	RS	Porto Alegre	RUA PROF. JOÃO DE SOUSA RIBEIRO, 859	YES
337	S2RSPAE0970	RS	Porto Alegre	ACESSO D SQ 3 1 UNIDADE VICINAL, 3146	YES
339	RSCSL0061	RS	Caxias Do Sul	Rua Sinimbu, 2682	YES
340	RSCSL0052	RS	Caxias Do Sul	Av Júlio de Castilhos, 4054	YES
341	RSCSL0054	RS	Caxias Do Sul	Rua Gen. Arcy da Rocha Nobrega, 504- Jardim América	YES
344	RSLHC0768	RS	Salvador Do Sul	LINHA COMPRIDA - CENTRO - SALVADOR DO SUL	YES
345	RSRGR0891	RS	Rio Grande	RUA REVOCATA DE MELLO, 51	YES
346	RSCSL0068	RS	Caxias Do Sul	Rua Remio Antônio Tomazzoni, 138 — Planalto — Caxias do Sul	YES

348	RSCAN0870	RS	São Gabriel	RUA FERNANDO FERRARI, S/N	YES
349	RSPAE0971	RS	Porto Alegre	ALAMEDA J - UNIDADE VICINAL, 183	YES
351	RSPAE0857	RS	Porto Alegre	AVENIDA CRUZEIRO DO SUL, 2323	YES
362	SCPRI0545	SC	Princesa	Rua Santa Catarina, s/n°	YES
364	3GSCFNS0405	SC	Florianópolis	Av. Rio Branco, 369	YES
365	SCWUM0274	SC	Witmarsum	Estrada Geral, s/n°	YES
366	SCFNS0333	SC	Florianópolis	R. do Calafate, 113 — BL2 - PANTANAL	YES
369	SCJGS0311	SC	Jaraguá Do Sul	Rua Gerhardt, s/n — Rio da Luz	YES
371	SCSOO0503	SC	São Jose	RUA AMBRÓSIO DA SILVA, 223	YES
373	3GSCFNS0382	SC	Florianópolis	RUA LACERDA COUTINHO,179	YES
374	SCBCU0351	SC	Balneário Camboriú	AV. BRASIL, 3080	YES
377	3GSCFNS0572	SC	Florianópolis	ROD. TERTULIANO BRITO XAVIER, 2352	YES
381	SPCAO4938	SP	Cubatão	RUA JOSE GONÇALVES TORRES S.N - CUBATÃO/SP	YES
382	SPWCV3688	SP	São Paulo	RUA PRO F VILMA VIEIRA, 21 - CASA VERDE	YES
383	SPWMM3911	SP	São Paulo	Av Moema, 81 - Planalto Paulista, São Paulo - SP	YES
384	SPWPC3942	SP	São Paulo	RUA MANOEL DA MATA SÁ 170	YES
385	SPWFO3735	SP	São Paulo	RUA ISANGA, Nº 521 - FREGUESIA DO Ó	YES
386	SPWCM3694	SP	São Paulo	Rua Pedro Rodrigues Beja, 111 — Vila Marari	YES
387	SPWVF4163	SP	São Paulo	RUA ANTONIO REIS N 11B — CHÁCARA BELENZINHO	YES
388	SPWPT3996	SP	São Paulo	Rua Dr João Toniolo 535 — Jardim São José	YES
389	SPWCM3695	SP	São Paulo	RUA FERRUCIO SANDOLI 232 — SÃO PAULO	YES
390	SPWST3117	SP	São Paulo	Av. Roque Petroni Júnior, 1089 - Chácara Santo Antônio (Zona Sul), São Paulo - SP	YES
392	SPWVD3048	SP	São Paulo	Rua Oscar de Medeiros, 164, Vila Medeiros	YES
393	SPWGR3743	SP	São Paulo	Rua DR. Antônio Simões de Carvalho,105 - Jd. São Bernardo -São Paulo - SP	YES

395	SPFAC4958	SP	Franca	Av Presidente Vargas 666 — Cidade Nova, Franca - SP	YES
396	SPWCS3728	SP	São Paulo	Rua Girolamo Dai Libri, 433 - Vila Moraes	YES
397	SPWAA3576	SP	São Paulo	Rua da Padroeira, 83 - Jardim Nordeste, São Paulo - SP	YES
398	SPOCO4536	SP	Osasco	Rua Rosa D'ângelo Pisapia, 156 — Vila Yara	YES
399	SPSBO4598	SP	São Bernardo Do Campo	Rua Ruy Barbosa, 260	YES
400	SPWBE3603	SP	São Paulo	RUA JOAO BATISTA DE LACERDA, 261	YES
401	SPSBO4607	SP	São Bernardo Do Campo	Rua Municipal, 193	YES
402	SPWBF3583	SP	São Paulo	Av. Francisco Matarazzo, 229 - Água Branca, São Paulo - SP	YES
403	SPWBF3585	SP	São Paulo	Rua Josef Kryss, 319 - Parque Industrial Tomas Edson	YES
404	SPSBO4630	SP	São Bernardo Do Campo	Rua Nelza Ascêncio, 128 — Ferrazópolis — São Bernardo do Campo	YES
405	SPSCN4637	SP	São Caetano Do Sul	Rua Archinto Ferrari, 118 — Santo Antônio — São Caetano do Sul	YES
406	SPWBV3589	SP	São Paulo	AVENIDA PAULISTA, 810	YES
407	SPWCB3641	SP	São Paulo	Rua Ribeiroles, 274 — Vila Paulista	YES
408	SPMMM5169	SP	Mogi Mirim	Rua Coronel Venâncio Ferreira Alves Adorno, 89 — Saúde	YES
409	SPRPO5341	SP	Ribeirão Preto	Rua Bernardino de Campos, 1236	YES
410	SPSTB5363	SP	Santa Barbara Doeste	RUA TREZE DE MAIO, 749 - CENTRO	YES
411	SPSJC5460	SP	São Jose Dos Campos	Rua Euclídes Miragaia, 26 — Centro	YES
412	SPPGE5272	SP	Praia Grande	AV. PRESIDENTE COSTA E SILVA, 13 — PRAIA GRANDE - SP	YES
414	SPSTS5379	SP	Santos	Rua Saturnino de Brito, 256 — Marapé — Santos	YES
415	SPSTS5396	SP	Santos	Av. Ana Costa, 121 - Vila Matias	YES
416	SPSTS5403	SP	Santos	Rua Vergueiro Steidel, 304 - Aparecida	YES

417	SPSUM5601	SP	Sumaré	Rua João Antônio Soares, 18 — Jd. Bom Retiro - Sumaré	YES
418	SPWPN3988	SP	São Paulo	Avenida Nove de Julho, 5624 — Jardim Paulista — São Paulo	YES
419	SPWTA4138	SP	São Paulo	Rua Cantagalo, 738 — Vila Gomes Cardim - SP	YES
420	SPWPZ3963	SP	São Paulo	Rua Dr. Homem de Melo, 1186 - Perdizes	YES
421	SPWPZ3964	SP	São Paulo	Rua Dr. Homem de Melo, 352 — Perdizes — São Paulo	YES
422	SPWJQ3800	SP	São Paulo	Rua das Oiticicas, 149 — Pq. Jabaquara - São Paulo - SP	YES
423	SPWJQ3801	SP	São Paulo	Rua Mianos, 104 — Vila Santa Catarina - SP	YES
425	SPWRE4007	SP	São Paulo	Alameda Barão de Limeira, 478 - Campos Elíseos, São Paulo - SP	YES
426	SPWRE4010	SP	São Paulo	Avenida Cásper Libero, 58 — Centro Histórico de São Paulo	YES
427	SPWRE4011	SP	São Paulo	Rua Conselheiro Ramalho, 104 — Bela Vista	YES
428	SPWVJ4178	SP	São Paulo	Rua Édipo Felíciano, 100 - Vila Jacuí, São Paulo — SP	YES
429	SPWLB3880	SP	São Paulo	Rua do Glicério, 731 — Liberdade — São Paulo	YES
430	SPWLB3881	SP	São Paulo	Rua Vergueiro, 168 — Liberdade — São Paulo	YES
433	SPWVM4212	SP	São Paulo	Rua Eça de Queiroz, 258 — Vila Mariana - SP	YES
434	SPWLP3868	SP	São Paulo	Rua Tonelero, 239 — Vila Ipojuca — São Paulo - SP	YES
435	SPWLP3870	SP	São Paulo	Rua Marcelina, 394 — Vila Romana - SP	YES
436	SPWSC4040	SP	São Paulo	Rua Albuquerque Lins, 374 - Santa Cecilia, São Paulo - SP	YES
437	SPWSD4078	SP	São Paulo	Av General Charles de Gaulle, 208 — Pq. São Domingos	YES
438	SPWVM4216	SP	São Paulo	R. Joel Jorge de Melo, 62 - Vila Mariana	YES
440	SPWSE4124	SP	São Paulo	Praça Doutor João Mendes, 62 — Liberdade	YES
441	SPWMM3898	SP	São Paulo	RUA SALTO, 70 — MOEMA	YES
442	SPWMM3901	SP	São Paulo	Av. dos Imarés, 88 - Indianópolis, São Paulo - SP	YES
443	SPWMM3903	SP	São Paulo	Alameda Jauaperi, 1083 — Moema - SP	YES
446	SPWMR3927	SP	São Paulo	Avenida Barão de Monte Mor, 75 — Vila Morumbi	YES
447	SPWSN4057	SP	São Paulo	Rua Voluntários da Pátria, 2798 - Santana, São Paulo - SP	YES
449	SPWSU4103	SP	São Paulo	Rua Estero Belaco 160 - Vila da Saúde, São Paulo - SP	YES

450	SPWSU4117	SP	São Paulo	Av. Jabaquara, 970 — Saúde — São Paulo	YES
452	SPWCO3064	SP	São Paulo	Av. Angélica, 1683 - Santa Cecilia, São Paulo - SP	YES
453	SPWSU3522	SP	São Paulo	Avenida Indianópolis, 1192 — (Clube Sírio Libanês) - Planalto Paulista, São Paulo	YES
454	SPWBL3618	SP	São Paulo	RUA EDUARDO COSTA, N°244 - VILA BRASILÂNDIA	YES
457	SPWCO3717	SP	São Paulo	Rua Minas Gerais, 388 - Higienópolis	YES
458	SPSBO4605	SP	São Bernardo Do Campo	Rua Marechal Badoglio, 15 — Vila Mussoline	YES
459	SPWJR3818	SP	São Paulo	Rua Conselheiro Amaral, 196 - Jaguará, São Paulo - SP	YES
460	SPWPN3987	SP	São Paulo	Rua Hungria, 664 - Jardins, São Paulo - SP	YES
461	SPWTA4136	SP	São Paulo	RUA GEN. GELÁS, 72 - TATUAPÉ, SÃO PAULO-SP - 03090-030	YES
462	SPWRE4012	SP	São Paulo	Rua Washington Luís, 392 — Centro Histórico — São Paulo	YES
463	SPWSC4044	SP	São Paulo	Praça Júlio Prestes, 30 - Campos Elíseos, São Paulo - SP	YES
464	SPWMM3907	SP	São Paulo	Alameda dos Tupiniquins, 1235 - Planalto Paulista	YES
465	SPWPN3972	SP	São Paulo	Rua Fradique Coutinho, 1590 — Pinheiros — São Paulo - SP	YES
466	SPWMM2564	SP	São Paulo	Rua inhambu, 763	YES
467	SPWVM3099	SP	São Paulo	Av. Brigadeiro Luís Antônio, 2819 — Jardim Paulista	YES
468	SPWLB3069	SP	São Paulo	Rua Topázio, 420 — Vila Mariana	YES
469	SPLRA5117	SP	Limeira	Rua Dr. José Carvalho Ferreira, s/n	YES
470	SPWJE3059	SP	São Paulo	Rua Caetanópolis, 930 - Jaguaré	YES
471	SPWCO3711	SP	São Paulo	Rua da Consolação, 2143	YES
473	SPWCO3721	SP	São Paulo	Av Angélica, N° 1189 — Consolação	YES
474	SPJAI4454	SP	Jundiaí	Rua do Rosário	YES
475	SPSTS5374	SP	Santos	Rua Augusto Paulino, 57	YES
476	SPWBV3596	SP	São Paulo	RUA LOURENCO GRANATO, 214	YES

477	SPWCY3637	SP	São Paulo	AVENIDA LACERDA FRANCO, 130, CEP:1536-000	YES
478	SPWAR3557	SP	São Paulo	RUA TEOFILO DIAS, 332	YES
479	SPWBS3612	SP	São Paulo	RUA JOSE MONTEIRO, 323	YES
480	SPWIT3794	SP	São Paulo	AVENIDA CAMPANELLA, EDIFÍCIO: CAMPANELLA 1297 — ITAQUERA	YES
481	SPSTS5387	SP	Santos	Rua Ana Costa, 555	YES
482	SPSTS5389	SP	Santos	Rua Barão de Paranapiacaba, 77	YES
483	SPSTS5397	SP	Santos	Rua Euclides da Cunha, 151	YES
487	SPWJP3833	SP	São Paulo	Rua José Maria Lisboa, 1187	YES
488	SPWJP3845	SP	São Paulo	Rua Pamplona, 825 - Jardim Paulista	YES
489	SPWPZ3957	SP	São Paulo	Rua Diana, 740 — Perdizes	YES
490	SPWJQ3807	SP	São Paulo	Rua das Grumixamas, 110 — Jardim Oriental	YES
491	SPWRT4003	SP	São Paulo	Rua Cândido Fontoura, 481	YES
492	SPWSC4035	SP	São Paulo	Rua Guaianases, 1197	YES
493	SPWMR3923	SP	São Paulo	Av. Lineu de Paula Machado, 1174 Jardim Everest	YES
494	SPWIR3748	SP	São Paulo	Rua Vasconcelos Drumond, 196, Ipiranga	YES
495	SPWIR3751	SP	São Paulo	Rua Costa Aguiar, 994 - Ipiranga	YES
496	SPWSU4108	SP	São Paulo	Av. Indianópolis, 2185	YES
497	SPARC3152	SP	Araçatuba	Rua Wenceslau Braz 120 - Ed Copacabana — Vila Santa Maria	YES
498	SPSTS3213	SP	Santos	Avenida Presidente Wilson, 66 — José Menino	YES
499	SPSZN3100	SP	Suzano	Rua Benjamin Constant, 1653	YES
500	SPWMM3096	SP	São Paulo	Alameda dos Anapurus, 1297 - Moema	YES
501	SPWJP3067	SP	São Paulo	Alameda Tietê, 301 — Jardim Paulista	YES
502	SPWJG3820	SP	São Paulo	Rua Sebastiao Laranjeiras 252	YES
503	SPWLJ3865	SP	São Paulo	RUA GIBOUCUS 62	YES
504	SPWCO3714	SP	São Paulo	Avenida Angélica, 2000 — Consolação — São Paulo	YES
505	SPWPN3974	SP	São Paulo	Rua Joaquim Antunes, 149 - Pinheiros — São Paulo	YES
506	SPWVM4205	SP	São Paulo	Rua Conselheiro Rodrigues Alves, 391 — Vila Mariana	YES

507	SPWLM3888	SP	São Paulo	Rua Antônio Ribeiro de Morais, 264 - Vila Carbone, São Paulo - SP	YES
508	SPWCO3719	SP	São Paulo	Rua Amaral Gurgel, 518 — Consolação	YES
509	SPWRE4004	SP	São Paulo	Rua Coronel Xavier de Toledo, 70 - República	YES
510	SPWRE4005	SP	São Paulo	Av. Prestes Maia, 702	YES
511	SPWIB3779	SP	São Paulo	Rua Jesuíno Arruda, 327 - Itaim Bibi	YES
512	SPWIR3753	SP	São Paulo	Rua Costa Aguiar, 2446 - Ipiranga	YES
513	SPWPZ3047	SP	São Paulo	Rua Campevas, n° 313/325 - Perdizes	YES
514	SPGJA4743	SP	Guarujá	Rua Mário Ribeiro, 886	YES
515	SPGJA5009	SP	Guarujá	Rua José Avelino de Oliveira, 260	YES
517	SPWJP3849	SP	São Paulo	Av. Paulista, 2006 — Bela Vista	YES
518	SPWJP3854	SP	São Paulo	Avenida Brasil, S/N - Jardim América (Praça Nossa Senhora do Brasil)	YES
519	SPWIR3760	SP	São Paulo	Rua Vemag 1, Vila Independência	YES
520	SPWJP3068	SP	São Paulo	Rua Batatais, 558	YES
521	SPPAA5239	SP	Piracicaba	Rua Praça da Catedral Dom Ernesto de Paula, 170 - Centro	YES
522	SPWCL3670	SP	São Paulo	Rua Caverá, 55 — Jardim Umarizal	YES
523	SPWAN3570	SP	São Paulo	AV. Felippo Sturba, S/N - Anhanguera	YES
524	SPCAS4848	SP	Campinas	Rua General Osório, 1233 — Centro	YES
525	SPCAS4871	SP	Campinas	Rua Júlio de Mesquita, 36	YES
526	SPCAS4876	SP	Campinas	Rua Vargem Grande do Sul, 506 - Esquina com Avenida Baden Powell	YES
527	SPITA4429	SP	Itatiba	Rua Quintino Bocaiúva, 343	YES
528	SPSOC5568	SP	Sorocaba	Rua Capitão Nascimento Filho, 171	YES
529	SPWMC3921	SP	São Paulo	Rua João Caetano, 379	YES
530	SPWCN3674	SP	São Paulo	R. Maria Angélica Franci, - Vila Franci	YES
531	SPWIP3017	SP	São Paulo	R Manuel Martins de Melo, S/N- Vila Itaim	YES
532	SPWIP3018	SP	São Paulo	Avenida Tomé Dias Laço, S/N — Itaim Paulista	YES
533	SPWJH3830	SP	São Paulo	Rua Tetiximira, 146 — Jardim Helena	YES

534	SPGRS3000	SP	Guarulhos	Rua Candelária, S/N - Jardim Jacy	YES
535	SPWMR2678	SP	São Paulo	Rua Jules Rimet, estádio Cícero P. de Toledo, portão04, Morumbi	YES
536	SPWBS5671	SP	São Paulo	RUA DO GASOMETRO, 741, CEP: 3004 - 000	YES
537	SPWBU3625	SP	São Paulo	RUA ARMANDO FAIRBANKS, 310	YES
538	SPWJP3841	SP	São Paulo	Avenida São Gabriel, 236 — Jardim Paulista	YES
539	SPWJP3851	SP	São Paulo	Alameda Campinas, 1600	YES
540	SPWSC4038	SP	São Paulo	Rua Albuquerque Lins, 92	YES
541	SPWIB3777	SP	São Paulo	Rua Jacurici, 184 — Itaim Bibi	YES
542	SPRPO5315	SP	Ribeirão Preto	Rua Ana Costa, 555	YES
543	SPPGE5273	SP	Praia Grande	Rua São Salvador, 50 - Guilhermina	YES
544	SPSTS5391	SP	Santos	Av. Presidente Wilson, 90 - José Menino	YES
546	SPWST4072	SP	São Paulo	Rua Rodrigues Pais, 193	YES
547	SPWPN3985	SP	São Paulo	Rua Diana, 740	YES
548	SPCAS4870	SP	Campinas	Rua José de Alencar, 675 - Centro	YES
549	SPWJQ3808	SP	São Paulo	Rua Porcelana, 193 — Vila do Encontro — São Paulo	YES
550	SPWIB3769	SP	São Paulo	AVENIDA HORÁCIO LAFER, 702	YES
551	SPWLP3869	SP	São Paulo	Rua Albion, 229 - Lapa, São Paulo - SP	YES
552	SPWBV3588	SP	São Paulo	ALAMEDA RIO CLARO, BELA VISTA - 157 SÃO -SP	YES
553	SPWPS3970	SP	São Paulo	R Antônio José Anacleto, S/N, Vila Fanton	YES
554	SPWMM3098	SP	São Paulo	Av. Jacutinga, 632 - Indianópolis	YES
555	SPWBU3628	SP	São Paulo	AVENIDA CAXINGUI, 175	YES
556	SPPGE5275	SP	São Paulo	RUA BALNEARIO CAMBURIU, 100	YES
557	SPWJP3853	SP	São Paulo	AL ITU,1067 VILA CARAGUATA CONDOMÍNIO EDIFÍCIO JOSÉ HACHEM	YES
558	SPWIB3784	SP	São Paulo	Av. Engenheiro Carlos Berrini, 801	YES
559	SPWJP3840	SP	São Paulo	AVENIDA PAULISTA, EDIFÍCIO ASAHI,1274	YES
560	SPWSN4046	SP	São Paulo	RUA, FRANCISCO RAMOS N, 28 SANTANA	YES
561	SPWSE4118	SP	São Paulo	AV DO ESTADO, Nº 2854, CENTRO	YES

562	SPOCO4514	SP	Osasco	Rua Dr. Mariano Jatahy Marcondes Ferraz, 215 - Centro	YES
563	SPRPO5325	SP	Ribeirão Preto	Rua Sete de Setembro,590	YES
564	SPWIB3761	SP	São Paulo	R. GUARAIUVA, 599	YES
565	SPWSL4085	SP	São Paulo	Avenida Sapopemba, 5435 — Sapopemba — São Paulo	YES
566	SPWSN4059	SP	São Paulo	Rua Valdemar Martins, 116	YES
567	SPWRE4015	SP	São Paulo	RUA AMARO, 418,	YES
568	SPWBV3594	SP	São Paulo	RUA PEIXOTO GOMIDE, 581	YES
569	SPWCB3644	SP	São Paulo	Rua Baronesa da Bela Vista, 801 (Hotel Ibis) - Vila Congonhas	YES
570	SPWRE4006	SP	São Paulo	RUA BENTO FREITAS 186	YES
572	SPGRS0436	SP	Guarulhos	Rua Caraguatatuba, 57	YES
573	SPWCS3725	SP	São Paulo	RUA KITIZU UTIYAMA, 314, CURSINO	YES
574	SPWJE3824	SP	São Paulo	Rua Marselha, 94, Jaguaré	YES
575	SPWVC3023	SP	São Paulo	AV. FLAMINGO, 1361 - VILA NOVA CURUÇÁ	YES
576	SPCAS4840	SP	Campinas	RUA BARRETO LEME, 2101	YES
577	SPCAS4857	SP	Campinas	Rua Doutor Sales de Oliveira, 463 - Vila Industrial - Campinas - SP	YES
578	SPCAS4882	SP	Campinas	Rua Doutor Quirino, 757 - Centro, Campinas - SP	YES
579	SPWCL3663	SP	São Paulo	Rua Odemis, 520	YES
580	SPWBF3579	SP	São Paulo	Largo Péricles, 0, Barra Funda	YES
581	SPSBO4601	SP	São Bernardo Do Campo	PÇA SAMUEL SABATINI, 200 - CENTRO - SÃO BERNARDO CAMPO	YES
582	SPSBO4608	SP	São Bernardo Do Campo	Rua Braga, 266 - Vila Lusitania, São Bernardo do Campo - SP	YES
583	SPSBO4613	SP	São Bernardo Do Campo	AVENIDA JOÃO FIRMINO, 1229 - ASSUNÇÃO - SÃO BERNARDO DO CAMPO - SP	YES
584	SPSCN4634	SP	São Caetano Do Sul	Rua Rafael Correia Sampaio, 977 - Santa Paula - S.C.S	YES
585	SPWBV3587	SP	São Paulo	Rua Treze de Maio, 1566 - Bela Vista - São Paulo	YES

587	SPSTS5383	SP	Santos	Av. Dona Ana Costa, 376 - Gonzaga - Santos	YES
590	SPWTA4137	SP	São Paulo	Rua Apucarana, 1563 - Tatuapé - São Paulo	YES
591	SPWJP3834	SP	São Paulo	Alameda Joaquim Eugênio de Melo, 1452 - Jd. Paulista - São Paulo	YES
592	SPWJP3844	SP	São Paulo	Alameda Jaú, 1167 - Jardim Paulista - SP	YES
593	SPWJP3848	SP	São Paulo	Rua Maestro Elias Lobo, 1040 - Jardim Paulista - São Paulo	YES
594	SPWPZ3959	SP	São Paulo	Rua Monte Alegre, 470 - Perdizes - São Paulo	YES
595	SPWRE4009	SP	São Paulo	Lago do Arouche, 96 - República - SP	YES
596	SPWSC4043	SP	São Paulo	Rua Dr Gabriel Dos Santos, 759 - Santa Cecilia	YES
597	SPWVM4217	SP	São Paulo	RUA MANUEL DA NÓBREGA, 181	YES
598	SPWMM3905	SP	São Paulo	Rua Gaivota, 22 - Moema	YES
599	SPWMM3910	SP	São Paulo	Rua Gaivota, 665 - Indianópolis - São Paulo	YES
600	SPWIB3773	SP	São Paulo	Av. Jurubatuba, 246 - Morumbi	YES
601	SPWTA4133	SP	São Paulo	RUA EMILIO MALLET, 1893, VILA GOMES CARDIM	YES
602	SPSTS2545	SP	Santos	Avenida Washington Luiz, 61 - Vila Matias- Santos - SP	YES
603	SPSCN3086	SP	São Caetano Do Sul	Rua Alameda São Caetano, 1738 - Santa Maria - S.C.S	YES
604	SPWRE3072	SP	São Paulo	Rua Vitória, 395	YES
607	SPWVS4235	SP	São Paulo	RUA JOAO GUILHERME, 239	YES
608	SPWVA4157	SP	São Paulo	Rua Chico Gomes, S/N°	YES
609	SPWLM3887	SP	São Paulo	R. Dirce Rodrigues 33	YES
610	SPCAS4865	SP	Campinas	Av. Barão de Itapura, 1518 - Jardim Guanabara, Campinas - SP	YES
611	SPGJA5000	SP	Guarujá	Rua José da Silva Figueiredo, 525 Enseada - Guarujá	YES
612	SPINE5044	SP	Itanhaém	Rua João Mariano, 193 - Centro - Itanhaém	YES
613	SPWBL3617	SP	São Paulo	ESTRADA DO SABÃO, 40	YES
614	SPRPO5311	SP	Ribeirão Preto	RUA AMÉRICO BRASILIENSE, 413	YES
616	SPPGE5280	SP	Praia Grande	Rua Nicarágua, 49 - Praia Grande - São Paulo	YES

618	SPWSC4034	SP	São Paulo	Rua Cruzeiro, 732, Barra Funda	YES
619	SPWMM3906	SP	São Paulo	R Dra. Neyde Aparecida Solito, 460 - Vila Clementino - Moema - SP	YES
620	SPCAS2549	SP	Campinas	RUA BARÃO DE JAGUARA, 707 - CONCEIÇÃO	YES
621	SPSCL2559	SP	São Carlos	Rua Dona Maria Jacinta, 241, São Carlos - SP	YES
622	SPGRS3002	SP	Guarulhos	Av. São Paulo, 352 - Cidade Brasil - Guarulhos - SP	YES
623	SPWCT3015	SP	São Paulo	Rua Jaguapeba, 100	YES
624	<b>SPWPT3039</b>	SP	São Paulo	Rua Cel. Gonçalves da Siqueira 145, Pirituba	YES
625	SPWRE3073	SP	São Paulo	Rua Augusta, 66	YES
626	SPJCI1883	SP	Jacareí	R. Cap. João Jose de Macedo, 340	YES
627	SPWPZ3961	SP	São Paulo	Rua Coari, 208	YES
628	SPWSL4088	SP	São Paulo	Rua Joaquim Ribeiro da Costa, 386	YES
629	SPWBF3580	SP	São Paulo	Rua Turiassú, 1863, Perdizes	YES
630	SPSJC5491	SP	São Jose Dos Campos	Rua Cel João Cursino, 240/242 - Ed. Toulon	YES
631	SPSTS5402	SP	Santos	RUA LACERDA FRANCO, Condomínio Ed. Icaraíma - Bairro Aparecida 137, CEP:11025180	YES
634	SPWPI3940	SP	São Paulo	Av. Cruzeiro do Sul, 1100 - Canindé	YES
635	SPJCI1882	SP	Jacareí	Rua Gilberto Moreira, 296 - Vila Formosa, Jacareí - SP	YES
636	SPWCB3651	SP	São Paulo	Rua Barão do Triunfo, 276	YES
637	TOPMJ2109	ТО	Palmas	AVENIDA LO-5, LT 01 - PLANODIRETOR SUL	YES

2	SRMGBPR4203	MG	Belo Horizonte	Avenida Augusto de Lima, 1549 - Fórum Lafayete	YES
3	RJSMI2621	RJ	São Joao De Meriti	Rodovia Presidente Dutra, 4200 - Shopping Grande Rio	YES
5	SPWCB3642	SP	São Paulo	Av. Washington Luís, s/n - Aeroporto Internacional de Congonhas	YES
6	SRCEFLA0142	CE	Fortaleza	Av. Senador Carlos Jereissati, 3000	YES
7	SPGRS2853	SP	Guarulhos	Rod. Presidente Dutra, Km 230 - Internacional Shopping Guarulhos	YES

8	SPGRS4359 / SPGRS4360	SP	Guarulhos	Rod. Hélio Schmidt, s/n - Aeroporto Internacional de São Paulo (Cumbica)	YES
9	SRBASCT0204	BA	Salvador	PRACA GAGO COUTINHO, 282,	YES
10	SRRJCTO3007	RJ	Rio De Janeiro	Praça Senador Salgado Filho, S/N - Aeroporto Santos Dumont	YES
11	DFIND0132	DF	Brasília	CSB 2, Lotes 1/4, Alameda Shopping	YES
12	SPBRE4271	SP	Barueri	Al. Rio Negro, 1033 - Alphaville Shopping	YES
14	SPARC2841	SP	Araçatuba	Av. Joaquim Pompeu de Toledo, 601 - Araçatuba Shopping	YES
17	SPBRU2845	SP	Bauru	R. Henrique Savi, 15-55 - Bauru Shopping	YES
18	SRRSIND0519	RS	Porto Alegre	AVENIDA PRAIA DE BELAS, 1181	YES
19	SRMGBEL4193	MG	Belo Horizonte	Rodovia BR-356, 3049 - BH Shopping	YES
20	MGCEM4198	MG	Contagem	Avenida João César de Oliveira, 1275 - Big Shopping (32260-250)	YES
21	SPWST4067	SP	São Paulo	R. Borba Gato, 59 - Boa Vista Shopping	YES
22	RJBOT2604	RJ	Rio De Janeiro	Praia de Botafogo, 400 - Botafogo Praia Shopping	YES
23	DFIND0726	DF	Brasília	SETOR TERMINAL NORTE, CONJUNTO J, BOULEVARD SHOPPING	YES
24	SPWBF2628	SP	São Paulo	Rua Turiassú, 2100	YES
25	DFIND0130	DF	Brasília	QD SCN Q 5 BL A, 0 - ASA NORTE	YES
27	SPGTA2632	SP	Guaratinguetá	Av. Juscelino Kubitscheck de Oliveira, 351 - Buriti Shopping Guará	YES
28	SRGOIND0199	GO	Goiânia	Av. Rio Verde, Qd. 102-104	YES
29	DFIND0133	DF	Brasília	Câmara dos Deputados - Praça dos Três Poderes	YES
32	SPCAS2846	SP	Campinas	R. Jacy Teixeira Camargo, 940 - Campinas Center	YES
33	SPCGT2849	SP	Caraguatatuba	Av. Arthur Costa Filho, 937 - Caraguá Praia Shopping	YES
34	PRIND0394	PR	Londrina	RODOVIA CELSO GARCIA CID, S/N	YES
36	SRRJTNQ3055	RJ	Rio De Janeiro	Avenida Geremário Dantas, 402 - Center Shopping	YES
37	SPSJC5473	SP	São Jose Dos Campos	Av. Deputado Benedito Matarazzo, 9403 - Center Vale Shopping	YES

38	SPWVP4227	SP	São Paulo	Av. Dr. Francisco Mesquita, 1000 - Central Plaza Shopping	YES
40	SRRJBOT3027	RJ	Rio De Janeiro	Praia de Botafogo, 501 - Centro Empresarial Mourisco	YES
41	SRRJBTJ3047	RJ	Rio De Janeiro	Av Ayrton Senna, 3000 - Via Parque - Citibank Hall	YES
43	SPWJE2880	SP	São Paulo	Av. Leão Machado, 100 - Continental Shopping	YES
45	SRMGLUR4196	MG	Belo Horizonte	Avenida Olegário Maciel, 1600 - Diamond Mall Shopping	YES
47	SPMIA2860	SP	Marilia	Av. Esmeraldas, 701 - Esmeralda Plaza Shopping	YES
48	SPSOC2891	SP	Sorocaba	Av. Izoraida Marques Peres, 401 - Esplanada Shopping	YES
49	SPWVG2881	SP	São Paulo	R: José Bernardo Pinto, 333 - Expo Center Norte	YES
50	SPWST0333	SP	São Paulo	R. Arquiteto Olavo Redig de Campos, 105 - EZ Towers — Torre A	YES
51	SPFAC2636	SP	Franca	Av. Rio Negro, 1100 - Franca Shopping	YES
52	SPCAS4833	SP	Campinas	Rodovia Dom Pedro I, Km 131,5 - Galeria Shopping	YES
53	SPCTD2850	SP	Catanduva	Av. José Nelson Machado, 280 - Garden Catanduva Shopping	YES
54	MTCBA0068	MT	Cuiabá	AV. LAVA PÉS	YES
55	SPSNE2870	SP	Santo André	Av. Industrial, 600 - Grand Plaza Shopping (Antigo ABC Plaza Shopping)	YES
56	SRRJBNF3050	RJ	Rio De Janeiro	Avenida Dom Hélder Câmara, 5474	YES
59	SRCEFLA0296	CE	Fortaleza	Av Dioguinho, 4189	YES
60	RJNIU2623	RJ	Nova Iguaçu	Av Governador Roberto da Silveira, 540 - Iguaçu Top Shopping	YES
61	SPSCL2874	SP	São Carlos	Passeio dos Flamboyants, 200 - Iguatemi São Carlos	YES
62	RJPTG2607	RJ	Rio De Janeiro	Avenida Maestro Paulo Silva, 400	YES
63	SPJCI2856	SP	Jacareí	R. Barão de Jacareí, 364 - Jacareí Shopping	YES
64	SPJAU2857	SP	Jau	Av. Dr. Quinzinho, 511 - Jaú Shopping	YES
65	SRRJJCP3042	RJ	Rio De Janeiro	Avenida Embaixador Abelardo Bueno, 3401	YES
66	SPPGE5270	SP	Praia Grande	AV AYRTON SENNA DA SILVA, 1500	YES
67	PBMAN4590	PB	Joao Pessoa	Avenida Gov Flavio Ribeiro Coutinho 805/TO	YES
69	SPWST4066	SP	São Paulo	Av. Dr. Chucri Zaidan, 902	YES

70	SPMAU2861	SP	Mauá	Av. Governador Mario Covas Junior, 01 - Mauá Plaza Shopping	YES
71	SPJAI2858	SP	Jundiaí	Av. Antônio Frederico Ozanan, 6000 - Maxi Shopping Jundiaí	YES
72	SRAMMNS3304	AM	Manaus	Avenida Djalma Batista, 1661 - Millennium Shopping	YES
73	SRMGUNI4194	MG	Belo Horizonte	Avenida Cristiano Machado, 4000 - Minas Shopping	YES
75	SPSTS2871	SP	Santos	Av. Euclides da Cunha, 21 - Miramar Shopping	YES
76	SPMCZ2862	SP	Mogi Das Cruzes	Av. Ver. Narciso Yague Guimarães, 1001 - Mogi Shopping	YES
77	SPWST4065	SP	São Paulo	AV ROQUE PETRONI JUNIOR, 1089	YES
78	RNCAN6201	RN	Natal	Avenida Senador Salgado Filho 2234	YES
79	SRCEFLA0297	CE	Fortaleza	Av. Bezerra de Menezes, no. 2450 (North Shopping)	YES
80	SPRPO2866	SP	Ribeirão Preto	Av. Presidente Kennedy, 1500 - Novo Shopping	YES
81	SPOCO0087	SP	Osasco	Rua Tenente Avelar Pires de Azevedo, 81 - Osasco Plaza Shopping	YES
82	PERCE3244	PE	Recife	Rua Cais da Alfadênga S/N	YES
84	MTCBA0111	MT	Cuiabá	AV. RUBENS DE MENDONÇA	YES
85	SRRJCGR3022	RJ	Rio De Janeiro	Estrada do Monteiro, 1200 - Park Shopping Campo Grande	YES
86	SPCAS4832 / SPCAS2673	SP	Campinas	Av. Guilherme Campos, 500 - Parque Dom Pedro Shopping	YES
87	PAICO2253	PA	Belém	Rodovia Augusto Montenegro, 4300 (Parque Shopping Belém)	YES
88	SPPPE2851	SP	Presidente Prudente	Rua Siqueira Campos, 1545 - Parque Shopping Prudente	YES
89	ALMCO2399	AL	Maceió	AV MENINO MARCELO, 3800 TABULEIRO MARTINS	YES
90	SRBAIGW0208	BA	Salvador	AVENIDA ANTONIO CARLOS MAGALHAES, 1034, PITUBA PARQUE CENTER	YES
92	SPSRR2840	SP	São Jose Do Rio Preto	Av. José Munia, 4775 - Plaza Avenida Shopping	YES
93	SPITU2855	SP	Itu	Av. Dr. Ermelindo Maffei, 1199 - Plaza Shopping Itu	YES

94	SPSRR2875	SP	São Jose Do Rio Preto	R. Bernardino de Campos, 2940 - Praça Shopping	YES
95	SPSTS2872	SP	Santos	R. Alexandre Martins, 80 - Praiamar Shopping	YES
98	SPPPE2865	SP	Presidente Prudente	Av. Manoel Goulart, 2400 - Prudenshopping	YES
99	SPWVS2253	SP	São Paulo	Rod. Raposo Tavares, Km 14,5 - Raposo Shopping (Antigo Shopping Off Price Raposo)	YES
100	SRRJJBT3041	RJ	Rio De Janeiro	Rua Jardim Botânico, 266 - Rede Globo	YES
101	SPRPO5314	SP	Ribeirão Preto	Av. Cel. Fernando Ferreira Leite, 1540 - Ribeirão Shopping	YES
103	SRRJBOT3021	RJ	Rio De Janeiro	Rua General Severiano, 97 - Rio Plaza Shopping	YES
104	SPSRR2876	SP	São Jose Do Rio Preto	Av. Brigadeiro Faria Lima, 6363 - Riopreto Shopping Center	YES
106	SRMGSAV4363	MG	Belo Horizonte	Rua Alagoas, 1314 - Shopping 5 <sup>a</sup> Avenida	YES
107	SPSNE4571	SP	Santo André	Av. Pereira Barreto, 42 - Shopping ABC (Antigo Mappin ABC)	YES
109	SRCEFLA0299	CE	Fortaleza	Av. Dom Luis, no. 500 (Shopping Aldeota)	YES
110	MGNLA4722	MG	Nova Lima	Rua Senador Milton Campos, 115 - (Shopping Alta Vila)	YES
111	SPWVF4169	SP	São Paulo	Av. Regente Feijó, 1739	YES
112	SRRJCPS3057	RJ	Campos Dos Goytacazes	Av. 28 de Março, 574 - Shopping Avenida 28	YES
113	SRBAPOK0201	BA	Salvador	AVENIDA CENTENARIO, 2992	YES
114	PEBVT3241	PE	Recife	Rua do Giriquiti 48	YES
116	SRGOGNA0304	GO	Goiânia	Rua 9,1855	YES
117	SRRJCPS3058	RJ	Campos Dos Goytacazes	Rua Doutor Silvio Bastos Tavares, 330 - Shopping Boulevard Campos	YES
118	SRRJSCR3028	RJ	Rio De Janeiro	Rua Barão de São Francisco, 236 - Shopping Boulevard RioShopping antigo Iguatemi	YES
119	ESVVA3847	ES	Vila Velha	Rodovia do Sol, Km 5000 (Shopping Boulevard Vila Velha)	YES
120	SPWVS4237	SP	São Paulo	Av. Prof. Francisco Morato, 2718 - Shopping Butantã	YES

121	SPWCL0085	SP	São Paulo	Estrada do Campo Limpo, 45 - Shopping Campo Limpo	YES
122	RSCAN0958	RS	Canoas	AV GUILHERME SCHELL 6750/ TE	YES
123	PABLM2198	PA	Belém	Rod. BR - 316, Km 1, s/n (Shopping Castanheira)	YES
124	SPWCO2634	SP	São Paulo	Av. Paulista, 2064 - Shopping Center 3	YES
125	SPFND1291	SP	Fernandópolis	Av. Litério Grecco, 600 - Shopping Center Fernandópolis	YES
126	SPWLP2883	SP	São Paulo	R. Guaicurus, esquina com a Rua Catão - Shopping Center Lapa	YES
127	SPWVG4173	SP	São Paulo	TV CASALBUONO, 120	YES
128	SPWPH3954	SP	São Paulo	R. Doutor João Ribeiro, 304 - Shopping Center Penha	YES
129	SPRCO2868	SP	Rio Claro	Av. Conde Francisco Matarazzo, 205 - Shopping Center Rio Claro	YES
131	PRIND0399	PR	Curitiba	AVENIDA MARECHAL FLORIANO PEIXOTO, 4880	YES
132	SPSJC5482	SP	São Jose Dos Campos	Av. São João, 2200 - Shopping Colinas	YES
133	SPWPI3937	SP	São Paulo	AV. CRUZEIRO DO SUL, 1100 - SHOPPING D	YES
134	SPWIB3783	SP	São Paulo	Av. das Nações Unidas, 12555 - Shopping D&D	YES
135	SRRJGAV3036	RJ	Rio De Janeiro	Rua Marquês de São Vicente, 52 - Shopping da Gávea	YES
136	SRMASLS1686	MA	São Luís	Av. Daniel de La Touche, 987	YES
137	SRMGCAI4197	MG	Belo Horizonte	Avenida Presidente Carlos Luz, 3001 - Shopping Del Rey	YES
138	SPWPN3976	SP	São Paulo	Av. Rebouças, 3970	YES
139	SPWBV3590	SP	São Paulo	R. Frei Caneca, 569 - Shopping Frei Caneca	YES
140	SPCOA0258	SP	Cotia	Rodovia Raposo Tavares, km 23	YES
141	PEJBO3326	PE	Jaboatão Dos Guararapes	Avenida Barreto de Menezes 800	YES
142	SPWMM3912 / SPWMM2797	SP	São Paulo	Av. Ibirapuera, 3103 - Shopping Ibirapuera	YES
143	SRCEFLA0298	CE	Fortaleza	Av. Washington Soares, no. 85 (Shopping Iguatemi)	YES
144	SPWPN0266	SP	São Paulo	Rua Angelina Maffei Vita, s/n° - Shopping Iguatemi - Expansão Fase III	YES

145	SPWPN3977	SP	São Paulo	Av. Brig. Faria Lima, 2232	YES
146	RSIND0529	RS	Caxias Do Sul	RST 453 2787 SN	YES
147	SRBABDA0202	BA	Salvador	AVENIDA TANCREDO NEVES, 42-152, SHOPPING IGUATEMI SALVADOR	YES
148	SPWCG3654	SP	São Paulo	AV. INTERLAGOS, 2255 - Shopping Interlagos	YES
149	SCIND0277	SC	São Jose	RUA GERONCIO THIVES	YES
151	SPARQ2842	SP	Araraquara	Acesso Engenheiro Heitor de Souza Pinheiro, 2270 - Shopping Jaraguá	YES
152	SPWVA4154	SP	São Paulo	Av. Giovanni Gronchi, 5819 - Shopping Jardim Sul	YES
153	SPGJA2852	SP	Guarujá	Av. Marechal Deodoro da Fonseca, 885	YES
154	SPWVG2884	SP	São Paulo	Av. Otto Baumgart, 500 - Shopping Lar Center	YES
155	SPWAC2885	SP	São Paulo	Av. Aricanduva, 5555 - Shopping Leste Aricanduva	YES
156	SPWRE2886	SP	São Paulo	R. Coronel Xavier de Toledo, 23 - Shopping Light	YES
157	SPWBS0086	SP	São Paulo	Rua Barão de Ladário, 566/670 - Shopping Mega Polo Moda	YES
158	ESSEA3846	ES	Serra	Av. João Palácios, 300 (SHOPPING MESTRE ALVARO)	YES
159	SPWTA2887	SP	São Paulo	R. Gonçalves Crespo, 78 - Shopping Metrô Boulevard Tatuapé	YES
160	SPWIT2888	SP	São Paulo	Av. José Pinheiro Borges, s/n - Shopping Metrô Itaquera	YES
162	SPWTA4134	SP	São Paulo	R. Domingos Agostim, 91 - Shopping Metrô Tatuapé	YES
163	SPSBO2624	SP	São Bernardo Do Campo	Praça Samuel Sabatini, 200 - Shopping Metrópole	YES
165	RNLSA6033	RN	Natal	Avenida Bernardo Vieira 3775.Shopping Midway mall.	YES
166	SRRJDCT3037	RJ	Rio De Janeiro	Avenida Pastor Martin Luther King Jr, 126	YES
168	PABLM2199	PA	Belém	Trav. Padre Eutíquio, no. 1078 (Shopping Pátio Belém)	YES
171	SPWBV3936	SP	São Paulo	R. Treze de Maio, 1947 - Shopping Pátio Paulista (Antigo Shopping Paulista)	YES
172	SRMGSAV4204	MG	Belo Horizonte	Avenida do Contorno, 6061 - Shopping Pátio Savassi	YES
173	SRBAASG0203	BA	Salvador	RUA CONS. JUNQUEIRA AYRES, 165/TO	YES

174	SPPAA2864	SP	Piracicaba	Av. Limeira, 722 - Shopping Piracicaba	YES
176	RJMCE2672	RJ	Macaé	Avenida Aluízio da Silva Gomes, 800 - Shopping Plaza Macaé	YES
177	SPWSU4114	SP	São Paulo	Praça Leonor Kaupa, 100 - Shopping Plaza Sul	YES
178	SRRJMAD3044	RJ	Rio De Janeiro	Estrada da Portela, 99 - Shopping Polo1	YES
181	PEBVG3243	PE	Recife	Rua Padre Carapuceiro 777	YES
182	SRMASLS1547	MA	São Luís	Av. São Luís Rei de França, no 08, bairro Turú. Shopping Rio Anil.	YES
184	SPRPO2867	SP	Ribeirão Preto	R. São José, 933 - Shopping Santa Úrsula	YES
185	RJSGO2908	RJ	São Gonçalo	Rua Nova Jersey, s/n - Lote 1 - Shopping São Gonçalo	YES
186	MGSLA2096	MG	Sete Lagoas	Rua Otávio Campelo Ribeiro, 2801 (Shopping Sete Lagoas)	YES
188	SPTBS4662	SP	Taboão Da Serra	Rod. Régis Bittencourt, Km 271,5 - Shopping Taboão	YES
189	PESAR3429	PE	Recife	Avenida Agamenon Magalhaes 153	YES
191	SRRJRCP3029	RJ	Rio De Janeiro	Avenida Maracanã, 987 - Shopping Tijuca	YES
192	PRIND0398	PR	Curitiba	RUA ITACOLOMI, 100	YES
193	PAPUP2033	PA	Parauapebas	Rodovia PA 275, Km 55 - Shopping Unique	YES
194	MGIIG4202	MG	Ipatinga	Rodovia BR-381 (Avenida Pedro Linhares Filho), Km 206 - Industrial - Shopping Vale do Aço	YES
195	SPVOS1097	SP	Valinhos	Rua Paiquerê, 200 - Shopping Valinhos	YES
196	SRRJBTJ3048	RJ	Rio De Janeiro	Avenida Ayrton Senna, 3000 - Shopping Via Parque	YES
197	SRRJBTJ3023	RJ	Rio De Janeiro	Avenida das Américas, 3900	YES
198	SPWAP3569	SP	São Paulo	Av. das Nações Unidas, 4777 - Shopping Villa-Lobos	YES
199	ESVTA3811	ES	Vitoria	Avenida Nossa Senhora dos Navegantes, 1440 - Shopping Vitória	YES
200	SPWBF3584	SP	São Paulo	Av. Antártica, 380 - Shopping West Plaza	YES
201	RJVRD2280	RJ	Volta Redonda	Rua Brás Magaldi Fernandes, 300	YES
202	SPSOC2892	SP	Sorocaba	Av. Afonso Vergueiro, 1700/1766 - Sorocaba Shopping	YES
203	SPOCO2640	SP	Osasco	Av. dos Autonomistas, 1828 - Super Shopping Osasco	YES
204	PRCTA0576	PR	Curitiba	Supermercado Angeloni - Avenida República Argentina,	YES

205	SPSZN2893	SP	Suzano	R. Sete de Setembro, 555 - Suzano Shopping	YES
206	DFIND0131	DF	Brasília	Rua 210, QS 1, Lt. 40	YES
207	SPTTE2894	SP	Taubate	Av. Charles Schneider, 1700 - Taubaté Shopping	YES
209	SPSTB2869	SP	Santa Barbara Doeste	Av. Santa Bárbara, 777 - Tívoli Shopping	YES
210	SRRJBOT3026	RJ	Rio De Janeiro	Rua Lauro Muller, 116 - Torre do Rio Sul - Complexo do RioSul (Shopping)	YES
211	SPWST2854	SP	São Paulo	Avenida Dr. Mário Villas Boas Rodrigues, 387 - Transamérica Expo Center	YES
217	DFIND0144	DF	Brasília	CPU DARCY RIBEIRO, 0 - ASA NORTE	YES
218	SPSJC2877	SP	São Jose Dos Campos	Av. Andrômeda, 227 - Vale Sul Shopping	YES
219	SRRJIRJ3062	RJ	Rio De Janeiro	Rua Itapera, Trevo das Margaridas - Via Brasil Shopping	YES
220	SRCEFLA0920	CE	Fortaleza	Av. Washington Soares, 4335	YES
221	SPWIB3785	SP	São Paulo	Av. das Nações Unidas, 12551 - World Trade Center São Paulo	YES
222	DFIND0138	DF	Brasília	PÇA TRES PODERESLOTE ÚNICO, 0 - ZONA CÍVICO ADMINIST	YES

## SHARE PURCHASE AGREEMENT VIA ISOLATED PRODUCTION UNIT AND OTHER COVENANTS

## EXHIBIT 4.2.1(VIII)

## Existing Sharing Agreements Subject to Prior Consent

1. Infrastructure Sharing Agreement No. CO-UNA-009/2008, entered into on 07/31/2008, by and between, on one side, TNL PCS S.A., and, on the other side, Nextel Telecomunicações Ltda., as amended and modified from time to time;

2. Infrastructure Sharing Agreement No. CO/TELEMAR/UNA-IE/024-2015, entered into on 05/18/2005, by and between, on one side, Telemar Norte Leste S.A., and, on the other side, Nextel Telecomunicações Ltda., as amended and modified from time to time;

3. Infrastructure Sharing Agreement No. DNO.060/INFRASTRUCTURE/2006, entered into on 11/01/2006, by and between, on one side, OI S.A. — Under Judicial Reorganization, and, on the other side, Nextel Telecomunicações Ltda., as amended and modified from time to time;

4. Infrastructure Sharing Agreement No. DNO/117/INFRA/2003, entered into on 10/31/2003, by and between, on one side, OI S.A. — Under Judicial Reorganization and OI Móvel S.A. — Under Judicial Reorganization, and, on the other side, AMERICEL S.A. and BCP S.A., as amended and modified from time to time;

5. Infrastructure Sharing Agreement entered into on 01/31/2008, by and between, on one side, BCP S.A., and, on the other side, Amazônia Celular S.A., as amended and modified from time to time;

6. Infrastructure Sharing Agreement No. CO/TELEMAR/UNA-IE/0001-2005, entered into on 05/20/2005, by and between, on one side, Telemar Norte Leste S.A., and, on the other side, ATL Telecom S.A., AMERICEL S.A., TELET S.A., STEMAR Telecomunicações Ltda., TESS S.A., and BSE S.A., as amended and modified from time to time;

7. Letter of Intent of Mutual Infrastructure Assignment Agreement, entered into on 09/08/2003, by and between, on one side, TNL PCS S.A, and, on the other side, ATL — Algar Telecom S.A., STEMAR Telecomunicações Ltda., and BSE S.A., as amended and modified from time to time;

8. Mutual Infrastructure Assignment Agreement No. DNO.129 — INFRA/2004, entered into on 09/30/2004, by and between, on one side, OI S.A. — Under Judicial Reorganization, and OI Móvel S.A. — Under Judicial Reorganization, and, on the other side, Vivo S.A., as amended and modified from time to time;

9. Infrastructure Sharing Agreement No. CO/TELEMAR/UNA-IE/025-2006, entered into on 05/02/2006, by and between, on one side, Telemar Norte Leste S/A, and TNL PCS S.A., and, on the other side, Vivo S.A., as amended and modified from time to time;

10. Infrastructure Sharing Agreement No. CO/TELEMAR/UNA-IE — 0004-2004, entered into on 07/01/2004, by and between, on one side, Telemar Norte Leste S.A., and TNL PCS S.A., and, on the other side, Vivo S.A., as amended and modified from time to time;

11. Infrastructure Sharing Agreement entered into on 01/31/2007, by and between, on one side, Amazônia Celular S.A., and, on the other side, Vivo S/A., as amended and modified from time to time;

12. Infrastructure Sharing Agreement entered into on 06/16/2004, by and between, on one side, Telemar Norte Leste S.A, and, on the other side, Telefônica Empresas S.A., as amended and modified from time to time;

13. Mutual Infrastructure Assignment Agreement entered into on 12/01/2001, by and between, on one side, TNL PCS S.A, Telemar Norte Leste S.A., and Tele Norte Leste Participações S.A., and, on the other side, Telemig Celular S.A., Amazônia Celular S.A. — Amazonas, Amazônia Celular S.A. — Pará, Amazônia Celular S.A. — Amapá, Amazônia Celular S.A. — Maranhão, Amazônia Celular S.A. — Roraima, as amended and modified from time to time;

14. Infrastructure Sharing Agreement No. CT/OI/RO — 181-2009, entered into on 11/25/2009, by and between, on one side, Tim Celular S.A., and TIM Nordeste S.A., and, on the other side TNL PCS S.A., and OI Móvel S.A. — Under Judicial Reorganization, as amended and modified from time to time;

15. Infrastructure Sharing Agreement No. CT/OI/RO — 185-009, entered into on 11/25/2009, by and between, on one side, Tim Celular S.A., and, on the other side, OI S.A. — Under Judicial Reorganization, as amended and modified from time to time;

16. Infrastructure Sharing Agreement No. CT/OI/RO — 183-2009, entered into on 11/25/2009, by and between, on one side, Tim Celular S.A., and TIM Nordeste S.A., and, on the other side, Telemar Norte Leste S.A., as amended and modified from time to time;

17. Infrastructure Sharing Agreement No. CO/TELEMAR/UNA-IE/016-2004, entered into on 03/28/2005, by and between, on one side, Empresa Brasileira de Telecomunicações S.A. — EMBRATEL, Vésper S.A., and Vésper São Paulo S.A., and, on the other side, Telemar Norte Leste S.A., and TNL PCS S.A., as amended and modified from time to time;

18. Infrastructure Sharing Agreement No. CO/TELEMAR/UNA-IE-0010-2006, entered into on 04/13/2006, by and between, on one side, Primesys Soluções Empresariais S.A., and, on the other side, Telemar Norte Leste S.A., and TNL PCS S.A., as amended and modified from time to time;

19. Mutual Infrastructure Assignment or Sharing Agreement No. DNO/138/INFRASTRUCTURE/2004, entered into on 06/01/2004, by and between, on one side, Empresa Brasileira de Telecomunicações - EMBRATEL, Vésper S.A., and Vésper São Paulo S.A., and, on the other side, OI S.A. - Under Judicial Reorganization, and OI Móvel S.A. - Under Judicial Reorganization, as amended and modified from time to time;

20. Infrastructure Sharing Agreement No. CO/OI/RO-024-2010 and CONT/CNO/0003/2010, entered into on 03/03/2010, by and between, on one side, OI S.A. - Under Judicial Reorganization, and, on the other side, CTBC Multimídia Data NET S.A., as amended and modified from time to time;

21. Colocation Agreement (Infrastructure Sharing) No. CO/TELEMAR/UNA-IE/0005-2004, entered into on 11/01/2004, by and between, on one side, Telemar Norte Leste S.A., and, on the other side, Companhia de Telecomunicações do Brasil Central, as amended and modified from time to time;

22. Colocation Agreement (Infrastructure Sharing Agreement) No. CO/TELEMAR/UNA-IE/0009-2004, entered into on 11/01/2004, by and between, on one side, Pegasus Telecom S.A., and, on the other side, CTBC Celular Participações S.A., as amended and modified from time to time; and

23. Infrastructure Sharing Agreement No. CO/OI/DA - 219-2015, entered into on 10/26/2015, by and between, on one side, OI Móvel S.A. - Under Judicial Reorganization, and, on the other side, Algar Multimídia S.A., Algar Telecom S.A., and Algar Celular S.A., as amended and modified from time to time.

\*\_\*\_\*

## SHARE PURCHASE AGREEMENT VIA ISOLATED PRODUCTION UNIT AND OTHER COVENANTS

EXHIBIT 4.2.1(IX)

				Address	
#	Site ID	State	Municipality	Address	Lease Agreements Identified
3	ALMCO2363	AL	Maceió	AV ASSIS CHATEAUBRIAND, 2998	YES
4	ALOUB2295	AL	Ouro Branco	RUA CORONEL LUCENA, S/N LUIZ G. DE CARVALHO	YES
6	AMMNS3146	AM	Manaus	Av. J, s/n - Rua Sete de Abril (Pró-menor Dom Bosco)	YES
8	APSQA2524	AP	Santana	AV. FRANCISCO DE MELO PARENTE, 147, REMEDIOS II	YES
12	S2BABRR0061	BA	Salvador	AVENIDA DOM JOAO VI, 466	YES
14	S2BAASG0093	BA	Salvador	RUA DA MOURARIA, 6	YES
15	S2BAPOK0008	BA	Salvador	RUA PROFESSOR ARISTIDES NOVIS, 12, FEDERAÇÃO	YES
16	S2BAIGW0058	BA	Salvador	RUA RUBENS GUELLI, 134	YES
22	BACXS4693	BA	Mata De São Joao	KM 76 - COMPLEXO COSTA DO SAUIPE	YES
23	BAPGU1174	BA	Porto Seguro	RUA SÃO PEDRO, 180	YES
24	BAUIA1294	BA	Ubaitaba	RUA DO CACAU S/N	YES
29	BAMJO0786	BA	Mata De São Joao	Rodovia BA099	YES
30	BAICM1146	BA	Ilha De Comandatuba	Rua de Comandatuba, s/n	YES
31	BABRM1286	BA	Brumado	Rod. BR 30	YES
32	S2CEFLA0232	CE	Fortaleza	Rua Vicente Leite, 630 - Meireles.	YES
33	S2CEFLA0114	CE	Fortaleza	Edifício San Lucas - Rua Prof. Silo Ribeiro, 420 Papicu	YES
34	S2CEFLA0198	CE	Fortaleza	RUA PINTO MADEIRA,757	YES
35	S2CEFLA0202	CE	Fortaleza	AVENIDA ANTONIO SALES, 1317,	YES
37	CEFLA0191	CE	Fortaleza	Avenida Washington Soares, n° 55	YES
38	S2CEFLA0188	CE	Fortaleza	AVENIDA LUIS, 300	YES
39	S2CEFLA0074	CE	Fortaleza	Rua Senador Pompeu, 649	YES

40	CEFLA0230	CE	Fortaleza	RUA VILEBALDO AGUIAR, ED. JARANA 2200	YES
41	CEFLA0187	CE	Fortaleza	Av. Vicente de Castro, 6043	YES
42	CEARJ0547	CE	Barbalha	DISTRITO SITIO FARIAS, 0	YES
47	DFBSA0236	DF	Brasília	SCES trecho 2 lote 32	YES
48	DFSAM0754	DF	Brasília	QR 411CONJ. 4, CASA 14, SAMAMBÁIA	YES
49	DFSAM0753	DF	Brasília	QS604, CONJ. B (CANDIDADTO B)	YES
50	DFBSA0225	DF	Brasília	Saan Quadra 3 Lote 870 - N S/N	YES
52	DFBSA0741	DF	Brasília	SET SAFS QD 4 LT 1 ZONA CIVICO ADMINIST	YES
54	DFBSA0189	DF	Brasília	SQS 307 BLOCO J - N° S/N	YES
55	DFROD0128	DF	Brasília	Rodovia Brasília - Unaí, BR251, Km 14 - n° S/N	YES
57	DFBSA0312	DF	Brasília	SHIS QI 11, BLOCO O. LAGO SUL	YES
59	ESCNA3115	ES	Colatina	AVENIDA GETULIO VARGAS, 500	YES
61	ESCIM3808	ES	Cachoeiro De Itapemirim	RUA JOSE FARIAS DE JESUS, 8	YES
62	ESVTA3002	ES	Vitoria	AV. NOSSA SENHORA DA PENHA,1495	YES
64	GOIUB0360	GO	Itumbiara	Setor Central - Praça da República. CEP. 75503-900	YES
65	GOGNA0425	GO	Goiânia	ARAGUAIA SHOPPING - RUA 44	YES
66	DFROD0119	GO	Cristalina	Rodovia Br 040, Brasília-Belo Horizonte, Km: 35 - N S/N	YES
67	GOGNA0393	GO	Goiânia	AV. 136. QUADRA 239. LOTE 9/10.	YES
68	GOACG0278	GO	Aparecida De Goiânia	RUA CAPIBARIBE Q 22, 0	YES
69	GOGNA0492	GO	Goiânia	RUA 04 ESQ. C RUA 02-A, QD.5A, LT.01, N°1.502. SETOR CENTRAL	YES
72	GOGNA0088	GO	Goiânia	Rua 17A Esquina Com 13A, 899 — ST. Aeroporto - GO	YES
73	GOGNA0462	GO	Goiânia	RUA BENEDITO ZUPELLI, 254-320	YES
74	S2MASLS1543	MA	São Luís	Rua Grande, n°490	YES
75	MASLS1557	MA	São Luís	Av. Senador Vitorino Freire, nº 1958 — Bairro Areinha	YES
76	S2MASLS1512	MA	São Luís	AVENIDA CASTELO BRANCO, 500	YES
77	MASLS1520	MA	São Luís	Rua do Egito, 283 - Centro	YES

79	MASMH1979	MA	São Mateus Do Maranhão	RUA BENU LAGO	YES
81	MGURA4872	MG	Uberaba	Rod. BR-050, Km 195	YES
82	MGASL5930	MG	Augusto De Lima	R Geraldo Lopes Alvim, 314	YES
85	3GMGGVS5325	MG	Governador Valadares	Av. JK, 1157	YES
86	MGJFA5020	MG	Juiz De Fora	Av. Barão do Rio Branco - 2288	YES
87	MGIJA5865	MG	Itajubá	Rd. Itajubá - Maria da Fé, Km 3	YES
88	MGSAV4348	MG	Belo Horizonte	Av. João Pinheiro, 495	YES
90	MGJFA5022	MG	Juiz De Fora	AVENIDA DO RIO BRANCO, 1441	YES
91	MGSLZ2004	MG	Belo Horizonte	Avenida Antônio Carlos, 8100 - Pampulha Mall	YES
92	MGBEL2018	MG	Belo Horizonte	Rua Afonso Costa Reis, 65	YES
93	MGCTO4130	MG	Belo Horizonte	Rua Espirito Santo - 1059	YES
94	MGCTO4308	MG	Belo Horizonte	RUA CARIJOS, 424 - SALA 08 (PRAÇA SETE DE SETEMBRO)	YES
96	MGSTO4302	MG	Belo Horizonte	Rua Carangola, 670 com Rua Cristina	YES
97	MGEPL4323	MG	Belo Horizonte	RUA BEGONIA, 72	YES
98	MGFUN4153	MG	Belo Horizonte	Av. Brasil, 709	YES
99	MGJFA5046	MG	Juiz De Fora	RUA MARECHAL DEODORO, 444	YES
100	MGURA5544	MG	Uberaba	Praça Rui Barbosa, 300, (ELVIRA SHOPPING)	YES
102	MGJFA5116	MG	Juiz De Fora	AVENIDA BARÃO DO RIO BRANCO	YES
105	MGBEL4055	MG	Belo Horizonte	Rodovia Br-356, 3049 - Bh Shopping	YES
106	MGFLO4125	MG	Belo Horizonte	Rua Jacuí, 1882	YES
107	MGIRO4867	MG	Itabirito	ESTRADA PRINCIPAL, S/N	YES
108	MGMRN4603	MG	Mariana	RUA JOSE MESQUITA, 167	YES
111	MGBET6472	MG	Betim	RUA ROMUALDA AUGUSTA DE MELO, 337	YES
116	MGSOE5014	MG	São Joao Del Rei	RUA GABRIEL PASSOS, 321	YES
118	MGGON5120	MG	Goiana	RUA JOAO DE SOUZA RIBEIRO, 0	YES

120	MGFVS2311	MG	Fronteira Dos Vales	RUA BAHIA, 198	YES
122	MGURA5723	MG	Uberaba	AVENIDA FILOMENA CARTAFINA, KM 17	YES
124	MGCAA6386	MG	Capitão Andrade	AVENIDA LEVINDO DIAS, S/N	YES
125	MGMEF6434	MG	Monte Formoso	MORRO DO CEMITERIO	YES
130	MGNLA2030	MG	Nova Lima	RODOVIA JUSCELINO KUBITSCHEK / BR -040, KM 545, S/N	YES
136	MGINH5259	MG	Itanhomi	AV. SÃO CRISTOVÃO, S/N	YES
141	MTCBA0021	MT	Cuiabá	AV HIST RUBENS DE MENDONCA - BAU	YES
142	MTCBA0164	MT	Cuiabá	AV. BEIRA RIO, 3100 JARDIM EUROPA	YES
144	PABLM2019	PA	Belém	Rua Santo Antônio	YES
153	PAPGN2432	PA	Paragominas	Avenida Júlia Kubitschek, s/n Quadra 19 Lote 06	YES
160	PAIAB2438	PA	Itaituba	Rua Santo Antonio - nº 1489	YES
161	PAVCB2387	PA	Barcarena	RODOVIA PA, KM21 - DISTRITO MURUCUPI (FABRICA ALUNORTE)	YES
162	PAPUP2026	PA	Parauapebas	MINA DE FERROSOS CARAJÁS N4 (PCE)	YES
164	PBMAN4510	PB	Cabedelo	AVENIDA FLAVIO RIBEIRO COUTINHO, N°143, MANAIRA	YES
165	PBCBO4516	PB	Joao Pessoa	RUA JOSE RAMALHO LEITE, 1233	YES
166	PBMAN4559	PB	Joao Pessoa	AVENIDA SAPE, 1800	YES
167	PBCTO4571	PB	Joao Pessoa	AVENIDA BARROSO, 433	YES
172	PEESP3020	PE	Recife	AV GOV. AGAMENON MAGALHÃES 2656- ESPINHEIRO	YES
173	PERCE3244	PE	Recife	RUA CAIS DA ALFÂNDEGA, S/N - SHOPPING ALFÂNDEGA	YES
175	PEBVG3037	PE	Recife	RUA RAUL AZEDO, 53	YES
176	PEBVG3036	PE	Recife	RUA JOSE ADERVAL CHAVES, 264	YES
177	PEBVG3044	PE	Recife	Boa Viagem Avenida Boa Viagem, 4000 Edf Vânia.	YES
178	PEBVG3087	PE	Recife	RUA ANISIO RODRIGUES COELHO, 423	YES

179	PEBVT3006	PE	Recife	RUA SETE DE SETEMBRO, 238	YES
180	PEGRX3016	PE	Recife	AV GOV. AGAMENON MAGALHÃES 2656- ESPINHEIRO	YES
181	PESTA3002	PE	Recife	RUA MARQUES DE HERVAL,150 - SANTO ANTONIO - RECIFE - PE	YES
182	PEIPJ4151	PE	Ipojuca	Rua Frei Vicente Salvador, 203	YES
184	PEPOX3225	PE	Recife	Estrada das Ubaias 634	YES
185	PEESP3039	PE	Recife	RUA QUARENTA E OITO N°434 - EDF. NICOLAU II	YES
186	PEIDL3038	PE	Recife	RUA CAP. JOSÉ DA LUZ,25	YES
187	PEOLD3424	PE	Olinda	Avenida José Augusto Moreira 2258 - Edf Cidade de Olinda	YES
188	PEPNM3128	PE	Recife	Avenida 17 de Agosto 500	YES
189	PEMAD3054	PE	Recife	Rua Real da Torre, 292 - Madalena, Recife - PE	YES
190	PEMAD3011	PE	Recife	Rua Demócrito de Souza Filho 71 - EDF SOLAR DO SICUPEMA	YES
191	PESTA3001	PE	Recife	RUA IMP D PEDRO II , 325	YES
192	PESTA3022	PE	Recife	Avenida Guararapes 154	YES
194	PEBVG3058	PE	Recife	Rua Major Armando de Souza Melo, 430 - Edf Jacunã	YES
195	PEAJP3079	PE	Recife	Rua Oscár Pinto, 293 - Casa Amarela, Recife - PE	YES
196	PEDIM3236	PE	Recife	Rua Manoel de Medeiros	YES
197	PEBVT3032	PE	Recife	AVENIDA CONDE DA BOA VISTA, 331 - CENTRO	YES
199	PECQL3149	PE	Recife	Rua Alto de Bela Vista, 0	YES
200	PETMR3031	PE	Recife	R. Jundiá, 242 - Edf Sisal	YES
201	PEBVG3082	PE	Recife	RUA SGT SILVINO MACEDO, 155	YES
203	PEBVG3222	PE	Recife	Rua Fernando Simões Barbosa 316	YES
204	PEJBO3310	PE	Jaboatão Dos Guararapes	Avenida Bernardo Vieira de Melo, 4804	YES
205	PEJBO3302	PE	Jaboatão Dos Guararapes	Avenida Beira Mar, 520	YES

215	PRCTA0667	PR	Curitiba	RUA UBALDINO DO AMARAL, 37, ESTÁDIO	YES
213	FICTA0007	ΓK	Cultuba	COUTO PEREIRA	
216	PRCTA0546	PR	Curitiba	AVENIDA PARANÁ 1758	YES
217	PRCTA0661	PR	Curitiba	RUA TREZE DE MAIO, 981	YES
218	PRMGA0556	PR	Maringá	AV. DONA SOPHIA RASGULAEFF,466	YES
219	PRCTA0678	PR	Curitiba	RUA COMENDADOR ARAUJO, 730, HOTEL IBIS	YES
221	PRCPM0685	PR	Campo Mourão	Rua Laurindo Borges, 1398 - Centro, Campo Mourão - PR	YES
222	PRFOZ0722	PR	Foz Do Iguaçu	Rua Rui Barbosa, 786	YES
224	PRCTA0651	PR	Curitiba	AVENIDA PARANA 3825	YES
225	PRLDA0424	PR	Londrina	Av Higienópolis, 70 - Jardim Higienópolis	YES
226	PRCTA0690	PR	Curitiba	Rua Mariano Torres, 916	YES
229	PRFOZ0721	PR	Foz Do Iguaçu	Rua Carlos Souttomayor, 393	YES
230	PRCTA0545	PR	Curitiba	RUA JOQUEI CLUBE, 437	YES
231	PRCTA0053	PR	Curitiba	Rua Belém, 43	YES
232	PRCTA0734	PR	Curitiba	Rua Visc. do Rio Branco, 803	YES
234	PRCPP3519	PR	Cornélio Procópio	AV. NOSSA SENHORA DO ROCIO,966	YES
235	PRCTA0658	PR	Curitiba	AVENIDA SILVA JARDIM, 1856	YES
236	PRCTA0523	PR	Curitiba	RUA PADRE ANCHIETA, 2128	YES
237	PRCTA0110	PR	Curitiba	Rod. Br 277 Cta - Orleans / Rod. do Café - Cidade	YES
257	PRCTAULIO	ΓK	Cultuba	Industrial De Curitiba	
238	PRCTA0143	PR	Curitiba	Rua da Paz, 202	YES
239	PRCTA0422	PR	Curitiba	Alameda Dr. Muricy, 650	YES
240	PRCTA0423	PR	Curitiba	Rua Dr. Manoel Linhares de Lacerda, 84 - Capão Raso	YES
242	PRCTA0151	PR	Curitiba	Rua XV de Novembro, 360	YES
243	PRLDA0214	PR	Londrina	Rua Sem. Souza Naves, 45	YES
244	PRCTA0126	PR	Curitiba	Rua Sete de Abril, 571	YES
245	PRCTA0142	PR	Curitiba	Av. João Gualberto, 623	YES
248	PRPGO0467	PR	Ponta Grossa	Rua Varnhagem, 366	YES
249	PRCTA0103	PR	Curitiba	Rua Marechal Deodoro, 950	YES

250	PRLDA0219	PR	Londrina	R. Astorga, 80- Hedy	YES
253	PRCTA0469	PR	Curitiba	Av Afonso Camargo, 849 - Cristo Rei - Curitiba - PR	YES
254	PRMGA0497	PR	Maringá	Rua Néo Alves Martins, 3176 - Zona 3	YES
258	PRFOZ0428	PR	Foz Do Iguaçu	AVENIDA FELIPE WANDSCHEER 1276	YES
260	PRCTA0454	PR	Curitiba	Av. Nossa Senhora da Luz, 1800 - Jardim Social - Curitiba - PR	YES
261	PRCTA0463	PR	Curitiba	Avenida da República, 6464 - Guaíra - Curitiba - PR	YES
262	PRLDA0505	PR	Londrina	AVENIDA SANTOS DUMONT, 500-BOA VISTA	YES
277	RJRCP0091	RJ	Rio De Janeiro	AVENIDA PAULO DE FRONTIN, 739	YES
284	RJMGE0314	RJ	Magé	RUA RAFAEL COZOLINO, QUADRA L, LOTE 3.	YES
286	RJCTO0960	RJ	Rio De Janeiro	Rua São Jose, 90	YES
288	RJBTJ0411	RJ	Rio De Janeiro	Av. das Américas, 3120	YES
295	RJLEB0130	RJ	Rio De Janeiro	Rua Dias Ferreira, N° 658	YES
298	RJTNQ2900	RJ	Rio De Janeiro	AVENIDA GEREMARIO DANTAS, 402	YES
303	RJGAL0141	RJ	Rio De Janeiro	AVENIDA VINTE DE JANEIRO, S/N° - ILHA DO GOVERNADOR - RIO DE JANEIRO	YES
315	3GRJVAS3332	RJ	Vassouras	ESTRADA BARÃO DE IPIABAS, S/N° - VASSOURAS	YES
325	RNMLU6004	RN	Natal	Rua Florestal 27	YES
330	RNLNA6017	RN	Natal	Rua N. Sra. de Lourdes 1376 - Ed. Manhatam	YES
332	ROPVO3036	RO	Porto Velho	AV. GOV. JORGE TEIXEIRA, 629	YES
334	S2RSPAE0651	RS	Porto Alegre	RUA PROF. JOÃO DE SOUSA RIBEIRO, 859	YES
337	S2RSPAE0970	RS	Porto Alegre	ACESSO D SQ 3 1 UNIDADE VICINAL, 3146	YES
339	RSCSL0061	RS	Caxias Do Sul	Rua Sinimbu, 2682	YES
340	RSCSL0052	RS	Caxias Do Sul	Av Júlio de Castilhos, 4054	YES
341	RSCSL0054	RS	Caxias Do Sul	Rua Gen. Arcy da Rocha Nobrega, 504- Jardim América	YES
344	RSLHC0768	RS	Salvador Do Sul	LINHA COMPRIDA - CENTRO - SALVADOR DO SUL	YES
345	RSRGR0891	RS	Rio Grande	RUA REVOCATA DE MELLO, 51	YES
346	RSCSL0068	RS	Caxias Do Sul	Rua Remio Antônio Tomazzoni, 138 — Planalto — Caxias do Sul	YES

348	RSCAN0870	RS	São Gabriel	RUA FERNANDO FERRARI, S/N	YES
349	RSPAE0971	RS	Porto Alegre	ALAMEDA J - UNIDADE VICINAL, 183	YES
351	RSPAE0857	RS	Porto Alegre	AVENIDA CRUZEIRO DO SUL, 2323	YES
362	SCPRI0545	SC	Princesa	Rua Santa Catarina, s/n°	YES
364	3GSCFNS0405	SC	Florianópolis	Av. Rio Branco, 369	YES
365	SCWUM0274	SC	Witmarsum	Estrada Geral, s/n°	YES
366	SCFNS0333	SC	Florianópolis	R. do Calafate, 113 — BL2 - PANTANAL	YES
369	SCJGS0311	SC	Jaraguá Do Sul	Rua Gerhardt, s/n — Rio da Luz	YES
371	SCSOO0503	SC	São Jose	RUA AMBRÓSIO DA SILVA, 223	YES
373	3GSCFNS0382	SC	Florianópolis	RUA LACERDA COUTINHO,179	YES
374	SCBCU0351	SC	Balneário Camboriú	AV. BRASIL, 3080	YES
377	3GSCFNS0572	SC	Florianópolis	ROD. TERTULIANO BRITO XAVIER, 2352	YES
381	SPCAO4938	SP	Cubatão	RUA JOSE GONÇALVES TORRES S.N - CUBATÃO/SP	YES
382	SPWCV3688	SP	São Paulo	RUA PRO F VILMA VIEIRA, 21 - CASA VERDE	YES
383	SPWMM3911	SP	São Paulo	Av Moema, 81 - Planalto Paulista, São Paulo - SP	YES
384	SPWPC3942	SP	São Paulo	RUA MANOEL DA MATA SÁ 170	YES
385	SPWFO3735	SP	São Paulo	RUA ISANGA, Nº 521 - FREGUESIA DO Ó	YES
386	SPWCM3694	SP	São Paulo	Rua Pedro Rodrigues Beja, 111 — Vila Marari	YES
387	SPWVF4163	SP	São Paulo	RUA ANTONIO REIS N 11B — CHÁCARA BELENZINHO	YES
388	SPWPT3996	SP	São Paulo	Rua Dr João Toniolo 535 — Jardim São José	YES
389	SPWCM3695	SP	São Paulo	RUA FERRUCIO SANDOLI 232 — SÃO PAULO	YES
390	SPWST3117	SP	São Paulo	Av. Roque Petroni Júnior, 1089 - Chácara Santo Antônio (Zona Sul), São Paulo - SP	YES
392	SPWVD3048	SP	São Paulo	Rua Oscar de Medeiros, 164, Vila Medeiros	YES
393	SPWGR3743	SP	São Paulo	Rua DR. Antônio Simões de Carvalho,105 - Jd. São Bernardo -São Paulo - SP	YES

395	SPFAC4958	SP	Franca	Av Presidente Vargas 666 — Cidade Nova, Franca - SP	YES
396	SPWCS3728	SP	São Paulo	Rua Girolamo Dai Libri, 433 - Vila Moraes	YES
397	SPWAA3576	SP	São Paulo	Rua da Padroeira, 83 - Jardim Nordeste, São Paulo - SP	YES
398	SPOCO4536	SP	Osasco	Rua Rosa D'ângelo Pisapia, 156 — Vila Yara	YES
399	SPSBO4598	SP	São Bernardo Do Campo	Rua Ruy Barbosa, 260	YES
400	SPWBE3603	SP	São Paulo	RUA JOAO BATISTA DE LACERDA, 261	YES
401	SPSBO4607	SP	São Bernardo Do Campo	Rua Municipal, 193	YES
402	SPWBF3583	SP	São Paulo	Av. Francisco Matarazzo, 229 - Água Branca, São Paulo - SP	YES
403	SPWBF3585	SP	São Paulo	Rua Josef Kryss, 319 - Parque Industrial Tomas Edson	YES
404	SPSBO4630	SP	São Bernardo Do Campo	Rua Nelza Ascêncio, 128 — Ferrazópolis — São Bernardo do Campo	YES
405	SPSCN4637	SP	São Caetano Do Sul	Rua Archinto Ferrari, 118 — Santo Antônio — São Caetano do Sul	YES
406	SPWBV3589	SP	São Paulo	AVENIDA PAULISTA, 810	YES
407	SPWCB3641	SP	São Paulo	Rua Ribeiroles, 274 — Vila Paulista	YES
408	SPMMM5169	SP	Mogi Mirim	Rua Coronel Venâncio Ferreira Alves Adorno, 89 — Saúde	YES
409	SPRPO5341	SP	Ribeirão Preto	Rua Bernardino de Campos, 1236	YES
410	SPSTB5363	SP	Santa Barbara Doeste	RUA TREZE DE MAIO, 749 - CENTRO	YES
411	SPSJC5460	SP	São Jose Dos Campos	Rua Euclídes Miragaia, 26 — Centro	YES
412	SPPGE5272	SP	Praia Grande	AV. PRESIDENTE COSTA E SILVA, 13 — PRAIA GRANDE - SP	YES
414	SPSTS5379	SP	Santos	Rua Saturnino de Brito, 256 — Marapé — Santos	YES
415	SPSTS5396	SP	Santos	Av. Ana Costa, 121 - Vila Matias	YES
416	SPSTS5403	SP	Santos	Rua Vergueiro Steidel, 304 - Aparecida	YES

417	SPSUM5601	SP	Sumaré	Rua João Antônio Soares, 18 — Jd. Bom Retiro - Sumaré	YES
418	SPWPN3988	SP	São Paulo	Avenida Nove de Julho, 5624 — Jardim Paulista — São	YES
418	SP W PIN 5966	SP	Sao Paulo	Paulo	
419	SPWTA4138	SP	São Paulo	Rua Cantagalo, 738 — Vila Gomes Cardim - SP	YES
420	SPWPZ3963	SP	São Paulo	Rua Dr. Homem de Melo, 1186 - Perdizes	YES
421	SPWPZ3964	SP	São Paulo	Rua Dr. Homem de Melo, 352 — Perdizes — São Paulo	YES
422	SPWJQ3800	SP	São Paulo	Rua das Oiticicas, 149 — Pq. Jabaquara - São Paulo - SP	YES
423	SPWJQ3801	SP	São Paulo	Rua Mianos, 104 — Vila Santa Catarina - SP	YES
425	SPWRE4007	SP	São Paulo	Alameda Barão de Limeira, 478 - Campos Elíseos, São	YES
423	SF W KE4007	51	Sao Faulo	Paulo - SP	
426	SPWRE4010	SP	São Paulo	Avenida Cásper Libero, 58 — Centro Histórico de São	YES
420	SF W KE4010	51	Sao Faulo	Paulo	
427	SPWRE4011	SP	São Paulo	Rua Conselheiro Ramalho, 104 — Bela Vista	YES
428	SPWVJ4178	SP	São Paulo	Rua Édipo Felíciano, 100 - Vila Jacuí, São Paulo — SP	YES
429	SPWLB3880	SP	São Paulo	Rua do Glicério, 731 — Liberdade — São Paulo	YES
430	SPWLB3881	SP	São Paulo	Rua Vergueiro, 168 — Liberdade — São Paulo	YES
433	SPWVM4212	SP	São Paulo	Rua Eça de Queiroz, 258 — Vila Mariana - SP	YES
434	SPWLP3868	SP	São Paulo	Rua Tonelero, 239 — Vila Ipojuca — São Paulo - SP	YES
435	SPWLP3870	SP	São Paulo	Rua Marcelina, 394 — Vila Romana - SP	YES
436	SPWSC4040	SP	São Paulo	Rua Albuquerque Lins, 374 - Santa Cecilia, São Paulo -	YES
437	SPWSD4078	SP	São Paulo	SP Av General Charles de Gaulle, 208 — Pq. São Domingos	YES
437	SPWVM4216	SP SP	São Paulo	R. Joel Jorge de Melo, 62 - Vila Mariana	YES
440	SPWSE4124	SP	São Paulo	Praça Doutor João Mendes, 62 — Liberdade	YES
440	SPW3E4124 SPWMM3898	SP SP	São Paulo	RUA SALTO, 70 — MOEMA	YES
441	SPWMM3901	SP SP	São Paulo	Av. dos Imarés, 88 - Indianópolis, São Paulo - SP	YES
442	SPWMM3903	SP SP	São Paulo	Alameda Jauaperi, 1083 — Moema - SP	YES
		SP		*	YES
446	SPWMR3927	SP SP	São Paulo	Avenida Barão de Monte Mor, 75 — Vila Morumbi	
447	SPWSN4057	SP	São Paulo	Rua Voluntários da Pátria, 2798 - Santana, São Paulo - SP	YES

					:
449	SPWSU4103	SP	São Paulo	Rua Estero Belaco 160 - Vila da Saúde, São Paulo - SP	YES
450	SPWSU4117	SP	São Paulo	Av. Jabaquara, 970 — Saúde — São Paulo	YES
452	SPWCO3064	SP	São Paulo	Av. Angélica, 1683 - Santa Cecilia, São Paulo - SP	YES
453	SPWSU3522	SP	São Paulo	Avenida Indianópolis, 1192 — (Clube Sírio Libanês) - Planalto Paulista, São Paulo	YES
454	SPWBL3618	SP	São Paulo	RUA EDUARDO COSTA, N°244 - VILA BRASILÂNDIA	YES
457	SPWCO3717	SP	São Paulo	Rua Minas Gerais, 388 - Higienópolis	YES
458	SPSBO4605	SP	São Bernardo Do Campo	Rua Marechal Badoglio, 15 — Vila Mussoline	YES
459	SPWJR3818	SP	São Paulo	Rua Conselheiro Amaral, 196 - Jaguará, São Paulo - SP	YES
460	SPWPN3987	SP	São Paulo	Rua Hungria, 664 - Jardins, São Paulo - SP	YES
461	SPWTA4136	SP	São Paulo	RUA GEN. GELÁS, 72 - TATUAPÉ, SÃO PAULO-SP - 03090-030	YES
462	SPWRE4012	SP	São Paulo	Rua Washington Luís, 392 — Centro Histórico — São Paulo	YES
463	SPWSC4044	SP	São Paulo	Praça Júlio Prestes, 30 - Campos Elíseos, São Paulo - SP	YES
464	SPWMM3907	SP	São Paulo	Alameda dos Tupiniquins, 1235 - Planalto Paulista	YES
465	SPWPN3972	SP	São Paulo	Rua Fradique Coutinho, 1590 — Pinheiros — São Paulo - SP	YES
466	SPWMM2564	SP	São Paulo	Rua inhambu, 763	YES
467	SPWVM3099	SP	São Paulo	Av. Brigadeiro Luís Antônio, 2819 — Jardim Paulista	YES
468	SPWLB3069	SP	São Paulo	Rua Topázio, 420 — Vila Mariana	YES
469	SPLRA5117	SP	Limeira	Rua Dr. José Carvalho Ferreira, s/n	YES
470	SPWJE3059	SP	São Paulo	Rua Caetanópolis, 930 - Jaguaré	YES
471	SPWCO3711	SP	São Paulo	Rua da Consolação, 2143	YES
473	SPWCO3721	SP	São Paulo	Av Angélica, N° 1189 — Consolação	YES
474	SPJAI4454	SP	Jundiaí	Rua do Rosário	YES
475	SPSTS5374	SP	Santos	Rua Augusto Paulino, 57	YES

176	CDWDW2506	CD	Cão Doulo	DUA LOUDENCO CDANATO 214	VEC
476	SPWBV3596	SP	São Paulo	RUA LOURENCO GRANATO, 214	YES
477	SPWCY3637	SP	São Paulo	AVENIDA LACERDA FRANCO, 130, CEP:1536-000	YES
478	SPWAR3557	SP	São Paulo	RUA TEOFILO DIAS, 332	YES
479	SPWBS3612	SP	São Paulo	RUA JOSE MONTEIRO, 323	YES
480	SPWIT3794	SP	São Paulo	AVENIDA CAMPANELLA, EDIFÍCIO: CAMPANELLA 1297 — ITAQUERA	YES
481	SPSTS5387	SP	Santos	Rua Ana Costa, 555	YES
482	SPSTS5389	SP	Santos	Rua Barão de Paranapiacaba, 77	YES
483	SPSTS5397	SP	Santos	Rua Euclides da Cunha, 151	YES
487	SPWJP3833	SP	São Paulo	Rua José Maria Lisboa, 1187	YES
488	SPWJP3845	SP	São Paulo	Rua Pamplona, 825 - Jardim Paulista	YES
489	SPWPZ3957	SP	São Paulo	Rua Diana, 740 — Perdizes	YES
490	SPWJQ3807	SP	São Paulo	Rua das Grumixamas, 110 — Jardim Oriental	YES
491	SPWRT4003	SP	São Paulo	Rua Cândido Fontoura, 481	YES
492	SPWSC4035	SP	São Paulo	Rua Guaianases, 1197	YES
493	SPWMR3923	SP	São Paulo	Av. Lineu de Paula Machado, 1174 Jardim Everest	YES
494	SPWIR3748	SP	São Paulo	Rua Vasconcelos Drumond, 196, Ipiranga	YES
495	SPWIR3751	SP	São Paulo	Rua Costa Aguiar, 994 - Ipiranga	YES
496	SPWSU4108	SP	São Paulo	Av. Indianópolis, 2185	YES
497	SPARC3152	SP	Araçatuba	Rua Wenceslau Braz 120 - Ed Copacabana — Vila Santa Maria	YES
498	SPSTS3213	SP	Santos	Avenida Presidente Wilson, 66 — José Menino	YES
499	SPSZN3100	SP	Suzano	Rua Benjamin Constant, 1653	YES
500	SPWMM3096	SP	São Paulo	Alameda dos Anapurus, 1297 - Moema	YES
501	SPWJP3067	SP	São Paulo	Alameda Tietê, 301 — Jardim Paulista	YES
502	SPWJG3820	SP	São Paulo	Rua Sebastiao Laranjeiras 252	YES
503	SPWLJ3865	SP	São Paulo	RUA GIBOUCUS 62	YES
504	SPWCO3714	SP	São Paulo	Avenida Angélica, 2000 — Consolação — São Paulo	YES
505	SPWPN3974	SP	São Paulo	Rua Joaquim Antunes, 149 - Pinheiros — São Paulo	YES

506	SPWVM4205	SP	São Paulo	Rua Conselheiro Rodrigues Alves, 391 — Vila Mariana	YES
507	SPWLM3888	SP	São Paulo	Rua Antônio Ribeiro de Morais, 264 - Vila Carbone, São Paulo - SP	YES
508	SPWCO3719	SP	São Paulo	Rua Amaral Gurgel, 518 — Consolação	YES
509	SPWRE4004	SP	São Paulo	Rua Coronel Xavier de Toledo, 70 - República	YES
510	SPWRE4005	SP	São Paulo	Av. Prestes Maia, 702	YES
511	SPWIB3779	SP	São Paulo	Rua Jesuíno Arruda, 327 - Itaim Bibi	YES
512	SPWIR3753	SP	São Paulo	Rua Costa Aguiar, 2446 - Ipiranga	YES
513	SPWPZ3047	SP	São Paulo	Rua Campevas, n° 313/325 - Perdizes	YES
514	SPGJA4743	SP	Guarujá	Rua Mário Ribeiro, 886	YES
515	SPGJA5009	SP	Guarujá	Rua José Avelino de Oliveira, 260	YES
517	SPWJP3849	SP	São Paulo	Av. Paulista, 2006 — Bela Vista	YES
518	SPWJP3854	SP	São Paulo	Avenida Brasil, S/N - Jardim América (Praça Nossa Senhora do Brasil)	YES
519	SPWIR3760	SP	São Paulo	Rua Vemag 1, Vila Independência	YES
520	SPWJP3068	SP	São Paulo	Rua Batatais, 558	YES
521	SPPAA5239	SP	Piracicaba	Rua Praça da Catedral Dom Ernesto de Paula, 170 - Centro	YES
522	SPWCL3670	SP	São Paulo	Rua Caverá, 55 — Jardim Umarizal	YES
523	SPWAN3570	SP	São Paulo	AV. Felippo Sturba, S/N - Anhanguera	YES
524	SPCAS4848	SP	Campinas	Rua General Osório, 1233 — Centro	YES
525	SPCAS4871	SP	Campinas	Rua Júlio de Mesquita, 36	YES
526	SPCAS4876	SP	Campinas	Rua Vargem Grande do Sul, 506 - Esquina com Avenida Baden Powell	YES
527	SPITA4429	SP	Itatiba	Rua Quintino Bocaiúva, 343	YES
528	SPSOC5568	SP	Sorocaba	Rua Capitão Nascimento Filho, 171	YES
529	SPWMC3921	SP	São Paulo	Rua João Caetano, 379	YES
530	SPWCN3674	SP	São Paulo	R. Maria Angélica Franci, - Vila Franci	YES
531	SPWIP3017	SP	São Paulo	R Manuel Martins de Melo, S/N- Vila Itaim	YES

532	SPWIP3018	SP	São Paulo	Avenida Tomé Dias Laço, S/N — Itaim Paulista	YES
533	SPWJH3830	SP	São Paulo	Rua Tetiximira, 146 — Jardim Helena	YES
534	SPGRS3000	SP	Guarulhos	Rua Candelária, S/N - Jardim Jacy	YES
535	SPWMR2678	SP	São Paulo	Rua Jules Rimet, estádio Cícero P. de Toledo, portão04, Morumbi	YES
536	SPWBS5671	SP	São Paulo	RUA DO GASOMETRO, 741, CEP: 3004 - 000	YES
537	SPWBU3625	SP	São Paulo	RUA ARMANDO FAIRBANKS, 310	YES
538	SPWJP3841	SP	São Paulo	Avenida São Gabriel, 236 — Jardim Paulista	YES
539	SPWJP3851	SP	São Paulo	Alameda Campinas, 1600	YES
540	SPWSC4038	SP	São Paulo	Rua Albuquerque Lins, 92	YES
541	SPWIB3777	SP	São Paulo	Rua Jacurici, 184 — Itaim Bibi	YES
542	SPRPO5315	SP	Ribeirão Preto	Rua Ana Costa, 555	YES
543	SPPGE5273	SP	Praia Grande	Rua São Salvador, 50 - Guilhermina	YES
544	SPSTS5391	SP	Santos	Av. Presidente Wilson, 90 - José Menino	YES
546	SPWST4072	SP	São Paulo	Rua Rodrigues Pais, 193	YES
547	SPWPN3985	SP	São Paulo	Rua Diana, 740	YES
548	SPCAS4870	SP	Campinas	Rua José de Alencar, 675 - Centro	YES
549	SPWJQ3808	SP	São Paulo	Rua Porcelana, 193 — Vila do Encontro — São Paulo	YES
550	SPWIB3769	SP	São Paulo	AVENIDA HORÁCIO LAFER, 702	YES
551	SPWLP3869	SP	São Paulo	Rua Albion, 229 - Lapa, São Paulo - SP	YES
552	SPWBV3588	SP	São Paulo	ALAMEDA RIO CLARO, BELA VISTA - 157 SÃO -SP	YES
553	SPWPS3970	SP	São Paulo	R Antônio José Anacleto, S/N, Vila Fanton	YES
554	SPWMM3098	SP	São Paulo	Av. Jacutinga, 632 - Indianópolis	YES
555	SPWBU3628	SP	São Paulo	AVENIDA CAXINGUI, 175	YES
556	SPPGE5275	SP	São Paulo	RUA BALNEARIO CAMBURIU, 100	YES
557	SPWJP3853	SP	São Paulo	AL ITU,1067 VILA CARAGUATA CONDOMÍNIO EDIFÍCIO JOSÉ HACHEM	YES
558	SPWIB3784	SP	São Paulo	Av. Engenheiro Carlos Berrini, 801	YES
559	SPWJP3840	SP	São Paulo	AVENIDA PAULISTA, EDIFÍCIO ASAHI,1274	YES

560	SPWSN4046	SP	São Paulo	RUA, FRANCISCO RAMOS N, 28 SANTANA	YES
561	SPWSE4118	SP	São Paulo	AV DO ESTADO, Nº 2854, CENTRO	YES
562	SPOCO4514	SP	Osasco	Rua Dr. Mariano Jatahy Marcondes Ferraz, 215 - Centro	YES
563	SPRPO5325	SP	Ribeirão Preto	Rua Sete de Setembro,590	YES
564	SPWIB3761	SP	São Paulo	R. GUARAIUVA, 599	YES
565	SPWSL4085	SP	São Paulo	Avenida Sapopemba, 5435 — Sapopemba — São Paulo	YES
566	SPWSN4059	SP	São Paulo	Rua Valdemar Martins, 116	YES
567	SPWRE4015	SP	São Paulo	RUA AMARO, 418,	YES
568	SPWBV3594	SP	São Paulo	RUA PEIXOTO GOMIDE, 581	YES
569	SPWCB3644	SP	São Paulo	Rua Baronesa da Bela Vista, 801 (Hotel Ibis) - Vila	YES
570		SP	C~- D1-	Congonhas	VEC
570	SPWRE4006		São Paulo	RUA BENTO FREITAS 186	YES
572	SPGRS0436	SP	Guarulhos	Rua Caraguatatuba, 57	YES
573	SPWCS3725	SP	São Paulo	RUA KITIZU UTIYAMA, 314, CURSINO	YES
574	SPWJE3824	SP	São Paulo	Rua Marselha, 94, Jaguaré	YES
575	SPWVC3023	SP	São Paulo	AV. FLAMINGO, 1361 - VILA NOVA CURUÇÁ	YES
576	SPCAS4840	SP	Campinas	RUA BARRETO LEME, 2101	YES
577	SPCAS4857	SP	Campinas	Rua Doutor Sales de Oliveira, 463 - Vila Industrial - Campinas - SP	YES
578	SPCAS4882	SP	Campinas	Rua Doutor Quirino, 757 - Centro, Campinas - SP	YES
579	SPWCL3663	SP	São Paulo	Rua Odemis, 520	YES
580	SPWBF3579	SP	São Paulo	Largo Péricles, 0, Barra Funda	YES
581	SPSBO4601	SP	São Bernardo Do	PÇA SAMUEL SABATINI, 200 - CENTRO - SÃO	YES
381	2L2R04001	5r	Campo	BERNARDO CAMPO	
592	SPSBO4608	SP	São Bernardo Do	Rua Braga, 266 - Vila Lusitania, São Bernardo do Campo	YES
582	SL2R04008	5r	Campo	- SP	
583	SPSBO4613	SP	São Bernardo Do	AVENIDA JOÃO FIRMINO, 1229 - ASSUNÇÃO - SÃO	YES
202	SLSD04012	ы	Campo	BERNARDO DO CAMPO - SP	

584	SPSCN4634	SP	São Caetano Do Sul	Rua Rafael Correia Sampaio, 977 - Santa Paula - S.C.S	YES
585	SPWBV3587	SP	São Paulo	Rua Treze de Maio, 1566 - Bela Vista - São Paulo	YES
587	SPSTS5383	SP	Santos	Av. Dona Ana Costa, 376 - Gonzaga - Santos	YES
590	SPWTA4137	SP	São Paulo	Rua Apucarana, 1563 - Tatuapé - São Paulo	YES
591	SPWJP3834	SP	São Paulo	Alameda Joaquim Eugênio de Melo, 1452 - Jd. Paulista - São Paulo	YES
592	SPWJP3844	SP	São Paulo	Alameda Jaú, 1167 - Jardim Paulista - SP	YES
593	SPWJP3848	SP	São Paulo	Rua Maestro Elias Lobo, 1040 - Jardim Paulista - São Paulo	YES
594	SPWPZ3959	SP	São Paulo	Rua Monte Alegre, 470 - Perdizes - São Paulo	YES
595	SPWRE4009	SP	São Paulo	Lago do Arouche, 96 - República - SP	YES
596	SPWSC4043	SP	São Paulo	Rua Dr Gabriel Dos Santos, 759 - Santa Cecilia	YES
597	SPWVM4217	SP	São Paulo	RUA MANUEL DA NÓBREGA, 181	YES
598	SPWMM3905	SP	São Paulo	Rua Gaivota, 22 - Mo ema	YES
599	SPWMM3910	SP	São Paulo	Rua Gaivota, 665 - Indianópolis - São Paulo	YES
600	SPWIB3773	SP	São Paulo	Av. Jurubatuba, 246 - Morumbi	YES
601	SPWTA4133	SP	São Paulo	RUA EMILIO MALLET, 1893, VILA GOMES CARDIM	YES
602	SPSTS2545	SP	Santos	Avenida Washington Luiz, 61 - Vila Matias- Santos - SP	YES
603	SPSCN3086	SP	São Caetano Do Sul	Rua Alameda São Caetano, 1738 - Santa Maria - S.C.S	YES
604	SPWRE3072	SP	São Paulo	Rua Vitória, 395	YES
607	SPWVS4235	SP	São Paulo	RUA JOAO GUILHERME, 239	YES
608	SPWVA4157	SP	São Paulo	Rua Chico Gomes, S/N°	YES
609	SPWLM3887	SP	São Paulo	R. Dirce Rodrigues 33	YES
610	SPCAS4865	SP	Campinas	Av. Barão de Itapura, 1518 - Jardim Guanabara, Campinas - SP	YES
611	SPGJA5000	SP	Guarujá	Rua José da Silva Figueiredo, 525 Enseada - Guarujá	YES
612	SPINE5044	SP	Itanhaém	Rua João Mariano, 193 - Centro - Itanhaém	YES

613	SPWBL3617	SP	São Paulo	ESTRADA DO SABÃO, 40	YES
614	SPRPO5311	SP	Ribeirão Preto	RUA AMÉRICO BRASILIENSE, 413	YES
616	SPPGE5280	SP	Praia Grande	Rua Nicarágua, 49 - Praia Grande - São Paulo	YES
618	SPWSC4034	SP	São Paulo	Rua Cruzeiro, 732, Barra Funda	YES
619	SPWMM3906	SP	São Paulo	R Dra. Neyde Aparecida Solito, 460 - Vila Clementino - Moema - SP	YES
620	SPCAS2549	SP	Campinas	RUA BARÃO DE JAGUARA, 707 - CONCEIÇÃO	YES
621	SPSCL2559	SP	São Carlos	Rua Dona Maria Jacinta, 241, São Carlos - SP	YES
622	SPGRS3002	SP	Guarulhos	Av. São Paulo, 352 - Cidade Brasil - Guarulhos - SP	YES
623	SPWCT3015	SP	São Paulo	Rua Jaguapeba, 100	YES
624	SPWPT3039	SP	São Paulo	Rua Cel. Gonçalves da Siqueira 145, Pirituba	YES
625	SPWRE3073	SP	São Paulo	Rua Augusta, 66	YES
626	SPJCI1883	SP	Jacareí	R. Cap. João Jose de Macedo, 340	YES
627	SPWPZ3961	SP	São Paulo	Rua Coari, 208	YES
628	SPWSL4088	SP	São Paulo	Rua Joaquim Ribeiro da Costa, 386	YES
629	SPWBF3580	SP	São Paulo	Rua Turiassú, 1863, Perdizes	YES
630	SPSJC5491	SP	São Jose Dos Campos	Rua Cel João Cursino, 240/242 - Ed. Toulon	YES
631	SPSTS5402	SP	Santos	RUA LACERDA FRANCO, Condomínio Ed. Icaraíma - Bairro Aparecida 137, CEP:11025180	YES
634	SPWPI3940	SP	São Paulo	Av. Cruzeiro do Sul, 1100 - Canindé	YES
635	SPJCI1882	SP	Jacareí	Rua Gilberto Moreira, 296 - Vila Formosa, Jacareí - SP	YES
636	SPWCB3651	SP	São Paulo	Rua Barão do Triunfo, 276	YES
637	TOPMJ2109	ТО	Palmas	AVENIDA LO-5, LT 01 - PLANODIRETOR SUL	YES

2	SRMGBPR4203	MG	Belo Horizonte	Avenida Augusto de Lima, 1549 - Fórum Lafayete	YES
3	RJSMI2621	RJ	São Joao De Meriti	Rodovia Presidente Dutra, 4200 - Shopping Grande Rio	YES
5	SPWCB3642	SP	São Paulo	Av. Washington Luís, s/n - Aeroporto Internacional de Congonhas	YES

6	SRCEFLA0142	CE	Fortaleza	Av. Senador Carlos Jereissati, 3000	YES
7	SPGRS2853	SP	Guarulhos	Rod. Presidente Dutra, Km 230 - Internacional Shopping Guarulhos	YES
8	SPGRS4359 / SPGRS4360	SP	Guarulhos	Rod. Hélio Schmidt, s/n - Aeroporto Internacional de São Paulo (Cumbica)	YES
9	SRBASCT0204	BA	Salvador	PRACA GAGO COUTINHO, 282,	YES
10	SRRJCTO3007	RJ	Rio De Janeiro	Praça Senador Salgado Filho, S/N - Aeroporto Santos Dumont	YES
11	DFIND0132	DF	Brasília	CSB 2, Lotes 1/4, Alameda Shopping	YES
12	SPBRE4271	SP	Barueri	Al. Rio Negro, 1033 - Alphaville Shopping	YES
14	SPARC2841	SP	Araçatuba	Av. Joaquim Pompeu de Toledo, 601 - Araçatuba Shopping	YES
17	SPBRU2845	SP	Bauru	R. Henrique Savi, 15-55 - Bauru Shopping	YES
18	SRRSIND0519	RS	Porto Alegre	AVENIDA PRAIA DE BELAS, 1181	YES
19	SRMGBEL4193	MG	Belo Horizonte	Rodovia BR-356, 3049 - BH Shopping	YES
20	MGCEM4198	MG	Contagem	Avenida João César de Oliveira, 1275 - Big Shopping (32260-250)	YES
21	SPWST4067	SP	São Paulo	R. Borba Gato, 59 - Boa Vista Shopping	YES
22	RJBOT2604	RJ	Rio De Janeiro	Praia de Botafogo, 400 - Botafogo Praia Shopping	YES
23	DFIND0726	DF	Brasília	SETOR TERMINAL NORTE, CONJUNTO J, BOULEVARD SHOPPING	YES
24	SPWBF2628	SP	São Paulo	Rua Turiassú, 2100	YES
25	DFIND0130	DF	Brasília	QD SCN Q 5 BL A, 0 - ASA NORTE	YES
27	SPGTA2632	SP	Guaratinguetá	Av. Juscelino Kubitscheck de Oliveira, 351 - Buriti Shopping Guará	YES
28	SRGOIND0199	GO	Goiânia	Av. Rio Verde, Qd. 102-104	YES
29	DFIND0133	DF	Brasília	Câmara dos Deputados - Praça dos Três Poderes	YES
32	SPCAS2846	SP	Campinas	R. Jacy Teixeira Camargo, 940 - Campinas Center	YES
33	SPCGT2849	SP	Caraguatatuba	Av. Arthur Costa Filho, 937 - Caraguá Praia Shopping	YES

34	PRIND0394	PR	Londrina	RODOVIA CELSO GARCIA CID, S/N	YES
36	SRRJTNQ3055	RJ	Rio De Janeiro	Avenida Geremário Dantas, 402 - Center Shopping	YES
37	SPSJC5473	SP	São Jose Dos Campos	Av. Deputado Benedito Matarazzo, 9403 - Center Vale Shopping	YES
38	SPWVP4227	SP	São Paulo	Av. Dr. Francisco Mesquita, 1000 - Central Plaza Shopping	YES
40	SRRJBOT3027	RJ	Rio De Janeiro	Praia de Botafogo, 501 - Centro Empresarial Mourisco	YES
41	SRRJBTJ3047	RJ	Rio De Janeiro	Av Ayrton Senna, 3000 - Via Parque - Citibank Hall	YES
43	SPWJE2880	SP	São Paulo	Av. Leão Machado, 100 - Continental Shopping	YES
45	SRMGLUR4196	MG	Belo Horizonte	Avenida Olegário Maciel, 1600 - Diamond Mall Shopping	YES
47	SPMIA2860	SP	Marilia	Av. Esmeraldas, 701 - Esmeralda Plaza Shopping	YES
48	SPSOC2891	SP	Sorocaba	Av. Izoraida Marques Peres, 401 - Esplanada Shopping	YES
49	SPWVG2881	SP	São Paulo	R: José Bernardo Pinto, 333 - Expo Center Norte	YES
50	SPWST0333	SP	São Paulo	R. Arquiteto Olavo Redig de Campos, 105 - EZ Towers — Torre A	YES
51	SPFAC2636	SP	Franca	Av. Rio Negro, 1100 - Franca Shopping	YES
52	SPCAS4833	SP	Campinas	Rodovia Dom Pedro I, Km 131,5 - Galeria Shopping	YES
53	SPCTD2850	SP	Catanduva	Av. José Nelson Machado, 280 - Garden Catanduva Shopping	YES
54	MTCBA0068	MT	Cuiabá	AV. LAVA PÉS	YES
55	SPSNE2870	SP	Santo André	Av. Industrial, 600 - Grand Plaza Shopping (Antigo ABC Plaza Shopping)	YES
56	SRRJBNF3050	RJ	Rio De Janeiro	Avenida Dom Hélder Câmara, 5474	YES
59	SRCEFLA0296	CE	Fortaleza	Av Dioguinho, 4189	YES
60	RJNIU2623	RJ	Nova Iguaçu	Av Governador Roberto da Silveira, 540 - Iguaçu Top Shopping	YES
61	SPSCL2874	SP	São Carlos	Passeio dos Flamboyants, 200 - Iguatemi São Carlos	YES
62	RJPTG2607	RJ	Rio De Janeiro	Avenida Maestro Paulo Silva, 400	YES
63	SPJCI2856	SP	Jacareí	R. Barão de Jacareí, 364 - Jacareí Shopping	YES

64		CD	т		MEG
64	SPJAU2857	SP	Jau	Av. Dr. Quinzinho, 511 - Jaú Shopping	YES
65	SRRJJCP3042	RJ	Rio De Janeiro	Avenida Embaixador Abelardo Bueno, 3401	YES
66	SPPGE5270	SP	Praia Grande	AV AYRTON SENNA DA SILVA, 1500	YES
67	PBMAN4590	PB	Joao Pessoa	Avenida Gov Flavio Ribeiro Coutinho 805/TO	YES
69	SPWST4066	SP	São Paulo	Av. Dr. Chucri Zaidan, 902	YES
70	SPMAU2861	SP	Mauá	Av. Governador Mario Covas Junior, 01 - Mauá Plaza Shopping	YES
71	SPJAI2858	SP	Jundiaí	Av. Antônio Frederico Ozanan, 6000 - Maxi Shopping Jundiaí	YES
72	SRAMMNS3304	AM	Manaus	Avenida Djalma Batista, 1661 - Millennium Shopping	YES
73	SRMGUNI4194	MG	Belo Horizonte	Avenida Cristiano Machado, 4000 - Minas Shopping	YES
75	SPSTS2871	SP	Santos	Av. Euclides da Cunha, 21 - Miramar Shopping	YES
76	SPMCZ2862	SP	Mogi Das Cruzes	Av. Ver. Narciso Yague Guimarães, 1001 - Mogi Shopping	YES
77	SPWST4065	SP	São Paulo	AV ROQUE PETRONI JUNIOR, 1089	YES
78	RNCAN6201	RN	Natal	Avenida Senador Salgado Filho 2234	YES
79	SRCEFLA0297	CE	Fortaleza	Av. Bezerra de Menezes, no. 2450 (North Shopping)	YES
80	SPRPO2866	SP	Ribeirão Preto	Av. Presidente Kennedy, 1500 - Novo Shopping	YES
81	SPOCO0087	SP	Osasco	Rua Tenente Avelar Pires de Azevedo, 81 - Osasco Plaza Shopping	YES
82	PERCE3244	PE	Recife	Rua Cais da Alfadênga S/N	YES
84	MTCBA0111	MT	Cuiabá	AV. RUBENS DE MENDONÇA	YES
85	SRRJCGR3022	RJ	Rio De Janeiro	Estrada do Monteiro, 1200 - Park Shopping Campo Grande	YES
86	SPCAS4832 / SPCAS2673	SP	Campinas	Av. Guilherme Campos, 500 - Parque Dom Pedro Shopping	YES
87	PAICO2253	PA	Belém	Rodovia Augusto Montenegro, 4300 (Parque Shopping Belém)	YES
88	SPPPE2851	SP	Presidente Prudente	Rua Siqueira Campos, 1545 - Parque Shopping Prudente	YES

				1	
89	ALMCO2399	AL	Maceió	AV MENINO MARCELO, 3800 TABULEIRO MARTINS	YES
90	SRBAIGW0208	BA	Salvador	AVENIDA ANTONIO CARLOS MAGALHAES, 1034, PITUBA PARQUE CENTER	YES
92	SPSRR2840	SP	São Jose Do Rio Preto	Av. José Munia, 4775 - Plaza Avenida Shopping	YES
93	SPITU2855	SP	Itu	Av. Dr. Ermelindo Maffei, 1199 - Plaza Shopping Itu	YES
94	SPSRR2875	SP	São Jose Do Rio Preto	R. Bernardino de Campos, 2940 - Praça Shopping	YES
95	SPSTS2872	SP	Santos	R. Alexandre Martins, 80 - Praiamar Shopping	YES
98	SPPPE2865	SP	Presidente Prudente	Av. Manoel Goulart, 2400 - Prudenshopping	YES
99	SPWVS2253	SP	São Paulo	Rod. Raposo Tavares, Km 14,5 - Raposo Shopping (Antigo Shopping Off Price Raposo)	YES
100	SRRJJBT3041	RJ	Rio De Janeiro	Rua Jardim Botânico, 266 - Rede Globo	YES
101	SPRPO5314	SP	Ribeirão Preto	Av. Cel. Fernando Ferreira Leite, 1540 - Ribeirão Shopping	YES
103	SRRJBOT3021	RJ	Rio De Janeiro	Rua General Severiano, 97 - Rio Plaza Shopping	YES
104	SPSRR2876	SP	São Jose Do Rio Preto	Av. Brigadeiro Faria Lima, 6363 - Riopreto Shopping Center	YES
106	SRMGSAV4363	MG	Belo Horizonte	Rua Alagoas, 1314 - Shopping 5 <sup>a</sup> Avenida	YES
107	SPSNE4571	SP	Santo André	Av. Pereira Barreto, 42 - Shopping ABC (Antigo Mappin ABC)	YES
109	SRCEFLA0299	CE	Fortaleza	Av. Dom Luis, no. 500 (Shopping Aldeota)	YES
110	MGNLA4722	MG	Nova Lima	Rua Senador Milton Campos, 115 - (Shopping Alta Vila)	YES
111	SPWVF4169	SP	São Paulo	Av. Regente Feijó, 1739	YES
112	SRRJCPS3057	RJ	Campos Dos Goytacazes	Av. 28 de Março, 574 - Shopping Avenida 28	YES
113	SRBAPOK0201	BA	Salvador	AVENIDA CENTENARIO, 2992	YES
114	PEBVT3241	PE	Recife	Rua do Giriquiti 48	YES
116	SRGOGNA0304	GO	Goiânia	Rua 9,1855	YES

115	Campos Dos Rua Doutor Silvio Bastos Tavares, 330 - Shopping		LID0		
117	SRRJCPS3058	RJ	Goytacazes	Boulevard Campos	YES
118	SRRJSCR3028	RJ	Rio De Janeiro	Rua Barão de São Francisco, 236 - Shopping Boulevard RioShopping antigo Iguatemi	YES
119	ESVVA3847	ES	Vila Velha	Rodovia do Sol, Km 5000 (Shopping Boulevard Vila Velha)	YES
120	SPWVS4237	SP	São Paulo	Av. Prof. Francisco Morato, 2718 - Shopping Butantã	YES
121	SPWCL0085	SP	São Paulo	Estrada do Campo Limpo, 45 - Shopping Campo Limpo	YES
122	RSCAN0958	RS	Canoas	AV GUILHERME SCHELL 6750/ TE	YES
123	PABLM2198	PA	Belém	Rod. BR - 316, Km 1, s/n (Shopping Castanheira)	YES
124	SPWCO2634	SP	São Paulo	Av. Paulista, 2064 - Shopping Center 3	YES
125	SPFND1291	SP	Fernandópolis	Av. Litério Grecco, 600 - Shopping Center Fernandópolis	YES
126	SPWLP2883	SP	São Paulo	R. Guaicurus, esquina com a Rua Catão - Shopping Center Lapa	YES
127	SPWVG4173	SP	São Paulo	TV CASALBUONO, 120	YES
128	SPWPH3954	SP	São Paulo	R. Doutor João Ribeiro, 304 - Shopping Center Penha	YES
129	SPRCO2868	SP	Rio Claro	Av. Conde Francisco Matarazzo, 205 - Shopping Center Rio Claro	YES
131	PRIND0399	PR	Curitiba	AVENIDA MARECHAL FLORIANO PEIXOTO, 4880	YES
132	SPSJC5482	SP	São Jose Dos Campos	Av. São João, 2200 - Shopping Colinas	YES
133	SPWPI3937	SP	São Paulo	AV. CRUZEIRO DO SUL, 1100 - SHOPPING D	YES
134	SPWIB3783	SP	São Paulo	Av. das Nações Unidas, 12555 - Shopping D&D	YES
135	SRRJGAV3036	RJ	Rio De Janeiro	Rua Marquês de São Vicente, 52 - Shopping da Gávea	YES
136	SRMASLS1686	MA	São Luís	Av. Daniel de La Touche, 987	YES
137	SRMGCAI4197	MG	Belo Horizonte	Avenida Presidente Carlos Luz, 3001 - Shopping Del Rey	YES
138	SPWPN3976	SP	São Paulo	Av. Rebouças, 3970	YES
139	SPWBV3590	SP	São Paulo	R. Frei Caneca, 569 - Shopping Frei Caneca	YES
140	SPCOA0258	SP	Cotia	Rodovia Raposo Tavares, km 23	YES

141	PEJBO3326	PE	Jaboatão Dos Guararapes	Avenida Barreto de Menezes 800	YES
142	SPWMM3912 / SPWMM2797	SP	São Paulo	Av. Ibirapuera, 3103 - Shopping Ibirapuera	YES
143	SRCEFLA0298	CE	Fortaleza	Av. Washington Soares, no. 85 (Shopping Iguatemi)	YES
144	SPWPN0266	SP	São Paulo	Rua Angelina Maffei Vita, s/n° - Shopping Iguatemi - Expansão Fase III	YES
145	SPWPN3977	SP	São Paulo	Av. Brig. Faria Lima, 2232	YES
146	RSIND0529	RS	Caxias Do Sul	RST 453 2787 SN	YES
147	SRBABDA0202	BA	Salvador	AVENIDA TANCREDO NEVES, 42-152, SHOPPING IGUATEMI SALVADOR	YES
148	SPWCG3654	SP	São Paulo	AV. INTERLAGOS, 2255 - Shopping Interlagos	YES
149	SCIND0277	SC	São Jose	RUA GERONCIO THIVES	YES
151	SPARQ2842	SP	Araraquara	Acesso Engenheiro Heitor de Souza Pinheiro, 2270 - Shopping Jaraguá	YES
152	SPWVA4154	SP	São Paulo	Av. Giovanni Gronchi, 5819 - Shopping Jardim Sul	YES
153	SPGJA2852	SP	Guarujá	Av. Marechal Deodoro da Fonseca, 885	YES
154	SPWVG2884	SP	São Paulo	Av. Otto Baumgart, 500 - Shopping Lar Center	YES
155	SPWAC2885	SP	São Paulo	Av. Aricanduva, 5555 - Shopping Leste Aricanduva	YES
156	SPWRE2886	SP	São Paulo	R. Coronel Xavier de Toledo, 23 - Shopping Light	YES
157	SPWBS0086	SP	São Paulo	Rua Barão de Ladário, 566/670 - Shopping Mega Polo Moda	YES
158	ESSEA3846	ES	Serra	Av. João Palácios, 300 (SHOPPING MESTRE ALVARO)	YES
159	SPWTA2887	SP	São Paulo	R. Gonçalves Crespo, 78 - Shopping Metrô Boulevard Tatuapé	YES
160	SPWIT2888	SP	São Paulo	Av. José Pinheiro Borges, s/n - Shopping Metrô Itaquera	YES
162	SPWTA4134	SP	São Paulo	R. Domingos Agostim, 91 - Shopping Metrô Tatuapé	YES
163	SPSBO2624	SP	São Bernardo Do Campo	Praça Samuel Sabatini, 200 - Shopping Metrópole	YES

165	RNLSA6033	RN	Natal	Avenida Bernardo Vieira 3775.Shopping Midway mall.	YES
166	SRRJDCT3037	RJ	Rio De Janeiro	Avenida Pastor Martin Luther King Jr, 126	YES
168	PABLM2199	PA	Belém	Trav. Padre Eutíquio, no. 1078 (Shopping Pátio Belém)	YES
171	SPWBV3936	SP	São Paulo	R. Treze de Maio, 1947 - Shopping Pátio Paulista (Antigo Shopping Paulista)	YES
172	SRMGSAV4204	MG	Belo Horizonte	Avenida do Contorno, 6061 - Shopping Pátio Savassi	YES
173	SRBAASG0203	BA	Salvador	RUA CONS. JUNQUEIRA AYRES, 165/TO	YES
174	SPPAA2864	SP	Piracicaba	Av. Limeira, 722 - Shopping Piracicaba	YES
176	RJMCE2672	RJ	Macaé	Avenida Aluízio da Silva Gomes, 800 - Shopping Plaza Macaé	YES
177	SPWSU4114	SP	São Paulo	Praça Leonor Kaupa, 100 - Shopping Plaza Sul	YES
178	SRRJMAD3044	RJ	Rio De Janeiro	Estrada da Portela, 99 - Shopping Polo1	YES
181	PEBVG3243	PE	Recife	Rua Padre Carapuceiro 777	YES
182	SRMASLS1547	MA	São Luís	Av. São Luís Rei de França, no 08, bairro Turú. Shopping Rio Anil.	YES
184	SPRPO2867	SP	Ribeirão Preto	R. São José, 933 - Shopping Santa Úrsula	YES
185	RJSGO2908	RJ	São Gonçalo	Rua Nova Jersey, s/n - Lote 1 - Shopping São Gonçalo	YES
186	MGSLA2096	MG	Sete Lagoas	Rua Otávio Campelo Ribeiro, 2801 (Shopping Sete Lagoas)	YES
188	SPTBS4662	SP	Taboão Da Serra	Rod. Régis Bittencourt, Km 271,5 - Shopping Taboão	YES
189	PESAR3429	PE	Recife	Avenida Agamenon Magalhaes 153	YES
191	SRRJRCP3029	RJ	Rio De Janeiro	Avenida Maracanã, 987 - Shopping Tijuca	YES
192	PRIND0398	PR	Curitiba	RUA ITACOLOMI, 100	YES
193	PAPUP2033	PA	Parauapebas	Rodovia PA 275, Km 55 - Shopping Unique	YES
194	MGIIG4202	MG	Ipatinga	Rodovia BR-381 (Avenida Pedro Linhares Filho), Km 206 - Industrial - Shopping Vale do Aço	YES
195	SPVOS1097	SP	Valinhos	Rua Paiquerê, 200 - Shopping Valinhos	YES
196	SRRJBTJ3048	RJ	Rio De Janeiro	Avenida Ayrton Senna, 3000 - Shopping Via Parque	YES
197	SRRJBTJ3023	RJ	Rio De Janeiro	Avenida das Américas, 3900	YES

198	SPWAP3569	SP	São Paulo	Av. das Nações Unidas, 4777 - Shopping Villa-Lobos	YES
199	ESVTA3811	ES	Vitoria	Avenida Nossa Senhora dos Navegantes, 1440 - Shopping Vitória	YES
200	SPWBF3584	SP	São Paulo	Av. Antártica, 380 - Shopping West Plaza	YES
201	RJVRD2280	RJ	Volta Redonda	Rua Brás Magaldi Fernandes, 300	YES
202	SPSOC2892	SP	Sorocaba	Av. Afonso Vergueiro, 1700/1766 - Sorocaba Shopping	YES
203	SPOCO2640	SP	Osasco	Av. dos Autonomistas, 1828 - Super Shopping Osasco	YES
204	PRCTA0576	PR	Curitiba	Supermercado Angeloni - Avenida República Argentina,	YES
205	SPSZN2893	SP	Suzano	R. Sete de Setembro, 555 - Suzano Shopping	YES
206	DFIND0131	DF	Brasília	Rua 210, QS 1, Lt. 40	YES
207	SPTTE2894	SP	Taubate	Av. Charles Schneider, 1700 - Taubaté Shopping	YES
209	SPSTB2869	SP	Santa Barbara Doeste	Av. Santa Bárbara, 777 - Tívoli Shopping	YES
210	SRRJBOT3026	RJ	Rio De Janeiro	Rua Lauro Muller, 116 - Torre do Rio Sul - Complexo do RioSul (Shopping)	YES
211	SPWST2854	SP	São Paulo	Avenida Dr. Mário Villas Boas Rodrigues, 387 - Transamérica Expo Center	YES
217	DFIND0144	DF	Brasília	CPU DARCY RIBEIRO, 0 - ASA NORTE	YES
218	SPSJC2877	SP	São Jose Dos Campos	Av. Andrômeda, 227 - Vale Sul Shopping	YES
219	SRRJIRJ3062	RJ	Rio De Janeiro	Rua Itapera, Trevo das Margaridas - Via Brasil Shopping	YES
220	SRCEFLA0920	CE	Fortaleza	Av. Washington Soares, 4335	YES
221	SPWIB3785	SP	São Paulo	Av. das Nações Unidas, 12551 - World Trade Center São Paulo	YES
222	DFIND0138	DF	Brasília	PÇA TRES PODERESLOTE ÚNICO, 0 - ZONA CÍVICO ADMINIST	YES

#### SHARE PURCHASE AGREEMENT VIA ISOLATED PRODUCTION UNIT AND OTHER COVENANTS

#### EXHIBIT 4.4.6(i)

Final Purchase Price Adjustment - Substantial Competitive Restrictions

The Final Purchase Price after the Competitive Restrictions shall be equal to the Final Purchase Price (i) <u>minus</u> the Annual Indoor Revenue Price of those Indoor Sites that have to be removed from the scope of the transaction due to the Competitive Restrictions and (ii) <u>minus</u> the Annual Outdoor Revenue Price of those Outdoor Sites that have to be removed from the scope of the transaction due to Competitive Restrictions.

#### SHARE PURCHASE AGREEMENT VIA ISOLATED PRODUCTION UNIT AND OTHER COVENANTS

#### EXHIBIT 5.2.1(V)

**Closing Instrument** 

### CLOSING INSTRUMENT OF THE SHARE PURCHASE AGREEMENT VIA ISOLATED PRODUCTION UNIT AND OTHER COVENANTS

This Closing Instrument of the Share Purchase Agreement via Isolated Production Unit and Other Covenants ("<u>Closing Instrument</u>") is entered into by and between, on one side,

1. **OI MÓVEL S.A. - UNDER JUDICIAL REORGANIZATION**, a corporation with its head offices in Brasília, Distrito Federal, at Setor Comercial Norte, Quadra 03, Bloco A, Edifício Estação Telefônica, Térreo, Parte 2, ZIP CODE 70.713-900, registered with the National Register of Legal Entities of the Ministry of Economy ("CNPJ/ME") under No. 05.423.963/0001-11, herein represented pursuant to its Bylaws (hereinafter referred to as "<u>Oi Móvel</u>");

2. **TELEMAR NORTE LESTE S.A. - UNDER JUDICIAL REORGANIZATION**, a company with its head offices in Rio de Janeiro, State of Rio de Janeiro, at Rua do Lavradio, n° 71, 2° andar, Centro, ZIP CODE 20.230-070, registered with the CNPJ/ME under No. 33.000.118/0001-79 and in the Commercial Registry of the State of Rio de Janeiro under NIRE 33.30015258 0, herein represented pursuant to its Bylaws (hereinafter referred to as "<u>TMAR</u>" and, together with Oi Móvel, "<u>Sellers</u>");

and, on the other side,

3. [**BUYER**], with its head offices at [=], registered with the CNPJ/ME under No. [=] (hereinafter referred to as "<u>Buyer</u>");

With Sellers and Buyer being jointly referred to as "Parties" and individually and indistinctly as "Party".

And, as intervening consenting party,

4. CALITEIA RJ INFRAESTRUTURA E REDES DE TELECOMUNICAÇÕES S.A., with its head offices in Rio de Janeiro, State of Rio de Janeiro, at Rua do Lavradio, n° 71, sl. 201/801, Centro, ZIP CODE 20.230-070, registered with the CNPJ/ME under No. 35.978.982/0001-75, herein represented pursuant to its Bylaws (hereinafter referred to as "Company");

and also as intervening consenting guarantor party,

5. **OI S.A. - UNDER JUDICIAL REORGANIZATION**, a publicly-held company, with its head offices at Rua do Lavradio n.º 71, Centro, in the City and State of Rio de Janeiro, ZIP CODE 20230-070, registered with the CNPJ/ME under No. 76.535.764/0001-43 (hereinafter referred to as "<u>Oi</u>" and together with the Company, the Intervening Consenting Sellers.

#### PREAMBLE

**WHEREAS**, on [=], the Parties and Intervening Consenting Sellers entered into the Share Purchase Agreement via Isolated Production Unit and Other Covenants, whereby, among other provisions, Buyer undertook to acquire all the shares issued by the Company, representing the totality of the Company's share capital ("<u>Agreement</u>"); and

**WHEREAS**, the Conditions Precedent for the Closing provided for in Clause 4 of the Agreement have been implemented and verified (or waived, as the case may be) until and/or on this date, so that the Parties wish to proceed with the Closing and consummate the legal transactions provided for in the Agreement to perform the Closing;

**NOW THEREFORE,** the Parties enter into this Closing Instrument, which shall be governed by the following terms and conditions that the parties mutually agree, as follows:

#### CLAUSE 1 DEFINED TERMS

**Clause 1.1.** <u>Defined Terms</u>. The terms used in this Closing Instrument in capital letters and not expressly defined herein, both in the singular and in the plural, and the verbal and nominal variations thereof, shall have the meaning ascribed to them in the Agreement.

#### CLAUSE 2 CONDITIONS PRECEDENT FOR THE CLOSING

**Clause 2.1.** <u>Compliance with the Conditions Precedent for the Closing</u>. The Parties represent and warrant (for themselves and, in the case of Sellers, also for the Company) that the Conditions Precedent for the Closing provided for in Clause 4 of the Agreement, have been implemented, complied with and verified (or waived, as the case may be), as applicable, prior to and/or on this Closing Date, as follows:

2.1.1 Conditions Precedent of the Parties. The Parties represent and warrant (for themselves and, in the case of Sellers, also for the Company) that the Conditions Precedent of the Parties provided for in Clause 4.1 of the Agreement, have been implemented, complied with and verified prior to and/or on this Closing Date, as provided for therein.

2.1.2 Conditions Precedent of Buyer. Sellers represent and warrant (for themselves and also for the Company) that the Conditions Precedent of Buyer provided for in Clause 4.2 of the Agreement, have been implemented, complied with and verified prior to and/or on this Closing Date, as provided for therein. Without prejudice to the foregoing, Sellers represent and warrant (for themselves and also for the Company) to Buyer that, on this Closing Date:

(i) all representations and warranties made by Sellers in the Agreement are true, complete and exact, both on the date of execution of this Agreement and on the Closing Date [(considering the representations updated in accordance with the provisions of Clause 14.4)]<sup>1</sup>;

(ii) Sellers and the Intervening Consenting Sellers are not in breach of the obligations undertaken by them under the Agreement, including those stipulated in Clause 8.1 of the Agreement; and

(iii) any event that constitutes a Material Adverse Effect has not occurred from the date of execution of the Agreement to this Closing Date;

2.1.3 Conditions Precedent of Sellers. Buyer represents and warrants that the Conditions Precedent of Sellers provided for in Clause 4.3 of the Agreement, have been implemented, complied with and verified prior to and/or on this Closing Date, as provided for therein. Without prejudice to the foregoing, Buyer represents and warrants to Sellers that, on this Closing Date:

(i) all representations and warranties provided by Buyer in the Agreement contained in Clause 7 are true, complete and exact, both on the date of execution of the Agreement and on this Closing Date; and

(ii) on this Closing Date, also, Buyer are not in breach of the obligations undertaken thereby in the Agreement.

**Clause 2.2.** <u>Closing Date</u>. The Parties, by mutual agreement, determine that the Closing is performed on the date hereof, and the date hereof is considered the Closing Date for all purposes and effects of the Agreement.

**Clause 2.3.** <u>Obligations at Closing</u>. As determined in Clause 5.2 of the Agreement, the Parties have complied with (and Sellers has caused the Company to comply with), on the date hereof, all the acts and procedures provided for in the Agreement for implementation of the Closing, including the acts and obligations contained in Clause 5.2 of the Agreement.

<sup>&</sup>lt;sup>1</sup> Note: to be filled in if applicable upon the execution of the Closing Instrument.

**Clause 2.4.** <u>Simultaneous Acts</u>. All acts and obligations indicated in Clause 5.2 of the Agreement and Clause 2.3 of this Closing Instrument are considered simultaneous.

**Clause 2.5.** <u>Representations and Warranties of Seller Error! Bookmark not defined</u>.. Without prejudice to the provisions of the Agreement, Sellers represent and warrant to Buyer that all documents and information provided by Sellers (and, as the case may be, by the Company) to Buyer, for (i) proof of compliance with the Conditions Precedent for the Closing referred in Clauses 2.1.1 and 2.1.2 of this Closing Instrument, and (ii) implementation and practice of the acts and procedures referred in Clause 2.3 of this Closing Instrument that are the initiative or responsibility of Sellers, are true, exact, updated and complete on the date hereof.</u>

**Clause 2.6.** <u>Representations and Warranties of Buyer</u>. Without prejudice to the provisions of the Agreement, Buyer represents and warrants to Sellers that all documents and information provided by Buyer to Sellers, for (i) proof of compliance with the Conditions Precedent for the Closing referred in Clauses 2.1.1 and 2.1.3 of this Closing Instrument, and (ii) implementation and practice of the acts and procedures referred in Clause 2.3 of this Closing Instrument that are the initiative or responsibility of Buyer, are true, exact, updated and complete on the date hereof.

### CLAUSE 3 MISCELLANEOUS

**Clause 3.1.** <u>Agreement Ratification</u>. Except as expressly provided for herein, this Closing Instrument does not replace the Agreement and is not intended to deprive, waive or mitigate the rights and obligations set forth therein, and all other clauses of the Agreement shall be expressly ratified. This Closing Instrument shall become an integral part of the Agreement, for all purposes, together with all its exhibits.

**Clause 3.2.** <u>No Waiver</u>. The execution by the Parties of this Closing Instrument does not constitute discharge or waiver of any of the Parties in relation to any breach or default of the Agreement that may have occurred before the date hereof, except as waived by any of the Parties, under the Agreement, if applicable.

**Clause 3.3.** <u>Confidentiality</u>. The confidentiality provisions provided for in Clause 14.12 (Confidentiality and Non-Disclosure) of the Agreement shall apply to this Closing Instrument, *mutatis mutandis*.

**Clause 3.4.** <u>Specific Performance</u>. Any of the Parties may seek the specific performance of the obligations contained in this Closing Instrument, under the terms and as provided for in Clause 14.21 (Specific Performance; Enforceable Instrument) of the Agreement.

**Clause 3.5.** <u>Applicable Law</u>. This Closing Instrument shall be governed by and interpreted pursuant to the Laws of the Federative Republic of Brazil.

**Clause 3.6.** <u>Conflict Resolution</u>: Any and all controversies or disputes related to or arising from this Closing Instrument shall be resolved in accordance with Clause 13 (Conflict Resolution) of the Agreement.

**Clause 3.7.** <u>Miscellaneous</u> The general provisions in Clause 14 of the Agreement (Miscellaneous) and its sub-clauses are, *mutatis mutandis*, incorporated into this Closing Instrument by reference, and shall apply to this Closing Instrument as if they were expressly transcribed herein.

IN WITNESS WHEREOF, the Parties and Intervening Consenting Sellers executed this Closing Instrument in two (2) counterparts of equal context and for one sole purpose, in the presence of the two (2) witnesses below.

São Paulo, [month] [day], 2020.

(Remainder of the page intentionally left blank)

[Signature page of the Closing Instrument of the Share Purchase Agreement via Isolated Production Unit and Other Covenants, entered into between Oi Móvel S.A. — Under Judicial Reorganization and [Buyer], and, as intervening consenting party, [SPE Towers], and further, as intervening consenting guarantor parties, Telemar Norte Leste S.A — Under Judicial Reorganization and Oi SA. — Under Judicial Reorganization, dated [=] [=], 2020]

# OI MÓVEL S.A. - UNDER JUDICIAL REORGANIZATION

By: [=] Title: [=]

By: [=] Title: [=]

# TELEMAR NORTE LESTE S.A. – UNDER JUDICIAL REORGANIZATION

By: [=] Title: [=] By: [=] Title: [=]

By: [=]

Title: [=]

# [BUYER]

By: [=] Title: [=]

# CALITEIA RJ INFRAESTRUTURA E REDES DE TELECOMUNICAÇÕES S.A.

By: [=] Title: [=] By: [=] Title: [=]

### **OI S.A. – UNDER JUDICIAL REORGANIZATION**

Witnesses:

Name: ID (RG): Individual Taxpayer's Register of the Ministry of Economy (CPF/ME): Name: ID (RG): Individual Taxpayer's Register of the Ministry of Economy (CPF/ME):

\*\*\*

#### SHARE PURCHASE AGREEMENT VIA ISOLATED PRODUCTION UNIT AND OTHER COVENANTS

EXHIBIT 5.2.1(VI)

Oi Sharing Agreement

#### SITE SHARING AGREEMENT

between

CALITEIA RJ INFRAESTRUTURA E REDES DE TELECOMUNICAÇÕES S.A. As Contractor

> OI MÓVEL S.A. - UNDER JUDICIAL REORGANIZATION as Principal

**OI S.A. – UNDER JUDICIAL REORGANIZATION** 

as Intervening Guarantor Party:

[=] [=], 2020

#### **TABLE OF CONTENTS**

		PAGE
CLAUSE 1	DEFINITIONS	5
Clause 1.1	Definitions	5
Clause 1.2	Other Definitions	8
CLAUSE 2	PURPOSE	8
	Space Lease	8
Clause 2.2	Consideration	9
Clause 2.3	Scope of the Space Lease	9
CLAUSE 3	EXHIBITS	9

Clause 3.1	Exhibits	9				
CLAUSE 4	SPACE LEASE	9				
Clause 4.1	Right of First Refusal of Excess Space	9				
Clause 4.2	Request Forms	10				
Clause 4.3	Reforms and Adjustments					
Clause 4.4	Right of Retention					
Clause 4.5	Lease Agreement	11				
CLAUSE 5	COMMON OBLIGATIONS OF THE PARTIES	11				
Clause 5.1	Common Obligations	11				
CLAUSE 6	CONTRACTOR'S OBLIGATIONS	11				
Clause 6.1	Contractor's Obligations	11				
CLAUSE 7	PRINCIPAL'S OBLIGATIONS	12				
Clause 7.1	Principal's Obligations	12				
Clause 7.2	Irregular Works, Installations and Services	14				
Clause 7.3	Indemnification of the Sale and Purchase Agreement	15				
CLAUSE 8	PAYMENT AND ADJUSTMENT	15				
Clause 8.1	Considerations	15				
•	nent of the Considerations	16				
	Tax Liability	16				
	Additional Requests	16				
	Guaranteed Consideration	17				
CLAUSE 9		17				
	Infrastructure Billing Statement	17				
	Payment Date	17				
CLAUSE 10	PENALTIES	18				
Clause 10.1	Default by Principal	18				
Clause 10.2	Exclusive Fault - Contractor	18				
CLAUSE 11	TOTAL OR PARTIAL TERMINATION OF THE	18				
CLAUSE 11	AGREEMENT					
Clause 11.1	AGREEMENT Termination	18				
Clause 11.1 CLAUSE 12	AGREEMENT Termination <u>MISCELLANEOUS</u>	18 <b>20</b>				
Clause 11.1 CLAUSE 12 Clause 12.1	AGREEMENT Termination <u>MISCELLANEOUS</u> Entire Agreement	18 <b>20</b> 20				
Clause 11.1 CLAUSE 12 Clause 12.1 Clause 12.2	AGREEMENT Termination <u>MISCELLANEOUS</u> Entire Agreement Entire Agreement	18 <b>20</b> 20 20				
Clause 11.1 CLAUSE 12 Clause 12.1 Clause 12.2 Clause 12.3	AGREEMENT Termination <u>MISCELLANEOUS</u> Entire Agreement Entire Agreement Damage	18 20 20 20 20				
Clause 11.1 CLAUSE 12 Clause 12.1 Clause 12.2 Clause 12.3 Clause 12.4	AGREEMENT Termination MISCELLANEOUS Entire Agreement Entire Agreement Damage Acts of God and Force Majeure	18 20 20 20 20 20				
Clause 11.1 CLAUSE 12 Clause 12.1 Clause 12.2 Clause 12.3 Clause 12.4 Clause 12.5	AGREEMENT Termination MISCELLANEOUS Entire Agreement Entire Agreement Damage Acts of God and Force Majeure Employer Liability	18 20 20 20 20 20 20 20				
Clause 11.1 CLAUSE 12 Clause 12.1 Clause 12.2 Clause 12.3 Clause 12.4 Clause 12.5 Clause 12.6	AGREEMENT Termination MISCELLANEOUS Entire Agreement Entire Agreement Damage Acts of God and Force Majeure Employer Liability Access Procedure	18 20 20 20 20 20 20 20 21				
Clause 11.1 CLAUSE 12 Clause 12.1 Clause 12.2 Clause 12.3 Clause 12.4 Clause 12.5 Clause 12.6 Clause 12.7	AGREEMENT Termination MISCELLANEOUS Entire Agreement Entire Agreement Damage Acts of God and Force Majeure Employer Liability Access Procedure Planning and Execution	18 20 20 20 20 20 20 20 21 21				
Clause 11.1 CLAUSE 12 Clause 12.1 Clause 12.2 Clause 12.3 Clause 12.4 Clause 12.5 Clause 12.6 Clause 12.7 Clause 12.8	AGREEMENT Termination MISCELLANEOUS Entire Agreement Entire Agreement Damage Acts of God and Force Majeure Employer Liability Access Procedure Planning and Execution Repairs	18 20 20 20 20 20 20 21 21 21				
Clause 11.1 CLAUSE 12 Clause 12.1 Clause 12.2 Clause 12.3 Clause 12.4 Clause 12.5 Clause 12.6 Clause 12.7 Clause 12.8 Clause 12.9	AGREEMENT Termination MISCELLANEOUS Entire Agreement Entire Agreement Damage Acts of God and Force Majeure Employer Liability Access Procedure Planning and Execution Repairs Notices	18 20 20 20 20 20 20 21 21 21 21				
Clause 11.1 CLAUSE 12 Clause 12.1 Clause 12.2 Clause 12.3 Clause 12.4 Clause 12.5 Clause 12.6 Clause 12.7 Clause 12.8 Clause 12.9 Clause 12.10	AGREEMENT Termination MISCELLANEOUS Entire Agreement Entire Agreement Damage Acts of God and Force Majeure Employer Liability Access Procedure Planning and Execution Repairs Notices Other Contracts	18 20 20 20 20 20 21 21 21 21 21 21				
Clause 11.1 <b>CLAUSE 12</b> Clause 12.1 Clause 12.2 Clause 12.3 Clause 12.4 Clause 12.5 Clause 12.6 Clause 12.7 Clause 12.8 Clause 12.9 Clause 12.10 Clause 12.11	AGREEMENT Termination MISCELLANEOUS Entire Agreement Entire Agreement Damage Acts of God and Force Majeure Employer Liability Access Procedure Planning and Execution Repairs Notices Other Contracts	18 20 20 20 20 20 20 21 21 21 21 21 21				
Clause 11.1 <b>CLAUSE 12</b> Clause 12.1 Clause 12.2 Clause 12.3 Clause 12.4 Clause 12.5 Clause 12.5 Clause 12.7 Clause 12.8 Clause 12.9 Clause 12.10 Clause 12.11 Clause 12.13	AGREEMENT Termination MISCELLANEOUS Entire Agreement Entire Agreement Damage Acts of God and Force Majeure Employer Liability Access Procedure Planning and Execution Repairs Notices Other Contracts Liability Atypical Relationship	18 20 20 20 20 20 20 21 21 21 21 21 21 21				
Clause 11.1 <b>CLAUSE 12</b> Clause 12.1 Clause 12.2 Clause 12.3 Clause 12.4 Clause 12.5 Clause 12.6 Clause 12.7 Clause 12.7 Clause 12.8 Clause 12.9 Clause 12.10 Clause 12.11 Clause 12.13 Clause 12.14	AGREEMENT Termination MISCELLANEOUS Entire Agreement Entire Agreement Damage Acts of God and Force Majeure Employer Liability Access Procedure Planning and Execution Repairs Notices Other Contracts Liability Atypical Relationship Guarantee	18 20 20 20 20 20 21 21 21 21 21 21 21 22				
Clause 11.1 <b>CLAUSE 12</b> Clause 12.1 Clause 12.2 Clause 12.3 Clause 12.4 Clause 12.5 Clause 12.5 Clause 12.6 Clause 12.7 Clause 12.8 Clause 12.9 Clause 12.10 Clause 12.11 Clause 12.13 Clause 12.14 <b>CLAUSE 13</b>	AGREEMENT Termination MISCELLANEOUS Entire Agreement Entire Agreement Damage Acts of God and Force Majeure Employer Liability Access Procedure Planning and Execution Repairs Notices Other Contracts Liability Atypical Relationship Guarantee ASSIGNMENT, TRANSFER AND SUBROGATION	18 20 20 20 20 20 21 21 21 21 21 21 21 22 22				
Clause 11.1 <b>CLAUSE 12</b> Clause 12.1 Clause 12.2 Clause 12.3 Clause 12.4 Clause 12.5 Clause 12.5 Clause 12.7 Clause 12.7 Clause 12.8 Clause 12.9 Clause 12.10 Clause 12.11 Clause 12.13 Clause 12.14 <b>CLAUSE 13</b> Clause 13.1	AGREEMENT Termination MISCELLANEOUS Entire Agreement Entire Agreement Damage Acts of God and Force Majeure Employer Liability Access Procedure Planning and Execution Repairs Notices Other Contracts Liability Atypical Relationship Guarantee ASSIGNMENT, TRANSFER AND SUBROGATION Assignment, Transfer and Subrogation	18 20 20 20 20 20 21 21 21 21 21 21 21 21 22 22 22				
Clause 11.1 <b>CLAUSE 12</b> Clause 12.1 Clause 12.2 Clause 12.3 Clause 12.4 Clause 12.5 Clause 12.6 Clause 12.7 Clause 12.7 Clause 12.8 Clause 12.9 Clause 12.10 Clause 12.11 Clause 12.13 Clause 12.14 <b>CLAUSE 13</b> CLAUSE 14	AGREEMENT Termination MISCELLANEOUS Entire Agreement Entire Agreement Damage Acts of God and Force Majeure Employer Liability Access Procedure Planning and Execution Repairs Notices Other Contracts Liability Atypical Relationship Guarantee ASSIGNMENT, TRANSFER AND SUBROGATION Assignment, Transfer and Subrogation ASSIGNMENT OF CREDIT RIGHTS	18 20 20 20 20 20 21 21 21 21 21 21 21 22 22 22 23				
Clause 11.1 <b>CLAUSE 12</b> Clause 12.1 Clause 12.2 Clause 12.3 Clause 12.4 Clause 12.5 Clause 12.6 Clause 12.7 Clause 12.7 Clause 12.8 Clause 12.9 Clause 12.10 Clause 12.11 Clause 12.13 Clause 12.14 <b>CLAUSE 13</b> Clause 13.1 <b>CLAUSE 14</b> Clause 14.1	AGREEMENT Termination MISCELLANEOUS Entire Agreement Entire Agreement Damage Acts of God and Force Majeure Employer Liability Access Procedure Planning and Execution Repairs Notices Other Contracts Liability Atypical Relationship Guarantee ASSIGNMENT, TRANSFER AND SUBROGATION Assignment, Transfer and Subrogation ASSIGNMENT OF CREDIT RIGHTS Assignment of Credit	18 20 20 20 20 20 21 21 21 21 21 21 21 21 22 22 22 23 23				
Clause 11.1 <b>CLAUSE 12</b> Clause 12.1 Clause 12.2 Clause 12.3 Clause 12.4 Clause 12.5 Clause 12.6 Clause 12.7 Clause 12.7 Clause 12.8 Clause 12.9 Clause 12.10 Clause 12.11 Clause 12.13 Clause 12.14 <b>CLAUSE 13</b> CLAUSE 14	AGREEMENT Termination MISCELLANEOUS Entire Agreement Entire Agreement Damage Acts of God and Force Majeure Employer Liability Access Procedure Planning and Execution Repairs Notices Other Contracts Liability Atypical Relationship Guarantee ASSIGNMENT, TRANSFER AND SUBROGATION Assignment, Transfer and Subrogation ASSIGNMENT OF CREDIT RIGHTS Assignment of Credit CONFIDENTIALITY	18 20 20 20 20 20 21 21 21 21 21 21 21 22 22 22 23				

CLAUSE 16	INDEPENDENT PARTIES	24			
Clause 16.1	Independence	24			
Clause 16.2	Relationship	24			
CLAUSE 17	TERM	24			
Clause 17.1	Term of Effectiveness	24			
CLAUSE 18	NOTIFICATIONS	24			
Clause 18.1	Notification Rules	24			
Clause 18.2	Term	24			
Clause 18.3	Change of Address	24			
Clause 18.4	Effect	24			
CLAUSE 19	ANTI-CORRUPTION REPRESENTATIONS AND	25			
CLAUSE 19	WARRANTIES				
Clause 19.1	Representations and Warranties	25			
CLAUSE 20	LAW AND CONFLICT RESOLUTION	25			
Clause 20.1	Specific Performance; Enforceable Instrument	25			
Clause 20.2	Applicable Law 2				
Clause 20.3	Arbitration	26			

### SHARING AGREEMENT ENTERED INTO BETWEEN CALITEIA RJ INFRAESTRUTURA E REDES DE TELECOMUNICAÇÕES S.A. AND OI MÓVEL S.A. - UNDER JUDICIAL REORGANIZATION

By this private instrument, entered into on [-] [-], 2020, on one side,

**1.** Caliteia RJ Infraestrutura e Redes de Telecomunicações, a corporation registered with the National Register of Legal Entities ("<u>CNPJ</u>") under No. 35.978.982/0001-75, with its head offices at Rua do Lavradio, n.º 71, sl. 201/801, Centro, ZIP CODE 20.230-070, Centro, in the City of Rio de Janeiro, State of Rio de Janeiro, herein duly represented pursuant to its Bylaws, hereinafter simply referred to as "<u>Contractor</u>";

and, on the other side,

2. Oi Móvel S.A. — Under Judicial Reorganization, a corporation with its head offices at Setor Comercial Norte, Quadra 03, Bloco A, Ed. Estação Telefônica, Térreo, Parte 2, in the City of Brasília, in the Federal District, registered with the CNPJ under No. 05.423.963/0001-11, herein duly represented pursuant to its Bylaws, hereinafter simply referred to as "<u>Principal</u>"; and

and also, as intervening guarantor party,

**3. Oi S.A.** — **Under Judicial Reorganization**, a publicly-held company, with its head offices at Rua do Lavradio n.° 71, Centro, in the City and State of Rio de Janeiro, ZIP CODE 20230-070, registered with the CNPJ under No. 76.535.764/0001- 43, herein duly represented pursuant to its Bylaws, hereinafter simply referred to as "<u>Guarantor</u>".

The Principal and the Contractor shall be hereinafter individually and indistinctly referred to as "Party" and together as "<u>Parties</u>".

# WHEREAS:

(A) Contractor owns certain Infrastructure Items (as defined below) identified in <u>Exhibit 1</u>, located on its own and/or third party land and/or properties, and has the right to assign the use thereof for the installation of Equipment (as defined below);

(B) the Parties are interested in negotiating the terms and conditions for the lease of certain spaces located in Contractor's Infrastructure Items, as requested by Principal, subject to the terms and conditions provided for in this Agreement;

(C) Principal is a provider of telecommunications services of collective interest, and uses infrastructure items to make its services available to its final clients; and

(**D**) the Parties are fully aware that Principal is subject to the rules imposed by the Brazilian Agency of Telecommunications (ANATEL),

The Parties hereby agree to enter into this Sharing Agreement ("<u>Agreement</u>"), which will governed by the following terms and conditions:

### **CLAUSE 1 - DEFINITIONS**

**Clause 1.1.** <u>Definitions</u>. When used in this Agreement, each of the terms defined in the Preamble and the Recitals will have the meaning indicated above, and the following terms will have the meaning indicated below:

"<u>Affiliate</u>" means, in relation to any of the Parties, any Person who directly or indirectly Controls, is Controlled by, or is under common Control with that Party, provided that Persons in the portfolio of funds managed by [name] will not be considered Affiliates of Contractor.

"<u>ANATEL</u>" means the Brazilian Agency of Telecommunications.

"<u>Regularization Cost Increase</u>" has the meaning ascribed thereto in the Sale and Purchase Agreement.

"<u>Governmental Authority</u>" means any governmental agency, independent agency, division, department, court or other direct or indirect federal, state or municipal government body, as well as any and all courts, judicial authorities and/or arbitration courts.

"Civil Code" means Brazilian Federal Law No. 10,406, dated January 10, 2002, as amended.

"<u>Brazilian Code of Civil Procedure</u>" means Brazilian Federal Law No. 13,105, dated March 15, 2015, as amended.

"<u>Sale and Purchase Agreement</u>" means the Share Purchase Agreement via Isolated Production Unit and Other Covenants entered into on *[date]* between Oi Móvel S.A. under Judicial Reorganization and Telemar Norte Leste S.A. - under Judicial Reorganization, as Sellers and [=], as Buyer (hereinafter referred to as the "<u>Buyer</u>"), with the intervention and consent of Contractor and the Guarantor, having as purpose the indirect purchase and sale of the Infrastructure Items by means of the direct acquisition of the total share capital of Contractor by Buyer.

"Lease Agreements" means the contractual instruments (written or verbal) related to the occupation of the respective spaces in which the Infrastructure Items are installed, such as specific lease agreements, loan for use (*comodato*), assignment for use of space, among others.

"<u>Control</u>" means, in relation to a Person, the power to, directly or indirectly, individually or jointly with other Persons, manage and order the direction of such Person's management and policies, by means of ownership of the majority of its voting capital, by operation of an agreement, or by any other means. Terms derived from Control, such as "Parent", "Controlling", "Controller" and other related words will have meanings similar to Control.

"<u>Closing Date</u>" means the date on which the parties to the Sale and Purchase Agreement close the legal transactions provided for in the Sale and Purchase Agreement.

"<u>Business Day</u>" means any day (i) that is not a Saturday or Sunday, or (ii) when commercial banks are open for business at regular business hours in the municipality of Rio de Janeiro, State of Rio de Janeiro and in the municipality of São Paulo, State of São Paulo, Brazil.

"Equipment" means, all together, the generators, telecommunications, radio and other equipment present in the structures of any Infrastructure Item and which are the property of Principal, its Affiliates, as well as their antennas and/or connection cables for such equipment that are apt to provide Personal Mobile Service and/or Specialized Mobile Service. Equipment include, without limitation, dedicated equipment, power cables for Principal's and its Subsidiaries' shelter/container, fiber optic cables, radio frequency and transmission antennas and their cables, exclusive and specific grounding system for these equipment associated grounding devices with coaxial and cables. shelter/container/outdoor Base Radio Station of Principal, and its Affiliates, all electronics and batteries associated with the shelter/container/outdoor Base Radio Station of Principal, and its Affiliates, specific internal and external grounding of the shelter/container/outdoor Base Radio Station of the Contractor, its Affiliates (including the physical grounding material and the physical connection between the Equipment and the general grounding system of the site), generators, diesel tanks, active and passive security system, telephone lines, COW (Cell on Wheels) base station, and any substations and power transformers dedicated to Principal, and its Affiliates, all energy contracts related to the Equipment, as well as the operating rooms and masonry shelters, but always excluding their respective foundations.

"<u>Base Radio Station</u>" means the equipment of the Personal Mobile Service or Specialized Mobile Service network located in an Infrastructure Item and responsible for communicating with the terminal equipment of the users of these services, through the transmission and reception of signals in the frequency bands authorized by ANATEL to Principal, as applicable, for the Personal Mobile Service or Specialized Mobile Service.

"<u>Excess Spaces</u>" means, with respect to an Infrastructure Item, any area available in that Infrastructure Item in which it is feasible to assign, lease or grant the right of use to any

Persons and, therefore, that is not being used, leased or assigned by Contractor in any capacity.

"<u>Adjustment Index</u>" means the Broad Consumer Price Index — IPCA, disclosed by the Brazilian Institute of Geography and Statistics — IBGE, or another index that may replace it.

"<u>Infrastructure Items</u>" means the infrastructure sites (including greenfield sites) described in <u>Exhibit 1</u>, which are comprised of the set including a Tower and/or Roof Top.

"<u>New Third-Party Sharing</u>" means a request made by any third party other than Principal and/or its Affiliates, for Contractor to share and/or allow the use, by such third party, of additional space in one or more Infrastructure Items, provided, however, that any sharing, leases and other space assignments (i) that are already in effect on the date hereof; and/or (ii) in Infrastructure Items in respect of which the Right of First Refusal of Excess Space (as defined below) has already been exercised or waived under this Agreement, will <u>not</u> be considered "New Third-Party Sharing" for the purposes hereof.

"<u>Person</u>" means any individual, legal entity, businessperson, general partnership or company (including, but not limited to, corporations, limited liability companies or other types of companies), foundation, investment fund, association, partnership, consortium, trust, fiduciary entity or any other entity or organization, with or without legal personality, or any Governmental Authority.

"<u>Oi Judicial Reorganization</u>" means the judicial reorganization process of Guarantor, together with other companies of its economic group, as a request filed on June 20, 2016 before the 7<sup>th</sup> Commercial Court of the Judicial District of the Capital City of Rio de Janeiro, under number 0203711-65.2016.8.19.0001.

"<u>Additional Consideration</u>" has the meaning ascribed thereto in <u>Exhibit 2</u> to this Agreement.

"Base Consideration" has the meaning ascribed thereto in Exhibit 2 to this Agreement.

"<u>Roof Tops</u>" means, in the infrastructure, the set of structures intended to support the radiofrequency and transmission antennas: support and fixation base for the vertical structures, lighting (including light barrier, photocell and wiring controls), pedalboards (including fall arrest safety cable), vertical or horizontal cable bed, grounding system (including lightening rod, wires and ground connections and the specific grounding mesh for the structure), beacons, access stairways and platforms, fixation/support structure for the equipment, switchboards, power piping and electrical cables. Roof Tops are usually assembled on: (i) slabs of building roofs; (ii) roofs; (iii) water tanks; (iv) facades (antennas only); and (v) any combination of the items above. Additionally, the characterization of the Roof Tops considers that they have a fully functional Base Radio Station owned by Principal and its Affiliates. The Roof Tops are listed in <u>Exhibit 1</u>, as "Roof Top" in the column "Site Type."

"<u>Irreversible Situation</u>" has the meaning ascribed thereto in Clause [10.2] of the Sale and Purchase Agreement.

"Towers" means every structural set capable of supporting the installation of transmission and radiofrequency antennas, including the tower structure, the foundation of the tower structure, the tower lighting (including light barrier, photocell and wiring controls, cables), the tower work platform, all antenna supports and tower equipment, tower rest platforms, stairways to the tower (including the fall arrest safety cable, body guard, bracings, vertical and horizontal stretchings, the general grounding system of the tower (including lightning rods, wires and ground connectors for the tower and the land grounding mesh), site grounding system (including the global grounding system for the site regarding fences, walls, doors, recipients, gates and energy inputs), power input panel where the meters are located, concrete foundations and/or metal shelters for power input, power infrastructure from the utility company's distribution network, the power input pattern, including ducts, posts and power piping and fiber optic, junction boxes as well as the materials related to the site perimeter (such as walls, fences, gates, etc.), metal skids for the Base Radio Station, concrete base for the Base Radio Station, metallic "eco box" (structure in metal profiles and checkered-plate floor and variable sizes) for the Base Radio Station, site lighting system, industrial outlet for a generator (steck), excluding any equipment that are installed in or attached to the Tower. The Towers are identified in Exhibit 1 in the "site type" column with the "greenfield" description.

"<u>Taxes</u>" mean all taxes, contributions, charges, rates, fees, duties, social contributions or other governmental charges of any nature, including, without limitation, all income taxes, withheld at the source or not, on capital gains, share capital, transfer, sale, use, occupation, ownership, consumption, franchise, severance, paid leave, payroll, federal, state, municipal and local taxes withheld at source and other taxes, as well as accretions, fines and interest in relation to any said amounts.

**Clause 1.2.** <u>Other Definitions</u>. Without prejudice and in addition to the terms defined in Clause 1.1 above, the following terms are defined in the wording of the Agreement:

" <u>Exhibits</u> "	9
" <u>Arbitration</u> "	26
" <u>Chamber</u> "	26
" <u>CNPJ</u> "	5
" <u>Contractor</u> "	5
"Principal"	5
" <u>Agreement</u> "	5
"Consideration Dilution"	17
"Right of First Refusal of Excess Space"	9
" <u>Dispute</u> "	26
" <u>Form</u> "	10
" <u>Guarantor</u> "	5
"Remaining Infrastructure Items"	17
" <u>Space Lease</u> "	8
" <u>NF/FS</u> "	17
" <u>Irregular Works</u> "	14
"Forbidden Payment"	25
" <u>Party</u> "	5
" <u>Parties</u> "	5
"Operating Procedures"	9
" <u>RAN Sharing</u> "	13

" <u>Regulation</u> "	26
"Base Consideration for RAN Sharing"	13
" <u>Considerations</u> "	15
"Rent Pass-Through"	15
"Regularization Cost Increase Pass-Through"	16
"Arbitral Tribunal"	26

#### **CLAUSE 2 - PURPOSE**

**Clause 2.1.** <u>Space Lease</u>. The purpose of this Agreement is the lease of spaces in the Infrastructure Items owned and/or held by Contractor, as listed in <u>Exhibit 1</u>, for the installation of the Equipment of Principal and its Affiliates, without resulting in a direct or indirect transfer of property and/or ownership of such Infrastructure "<u>Space Lease</u>").

**Clause 2.2.** <u>Consideration</u>. The Space Lease provided for in this Agreement will be remunerated individually for each Infrastructure Item, according to the metrics, values and conditions set out in <u>Exhibit 2</u> in and Clause 8.1, and must comply with the Operating Procedures detailed in <u>Exhibit 3</u>.

**Clause 2.3.** <u>Scope of the Space Lease</u>. The provisions in this Agreement apply only and solely to the Infrastructure Items listed in <u>Exhibit 1</u>, so that any agreements or other instruments entered into or that may be entered into by the Parties and/or their Affiliates and which have a purpose similar to that of the this Agreement, will remain (or will become, as the case may be) in force in accordance with their respective terms and conditions.

### **CLAUSE 3 - EXHIBITS**

**Clause 3.1.** <u>Exhibits</u>. The following exhibits (together, the "<u>Exhibits</u>") are part to this Agreement, as if they were transcribed herein:

3.1.1 List of Exhibits:

- a. <u>Exhibit 1</u> List of Infrastructure Items contracted;
- b. <u>Exhibit 2</u> Consideration;

c. <u>Exhibit 3</u> - Minimum Requirements for Operating Procedures ("<u>Operating</u> <u>Procedures</u>");

- d. <u>Exhibit 4</u> Form;
- e. <u>Exhibit 5</u> Letter Indicating the Checking Account; and

f. <u>Exhibit 6</u> - RAN Sharing Leases.

3.1.2 In the event of any divergence between the Exhibits and this Agreement, the provisions in this Agreement shall prevail.

3.1.3 The content and form of the exhibits will be kept up to date by mutual agreement between the Parties, subject to the provisions of this Agreement, provided that Contractor may update Exhibit 5 unilaterally, at its sole discretion.

### **CLAUSE 4 - SPACE LEASE**

**Clause 4.1.** <u>Right of First Refusal of Excess Space</u>. Contractor hereby grants, in an irrevocable and irreversible manner, the right of first refusal to Principal and any of its Affiliates to lease additional space in Infrastructure Items whose Excess Space is equal to or less than five (5) square meters, in order to allow the expansion of the Equipment of Principal and/or its Affiliates already installed or for the installation of new Equipment necessary to provide the telecommunications services of Principal or its Affiliates ("Right of First Refusal of Excess Space").

4.1.1 For the avoidance of doubt: (i) in the event that Contractor receives a request for New Third-Party Sharing that, if implemented, will reduce the Excess Space in any Infrastructure Item to an area equal to or less than five (5) square meters, the Right of First Refusal of Excess Space will apply to such New Third-Party Sharing, and (ii) the Right of First Refusal of Excess Space shall be exercised by Principal only once for each Infrastructure Item, upon receipt of the notification referred to in Clause 4.1.3.

4.1.2 If Contractor (i) exercises the Right of First Refusal of Excess Space, as established in Clause 4.1 above, or (ii) otherwise, even if not subject to the Right of First Refusal of Excess Space, requests the expansion of the respective occupied area and/or the load of the respective Equipment, the Space Lease in relation to the Excess Spaces subject matter of said exercise or expansion will be subject to the payment of the Additional Consideration related to the expansion of the space, as established in Exhibit 2. In the event of exercise of the Right of First Refusal of Excess Space, Contractor may not deny the expansion of the Space Lease of such Infrastructure Item to Principal and its Affiliates under any circumstances, except in cases where such expansion is in disagreement with the terms of this Agreement, subject, in any case, to the provisions of Clause 4.1 above.

4.1.3 If Contractor receives a request for a New Third-Party Sharing referring to a specific Infrastructure Item whose Excess Space is equal to or less than five (5) square meters and, therefore, is subject to the Right of First Refusal of Excess Space, Contractor must deliver a notification to Principal, for the latter to inform, within ten (10) Business Days of receipt of such notification, if Principal: (i) wishes to exercise the Right of First Refusal of Excess Space; or (ii) does not wish to exercise the Right of First Refusal of Excess Space, being agreed that the silence or the untimely sending of the notification of its right of first refusal will be interpreted as a decision not to exercise the Right of First Refusal of Excess Space.

4.1.4 Contractor shall guarantee the availability of Excess Spaces subject to the Right of First Refusal of Excess Space in accordance with the provisions above, provided that (i) there is technical feasibility of the installation project in the respective Infrastructure Item; and (ii) Principal fully bears any investments of Contractor necessary for that purpose; provided that any applicable costs (including, but not limited to, the cost of structural analysis) are duly, formally and previously approved by Principal.

4.1.5 For the avoidance of doubt, Contractor will have no obligation to ensure or guarantee availability of Excess Space in Infrastructure Items in relation to which the Right of Refusal of Excess Space has already been exercised or waived pursuant to this CLAUSE 4.

**Clause 4.2.** <u>Request Forms</u>. New requests for expansion or reduction regarding each Infrastructure Item indicated in <u>Exhibit 1</u> to this Agreement shall be agreed upon by the Parties, in writing, through a Space Lease Form ("<u>Form</u>"), according to the model contained in <u>Exhibit 4</u> and are subject to the provisions of Clause 6.1.6 below for cases of expansion.

4.2.1. For the avoidance of doubt: (i) the mere replacement of Equipment, as long as it does not represent the need to expand the respective occupied area and/or the load of the respective Equipment, shall not constitute a new request for the purposes of Clause 4.2 above, and (ii) any request by Principal that causes expansion of the respective occupied area and/or the load of the respective Equipment will be considered a new request for service purposes, and it will imply Additional Consideration under the terms of Clause 4.1.2 below and Exhibit 2.

4.2.2. If Principal submits a request for expansion/Lease of Excess Space (or exercises the <u>Right of First Refusal of Excess Space</u>) in relation to an Infrastructure Item in relation to which, in Contractor's reasoned understanding, there is a reasonable risk of declaring an Irreversible Situation, Contractor reserves the right to condition the acceptance of such requests and/or exercise to the assumption, by Contractor, of the obligation indicated in Clause 8.5.2.

**Clause 4.3.** <u>Reforms and Adjustments</u>. Principal shall design, execute, contract and inspect the reforms and adaptations necessary for the installation of its Equipment, in accordance with Contractor's criteria and specifications, with all expenses being borne exclusively by Principal, provided that these expenses do not relate to maintenance and repair of the Infrastructure Items (which will be the responsibility of Contractor).

**Clause 4.4.** <u>Right of Retention</u>. Upon termination of this Agreement, for any reason, Principal will not be entitled to retain or be indemnified for any improvements made to the Infrastructure Items, which will be incorporated into the Infrastructure Items and/or the land or properties on which said Infrastructure Items are located/installed, subject to the provisions of Clause 4.4.1 below.

4.4.1 Without prejudice to the provisions of Clause 4.4 above, at the end of the term of this Agreement, Principal shall have the right to remove the Equipment installed in the Infrastructure Items and that are exclusively owned by Principal, provided that (i) Contractor shall grant Principal access to the Infrastructure Items, without any charge or additional cost, subject to the terms and conditions provided herein, as applicable, and (ii) all costs and expenses related to the removal of Principal's Equipment will be borne exclusively by Principal.

**Clause 4.5.** <u>Lease Agreements</u>. The Parties agree to be bound by the rights and obligations contained in this Agreement and shall comply, where applicable, with the terms and conditions contained in the Lease Agreements.

### **CLAUSE 5 - COMMON OBLIGATIONS OF THE PARTIES**

**Clause 5.1.** <u>Common Obligations</u>. In addition to other obligations provided for in this Agreement, the Parties undertake to fulfill and comply with the following commitments, as applicable:

5.1.1 Communicate to the other Party, immediately after identification, any relevant abnormalities or changes detected to the Infrastructure Items and/or the Equipment of Principal and its Affiliates, as the case may be, that may affect the rights of the other Party and/or third parties, and such communications shall be in writing within twenty-four (24) hours after the verbal communication to the other Party.

5.1.2 Correct, within the maximum period of seventy-two (72) hours and at its own expense, any interference that may be caused by the equipment under its responsibility (including the Equipment of Principal and its Affiliates), to the systems of the other Party or third parties under this Agreement or other sharing agreements with third parties.

5.1.3 Maintain, when technically possible, the equipment under its responsibility (including the Equipment of Principal and its Affiliates) installed within each Infrastructure Item in separate environments and with independent access.

### CLAUSE 6 - CONTRACTOR'S OBLIGATIONS

**Clause 6.1.** <u>Contractor's Obligations</u>. Contractor's obligations include, in addition to others provided for in this Agreement:

6.1.1 Provide Principal, under this Agreement (and subject to the terms of the Lease Agreements), with the Space Lease, in accordance with the procedure and term agreed between the Parties.

6.1.2 Grant free and unconditional access, at any time and day, twenty-four (24) hours a day, three hundred and sixty-five (365) days a year, to the Infrastructure Items in which Space is leased by Principal or its Affiliates, provided that this is permitted by the terms of the Lease Agreements, to Principal's professionals and/or representatives, as duly identified and accredited, so that they proceed with the installation, maintenance, conservation and repair in the Equipment of Principal and its Affiliates, provided that Principal shall comply with the procedures for accessing the Infrastructure Items previously and formally informed by Contractor.

6.1.3 Inform Principal in advance of any changes to its operating and safety procedures.

6.1.4 Provide Principal, within five (5) days, counted from the date of receipt of such request, with the information and documents that are in its possession and that are necessary to obtain any licenses and permits, as well as any other documents required by the applicable legislation for that purpose.

6.1.5 Except for corrective maintenance, Contractor shall previously communicate to Principal, within a maximum period of seventy-two (72) hours, of any interruptions in the power supply or air conditioning, due to the need of maintenance of the Infrastructure

Items, in case electricity or air conditioning have been contracted and are the responsibility of Contractor and/or its Affiliates with respect to such Infrastructure Item.

6.1.6 Verify, within thirty (30) days from the date of each request sent by Principal in this regard, the feasibility and technical availability of the Infrastructure Items in relation to requests that may be sent by Principal under the terms of Clause 4.2.1.

6.1.7 Maintain, at its expense for the entire term of this Agreement, with a first-rate insurance company, general and professional liability insurance and operating risk insurance for the Infrastructure Items subject matter of this Agreement and provide Principal, if so requested and in good faith, a copy of the respective policies contracted within thirty (30) days from the date of such request.

6.1.8 Be liable for the payment of any expenses resulting from fines or infractions to which it gives rise, as of the date of execution of this Agreement, for non-compliance with any laws, decrees or regulations that may affect the Infrastructure Items subject matter of this Agreement, except if (i) such expenses result from the absence of any license that is the obligation of Principal and/or its Affiliates, (ii) such expenses result from any facts, acts or omissions prior to the Closing Date; or (iii) Principal and/or its Affiliates are required to bear such expenses in accordance with this Agreement or the Sale and Purchase Agreement.

6.1.9 Bear, when requesting any licenses, permits, certificates and any other documents required for the execution and legalization of the facilities, equipment, works or services under its liability, including, but not limited to, the Infrastructure Items, all expenses arising from such request, as well as be liable for such request with the Governmental Authorities, except in the cases in which Principal and/or its Affiliates are required to bear such expenses and/or implement such procedures, in accordance with this Agreement or the Sale and Purchase Agreement.

6.1.10 Be liable for any direct damage that may be caused by its representatives, employees, workers, agents or contractors to the Equipment, assets and/or facilities of Principal and/or its Affiliates, as provided such damage is duly proven, subject to the provisions of this Agreement.

6.1.11 Keep Principal's Equipment free from any radio frequency interference, under penalty of bearing all costs arising therefrom, provided, however, that nothing in this Agreement shall compel Contractor to bear any costs related to radio frequency interference caused by facts, acts, omissions or circumstances prior to the Closing Date, which shall be fully borne by Principal, provided that such costs are duly proven by Contractor through technically substantiated reports.

6.1.12 Correct, at its expense, within seventy-two (72) hours, counted from the date of receipt of a written notification to this end, any interferences that its equipment or third party equipment under its liability is causing to the Equipment of Principal and its Affiliates installed in the Infrastructure Items, provided that the notification indicated in this Clause must contain reasonable details of the interferences to be corrected.

6.1.13 Observe and comply with the Operating Procedures, as provided for in this Agreement.

6.1.14 Be liable for the execution and payment of expenses resulting from repairs for conservation and corrective and preventive maintenance of the premises, installations and other tools associated with the use of the contracted Infrastructure Items, except in the cases in which Principal and/or its Affiliates are required to bear such expenses and/or implement such procedures in accordance with this Agreement or the Sale and Purchase Agreement.

# CLAUSE 7 - PRINCIPAL'S OBLIGATIONS

**Clause 7.1.** <u>Principal's Obligations</u>. Contractor's Obligations include, in addition to others provided for in this Agreement:

7.1.1 Formalize the new uses of the Infrastructure Items by means of a correctly completed Form, with the specifications, technical data, usage characteristics, installation project and other required information, pursuant to Clause 4.2 of this Agreement.

7.1.2 Provide all clarifications and technical information on the Space Lease and the Equipment belong to Principal and its Affiliates, which may be requested by Contractor.

7.1.3 Not to assign, transfer, donate, exchange, sublease, encumber, pledge as security, lend for use, negotiate or lend, at any title, the spaces leased in the Infrastructure Items subject matter of this Agreement, totally or partially, without the prior written consent of Contractor, except as set forth in Clause 13.1.

a. Without prejudice to the provisions set forth above, the sharing of Equipment, access network and spectrum ("<u>RAN Sharing</u>") may be performed by Principal with other telecommunications carriers, upon payment, for each Infrastructure Item with RAN Sharing, of the Base Consideration for RAN Sharing per operator, as provided in <u>Exhibit 2</u> ("<u>Base Consideration for RAN Sharing</u>").

b. On the date hereof, Principal represents that there are fifty-four (54) Infrastructure Items with RAN Sharing leases (representing a total of sixty-five (65) cases of RAN Sharing), as listed in <u>Exhibit 6</u>, which are hereby considered authorized by Contractor and on which the Base Consideration for RAN Sharing will apply.

c. Any additional RAN Sharing leases to be made by Principal with other telecommunications carriers will be subject to (i) Base Consideration for RAN Sharing, pursuant to <u>Exhibit 2</u>; and (ii) the notification procedures provided for in Clause 7.1.18.

7.1.4 Execute, at its expense, any works, services or facilities necessary for the use, installation and/or expansion of Equipment in the Infrastructure Items, provided that Contractor's prior written consent is obtained, accompanied by presentation and approval of the applicable technical projects, provided that if Principal does not respond within five (5) Business Days of receiving the request for approval sent under the terms above, then the works, services or facilities mentioned in the communication sent by Principal will be considered accepted for all purposes of this Agreement.

7.1.5 Ensure Contractor the right to inspect, by itself or by its duly accredited representatives and together with representatives or employees of Principal and/or its

Affiliates, by giving notice at least five (5) days in advance, the works, services and installations carried out or in progress, related to the use, installation and/or expansion of Equipment in the Infrastructure Items contracted, in order to check if the obligations undertaken by Principal are being fulfilled.

7.1.6 Maintain, at its expense and throughout the term of this Agreement, with the firstrate insurance company, general and professional liability insurance and operating risk insurance for the facilities and installed equipment held by Principal, which are allocated or located in the Infrastructure Items contracted, with adequate coverage, and provide Contractor, if so requested in good faith, with a copy of the respective policies contracted within thirty (30) days from such request.

7.1.7 Inform Contractor, at least seven (7) days in advance, the date, location and conditions of arrival of Equipment and other related materials to be installed, allocated and/or destined for Space Lease in the Infrastructure Items by Principal and, if applicable, its Affiliates, except in cases of need for installation or emergency use, in which case the aforementioned period of seven (7) days will be reduced, upon reasoned justification by Principal, to two (2) Business Days.

7.1.8 Be liable for the payment of any expenses arising from fines or infractions that it gives rise to due to noncompliance with any laws, decrees or regulations, as they may be levied on (i) its Equipment and service provided; and/or (ii) provided they are exclusively attributable to Principal, on the Infrastructure Items subject matter of this Agreement, without prejudice to the relevant rights and obligations stipulated in the Sale and Purchase Agreement.

7.1.9 Bear all expenses resulting from obtaining, as well as the liability for requesting from the Governmental Authorities, any licenses, permits, certifications, exemptions and any other documents required for the execution and legalization of its Equipment, installations, works and/or services that are under its liability, including, but not limited to, the Equipment of Principal and its Affiliates.

7.1.10 Do not put any advertising or communication materials of an institutional or marketing nature in the Infrastructure Items without the prior written authorization of Contractor.

7.1.11 Maintain and observe, in relation to its Equipment installed in the Infrastructure Items and throughout the term of this Agreement, all licenses, permits, exemptions and other certificates related or required for the operation of its Equipment and those of its Affiliates, as required by pertinent law, including, but not limited to, the requirements in the regulation established by ANATEL.

7.1.12 Require visible identification of its representatives, employees, workers, agents or contractors, in compliance with the provisions of this Agreement, to be allowed to enter and circulate at Contractor's premises, including the Infrastructure Items.

7.1.13 Be liable for any direct damage that may be caused by its representatives, managers, employees, workers, agents or contractors, provided such damage is duly proven, subject to the provisions of this Agreement.

7.1.14 Observe and comply with the access and security procedures related to the premises where the Infrastructure Items and other facilities of Contractor are located.

7.1.15 Observe and comply with the Operating Procedures, as provided for in this Agreement.

7.1.16 Install, allocate and assign to (as well as allow the installation, allocation or assignment only in) the Infrastructure Items, Equipment and materials that have been duly approved and licensed by ANATEL, under the terms provided for in the applicable legislation.

7.1.17 Eliminate, at its expense, undue interference caused by its Equipment in the Infrastructure Items and other equipment of Contractor or third parties previously installed or allocated in the Infrastructure Items within seventy-two (72) hours counted from the receipt of a written notification from Contractor in this regard, provided that the notification provided for in this Clause must contain reasonable details of the interference to be corrected.

7.1.18 Notify Contractor fifteen (15) days in advance of the intention to enter into additional RAN Sharing leases and provide the applicable supporting documentation.

7.1.19 Be liable for any direct damage that may be caused in any of the Infrastructure Items by the telecommunications carriers that have carried out RAN Sharing with Principal and/or its Affiliates, provided they are duly proven, subject to the provisions hereof.

7.1.20 Be liable to Contractor for the use of Infrastructure Items by its Affiliates, in compliance with the terms and conditions established in this Agreement, and undertake to indemnify Contractor for any noncompliance by its Affiliates with the terms and conditions set forth in this Agreement.

**Clause 7.2.** <u>Irregular Works, Installations and Services</u>. Without prejudice to Clause 7.1.4 above, in the event that the services, facilities and/or works performed by Principal under this Agreement are in disagreement with their technical implementation projects (together, "<u>Irregular Works</u>"), then Contractor may request Principal that the Irregular Works be undone or redone, if said inadequacies of the Irregular Works imply any damage or imminent risk to the Infrastructure Items and/or its facilities, systems and other equipment allocated/installed there. On the other hand, if said inadequacies do not imply any damage or imminent risk to the Infrastructure Items and/or to its facilities, systems and other equipment allocated/installed there, it is hereby established that Principal, after being duly notified, will have a period of up to twenty (20) Business Days to carry out the due regularization and adaptation of the Irregular Works to their technical implementation projects.

7.2.1 All damages, costs and other expenses associated with the Irregular Works, as provided for in Clause 7.2 above, including those that are necessary for the undoing, redoing, adequacy and regularization of the Irregular Works, shall be the sole liability of Principal. In this regard, Principal undertakes to indemnify and keep Contractor harmless from any damage, costs and other expenses associated with the Irregular Works and that, for any reason, will be borne or incurred by Contractor and/or its Affiliates.

**Clause 7.3.** <u>Indemnification of the Sale and Purchase Agreement</u>. Nothing in the provisions of this Agreement shall impact, diminish, limit and/or impair, in any way, any of the indemnification obligations provided for in the Sale and Purchase Agreement, which will remain in full force and effect under the terms agreed therein. In addition, the terms and conditions in the Clauses of this Agreement shall not be construed, under any circumstances, as novation, modification and/or alteration of said indemnification obligations provided for in the Sale and Purchase Agreement, and any and all losses and damages incurred by Contractor pursuant to this Agreement will not be subject to the limits provided for in the Sale and Purchase Agreement, provided, however, that a loss in respect of which an indemnification will be duly paid under the terms of the Sale and Purchase Agreement, then such loss will not be indemnified again under this Agreement.

7.3.1 For the avoidance of doubt, and pursuant to Clause [10.2.3] of the Sale and Purchase Agreement, an Irreversible Situation may give rise to the application of the provisions of Clause 8.5 and also give rise to additional losses and damages, which will remain recoverable under the terms, limits and conditions of the Sale and Purchase Agreement.

## CLAUSE 8 - PAYMENT AND ADJUSTMENT

**Clause 8.1.** <u>Considerations</u>. In consideration for the Space Lease in the Infrastructure Items, Principal shall pay Contractor the amounts specified below, without prejudice, however, to the amounts due pursuant to <u>Exhibit 2</u> and other considerations expressly provided for in this Agreement (together, the "<u>Considerations</u>"):

8.1.1 Principal shall pay the Base Consideration on a monthly basis, as set out in Exhibit 2 to this Agreement, which refers to the lease and use of space in each of the Infrastructure Items. On the date of execution of this Agreement, the total amount of the Base Consideration due by Principal to Contractor is provided for in Exhibit 2.

a. Principal hereby acknowledges and agrees that (i) for each Infrastructure Item, the amount due by Contractor to the counterpart of the Lease Agreement related to the Infrastructure Item in question shall be added to the amount of the Base Consideration of such Infrastructure Item, according to the amounts stipulated in Exhibit 2 ("Lease Pass-<u>Through</u>"), subject to the provisions of Clause 8.1.4; and (ii) for certain Infrastructure Items, Contractor shall reimburse Principal for certain expenses; in any case, as set forth in Exhibit 2, without prejudice, however, to the provisions in Clause 8.1.4 below.

b. As detailed in Exhibit 2, the Base Consideration will not be affected or decreased due to any requests from Principal that imply a reduction in the respective areas occupied in each Infrastructure Item.

8.1.2 In addition to the Base Consideration, Principal shall pay the Additional Consideration on a monthly basis, as set forth in <u>Exhibit 2</u> to this Agreement, which will be applicable for all Infrastructure Items, should this be due by Principal hereunder.

8.1.3 In addition to the Base Consideration and the Additional Consideration (if applicable), Principal shall pay the Base Consideration for RAN Sharing on a monthly basis, as set forth in <u>Exhibit 2</u> to this Agreement, which will be due (a) in relation to all

Infrastructure Items with RAN Sharing leases made by Principal; and (b) by telecommunications operator with RAN Sharing in the Infrastructure Item, so that more than one monthly installment of Base Consideration for RAN Sharing per Infrastructure Item may be due, if Principal has performed RAN Sharing of such Infrastructure Item for more than one telecommunications operator, pursuant to Clause 7.1.3 above.

8.1.4 Without prejudice to the Considerations due under this Agreement, the Parties agree and covenant that the amount of Regularization Cost Increase determined in relation to one or more Infrastructure Items will be added in full to the Lease Pass-Through of the Infrastructure Item in question, provided that the Regularization Cost Increase will be fully due by Principal, without applying any type of proportion or division with any other Persons (other than Principal and/or its Affiliates) who may have valid Space Leases in relation to the Infrastructure Item in question ("<u>Regularization Cost Increase Pass-Through</u>").

a. <u>Exhibit 2</u> provides for (i) the Infrastructure Items in which the Regularization Cost Increase Pass-Through is due on the date hereof, as well as the amount due as Regularization Cost Increase Pass-Through for such Infrastructure Items; and (ii) a merely illustrative example of the calculation of the Regularization Cost Increase Pass-Through that may be due under the terms of Clause 8.1.4 above for a given Infrastructure Item.

b. The Regularization Cost Increase Pass-Through will be due (i) while the Lease Agreement for the Infrastructure Item that caused the Regularization Cost Increase Pass-Through remains in force; (ii) in full, regardless of the number of days in the month in which (y) the Infrastructure Item was actually used by Principal and/or its Affiliates; and/or (z) the Equipment of Principal and/or its Affiliates were installed, allocated or located in the Infrastructure Item in question; and (iii) even if the Space Lease has been terminated for the Infrastructure Item in question (and/or the Equipment of Principal and/or its Affiliates are no longer installed, allocated or located in such Infrastructure Item).

c. Principal hereby acknowledges and agrees that the Regularization Cost Increase Pass-Through will be applied regardless of the number of Infrastructure Items that may be the object of a Regularization Cost Increase.

**Clause 8.2.** <u>Payment of the Considerations</u>. The Considerations related to the Infrastructure Items provided for in <u>Exhibit 1</u> will be due by Principal as of the date of execution of this Agreement and, in the case of leases of additional space in the Infrastructure Items, from the first day of availability thereof, by Contractor, of the additional space leased in the Infrastructure Items.

8.2.1 Subject to the provisions of Clause 8.1.4 above, the Base Consideration to be paid in return for the use and Space Lease of each Infrastructure Item will be proportional to the number of days in the month in which (i) this Infrastructure Item was effectively used Principal and/or its Affiliates; and/or (ii) Equipment of Principal and/or its Affiliates were installed, allocated or located in the Infrastructure Item in question, by taking into account, for this calculation, that each month is always composed of thirty (30) consecutive days. 8.2.2 The Considerations and the prices provided for in <u>Exhibit 2</u> will be adjusted every twelve (12) months or less, provided that there is no legal impediment, counted from the date of execution of this Agreement, by the positive variation of the Adjustment Index and, for the avoidance of doubt, provided that the Lease Pass-Through and the Regularization Cost Increase Pass-Through will be subject to the amount effectively due to the Counterparty in the Lease Agreement and the adjustments provided for therein.

8.2.3 Principal will be liable for and shall bear all Considerations due by its Affiliates under the terms of this Agreement, including those resulting from the Space Lease.

**Clause 8.3.** <u>Tax Liability</u>. Principal shall bear the financial burden of any Taxes that are levied or will be levied on the object of this Agreement and associated procedures, including PIS, COFINS or other Tax, contribution or charge of this nature, in accordance with the provisions of federal, state and local tax laws.

8.3.1 The applicable percentages or the calculated amounts of the Taxes, social contributions and charges mentioned above will be charged to Principal, together with the Considerations and other payments related to the Space Lease, so that the amounts receivable by Contractor are net of said Taxes.

**Clause 8.4.** <u>Electricity</u>. Principal shall contract electricity directly from the local electricity utility company or provide its own energy generators in relation to all its Equipment and/or those of its Affiliates. Contractor will be responsible for providing electricity to its tenants after the Closing Date, subject to the terms of the Clause below:

8.4.1 In situations in which, prior to the Closing Date, Contractor supplied electricity to other lessees of the Infrastructure Items, Contractor shall use its best efforts to make these lessees directly contract the supply of electricity. For the cases above, Contractor further undertakes to charge the other lessees of the Infrastructure Items for electricity costs related to the consumption of said lessees that were incurred directly by Principal and to reimburse such amounts to Principal. If such refund is not made, Principal is hereby authorized to offset said amount against any payments due by Principal to Contractor as a result of this Agreement.

**Clause 8.5.** <u>Guaranteed Consideration</u>. Without prejudice to the Considerations and other applicable payments, as provided for herein, Principal hereby acknowledges and agrees that in the event of an Irreversible Situation, the amount of the Base Consideration (added, if applicable, to the Additional Consideration as provided for in Clause 8.5.2) in relation to the Infrastructure Item(s) subject to the Irreversible Situation will be (i) divided by the total number of Infrastructure Items subject matter of this Agreement, excluding those that are the object of the Irreversible Situation (the "Remaining Infrastructure Items"), and (ii) the result of such division will be added to the value of the Base Consideration of the Remaining Infrastructure Items, so that the total amount due by Principal to Contractor remains equal to the amount that would be due if such an Irreversible Situation had not occurred ("Consideration Dilution").

8.5.1 The Consideration Dilution will remain applicable until the end of this Agreement, so that Contractor will be entitled, through the Consideration Dilution, to the original economic benefit that the Infrastructure Item object of the Irreversible Situation should provide to Contractor until the final term originally contracted and adjusted hereunder.

8.5.2 In situations where the Additional Consideration is due in order to enable the Additional Space Lease under the terms of Clause 4.2.2 above, then the Additional Consideration will be added to the Base Consideration for purposes of Consideration Dilution.

8.5.3 <u>Exhibit 2</u> contains an example of a hypothetical calculation for the application of the Consideration Dilution.

8.5.4 Principal hereby acknowledges and agrees that (i) the Consideration Dilution will be applied regardless of the number of Infrastructure Items that may be the subject of an Irreversible Situation, (ii) the application of the Consideration Dilution was agreed between the Parties so as to fully allocate the risk of an Irreversible Situation to Principal, and (iii) the Consideration Dilution was an essential element for Buyer to enter into with Principal and its Affiliates the Sale and Purchase Agreement, by means of which Buyer indirectly acquired the Infrastructure Items of Contractor and its Affiliates.

## CLAUSE 9 - BILLING

**Clause 9.1.** <u>Infrastructure Billing Statement</u>. Until the first (1<sup>st</sup>) Business Day of each month, Contractor shall send to Principal, an electronic spreadsheet called "<u>Infrastructure Billing Statement</u>" in order for Principal to check and verify the leased/shared Infrastructure Items and the financial amount related to the Consideration due for the previous calendar month. Principal shall (a) confirm its agreement with the "Infrastructure Billing Statement" for that month; or (b) provide its comments and/or present its disagreements with the "Infrastructure Billing Statement" for that month; or (15) days after receiving such "Infrastructure Billing Statement". The absence of a Principal's statement within the period provided for above will be interpreted as Principal's agreement in relation to all aspects of said "Infrastructure Billing Statement" for that month, subject to the provisions of Clause 8.2.2 above.

**Clause 9.2.** <u>Payment Date</u>. Subject to the provisions of Clause 8.1.4b above, the payment of the Considerations shall be made to Contractor until the due date indicated in the respective bills and/or invoices issued by Contractor ("<u>NF/FS</u>"), which shall be issued and sent to Principal at least ten (10) days in advance of the due date, subject to the provisions of Clause 9.2.1.

9.2.1 In the event that Principal and/or its Affiliates lease space in more than one hundred (100) Infrastructure Items contracted, the NF/FS shall be issued and sent to Contractor at least fifteen (15) days in advance of their due dates.

9.2.2 Contractor may assign the credit rights related to this Agreement on a fiduciary basis, including those arising from the Considerations, regardless of Principal's prior or later approval or consent. In the event of a fiduciary assignment of credit rights related to this Agreement by Contractor, Principal undertakes to make the payments due only in the bank account to be indicated in writing by Contractor in accordance with the letter that forms part of this Agreement according to the model contained in <u>Exhibit 5</u>.

9.2.3 If the term mentioned in Clause 9.2 or, if applicable, Clause 9.2.1, is not met, or even if the NF/FS and other collection documents present any errors and/or missing information as required by law or by this Agreement, then the Parties agree that the due dates indicated in the corresponding NF/FS will automatically be extended by the same number of days that the delivery of such NF/FS and other applicable collection documents was delayed, without any penalty on Principal.

## **CLAUSE 10 - PENALTIES**

**Clause 10.1.** <u>Default by Principal</u>. The Parties agree and covenant, without prejudice to the rights granted to Contractor in Clause 11.1 below, that any breach by Principal of its obligations provided for in this Agreement (including the non-payment of the Considerations on the due dates indicated in the respective NF/FS, as provided for in CLAUSE 9 above) will give rise to the following penalties, regardless of any prior notice or communication and without prejudice to any applicable losses and damages:

Default fine of two percent (2%) on the amount due to charged only once as of the day following the due date, without prejudice to the obligation to pay the full arrears;

Late-payment interest of one percent (1%) per month on the overdue amount, calculated *pro rata die*, accrued from the day after the due date until the date of the actual settlement of the debt; and

10.1.3 Adjustment for inflation calculated by the variation in the Adjustment Index, determined from the day after the due date until the date of the actual settlement of the debt.

Clause 10.2. Exclusive Fault - Contractor. During the term of this Agreement, if, due to Contractor's sole fault, (thus excluding facts attributed exclusively to third parties, including, but not limited to, Principal, its Affiliates, electricity utility companies and lessors in the Lease Agreements or beyond its control), Principal will be unable to access or use the spaces leased under the Space Lease of any of the Infrastructure Items in the manner provided for in this Agreement, and this affects or jeopardizes the quality and continuity of the services provided by Contractor, for a period longer than seventy-two (72) hours counted from the receipt of the notification sent by the Contractor in this sense, then Contractor will be required to pay a daily fine in the amount corresponding to the amount equivalent to 1/30 of the Considerations due by Principal in consideration for the Space Lease in the Infrastructure Item affected, limited to thirty (30) days. The fine mentioned in this Clause will not be due if the impossibility of using or accessing the Infrastructure Items results from (i) the terms of the Lease Agreements in force, (ii) an act, fact or omission that occurred prior to the Closing Date (including the Irregularities of the Sites, as defined in the Sale and Purchase Agreement), (iii) an Irreversible Situation, (iv) unforeseeable circumstances or force majeure, under the terms of the applicable legislation.

**CLAUSE 11 - TOTAL OR PARTIAL TERMINATION OF THE AGREEMENT Clause 11.1.**<u>Termination</u>. This Agreement may only be terminated, in its entirety or in relation to one or more Infrastructure Items, in the following cases: (i) proven noncompliance with the Parties' obligations, as set forth in Clause 11.1.1 below, or (ii) as set forth in Clause 11.1.3 below. 11.1.1 In the event of material breach of this Agreement by either Party, the innocent Party may terminate this Agreement, by operation of law, in relation to the Infrastructure Item(s) object hereof and/or that is/are affected by the default, by simple notification in writing to the defaulting Party, provided that such default is not remedied by the defaulting Party within sixty (60) days from the date of the notification sent by the innocent Party (as provided for in CLAUSE 18) of the noncompliance in question, without prejudice to the Parties initiating the relevant procedure under Clause 20.3 below.

11.1.2 The Parties hereby acknowledge and agree that, except for Overload Irregularities not remedied by Contractor within twelve (12) months from the Closing Date (as defined and regulated in Clause 10.4 of the Sale and Purchase Agreement) that results in a breach of this Agreement, no event arising from an act, fact or omission occurring prior to the Closing Date (including the Irregularities of the Sites, as defined in the Sale and Purchase Agreement or an Irreversible Situation) may be used to characterize a breach of this Agreement.

11.1.3 This Agreement may also be terminated in its entirety by operation of law and without prejudice to compensation for any losses and damages to the innocent Party in the following cases:

a. By Contractor, at its sole discretion, (i) in the event of adjudication of bankruptcy or even dissolution or liquidation of Principal, while Principal is Oi Móvel S.A. or any of its Affiliates, or (ii) in the event of adjudication of bankruptcy, a request for judicial or extrajudicial reorganization (except for Oi Judicial Reorganization currently underway), or even dissolution or liquidation of Contractor (in case Contractor ceases to be Oi Móvel S.A. or any of its Affiliates);

b. by Principal, at its sole discretion, in the event of adjudication of bankruptcy, request for judicial or extrajudicial reorganization, or yet dissolution or liquidation of Contractor;

c. by a judicial or arbitration decision that became final and unappealable;

d. by the innocent Party, in the event of an attempt to assign this Agreement without the consent of the other Parties, subject to the provisions of Clause 13.1 hereof.

11.1.4 This Agreement may also be terminated by operation of law and without prejudice to compensation for losses and damages to the innocent Party, in relation to one or more Infrastructure Items, in the following cases:

a. by Contractor, in relation to a given Infrastructure Item, in the event of (i) termination of the respective Lease Agreement that has not been caused by Contractor's contractual breach, or (ii) definitive expiration of the term of the Lease Agreement for the Infrastructure Item in question;

a. by Contractor, in relation to a given Infrastructure Item, if an Irreversible Situation is configured in such Infrastructure Item, provided that the exercise of the right of termination in relation to said Infrastructure Item will not remove or otherwise affect the Consideration Dilution; and/or

b. by Principal, in case the interference described in Clause 6.1.11 is not remedied within fifteen (15) Business Days of the notification in this sense sent to Contractor.

11.1.5 If this Agreement is terminated in its entirety or in relation to one or more Infrastructure Items due to Contractor's sole fault, the payment by Principal will not be due regarding any Consideration that would be due until the original expiration date provided for in Clause 17.1 of this Agreement, in which case the payment of any amounts referring to the period in which the lease and effective use of the space leased pursuant to this Agreement took place and which have not yet been paid.

11.1.6 When the termination affects only part of the Infrastructure Items covered by this Agreement, without compromising the maintenance of the legal relationship under the terms set forth herein in relation to the other Infrastructure Items, the effects of such termination will be partial, and the Agreement will remain valid and effective, where applicable, for the other Infrastructure Items not affected by the assumptions described above.

11.1.7 If Contractor becomes aware that an Infrastructure Item is at risk of being extinguished, Contractor shall notify Principal to jointly discuss the possibility of reallocating Principal's Equipment to another telecommunications site. Principal may, at its sole discretion, choose whether or not to relocate and, if the Parties agree upon such relocation, the terms and conditions of this Agreement will apply to the new site, by completing and signing a Form under the terms set forth herein, *mutatis mutandis*.

## **CLAUSE 12 - MISCELLANEOUS**

**Clause 12.1.** <u>Entire Agreement</u>. This Agreement revokes and fully replaces any instruments previously executed between the Parties with the same purpose contracted exclusively in relation to the Infrastructure Items indicated in <u>Exhibit 1</u> hereto, it being agreed that all agreements previously executed by the Parties or by the Parties and their Affiliates having as purpose other sites or infrastructure items will remain in full force and effect in accordance with their terms and conditions.

**Clause 12.2.** <u>Waiver</u>. Any tolerance and/or non-exercise, by either Party, of the its right or faculty by operation of this Agreement, or any agreement with any delay in fulfilling the obligations of the other Party, will not affect those rights or faculties that may be exercised, at any time, at the sole discretion of the Party that owns them, nor will it change the conditions set out in this Agreement, therefore not configuring novation.

**Clause 12.3.** <u>Damage</u>. Provided their fault and/or intent is proven, each Party will be exclusively and fully liable for compensation of any direct damages caused to the another Party, except in the event of Force Majeure or Acts of God, pursuant to article 393, sole paragraph, of the Civil Code. Neither Party will be liable to the other Party for any indirect damages, moral damages, lost profits, loss of business opportunity or commercial failures of the other Party.

12.3.1 Without prejudice to the foregoing, the Party that causes any damage, and its fault and/or intent is proven, to the facilities and Equipment of the other Party or third parties, including in the pre-installation, installation, operation and decommissioning phases, will

be solely and fully liable for compensation of such damages, limited to the replacement value of the equipment and facilities actually damaged, not being responsible for any other damage, loss or value.

12.3.2 The provisions in this Clause shall not impact, diminish, limit and/or impair, in any way, any of the indemnification obligations provided for in the Sale and Purchase Agreement, which remain in full force and effect under the terms agreed therein, nor shall they be interpreted, in any event, as novation, modification and/or alteration of said indemnification obligations provided for in the and Sale and Purchase Agreement.

**Clause 12.4.** <u>Acts of God and Force Majeure</u>. Acts of God or Force Majeure events shall be exempt from liability pursuant to Article 393 of the Civil Code.

12.4.1 The Party affected by an Act of God or Force Majeure shall immediately notify the other Party of the occurrence of the fact in question and the estimated period during which such Party will be unable to comply with or be required to delay the performance of its obligations under this Agreement.

12.4.2 Once the effects of Act of God or Force Majeure event have ceased, the affected Party shall immediately notify the other Party of this fact, restoring the original situation.

12.4.3 If the occurrence of an Act of God or Force Majeure only partially impairs the performance of any obligations under this Agreement by one of the Parties, the affected Party shall fulfill the obligations that have not been affected by the occurrence of such Act of God or Force Majeure.

**Clause 12.5.** <u>Employer Liability</u> Each Party is fully liable as the sole employer, and must therefore fulfill all its social and labor obligations, including salaries, social benefits, bonuses, social and social security charges, severance payments and indemnities and any other labor rights, as well as other expenses such as daily allowances, transportation, accommodation and meals for its employees or agents, with no jointly and several liability existing between the Parties in respect thereof.

**Clause 12.6.** <u>Access Procedure.</u> Without prejudice to the provisions of Clause 7.2 above, if the services, installations and/or works performed by Principal are proven to imply disobedience to the procedures for accessing Contractor's premises, to be previously and formally informed, or yet if they violate the security of persons or property of Contractor and/or third parties, then Contractor may order to undo, redo or suspend such services, installations or works, provided that, under no circumstances, Contractor's lack of inspection or pronouncement will relieve Principal from its responsibilities.

**Clause 12.7.** <u>Planning and Execution</u> The Parties will be responsible for planning and executing all activities that, by means of this Agreement or normative decisions of the competent Governmental Authority, are imposed on them, in order to properly protect the workers and/or employees of both Parties and/or against accidents, as well as to prevent damage to the property of the Parties and/or third parties.

**Clause 12.8.** <u>Repairs</u> Requests for any repairs covered under this Agreement must be included in the Operating Procedures and must be forwarded to Contractor through the procedures set out in <u>Exhibit 3</u>.

**Clause 12.9.** <u>Notices.</u> Notices, notifications, reports and other communications related to this Agreement shall be, unless specifically and expressly provided otherwise herein, made in writing and submitted in person, or sent by mail and/or e-mail, with proof of receipt, to the addresses and recipients provided for below, subject to the rules set out in Clause 18.1, *et seq*:

If to Contractor:

[=]

If to Principal:

[=]

**Clause 12.10.** <u>Other Contracting.</u> The provision of any other services using the Infrastructure Items described in <u>Exhibit 1</u> of this Agreement will be carried out by executing a specific agreement between the Parties, in compliance with the rules in force for the intended services.

**Clause 12.11.** <u>Liability</u> The Parties agree that Contractor shall not be liable for any interruption in the supply of electricity related to the Space Leases subject matter of this Agreement, which is the liability of the electricity utility company or supplier.

12.11.1 In any event, Contractor will not be liable for any interruption in the supply of electricity for the Space Leases in case of preventive maintenance previously informed to Principal, or in case of any interruption due to failures of the utility company responsible for supplying electricity for the Infrastructure Items.

**Clause 12.12.** The Parties represent that compliance with the quality parameters and obligations associated with the concessions, permissions or authorizations granted by the granting authority for the provision of the respective telecommunications services by Principal must not be compromised by the obligations agreed upon herein, always subject to the principles of strict good faith.

**Clause 12.13.** <u>Atypical Relationship</u> The Parties acknowledge that, for the regulation of this Agreement, the conditions freely agreed upon between the Parties herein will prevail, and the Parties expressly and irrevocably recognize that (i) the contractual review intended to reduce the price related to the monthly use of each Infrastructure Item established will not be allowed during the term of this Instrument; (ii) the renewal of this Agreement may take place under the terms of Clause 17.1.1 below. For all legal purposes, exclusively with regard to the provisions of items "(i)" and "(ii)" of this Clause, the Parties acknowledge that this leasing relationship will be considered atypical and the events of termination of this Agreement will be only those set forth in CLAUSE 11 above.

**Clause 12.14.** <u>Guarantee</u>: Guarantor enters into this Agreement as guarantor, joint and several debtor and principal payer of all obligations undertaken by Principal under this Agreement, and grant a personal guarantee in the form of a suretyship in favor of Contractor.

12.14.1 Guarantor's obligations undertaken herein will not be affected by any acts or omissions that could exempt Guarantor from its obligations or affect them, including, but not limited due to: (a) any extension of term or agreement between the Parties; (b) any novation or non-exercise of any right of Contractor; and (c) any limitation or disability, including arising from a request for extrajudicial reorganization, a request for judicial reorganization, intervention or bankruptcy.

12.14.2 Guarantor expressly waives, in an irrevocable and irreversible manner, the benefits of order, division, rights and exemption powers of any nature provided for in articles 333, sole paragraph, 365, 366, 368, 821, 824, 827, 834, 835, 837, 838 and 839, all of the Civil Code, and articles 130 and 794 of the Brazilian Code of Civil Procedure.

12.14.3 Guarantor hereby expressly waives the right to subrogate to the Contractor's credit rights, if it will honor, in whole or in part, the Principal's obligation under this Agreement.

12.14.4 This suretyship is provided by Guarantor on an irrevocable and irreversible basis, and it will enter into force on the date of execution of this Agreement, remaining valid in all its terms until full payment of all the obligations guaranteed by it hereunder and in accordance with Article 818 of the Civil Code.

12.14.5 Guarantor hereby acknowledges that the suretyship is provided for a term of two hundred and forty (240) months, which may be extended for another two hundred and forty (240) months in the event of beginning of foreclosure of the guarantees and therefore, article 835 of the Civil Code shall not apply.

12.14.6 In view of the suretyship provided herein, Principal undertakes to file this Agreement and any amendments thereto with the competent Titles and Documents Registry Offices within five (5) Business Days from the date of execution of this Agreement or within five (5) Business Days from the date of any such amendments, as the case may be.

**Clause 12.15.** <u>Merger of Principal</u>. Subject to the provisions of Clause 13.1.1 below, no provision in this Agreement shall be interpreted or construed as a limitation, express or implied, to a company merger, pursuant to article 223 of Law 6,404 of 1976, of Principal by Guarantor, provided that after the conclusion of such merger, all obligations of Principal under this Agreement will be succeeded by Guarantor.

## **CLAUSE 13 - ASSIGNMENT, TRANSFER AND SUBROGATION**

**Clause 13.1.** <u>Assignment, Transfer and Subrogation</u> Except as provided for in this Agreement, neither Party may assign or otherwise transfer, in whole or in part, this Agreement or any rights arising therefrom without the prior and express written consent of the other Party. However, regardless of prior or subsequent approval, upon mere notification, the assignment and/or transfer of the Agreement will be permitted to Affiliates, subject, in any event, to the applicable regulations, provided that (i) the assignee company has sufficient capacity to honor the obligations provided for in this Agreement, and (ii) assignor and assignee remain jointly and severally liable for complying with the obligations provided for herein, except in the event provided for in Clause 13.1.1 below.

13.1.1 The Parties acknowledge and agree that, in the event of a contribution by Principal of all or a substantial part of its mobile telephony assets to the share capital of other entity or entities to be subsequently sold as an isolated production unit, or otherwise of a transfer of said assets in preparation for sale thereof as an isolated production unit: (i) this Agreement shall form part of said isolated production unit and be assigned to the entity or entities that will receive part or all of Principal's mobile telephony assets, and (ii) upon conclusion of the sale of said entity or entities to a third party, as an isolated production unit, the obligations undertaken by Guarantor in this Agreement will be automatically extinguished.

13.1.2 The Parties agree that, in any case of change of Control, directly or indirectly, of Contractor or of Principal, this Agreement will remain in full force, especially regarding the obligations of Principal related to the Space Leases in the Infrastructure Items and to pay the Consideration and other amounts due under this Agreement, as well as regarding Contractor's obligation to lease spaces in the Infrastructure Items to Principal hereunder, including in the event that a change in Control arises from an eventual exclusion by guarantee creditors of Contractor's shares, provided that Contractor continues to have sufficient capacity to honor its obligations as provided for herein.

13.1.3 This Agreement binds the Parties for themselves and by their successors.

13.1.4 In case of transfer of Principal's grant for the provision of telecommunications services, the succeeding entity is subrogated in all rights and obligations undertaken in this Agreement.

# **CLAUSE 14 - ASSIGNMENT OF CREDIT RIGHTS**

**Clause 14.1.** <u>Assignment of Credit</u> Principal is fully aware of and agrees that Contractor may raise, in the financial and/or capital market, funds by assigning its credit rights arising from the Space Lease subject matter of this Agreement, in accordance with the provisions of article 286 of the Civil Code, provided that Contractor remains the unquestionable holder of the other rights and all obligations undertaken under this Agreement.

# **CLAUSE 15 - CONFIDENTIALITY**

**Clause 15.1.** <u>Confidentiality</u>. As of the date hereof and for a period of two (2) years from the date on which this Agreement is no longer in force for any reason, the Parties undertake to keep the secrecy and restricted the use to the performance of this Agreement of all information related to technical, operating, commercial, legal and financial aspects of the other Parties to which they have access as a result of the performance of this Agreement.

15.1.1 The confidentiality obligations set out above will not apply to any information disclosed by the disclosing Party, provided that such information:

a. is made available to the general public or has become, after its disclosure, part of the public domain through publication or by any other means, without any fault by the receiving Party in respect thereto;

b. becomes public domain, after receipt thereof by the receiving Party, for reasons not attributable to the receiving Party's actions or omissions;

c. was already known to the receiving Party prior to its disclosure by the disclosing Party, and therefore it has not been acquired, directly or indirectly, from the disclosing Party and/or any third party that is under no obligation of confidentiality to the disclosing Party;

d. was acquired, after disclosure thereof, in good faith, without any restriction of confidentiality, from a third party that is not bound by any confidentiality provision with the disclosing Party;

e. is no longer treated as confidential by the disclosing Party, as so communicated in writing by the disclosing Party; and/or

f. is required from the receiving Party by a Governmental Authority or any selfregulatory entity having jurisdiction over the Party (including, for the avoidance of doubt, any requests for information in the scope of due diligence, inspections or verifications carried out in the ordinary course of business, having as scope the activities and businesses of said Party), limited, in any case, to what is strictly necessary, provided that the receiving Party informs the disclosing Party in advance or immediately, if possible , of the need for such disclosure, so that the disclosing Party can take the applicable measures with the applicable Governmental Authority (or self-regulatory entity) that prevent disclosure of such information.

15.1.2 The Confidential Information related to this Agreement may be disclosed to Affiliates of the Parties and/or to investors, co-investors, creditors, potential buyers and managers of the Parties, to the strict extent of their need and provided that they are also previously required to guarantee the confidentiality of the information disclosed.

15.1.3 Contractor is hereby authorized to make a copy of this instrument available to financial institutions that participate in the fiduciary assignment process of credit rights, as set out in Clause 9.2.2, provided that the aforementioned financial institutions undertake to guarantee the confidentiality of this Agreement under the terms hereof.

## **CLAUSE 16 - INDEPENDENT PARTIES**

**Clause 16.1.** <u>Independence.</u> This Agreement will not create, under any circumstances, a partnership or commercial agency relationship between the Parties, or any relationship other than the lease agreed herein, and each Party is solely liable for its acts and obligations.

**Clause 16.2.** <u>Relationship</u>. Nothing in this Agreement may be interpreted so as to create (i) any corporate relationship between Contractor and Principal, and/or (ii) any employment relationship between the Parties or between a Party and the employees and contractors of the other Party.

## CLAUSE 17 - TERM

**Clause 17.1.** <u>Term of Effectiveness</u> The Agreement shall become effective as of its execution date and shall remain in force for a period of fifteen (15) years.

17.1.1 The term of this Agreement shall be renewed automatically and successively for equal periods of twenty-four (24) months, except when otherwise notified by any of the Parties, in writing, at least sixty (60) days prior to the expiration of such term or of its successive renewals.

## **CLAUSE 18 - NOTIFICATIONS**

**Clause 18.1.** <u>Notification Rules</u>. Any notices, notifications, reports and other communications related to this Agreement shall, unless specifically and expressly provided for in this Agreement, be made in writing and delivered in person or sent by mail, return receipt requested, being considered received on the date of delivery to such recipients, according to the addresses provided for in Clause 12.9.

**Clause 18.2.** <u>Deadline</u>. Communications between the Parties shall always be in writing and when verbal, for practical or urgent reasons, they must be confirmed in writing within five (5) days.

**Clause 18.3.** <u>Change of Address</u>. Each Party, through its legal representative, may designate, by letter sent to the other Party, new contacts and new addresses to replace those designated in Clause 12.9 above.

**Clause 18.4.** <u>Effect</u>. The notifications or communications referred to in this Clause, unless expressly provided otherwise in this Agreement, shall be deemed received, thus taking their effects on the date on which they are effectively and provenly received by their recipients, regardless of the date of dispatch by the sending Party or actual knowledge thereof by the receiving Party.

## CLAUSE 19 - ANTI-CORRUPTION REPRESENTATIONS AND WARRANTIES

**Clause 19.1.** <u>Representations and Warranties</u>. The Parties hereby represent to be aware of and to understand the terms of the Brazilian anti-corruption laws or of any other laws applicable to the purpose hereof, particularly the Foreign Corrupt Practices Act, - Act, 15 U.S.C. Paragraphs 78dd-1 *et seq.* - ("<u>FCPA</u>") of the United States of America ("<u>Anti-Corruption Rules</u>"), and undertake to abstain from any activity that may constitute a violation of the provisions of such Anti-Corruption Rules.

19.1.1 *Forbidden Payments*. The Parties, on their behalf and on behalf of their Affiliates, managers, officers, employees and agents, and any other Persons who may act on their behalf, undertake to conduct their commercial practices during the term of this Agreement in an ethical manner and in accordance with the applicable legal precepts. Upon performance of this Agreement, neither Party or their Affiliates nor any of their officers, employees, agents and any other Persons acting on their behalf shall give, offer, pay, promise to pay or authorize the payment, whether directly or indirectly, of any money or any valuable thing to any Governmental Authority, consultant, representative, partner or any other third party with the purposes of influencing any act or decision of any

Governmental Authority, or to assure any undue advantage, or to direct the businesses to any Person and/or which violates the Anti-Corruption Rules ("Forbidden Payment").

a. a Forbidden Payment does not include the payment of reasonable expenses incurred in good-faith, such as, for example, travel and hotel expenses, which are directly related to the promotion, explanation, demonstration or to products and services, or to performance of an agreement with a government or its agencies, provided that the payment is permitted by the applicable laws.

b. upon prior notice, each Party agrees that the other Party will have the right to carry out a due diligence procedure solely to ensure continued compliance with the representations and warranties provided by each Party in Clause 19.1.2 below. Each Party must fully cooperate in any due diligence carried out under this item.

19.1.2 *Additional Representations*. For the purposes of this Clause, each Party represents that:

a. it has not violated or is in violation of the Anti-Corruption Rules;

b. it has already implemented or undertakes to implement, during the term hereof, a compliance and training program reasonably efficient to prevent and detect violations of the Anti-Corruption Rules and the requirements set forth in this Clause; and

c. it is aware that any activity in violation of the Anti-Corruption Rules is prohibited and is aware of the possible consequences of such violation.

d. any breach of the Anti-Corruption Rules by any Party, in any of its aspects, will entail an obligation on the offending Party to pay the innocent Party for any direct losses and damages that the offending Party may cause.

# **CLAUSE 20 - LAW AND CONFLICT RESOLUTION**

**Clause 20.1.** Specific Performance; Enforceable Instrument. Each Party acknowledges that the remedies provided by law for non-compliance or threat of non-compliance with this Agreement may be inappropriate and, in acknowledging this fact, the Parties will have the right, without the need to provide security and in addition to any other remedies that may be available, to seek relief in the form of specific performance of their mandatory injunctions and/or obligations to pay, temporary restrictive measures, temporary or permanent remedy that may be available under the terms of articles 294 to 311, 497 *et. seq.* of Brazilian Code of Civil Procedure. Specific performance will not, under any circumstances, be limited and/or impaired by any provision of this Agreement. Each Party further agrees that this Agreement, which is duly signed by two (2) witnesses, is an extrajudicial enforceable instrument for the purposes of Article 784, III, of the Brazilian Code of Civil Procedure.

**Clause 20.2.** <u>Applicable Law</u>. This Agreement shall be governed by and construed pursuant according to the laws of the Federative Republic of Brazil.

Clause 20.3. <u>Arbitration</u>. Any disputes arising from this Agreement or otherwise related to it, including as to its existence, validity, effectiveness, interpretation of its terms,

conditions, execution or termination ("<u>Dispute</u>") shall be resolved by arbitration as provided for in this Clause 20.3 ("<u>Arbitration</u>").

20.3.1 The Parties agree that, before initiating Arbitration to resolve any Dispute, they shall attempt to negotiate an amicable settlement agreement for such Dispute, within a period not exceeding fifteen (15) Business Days after receipt by a Party of notification of the existence of such Dispute, as sent by the other Party. The Parties agree that their obligation to resolve any Disputes amicably is an ancillary obligation that does not preclude the immediate initiation of Arbitration at any time, at the sole and discretionary criteria of either Party.

20.3.2 At the end of this period or if, at the discretion of either Party, it is impossible to reach an amicable solution, the Interested Party may submit the Dispute to Arbitration before the International Chamber of Commerce (ICC) Arbitration Chamber ("<u>Chamber</u>"), in accordance with its arbitration regulation ("<u>Regulation</u>") in force on the request date for the establishment of Arbitration, with the exception of the changes provided for herein.

20.3.3 The Arbitration will be conducted by three (3) arbitrators ("<u>Arbitral Tribunal</u>"), one nominated by the requesting party and the other nominated by the requested party, in accordance with the Regulations. If there is more than one requesting party and/or more than one requested party, then the requesting parties and/or requested parties must together indicate their respective arbitrator. In the absence of agreement between the requesting parties or requested parties to indicate their respective co-arbitrator, all arbitrators must be appointed by the Chamber. The two arbitrators thus appointed shall nominate, by mutual agreement, the third arbitrator, who will act as president of the Arbitral Tribunal, within the term provided for in the Regulations. If any of the three arbitrators is not appointed within this period, then the Chamber shall appoint them, as provided for in the Regulation. Any and all controversies regarding the appointment of arbitrators by the parties, as well as the choice of the third arbitrator, will be settled by the Chamber. The Parties, by mutual agreement, remove the application of the provisions in the Regulation that limit the choice of a sole arbitrator, co-arbitrator or president of the arbitrat tribunal to the list of arbitrators of the Chamber.

20.3.4 The Arbitration will be based in the City of São Paulo, State of São Paulo, Brazil, where the arbitration award will be issued, and it will be conducted in the Portuguese language. The Arbitral Tribunal shall judge the merits of the Dispute in accordance with Brazilian law and shall not judge by equity.

20.3.5 Any order, decision, determination or award handed down by the Arbitral Tribunal will be final and binding on the parties and their successors, who expressly waive any appeal. The arbitral award may be enforced before any judicial authority with jurisdiction over the parties and/or their assets.

20.3.6 Each party will bear the costs and expenses that it causes in the course of the Arbitration and the Parties will share equally the costs and expenses whose cause cannot be attributed to one of them. The arbitral award will assign to the losing party, or to both parties to the extent that their claims are not upheld, the final liability for the cost of the proceeding, including the reimbursement of attorneys' and other advisers' fees of

reasonable value. The arbitration award shall not impose the payment of attorneys' fees due to loss of suit.

20.3.7 The parties choose the jurisdiction of the central courts of São Paulo, State of São Paulo, Brazil, to settle any issues arising out of this Agreement, and waive any other, however privileged, for the exclusive purposes of obtaining urgent relief to protect or safeguard their rights prior to the establishment of the Arbitral Tribunal, without this being considered a waiver of Arbitration. Any measure granted by the Judiciary Branch shall be promptly notified by the party that requested such measure to the Chamber. The Arbitral Tribunal, once constituted, may review, maintain or revoke the measures granted by the Judiciary Branch.

20.3.8 The parties agree that all aspects related to Arbitration, including its own existence, must be kept confidential. All its elements (including, but not limited to, the Parties' allegations, evidence, reports and other statements of third parties and any other documents presented or exchanged in the course of the arbitration proceeding) will only be disclosed to the Arbitral Tribunal, to the parties, their lawyers, Chamber officials, and any person needed to develop the Arbitration, unless disclosure is required to comply with any obligations imposed by applicable law, or by any Governmental Authority.

20.3.9 Any and all disputes related to the confidentiality obligation will be settled in a final and binding manner by the Arbitral Tribunal, which may take any measure to safeguard the confidentiality of the arbitration procedure, or any other matter related to Arbitration.

In the event that two or more disputes arise in relation to this Agreement, 20.3.10 or in any way related to it, its resolution may occur through a single arbitration proceeding, pursuant to the Regulation. Before the constitution of the Arbitral Tribunal, the Chamber will be responsible for consolidating the referred disputes in a single arbitral procedure, in accordance with the Regulation. After the constitution of the Arbitral Tribunal, in order to facilitate the resolution of related disputes, the Arbitral Tribunal may, at the request of one of the parties, consolidate the arbitration procedure with any other pending arbitration procedure that involves the resolution of disputes arising from this Agreement, or otherwise related to it. The Arbitral Tribunal shall consolidate the procedures, provided that (i) they involve the same parties; (ii) there are factual and/or legal issues that are common between them; and (iii) the consolidation in these circumstances does not result in losses arising from unjustified delays in resolving such disputes. The jurisdiction to determine the consolidation of procedures and conduct the consolidated procedure will be the first arbitral tribunal constituted. The consolidation decision will be final and binding on all parties involved in the disputes and arbitration procedures object of the consolidation order.

20.3.11 Guarantor is expressly bound by this arbitration clause for all legal purposes.

IN WITNESS WHEREOF, the Parties executed this Agreement in two (2) counterparts of equal content and form, in the presence of two (2) undersigned witnesses, for all legal purposes.

		ring Agreement ibit 1 - Infrastructure Items	5						
	Keycode 530684	RJTQR0354	-22,93127778	Longitude (decimaî) -43,37391667	State Municipality R] Rio De Janeiro	Estrada do Guerengué, 96	Type of Site Greenfield	Structure Type of Structure Pole	Height 30
	531174 531686 531712	SPWFO3737 SPCAO4938 RJL]R2002	-23,46780556 -23,87822	-46,70027778 -46,41487 -43,184782	SP Sao Paulo SP Cubatao R] Rio De Janeiro	ESTRADA DO SARÃO, Nº 456 - FREGUESIA DO Ó RUA JOSE GONÇALVES TORRES SN - CUBATÃO/SP Rua Álvaro Chaves, 44 - Larméiras	Greenfield Greenfield Roofton	Self-supporting Self-supporting Mast	40 50 5
S. M. M. M. A. M.	532142 532672	SPWMM3911 SPWPC3942	-23,5711	-46,661	SP Sao Paulo SP Sao Paulo	Av. Moorna, 81 - Planalto Pauliota, São Paulo - SP RUA MANOEL DA MATA SÁ 170	Rooftop Greenfield	Mast Self-supporting	40 (2,2+2,2+2,2) 60
A. Mode	532739 532850	SPWCM3694 SPWVF4163	-23,67365556 -23,56755556	-46,5475	SP Sao Paulo SP Sao Paulo	Rua Padro Rodrigaes Beig, 111 – Vila Marari RUA ANTONIO REIS N 11B – CHÁCARA BELENZINHO	Greenfield Greenfield	Self-supporting Self-supporting	40 (0),5 40 40
	533034 533379	SPWCM3695 SPWST3117	-23,6803 -23,623728	-46,6496 -46,608739	SP Sao Paulo SP Sao Paulo	RUA FERRUCIO SANDOLI 232 – SÃO PAULO Av. Roque Petroni Júnior, 1089 - Chácara Santo António (Zona Sal), São Paulo - SP	Greenfield Rooftop	Self-supporting Mast	40 40 10 40
C No.	533543 533567	SPWVD3048 SPWGR3743	-23,4941 -23,7554	-46,5874 -46,701	SP Sao Paulo SP Sao Paulo	Rua Oscar de Medeiros, 164, Vila Medeiros Rua DR. Antônio Simões de Carvalho,105 - Jd. São Bernardo -São Paulo - SP	Greenfield	Self-supporting Self-supporting	40 40 30
2 No. </td <td>579525 579606 579717</td> <td>S2CEFLA0114</td> <td>-3,739483</td> <td>-38,48196</td> <td></td> <td>Edificio San Lucas - Rua Prof. Silo Ribeiro, 420 Papica</td> <td></td> <td></td> <td>(3+3+1,5) 2,2 4</td>	579525 579606 579717	S2CEFLA0114	-3,739483	-38,48196		Edificio San Lucas - Rua Prof. Silo Ribeiro, 420 Papica			(3+3+1,5) 2,2 4
		CEMUL0416	-4,31039	-39,0023	CE. Mulungu	Rodovia CE-065 (trecho Mulungú-Aratuba), s/n - Morro da CETEL	Greenfield	Self-supporting	4 40 36
	647841 648707	MGURA4872 MGASL5930 PRPGO0516	-18,11226 -25,1275278	-47,82707 -44,25524 -50,1540556	MG Augusto De Lima PR Ponta Grossa	R. Geraldo Lopes Alvim, 314 RUA THAUMATURGO DE AZEVEDO - 82.	Greenfield Rooftop	Self-supporting Mast	70 40 6
S         S	649623 652212	PRCTA0667 PRCTA0546	-25,4219167 -25,3980556	-49,2595 -49,2444167	PR Curitiba PR Curitiba	RUA UBALDINO DO AMARAI, 37, ESTÁDIO COUTO PEREIRA AVENIDA PARANÁ 1758	Rooftop Rooftop	Mast Mast	40 (4,2+4,2+4,2) (3+3+3)
N <t< td=""><td>654565 655624</td><td>S2BABRR0019 MGBCA5006</td><td>-12,970916 -21,23776</td><td>-38,489489 -43,79795</td><td>BA Salvador MG Barbacena</td><td>RUA BERNARDO SPECTOR, 94 Morro monte Mário - s/nº</td><td>Greenfield Greenfield</td><td>Self-supporting Self-supporting</td><td>35 30 61,5</td></t<>	654565 655624	S2BABRR0019 MGBCA5006	-12,970916 -21,23776	-38,489489 -43,79795	BA Salvador MG Barbacena	RUA BERNARDO SPECTOR, 94 Morro monte Mário - s/nº	Greenfield Greenfield	Self-supporting Self-supporting	35 30 61,5
	657145 658575	PRMGA0556 BAIL10606	-23,39575 -14,763417	-51,91925 -39,063639	PR Maringa BA Ilheus	AV. DONA SOPHIA RASGULAEFF,466 AVENIDA PROCLAMACAO, S/N	Rooftop Greenfield	Mast Self-supporting	6 6 40
1 10. 10. 10.0 <th< td=""><td>658896 659280</td><td>PRORG0252 PRCPM0685</td><td>-24,2113889 -24,037</td><td>-50,9252778 -52,309</td><td>PR Ortigueira PR Campo Mourao</td><td>RUA SÃO PAULO, 99 Rua Laurindo Borges, 1398 - Centro, Campo Mourão - PR</td><td>Greenfield Rooftop</td><td>Self-supporting Mast</td><td>(4+4) 46 (4+4) 23</td></th<>	658896 659280	PRORG0252 PRCPM0685	-24,2113889 -24,037	-50,9252778 -52,309	PR Ortigueira PR Campo Mourao	RUA SÃO PAULO, 99 Rua Laurindo Borges, 1398 - Centro, Campo Mourão - PR	Greenfield Rooftop	Self-supporting Mast	(4+4) 46 (4+4) 23
	660433 660434	PRFOZ0722 PRFOZ0706	-25,53894 -25,53797	-54,57558 -54,58844	PR Foz Do Iguacu PR Foz Do Iguacu	Run Rui Barbooa, 786 Run Rui Barbooa, 475	Rooftop Rooftop	Mast Mast	23 (4+4) (5+5) (6+4+4+4+4)
No.         No. <td>661251</td> <td>BABES1110</td> <td>-12,14975</td> <td>-44,99713889 -51,1653</td> <td>BA Barreiras PR Londrina</td> <td>Av. Higenopolis, 70 - Jardim Higenopolis</td> <td>Greenfield</td> <td>Self-supporting Mast</td> <td>(0+4+4+4+4) 30 (6+6+6) 36</td>	661251	BABES1110	-12,14975	-44,99713889 -51,1653	BA Barreiras PR Londrina	Av. Higenopolis, 70 - Jardim Higenopolis	Greenfield	Self-supporting Mast	(0+4+4+4+4) 30 (6+6+6) 36
Control         Non- Control         Non- Contro         Non- Control         Non- Control	662983 878570 879478	S2BAASG0093	-12,980628	-49,2610611	PR Curitiba BA Salvador	Rua Mariano Torres, 916 RUA DA MOURARIA, 6	Rooftop	Mast Mast	(4+4) 3,5 (7+7+7)
Desc         Desc <thdesc< th="">         Desc         Desc         <thd< td=""><td>7140610</td><td>PRSLF0788 ALPYH2286</td><td>-26,1766944 -9,622775</td><td>-37,767747</td><td>PR Salgado Filho AL Piranhas</td><td>Rus Ruy Barbosa, s/n" AVENIDA SERGIPE, S/N</td><td>Greenfield Greenfield</td><td>Self-supporting Self-supporting</td><td>50 60 (5+5+5)</td></thd<></thdesc<>	7140610	PRSLF0788 ALPYH2286	-26,1766944 -9,622775	-37,767747	PR Salgado Filho AL Piranhas	Rus Ruy Barbosa, s/n" AVENIDA SERGIPE, S/N	Greenfield Greenfield	Self-supporting Self-supporting	50 60 (5+5+5)
	7140711 7140780 7141039	PEESP3020 PERCE3244 SPEAC4955	-8,034944444 -8,065225	-43,892025 -34,873444 -47,40859	PE Recife PE Recife	AV GOV. AGAMENON MAGALHÅES 2656-FSPINHEIRO RUA CAIS DA ALFÅNDEGA, S/N – SHOPPING ALFÅNDEGA	Rooftop Rooftop Rooftop	Mast Mast Mast	(4+4) 3 (3,02+3,02+3,02)
	7141041 7141226 7141263	SPFAC4958 SPWCS3728 SPWAA3576	-20,529 -23,636 -23,534	-47,392 -46,614 -46,484	SP Franca SP Sao Paulo	Av. Presidente Vargas 666 – Calade Nova, Franca - SP Rua Girolamo Dai Libri. 433 - Vih Moraes	Rooftop Rooftop Rooftop	Mast Mast Mast	(5+5+5) 10 6
No.       No.     No.     No.     No.     No.     No.     No.     No.     No.       No.     No.     No.     No.     No.     No.     No.     No.     No.       No.     No.     No.     No.     No.     No.     No.     No.       No.     No.     No.     No.     No.     No.     No.     No.       No.     No.     No.     No.     No.     No.     No.     No.       No.     No.     No.     No.     No.     No.     No.     No.       No.     No.     No.     No.     No.     No.     No.     No.       No.     No.     No.     No.     No.     No.     No.     No.       No.     No.     No.     No.     No.     No.     No.     No.       No.     No.     No.     No.     No.     No.     No.     No.       No.     No.     No.     No.     No.     No.     No.     No.       No.     No.     No.     No.     No.     No.     No. <t< td=""><td>7141286 7141345 7141350</td><td>SPOCO4536 SPSBO4598 SPWBE3603</td><td>-23,549855 -23,701033 -23,54937</td><td>-46,760417 -46,555151 -46,58456</td><td>SP Oxasco SP Sao Berrardo Do Campo SP Sao Paulo</td><td>Ruz Ruy Barbonz, 260 RUA JOAO BATISTA DE LACERDA, 261</td><td>Rooftop Rooftop Rooftop</td><td>Mast Mast Mast</td><td>6 (5+5+5) (4+4+1)</td></t<>	7141286 7141345 7141350	SPOCO4536 SPSBO4598 SPWBE3603	-23,549855 -23,701033 -23,54937	-46,760417 -46,555151 -46,58456	SP Oxasco SP Sao Berrardo Do Campo SP Sao Paulo	Ruz Ruy Barbonz, 260 RUA JOAO BATISTA DE LACERDA, 261	Rooftop Rooftop Rooftop	Mast Mast Mast	6 (5+5+5) (4+4+1)
Desc.         Desc. <th< td=""><td>7141378 7141379</td><td>SPWBF3583 SPWBF3585</td><td>-23,531 -23,517</td><td>-46,666 -46,677</td><td>SP Sao Paulo SP Sao Paulo</td><td>Av. Francisco Matarazzo, 229 - Água Branca, São Paulo - SP Rua Josef Kryss, 319 - Parque Industrial Tomas Edson</td><td>Rooftop Rooftop</td><td>Mast Mast</td><td>(4+4+4) (5+5+5) (6+6+6+3)</td></th<>	7141378 7141379	SPWBF3583 SPWBF3585	-23,531 -23,517	-46,666 -46,677	SP Sao Paulo SP Sao Paulo	Av. Francisco Matarazzo, 229 - Água Branca, São Paulo - SP Rua Josef Kryss, 319 - Parque Industrial Tomas Edson	Rooftop Rooftop	Mast Mast	(4+4+4) (5+5+5) (6+6+6+3)
No.         No. <td>7141466 7141473</td> <td>SPSBO4630 SPSCN4637 SPWBV3589</td> <td>-23,726927 -23,619705 -23,56009</td> <td>-46,574405 -46,65031</td> <td>SP Sao Bernardo Do Campo SP Sao Cartano Do Sal SP Sao Paulo</td> <td>Rua Nelza Ascincio, 128 – Ferrarolyonia – São Bernado do Campo Rua Archinto Ferrari, 118 – Santo Antônio – São Caetano do Sul AVENIDA PAULISTA, 810</td> <td>Rooftop Rooftop</td> <td>Mast Mast Mast</td> <td>(4+10) (4+4+3) (4+4+4)</td>	7141466 7141473	SPSBO4630 SPSCN4637 SPWBV3589	-23,726927 -23,619705 -23,56009	-46,574405 -46,65031	SP Sao Bernardo Do Campo SP Sao Cartano Do Sal SP Sao Paulo	Rua Nelza Ascincio, 128 – Ferrarolyonia – São Bernado do Campo Rua Archinto Ferrari, 118 – Santo Antônio – São Caetano do Sul AVENIDA PAULISTA, 810	Rooftop Rooftop	Mast Mast Mast	(4+10) (4+4+3) (4+4+4)
	7141532 7141620 7141726	SPWCB3641 SPMMM5169 SPRPO5341	-22,439791	-46,66083 -46,967025 -47,80719	SP Mogi Mirim SP Ribeirao Preto	Rus Ribeiroles, 274 – Vila Paulista Rus Coronel Venáncio Ferneira Alves Adorno, 89 – Saúde	Rooftop Rooftop Rooftop	Mast Mast Mast	(4+4+4) (6+6+6+2) (3+5+5)
No.         No. <td>7141806</td> <td>SPS[C5460</td> <td>-22,7586 -23,19207 -24,013</td> <td>-45,89044</td> <td>SP Sao Jose Dos Campos</td> <td>Run Fundides Miraguia, 26 - Centro</td> <td>Rooftop</td> <td>Mast Mast</td> <td>(4+4+4) 1,5 (6+6+6)</td>	7141806	SPS[C5460	-22,7586 -23,19207 -24,013	-45,89044	SP Sao Jose Dos Campos	Run Fundides Miraguia, 26 - Centro	Rooftop	Mast Mast	(4+4+4) 1,5 (6+6+6)
No.   No.	7141905	SPSTS5379 SPSTS5396	-23,949	-46,346 -46,33	SP Santos SP Santos	Rus Saturnino de Brito, 256 – Marapé – Santos Av. Ana Costa, 121 - Vila Matias	Rooftop	Mast Mast	(4+4+4+4) (3+3+6) (3+3+3+3)
	7141970 7142017	SPSUM5601 S2BAIGW0058	-22,84806 -12,9948889	-47,18647 -38,4708889	SP Surrare BA Salvador	Rua João Antônio Soares, 18 – Jd. Bom Retiro - Sumaré RUA RUBENS GUELLI, 134	Rooftop Rooftop	Mast Mast	(3+3+3) 8 10
	7142202 7142381	S2BAGGG1233 BAROL0440	-14,32689 -11,4861111	-39,46868 -37,9313889	BA Gongogi BA Rio Real	Rua da Igreja, S/N – Centro RUA CORONEL BENVENUTO, S/N	Greenfield Greenfield	Self-supporting Self-supporting	30 40 70
No.     No. </td <td>7142403 7142448</td> <td>SPWTA4138 SPWPZ5963</td> <td>-23,54583</td> <td>-46,556094 -46,67489</td> <td>SP Sao Paulo SP Sao Paulo</td> <td>Rua Dr. Homem de Melo, 1186 - Perdizes</td> <td>Rooftop Rooftop</td> <td>Mast Mast</td> <td>10 (3+3+3) (3+3+3+4+4) (4+4+4)</td>	7142403 7142448	SPWTA4138 SPWPZ5963	-23,54583	-46,556094 -46,67489	SP Sao Paulo SP Sao Paulo	Rua Dr. Homem de Melo, 1186 - Perdizes	Rooftop Rooftop	Mast Mast	10 (3+3+3) (3+3+3+4+4) (4+4+4)
	7142464	SPWJQ3800	-23,6589 -23,652948 -73,656	-46,6462	SP Sao Paulo	Rua das Oricicas, 149 – Pa, Jabacuara - São Paulo - SP	Rooftop	Mast	(8+8) (3+3+3) (4+4+4)
b)     Disk	7142473 7142476	SPWRE4007 SPWRE4010	-23,536667 -23,54105	-46,645579 -46,636398	SP Sao Paulo SP Sao Paulo	Alameda Barão de Limeira, 478 - Campos Elíseos, São Paulo - SP Avenida Cásper Libero, 58 - Centro Hitórico de São Paulo	Rooftop Rooftop	Mast Mast	(6+6+6+3) (4+4+4) (4+4+4+4)
No.     Appendix     Appendix<	7142514 7142554	SPWVJ4178 SPWLB3880	-23,502 -23,55867	-46,45 -46,63043	SP Sao Paulo SP Sao Paulo	Rus Édipo Feliciano, 100 - Vila Jacui, São Paulo – SP Rua do Glicério, 731 – Liberdade – São Paulo	Rooftop Rooftop	Mast Mast	(3+3+3+3) (4+4+4) (4+4+4)
	7142556 7142570	SPWLB3882 SPWVM4207	-23,57@4 -23,58472 -23,5788	-46,64028 -46,61861	SP Sao Paulo SP Sao Paulo	Rus Venyueiro, 1661 – Vila Mariana Rus Miriam, 310 – Jardim da Gloria	Rooftop Rooftop	Mast Mast	(3+3+3) (4+4+4) (3+3+3)
	7142587	SPWLP3870 SPWSC4040	-23,53 -23,534	-46,696 -46,655	SP Sao Paulo SP Sao Paulo	Ruz Marcelina, 394 – Vila Romana - SP Rus Albuquerque Lins, 374 - Santa Cecilia, São Paulo - SP	Rooftop Rooftop	Mast Mast	(3+3+3+6+6+6) (6+6+6) (4+4+4+4)
Desc         Desc <thdesc< th="">         Desc         Desc         <thd< td=""><td>7142601 7142631</td><td>SPWVM4216 SPWSE4123</td><td>-23,598 -23,54796</td><td>-46,634 -46,631055</td><td>SP Sao Paulo SP Sao Paulo</td><td>R. Joel Jorge de Melo, 62 - Vila Mariana Rua 25 de março, 197 - Centro Historico de São Paulo</td><td>Rooftop Rooftop</td><td>Mast Mast</td><td>4 (4+4+4) (6+6+3)</td></thd<></thdesc<>	7142601 7142631	SPWVM4216 SPWSE4123	-23,598 -23,54796	-46,634 -46,631055	SP Sao Paulo SP Sao Paulo	R. Joel Jorge de Melo, 62 - Vila Mariana Rua 25 de março, 197 - Centro Historico de São Paulo	Rooftop Rooftop	Mast Mast	4 (4+4+4) (6+6+3)
Description         Appendix	7142645 7142647	SPWMM3898 SPWMM3901	-23,575465 -23,611	-46,656397 -46,668	SP Sao Paulo SP Sao Paulo	RUA SALTO, 70 – MOEMA Av. dos Imarés, 88 - Indianópolis, São Paulo - SP	Rooftop Rooftop	Mast Mast	(6+6+6) (3+4+4+4) (10+10+6) (3+3+3)
10.     10.00     2000	7142669	SPWIB3768	-23,59734 -23,596901	-46,66923 -46,678674		Rus Findeiras, 479 – Itaim Bhi	Rooftop	Mast	(6+9) (6+6+6)
i         book         product         status         status         base	7142710 7142763	SPWSN4057	-23,4956 -23,63194	-46,6255 -46,71083	SP Sao Paulo	Rua Voluntários da Pátria, 2798 - Santana, São Paulo - SP	Rooftop Rooftop Rooftop	Mast	(4+4+4+3) (3+3+3+6) (4+4+4) (4+4+4)
111 <th< td=""><td>7142812</td><td>SPWSU4117</td><td>-23,612922</td><td>-46,638404</td><td>SP Sao Paulo</td><td>Av. Jahaquara, 970 – Saide – São Paulo QUDRA 210, CONJ. 1, CASA 17 - SANTA MARIA</td><td>Rooftop Rooftop Rooftop</td><td>Mast</td><td>3 5 (1+1+0</td></th<>	7142812	SPWSU4117	-23,612922	-46,638404	SP Sao Paulo	Av. Jahaquara, 970 – Saide – São Paulo QUDRA 210, CONJ. 1, CASA 17 - SANTA MARIA	Rooftop Rooftop Rooftop	Mast	3 5 (1+1+0
No.         No. <td></td> <td>SPWSU3522</td> <td>-23,60914</td> <td>-46,65231</td> <td>SP Sao Paulo</td> <td>Av. Angelica, 1683 - Santa Cecilia, São Paulo - SP Avenida Indianópolis, 1192 - (Clabe Sirio Libanés) - Planalto Paulista, São Paulo</td> <td>Rootiop</td> <td>Mast</td> <td>(6,75+5,4) (3+3+3) 56,27</td>		SPWSU3522	-23,60914	-46,65231	SP Sao Paulo	Av. Angelica, 1683 - Santa Cecilia, São Paulo - SP Avenida Indianópolis, 1192 - (Clabe Sirio Libanés) - Planalto Paulista, São Paulo	Rootiop	Mast	(6,75+5,4) (3+3+3) 56,27
1         1001         10000         1000         10000	7144692 7145243	PERCE3073 SCFNS0458 SCFNS0500	-8,056138889 -27,58991 -27,5913	-34,8725 -48,51513	PE Recife SC Piorianopolis	RUA BRUM 463/485 RECIFE ANTIGO Av. Madre Bervenutz. 687	Rooftop Rooftop	Mast Mast Mast	4 (2,5+2,5) (4+4+4)
Nome         Nome <th< td=""><td>7145310</td><td>SCPRI0545 SCRCP0546</td><td>-26,94797</td><td>-50,14</td><td>SC Princesa SC Rio Do Campo</td><td>Rus Tiradentes, s/nº, Centro</td><td>Greenfield</td><td>Self-supporting Self-supporting</td><td>50 50 50</td></th<>	7145310	SCPRI0545 SCRCP0546	-26,94797	-50,14	SC Princesa SC Rio Do Campo	Rus Tiradentes, s/nº, Centro	Greenfield	Self-supporting Self-supporting	50 50 50
Desc         Desc <thdesc< th="">         Desc         Desc         <thd< td=""><td>7145498</td><td>3GSCFNS0405</td><td>-27,591944</td><td>-48,554444</td><td>SC Florianopolis</td><td>Av. Rio Branco, 369</td><td></td><td></td><td>4 60 5,2</td></thd<></thdesc<>	7145498	3GSCFNS0405	-27,591944	-48,554444	SC Florianopolis	Av. Rio Branco, 369			4 60 5,2
1         No.00         No.	7145669 7146168 7146296	SCFNS0340 PRFOZ0721 PRCTA0545	-27,61175 -25,50586 -25,4520556	-48,5934167 -54,39086 -49,2535	SC Florianopolis PR Foz Do Iguacu PR Curitiba	Rua Antenor Morais, nº 42 Rua Carlos Soutromayor, 393 RUA JOQUEI CLUBB, 437	Rooftop Greenfield Rooftop	Mast Self-supporting Mast	(4+4+4) 15 7,5
1         1.110         1.0701         0.0701         0.0         0.0001	7146310 7146357	PRCTA0053 PRCTA0734 CEFLA0191	-25,407857 -25,429(0278 -3,73354	-49,2554125 -49,2814722 -38,49642	PR Caritha PR Caritha CE Fortaleza	Run Belérn, 43 Run Visc. do Bernero, 803 Aveniak Washington Scarco, nº 55	Rooftop Rooftop Rooftop	Mast Mast Mast	(3+3) 6 1,46
1         1	7147385	BAPXD 4536 BACIS1208	-10,09805 -12,73811	-38,04745 -38,44071	BA Pedro Alexandre BA Candeias	Estrada Para Barreiras, S/N Rua Alto da Boa Vista	Greenfield	Self-supporting Self-supporting	(4+4+4) 30 18,1
1         10.000         LO.200         20.001         20.001         20.001         Additional and a state of the state of	7148941 7148988	3GMGGVS5325 S2RSPAE0651	-18,87185 -29,979376	-41,95713 -51,189987	MG Governador Valadares RS Porto Alegre	Av. JK, 1157 RUA PROF. JOÃO DE SOUSA RIBEIRO, 859	Rooftop Rooftop	Mast Mast	9,2 (2,5+2+1,5+3+1,5+2,4 1,77
88962 </td <td>7150716 7150884</td> <td>BACXS4693 R[BGT0254 3GPRSIV0857</td> <td>-12,43817 -23,025 -22,700312</td> <td>-37,91875 -43,556 -51,792436</td> <td>BA Mata De Sao Joao R] Rio De Janeiro PR Santo Inacio</td> <td>KM 76 - COMPLEXO COSTA DO SAUIPE Estrada Roberto Burle Mars, 2467 - Barra Guaraba RUA APRÍCIO ALVES FILHO, 831</td> <td>Rooftop Greenfield</td> <td>Mast Mast Self-supporting</td> <td>5 10 41 30</td>	7150716 7150884	BACXS4693 R[BGT0254 3GPRSIV0857	-12,43817 -23,025 -22,700312	-37,91875 -43,556 -51,792436	BA Mata De Sao Joao R] Rio De Janeiro PR Santo Inacio	KM 76 - COMPLEXO COSTA DO SAUIPE Estrada Roberto Burle Mars, 2467 - Barra Guaraba RUA APRÍCIO ALVES FILHO, 831	Rooftop Greenfield	Mast Mast Self-supporting	5 10 41 30
1         1058         2011,00         1059         2011,00 <td>660232 648290</td> <td>PRCPP3519 MGJFA5020</td> <td>-23,1832222 -21,76208</td> <td>-50,6573056 -43,34952</td> <td>PR Cornelio Procopio MG Juiz De Fora</td> <td>AV. NOSSA SENHORA DO ROCIO.966 Av. Bario do Ros Branco - 2288 Rus Grande. nº 400</td> <td>Rooftop Rooftop</td> <td>Mast Mast</td> <td>30 (6+6+6) (5+1,60+3) (3+3+3)</td>	660232 648290	PRCPP3519 MGJFA5020	-23,1832222 -21,76208	-50,6573056 -43,34952	PR Cornelio Procopio MG Juiz De Fora	AV. NOSSA SENHORA DO ROCIO.966 Av. Bario do Ros Branco - 2288 Rus Grande. nº 400	Rooftop Rooftop	Mast Mast	30 (6+6+6) (5+1,60+3) (3+3+3)
1         1000         MG0101         3.4850         4.2801         0.0         10 International Activity State Activity         Internativity State Activity State Activity Activity Activity Activity	578305 657208	S2CEFLA0188 PRCTA0658	-3,7327778 -25,4435556	-38,4986111 -49,2787222	CE Fortaleza PR Curitiba	AVENIDA LUIS, 300 AVENIDA SILVA [ARDIM, 1856	Rooftop Rooftop	Mast Mast	(5+3+3) 4 (5,5+4,5) (4+1,80+1,53)
4         1010         AUREDS         AURD         20000         AURD         PLACEMENT LICEN LATURAL DE LANAURE         Conclud         September           1         1010         AUREDS         AURD         AURD         PLACEMENT LICEN LATURAL DE LANAURE         Conclud         September           1         1052         ATHAD"         ATHAD         ATHAD         ATHAD"         AURD         BAR AURD         Restaure Res	578327 510938	MASLS1557 ALMCO2363	-2,540556 -9,672061	-44,286944 -35,745839	AL Macrio	Av. Senador Vitorino Freire, nº 1958 – Bairro Arcinha AV ASSIS CHATEAUBRIAND, 2998	Rooftop Rooftop	Mast Mast	(3+3+3) (5+5+5+5+5) (3+3+3) 16
11002SURLAUPSURLAUPSURLAUPSURLAUPSURLAUPNumberNumberNumber11002SURLAUPSURLAUPSURLAUPSURLAUPSURLAUPNumber <td< td=""><td>511084 878726 658976</td><td>ALOUB2295 MGSAV4348 RSSAN0741</td><td>-9,162167 -19,92969 -29,54889</td><td>-37,359192 -43,93725 -56,82975</td><td>AL Ouro Branco MG Belo Horizonte RS Uraguaiana</td><td>RUA CORONEL LUCENA, S/N LUIZ G. DE CARVALHO Av. Join Patheiro, 405 RB 472 KM 5405</td><td>Greenfield Rooftop Greenfield</td><td>Self-supporting Mast Self-supporting</td><td>40 (5+2,50+2,50) 60</td></td<>	511084 878726 658976	ALOUB2295 MGSAV4348 RSSAN0741	-9,162167 -19,92969 -29,54889	-37,359192 -43,93725 -56,82975	AL Ouro Branco MG Belo Horizonte RS Uraguaiana	RUA CORONEL LUCENA, S/N LUIZ G. DE CARVALHO Av. Join Patheiro, 405 RB 472 KM 5405	Greenfield Rooftop Greenfield	Self-supporting Mast Self-supporting	40 (5+2,50+2,50) 60
1         Disk         Disk         Quark         Disk         Quark         Disk         Quark         Disk         Disk <thdisk< th="">         Disk         Disk         <thd< td=""><td>7146622 7145782 7144305</td><td>S2CEFLA0074 PRCTA0523 DFBSA0236</td><td>-3,724462 -25,432907 -15,819</td><td>-38,528741 -49,299796 -47,875</td><td>CE Fortaleza PR Curitika DF Brasilia</td><td>Rus Senador Pompeu, 649 RUA PADRE ANCHIETA, 2128 SGES trecho 2 Jos 32</td><td>Rooftop Rooftop Rooftop</td><td>Mast Mast Mast</td><td>7,5 (6+3+6+3) (4+6+2,3)</td></thd<></thdisk<>	7146622 7145782 7144305	S2CEFLA0074 PRCTA0523 DFBSA0236	-3,724462 -25,432907 -15,819	-38,528741 -49,299796 -47,875	CE Fortaleza PR Curitika DF Brasilia	Rus Senador Pompeu, 649 RUA PADRE ANCHIETA, 2128 SGES trecho 2 Jos 32	Rooftop Rooftop Rooftop	Mast Mast Mast	7,5 (6+3+6+3) (4+6+2,3)
Storie         SPREND         34,000         45,001         9         ks. Pub.         BLED         BLED         General Mathematication           11010         SPREND         34,007         45,007         39         ks. Pub.         BLED	7147166 7147245 7146650	DF8AM0754 DF8AM0753 \$2RSPAE0850	-15,87916667 -15,86166667 -30,101	-48,08861111 -51,2678	DF Brasilia DF Brasilia RS Porto Alegre	QR 411CONJ. 4, CASA 14, SAMAMBÁIA OS694, CONJ. B (CANDIDADTO B)	Rooftop Rooftop	Gured Bracket Mast	17 17 6
Intel         OPRIADS         -3.0000         -4.001         -9         Intelnability Corp.         Intelnability Made, 11. Vol Manda,         Node         Material           11010         OPRIADS         -3.0000         -4.0011         -4.001         -9         Intelnability Made, 11. Vol Manda,         Node         Material           11020         OPRIADS         -3.0000         -9         Intelnability Made, 11. Vol Manda,         Node         Material           11020         OPRIADS         -3.0000         -9         Intelnability Made, 10. Vol Manda,         Node         Material           11020         OPRIADS         -3.0000         -9         Intelnability Made, 10. Vol Manda,         Node         Material           11020         OPRIADS         -3.0000         -9         Intelnability Made, 10. Vol Manda,         Node         Material           11020         OPRIADS         -3.0001         -4.000         -9         Intelnability Material         Node         Material           11020         OPRIADS         -3.0011         -4.002         -9         Intelnability Material         Node         Material           11021         OPRIADS         -3.0011         -4.002         -9         Intelnability Material         Node         Material	532890 533333	SPWPS3967 SPWLM3890	-23,42997 -23,50641	-46,75923 -46,67416	SP Sao Paulo SP Sao Paulo	RUA RIO JUNDIAI Nº 157 – PERUS.	Greenfield Greenfield Greenfield	Self-supporting Pole	40 60 40,6 (3+3+3)
Trichon         SPENDED         Statility         40,053         9         No Pair         Name         Name         Name           Trichon         SPENDED         Statility         40,053         9         No Pair         Name         Name <t< td=""><td>7141374 7142393</td><td>SPWC03717 SPSB04605 SPWJR3818</td><td>-22,55527 -23,660059 -23,51361</td><td>-46,665 -46,569171 -46,74555</td><td>SP         Sao Paulo           SP         Sao Bernardo Do Campo           SP         Sao Paulo</td><td>Rua Marechal Badoglio, 15 – Vila Mussoline Rua Conselheiro Amaral, 196 - Jaguará, São Paulo - SP</td><td>Rooftop Rooftop</td><td>Mast Mast Mast</td><td>(4+4+4+4+4+4) (3+3+3)</td></t<>	7141374 7142393	SPWC03717 SPSB04605 SPWJR3818	-22,55527 -23,660059 -23,51361	-46,665 -46,569171 -46,74555	SP         Sao Paulo           SP         Sao Bernardo Do Campo           SP         Sao Paulo	Rua Marechal Badoglio, 15 – Vila Mussoline Rua Conselheiro Amaral, 196 - Jaguará, São Paulo - SP	Rooftop Rooftop	Mast Mast Mast	(4+4+4+4+4+4) (3+3+3)
1         11/27         3/97/K07         2.5540         4.005         69         has had.         Bas hadas         Bas hadas <th< td=""><td>7142396 7142401 7142498</td><td>SPWPN3987 SPWTA4136 SPWRE4012</td><td>-23,58111 -23,53055556 -23,537794</td><td>-46,6925 -46,55916667 -46,636868</td><td>SP Sao Paulo SP Sao Paulo SP Sao Paulo</td><td>Run Hurgrin, 664 - Lundins, Sio Paulo - SP RUA GEN, GELÁS, 72 - TATUAPÉ, SÃO PAULO-SP - 05090-050 Run Wohintoru Luis, 392 - Cantro Histórico - São Paulo</td><td>Rooftop Rooftop Rooftop</td><td>Mast Mast Mast</td><td>(4+4+6) (4+4+4+4+4) (4+4+6)</td></th<>	7142396 7142401 7142498	SPWPN3987 SPWTA4136 SPWRE4012	-23,58111 -23,53055556 -23,537794	-46,6925 -46,55916667 -46,636868	SP Sao Paulo SP Sao Paulo SP Sao Paulo	Run Hurgrin, 664 - Lundins, Sio Paulo - SP RUA GEN, GELÁS, 72 - TATUAPÉ, SÃO PAULO-SP - 05090-050 Run Wohintoru Luis, 392 - Cantro Histórico - São Paulo	Rooftop Rooftop Rooftop	Mast Mast Mast	(4+4+6) (4+4+4+4+4) (4+4+6)
Theory         SPANDS54         Status         Adda2         SP         See Pair         Readed by Team Pairs         Reader         Mader         Mader           11400         CV1000         20010         40.0021         9         See Pair         The book Pairs         Reader         <	7142672 7142778	SPWPN3972	-23,614 -23,55416	-46,662 -46,6925	SP Sao Paulo SP Sao Paulo SP Sao Paulo	Prace Jolin Pretex, 30 Campos Hiscos, Sio Paulo - SP Alameda dos Tupiniquins, 1235 - Planalo Paulosta Rui Fraidgus Contribo, 1500 - Pinheirus - Sio Paulo - SP	Rooftop	Mast Mast	(3+3+3) (5+5+5) (3+3+3)
Status         PAPPYN         2.500000         -0.4400000         PA         P	7144206 7144293	SPWVM3099 SPWLB3069	-23,57444	-43,653 -46,632733	SP Sao Paulo SP Sao Paulo	Av. Beigudeiro Luis António, 2819 – Jardim Paulista Rua Topizio, 420 – Vin Mariana	Rooftop	Mast Mast	(3+3+3+3+3) (6+3) (4+4+4+3)
1         16400         PRC1A010         35,4530         Ap/(15)         PR         Conduct         Red         Adapt         Max         Adapt         Max           1         0450         PRC1A010         35,4530         Ap/(15)         PR         Conduct         Adapt         Max         Adapt         Max         Adapt         Max	521163 521528 7144021	PAIPP2070 PABLM2019 PABLM2136	-1,41111111 -1,40722222	-48,44083333 -48,44527778	PA Ipicuna Do Para PA Belem PA Belem	Rua Sangento Simplicio, 371 Rua Santo António Av. Tavares Bastos, 632 - Marambaia	Greenfield Greenfield Greenfield	Self-supporting Pole Pole	80,29 44,79 37,91
5         55006         974X511         22,5206         -12,009         52         Imm.         Ban De for Combining Am         Generation Systems           6         53305         97105109         -22,54052         -67,55472         58         Sam Dai         Ban De for Combining Am         End Generation Systems         End Sam Dai         End Sam Dai<	660280 7146242	PRCTA0143 PRCTA0422	-25,43077867 -25,4318749	-49,25804193 -48,27251823	PR Curitha PR Curitha	Rod. Be 277 Cta - Orleans / Rod. do Café - Catade Industrial De Cunitba Rua da Paz, 202 Alameda Dr. Mariey, 650	Rooftop Rooftop	Mast Mast Mast	(3+3) 5 (3+3+3) (1+2+3)
88 7144274 SPWC05715 -23,553 -46,6509 SP Sao Paulo Rua Reh Cnita, 450 Rooftop Mast	533008 533395	SPLRA5117 SPWJE3059	-25,49939318 -22,54296 -23,54334722	-47,38998 -46,75434722	SP Limeira SP Sao Paulo	Rua Dz. José Carvalho Fernéira, s/n Rua Caetanópolis, 930 - Jaguaré	Greenfield Rooftop	Self-supporting Mast	(3+3+3) 40 (3+3+3) (3+3+3)
89 7141357 SPWC03721 -235433 -46,6564 SP Sao Paulo Av. Angdica, N°1189 - Consolução Roofrop Mast	7141274 7141357	SPWCO3715 SPWCO3721	-23,553 -23,5433	-46,6569 -46,6564	SP Sao Paulo SP Sao Paulo	Ruz Bela Cintra, 450 Av. Angélica , Nº 1189 – Consolação	Rooftop	Mast Mast	(3+3+3) (5+5+3)
0         74435         SP14454         2,1998         44,8039         52         Imlia         Backshown         Isolay         Mar           1         74455         SVTS14         2,555         4,614         52         Same         Backshown         Backshown <td< td=""><td>7141816 533003</td><td>SPSTS5374 SPWBV3596</td><td>-23,9542 -23,55803</td><td>-46,64891</td><td>SP Santos SP Sao Paulo</td><td>Rua Augusto Paulino, 57 RUA LOURENCO GRANATO, 214</td><td>Rooftop</td><td>Mast Mast</td><td>(4+4+4+4+4+4) (3+3+3+3) (4+4+4+3) (4+4+4)</td></td<>	7141816 533003	SPSTS5374 SPWBV3596	-23,9542 -23,55803	-46,64891	SP Santos SP Sao Paulo	Rua Augusto Paulino, 57 RUA LOURENCO GRANATO, 214	Rooftop	Mast Mast	(4+4+4+4+4+4) (3+3+3+3) (4+4+4+3) (4+4+4)

_			Geolocation			Address		Structure	
41 195 196	Keycode 7141411 7142770	Site ID L SPWBS3612 SPWIT3794	atitude (decimal) Lo -23,54216667 -23,53438	ngitude (decimal) Stat -46,60605556 SP -46,46393 SP		Address RUA JOSE MONTERO, 323 AVENIDA CAMPANELLA, EDIFÍCIO: CAMPANELLA 1297 – ITAQUERA	Type of Site Rooftop Rooftop	Type of Structure Mast Mast	Height (4+4) 4
197 198 199	7148679 7141871 7141873	RJPTS0074 SPSTS5387 SPSTS5389	-22,30964 -23,96904 -23,9571	-43,12962 RJ -46,3326 SP -46,32503 SP	Petropolis	Eas. da União e Indústria, 22565 Rua Ana Conta, 555 Rua Bario de Pararegiación, 77	Greenfield Rooftop Rooftop	Pole Mast Mast	9,4 (3+3+3) (6+6+6+6)
200 201 202	7141906 7141908 7141936	SPSTS5397 SPSTS5399 SPSTS5406	-23,9656 -23,93542 -23,96493	-46,33925 SP -46,33177 SP -46,3199 SP	Santos	Rua Euclides da Camba, 151 Praca dos Andradas, 91/93 Rua Dr. Lobo Vians, 68	Rooftop Rooftop Rooftop	Mast Mast Mast	(4+4+4) (3+3) (3+3+3+3)
203 204 205	7141938 7142414 7142439	SP5T55408 SPWJP3833 SPWJP3845	-23,96132 -23,564865 -23,565088	-46,34091 SP -46,661963 SP -46,655097 SP	Santos Sao Paulo	Real Viscondi de l'aria, 250 Real Viscondi de l'aria, 250 Ran José Maria Lisbon, 1187 Rea Paraphona, 825 – Jardim Paulosa	Rooftop Rooftop Rooftop	Mast Mast Mast	(3+3+3) (6+6) (4+4)
206	7142443 7147858	SPWPZ3957 S2RSPAE0970 ESCNA3115	-23,535286 -30,149231 -19,5345	-46,65307/ SP -46,682718 SP -51,14314 RS -40,62714 ES	Sao Paulo Porto Alegre	Rai Famjuna, ASS - Jaram Fatawa Rai Dama, 740 - Pendara ACESSO D SQ 31 UNIDADE VICINAL, 3146 AVENDA GETULUO VARGAS. 500	Rooftop Rooftop Rooftop Rooftop	Mast Mast Mast	(3+3+3) (3+3+3) 4 (6,7+6,7)
208 209 210	646650 647847	MGORP4601 MGJFA5022	-20,39247222 -21,754453	-43,49375 MC -43,35104 MC	Ouro Preto	RUA DO CRUZEIRO, 248 AVENIDA DO RIO BRANCO, 1441	Rooftop Rooftop	Mast Mast Mast	(6,/+6,/) 10 (4+4)
211 212 213	7142492 7142533 7142567 7142702	SPWIQ3807 SPWRT4003 SPWSC4035 SPWMR3923	-23,647357 -23,586326 -23,532194 -23,583254	-46,641674 SP -46,7900818 SP -46,64689 SP		Rux das Geranistanas, 110 – Janlim Oriental Rus Cindido Fontoura, 481 Rus Gunianase, 1197 Av. Lineu de Pauls Machado, 1174 Janlim Everest	Rooftop Rooftop Rooftop Rooftop	Mast Mast Mast Mast	(5+5+5) (4+4+4) (3+3+3)
214 215 216	7142702 7142746 7142748	SPWIR3748 SPWIR3751	-23,583254 -23,577905 -23,582762	-46,612124 SP -46,612124 SP -46,607048 SP	Sao Paulo Sao Paulo	Rus Vasconcelos Drumond, 196, Ipiranga Rus Costa Aguiar, 994 - Ipiranga	Roofsop Roofsop	Mast Mast	(6+6+6) (4+4+4) 8
217 218 219	7142785 7144449 7144581	SPWSU4108 SPARC3152 SPSTS3213	-23,62198056 -21,21879 -23,96765	-46,64914722 SP -50,43917 SP -46,33975 SP		Av. Indianipolis, 2185 Rus Wencolau Braz 120 - Ed Copacabara - Vila Santa Maria Averida Presidente Wilson, 66 – José Menino	Rooftop Rooftop Rooftop	Mast Mast Mast	(6+6+6) (3,20+4,23+3,20) (4+4+4)
220	7144583 7144620 7144683	SPSZN3100 SPWMM3096 SPWIP5067	-23,64832858 -23,609865 -23,56194444	-46,31464501 SP -46,660513 SP -46,666411 SP	Sao Paulo Sao Paulo	Rus Beriganin Constant, 1653 Alameda dos Anaganas, 1297 - Moerna Alameda Tixti, 301 - Jardin Paulsia	Rooftop Rooftop Rooftop	Mast Mast Mast	(4+4+4+6+6) (4+4+4) (4+4+4)
223 224 225		PITAQ1372 BAUIA1294 BAJGQ1239	-6,597028 -14,308056 -13,52977778	-42,284167 PI -39,326889 BA -39,97066667 BA	Ubsitaba Jagaaquara	AV. Principal, s/m RUA DO CACAU S/N RUA 15 DE NOVEMBRO, S/N	Greenfield Greenfield Greenfield	Self-supporting Self-supporting Self-supporting	40 50 50
226 227 228	7142180 7142282 7142302 7147029	BAPPK1289 BAPGU1173 BASUX1312	-14,94391667 -16,43944444 -10,943083	-41,71886111 BA -39,06722222 BA -40,407972 BA	Porto Seguro	PRACA IOAQUIM IOSE RIBEIRO, S/N Rea Projezda, Sn (Cadad Hatorica) NÃO ENCONTRADO	Greenfield Greenfield Greenfield	Self-supporting Self-supporting Self-supporting	100 30 40
229 230 231		RSCEN0022 S2MASLS1512 PAICO2234	-29,35931 -2,5122222 -1,20317	-50,82506 RS -44,303611 MA -48,28411 PA		RUA JOSÉ PEDRO PIVA, 220, CENTRO AVENIDA CASTELO BRANCO, 500 RODOVIA ARTHUR BIZEARDES, 9001 – KM 10	Rooftop Rooftop Greenfield	Mast Mast Self-supporting	(6+6+4) (1+2+3) 40
232 233 234		PABIM2135 PABIM2139 PABIM2140	-1,20317 -1,279585 -1,2544 -1,4413889	-48,29108 PA -48,28391 PA -48,464,5056 PA	Belem	AVENDA FERNANDO GULIDON, 1520 TRAV. LOMAS VALENTINAS, 40 TRAVESSA CHACO, 3029	Greenfield Greenfield Greenfield	Self-supporting Self-supporting Self-supporting	40 40 40
235 236 237	7147345 578993 579409	PAPDA2397 PBMAN4510 PBCBO4516	-1,226821 -7,0978 -7,123997	-48,29312 PA -34,846469 PB -34,825 PB	Santa Barbara Do Para Cabedelo	RODOVIA AUGUSTO MEERA FILHO, S/N - CENTRO AVENIDA FLAVIO RIBEIRO COUTINHO, N°143, MANAIRA RUA JOSE RAMALIJO LETIE, 1233	Greenfield Rooftop Rooftop	Self-supporting Mast Mast	60 4 5
238 239 240	579912 7143884 579505	PBMAN4559 PBCTO4571 PEBVG3036	-7,10013 -7,120231 -8,108136	-34,83983 PB -34,87445 PB -34,891775 PE	Jono Pesson	AVENIDA SAPE, 1800 AVENIDA BAROSO, 433 RUA JOSE ADERVAL CHAVES, 264	Rooftop Rooftop Rooftop	Mast Mast Mast	12 4 (4+4+4)
241 242 243	7140674 7140678 7140684	PEBVG3044 PEBVG3087	-8,1244444 -8,127947 -8,060886	-34,897222 PE -34,90(9)(9) PE	Recife	Boa Viagem Averida Boa Viagem, 4000 Edf Vinia. RUA ANISIO RODRIGUES COELHO, 423	Rooftop Rooftop Rooftop	Mast Mast Mast	(6+6+6) (4+4+4)
244	7140715 7140822 7146731	PEBVT3006 PEGRX3016 PESTA3002 PICZE1359	-8,051383 -8,0659 -6,793014	-34,883081         PE           -34,902306         PE           -34,8818         PE           -42,2433         PI	Recife Recife	RUA SETE DE SETEMBRO, 238 AV GOV. AGAMENON MAGAULÁES 265-ESPINHEIRO RUA MARQUES DE HERVAL, 159 – SANTO ANTONIO – RECIFE – PE Rea Projetala S/N	Rooftop Rooftop Greenfield	Mast Mast	(2+2+2) (2+2+3+2,5) (5+5+5)
246 247 248	7146757 532756 530725	PISSP1369 RJBGU0417	-7,012652 -22,874815	-41,38302 Pl -43,466389 RJ	Sussingura Rio De Janeiro	Rua Joaquim Percint Leal 1048 / Centro RUA CORONEL TAMARINO,1986	Greenfield Greenfield	Self-supporting Pole	40 42 20
249 250 251	530865 7147333	RJPDE0297 RTIQR0140 SCJ080311 C0005C3030	-22,90638889 -22,9234763 -26,542275	-43,31388889 R] -43,3681838 R] -49,182278 SC -46,72859 SP	Rio De Janeiro	Travesa Source Pereira, 22 Run Masquis De Learenpagi, Lois 4 Run Gerhandt, s/n – Rio da Luz	Greenfield Greenfield Greenfield	Self-supporting Pole Self-supporting	30 30 30
252 253 254 355	532784 7142557 7141273 7141273	SPWJG3820 SPWCJ3865 SPWCC03714 SPWCD07167a	-23,54926 -23,54242 -23,54972 23,546777	-46,39481 SP -46,65992 SP	Sao Paulo Sao Paulo	Rus Schwäris Larzajörze 252 RUA GIBOUCUS 62 Avernin Angelice, 2000 – Consolução – Sino Paulo Ren International Machanico – Sino Paulo	Greenfield Greenfield Rooftop	Self-supporting Self-supporting Mast	50 50 (4+4+4+3) (4+4)
255 256 257	7142780 7142569 7142581	SPWPN3974 SPWVM4205 SPWLM3888 DAMICSIE	-23,566777 -23,58381 -23,49(81	-46,681700 SP -46,64182 SP -46,68186 SP -7569570 Ph	Sao Paulo Sao Paulo Sao Paulo	Run Jongaimi Antunes, 149 - Finheiron - Sino Paulo Run Conselvitor Rodrigues Advers, 301 - Vila Mariana Run Antônio Rheiro de Morzín, 264 - Vila Carbone, Silo Paulo - SP	Rooftop Rooftop Rooftop	Mast Mast Mast	(4+4) (4+3+3) (4+4+4)
258 259 260	881812 7142377 7143934	BAMJO0786 BAICM1146 PAARU2211 BARLM2016	-12,54906 -15,35298 -1,38777 -1,28777	-37,98879 BA -38,98098 BA -48,38159 PA	Ilba De Comandatuba Ananindeus	Rodovia BA099 Ran de Comandatula, s/n Ran Coletora Late, s/n BUL CHURTON LATE, s/n	Rooftop Rooftop Greenfield	Mast Mast Self-supporting	(5+5+5) (3,5+3,5+3,5+3,5) 50
261 262 263	7144002 7147656 580927	PABLM2010 PAPGN2432 PEIP[4151	-1,39937 -2,978056 -8,397003	-48,47451 PA -47,363611 PA -35,063306 PE	Paragominas Ipojaca	RUA CUMBICA, Nº13 (ESQ. AV. [ULIO CEZAR) Avenida Júňa Kubitschek, v/n Quadra 19 Lote 06 Rua Frei Vicente Salvadore, 203	Greenfield Greenfield Greenfield	Self-supporting Self-supporting Self-supporting	30 60 30
264 265 266 267	7140626 7140737 7140799	PBCB04517 PEJB03306 PEP0X3225	-7,144494 -8,176417 -8,033194	-34,813 PB -34,917167 PE -34,920806 PE	Jaboatao Dos Guararapes Recife	Ran Mirtsu Bichara Solweira, 66 Avosida Bernardo Vicin de Meho, 26(0) Estada das Ubaise 654	Rooftop Rooftop Rooftop	Mast Mast Mast	4 (8+8+8+8+8) (11+11+11)
267 268 269	7147089 7147090 879926	PEIDL3038 PEOLD3424	-8,04175 -8,0683611 -7,976169	-34,8928 PE -34,892722 PE -34,834581 PE	Recife Recife Olinda	RUA QUARINTA E OITO N°454 - EDE NICOLAU II RUA CAP. JOSÉ DA LUZ25 Avenida José Augusto Moreira 2288 - Edf Gabade de Olinda	Rooftop Rooftop Rooftop	Mast Mast Mast	7 5 (2+2+2)
270 271 272	576778 7140797 7141356	RNPTL6005 PEPNM3128 SPWCO3719	-5,790003 -8,036028 -23,5465	-35,202139 RN -34,913778 PE -46,6481 SP	Natal Recife Sao Paulo	AVENIDA PRUDENTE DE MORAES, 757 Avenida 17 de Agosto 500 Rua Amaral Gengel, 518 – Consoloção	Rooftop Rooftop Rooftop	Bracket Mast Mast	17 (12+12+12) (5+5+5)
273 274 275	7142470 7142471 7142696	SPWRE4004 SPWRE4005 SPWIB3779	-23,54646667 -23,538(0)722 -23,582321	-46,63949444 SP -46,63580833 SP -46,675807 SP	Sao Paulo Sao Paulo	Run Coronel Xiwier de Tokelo, 70 - República Av. Prostes Maia, 702 Run Jevino Arrada, 327 - Inim Bibi	Rooftop Rooftop Rooftop	Mast Mast Mast	(6+6+6) (4+4+4+4) (3+4+4)
276 277 278	7142750 7144623 7141055	SPWIR3753 SPWPZ3047 SPG]A4743	-23,5957 -23,537537 -23,99758	-46,60534 SP -46,676035 SP -46,26031 SP	Sao Paulo Sao Paulo Guaraja	Rua Conta Anguir, 2446 - Injunza Rua Camperas, nº 313/325 - Perduen Rua Mário Risteiro, 886	Rooftop Rooftop Rooftop	Mast Mast Mast	(4+4+4) (5+5+5) (3+3+3)
279 280 281	7141064 7141856 7142458	SPG[A5009 SPPGE5277 SPW[P3849	-24,00209167 -24,03057 -23,55875278	-46,26845833 SP -46,47799 SP -46,65911111 SP	Guareja Prais Grande	Rua Ioné Avelino de Oleveire, 200 Rua Afonno Clavere, 43 Av. Pañisez, 2006 – Bela Vista	Rooftop Rooftop Rooftop	Mast Mast Mast	(4+4+4) (3+3+3+3) (3+3)
282 283 284	7142463 7142769 7144475	SPWJP3854 SPWJR3760 SPWJP3068	-23,56957 -23,5946 -23,56922	-46,67254 SP -46,5872 SP -46,65798 SP	Sao Paulo Sao Paulo	Avenida Braol, S/N - Jardim América (Praça Nosas Senhora do Braol) Rea Vernag I, Vila Independência Rea Bratatio, 558	Rooftop Rooftop Rooftop	Mast Mast Mast	(3+3+3) (3+3+3) (4+4+4)
285 286 287	7141723 533391 532140	SPPAA5239 SPWCL3670 SPWAN3570	-22,72492 -23,62315 -23,4394	-47,64866 SP -46,75023 SP -46,7901 SP	Piracicaba	Rua Praça da Gasedaral Dom Ernesto de Paula, 170 - Centro Rua Caserrá, 55 – Jardim Umarizal AV. Fedipos Starba, 50, N Anlanguera	Rooftop Greenfield Greenfield	Mast Pole Self-supporting	(3+3+3+3) 40 50
288 289 290	7140974 7141035 7141038	SPCAS4848 SPCAS4871 SPCAS4876	-22,90313 -22,90455 -22,94	-47,05807 SP -47,05212 SP -47,06226 SP	Campinas Campinas	Rus Grennal Owins, 1233 – Centro Rus Jölino de Mraguita, 36 Rus Vangern Geznie de 054, 506 – 504, 507 – 504, 507 – 504, 507 – 504, 507 – 504, 507 – 504, 507 – 504, 507 – 504	Rooftop Rooftop Rooftop	Mast Mast Mast	(3+3+5+5+5) (3,3+3,3+3,3) (8+8+8)
291 292 293	7141421 7141986 7142644	SPITA4429 SPSOC5568 SPWMC3921	-23,006 -23,50116389 -23,54983267	-46.84016 SP -47,45195278 SP -46,6034218 SP	Intiba Sorocaba	Ruz Quintino Bocaiúra, 343 Ruz Capitão Nascimento Filho, 171	Rooftop Rooftop Greenfield	Mast Mast Pole	(6+6+4+3) (4+4+4+4) 30
294 295	644859 660242	SCBCU0005 PRCTA0054	-26,98771144 -25,59061389	-48,63645491 SC -49,27764722 PR	Balacario Camboria Curitiba	Rui Iolo Castano, 379 Av. Alvin Ruser, 379 Ran Prof. Lycio G. de C.Vellozo, 191	Rooftop Rooftop	Mast Mast	(1,5+1,5+1,5) (2+2+2)
296 297 298	651280 879549 650165	PRCTA0151 RSCSL0061 PRLDA0214	-25,4303056 -29,16858 -23,31271389	-49,27025 PR -51,18822 RS -51,1759 PR	Caxias Do Sul Londrina	Ran XV de Norvenhero, 560 Ran Smirnbac, 2682 Ran Serra, Sozza Narva, 45	Rooftop Rooftop Rooftop	Mast Mast Mast	4 (3,70+3,70) 4
299 300 301 302 303	649224 658738 659790	PRCTA0126 PRCTA0142 PRCTA057	-25,4263889 -25,4197222 -25,44751944	-49,2561111         PR           -49,2663889         PR           -49,29173333         PR           -49,282341         PR	Curitiba	Rua Sette de Abel, 571 Av. João Guilberto, 623 Targos Alfredo Parado, 1441 Alameda Dr. Carlos de Carcalho, 771 - Centro	Rooftop Rooftop Rooftop	Mast Mast Mast	(3+3+3+3) (2+2+2)
304	660390 7144833 7147523	PRCTA0123 RSCSL0052 PRPGO0467	-25,434067 -29,168377 -25,07564622	-51,203406 RS -50,176363 PR	Caxias Do Sul Ponta Grossa	Av. Júko de Castilhos, 4054 Rus Varnhagem, 366	Rooftop Rooftop Rooftop	Mast Mast Mast	3 6 (6+6+6)
305 306 307	879216 879542 532116	PRCTA0103 RSCSL0054 SPWCN3674	-25,4291667 -29,154564 -23,50513	-49,2633332 PR -51,171622 RS -46,50943 SP	Caxins Do Sul Sao Paulo	Ran Marechal Doodoro, 950 Ruz Gen. Arev da Rocha Nobrega, 504- Jardim América R. Maria Angélica Franci, - Vils Franci	Rooftop Rooftop Greenfield	Mast Mast Self-supporting	(3+3) 6 40,35
308 309 310	533252 533269 650163	SPWIP5017 SPWIP5018 PRLDA0219	-23,49238 -23,50641 -23,31101024	-46,3926 SP -46,385 SP -51,17869713 PR	Sao Paulo	R. Marnut Martine de Melo, S/N- Vila Itaim Aversida Tomé Disa Laço, S/N – Itaim Poulata R. Astorg, 80 Hedy	Greenfield Greenfield Rooftop	Self-supporting Self-supporting Mast	50 60 (3+3)
311 312 313	660363 531042 533328	PRASS3015 SPW[H3830 SPGRS3000	-23,3735 -23,48289 -23,47245	-50,843 PR -46,41926 SP -46,3944 SP	Assai Sao Paulo Guaralhos	Run Manoel Rinx, 950 Run Teticiarinr, 146 – Jandim Hekma Run Candelrinz, S/N. – Jardim Jacy	Greenfield Greenfield Greenfield	Self-supporting Self-supporting Self-supporting	30 46 40
314 315 316	7144621 880077 7146853	SPWMR2678 RSPAE0656 RSPAE0013	-23,60001 -30,118054 -30,065389	-46,71906 SP -51,253761 RS -51,186222 RS	Sao Paulo Porto Alegre Porto Alegre	Run Jules Rimet, cotidioCierco P. de Coledo, portico64, Morambi AV. Wencershan Essolar, 5407 Run Conosel Aparicio Borges, S/N - Giória	Rooftop Rooftop Rooftop	Mast Mast Mast	32 6 (6+6)
317 318 319	532881 532560 531670	RJCPB2225 RJSCT0038 RJVPN0027	-22,96206 -22,899465 -22,841119	-43,174784 RJ -43,200305 RJ -43,312 RJ	Rio De Janeiro Rio De Janeiro	Av. Princesa Isabel, n.º 320 Rua Cardoso Marinho, 18 Rua Egré Lafaire Stockler 961, Vila Penha	Rooftop Rooftop Rooftop	Mast Mast Mast	(5+2+3) 6 (6+6)
320 321 322	656155 648465	GOIUB0360 MGSLZ2004 MGBEL2018	-18,418944 -19,85048 -19,968181	-49,216222 GC -43,96363 MC -43,93449 MC	Belo Horizonte Belo Horizonte	Setor Central - Praça da República. CEP. 75503-000 Avenida Antóinio Carlos, 8100 - Pampalha Mall Raa Aforno Corta Reit, 65	Greenfield Rooftop Rooftop	Self-supporting Mast Mast	9 6 (6+4)
323 324 325	647482	MGCTO4150 MGCTO4308 MGDVL3098	-19,923924 -19,91892 -20,1419	-43,938501 MG -43,93906 MG -44,8893 MG	Belo Horizonte Belo Horizonte	Run Espirino Santo - 1059 RUA CARIJOS, 424 - SALA 08 (PRACA SETE DE SETEMBRO) RUA CARIJOS, 424 - SALA 08 (PRACA SETEMBRO)	Rooftop Rooftop Rooftop	Mast Mast Mast	6 (2,5+2,5+2,5+2,5+2,5+2,5) (6+6)
326 327 328	7150749 531226 531259	3GR]CT03347 NULL RIRCP0091	-22,898906 -22,7575837 -22,930766	-43,180778 RJ -43,4465363 RJ -43,208619 RJ		RUA BENEDITINOS, 23 AVENIDA NILO PEGANHA,450 AVENIDA PAULO DE FRONTRN, 739	Rooftop Rooftop Rooftop	Mast Mast Mast	6 4 6
329 330	531612 532663 71.07639	RJBT10728 RJJBT0343	-22,997497 -22,968802 14,4508990	-43,358574 RJ -43,234633 RJ -49,2595833 GC	Rio De Janeiro	AVENIDA DAS AMERICAS, 4666, CEP-22631903 Rus Pacheco Leio, 125 - Fundon - Jundum Roninico, Ros de Janeiro - RJ, 22460-905 ARACUALS SHOPMOL - RUA 44	Rooftop Rooftop	Mast Mast Mast	9 10
331 332 333 334	/14/639 532340 533025 7146525	GOGNA0425 RJBTJ0466 RJCBF1002 RJBTJ2675	-16,559889 -22,987643 -22,816308 -23,001226	-43,358213 RJ -41,980516 RJ	Cabo Frio	ARAGLAIA SHOPPING - ROA 44 RUA BAHUINEAS DA PENISULA, 200 ESTRADA DE CABO FRIO E BUZZOS, S/N AVENIDA DAS AMERICAS, 7777	Rooftop Greenfield Rooftop	Mast Self-supporting	13,5 (7+7) 60 21
335 336	7149019 7144398	R[B1[20/5 R]MGB0066 DFROD0119 R[MGE3514	-23,001226 -22,950579 -16,944194 -22,595189	-43,38578 RJ -44,073359 RJ -47,315306 GC -43,187617 RJ	Mangaratiba O Cristalina	AVESHDA DAS AMERICAS, //// RODOVIA ROS SANTOS, 0 Rodovia Re 040, Braski-Beh Horizonis, Knr: 55 - N S/N RUA RAFAEL COZCIENCO OLIADRA L. LOTE 3.	Roottop Roottop Greenfield Greenfield	Self-supporting Mast Self-supporting Self-supporting	21 12 60 40
337 338 339 340	531614 531770	RIMGE03514 RJRTJ0412 RJCT00960 GOGNA0393	-22,595189 -22,99681 -22,59602 -16,078553	-43,187617 RJ -43,3462 RJ -43,177014 RJ -49,259136 GC	Rio De Janeiro Rio De Janeiro	RUA RAFAEL COZOLENO, QUADRA L, LOTE 3. Av. Das Américas, 3R atos 5. Condeminio Mirio Henrique Simonsen Rus Sao Jose, 90 AV. 136. QUADRA 29. LOTE 9/10.	Greenfield Rooftop Rooftop Rooftop	Self-supporting Mast Mast Mast	40 5 (6+6+6) 10
340 341 342 343	663318	GOGNA0393 GOACG0278 GOGNA0492 RIRAM054	-16,078553 -16,736694 -16,6756099 -22,86132	-49,259136 GC -49,260222 GC -49,2660699 GC -43,280486 RI	Aparecida De Goiania Geiania	RUA CAPIBARIBE Q 22, 0 RUA 04 ESQ. C RUA 02-A, QD.5A, LT.01, N°1.502. SETOR CENTRAL	Rooftop Rooftop Rooftop Rooftop	Mast Pole Mast Mast	10 6 4 2
344 345	531356 7143247 7143674 200774	RJBTJ0411 RJIRJ0023	-22,86132 -22,998897 -22,825248	-43,280486 RJ -43,34343 RJ -43,325317 RJ	Rio De Janeiro	PROXIMO AO MERCADINHO FRANKFULI. Av. das Amiricas, 3120 Ran Hannibal Porto, 450	Rooftop Rooftop Rooftop	Mast Mast	2 6 6
346 347 348	879574 530789 531862 531862	RJCT00218 RJDQX2829 RJBTJ0403 BJC0B0756	-22,908718 -22,784744 -22,987281	-43,184914 R] -43,507403 R] -43,566802 R]	Rio De Janeiro R	AVENIDA GOMES FREIRE, 196 RUA PORTO ALEGRE, 515 un L, 4 - Barra da Tijaca, Rio de Janciro - RJ (ATRAS DA CHURRASCARIA RIO BRASA)	Rooftop Rooftop Greenfield	Mast Mast Pole	6 (5+5) 30
349 350 351	532459 532704 647499	RJSQR0550 RJMRC0622 MGSTO4302	-22,9138 -22,914911 -19,94494	-42,4685 RJ -42,9326 RJ -43,94331 MK	Itaborai Belo Horizonte	Rus Sepifindo Berro, 9 RUA GILMAR DOS SANTOS DUARTE, S/N - QU14 LT12 Rus Carangola, 670 com Rus Cristina	Greenfield Greenfield Rooftop	Pole Pole Mast	40 30 20
352 353 354	648287 7142957 521576	MGEPL4323 MGFUN4153 PAMBA2346	-19,902233 -19,92711 -5,2052	-43,906677 MG -43,9262 MG -49,613 PA	Belo Horizonte Belo Horizonte Maraba	RUA BEGONIA, 72 Av. Bessi, 709 FOLHA 32, QD. 13, LOTE 8, S/N – NOVA MARABÁ	Greenfield Rooftop Greenfield	Pole Mast Self-supporting	30 6 60
355 356 357	7141440 7141443 521825	SPWBS5671 SPWBU3625 PAMBA2311	-23,54322 -23,57189 -5,34899	-46,61921 SP -46,70561 SP -49,13479 PA	Sao Paulo Maraha	RUA DO GASOMETRO, 741, CEP. 3004 - 000 RUA ARMANDO FAIRIANKS, 310 RUA CINCO DE ABRIL, 3410	Rooftop Rooftop Greenfield	Mast Mast Self-supporting	(4+4+4) (4+4+4) 40
357 358 359 360	662011 648835 647608	ESVVA3045 MGJFA5046 MGURA5544	-5,34899 -20,334249 -21,76003 -19,74954	-40,29406 ES -43,34818 MG -47,23671 MG	Vila Velha Juiz De Fora Uberaba	RUA ALISON SIMOFS, 306 RUA MARECHAL DEODORO, 444 Pras Rei Barbosa, 300. (ELVIRA SHOPPING.)	Greenfield Rooftop Rooftop	Self-supporting Mast Mast	3 (3+4) (6+6+6)
361 362 363	651384 647757 659049	MGBSM6267 MGJFA5116 MGCRY6387	-17,01712 -21,766537 -17,187222	-46,09017 MC -43,347744 MC -41,096389 MC	Brasilandin De Minas Juiz De Forn Carai	FAZENDA GLEBA DA BARRA, S/N AVENIDA BARÁO DO RIO BRANCO RUA JORE VICENTE COMBRA, 260	Rooftop Rooftop Greenfield	Mast Mast Pole	2,3 (2+6) 42
364 365 366	656975 658852 7149948	GOVPI5274 MGRPN5688 R[LEB0130	-16,056833 -19,07736 -22,982028	-47,983083 GC -46,20436 MC -43,224083 RI	<ul> <li>Valparaiso De Goias</li> <li>Rio Paranaiba</li> <li>Rio De Janeiro</li> </ul>	QUADRA 5, 0, CEP.72871005 RODOVIA BR 334, 0 Rua Dais Ferrier, N° 658	Rooftop Greenfield Roofton	Mast Self-supporting Mast	6 50 3
367 368 369	658278 7146590 647506	DFBSA0225 DFBSA034 MGBEL4055	-15,76248 -15,790533 -19,8894444	-47,93435 DF -47,875706 DF -43,9680556 MC	Brasila Brasila	International Activity of the Activity of the Activity of Activity	Rooftop Rooftop Rooftop	Gurred Mast Mast	10 4 3
370 371	661823 530741	MGFLO4125 RJBER0244	-19,89663 -22,75329	-43,93488 MG -43,403386 RJ	Belo Horizonte Belford Roxo	Rua Jacui, 1882 AVENIDA RETIRO DA IMPRENSA, 595	Rooftop Rooftop	Mast Mast (i	6 6+6+6+6+6+6+6+6+6+6+6+6+6+6+6+
372 373 374 375	659486 532987 7145458 7145458	MGIRO4867 RJPRT0526 SCSO00503 SUBURD5841	-20,21091667 -22,94684 -27,559494 -23,5514	-43,72011111 MG -43,05836 RJ -48,619475 SC -46,6704 SR	Niteroi Sao Jose	ESTRADA PRINCIPAL, S/N Ran Ábla Numes, Nº 14 RUA AMBRÓSIO DA SILVA, 223 Auraide SIC Guineir 204	Greenfield Greenfield Brooken	Self-supporting Self-supporting Mest	40 30 18 (2+2+2+2+0)
375 376 377 378	7142436 7142460 7142591 7142694	SPW1P3841 SPW1P3851 SPW5C4038 SPW1B3777	-23,5811 -23,57346 -23,53759167 -25,06333333	-46,6704 SP -46,6606612 SP -46,65806111 SP -46,687264 SP	Sao Paulo Sao Paulo	Avenida São Gabriel, 226 – Jandim Paulista Alameda Campiana, 1600 Rua Albuquengar Lins, 92 Rua Jacuarici, 184 – Jiarim Ribi	Rooftop Rooftop Rooftop Rooftop	Mant Mant Mant Maot	(3+3+3+3+4) (3+3+3+3+6) (4+4+4)
378 379 380	7142694 7141657 7141835	SPWIB3777 SPRPO5315 SPPGE5273	-25,06333333 -21,1827 -24,01348	-46,687264 SP -47,8119 SP -46,42288 SP	Ribeirao Preto	Rus Jacunici, 184 – Itain Bhi Run Anea Costa, 555 Rus Silo Salvador, 50 - Guilhermina	Rooftop Rooftop Rooftop	Mast Mast Mast	(4+4+4) (5+5+5+5+5) (4+4+4)
381 382 383	7141875 7141907 7142762	SPSTS5391 SPSTS5398 SPWST4072	-23,96798 -23,93713 -23,632337	-46,35262 SP -46,32359 SP -46,703938 SP	Sao Paulo	Av. Presidente Wilson, 90 - José Marino Przez Dom Idlino José Starzo, 42 - Centro Ruz Bodriguez Pin, 193	Rooftop Rooftop Rooftop	Mast Mast Mast	(4+4+4+4+4+4+4+2) (3+3+3+2,3+2,3) (5+5+5+5)
	7142808 7141034	SPWPN3985 SPCAS4870 SPWJQ3808	-23,55236111 -22,90766 -23,65476	-46,68951944 SP -47,06089 SP -46,63874 SP	Campinas Sao Paulo	Rus Davas, 740 Rus José de Alemar, 675 - Centro Rus Percelut, 193 – Via de Encontro – Súo Paulo	Rooftop Rooftop Rooftop	Mast Mast Mast	(4+4+4) (3+3+3+5+5) (4+4+4)
384 385 386	7142493			-49,54305556 PR		EST. COL. JD. ANA M. CORDEIRO, S/N - ND	Greenfield	Self-supporting Mast	76
384 385 386 387 388 389	649446 7147710 7142670	PRCAK0048 PEMAD3054 SPWIB3769	-25,67638889 -8,05444 -23,5884	-34,50888 PE -46,6881 SP	Recife Sato Paulo	Rus Real da Torre, 292 - Madalena, Recife - PE AVENIDA HORÁCIO LAFER, 702	Rooftop Rooftop	Mast	(5+5+5) (3+3+2)
384 385 386 387 388	649446 7147710	PEMAD3054	-8,05444	-34,90888 PE	Recife Sao Paulo Sao Paulo Sao Paulo Sao Paulo		Rooftop		

			Greie	ecution		Address		Structure	
		PAICO2240		Longitude (decimal) -48,45756 46,71277		Address ESTRADA DO TAPANÁ, S/N, KM 1 L725 – TAPANÁ AVENDA C A VINETUL 155		Self-supporting	Height 40 (4+4+4)
	398 7141854	SPPGE5275	-24,01983	-46,44942 -46,66273 -46,69205	SP Sao Paulo	RUA BALNEARIO CAMBURIU, 100 AL ITU,1067 VILA CARAGUATA CONDOMINIO EDIFICIO JOSÉ HACHEM AN Enzymberio Torko Revisio 1811	Rooftop Rooftop Rooftop	Mast	(6+3+3) (4+4+4+3) (4+4+4)
Cond	402 7143964	PAAIU2221	-1,34001	-38,469693 -48,35395	CE Fortakza PA Belem	RUA DISTRITO INDUSTRIAL DE ANANINDEUA, S/N	Rooftop Greenfield	Self-supporting	(4+5+5) 60
	403 /140821 404 /140823 405 /149558	PESTA3022 PEAFO3382	-8,06311 -8,510037	-34,879278 -41,010361	PE Recife PE Afranio	Avenida Guararapes 154 RUA JOSE VICENTE DE ARAUJO, 0	Rooftop Greenfield	Mast Self-supporting	5 4 40
	40/ 580400 408 520993	PIGMN1360 PABLM2134	-7,362638889 -1,45478	-41,36263889 -48,49011	PI Gemisiano PA Belem	AV BRASIL, Nº 230 AV. DR MORAES 121 NAZARÉ	Greenfield Rooftop	Self-supporting Mast	(6+6+6) 50 (4+4+3)
		RNPWM5701	-5,978611111	-35,11871667	RN Parnamirim	Run Mişiri Armando de Souza Melo, 430 - Edil Jacunà RUA CENTRAL, 0, COHABINAL RUA JOSÉ LEAL, 61 - CENTRO - SÃO JOÃO DO CARIRI	Greenfield	Self-supporting	(7+7+7) 48 40
A. D.	412 532477	SPWSN4046 CEFLA0187 SPWSE4118	-3,71269	-38,47431	CE Fortaleza	Av. Vicente de Castro, 6043	Greenfield Rooftop Reaftern	Self-supporting Mast Must	40 (3+5) (4+4+4)
	416 533079	RJTNQ2900 RJDEO0280	-22,9247 -22,85537	-43,3612 -43,38575	R] Rio De Janeiro	RUA MANACA, 0	Rooftop	Mast	21 40
A. B. A	418 7141236	SPOCO4514	-23,5277	-46,78	SP Osasco	Rua Dr. Mariano Jatahy Marcondes Ferraz, 215 - Centro		Mast	50 10 (1,3+1,3)
Des         Des <thdes< th=""> <thdes< th=""> <thdes< th=""></thdes<></thdes<></thdes<>	422 656355		-22,9571 -25,43555		PR Curitiba	Av. Afonso Camargo, 849 - Cristo Rei - Curitiba - PR	Greenfield Rooftop Greenfield	Self-supporting Mast Self-supporting	12,5 (3+3+3) 40
And And And And And And And And And   And	423 7141331 424 7147893 425 7142657	SPWIB3761	-23,6103	-46,6884	SP Sao Paulo		Rooftop Rooftop Rooftop	Mast	(4+4+4) (3+3) (8+8)
C <td>420 /142/11 427 576000</td> <td>SPWSN4059 RNMLU6004</td> <td>-5,7905</td> <td>-46,64833 -35,19144444</td> <td>SP Sao Paulo RN Natal</td> <td>Rux Valdemar Martins, 116 Rux Florestal 27</td> <td>Rooftop Greenfield</td> <td>Mast Pole</td> <td>(4+4+4) 30</td>	420 /142/11 427 576000	SPWSN4059 RNMLU6004	-5,7905	-46,64833 -35,19144444	SP Sao Paulo RN Natal	Rux Valdemar Martins, 116 Rux Florestal 27	Rooftop Greenfield	Mast Pole	(4+4+4) 30
Set         Set <td>429 522045 430 7142501</td> <td>APSQA2524 SPWRE4015</td> <td>-3,18472 -23,55342</td> <td>-51,991389 -46,64087</td> <td>AP Santana SP Sao Paulo</td> <td>AV. FRANCISCO DE MELO PARENTE, 147, REMEDIOS II RUA AMARO, 418,</td> <td>Greenfield Rooftop</td> <td>Self-supporting Mast</td> <td>40 (4+4+3+3)</td>	429 522045 430 7142501	APSQA2524 SPWRE4015	-3,18472 -23,55342	-51,991389 -46,64087	AP Santana SP Sao Paulo	AV. FRANCISCO DE MELO PARENTE, 147, REMEDIOS II RUA AMARO, 418,	Greenfield Rooftop	Self-supporting Mast	40 (4+4+3+3)
	431 3121443 432 7141477 433 7141534	SPWBV3594 SPWCB3644	-23,55916 -23,62477	-46,65517 -46,66218	SP Sao Paulo SP Sao Paulo	RUA PEIXOTO GOMIDE, 581 Rua Baronesa da Bela Vista, 801 (Hotel Ibin) - Vila Congonhas	Rooftop	Mast Mast	11 (3+3+3+3) (3+3+3)
D         D <thd< th=""> <thd< th=""> <thd< th=""> <thd< th=""></thd<></thd<></thd<></thd<>	434 7142472 435 7146796 436 520570	SPWRE4006 SPWLB0177 AMM/NS3146	-23,57214	-46,6457 -46,63902 -60,0513611	SP Sao Paulo SP Sao Paulo AM Menun	RUA BENTO FREITAS 186 RUA APENINOS, 59 – ACLIMAÇÃO Av. L. (a., Rus State de Abel (Vésneur Dom Baser)	Rooftop	Mast	(5+5+6) (5+5+5) 50
Del <t< td=""><td>437 530842 438 531201</td><td>R[SGO0514 R[BRB0025</td><td>-22,817135 -22,8650382</td><td>-43,032304 -43,3549325</td><td>R  Sao Goncalo R  Rio De Janeiro</td><td>AVENIDA SÃO MIGUEL, 44 Rua Henrique Ferreira, 101 – Bento Ribeiro</td><td>Rooftop Rooftop</td><td>Mast Mast</td><td>(1,80+7,00) (7,5+7,5+3+3+3+2+1,20) 30</td></t<>	437 530842 438 531201	R[SGO0514 R[BRB0025	-22,817135 -22,8650382	-43,032304 -43,3549325	R  Sao Goncalo R  Rio De Janeiro	AVENIDA SÃO MIGUEL, 44 Rua Henrique Ferreira, 101 – Bento Ribeiro	Rooftop Rooftop	Mast Mast	(1,80+7,00) (7,5+7,5+3+3+3+2+1,20) 30
	441 531606	RJPII1832	-22,836081 -22,690624	-43,874674	R] Pirai	ROD. PRESIDENTE DUTRA, S/N° - KM 232 - SENTIDO RJ			(4+6+6+2+2+2+2+2+2+2) 28,5
D </td <td>442 531829 443 531846 444 531869</td> <td>R]PTG0164</td> <td>-22,8004 -23.007368</td> <td>-43,2015</td> <td>RJ Rio De Janeiro</td> <td>AV. MAESTRO PAULO E SILVA</td> <td></td> <td>Bracket Mast Self-supporting</td> <td>16,2 (9+5+6) 35</td>	442 531829 443 531846 444 531869	R]PTG0164	-22,8004 -23.007368	-43,2015	RJ Rio De Janeiro	AV. MAESTRO PAULO E SILVA		Bracket Mast Self-supporting	16,2 (9+5+6) 35
	446 532629	SPWCS3725	-23,629	-46,6181	SP Sao Paulo	RUA KITIZU UTIYAMA, 314, CURSINO		Mast Self-supporting Self-supporting	(3+3+3+3) 40 40
	448 533109 449 533454	RJJSL0059 SPWVC3023	-22,88662 -23,51975	-43,392103 -46,41814	RJ Rio De Janeiro SP Sao Paulo	Rua Alberico Diniz, 1632 - Vila Militar Jardim Solacap – Rio de Janeiro AV. FLAMINGO, 1361 – VILA NOVA CURUÇÂ	Greenfield	Self-supporting	40 30 40
D         D	451 576910	RNSKA5755 RNJUC5737	-6,7705556 -6,27025	-36,7333333 -35,32803	RN Santa Do Senido RN Jundia	AV. ZEZÉ APRIGIO, 27 GRUPO JUNDIA, 0	Greenfield	Self-supporting Self-supporting	25 40 40
Desc         Desc <thdesc< th="">         Desc         Desc         &lt;</thdesc<>	454 577579 455 578144	PEDIM3236 MASMH1979 PBAHA4901	-8,020886 -4,041244	-34,953303 -44,470092 -35,54775	PE Recife MA Sao Matcus Do Maranhao PB Alagoinha	Run Manoel de Medeiros RUA BENU LAGO RUA BELARMINO COSTAS/N	Greenfield	Self-supporting Self-supporting	30 42 50
S     S <td>456 647604</td> <td>MGMRN4603 PRMGA0497</td> <td>-20,38588889 -23,42333333</td> <td>-43,41969444 -51,94472222</td> <td>MG Mariana</td> <td>RUA JOSE MESQUITA, 167 Rua Néo Alves Martins, 3176 – Zona 3</td> <td>Greenfield Roofsop</td> <td>Pole Mast</td> <td>30 4 5</td>	456 647604	MGMRN4603 PRMGA0497	-20,38588889 -23,42333333	-43,41969444 -51,94472222	MG Mariana	RUA JOSE MESQUITA, 167 Rua Néo Alves Martins, 3176 – Zona 3	Greenfield Roofsop	Pole Mast	30 4 5
d         d	460 65/504	MGURN4265 SCSOO0443	-20,3655 -27,59111	-42,7166 -48,60972	MG Urucania SC Sao Jose	RODOVIA MG 265, KM 120 RUA JOAO SATURNINO OURIQUES, 619	Greenfield Rooftop	Self-supporting Mast	20 4
A. B. Control     A. B. Contro     A. B. Control     A. B.	462 660106 463 660435	PRAUN0434 PRFOZ0428	-23,9341667 -25,544	-52,4961111 -54,57	PR Ararana PR Fox Do Iguacu	RUA ROCHA POMBO, 605 – JUVEVÉ – CURITIBA AVENIDA FELIPE WANDSCHEER, 1276	Greenfield Rooftop	Self-supporting Mast	40 100 4
2         30         300 <td>465 /140813 466 7140855</td> <td>PEBVT3032 RNBRV6009 RNLNA6017</td> <td>-8,06075 -5,801359 -5,809052</td> <td>-35,210722 -35,2015</td> <td>PE Recife RN Natal RN Natal</td> <td>AVENIDA CONDE DA ROA VISTA, 331 - CENTRO Ren Noel Rose 856 Barro Vermelho Run N. Sza, de Lourdes 1376 - Ed. Manbaran</td> <td>Greenfield Rooftop</td> <td>Self-supporting Mast Self-supporting</td> <td>44 2,7 32</td>	465 /140813 466 7140855	PEBVT3032 RNBRV6009 RNLNA6017	-8,06075 -5,801359 -5,809052	-35,210722 -35,2015	PE Recife RN Natal RN Natal	AVENIDA CONDE DA ROA VISTA, 331 - CENTRO Ren Noel Rose 856 Barro Vermelho Run N. Sza, de Lourdes 1376 - Ed. Manbaran	Greenfield Rooftop	Self-supporting Mast Self-supporting	44 2,7 32
D         D	467 7140970 468 7140991	SPCAS4840 SPCAS4857	-22,89805 -22,91116	-47,0518 -47,06472	SP Campinas SP Campinas	RUA BARRETO LEME, 2101 Rua Doutor Sales de Oliveira, 463 - Vila Industrial - Campiras - SP	Rooftop Rooftop	Mast	(3+3) (6+6+6+6) (3+3+3+3)
	470 7141173 471 7141351	SPWCL3663 SPWBF3579	-23,646764 -23,53233	-46,772889 -46,6631	SP Sao Paulo SP Sao Paulo	Rua Odemis, 520 Largo Péricles, 0, Barra Funda	Rooftop	Mast Mast	(3+3+3+3) (6+6+6+3) 3+3+3
Cond C	474 7141402	SPSBO4613 SPSBO4613	-23,7098911 -23,7229	-46,5565913	SP Sao Bernardo Do Campo SP Sao Bernardo Do Campo	Rua Braga, 266 - Vila Lusitania, Sio Bernardo do Campo - SP AVENIDA JOÃO FIRMINO, 1229 – ASSUNÇÃO – SÃO BERNARDO DO CAMPO - SP		Mast	15,2 (3+3+3+3) 3
D         D	475 7141465 476 7141471	SPWBV3587	-23,56664	-46,64474	SP Sao Paulo	Rua Treze de Maio, 1566 – Bela Vista – São Paulo		Mast	5 5 (3+3+3)
a)     a)     a)     b)     <	478 7141845 479 7141878	SPSTS5383 SPSTS5395	-23,960502 -23,94431	-46,332308 -46,32687	SP Santos SP Santos	Av. Dona Ana Costa, 376 - Gonzaga – Santos RUA JULIO DE MESQUITA, 158 – VILA MATIAS- SANTOS– SP	Rooftop	Mast	(4+4) (6+6+6+4+4) (3+3+3+4)
B         B	481 7142402	SPWTA4137 SPWJP3834	-23,55196 -23,572762	-46,56205 -46,657295	SP Sao Paulo	Rui Apucarara, 1563 -Tatungé - São Paulo Alameda Joaquím Eugénio de Melo, 1452 – Jd. Paulista – São Paulo	Rooffop Rooffop	Mast Mast	(6+4+4) (3+3+3+3)
D         D	485 /142445	SPWIP3848 SPWPZ3959	-23,5772 -23,534548	-46,6705 -46,66823	SP Suo Paulo SP Suo Paulo	Rua Maestro Filas Lobo, 1040 – Jardim Paulista – São Paulo Rua Monte Alegre, 470 – Perelizes – São Paulo	Rooftop	Mast Mast	(6+5+5) (4+4) (4+4+4+4)
Jac         Sol         Sol <td>486 7142475 487 7142596</td> <td>SPWRE4009 SPWSC4043</td> <td>-23,54046 -23,5393</td> <td>-46,64457 -46,6605</td> <td>SP Sao Paulo SP Sao Paulo</td> <td>Lago do Arouche, 96 – República - SP Rua Dr Gabriel Dos Santos, 759 – Santa Cecilia</td> <td>Rooftop</td> <td>Mast Mast</td> <td>(4+4+4) (3+3+3+3) (6+6+6)</td>	486 7142475 487 7142596	SPWRE4009 SPWSC4043	-23,54046 -23,5393	-46,64457 -46,6605	SP Sao Paulo SP Sao Paulo	Lago do Arouche, 96 – República - SP Rua Dr Gabriel Dos Santos, 759 – Santa Cecilia	Rooftop	Mast Mast	(4+4+4) (3+3+3+3) (6+6+6)
Add       Add     Add     Add     Add     Add     Add     Add     Add     Add       Add     Add     Add     Add     Add     Add     Add     Add     Add       Add     Add     Add     Add     Add     Add     Add     Add       Add     Add     Add     Add     Add     Add     Add     Add       Add     Add     Add     Add     Add     Add     Add     Add       Add     Add     Add     Add     Add     Add     Add     Add       Add     Add     Add     Add     Add     Add     Add     Add       Add     Add     Add     Add     Add     Add     Add     Add       Add     Add     Add     Add     Add     Add     Add     Add       Add     Add     Add     Add     Add     Add     Add       Add     Add     Add     Add     Add     Add     Add       Add     Add     Add     Add     Add     Add       Add     Add     Add <t< td=""><td>490 7142675</td><td>SPWMM3905 SPWMM3910</td><td>-23,59882 -23,60222</td><td>-46,66286 -46,66611</td><td>SP Sao Paulo SP Sao Paulo</td><td>Rus Gaivota, 22 – Moema Rus Gaivota, 665 – Indianópolis – São Paulo</td><td>Rooffop Rooffop</td><td>Mast Mast</td><td>(4+4+4) (4+4+4)</td></t<>	490 7142675	SPWMM3905 SPWMM3910	-23,59882 -23,60222	-46,66286 -46,66611	SP Sao Paulo SP Sao Paulo	Rus Gaivota, 22 – Moema Rus Gaivota, 665 – Indianópolis – São Paulo	Rooffop Rooffop	Mast Mast	(4+4+4) (4+4+4)
	492 7142816 493 7143549	SPWTA4133 CEARJ0547	-23,54451 -6,338194	-46,55553 -39,40525	SP Sao Paulo CE Barbalha	RUA EMILIO MALLET, 1893, VILA GOMES CARDIM DISTRITO SITIO FARIAS, 0	Rooftop Greenfield	Mast Self-supporting	(3+6+4) (4+4+4) 33
	495 7144557 496 7144635	SPSCN3086	-23,633	-46,553	SP Sao Cactano Do Sul	Rua Alameda Sio Caetano, 1738 – Santa Maria – S.C.S	Roofiop	Mast	(4+4+5+5+5) (3+3+3) (6+6+6)
	497 7145465 498 7146262	3GSCFNS0382 PRCTA0453	-25,4	-48,5499 -49,2625	SC Florianopolis PR Curitiba	RUA LACERDA COUTINHO,179 RUA ALBERTO FOLLONI, 1673, JUVEVÉ, CURTIRA	Rooftop Rooftop	Mast Mast	5 (6+1,5) (1,4+2+2+2)
B         C	500 /146268 501 7146339	PRCTA0463 PRLDA0505	-25,474 -23,319355	-49,2766 -51,147805	PR Curitha PR Londrina	Avenida da República, 6464 – Guaira – Curitha - PR AVENIDA SANTOS DUMONT, 500-BOA VISTA	Rooftop	Mast Mast	6 (4+4)
B         C	502 7146549 503 7146946 504 7146956	SPCIV0221	-23,5508	-46,8446	SP Carapiculta	RUA REVOCATA DE MELLO, 51		Mast	(3+1,30) 10 6
B         B	505 /14/000	PAIAB2438	-29,18667 -4,245833 -26,999605	-51,17056 -55,985556 -48,625056	PA Itaituba	Rua Remio Antônio Tomazzoni, 138 – Planalto – Caxias do Sal Rua Santo Antonio - nº 1489	Rooftop Greenfield Rooftop	Mast Self-supporting Mast	6 40 (3+3)
B         B	508 7147669 509 7148511	DFBSA0741 PECGD3112	-8,032888889	-34,87069444	DF Brasilia PE Recife	SET SAFS OD 4 LT 1 ZONA CIVICO ADMINIST AVENIDA PROFESSOR ANDRADE BEZERRA, S/N - CENTRO	Greenfield	Pole	4 40 00
D         D <thd< th="">         D         <thd< th=""> <thd< th=""></thd<></thd<></thd<>	511 520994 512 531439	PAPUP2026 SPWTR4142	-6,0469444 -23,45884	-50,1475 -46,58716	PA Paratapobas SP Sao Paulo	MINA DE FERROSOS CARAJAS N4 (PCE) R. José Camargo Aranha, Jd. Guapira	Greenfield	Self-supporting Self-supporting	20 70 40
D1         D33         D4000         D4000         D4000         D500         D5000         D5000 <thd5000< th=""> <thd5000< th=""> <thd5000< t<="" td=""><td>513 531707 514 532138 515 532508</td><td>RJIC10950</td><td>-22,90774</td><td>-43,110038</td><td>R] Niteroi</td><td>Run Dontor Camha Ferreira, 70 Run Cel. Moneira Geour, 264, Icansi RUA [OAO GUILJERME, 299</td><td></td><td></td><td>(6+6+6+3+3+3+3+3+4+4+4+4+4+4+4+4+4+4+4+4+</td></thd5000<></thd5000<></thd5000<>	513 531707 514 532138 515 532508	RJIC10950	-22,90774	-43,110038	R] Niteroi	Run Dontor Camha Ferreira, 70 Run Cel. Moneira Geour, 264, Icansi RUA [OAO GUILJERME, 299			(6+6+6+3+3+3+3+3+4+4+4+4+4+4+4+4+4+4+4+4+
B         B	516 532529 517 532828 518 533036	RJVLC1843	-23,63707 -22,24666667 -23,49714	-43,70416667		Rus Chico Gomes, S/N* AV. NILO PECANHA, 504 - ESPLANADA DO CRUZEIRO - VALENÇA		Mast	40 21 41
0         0	520 581645	R[TRL0612 CECEA0252	-22,416826 -3,55013889	-42,983676 -40,65647222	R] Teresopolis CE Coreau	Rus Guandu 680, Pimenteirus Rus Tabelião Angelin, S/N - Centro	Rooftop Greenfield	Mast Self-supporting	10 30 70
B         B	522 655005	MGBET6472 MGRPC4883	-19,96729 -19,93547	-44,19717 -43,16867	MG Betim MG Rio Piracicaba	RUA ROMUALDA AUGUSTA DE MELO, 337 ESTAÇÃO TELEMAR, MORRO DAS ANTENAS	Rooftop Greenfield	Mast Self-supporting	(4+4+4+4+4+4) 30
D3         N100         D50         D50 <thd50< th=""> <thd50< th=""> <thd50< th=""></thd50<></thd50<></thd50<>	525 662547	RSPAE0151 RSCAN0870	-30,03375	-51,18645833	RS Porto Alegre	RUA ENGENHEIRO AFONSO CAVALCANTI, 100 RUA FERNANDO FERRARI, S/N	Rooftop	Bracket Mast	(1,5+1,5+4) 21,5 3
10.1         10.00	527 7140706 528 7141030 529 7141059	SPCAS4865	-8,085 -22,89194 -23,981553	-34,968597 -47,06333 -46,235457	SP Campinas	Av. Barão de Itapura, 1518 - Jardim Guanabara, Campinas - SP		Mast	11 (3+3+3+3) (3+3+3+3+3)
N1       N1     N1     N1     N1     N1     N1     N1     N1     N1       N1     N1     N1     N1     N1     N1     N1     N1     N1       N1     N1     N1     N1     N1     N1     N1     N1     N1       N1     N1     N1     N1     N1     N1     N1     N1     N1     N1       N1     N1     N1     N1     N1     N1     N1     N1     N1     N1     N1     N1       N1     N1     N1     N1     N1     N1     N1     N1     N1     N1     N1     N1     N1       N1 </td <td>530 7141091 531 7141381</td> <td>SPINE5044 SPWBL3617</td> <td>-24,18441 -23,46861</td> <td>-46,78722</td> <td>SP Itanharm SP Sao Paulo</td> <td>Rua João Mariano, 193 – Centro - Itanhaém ESTRADA DO SABÃO, 40</td> <td>Rooftop</td> <td>Mast Mast</td> <td>(6+6) (6+6+3) 4</td>	530 7141091 531 7141381	SPINE5044 SPWBL3617	-24,18441 -23,46861	-46,78722	SP Itanharm SP Sao Paulo	Rua João Mariano, 193 – Centro - Itanhaém ESTRADA DO SABÃO, 40	Rooftop	Mast Mast	(6+6) (6+6+3) 4
11         1420         370-00         370-00         470-00	533 7141844 534 7141859	SPSTS5380 SPPGE5280	-23,971389 -24,01515	-46,43055	SP Santos SP Prais Grande	Rua Vicente Carvalho,31 - Boqueirio Rua Nicaragua, 49 - Prais Grande - São Pualo	Rooftop	Mast Mast	4 5 (6+6+6)
13     100     MCDN11     1000     4007     100     Department of the	535 7141909 536 7142566 537 7142652	SPWSC4034 SPWMM3906	-23,520913 -23,59277	-46,654236 -46,65	SP Sao Paulo SP Sao Paulo	Av. Washington Luiz, nº 505 Rua Gruzeino, 732, Barra Funda R. Drz. Nevde Apareida Solito, 460 - Vila Clementino – Moerna - SP	Rooftop	Mast Mast	(5+5) (4+4+4) (5+5+5)
11     1043     PAX30     PAX300     PAX30     PAX30 <td< td=""><td>538         7143020           539         7143384           540         7143675</td><td>MGCON4519 MGUNA5662 RJNCT0438</td><td>-19,51876389 -22,896029</td><td>-43,96276 -50,33352778 -43,12295</td><td>MG Confins MG Uniao De Minas</td><td>RODOVIA MG-010,KM (8 - AEROPORTO INTERNACIONAL RUA DOZE, N° 684 Rua Quinze de Novembro, n°49 - Icarai</td><td>Greenfield Rooftop</td><td>Mast Self-supporting Mast</td><td>(5,5+5,5+5,5+5,5) 80 (6+6)</td></td<>	538         7143020           539         7143384           540         7143675	MGCON4519 MGUNA5662 RJNCT0438	-19,51876389 -22,896029	-43,96276 -50,33352778 -43,12295	MG Confins MG Uniao De Minas	RODOVIA MG-010,KM (8 - AEROPORTO INTERNACIONAL RUA DOZE, N° 684 Rua Quinze de Novembro, n°49 - Icarai	Greenfield Rooftop	Mast Self-supporting Mast	(5,5+5,5+5,5+5,5) 80 (6+6)
11         11/10/2         VIRUNDS         0.100/0         0.0	541 7143835 542 7143856	SPCAS2549 PETMR3031	-23,52305556 -8,03075	-47,10361111 -34,90425	SP Campinas PE Recife	RUA BARÃO DE JAGUARA, 707 - CONCEIÇÃO R. Jundië, 242 - Edf Ssal	Rooftop	Bracket Pole	6 58 (4+4+4)
14         1188         978100         128         44.04         9         Sorba         Internation         Internation         Internation           45         1516         GAMN01         14.0111         44.0121         GD         Angel         11.0140         GAMN013 MULTIANDER	545 /144604	SPGRS3002 SPWCT3015	-23,47626 -23,56721386	-46,42094483	SP Guaralhos SP Sao Paulo	Rus Jaguapeba, 100	Rooftop	Mast Mast	(6+6+6) (4+6)
94         70.90         KAURO         71.90         71	547 7144688	SPWRE3073 GOANS0010	-23,55	-46,64684 -48,94197222	SP Sao Paulo GO Anapolis	Rus Augusta, 66 VIELA VP-3D – DISTRITO AGRO-INDUSTRIAL DE ANÁPOLIS (DAIA) – ANÁPOLIS	Rooftop Greenfield	Mast Self-supporting	40,7 (3+3+3) 80
92         Territy         BOXBND         404001         404001         Perspective Register         AV BURNON Set         Conside         Mergeneral           81         Trity         PELN46         4.0801         AUSO         F         Inset         Table (Sale, Sec.)         General         General <td< td=""><td>550 7146416 551 7146751</td><td>DFBSA0676</td><td>-27,1285 -15,795 -6,814861</td><td>-47,932222</td><td>DF Brasilia PI Sao Joao Da Carabrava</td><td>Sadoeste Octogoral QRSW 1 Bioco A5 Av. Nossa Sra. de Fátirna, 0</td><td>Rooftop</td><td>Bracket</td><td>60 15 84</td></td<>	550 7146416 551 7146751	DFBSA0676	-27,1285 -15,795 -6,814861	-47,932222	DF Brasilia PI Sao Joao Da Carabrava	Sadoeste Octogoral QRSW 1 Bioco A5 Av. Nossa Sra. de Fátirna, 0	Rooftop	Bracket	60 15 84
11       11 <t< td=""><td>552 7147044</td><td>ROCMN3200 PIUIO1436</td><td>-10,56830556 -4,589551</td><td>-63,62225 -42,860702</td><td>RO Campo Novo De Rondonia PI Uniao</td><td>AV. RIO BRANCO, S/N Rua Davi Caldas, S/N° - Centro</td><td>Greenfield Greenfield</td><td>Self-supporting Self-supporting</td><td>48 86 (4+4+4)</td></t<>	552 7147044	ROCMN3200 PIUIO1436	-10,56830556 -4,589551	-63,62225 -42,860702	RO Campo Novo De Rondonia PI Uniao	AV. RIO BRANCO, S/N Rua Davi Caldas, S/N° - Centro	Greenfield Greenfield	Self-supporting Self-supporting	48 86 (4+4+4)
iii         'ison         Add with         -4 labor         4 labor         Number         PRADA MALO III, PRADA NALO IV, P	555 7147709	PEBVG3082 PESAR3274	-8,109444	-34,905833 -34,875658	PE Recife PE Recife	RUA SGT SILVINO MACEDO, 155	Rootiop	Mast Mast	(4+4+4) 5 18
bit       Sing       PMPPoint       Approx 2       Space in PMPPoint       Constant in PMPPoint </td <td>557 7147859 558 7150004 559 10706093</td> <td>3GRJVAS3332 PIVLN1349</td> <td>-7,143333</td> <td>-43,554974 -40,938889</td> <td>R] Vassouras PI Vila Nova Do Piani</td> <td>ESTRADA BARÃO DE IPIABAS, S/Nº – VASSOURAS Av Central, s/n</td> <td>Greenfield</td> <td>Self-supporting Self-supporting</td> <td>9 10 46</td>	557 7147859 558 7150004 559 10706093	3GRJVAS3332 PIVLN1349	-7,143333	-43,554974 -40,938889	R] Vassouras PI Vila Nova Do Piani	ESTRADA BARÃO DE IPIABAS, S/Nº – VASSOURAS Av Central, s/n	Greenfield	Self-supporting Self-supporting	9 10 46
b1         B1094         B170918         234209         44.027         B         B10 hom         Ine Machador Mach	560 521096 561 531060 562 531197	SPWPZ3961	-23,53652	-46,69156	SP Sao Paulo	Rua Coari, 208		Self-supporting	NÃO INFORMADO 40 3
Solution         1502	563 531604 564 532022	RJCD50381 RJPMG0058	-22,945309 -22,879088	-43,362971 -43,445684	RJ Rio De Janeiro RJ Rio De Janeiro	Estr. Mal. Miguel Salazar Mendes de Moraes, 1409 Rea Cajaba, 245, Padre Miguel	Rooftop	Mast	5 (6+4+5) (4+4+4+4)
bit         delDa         Magnetics         Autom C2         Au	565 532520 566 532582 567 645511	SPWSL4088 TOPM[2109	-23,58681 -10,197556	-46,54851 -48,349167	SP Sao Paulo TO Palmas	Rua Joaquim Ribeiro da Costa, 386 AVENIDA LO-5, LT 01 – PLANODIRETOR SUL	Greenfield Rooftop	Self-supporting Mast	(4+4+4+4) 40 9
P1         dbb/s         EXPLOY         33,7431         S2,1431         S2         Palar         A Descript Model (PM Model)         Control (M Model)         Mediance (M Model)           P1         dbb/s         EXPLOY         33,7431         S2,1431         S4         Palar         A Descript Model (M Model)         Control (M Model)         Schlammer	568 647320 569 648887 570 650088	PRIDA0215 MTCBA0021	-23,301657 -15,592444	-56,0915	PR Londrina MT Cuiaba	RUA SALVINA MARIA DO CARMO, S/N Rua Tiradentes, 501 AV HIST RUBENS DE MENDONCA - BAU	Rooftop	Bracket Gurred	40 15,33 10
S1         4405.         ModelWork         3,1199         44,254         Mat         Match (MML) Processing         Konder         Mater           S1         4405.         ModelWork         3,1199         44,254         Mat         Match (MML) Processing         Konder         Mater           S1         Will S2         ModelWork         3,1199         44,254         Mater	571 650620 572 651160	RSPLT0139 DFBSA0189	-31,754321 -15,811593	-52,31574 -47,903387	RS Pelotas DF Brasila	Av. Domingos de Almeida, 1902- Areal SQS 307 BLOCO J - Nº S/N	Greenfield Rooftop	Self-supporting Mast	40 5 6
97         48162         MTCMM64         -1.65286         -56,000         -56,	575 661926	ESVTA3002 MGSOE5014	-20,29791 -21,13589	-40,30039 -44,25834	ES Vitoria MG Sao Joao Del Rei	AV. NOSSA SENHORA DA PENHA,1495 RUA GABRIEL PASSOS, 321	Rooftop Rooftop	Mast Mast	6 5
60         74/26         Ph[0100         A1902         Multiple         Philos         August         Restar         Multiple         August         Phil           60         74/26         Phil         Philos         A1902         Multiple         A1902         Multiple         A1902         Multiple         August         Multiple         M	577 881342	MTCBA0164 RJGAL0976	-15,623667 -22,80670556	-56,087222 -43,23270833	MT Cuisba RJ Rio De Janeiro	AV. BEIRA RIO, 3100 JARDIM EUROPA Estrada das Canúrias, S/N	Rooftop Greenfield	Mast Self-supporting	7 5 20
52         714132         947087360         -31,2315         -44,6021         59         San Pado         Real Twinit (61, Padies)         Konlog         Mat           54         714173         985(544)         -25,07111         -45,9555         59         San Long         Real (46) Andias         Konlog         Mat           54         714170         985(544)         -25,0710         -46,062         59         San Rea         REA (14) Andia (craims, 20/42,0714)         Konlog         Mat           54         71425         98707446         -25,0710         -46,082         59         San Rea         REA (14) ADDIA (Condumine 14), C21 ADDIA (15), C21 ADDIA	580 7140740 581 7140761	PEBVG3222 PEJBO3310	-8,120139 -8,195972	-34,901611 -34,919581	PE Recife PE Jaboatao Dos Guararapes	Rua Fernando Simors Barbosa 316 Averida Bernardo Vieira de Melo, 4804	Rooftop	Mast Pole	3 6 5
St.         1425.         SPRTMO         2,001         4,603         SP /	582 7141352	SPWBF3580 SPSJC5491	-23,52815	-46,68023 -45,8955556	SP Sao Paulo SP Sao Jose Dos Campos	Rus Turiassá, 1863, Perdíaes Rus Cel Icáo Curaino, 240/242 - Ed Tealers	Rooftop Rooftop	Mast Mast	5 (3+3+3+3) 6 (4+4+4)
yr/         114//b         SPPUP196         -23,2055         -44,233         SP         San Pado         As: Crawins dod, 110: Cainidi         Rodes         Mar         6           58         71673         RIGALING         -22,0067         -10,40678         RI         Roh Pado         As: Crawins dod, 110: Cainidi         Rodes         Nat         6	JUN /141210	SPWTU4150 SPWMM3899	-23,59191	-46,58804 -46,67088	SP Sao Paulo	RUA DOMINGOS FERNANDES, 588 – MOEMA – SÃO PAULO	Rootiop	Mast Mast	(6+6+6) (2,5+2)
589 714142 CEFLA0171 3,72978 -38,52563 CE Fortaleza Rus Sena Madureira, 1070 Roofforp Mast	585 /142425 586 7142646	SPWP13940		-43,246878		Av. Vinte de Janeiro, S/Nº	Rooftop Greenfield	Pole	(4+4+4+4+4) 30
9 1415 0 00001 1415 0 00001 0 4000 0 4000 0 8 min 0 4000 0 1410 0 8 min 0 4000 0 14000 0 14000 0 14000 0 14000 0 14000 0 140000 0 140000 0 140000 0 140000 0 140000 0 140000 0 140000 0 140000 0 140000 0 140000 0 140000 0 140000 0 1400000 0 1400000 0 1400000 0 1400000 0 1400000 0 1400000 0 1400000 0 1400000 0 1400000 0 14000000 0 14000000 0 14000000 0 140000000 0 140000000 0 140000000 0 1400000000	385 /142425 586 7142646 587 7142776	RJGAL0142 CEFLA0171	-22,830507 -3,72978	-38,52563	CE Fortaleza		Rooftop	Mast	(2+2+2)
9)         714451         AMPGA24         -1.95044         -59,21655         AM         Products Expendes         Bar 16, 21. Annuit         Generalial         Self segmenting           94         714495         GOCKAM088         -1.66718         -92,6244         GO         Genina         Bar 176, 2995. Annupris-GO         Borling         Mat	385         7142445           586         714276           587         7142776           588         7143713           589         7144174           590         7144176           591         7144425           592         714403	RIGAL0142 CEFLA0171 DFROD0128	-16,194(039	-47,454194 -47,959275	DF Braslia	Rodovia Brasilia - Unai, BR251, Km 14 - nº S/N	Greenfield	Self-supporting	(2+2+2) 60 5 9
95         1612         ORGANS2         4.605         4.001         60         Gene         REREDUCT/PULL 94-Db         Refer         Mar           95         1602         RSGADS         4.005         4.005         6.005         Refer         Refer         Mar           95         1602         RSGADS         4.005         4.005         8.5         Soc Gale         RE/S0.1010.1018.RED.101.5.         Refer         Mar	385         7142425           386         7142646           587         7142676           588         7143713           589         7144142           590         7144142           591         7144252           592         7144403           593         7144513           594         7144955	R[GAL0142 CEFLA0171 DFROD0128 DFRSA0015 DFRSA0015 DFRSA0012 AMPRG3424 GOGNA0008	-16,194(0)39 -15,81109 -15,835583 -1,2359444 -16,67138	-47,454194 -47,959275 -47,867447 -59,4210556 -49,26604	DF Braslia DF Braslia DF Braslia AM Presidente Figueiredo GO Goiania	Redovia Brazin, Unij BB251, Km 14. uř S/N (#IL.DP) SIA SAPS AE D SHIS Q111, BLOCO O. LAGO SUL Ra Tefe, 23 Amorai Ru 172, Espaine Con ISA, 899 – ST. Aeroporto - GO	Greenfield Rooftop Rooftop Greenfield Rooftop	Self-supporting Mast Bracket Self-supporting Mast	(2+2+2) 60 5 9 75,6 (4+2) 9

			Groto	zation		Address		Structure	
	598 7147472	SP[C11882	Latitude (decimal) -29,97833 -23,30361	-46,95638	SP Jacarei	Addreas Av. Fernando Fernari, 1001 Rza Gilberto Movira, 296 - Vili Formosa, Jacasi - SP	Rooftop	Mast	Height (4+4+4) (3+3+3)
	599 7147530 600 7149041 601 7141538	RSPAE0837 SPWCB3651	-23,62529	-51,2107083	RS Porto Alegre SP Sao Paulo	AVENIDA JOÃO PESSOA, 2050 - SANTANA Rua Barão do Triunfo, 276	Rooftop	Mast	5 (4+3+3+3)
	603 656692	MGGON5120	-21,53665		MG Goiana	RUA JOAO DE SOUZA RIBEIRO, 0		Self-supporting	40 32 70
		MGCSH5763	-18,578	-47,876	MG Cascalho Rico	RUA IOLANDA M. CARDOSO, 225	Greenfield	Self-supporting Self-supporting	
	<u>608</u> 7149334 609 7146773	RSPAE00381 RSPAE0037	-30,084 -30,086365	-51,2459 -51,177371	RS Porto Alegre RS Porto Alegre	AVENIDA DIARIO DE NOTICIAS, 300 Estr. Embratel, 114	Greenfield	Mast Self-supporting	4 50
	611 /146305 612 658009	RSPAE0056 MGCAA6386	-30,10923 -19,0726	-51,26387 -41,8649	RS Porto Alegre MG Capitao Andrade	RUA INTERNA DO CLUBE DOS JANGADEIROS AVENIDA LEVINDO DIAS, S/N	Greenfield Greenfield	Self-supporting Self-supporting	24 30
	614 7142823 615 7146539	MGORP4604 RSPAE0047	-20,3109 -30,189	-43,71095 -51,162	MG Ouro Preto RS Porto Alegre	AVENIDA VER. JOAQUIM DIAS TOLĖDO, S/N ESTRADA CHAPEU DO SOL, 481	Greenfield Rooftop	Pole Mast	42 6
	618 7147825	MGIFA5035 RSPAE0533	-21,7972 -29,9869	-43,3834 -51,1809	MG Juiz De Fora RS Porto Alegre	RUA LEONILDO GONÇALVES REGADO, 125 AVENIDA DOS ESTADOS, 1353	Greenfield Rooftop	Pole Mast	40 12 6
D     D     D     No.	619 /14/062 620 7144840	ROPVO3036	-8,76386	-63,88491	RO Porto Velho	AV. GOV. JORGE TEIXEIRA, 629	Greenfield	Self-supporting	40 30 30
	623 647651	MGADS4717	-19,6509	-42,8684	MG Antonio Dias	RUA BELO HORIZONTE, S/N		Self-supporting	6 43 60
	625 7143335 626 878575	S2MGSER4079 MGPOU6188	-19,94466 -22,248317	-43,92221 -44,934313	MG Belo Horizonte MG Pouso Alto	RUA AFONSO PENA, 4001 RUA JOÃO LINO DA SILVA, S/N	Rooftop	Mast Self-supporting	
	628 646972 629 648120	MGENF6404 MGINH5259	-19,62155 -19,17572	-42,22775 -41,86441	MG Entre Folhas MG Itanhomi	ALTO DO MORRO DA ESTAÇÃO TELEMAR AV. SÃO CRISTOVÃO, S/N	Greenfield	Self-supporting Self-supporting	40
D         D	631 661189 632 7145534	MGUBU5904 3GSCPNS0572	-17,255 -27,43006	-40,571 -48,47656	MG Umburatiba SC Florianopolis	RUA PEDRO GANGA, Nº 47 ROD. TERTULIANO BRITO XAVIER, 2352	Greenfield Rooftop	Self-supporting Mast	40 4
	635 655458	SCGPR1117 MGMPL6433	-26,9324 -18,6581	-48,9535 -41,3974	SC Gaspar MG Mendes Pimentel	RUA DORALÍCIO GARCIA, 330 RUA DA TORRE, S/N	Rooftop Greenfield	Mast Pole	(10+10+10+7+7) 30
D         D	638 40700	SCCCO0052 SRR[CTO3024	-27,0742	-52,62663 -43,178045	SC Chapeco R] Rio De Janeiro	AVENIDA FERNANDO MACHADO, 3639 Rus Semular Dantus 105 - BR SEDAN 23 andreas	Roothop	Mast Corporate Building	40 (3+3)
B         B	640 17779 641 22803	RJSMI2621 SPWST2671	-23,6232	-43,3511	R] Sao Joao De Meriti	Rodovia Presidente Dutra, 4200 - Shopping Grande Rio	Indoor Indoor Indoor	Shopping Mall	
	642 6176 643 44270		-3,7792	-46,6589 -38,54085 -46,5487	CE Fortaleza	Av. Senador Carlos Jercisati, 3000	Indoor	Airport	
	645 5766 / 5767	SRBASCT0204	-12,9145	-38,334916	SP Guruhos BA Salvador	Rod. Hélio Schmidt, s/n - Aeroporto Internacional de Sio Paulo (Cumbica) PRACA GAGO COUTINHO, 282,	Indoor	Airport Airport	
Disk         Allow	648 24395 649 5686	DFIND0132 SPBRE4271	-15,8361 -23,4965	-48,057111 -46,8491	DF Brasila SP Baracri	CSB 2, Lotes 1/4, Alameda Shopping Al. Rio Negro, 1033 - Alphaville Shopping	Indoor Indoor	Shopping Mall Shopping Mall	
D         D	651 4635 652 25798	SPARC2841 SRRSINID0525	-21,222 -30,0321	-50,43551 -51,23039	SP Aracatuba RS Porto Alegre	Av. Joaquim Pompeu de Tokdo, 601 - Araçatuba Shopping PRACA DA MATRIZ SN	Indoor Indoor	Shopping Mall Public Building	
	653 17918 654 4709 655 25793	SPBRU2845 SRRSIND0519	-22,3423 -30,0501	-49,0495 -51,2290278	SP Bauru RS Porto Alegre	Averida das Amiricas, 4666 R. Henrique Sovi, 15-55 - Bauru Shopping AVENIDA PRAIA DE BELAS, 1181	Indoor Indoor	Shopping Mall Shopping Mall	
	656 13727 657 13836 658 6650	SRMGBEL4193 MGCEM4198 SPWST4067	-19,974 -19,943 -23,6543	-43,944548 -44,039912 -46,70053	MG Belo Horizonte MG Contagem SP Sw Paulo	Rodovia BR-356, 3049 - BH Shopping Avenida Joino Gara de Oliveira, 1275 - Big Shopping (32260-250) R. Borda Gato, 59 - Bao Vinta Shoronia	Indoor Indoor	Shopping Mall Shopping Mall Shopping Mall	<u> </u>
B         B         B         A	660 45016	RJBOT2604 DFIND0726 SPWBF2628	-22,9473 -15,7334	-43,1833 -47,89914	RJ Rio De Janeiro DF Brasila	Praia de Botafogo, 400 - Botafogo Praia Shopping SETOR TERMINAL NORTE: CONIUNTO I. BOULEVARD SHOPPING	Indoor Indoor	Shopping Mall Shopping Mall	
M         M	662 24393 663 5556	DFIND0130 SPSVE2890	-15,7873 -23,9675	-47,888806 -46,3836	DF Brasilia SP Sao Vicente	OD SCN Q 5 BL-A, 0 - ASA NORTE R. Prij Gapar, 3G- Briomar Shopping Ar, Jusefin Kahrebevek JC (Ditoin) 3(3) - Bavid Storaging Court	Indoor Indoor	Shopping Mall Shopping Mall	
D         D <thd< th="">         D         <thd< th=""> <thd< th=""></thd<></thd<></thd<>	665 27674	SRGOIND0199	-16,7412	-49,275944	GO Goiania	Av. Rio Verde, Od. 102-104	Indoor	Shopping Mall Public Building	
Del         Del <thdel< th=""> <thdel< th=""> <thdel< th=""></thdel<></thdel<></thdel<>	668 17064 669 4761	SRRJCTO3019 SPCAS2846	-22,9323	-47,0781	SP Campinas		Indoor	Public Building Shopping Mall	
Dis         Dis <thdis< th=""> <thdis< th=""> <thdis< th=""></thdis<></thdis<></thdis<>	671 27204 672 50578	PRIND(394 SRRJDQX3(59	-23,3436 -22,7874	-51,18611111 -43,286602	PR Londrina R] Daque De Caxias	RODOVIA CELSO GARCIA CID, S/N Rodovia Washington Luis, 2895 - Caxias Shopping	Indoor Indoor	Shopping Mall Shopping Mall	
D         Solution         So	673 53200 674 5350 675 6759	SPSJC5473 SPWVP4227	-23,1994 -23,5935	-45,88091667 -46,58554	R] Rio De Janeiro SP Sao Jose Dos Campos SP Sao Paulo	Av. Deputado Benedito Matarazoo, 9403 - Center Valc Shopping Av. Dr. Francisco Mesquita, 1000 - Central Plaza Shopping	Indoor Indoor	Shopping Mall Shopping Mall	
B         B         AD         AD        AD        AD	676 18948 677 44517	SRRJBOT3027	-22,9504	-38,53233333 -43,180644	CE Fortaleza R] Rio De Janeiro	AV DR SILAS MUNGUBA, 5700 Praia de Botafogo, 501 - Centro Empresarial Mourisco	Indoor Indoor	Corporate Building Business Center Corport Venue	
D         D <thd< th=""> <thd< th=""> <thd< th=""> <thd< th=""></thd<></thd<></thd<></thd<>	680 6350	SPBRE0581 SPWJE2880	-23,5059 -23,5476	-46,83716 -46,757	SP Barueri SP Sao Paulo	Av Marcos P de Ulhoa Rodrigues, 939 - AZUL Av. Leão Machado, 100 - Continental Shopping	Indoor Indoor	Corporate Building Shopping Mall	-
	682 14425	SRMGLUR4196 SRRJDQX4007	-19,9278	-43,946889 -43,31112	MG Belo Horizonte R] Duque De Caxias	Avenida Olegirio Maciel, 1600 - Diamond Mall Shopping AV GOV LEONEL DE MOURA BRIZOLA 1777	Indoor	Shopping Mall Shopping Mall	
B         B		SPSOC2891	-23.511	-47,4628	SP Sorocaba	Av. Izoraida Marques Peres, 401 - Esplanada Shopping	Indoor Indoor Indoor	Shopping Mall Shopping Mall	
	<u>688</u> 4894 689 4780	SPFAC2636	-20,548	-47,4169	SP Franca	Av. Rio Negro, 1100 - Franca Shopping	Indoor	Shopping Mall	
D         D <thd< th=""> <thd< th=""> <thd< th=""> <thd< th=""></thd<></thd<></thd<></thd<>		SPCTD2850 MTCBA0068 SPSNF2870	-15.5882	-48,9775 -56,112361 -46,5315		Av. José Nelson Machado, 280 - Garden Catandura Shopping AV. LAVA PÉS Av. Industrial, 600 - Grand Plaza Shopping (Antigo ABC Plaza Shopping)	Indoor Indoor Indoor	Shopping Mall Shopping Mall Shopping Mall	
B         B	694 14431 695 44972	SRMGLUX6493	-22,8865 -19,9507	-43,285139 -43,95887	R] Rio De Janeiro MG Belo Horizonte	Avenida Dom Hilder Cimara, 5474 Rua Gentios, 1350 - Hospital Laxenburgo/Mário Penna Averida Beirs-Mar. No 3980	Indoor Indoor	Shopping Mall Hospital Hotel	
D         D <thd< th="">         D         <thd< th=""> <thd< th=""></thd<></thd<></thd<>	696 23412 697 17469	SRCEFLA0296 RJNIU2623	-3,75023 -22,7536	-38,448089 -43,454205	CE Fortaleza RJ Nova Igaacu	Av Diogainho, 4189 Av Governador Roberto da Silveira, 540 - Igança Top Shopping	Indoor Indoor	Hotel Shopping Mall	
No.         No.         No.         No.         No.         No.         No.         No.           No.         No.         No.         No.         No.         No.         No.         No.           No.         No.         No.         No.         No.         No.         No.         No.         No.           No.	609 17559 700 5016	RJPTG2607 SPJCI2856	-22,8018 -23,3042	-43,201679 -45,9633	RJ Rio De Janeiro SP Jacarei	Avenida Maestro Paulo Silva, 400 R. Bario de Jacarei, 364 - Jacarei Shopping	Indoor Indoor	Shopping Mall Shopping Mall	
No.         Del         Apple         App	702 17296 703 5191	SRRJJCP3042 SPPGE5270	-22,9754	-43,390507 -46,40744659	RJ Rio De Janeiro SP Praia Grande	AV AYRTON SENNA DA SILVA. 1500	Indoor	Stadium Shopping Mall	
No.     A. D.     A. D.     A. D.     A. D.     A. D. D.		SPWST3333	- /37/003 -23,6214 -23,6214	-46,09962 -46,09962	SP Sao Paulo	Av. Dr. Chueri Zaidan, 902 (Market Place Offices - Torres de escritório)	Indoor Indoor	Corporate Building Shopping Mall	
No.         Ope         Ope <td>708 5854 709 19690</td> <td>SP[AI2858 SRAMMNS3304</td> <td>-23,168 -3,10138</td> <td>-46,89205556 -60,024722</td> <td>SP Jundiai AM Manaus</td> <td>Av. Antônio Frederico Ozaran, 6000 - Maxi Shopping Jundiai Averida Djalma Batista, 1661 - Millenrium Shopping</td> <td>Indoor Indoor</td> <td>Shopping Mall Shopping Mall</td> <td></td>	708 5854 709 19690	SP[AI2858 SRAMMNS3304	-23,168 -3,10138	-46,89205556 -60,024722	SP Jundiai AM Manaus	Av. Antônio Frederico Ozaran, 6000 - Maxi Shopping Jundiai Averida Djalma Batista, 1661 - Millenrium Shopping	Indoor Indoor	Shopping Mall Shopping Mall	
No.         A. D.         A	710 15097 711 24402	SRMGUNI4194 DFIND0139 SPSTS2871	-19,8746	-43,925672		Averida Gristiano Machado, 4000 - Miras Shopping Itamaraty - Explanada dos Ministérios, Bloco II, Ministério das Relações Exteriores Av. Eucledes da Canha, 21 - Miramar Shopping	Indoor Indoor Indoor	Shopping Mall Public Building Shopping Mall	
No.         No. <td></td> <td>SPWST4065</td> <td>-23,6232</td> <td>-46,1796 -46,69843</td> <td>SP Sao Paulo</td> <td>Av. Ver. Narciso Yagae Guimarko, 1001 - Mogi Shopping AV ROQUE PETRONI JUNIOR, 1089</td> <td>Indoor</td> <td>Shopping Mall Shopping Mall Shopping Mall</td> <td></td>		SPWST4065	-23,6232	-46,1796 -46,69843	SP Sao Paulo	Av. Ver. Narciso Yagae Guimarko, 1001 - Mogi Shopping AV ROQUE PETRONI JUNIOR, 1089	Indoor	Shopping Mall Shopping Mall Shopping Mall	
Dis         Dis <thdis< th=""> <thdis< th=""> <thdis< th=""></thdis<></thdis<></thdis<>	717 5269	SPRPO2866	-3,73527 -21,2072	-47,76361111	CE Fortaleza SP Ribeirao Preto	Av. Bezerra de Menezes, no. 2450 (North Shopping) Av. Presidente Kennedy, 1500 - Novo Shopping	Indoor Indoor	Shopping Mall	
10         10<	719 4403 720 72069	PERCE3244 PESTA3052	-8,06006	-34,87816	PE. Recife	Rua Cais da Alfadênga S/N Praça da Republica, S/N	Indoor Indoor	Shopping Mall Public Building	
B         BO         BOO         BOO         ADD	722 50780 723 4779 / 23280	SRRJCGR3022	-22,9271	-43,575127	RJ Rio De Janeiro SP Campinas	Estrada do Monteiro, 1200 - Park Shopping Campo Grande Av. Guilherme Campos, 500 - Parque Dom Pedro Shopping	Indoor Indoor	Shopping Mall Shopping Mall	
Sol         Sol <td>725 5218</td> <td>ALMCO2399</td> <td>-9,559894</td> <td>-35.744475</td> <td>SP Presidente Prudente</td> <td></td> <td>Indoor</td> <td>Shopping Mall Shopping Mall</td> <td></td>	725 5218	ALMCO2399	-9,559894	-35.744475	SP Presidente Prudente		Indoor	Shopping Mall Shopping Mall	
b.         b.         dependence         dependence         b.         dependence         b.         dependence         dependence           10         100         0.00         0.0000	729 5446	RJGAV0727 SPSRR2840	-22,9783 -20,8279	-43,230293 -49,3863	R] Rio De Janeiro SP Sao Jose Do Rio Preto	Rua Vice-Governador Rúbens Berardo, 100 Av. José Munia, 4775 - Plaza Avenida Shopping	Indoor	Corporate Building Shopping Mall	
Description         Description <thdescription< th=""> <thdescription< th=""></thdescription<></thdescription<>	731 5447 732 5499	SPSRR2875 SPSTS2872	-20,8117 -23,9777	-49,38147222 -46,3095	SP Sao Jose Do Rio Preto SP Santos	R. Berrardino de Campos, 2940 - Praça Shopping R. Alexandre Martins, 80 - Praiamar Shopping	Indoor Indoor	Shopping Mall Shopping Mall	
No.     All       No.     All     All<	733 17436 734 24400 755 5210	DFIND0137 SPPPE2865	-15,8085 -22,1161	-43,118833 -47,859389 -51,4076	DF Brasilia SP Presidente Pradente	Rus da Conneciçio, 188 SET SAFS QUADRA 4 LOTTE 3, 0 - ZONA CÍVICO ADMINIST Av. Manoel Goulart, 2400 - Prudenshorping	Indoor Indoor Indoor	Corporate Building Public Building Shopping Mall	
91     92     92     92     92     92     94     <	738 5281	SPWVS2253 SRRJJBT5041 SPRPO5314	-23,5882 -22,9606 -21,21,33	-46,75102 -43,20939 -47,81638889	SP Sao Paulo RJ Rio De Janeiro SP Ribeirao Preto	Av. Cel. Fernando Ferneira Leite, 1540 - Ribeirão Shopping	Indoor Indoor	Shopping Mall Corporate Building	
30     500     500     5000 <t< td=""><td>739 17298 / 73191 740 50509 741 5448</td><td>SRRJ[CP3043 / SRR][CP3053 SRRJBOT3021</td><td>-22,9766 -22,9537</td><td>-43,4109 -43,178543</td><td>R] Rio De Janeiro R] Rio De Janeiro</td><td>Avenida Salvador Allende, 6555 Rua Gererul Severinto. 97 - Rio Plaza Shomine</td><td>Indoor Indoor</td><td>Convention Center Sheeping Mall</td><td></td></t<>	739 17298 / 73191 740 50509 741 5448	SRRJ[CP3043 / SRR][CP3053 SRRJBOT3021	-22,9766 -22,9537	-43,4109 -43,178543	R] Rio De Janeiro R] Rio De Janeiro	Avenida Salvador Allende, 6555 Rua Gererul Severinto. 97 - Rio Plaza Shomine	Indoor Indoor	Convention Center Sheeping Mall	
No.         Print         Print         Print         Print         Print         Note         Name of the print         Name of the prin         Name of the prin         Name of the	742 26585 743 14784	SRMGSAV4363	-19,9384	-43,937148	SC Florianopolis MG Belo Horizonte		Indoor	Shopping Mall	
Mai         Colo         SWARD         Colo         SWARD         Colo         SwarD         Advance	745 27591 746 18946 747 445**	PRCTA0809	-25,4635 -3,73333	-49,29051111 -38,4977778	PR Curitha	SHOPPING ÁGUA VERDE - AVENIDA REPÚBLICA ARGENTINA, 1927	Indoor	Shopping Mall	
11         40°         MOVE         40°         MOVE         Ref         Mach (main and (	748 6721	SPWVF4169 SRRJCPS3057	-23,5618 -21,7636	-46,560542 -41,33308	SP Sao Paulo R  Campos Dos Goytacazes		Indoor Indoor	Shopping Mall Shopping Mall	
Sint         Barro	752 38449	SRBAPOK0201 PEBVT3241 SPGR50184	-13,0067 -8,0599 -23,4341	-38,524524 -34,887764 -46,40342	BA Salvador PE Recife SP Guardhos	Rua do Giriquiti 48 Avenida Juscelino Kubitschek de Oliveira, 5308- Shopping Bomsucesso	Indoor Indoor	Shopping Mall Shopping Mall	
1.3         1.0.1         1.0.10	753 24800	SRGOGNA0304 SRRJCPS3058 SRRJSCR3028	-16,695 -21,7549 -22,9193	-49,265389 -41,34913 -43,25045	R] Campos Dos Goytacazes R] Rio De Janeiro	Rus 9, 1855 Rus Doutor Silvio Ratos Tavares, 330 - Shopping Boulevard Campos Rus Bario de São Francisco, 246 - Shopping Boulevard RosShopping antipo Iguaterri	Indoor Indoor	Shopping Mall Shopping Mall Shopping Mall	
90         480 <sup>2</sup> BCAMP         ADD         ALASS         B         Game         ACCULURANC SOLEL (%) T.         Adds         Adds         A           91         972         PLAD29         4.040         Adds         A         Machine	758 38438	ESVVA3847 SPWVS4237 SPWCL0085	-20,3924 -23,5844 -23,6495	-40,318042 -46,72507 -46,75729	ES Vila Velha SP Sao Paulo SP Sao Paulo	Rodovia do Sol, Km 5000 (Shopping Boulevard Vila Velha) Av. Prof. Francisco Morato, 2718 - Shopping Batantá Estratada do Campo Limpo, 45 - Shopping Campo Limpo	Indoor Indoor	Shopping Mall Shopping Mall	
Sol         Allow         SNADD         SOL         SADD         SADD <th< td=""><td>759 44607</td><td>RSCAN0958 PABLM2198</td><td>-29,9103 -1,40388</td><td>-51,18295 -48,430556</td><td>RS Canoas PA Belem</td><td>AV GUILHERME SCHELL 6750/ TE Rod. BR - 316, Km 1, s/n (Shopping Castanheira)</td><td>Indoor Indoor</td><td>Shopping Mall Shopping Mall</td><td></td></th<>	759 44607	RSCAN0958 PABLM2198	-29,9103 -1,40388	-51,18295 -48,430556	RS Canoas PA Belem	AV GUILHERME SCHELL 6750/ TE Rod. BR - 316, Km 1, s/n (Shopping Castanheira)	Indoor Indoor	Shopping Mall Shopping Mall	
Model         All Monthle         All Monthle         Solution         All Monthle         Al	762 44758 763 6426	SPFND1291 SPWLP2883	-20,2714 -23,5212	-50,23365 -46,7008	SP Fernandopolis SP Sao Paulo	Av. Litério Grecco, 600 - Shopping Center Fernandopolis R. Guaicurus, esquina com a Rus Catão - Shopping Center Lapa	Indoor Indoor	Shopping Mall Shopping Mall	
50         10%         AbbC/10%         400%         400         Abb Absence         In the face of the face	765 6495 766 5254	SPWPH3954 SPRCO2868	-23,5238 -22,412	-46,54694 -47,55475833	SP Sao Paulo SP Rio Claro	R. Doutor João Ribeiro.304 - Shopping Center Penha	Indoor	Shopping Mall Shopping Mall	
11         400         MBEND         300         4000         90         4000         An one base base base base base base base bas	767 13995 768 27208 769 5359	PRIND(399 SPS)(55482	-25,4724 -23,2057	-49,25280556 -45,90733333	PR Curitiba	AVENIDA MARECHAL FLORIANO PEIXOTO, 4880	Indoor Indoor	Shopping Mall Shopping Mall	
The State         State Add Add Add State         State Add Add Add Add Add Add Add Add Add Ad	771 6307	SPWP19937 SPWIB3783 SRRJGAV3036	-23,5202 -23,6088 -22,9751	-46,62322 -46,69095 -43,228487	SP Sao Paulo R] Rio De Janeiro	Av. das Naçoes Unidas, 12555 - Shopping D&D Run Marqués de São Vicente, 52 - Shopping da Gávea	Indoor	Shopping Mall Shopping Mall	
171         180         90.0085         2.02         44.021         9         Con         Booked layer Trans, b.21         Idea         Borger Mit         -           75         466         PIRONS         4.0403         4.0408         8         Idea         Marger Mit         -         Booked layer Trans, b.21         Idea         Borger Mit         -           75         466         PIRONS         4.0403         4.0408         8         Idea         Suppose Mit         -         -         -         -         -         Idea         Suppose Mit         -         <	773 53050 774 13810 775 4507	SRMASLS1686 SRMGCAI4197	-2,52753	-44,255088 -43,967975	MA Sao Lais MG Belo Horizonte	Av. Daniel de La Touche, 987 Avenida Presidente Carlos Luz, 3001 - Shopping Del Rey	Indoor Indoor	Shopping Mall Shopping Mall Shopping Mall	
17:         6407 280         9FWAMD2 2 9FWAMD2         34007         44,047         9         46,048         9         48,048         1.0 Integrate 100 (Strengt Integrate	777 50637	SPWBV3590 SPCOA0258	-23,5539 -23,5922	-46,65269 -46,83274	SP Sao Paulo SP Cotia	R. Frei Caneca, 569 - Shopping Frei Caneca Rodovia Raposo Tavares, km 23	Indoor Indoor	Shopping Mall Shopping Mall	
Sci.         Solution         State of Solution	779 6459 / 22802	SPWMM3912 / SPWMM2797 SRCEFLA0298	-23,6099 -3,7575	-46,66645 -38,48805556	SP Sao Paulo CE Fortaleza	Av. Ibirapaera, 3103 - Shopping Ibirapaera Av. Washington Soares, no. 85 (Shopping Iguatern)	Indoor Indoor	Shopping Mall Shopping Mall	
55         494         970C/544         24/54         44/20*         92         Sor Pads         ALVERIZADOS (25) Supergic landargo         Indoor         Supergic land	782 6508 783 25802	SPWPN3977 RSIND0529	-23,5769 -29,1766	-46,688 -51,21786	SP Sao Paulo RS Caxias Do Sal	Av. Brig. Faria Lima, 2232 RST 453 2787 SN	Indoor Indoor	Shopping Mall Shopping Mall	
57         662         98B (1000)         (1)764         3.6 (20)         NA         Observation         (1)701	785 6194 786 26628	SPWCG3654 SCIND0277	-23,674 -27,5855	-46,6768 -48,614708	SP Sao Paulo SC Sao Jose	AV. INTERLAGOS, 2255 - Shopping Interlagos RUA GERONCIO THIVES	Indoor	Shopping Mall Shopping Mall	
92         068         976 (208)         20,00         46,011         97         Sar Pala         A Annabes, 2013< Segregation Annabes         Index         Sterrage Mat         .           93         648         976 (208)         20,000         46,0101         97         Sar Pala         A Konzules, 2013         Non-markes, 2013< Segregation Annabes	787 9082 788 4652	SRBAIGW0096 SPARQ2842	-21,7827	-38,463821 -48,20135556	BA Salvador SP Araraquara	AVENIDA ANTONIO CARLOS MAGALHAES, 848, SHOPPING ITAIGARA Acesso Engenheiro Heitor de Souza Pinheiro, 2270 - Shopping Jaragui	Indoor Indoor	Shopping Mall	
75         648         SPRED86         2.546         4.6(35)         59         Sor Path         I. Comed Nation of Tables, 32-Stopping Light         Indoor         Sherpeng Light         . Sherpeng Light         Indoor         Sherpeng Light         . Sherpeng Light <th. light<="" sherpeng="" th="">         . Sherpeng Light</th.>		SPGJA2852 SPWVG2884	-23,9977 -23,5146	-46,25663889 -46,6153	SP Guruji SP Sao Paulo	Av. Marechal Deodoro da Fonseca, 885 Av. Otto Baumgart, 500 - Shopping Lar Center	Indoor	Shopping Mall	
76         67.6         SP012387         24,59         46,576         89         Sp.Pail         R. Complex Company, 200,000,000,000,000,000,000,000,000,00	793 6548	SPWRE2886 SPWRS0086	-23,5466 -23,5364	-46,63857 -46,61826	SP Sao Paulo	R. Coronel Xavier de Toledo, 23 - Shopping Light Rus Barilo de Ladirio, 566/670 - Shopping Mega Polo Moda	Indoor	Shopping Mall Shopping Mall Shopping Mall	
	796 6676 797 6334	SPWTA2887	-23,539	-46,5763	SP Sao Paulo	R. Gonçalves Crespo, 78 - Shopping Metró Boulevard Tatuapé	Indoor	Shopping Mall	

			C	American Contraction			Address		\$5	
	Keycode		Latitude (decimal)	Longitude (decimal)	State	Municipality	Address		Type of Structure	Heirht
798	6743	SPWVM4203	-23.599	-46.63708	SP	Sao Paulo	Av. Domineos de Morais, 2564 - Shorreine Metró Santa Cruz	Indoor	Shopping Mall	
799	6682	SPWTA4134	-23,5413	-46.57633	SP	Sao Paulo	R. Domineos Arostim, 91 - Shopping Metro Tatuapé	Indoor	Shopping Mall	
800	5963	SPSBO2624	-23,6931	-46.5503	SP	Sao Bernardo Do Campo	Praça Samuel Sabatini, 200 - Shopping Metropole	Indoor	Shopping Mall	-
801	27593	PRIND0574	.25.4353	.49 274595	PR	Curitila	SHOPPING METROPOLITAN - PRACA BUI BARBOSA, S/N	Indone	Sheeping Mall	
802	4522	RNLSA6033	-5.81236	-35.20513	RN	Nata	Avenida Bernardo Vicira 3775.Shorreine Midway mall.	Indoor	Shopping Mall	-
803	17081	SRRIDCT3037	-22.8785	-43.2729	RI	Rio De Janeiro	Avenada Pastor Martin Lather King Jr. 126	Indoor	Shopping Mall	-
804	5500	SPSTS2873	-23.962	-46.3327	SP .	Santos	Av. Ann Courts 549 - Shorming Parene Balmeirin	Indoor	Showing Mall	
805	19773	PABI M2199	.1.459	-48.495	PA	Belem	Tray, Padra Eutiquio, no. 1078 (Shorreing Pino Belem)	Indoor	Shopping Mall	-
806	6240	SPWC03720	-23 5421	-46 65741	SP.	Suo Paulo	Av. Havenórolis, 618 - Shorring Patio Havenórolis	Indoor	Shopping Mall	
807	40604	SPSTS1165	-23.965	-46.33217	SP	Santos	Ay, Ana Costa, 465 - Shopping Patao Irooranga	Indoor	Shopping Mall	-
808	6168	SPWBV3936	-23.5706	-46.64344	CP CP	Sao Paulo	R. Treze de Maio, 1947 - Shopping Patio Paulista (Antigo Shopping Paulista)	Indoor	Shopping Mall	-
809	14782	SRMGSAV4204	-12,9403	-43.93356	MG	Belo Horizonte	Avenida do Contorno, 6061 - Shorring Pátio Sayasi	Indoor	Shopping Mall	
810	8877	SRBAASG0203	-12.9834	-38.512943	BA	Salvador	RUA CONS. JUNOUEIRA AYRES. 165/TO	Indoor	Shopping Mall	
811	5140	SPPAA2864	-22,7042	-47,6502	SP	Piracicaba	Av. Limeira, 722 - Shoreing Piraciaha	Indoor	Shopping Mall	
812	23302	SPLRA3541	-22.571	-47,40854	SP	Limeira	Ran Carlos Gomes, 1321 à" Centro - Shopping Plaza Limeira	Indoor	Shopping Mall	
813	51022	RIMCE2672	.22 3924	-41 807117	RI	Marse	Avenida Alaizio da Silva Gomes, 800 - Shopping Hara Macaé	Indoor	Shopping Mall	
814	6672	SPWSU4114	-23,6198	-46.62783	SP .	Sao Paulo	Praca Leonor Kauna. 100 - Shorning Plaza Sal	Indoor	Shopping Mall	
815	17347	SPWSU4114 SRRIMAD3044	-23,0176 -22,8725	-46,62783	RI	Rio De lanciro	Estrada da Portela. 99 - Shopping Pazz Sul	Indoor	Shopping Mall Shopping Mall	-
815	25424	ROPVO3087	-22,0723	-43,33892	RO	Porto Velho	Ay, Rio Madaira - Shopping Porol	Indoor	Shopping Mall	
817	30998	SPDDA0082	-0,/4024 -23,6914	-46.62315	SP	Diadema	Av. no statura - snopping rono veino Rus Graciosa, s/n	Indoor	Shopping Mall Shopping Mall	-
818	4151	PEBVG3243	-23,0914 -8.11721	-46,62313	PE	Recife	Rus Graewa, s/n Rus Padre Caravacsiro 777	Indoor		-
									Shopping Mall	-
819 820	19100	SRMASLS1547 SRRIBOT3052	-2,53375	-44,225756 -43,1766	MA	Sao Luis Rio De Ianeiro	Av. São Luis Rei de França, no (8, bairro Turú. Shopping Rio Anil. Ren Lauro Muller. 116	Indoor	Shopping Mall Shopping Mall	
	16828	SERIBOT3052 SPRPO2867		-43,1766 -47.80805556	K] SP					
821			-21,1816 -22,8144			Ribeirao Preto	R. São José, 933 - Shopping Santa Úrsula	Indoor	Shopping Mall	
822	17755	RJSGO2908 MGSI A 2095		-43,070443 -44,27115	RJ	Sao Goncalo	Rua Nova Jersey, s/n - Lote 1 - Shopping Sio Gonçalo	Indoor	Shopping Mall	
			-49,475			Sete Lagoas	Rua Otivio Campelo Ribeiro, 2801 (Shopping Sete Lagoas)		Shopping Mall	
824	6196	SPWCG3656 SPTBS4662	-23,6773	-46,69782 -46,77583	SP	Sao Paulo	Av. das Nações Unidas, 22540 - Shopping SP Market	Indoor	Shopping Mall	
825	6078				SP	Taboao Da Serra	Rod. Régis Bittencourt, Km 271,5 - Shopping Taboão	Indoor	Shopping Mall	
826	4409	PESAR3429	-8,03845	-34,871772	PE	Recife	Averida Agamenon Magalhaes 153	Indoor	Shopping Mall	
827	5675	SPBRE2844	-23,5043	-46,8346	SP	Barueri	Av. Piracema, 669 - Shopping Tamboré	Indoor	Shopping Mall	-
828	17643	SRRJRCP3029	-22,9217	-43,235102	RJ	Rio De Janeiro	Avenida Maracarii, 987 - Shopping Tijaca	Indoor	Shopping Mall	
829	27207	PRIND(398	-25,4786	-49,29361111	PR	Curitiba	RUA ITACOLOMI, 100	Indoor	Shopping Mall	
830	50922	PAPUP2033	-6,09166	-49,855555	PA	Parauspebas	Rodovia PA 275, Km 55 - Shopping Unique	Indoor	Shopping Mall	
831	14210	MGIIG4202	-19,4962	-42,56532	MG	Ipatinga	Rodovia BR-381 (Avenida Pedro Linhares Filho), Km 206 - Industrial - Shopping Vale do Aço	Indoor	Shopping Mall	
832	38688	SPVOS1097	-22,9816	-47,01263	8	Valinbos	Rua Paiqueré, 200 - Shopping Valinhos	Indoor	Shopping Mall	-
833	16875	SRR[BT]5048	-22,982	-43,361778	RJ	Rio De Janeiro	Avenida Ayrton Senna, 3000 - Shopping Via Parque	Indoor	Shopping Mall	
834	50781	SRR[BT]3023	-22,9987	-43,351193	RJ	Rio De Janeiro	Avenida das Américas, 3900	Indoor	Shopping Mall	
835	6104	SPWAP3569	-23,5501	-46,72125	SP	Sao Paulo	Av. das Nações Unidas, 4777 - Shopping Villa-Lobos	Indoor	Shopping Mall	-
836	16692	ESVTA3811	-20,3123	-40,287093	ES	Vitoria	Avenida Nossa Senhora dos Navegantes, 1440 - Shopping Vitória	Indoor	Shopping Mall	
837	6125	SPWBF3584	-23,5265	-46,67356	SP	Sao Paulo	Av. Antirtica, 380 - Shopping West Plaza	Indoor	Shopping Mall	-
838	44677	R[VRD2280	-22,5166	-44,107043	RJ	Volta Redonda	Rus Bris Magakli Fernandes, 300	Indoor	Shopping Mall	
839	5391	SPSOC2892	-23,4962	-47,4676	SP	Sorocaba	Av. Afonso Vergueiro, 1700/1766 - Sorocaba Shopping	Indoor	Shopping Mall	
840	5923	SPOCO2640	-23,5358	-46,7711	SP	Osasco	Av. dos Autonomistas, 1828 - Super Shopping Osasco	Indoor	Shopping Mall	-
841	27594	PRCTA0576	-25,4537	-49,288556	PR	Curitiba	Supermercado Angeloni - Avenida República Argentina,	Indoor	Supermarket	
842	6066	SPSZN2893	-23,5433	-46,3185	SP	Suzano	R. Sete de Setembro, 555 - Sazano Shopping	Indoor	Shopping Mall	
843	24394	DFIND0131	-15,841	-48,045194	DF	Brasilia	Rm 210, QS 1, Lt. 40	Indoor	Shopping Mall	-
844	5585	SPTTE2894	-23,0248	-45,5823	SP	Taubate	Av. Charles Schneider, 1700 - Taubaté Shopping	Indoor	Shopping Mall	
845	50759	DFIND0744	-15,8053	-47,86264	DF	Brasilia	SET SAFS QD 4 LT 1,0 ZONA CÍVICO ADMINIST	Indoor	Public Building	-
846	5489	SPSTB2869	-22,7489	-47,3677	SP	Santa Barbara Doeste	Av. Santa Barbara, 777 - Tiroli Shopping	Indoor	Shopping Mall	
847	45106	SRRJBOT3026	-22,9569	-43,176028	RJ	Rio De Janeiro	Rua Lauro Maller, 116 - Torre do Rio Sal - Complexo do RioSal (Shopping)	Indoor	Shopping Mall	
848	6643	SPWST2854	-23,6483	-46,7187	SP	Sao Paulo	Avenida Dr. Mario Villas Boas Rodrigues, 387 - Transamérica Espo Center	Indoor	Convention Center	
849	17063	SRR[CTO3033	-22,9045	-43,1736	RJ	Rio De Janeiro	Raa Dom Manoel, s/n - Tribunal de Justiça do Estado	Indoor	Public Building	
850	19113	SRMASLS0564	-2,5281	-44,3057	MA	Sao Luis	Praça Pedro II s/n	Indoor	Public Building	-
851	6461	SPWMM3914	-23,5872	-46,6585	SP	Sao Paulo	Tunel Ayrton Senna / Av. Vinte e Três de Maio - Av. Antonio Joaquim de Moura Andrade	Indoor	Tunnel	
852	6296	SPWIB3772	-23,5905	-46.6897	SP	Sao Paulo	Tunel Janio Ouadros / Ay, Presidente Juscelino Kubitscheck, s/n	Indoor	Tunnel	
853	15035	MGULA4250	-18.9107	-48.2588	MG	Uberlandia	Avenida João Naves de Ávila. 1331 - Uberlándia Center Shoroine	Indoor	Shopping Mall	-
854	24407	DFIND0144	-15.7608	-47.8705	DF	Brasilia	CPU DARCY RIBEIRO. 0 - ASA NORTE	Indoor	Pablic Building	
855	5331	SPSIC2877	-23.2169	-45.89305556	SP	Sao Jose Dos Campos	Av. Andrömeda. 227 - Vale Sul Shorring	Indoor	Shopping Mall	
856	45169	SRRIR15062	-22.823	.43 319779	RI	Rio De Ianeiro	Rua Itanera. Trevo das Marsuridas - Via Brasil Shorning	Indoor	Shopping Mall	
857	73212	SRCEELA0920	-3 79479	-38.479615	CE	Fortakza	Av. Washington Soares, 4335	Indoor	Shopping Mall	
858	6309	SPWIB3785	-3,19419 -23,6086	-36,4/3613 -46,02703	SP	Sao Paulo	Av. das Nações Unidas, 12551 - World Trade Center São Paulo	Indoor	Corporate Building	
950	24401	DEINIDO 18	15 9075	47.962390	DE	Banalia	BCA TRES DODERESLOTE ÚNICO 0. ZONIA CÍUICO ADMINIST	ladout	Bold's Boldiers	

Annex 2 - Remuneration												
Centrovice		Alderes	line and		Base Per Base Rever	erreins kilding Konservice F	er a Dur of Conclusion	Eas Sharing		Paulongh	Several Inc. (1971)	Increase of Regulation from Cost
1         AV0004         AV2/300/04         ZZ2/302/04         ZZ2/302/04         AUX           2         AV1171         AV405/07/157         Z/3/46/301         -45/3           3         SV2/300         AV405/07/157         -2/3/46/301         -45/3	27.92 R R Lawrence 27.92 R R Lawrence 29.92 R R Lawrence 20.97 R R R R R R R R R R R R R R R R R R R	Entrance Entranced Generative EDELATE CONTRACTOR FOR THE ENTRACTOR BUT ADDRESS OF THE STREET STREET STREET	Correlated Proc.	No. State No. All No.	1244 Exercises * Exactle Reservers 1244 64000 3.54 64000 2.19	00000000000000000000000000000000000000	1 0	printer and a	285.00 3.00 3.00	David Farth Promoty David Farth Promoty David Farth Programy David Farth Programy	NOT DEPORTATION	na toʻgʻalerresten blan Dander (BBL)(Mondo) 0.00 0.00 0.00
I         Control         Cont	Bits         Bits <th< td=""><td>Reception Society (2013) TORESSON - CERTER/OP Real-Mean Cherry, E Learnprise REAL PROF VELOC VELOC, 21, CASA VERIER As Means R. J. Reads, Works Velocity (2014)</td><td></td><td>30 8 40 (2+22=15)</td><td></td><td> 0. 80 0. 80 0.</td><td></td><td>(PRINCIPAL CO) 1</td><td>245.50 3.60 285.50 3.60 3.60 3.60 3.60 3.60 3.60 3.60 3.60 3.60 3.60 3.60 3.60</td><td>Third Farly Property Third Farly Property Third Farly Property Third Farly Property</td><td>Constraint (Constraint)     Constraint(Constraint)     Constraint(Constraint)     Constraint)     Constraint(Constraint)     Constraint)     Constraint(Constraint)     Constraint)     Constraint(Constraint)     Constraint)     Constraint(Constraint)     Constraint)     Constraint(Constraint)     Constraint)     Constraint(Constraint)     Constraint(Constraint)     Constraint)     Constraint(Constraint)     Constraint)     Constraint(Constraint)     Constraint)     Constraint(Constraint)     Constraint(Constraint)     Constraint)     Constraint(Constraint)     Constraint)     Constraint(Constraint)     Constraint)     Constraint(Constraint)     Constraint)     Constraint(Constraint)     Constraint)     Co</td><td>430 030 030 030</td></th<>	Reception Society (2013) TORESSON - CERTER/OP Real-Mean Cherry, E Learnprise REAL PROF VELOC VELOC, 21, CASA VERIER As Means R. J. Reads, Works Velocity (2014)		30 8 40 (2+22=15)		0. 80 0. 80 0.		(PRINCIPAL CO) 1	245.50 3.60 285.50 3.60 3.60 3.60 3.60 3.60 3.60 3.60 3.60 3.60 3.60 3.60 3.60	Third Farly Property Third Farly Property Third Farly Property Third Farly Property	Constraint (Constraint)     Constraint(Constraint)     Constraint(Constraint)     Constraint)     Constraint(Constraint)     Constraint)     Constraint(Constraint)     Constraint)     Constraint(Constraint)     Constraint)     Constraint(Constraint)     Constraint)     Constraint(Constraint)     Constraint)     Constraint(Constraint)     Constraint(Constraint)     Constraint)     Constraint(Constraint)     Constraint)     Constraint(Constraint)     Constraint)     Constraint(Constraint)     Constraint(Constraint)     Constraint)     Constraint(Constraint)     Constraint)     Constraint(Constraint)     Constraint)     Constraint(Constraint)     Constraint)     Constraint(Constraint)     Constraint)     Co	430 030 030 030
7 532672 57937C3542 2337180 444 8 53296 5792503735 2348425 447 8 657755 5792503735 2348425 447	47542 34P San Pado 21392 34P San Pado 21392 34P San Pado	BUA MANGEL DA MARANA PR BUA KANGA, Nº 521, HEROLEBRA DO O REALEMANT DE	Georgical MdCopporing Georgical XelCopporing Classifield RelCopporing	60 40	2.792 4,000 3.67 4,000	0 0. 0 0.	0 0 0 0		8.00 8.00	Third Party Property Third Party Property Third Party Property	5000000 5,44731 500000	0.80
10 X250 399719423 23.0796 463 11 X250 39921255 23.6407 463 12 X350 39921255 23.6407 453	A2700 SP San Pado 71275 SP San Pado 4000 SP J Pado	EA ANTONIO BERN HE. OLICARA BELINZENDO BEN DELENTRENE NO. JARÍO NO. BEN DE JUE TRUER NO. JARÍO NO. DO	Georgeoidal Bolt apparting Georgical Bolt apparting Georgical Bolt apparting	41 41 41	4,014 4,000 3.647 4,000 3.64	0.000000000000000000000000000000000000	0 0 0 0 0		8.00 9.00 9.00	Table Facty Property Table Facty Property Table Facty Property	473332 2,474.47 3,281.60	0.00 0.00 0.00
12         11356         970723400         21.6070         464           13         13177         97037311         21.4217         464           16         13170         97037311         21.4217         464           16         13170         19703731         21.6480         464           16         13170         197047048         21.6480         464           16         13175         197047048         21.6480         464           16         13175         197047048         21.6480         464           17         76405         ENJ14778         47113         305	2007 1 37 Sar Pade 2007 1 37 Sar Pade 2702 37 Sar Pade 372 37 Sar Pade	Jac Response 1997 Characterized and Constraints of the Constraint of the Constrai	Reacher Mass Generated Mill segmenting Generated Mail segmenting	30 30 40	2008 6480 3041 6480 12 6480	8	0 0 0 0 0 0 0 0 0		5.00 5.00 5.00 5.00	Table Forty Property Table Forty Property Table Forty Property	NOT INFORMED NOT INFORMED LLONGO	0.8
16 53367 5PEC00743 237360 445 17 87603 KNJ3PT38 447113 355 18 87903 8X3PLA022 37362 364	2020 SP San Paulo 2023 KN Japi 42388 CE Portulos	Ran D.R. Analonis Number de Carendos, 103 - 54 Nin Homando: Nin Paulo: 307 Ran Rick Vens, Qui, Jupi Ran Venemi, Lini, Q20 - Mennico.	Gueseleld Mdf sepporting Gueseleld Xelf sepporting Resellery New	40 30 (3+3+1,8)	5.53         4,000           13         4,000           30.156         4,000           200008         4,000           NOT DOPOSIMID         4,000           50.71         4,000           54.71         4,000           54.77         4,000           54.77         4,000	10 0. 10 0. 10 0.	0 0 0 0		9.00 9.00 9.00	Thai Fely Negry Thai Fely Negry National Million Thai Fely Negry Thai Fely Negry Thai Fely Negry	1201.00 NOT 109-CROMID 11(08.00 11(08.00 14(08.03 N/A 4(08.03 4(08.03 4(08.03 4(08.03) 4(08.03)	6.00 6.00 6.00
19 57966 322391,0114 3,7598 364 28 57917 302391,0139 3,7523 360 28 57958 302391,0120 3,7523 360	0076 CE Portelese NIRO CE Portelese NIRT CE Portelese	Islifies has Low - Bas Post Shi Rhom, OP Epies Bit A PENTO SLIDE BA737 AVENDES AVENTO NE SALIS, 197,	Reacher Mass Reacher Mass Reacher Mass Reacher Mass	2.2 4 4	2.2008 4,000 NOT POPORABLE 4,000 NOT INFORMED 4,000	8 0. 8 0. 8 0.	0 0 0 0		9.00 9.00 9.00	Thind Party Property Thind Party Property Thind Party Property	400333 NOT DPG808D NOT DPG808D	0.30 0.30 0.30
22 MEMO CEMETADE ADMIN 396 29 64090 MCMEDO'N 1933MAN 416 38 64793 MCREGEO 1938AN 475	00200 CE Mulanga 41795 MC Samana Do Earlos 82797 MC Usardos	Rodevic GL 000 (broken Makaga) Assumbia, tyle: Masses do GETEL Bar de Dargebra, tyle Rod 100, 084, K= 125	Gravital Add appoint Gravital Add appoint Gravital Add appoint	40 36 70	1471 4,000 56,87 4,000 6,73 4,000	8 6 8 6 8 6	0 0 0 0 0 0		8.00 9.00 9.00	Third Party Property Not Informed Third Party Property	UNUT INCREMENT NOT INCREMENT NOT INCREMENT NOT INCREMENT AURIENT AURIENT AURIENT AURIENT AURIENT AURIENT	6.30 6.30 6.30
20         547561         385304,0336         0311256         443           20         649707         79255,68156         2512753         561           21         649716         7925,68156         2512753         561           22         649716         7925,68156         2512753         552	20026 MG Auguste De Linu. 15806 PE Plonte Connu. 00305 PE Pulsua	R. Coreido Lores Alon, Ma BEA, MAMMATTROO DE ARANDO A AZ BEA MERCITÃO CORREA MOUTRA AN	Greenield Roll appendig Resolute Mint Greenield Roll appendig	40	3.76 6.800 2.47 6.800 11.98 6.800	8	0 0 0 0 0 2	(PENDENCO) 2,	9.00 9.00 900.30	Table Party Property Table Party Property Net Informed	1140.71 2.5600 NOT INFORMED	0.30 0.30 0.30
2 60403 PRC3067 254203 40 2 60403 PRC3067 254203 40 2 64232 PRC3066 254306 40 3 64234 R644473 55603 556	2000 PE Carilla 2042 PE Carilla 6802 EA Ingratus	RUA URALINA DI DI ARABAL, IN JUNITARI DI CONTO PREFIRA ATTENDA PREMA TENI AV JUTATI MAGDALISTIS, ON	kuches bar Grandial & Korperen Grandial & Korperen Kuches & Korperen Grandial & Korperen Kuches & Korperen Grandial & Korperen Grandial & Korperen Grandial & Korperen Grandial & Korperen Grandial & Gorperen Grandial & Gorperen G	(1,2+4,2+4,2) (3+3+3) 30	C7         Δβ           15         5.5           15         5.6           16         6.6           17         6.6           18         6.6           19         6.6           10         6.6           10         6.6           11         6.6           12         6.6           13         6.6           14         6.6           15         6.6           16         6.6           17         6.6           18         6.6           19         6.6           10         6.6           10         6.6           10         6.6           10         6.6           10         6.6           10         6.6           10         6.6           10         6.6           10         6.6           10         6.6           10         6.6           10         6.6           10         6.6           10         6.6           10         6.6           10         6.6	0 0. 0 0.	0 0 0 0		440 440 440 440 440 450 450 450 450 450	Nut i kay Kapagi Tan Lihanadi Kan Jianadi Kaya Kapangi Kaya Kapangi Kana Kapangi Ka	NUL DOCUMUND NUT DOCUMUND NUT DOCUMUND N(A N(A 2)0844 NUT DOCUMUND	6.80 6.80 6.30
N         DFMON         Addressen         Figure         Addressen         Addressen <td>27975 EA Sarany 27975 MG Barbarea 27964 FE Casible</td> <td><math display="block">\begin{tabular}{lllllllllllllllllllllllllllllllllll</math></td> <td>Gweeldd Adrageering Beeling Mas Roches Mas</td> <td>413 4</td> <td>2.17 E,000 142.57 4,000 2.19 4,000</td> <td>8 6 8 6</td> <td>0 2 0 0</td> <td>IPENDENC CR 2/</td> <td>530 500.30 5.00</td> <td>Third Farty Property Third Farty Property Third Farty Property</td> <td>2,05.44 NOT DPCRMID</td> <td>0.80</td>	27975 EA Sarany 27975 MG Barbarea 27964 FE Casible	$\begin{tabular}{lllllllllllllllllllllllllllllllllll$	Gweeldd Adrageering Beeling Mas Roches Mas	413 4	2.17 E,000 142.57 4,000 2.19 4,000	8 6 8 6	0 2 0 0	IPENDENC CR 2/	530 500.30 5.00	Third Farty Property Third Farty Property Third Farty Property	2,05.44 NOT DPCRMID	0.80
S         OPTIM         PROJ. NAM         23.200%         111           S         GATM         RALE AND         23.200%         114           S         GATM         RALE AND         24.10%         114           S         GATM         RALE AND         24.10%         21.00%           S         GATM         RALE AND         24.10%         20.00%           S         GATM         RALE AND         24.10%         20.00%           S         GATM         RALE AND         24.10%         20.00%           S         GATM         RALE AND         20.00%         20.00%         20.00%           S         GATM         RALE AND         20.00%         20.00%         20.00%         20.00%           S         GATM         RALE AND         20.00%         20.00%         20.00%         20.00%         20.00%         20.00%         20.	00344 EA Boos 2020 PE Cambo 2020 PE Onicato	ANTINERA PROCEAMACAO, A/N BUA COMENDARIOR ARCENT, TRA HETTEL BES BUA NO FRANCO VI	Gueselidd Self sepporting Reacher Mass Gweselidd Self sepporting	40 (9+4) 45	8.12 4,880 2.55 6,880 13.57 6,880	8 0. 8 0.	8 0 8 0		0.00 0.00 0.00	Not Informed Third Party Property Own Property	NOT DAVGRADD A/7278 NOT DAVGRAD NOT DAVGRAD N/A	0.30 0.30 0.30
9 64020 29625000 244030 523 30 64030 3204000000 1230000 364 8 64033 20040000122 255500 564	M000 PE Comm. Mouras A006 E3 Micador 2008 PE Par Da Isaan	Em Larendo Borros 1930 Como Carros Muneto, FE AMARIA DOM KARO V. 686 Bas Relations 700	Rection Meet Greenheid Mill segrenting Rection Meet	B           0+4           3           0+1           0+1           0+1           0+1           0+1           0+1           0           0           0           0           0           0           0           0           0           0           0           0           0           0           0           0           0           0           1	6,320 + 6,420 4,490 844 4,490 9,72+1,475 4,490	80 00 80 00 80 00	0 0 0 0		3.00 9.00 9.00	Third Party Property Third Party Property Third Party Property	N/A NOT ENCOMME ANNUA NOT ENCOMMED	0.80 0.80 0.80
60.0014 PED/20106 23.33777 343     60.013 PECT0051 23.3597 413     60.013 PECT0051 23.3597 413     60.123 PARE01100 123.079 443	0014 PE For Dr. Ignue 2020 PE Carilla 20214 EA Eserviso	Ban Rad Bardwar, 475 AVENEAA PARANA, 1925 BUA AREE 20 FAREAG, 485	Reacher Mass Reacher Mass Gweenleid Self-segreering	(1+1) (1+4+4+4+4) 30	(0,77 + 1,32) 4,400 4,00 4,000 3.48 4,000	80 0. 30 0. 30 0.	0 0 0 0 0 2	presson or z	9.00 9.00 900.30	Thind Party Property Titled Party Property Net Informed	NUT DOCIMUM UNIT DOCIMUM NUT DOCIMUM	0.80 0.80 0.90
48 64236 PEL3A424 233000 311 48 642071 E0050073 2537809 334 48 64293 PEC330090 2543542 493	14330 PE Londona 43772 ES Pasied/Do-Jonamo 26356 PE Custolia	As Digenspela, 29 - Joshin Highwayala Bir Ajtulari Dir. Castributi B, 8/N Bir Manna Tores, 70	Reschap Mass Gweedidd Rell apporting Reschap Mass	(1+4) 2/ M (1+4)	43 (5)60 + 6)60 + 6)603 4,080 18:33 4,080 (5)8 + 5,47) 4,080	80 0. 80 0. 80 0.	0 0 0 0 0 0		9.00 9.00 9.00	Thind Farty Property Own Property Thind Party Property	NOT INFORMED N/A 429671	0.80 0.80 0.90
C         Party         Education         12082         12082           0         KKEP         KKEP         XXXV         XX           0         KKEP         XXXV         XX         XX           0         KKEP         XXXV         XX         XX           0         TABLE         AVESS         XXXV         XX           0	N106 EA Salvador M111 EA Salvador M135 PE Salgado Hibo	RUA DIA MACHEMARA, 4 RUA PROPERSION AND THE SAME AND A DESCRIPTION OF A DE	Reachen Mani Reachen Mani Gweedeld Melf segmenting	3.8 (7+7+7) 30	3.22 6,000 3.54 6,000 3.75 6,000	8 0 8 0 8 0	8 0 0 0 8 0		8.00 9.00 8.00	Third Party Property Third Party Property One Property	NOT DECEMBED NOT DECEMBED N/A	0.30 0.30 0.30
N         T4950         MP00236         5422%         111           N         T4970         PURCON         AUX8         MA           N         T4971         PURCON         AUX8         MA           N         T4971         PURCON         AUX84         AU	24773 AL Presiden 8987 P3 Revie 8928 P3 Revie	MYDNDA MOREPP, AN Bes Veren Bes Pele Consumers 2012 (2012) W GCPC ACAMENNON MACHINERS 2016 (1997) IEEO W GCPC ACAMENNON MACHINERS 2016 (1997) IEEO	Greenield Reit segmenting Reschert Mass Reschert Mass	00 (3+3+5) (3+6)	24 4.000 NOT INFORMED 4.000 NOT INFORMED 4.000	8 0 8 0 8 0	0 0 0 0 0 0		9.00 9.00 9.00	Not Informed Third Party Property Third Party Property	NOT DECRMED MILLS NOT DECRMED	0.80 0.30 0.30
N 74307 3793639 2033622 67 N 74308 37936 203200 67 N 74420 97955728 235500 46	4007 SP Frame S200 SP Frame 61400 SP San Pade	Res Volumins Marie Marie 2011 - Biglewiysle An Frendens Varyn 601 - Gladar Nova, Franz S.F. Res Gimbers Die Jahri 531 - Ville Moran	Reachen Mass Reachen Mass Reachen Mass	(3//2+3//2+3//2) (9+3+9) 2 19	(1,58 + 1,58 + 1,50) 4,000 (211 (3,040-1,200) 4,000 2,007 4,000	0 0. 0 0. 0 0.	0 0 0 0		8.00 9.00 9.00	Talad Party Property Talad Party Property Talad Party Property Talad Party Property Talad Party Property Talad Party Property	3/9/121 26/7538 332546	0.30 0.30 0.30
S         Tester         Match         M	Att 00         SP         Sam Paulo           5062         SP         Owners           5053         SP         Sam Enrounds The Campo	Ras da Padrovies, 83 - Sardan Nondorsu, Nie Pada - 59 Ras Rosa Dringde Phopis, 158 - Vils Van Ras Rey Badron, 200	Reachery Mass Reachery Mass Reachery Mass	4 6 (3+3+3)	3.078 6,400 2.809 6,000 (5,04 + 1,08 + 0,03) 6,000	8 6. 8 6. 8 6.	8 0 8 0 8 0		9.00 9.00 9.00	Think Party Property Think Party Property Think Party Property	Nor aviewable 1615.00 2615.00 2617.0	0.30 0.30 0.30
40 7141300 307679-2003 2334097 443 40 7141375 305604637 2317126 443 42 7141375 307604637 2317126 443	MDM         NP         Nam Paulo           MDMS         NP         Nam Research The Campo           MDMS         NP         Nam Paulo           MDMS         NP         Nam Paulo	BUA (GAO RATECTA DE LAC BEDA, 201 Euro Manisted, 193 An Promiser Manusco, 227, Ann Brown, Nr. Buds ; 52	Reachap Mass Reachap Mass Reachap Mass	(8+0+1) (8+0+0) (8+3+9)	NOT INFORMED         4,000           (2,04 + 1,67 + 0,03)         4,000           2,17         4,000	80 0. 30 0. 80 0.	0 0 0 0		9.00 9.00 9.00	Tani Pery Pepery Thial Pery Pepery Tani Pery Pepery	3/963.82 12/976.36 4/7536	0.30 0.30 0.30
	6750 SF San Fields MICI SF San Fiernals De Carrens 27411 SF San Castano De Sul	Bas Bood Kenso, 101 y Penner Balannin Tomas Dahon Ran Noles Annines, 128 y Denselinde y Xie Bennelis die Campo Ran Andreas Pennes, 118 y Same Annines - Sin Cantaes die Sul	Resilver Meet Resilver Meet Resilvey Meet	(0+0+1) 7.87 (0+20) (0+0+7)	1.0.00+0202+0.224+1.7% 4.000 0.003+0.228 4.000 0.002 4.000	8 0 8 0 8 0	0		9.00 9.00 9.00	Table Farty Property Table Farty Property Table Farty Property	K16423 629335 7,98626	0.30 0.30 0.30
46 741073 3PERVISO 233409 440 47 741032 3PECEN41 234366 444 48 744342 3PECEN41 234396 444	0011 3P San Pado 0083 3P San Pado 9030 3P Mag Mana	ANTENDAL POLITATIA, 600 Ban Ekieven, 274 - Vale Pauline Ban Convert Uterinisis Formities Mores Adverse, 80 - Italile	Realisy Mas Realisy Mas Realisy Mas	(1+1+1) (1+1+1) (1+1+1+2)	3.60 Kernel Control Co	8 6 8 6 8 6	0 0 0 0 0 0	(PENDENC OF 1.	9.00 9.00 9.00	Third Party Property Third Party Property Third Party Property	NOT INFORMED 2,71841 N/DL47	6.80 6.80 6.90
0 14175 000000 20000 475 70 14179 000000 201000 475 1 14180 000000 201000 450	80707 SP Ellower Perm 41877 SP Sam Farlian Dorner 8044 SP San Jone Don Campon	Kan Brownshow do Campon, 2255 BAA THE OF DOM: NAME, YM - CONTROL NAME AND A STREET AND A STREET AND A STREET BAR Fundation Minington, 2014 Control	Realisy Mas Realisy Mas Realisy Mas	()+3+8) ()+4+6) 4 1.5	(0,47 + 3,06 + 1,71) 6,480 (0,3 5,072 - 0,026 + 1,712) 6,480 1,47 6,480	0 0. 30 0. 30 0.	0 1 0 0 0 1	IPENDENCION 1,	280.80 0.00 280.80	Third Party Property Third Party Property Third Party Property	NOT DEVOLUED 1/X12311 NOT DEVOLUED	6.80 6.80 6.90
7 74100 3P00072 211701 440 7 74100 3P00072 211701 440 7 74100 3970079 210000 40		Jornals No. J. (2014). BULLY (READER: 3P Jornals No. Jon, (2014). Jones Pair War Res Suscession de Franç 2014 - Marque - Samer Jay Jones VIII, VIII, N.	Institution         Institution           1         1         1           1         1         1         1           1         1         1         1         1           1	$\begin{array}{c} \rho_{1} = \rho_{1} \\ \rho_{1} = + \rho_{1} \\ \rho_{1} = + \rho_{1} \\ \rho_{1} = + \rho_{2} \\ \rho_{1} = + \rho_{2} \\ \rho_{1} = + \rho_{2} \\ \rho_{2} = + \rho_{2} \\ \rho_{1} = + \rho_{2} \\ \rho_{2} = + \rho_{2} \\ \rho_{1} = + \rho_{2} \\ \rho_{2} = + \rho_{2} \\ \rho_{1} = + \rho_{2} \\ \rho_{2} = + \rho_{2} \\ \rho_{1} = + \rho_{2} \\ \rho_{2} = + \rho_{2} \\ \rho_{1} = + \rho_{2} \\ \rho_{2} = + \rho_{2} \\ \rho_{1} = + \rho_{2} \\ \rho_{2} = + \rho_{2} \\ \rho_{1} = + \rho_{2} \\ \rho_{2} = + \rho_{2} \\ \rho_{1} = + \rho_{2} \\ \rho_{1} = + \rho_{2} \\ \rho_{2} = + \rho_{2} \\ \rho_{1} =$	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	0 0 0 0 0	1 1 0	(PENDENC OF 1.	1200 1200 1200 1200 1200 1200 1200 1200	mini for fuery and for fuery basil for a fuery for a f	NIT PAPENABD TO ALL TO ALL	00 0.00 0.00 0.00 0.00
101000         101000         40           7         741700         19713400         231700         440           7         741700         39113461         223486         411           7         741700         39113461         223486         411           7         74207         3244600         127040         440	1200 XF News 1200 XF News 1207 XF Newser 12097 EA Salvador	Kay Terreteri, Madel, M. J. Manne, Kay Terreteri, Madel, M. J. Manne, Ras Join Antonio Stano, N. M. Born Koro, Samari Rich Rick Net, R. N. GUTZLA, 134	Easting Mass Easting Mass Easting Mass	(1+3+3) (1+3+3) 3 10	1.249 4.480 3.249 4.480 3.78 4.480 2.72 4.480	0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0		9.00 9.00 9.00	Thind Farry Property Thind Farry Property Thind Farry Property	NOT INFORMED 5,0006 7,004.73	030 030
79 754209 X28X8(79132 12109)2 34 80 754250 X28X8(79132 141309) 370 80 754250 8348(1440) 114821 375	MUIT EA Michaelor MURE EA Grouppy 191379 EA Ein End	Alamada Preiro de Conseptes, MD.A, Buels Mans Bara da Tgeng, S/N = Conse BUA CORDINITA EDINITINATION/TROJ/N	Gaussield Xelf organing Gaussield Xelf opposing Gaussield Xelf opposing	NI 41 71	11.76 4,800 24 4,800 34.36 4,800	8 6 8 6 8 6	0 0 0 0 0 2	pressure og 2	3.00 3.00 300.30	Third Party Preparty One Perpety No Informed One Perpety	N/A NOT INCOMED N/A	0.30 0.30 0.30
80 7142007 39987907008 23.88058 444 80 7142003 3999744158 23.8403 443 80 7142008 3998742555 23.8500 444	67072 37 New Fash Mid74 37 New Fash 67497 37 New Fash	Armida Nove de Juleo, N2H - Junion Paulon - Nos Paulo Ros Campple, 728 - Vile Gromes Cardon - 30 Ros Dis Homeon de Mede, 1286 - Paulons	Reachop Mass Reachop Mass Reachop Mass	10 (3+3+3) (3+3+4+4)	2.654 6,885 2.599 6,885 3.561 6,885	0 0. 0 0. 0 0.	0 0 0		0.00 0.00 0.00	Think Furly Property Think Furly Property Think Furly Property	11,031.97 3,038.30 8,024.28	0.00 0.00 0.00
xx         7142449         387072314         2333722         464           36         7142344         370702000         2345000         444           37         7142343         370702000         2345000         444           37         7142343         370702011         2345203         444	10111 37 Jun Pado 0020 37 Jun Pado 01211 37 Jun Pado	Han Dis Horsewe de Moles, 352 - Freders - Mar Paule Kan das Ommens, 100 - Pig Jahagara - Nar Paule - 52 <sup>o</sup> Kan Marces, 201 - Vie Steets Conview. 52 <sup>o</sup>	Realing Mast Realing Mast Realing Mast	(1+1+4) (1+3) (1+3)	2.493 4,486 2.499 4,486 2.549 4,486	a 6. 20 6. 20 0.	0 0 0 0		1.00 1.00	salal Party Poperty Third Party Poperty Third Party Poperty Wald Party Poperty	Kartini Grad Sa Ganana NOT Bercecher	0.00 0.00 0.00
m         714266         STROPS2         214300         460           R         714273         SPRELBONC         214300         460           R         714275         SPRELBONC         214300         460           R         714275         SPRELBONC         234300         460           R         714275         SPRELBONC         234400         460	1000 27 Jan Pado 1000 27 Jan Pado 1000 27 Jan Pado 1000 27 Jan Pado	An Lerennie de Yinei 772, Vile Coursei Alemais Barie de Lerenne (72), Careren Elizens Nie Pade - 39 Arende Cargos Edense, 58 - Course Histories de Nie Pade Receit	Exclose Mass Exclose Mass Exclose Mass Factors	(1+1+1) (1+1+1+1) (1+1+1)	4387 4480 4483 4480 2094 4480 7497	M 0. 00 0.	0 0 0		1.00 0.00 0.00	Third Party Property Third Party Property Third Party Property Third Party Property		0.00 0.00 0.00
		nan Compression namellin, 164 – Molt Yam Ron Eulyre Felinwer, 2007 - Via Jawei, Sin Paulo - M Ban die Glacine, 701 – Laboraturk - Sins Paulo Ran Dromeine, 2021 – Laboratur Via Maria	Ranhon Man Ranhon Man Ranhon Man Ranhon Man Ranhon Man Ranhon Man Ranhon Man Ranhon Man Ranhon Man Ranhon Man	(3+3+3+3) (3+3+4+3) (3+4+4) (3+4+4)	25.5% 4,8% 27.5% 4,8% 2.52 4,8% 2.0%	- 0. 30 0. 80 0.	0 0 0 0 0		3.00 3.00 3.00	Think Party Property Think Party Property Think Party Property Think Party Property	NET INCOMED A/98/32 128279	0.80 0.80 0.80 0.80 0.80 0.80 0.80 0.80
	6020 NP Nor Pade 6020 NP Nor Pade 6020 NP Nor Pade	Ran Vergueren, - La Martina - Ann Paran Ran Vergueren, 1661 - Vin Martan Ran Vergueren, 200 - Judien da Cloran Ran Feynder Queren, 200 - Vin Martana, - MP	Easing Mass Easing Mass Easing Mass	(3+3+3) (3+3+3) (3+3+3) (3+3+3)	1,00, 0, 27 + 1,03 + 1,14 4,400 (76, 0,778+1,230+1,770) 4,400 2,050 4 + e=	0. 20 0. 20 0.	0 0 0 0 0 0 0		9.00 9.00 9.00	Ne klassed Gen Papery Back Tarp Speep Back Tarp Speep Stack Tarp Speep Tark Tarp Speep	UIXIN UIXIN NOT DOCEMED	0.00 0.00 0.00
9 71256 579217565 233350 443 9 712567 579239575 233380 463 30 712567 579239575 233380 463	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Eas Teneless, 201 - Via Iprozes - No Paule - 57 Has Manufan, 351 - Via Romans - 57 Eas Observation Ann. 751, Janes Gaulia No. Paule - 37		Byteley         β + 4 + 6           β + 4 + 6         β + 4 + 6           β + 4 + 6         β + 4 + 6           β + 4 + 6         β + 4 + 6           β + 4 + 6         β + 4 + 6           β + 4 + 6         β + 4 + 6           β + 4 + 6         β + 4 + 6           β + 4 + 6         β + 4 + 6           β + 4 + 6         β + 4 + 6           β + 4 + 6         β + 4 + 6           β + 4 + 6         β + 4 + 6           β + 4 + 6         β + 4 + 6           β + 4 + 6         β + 4 + 6           β + 4 + 6         β + 4 + 6           β + 4 + 6         β + 4 + 6		80 00 80 00 80 00			8.00 8.00 8.00	Thind Farty Preparty Thind Farty Preparty Thind Farty Preparty	With West State           With West State           With State	0.30 0.80 0.97
1         2 <th2< th=""> <th2< th=""> <th2< th=""> <th2< th=""></th2<></th2<></th2<></th2<>	2005 37 New Paulo 2000 37 New Paulo 2000 37 New Paulo 2000 37 New Paulo	In closed Charles de Garde 200 , Da Sta Dissioner In Convert Charles de Garde 200 , Da Sta Dissioner E Jud Jugo de Males, G. Vila Manan Ras 20 de senzo, 107 , Como Hancone de Sta Parlo	Easthean Mass Easthean Mass Easthean Mass	1 (1+1+4) (1+1+4) (1+1+7)	1000 (3.54=3.555+1.000) 4.400 3.000 (3.54=3.555+1.000) 4.400 3.500 4.400	0 0,00	0 0 0 0 0		5.00 5.00 5.00	nam Fariy Property Think Fariy Property Think Fariy Property Think Fariy Property Think Fariy Property	X/65.M8 X/56.4.5 3,928.83	030 030 030 030
188 714262 878584124 - 234879 - 444 188 714268 378584298 - 234787 - 444 188 714268 3785842998 - 234189 - 444	43343 3P New Paulo 05640 3P New Paulo 06600 3P New Paulo	Paga Dourse Juin Mondos, G. Liberalade BUA MAIDO, 70 - 38000MA Av dos Insario, 88 - Industrycels, Nar Paulo - 8P	Rankap Mas Rankap Mas Rankap Mas	(5+6+6) (5+6+4+6) (30+10+6)	1,62,03,48+0,39+0,39) 4,480 2,80 4,480 1,714 4,480	80 00 80 00 80 00			8.80 8.80 8.80	Thind Futy Property Thind Futy Property Thind Futy Property	NOT INFORMED B(2028) T(N274	0.30 0.30 0.30
bit         Ti-42647         SPECMATING         23.0003         444           bit         Ti-42646         SPECMATING         23.09751         464           bit         Ti-42646         SPECMATING         23.09751         464           bit         Ti-42647         SPECMATING         23.09020         464	0005 37 Net Pado 0023 37 Net Pado 0787 39 Net Pado	Alamada Jangyon, 2003 - Massea - 30° Ban Pendenor Marsies Januara, 200 - VI. News Concepto Kan Pendanoy, C77 - Janie Blas	Reachop Mass Reachop Mass Reachop Mass	(3+3+3) (1+1) (3+6+6)	2.429 4,880 2,129 (9,629 + 1,50) 4,480 3.714 4,480	80 00 80 00 80 00	0 0 0 0		0.00 0.00		25,075.56 3,083.00 6,000.00	0.00 0.00 0.00
10 714250 07050072 2141167 447 10 714250 07050407 214760 444 10 714250 07050407 214760 444	20403 34P San Paulo 42000 34P San Paulo 71083 34P San Paulo	Aronial Reside de Moore 2006, '5 – Vila Mesanda Res Undersitere de Paris, 2718 - Januar, Nac Paris - 597 Res Perciever Marcolate de Cherdin, 2011, Chinese Janes Aufrie - 597	Reachop Mass Reachop Mass Reachop Mass	(1+1+1+3) (3+3+3+4) (1+1+1-1)	5081 6,895 238 6,895 2462 6,895	xi 6. xi 6. Xi 6.	0 0 0 0		8.00 9.00 9.00	Third Party Property Third Party Property Third Party Property	1,500.78 NOT INFORMED DATAS	0.30 0.30 0.30
113         754290         87896,8103         254180         444           114         754292         87896,8117         2541292         444           115         754298         D99358777         1645139         466	4330 SP Sav Pade 4380 SP Sav Pade 0584 DP Break	Bas Brawn Brian, 100. This do Saide, Sin Pader, M. Ar. Islandson, V. S. Jaide, J. Str. Pade QUDEA 214, CONG 1, CASA 17, ANNUAL MARKS.	Reschen Mass Reschen Mass Reschen Mass	3 3 3	1061 6.895 (2.72 + 166) 6.895 1.491 6.895	8	0 0 0 0		9.00 9.00 9.00	Third Party Property Third Party Property Third Party Property	4206.70 7208200 1,288200	0.30 0.30 0.30
116 74400 3P0C1002 214704 46 117 744426 3P0C0064 2354700 464 118 744802 3P0C0064 454	66627 3P San Pado 65800 3P San Pado 65211 3P San Pado	Ar. Panala Press, 2011 VI. Nerve Carbonissina Nor Panis An Angelin, 2015. Name Cardin, Sair Panis. 20 Avanish Induscipatio, 1112 Elabor New Librario, Plansis, Panis, No. Panis.	Realisy Mass Realisy Mass Realisy Mass	$\begin{array}{c} 3\\ \beta+4+0\\ (0,7)+3,0\\ (0,7)+3,0\\ M,27\\ M,27\\ M,27\\ M,27\\ M\\ (2,1+2,0)\\ (3+4+0)\\ M\\ M\\ M\end{array}$	(3,12+6,03+6,07) 6,000 (3,623 + 6,638) 6,000 2,254 6,000	0 0. 30 0. 30 0.	0 0 0 0 0 0		9.00 9.00 9.00	Third Party Property Third Party Property Third Party Property	6,034.09 Bi,835.41 13,738.83	6.80 6.80 6.30
107 14407 2000 2007 3000 3000 30 128 714002 PER2013 30001 30 128 714023 8270608 273000 40 179 714026 8270608 273070 40	2020 PE Resile 2020 PE Resile 2020 E Prosecolo	BAR A BROCH MAN, HIS , H	Reachery Mass Reachery Mass Reachery Mass	4 (23+2,0) Redect	NOT INFORMED 4,000 (0,00+6(2) 4,000 (0,00+6(2) 4,000				8.00 9.00 9.00	One Property One Property Third East Descate	N(3 N(3 N(3)	0.80
121 714200 80792048 204400 353 121 714006 8010748 20407 803 120 714001 PROJECTS 252978 313	XIXI         SC         Planneyeds           XIXI         K         Fanneyeds           XIXI         K         Planneyeds           XIXI         FK         Camba           XIXI         FK         Camba           XIXI         FK         Camba           XIXI         FK         Fanneyeds           XIXI         FK         Fanneyeds           XIXI         FK <td>Ran Xuein Caminin, s/s<sup>a</sup> Ban Tanahonin, s/s<sup>a</sup> Commo Ran Krish Camilton Mitter PAREN S/N</td> <td>Generalishi Meli seggenting Generalishi Meli seggenting Generalishi Meli seggenting</td> <td>N N</td> <td>1.55 4,400 NOT DECEMBED 6,400 2.72 6,400</td> <td>8 0. 8 6. 8 0.</td> <td>0 0 0 0 0 0</td> <td></td> <td>3.00 3.00 3.00</td> <td>Pontaine Pore of Charge One Persporty Net Informed</td> <td>N/A N/A NOT DOCKMED</td> <td>0.30 0.30 0.30</td>	Ran Xuein Caminin, s/s <sup>a</sup> Ban Tanahonin, s/s <sup>a</sup> Commo Ran Krish Camilton Mitter PAREN S/N	Generalishi Meli seggenting Generalishi Meli seggenting Generalishi Meli seggenting	N N	1.55 4,400 NOT DECEMBED 6,400 2.72 6,400	8 0. 8 6. 8 0.	0 0 0 0 0 0		3.00 3.00 3.00	Pontaine Pore of Charge One Persporty Net Informed	N/A N/A NOT DOCKMED	0.30 0.30 0.30
10         Ποριος         Λ. Κ.	Milli MC Plenaesersik MIRI MC Timaman MIRI MC Plenaesersik	As for Brown, 307 E-mails Grind, star R. do Cableloy, T. H.Z. ENNTINGL	Resolute Mass Gweneled Pros Resoluty Mass	6 60 3.2	143 4486 3.3 4486 1.45 4,466	8 0 30 0 30 0	0 0 0 0		5.00 5.00 5.00	Think Party Responsy Think Party Responsy Think Party Responsy	NUT DEGRAMED NUT DEGRAMED NUT DEGRAMED NUT DEGRAMED NUT DEGRAMED NUT DEGRAMED NUT DEGRAMED AUTOR AUTOR AUTOR AUTOR AUTOR NUT DEGRAMED	0.30 0.30 0.30
120         T44645         SCINSING         2741175         400           130         T14048         PEO/20221         253096         543           131         T14028         PEO/20221         253096         543           131         T14028         PEC230143         254526         493	20342 SC Flostavepols 2086 PE For Do Igano 2030 PE Cambo	Eas Automor Marsais, of 42 Base Carlos Sanatamayon, 2015 Bit A (COQUER) CALFER, 437	Reselver Mass Gweedeld Mill segresting Reselver Mass	(1+0+0) 13 7.5	(1,28+4,28+4),29) 6,486 3.08 6,486 2.18 6,486	10 0. 10 0. 10 0.	0 0 0 0 0 0		5.00 5.00 5.00	Think Party Property Think Party Property Think Party Property	NOT INCOMED NOT INCOMED NOT INCOMED	0.30 0.30 0.30
132         754020         PBC23003         .25.4036         .49.2           131         7540307         PBC230734         .25.42303         .49.2           131         7546423         C391.40791         .3.73784         .36.4	2001 PE Cambo 2017 PE Cambo 2012 CE Portdoo	Ros Fedéra, 63 Ros Ven de Ros Finance, 803 Avendale Wangers Nource, 4 <sup>6</sup> 30	Reachop Mass Reachop Mass Reachop Mass	(1+3) 6 1.46	[0,47 + 1,17] 4,400 1.42 4,400 3.29 4,400	80 0. 80 0. 80 0.	0 0 0 0 0 0		9.00 9.00 9.00	Think Forty Property Think Forty Property Think Forty Property	32,548.78 4,000.00 8,310.83	0.30 0.30 0.30
118         T4467         PERPCONT         AU025         342           1%         T47108         BAPEDA16         Jan980         360           1%         T47108         BAPEDA16         Jan980         360           1%         T47108         BAPEDA16         Jan980         360           1%         T47109         BAPEDA16         Jan980         360           1%         T47100         BAPEDA16         Jan980         Jan980         Jan980           1%         T48702         BAEDA1700         ZAL1011         460	80756 PE Revie 00785 EA Palse Assaulte 48071 EA Condison	BULK RAAT, ANDERO, 85 Evental Pers Barriera, 8/05 Ray She de Bes Yong	Reachery Mass Gweedield Rell segmenting Gweedield Rell segmenting	(1+4+4) 30 18.1	NOT DECEMBED 4,855 355 4,855 3 4,855	8 0. 8 0. 8 0.	8 0 6 0 8 0		9.00 9.00 9.00	Third Fully Property One Property Net Informed	4,00.56 N/A NOT INFORMED	0.30 0.30 0.30
19 74692 MOMBER 21001 46 19 74690 MOMBERSE 54700 40 10 74698 MERVICE 207708 51	07080 MG San Lowense 19733 MG Governation Valuation 19779 RS Print Alegor	Namola de Oranilaro, 383 de 36, 107 BELN PRCE, JOLIO DE ECULAS REPORTO, 200	Greenied Post Resolver Mass Resolver Mass	52 (23+2+13+5+13+24) 2+ 1.77	6.73 6.885 148 + 0.30 + 1.81 + 0.33 + 1, 6.890 2.87 6,480	8 0 8 0 8 0	0 0 0 2 0 0	(PENDENCION) 2,	5.00 800.30 5.00	Net Informed Perciation Proce of Charge Think Porty Progressy Think Porty Progressy	NOT INCOMED N/A NOT INCOMED	0.30 0.30 0.30
10         100 (2000)         122 (200)         31           141         718/716         REGULEN         236200         453           141         718/016         REGULENCY         22.50/01         313           141         718/016         REGULENCY         22.50/01         313           146         718/016         REGULENCY         22.50/01         313           146         718/016         REGULENCY         22.50/01         313		East in Constrainting of the Alexandria Found to Constrainting of the Alexandria BULS APPENDED ALEXAND HELITO, NO BULS APPENDED AND HELITO, NO	Reachop Mass Reachop Mass Gwordeld Roll appearing Gwordeld Roll appearing	10 41 30	1.008 4,000 1.008 4,000 1.25 4,000 3.09 4,000	8 6 8 6 8 6			0.00 0.00 0.00	One Property One Property One Property One Property	MIT DROMAND The Second Second MIT DROMAND MIT DROMAND	0.00 0.
10 66022 PROPERTY 213802 No. 16 68029 MCP/M29 215828 453 16 87902 820333348 23596 442	ATTI PE Canada Processo MINI MG Jaio De Fora 2007 MA Sec Lain	AV. NCR8A SENSEDEA INO EXCENSION As. Barico de Ere Brennes - 2205 Esta Conselo, aº 230	Reading Mass Reading Mass Reading Mass	(3+5+6) (3+1,60+3) (3+3+3)	1.77 4,800 (2,6+0,67+0,74) 4,800 (0,7+0,7+0,74) 4,800	8 0. 0 0. 0 0.	8 0 8 0 8 0		8.00 9.00 9.00	Third Party Property Third Party Property Third Party Property	NOT DOCKMED KUISNE BURKEII	0.30 0.30 0.30
18 17530 302024,4108 3,732% 384 18 0758 96213458 254096 802 19 0807 30203080 224087 455	2781 CE Portaleo 2792 PE Combs 4228 MC Inches	ANTENEDA LUS, NO ANTENEDA UN DAUDOL, DES Rel. Antelio: Nacorda P. K. Key, 3	Earthop Mass Earthop Mass Earthop Mass	4 (((+1,0)) ((+1,0)+1,0))	NOT INPORMED 6,400 1.49 6,000 0.09+6(0)+0 6.400	8 6 8 6 8 8	0 0 0 0		5.00 5.00 5.00	Third Party Preparty Third Party Preparty Third Party Preparty	6,053.00 NOT (NOCOMBED 801.13	0.30 6.30 0.30
151         27012*         MARE \$1337         2.54096         442           152         50098         34.54052505         5.67296         5.87           150         7542264         BARGU1174         .5645157         .596	2026 M.S. Jan Lais 2020 M. Marsie 00205 BA Press Signs	An Annahor Visionine Preiris, ef 1998 - Baines Annaha Art Andre CAUCH AN BRIAND, 2019 BULK AND PREIRING, 160	Benkrey Man Benkrey Man Benkrey Man Benkrey Man Benkrey Man Grendeld Margenerg Benkrey Man Benkrey Man	(4-149-130) (4-149-130) (3-1-1-1-1) (3-1-2) (4-1-2) (4-1-2) (3	42 4,000 0.75 4,000 3.21 4,000	20 0. 20 0. 20 0.	0 0 0 0		500 500 500 500 500 500 500 500 500 500	Thind Farty Property	NOT DOCOMED 221341 304300	0.30 0.30 0.30
154         51.084         ALCO/0228         51.6217         37.3           156         879726         MCGANUSM         155.260         455           156         65975         BOALNI'HI         -255.6697         365	33937 AL Com Buses 93728 MG Rele Houses 82978 R8 Urgunians	BUA CORONEL LICENA, A/N LICE G. DR CAPULLED Ar. Juin Websin, 403 BR 472, XM Mann	Greenicki Mell suggesting Reachup Mass Greenicki Mell suggesting	40 (1+2,80+2,80) 60	3.13 4,000 (0,71 + 0,32 + 0,87) 4,000 4.32 4,000	8 6. 30 6. 80 6.	0 0 0 0 0 0		0.00 I 0.00 0.00	Tada Bang Nagang Sengar Sengar	N/A NOT INFORMED NOT INFORMED	0.30 0.30 0.30
10 14422 30304.4674 37366 363 18 14752 PEC10033 24420 402 19 14008 DPEC2025 134790 75	32024 CE Postalou 2020 PE Cambo 2020 DF Brooks	Eas Jonales Pongos, 60 RUA PATRO, NOCHERZA, 2128 RUA Standard Zine 12	Reachery Mass Reachery Mass Reachery Mass	7.8 (8+3+6+3) (8+6+2,3)	137 4,000 433 4,000 4,201 (7,732+1,200+1,90) 4,000	8 0. 8 0. 8 0.	0 0 0 0 0 1	(PRINCEING COL 1,	0.00 0.00 280.00	Third Party Preparty Third Party Preparty Third Party Preparty	3,623.82 3(050.00 NUT INFORMED	0.30 0.30 0.31
16 14716 DECORTS 150767 40 16 14720 DECORTS 150007 40 16 1420 DECORTS 15000 50	1914 DF Breaks 0981 DF Breaks 2020 RS Dates Alexer	QUE ETECTINE & CANA DA LA MARCHERTE Quera, CONS E SCANDERALITERO IN AV. GUEVARA, 1927	Reading Cloped Reading Tabata Meson Reading Meso	17 17 4	1.45 6,460 1.451 6,460 2.567 6,460	0 0 0 0 0 0	0 0 0 0 0 0		5.00 5.00	Takai Party Poperty Takai Party Poperty Takai Party Poperty Takai Party Poperty	2,793,77 2,393,71 NOT INFORMED NOT INFORMED	0.80 0.30 0.00
101         202001         202001         204000         40           104         A12001         3PWFNWT         2142007         467           104         S13335         SPWFAWN         2145007         467           105         T14154         SPWFAWN         2155601         464           106         T14154         SPWFAWN         2155601         464	State         Box         Pres         Array           27         28         Pres         Array           28         28         Array         Array           29         28         Array         Array           200         29         Array         Array           201         29         Array         Array           202         29         Array         Array           203         29         Array         Array           204         29         Array         Array           205         29         Array         Array           206         20         Array         Array           207         29         Array         Array           208         29         Conta         Array           208	BUCKERS OF A DESCRIPTION OF A DESCRIPTIO	Converted Mill segmenting Converted Proc Resolution Mass	60 016 (3+3+3)	44 6400 2383 6400 240 6400	N 0 N 0 N 0 N 0	0 0 0 0 0 0 0		5.00 5.00 5.00	Third Party Property Third Party Property Third Party Property	502330 9(12379 6/2343	0.80 0.80 0.30 0.30
107 7341374 30500000 .256600 .445 168 734203 30505018 .2551361 .465 160 734200 30507007 .255011 .466	MUT7 3P San Barnanio Do Campo 2003 3P San Paulo 0220 3P San Paulo	Ran Manushai Badoglee, 21 – Yila Manushaw Ran Canadhaine Annand, 220 - Joganei, 310 - Paule - 357 Ran Hangyu, 641 - Jandan, 380 - Paules - 387	Reading Mass Reading Mass Reading Mass	$\begin{array}{c} 0\\ 0\\ 0\\ 0\\ 0\\ 0\\ 0\\ 0\\ 0\\ 0\\ 0\\ 0\\ 0\\ $	64         6.400           283         6.400           243         6.400           247         6.400           6.717         6.400           3.541         6.400           3.542         6.400           3.543         6.400           3.644         6.400           3.644         6.400           3.644         6.400           3.644         6.400           3.644         6.400           3.644         6.400           3.644         6.400           3.644         6.400           3.645         6.400           3.644         6.400           3.645         6.400           3.645         6.400           3.645         6.400           3.645         6.400           3.645         6.400           3.645         6.400	10 0 10 0 10 0	0 0 0 0 0 0		9.00 9.00 9.00	Third Party Property Not Informed Third Party Property	NOT DEPORMED NOT DEPORMED X316.38	0.00 0.00 0.00
178         7142841         SPRTAUM         233306         463           171         7142808         SPREE8012         2333779         464           172         7142807         SPREE8012         2333588         464	SITIT         SF         Sam Finds           45087         SF         Sam Finds           68085         SF         Sam Finds	BUAGING GERLÁN, CZ. – XXEVARÝ, AÚCI PREZI / SP. – 60090-020 Kos Wolnigens Lais, HZ. – Contex Hindraws - Nie Paulo Diago Julio Porene, SI. – Compto Ellinov, Silo Paulo - SP.	Reachap Mass Reachap Mass Reachap Mass	(4+4+4+4) (3+4+4) (3+3+3)	5404 4,000 2,277 4,000 3,474 4,000	80 0. 80 0. 80 0.	0 0 0 0		9.00 9.00 9.00	Third Party Preperty Third Party Preperty Third Party Preperty	NOT DECOMBLD 4,325.08 1,627.42	0.80 0.30 0.30
171 742672 SPEMMENT 214140 444 171 74275 SPERSOT2 215542 444 173 74397 SPENSOT2 216540 444	60200 34P San Paulo 00200 34P San Paulo 00842 34P San Paulo	Atomola des Toponquins, 1233 - Plendos Paulara Ras Fuelopo Constello, 1390 - Plendos - NP Esta industria, 1390 - Esta industria, 733	Reachen Mana Reachen Mana Reachen Mana	(3+3+3) (3+3+3) (3+3+3)	2.5 6,0% 2.5 6,0% 4.26 6,0%	8 0. 8 6. 8 0.	0 0 0 0		9.00 9.00 9.00	Third Party Property Third Party Property Third Party Property	XIMIT NOT DATORNED 200300 NOT DATORNED	0.30 0.30 0.30
178 73428 97554899 233736 454 177 734293 978128869 2337444 444 178 82163 9742950 23869 414	40300 SP Sar Pade 43273 SP Sar Pade 44074 PA Spream Do Para	An Brigadien Lab Areiten 2017 - Darien Paden Ras Topian, GD - Vin Marine Paden Ras Support Simplein, 321	Rescher Mass Rescher Mass Gweekel Kell opposing	(1+3) (1+1+1+3) 80.29	1,209 6,400 2,423 6,400 36,73 6,400	8 0 8 0 8 0	0 0 0 0 0 1	PRINDENCI CIT 1,	9.00 9.00 2353.00	Third Party Property Third Party Property Not Informed	NOT DEVEKIED 13,2014 NOT DEVEKIED	0.30 0.30 0.30
175 12128 FAR2010 3.41111 .443 180 744021 FAR2015 .44722 .444 180 66428 PRC20020 .254328 .473	4483 PA Below 4803 PA Below MITA PE Carible	Ean Santo Antoine Arc. Tarranto Batano, Ali A. Mananthaia Real: Br 277 Cas : Orlanzo / Real: do Cafe : Calulor Industrial Dir Carellas	Georgical Pros Georgical Pros Reschop Mass	44.79 37.94 (3+3)	5.55 6,400 6.92 6,400 (5,64+1,92) 6,400	8 6. 8 6. 8 6.	0 0 0 0 0 0		8.00 9.00	Third Pariy Despery Na Islammi Third Pariy Deproty One Perpety Third Pariy Perpety Third Pariy Deproty Third Pariy Deproty Third Pariy Deproty Third Pariy Deproty Third Pariy Deproty	2,194.50 N/A NOT INFORMED	6.80 6.80 6.30
10 74232 PRC10022 234307 402 16 74233 PRC10023 234979 402 10 53508 97424317 225425 472		Alamota Die Massey (20) Ran Die Massey Liebarre de Lacerda, NI - Optie Ran Ran Die John Correlto Francesso vie	Reachen Mass Reachen Mass Gweekleit Mell sammeter				0 0 0 0		8.00 9.00 9.00	Table Party Property Table Party Property Table Party Property	433446 433446 578556 578556 578546 NOT DECEMBED	0.30 0.30 0.30
16         53359         3P820300         2354203         443           16         734327         3P8050731         235390         463           18         734323         SP8050733         2353300         464	75435 SF Net Fash 0000 SF Net Fash 0000 SF Net Fash	Bas Cartantysis, 200 Japané Kon de Carendonis, 2143 Bas Bele Caren, 60	Reachop Mass	(3+3+3) (3+3+3) (3+3+3)	(04+6,2*+6,2*) 6,480 (121+1.31+1.16) 6,480 (029+1.12+6.76) 6,480	30 0. 30 0. 30 0.	B		5.00 5.00 5.00	Think Party Property Think Party Property Think Party Property	NOT INCOMED NOT INFORMED TENAT	0.00 0.00 0.00
10         T4130         STRUCHUZI         23.6470         646           19         74.8130         SIR513414         23.1980         463           19         74.8165         SIR513474         23.0962         463           19         74.8166         SIR513874         23.09620         463	0000 37 Sur Pada 8033 37 Jonán 3340 37 Janua 199 Janua	Are, Amplies, N. 1109 - Convolution For the Residen Raw Argunite Pauline, 57	Rearing Mass Rearing Mass Rearing Mass	(1+3+3) (1+2+4+4+4+4+4) (1+3+3+3)	(1,58+6,79+6,38) 4,880 (78+1,91+2,31+6,78+6,80+2 4,880 (5,58+1,61+6,71+6,76) 4,880	0	9 0 0		5.00 5.00 5.00	salal Parij Poperij Thail Parij Poperij Thail Parij Poperij Thail Parij Poperij	N78540 X/32274 420600	030 030 030 030
	12271 SF Non Pado 12271 SF Non Pado 17275 SF Non Pado 10000 SF Y Y	NOVE CONTRACTOR OF A CONTRACT	Keening Mas Keeling Mas Keeling Mas Keeling ***	(4+4+3) (3+4+4) (3+4+4) (3+4+4)	NOT INFORMED 4,000 NOT INFORMED 4,000 NOT INFORMED 4,000 NOT INFORMED 4,000	0. 80 0. 80 0.			3.00 3.00 3.00	Think Party Property Think Party Property Think Party Property	NOT INFORMED ATMIN ATMIN	480 0.00 0.00 0.00
1% 7427% 3/9877% 233427 444 1% 7427% 3/9877% 233368 443 1% 74467% 8/973674 223364 433 1% 74467 8/973667 *******	6m Pado 4533 3P San Pado 12942 8J Principals 3520 5P 5	John June Marken and San	Reachery Mass Reachery Mass Generalish Proc Reachery Mass	4 14 (3+3+3)	KOT INFORMED     4,000     6,79     4,000     2,02	0 0 0 0 0 0 0 0 -	0 0 0 0 0 0		0.00 0.00 0.00	Thind Party Property Net Informed Thind Party Property	600374 NOT INFORMED NOT INFORMED	0.00 0.00 0.00 0.00
199 701279 19970009 200716 440 200 704798 19700077 20000 440 200 704799 19700079 200002 440	2283 3P Keen 2703 3P Keen 2007 3P Keen	Kan Barin de Normepansha, 77 Esa Dadaire de Cardon 191 Done des Andrades (11/20	Earling Mass Earling Mass Earling Mass	(1+1+1+1) (1+1+1) (1+3)	3.32 6,400 2.28 6,400 3.36 4.400	0 0 0 0 0 0			3.00 5.00 5.00	Thind Farty Preparty Thind Farty Preparty Thind Farty Preparty	8(337.42 X333.50 4.500.22	638 638 538
202         7147956         3P9T35886         2356493         443           201         714798         3P9T35888         2356132         463           201         714798         3P9T35888         2356132         463           201         7142194         3P9079533         2356497         464	51990 37 News 5001 37 News 61756 37 New Pails	Em Die Lohne Vuon, 68 Em Voorwele de Funi, 280 Em José Mann Lohne, 1107	Reachop Mass Reachop Mass Reachop Mass	(3+3+3) (3+3+3) (3+4)					0.00 0.00 0.00	salel Party Property Thiel Party Property Thiel Party Property Thiel Party Property		0.00 0.00 0.00
	MID         SP         Sam Paulo           AMD         3P         Sam Paulo           AMD         3P         Sam Paulo           AMD         3P         Sam Paulo           AMD         AMD         Parmer Alagor           AMD14         BO         Parmer Alagor           AMD14         BO         Outers Parmer           AMD15         AMG         Outer Parmer           AMD16         MGC         Jaur Dar Parmer           AMD17         AP         Sam Paulo	nan Pomplon, 823 - Justine Pauline Ban Duan, 949 - Paulan ACISSO D NG 51 UNIDARIS VICENAL, 5146 ACISSO D NG 51 UNIDARIS VICENAL, 5146	Keening Mas Keeling Mas Keeling Mas Keeling	(8+6) (3+3+3) 4 (9 *****)	1.00 4,000 1.00 4,000 NOT INFORMED 6,000 0.04	0. 8 0. 8 0. 8	. 0 0 0	(PENDENC OI) 2	5.00 5.00 5.00	Think Party Property Think Party Property Think Party Property	801 801-8680 8,11786 2,0188 6,0188	480 0.00 0.00 0.00
200 66658 MCOPPOR 20,7527 453 201 66658 MCOPPOR 20,7527 453 201 67567 MCPM02 21,7549 453 211 74202 57920507 744294		AVENUES DO CREATE NO. (10) BUA DO CREATERN (10) AVENUESA DO REO REANCES, 1443 Buodo Creationna 110 - Indio - Viscont	Rening Mas Rening Mas Rening Mas Rening Mas	(0+4) (0+4) (0+5+3)	4,000 3,492 4,000 2,26 4,000 (3,73 + 6,03 + 6,00 ) * ****	0 20 0 20 0 20 0 20 -	2 0 1 0 -	PRINDING OF 1	0.00 230.30 0.00	Not Informed Think Party Property Think Party Property	NOT INFORMED MALM NOT DECOMIN	0.00 0.00 0.00 0.00
212         74200         27827000         210000         400           213         74260         27827000         210000         400           213         74260         27826000         210000         400           214         74260         27826000         210000         400           214         742700         878040000         210000         400	7000 37 Not Fails 10070 37 Not Fails 10070 37 Not Fails	Eas Caising to Tamina Million Eas Caising of Paris Bas Caising of 1977 Arc Lange de Paris Machada, 1974 (1996) - 9	Review Mass Review Mass Review Mass	(1+4+4) (1+3+3) (3+3+3) (3+4+6)	(0.28 + 0.42 + 0.28 ) 4.400 (1.28 + 0.42 + 0.28 ) 4.400 (1.28 + 1.23 + 0.92 ) 4.400 (1.44 + 1.79 + 0.48 )	81	0 8 9 8 9		9.00 9.00 9.00	Think Purey Property Think Purey Property Think Purey Property	1,790,00 T,006,00 30,000,00	030 030 030 030
215         7142746         XPERCINS         233/7701         444           216         7142748         XPERCINS         233/8276         444           217         7142748         XPERCINS         233/8276         444           217         7142748         XPERCINS         233/8276         444	43212 3P New Paulo 60708 3P New Paulo 60703 3P New Paulo	Ras Yanowelo Donoodi, 196, Ipinogo Ras Certo Aguin, Wei Apinogo Ar. Italianipolo, 2183	Reading Mass Reading Mass Reading Mass	(1+1+1) X (1+1+1)	(5,0) + 6,73 + 6,73 4,400 2.7 4,400 (3,03 + 6,03 + 1,73) 4,400	80 00 80 00 80 00			9.00 9.00 9.00	Thind Fury Property Thind Fury Property Thind Fury Property	NOT INFORMED 805311 12,14335	030 030 030
218         T44449         MPARCH153         21,21679         564           219         T44840         MPARCH153         23,5676         443           229         T44840         MPARCH190         23,5676         443           229         T44840         MPARCH190         23,54803         443	42037 34 Amendus 33073 34 Series 32463 34 Series	Eau Wonersku Bruz (20 - 16 Coparabana - Via hara Masa Asamila Perakara Wilcon, 66 - 3nai Manine Eau Benjamin Comtana, 1033	Reachop Mass Reachop Mass Reachop Mass	(1,20+4,23+3,20) (3+4+4) (3+4+4+6+6) 2,76	(1,05 + 1,05 + 0,03) 4,000 (0,75 + 0,75 + 1,72) 4,000 +1,71 + 2,64 + 0,71 + 2,00 4,000	0 0. 0 0. 0 0.	0 0 0 0		0.00 0.00 0.00	said Party Property Think Party Property Think Party Property	NOT INCOMED NOT INCOMED ADDITION	0.00 0.00 0.00
Jan         T44626         JPEMI2006         Z16090*         460           ZZ2         T44626         37%         23.6034         460           ZZ3         84028         PTD/Q1072         23.6034         462           ZY3         84028         PTD/Q1072         43.0733         422	10000 37 San Fash 10001 37 San Fash 20017 F1 Tangar Do Fasa 10000 81 San San San San	Alamada des Auspans, 1207 - Misana Alamada Tesi, 305 – Joshim Paulas AY, Panspel, da	Reachop Mass Reachop Mass Greenhal Ball supporting Classifield W.W.	(1+1+1) (1+1+1) 40	$(\gamma_1 d^2 + 0_1 (d + 1) (2))$ 4,000 $(\beta_1 d^2 + 0_1 d^2 + 0_1 (d - 0_1 (d + 0)))))))))))))))))))))))))))))$	M 05 30 05 30 05	0 0 0 0		1.00 3.00 3.00	salid Party Poperty Third Party Poperty Not Informed World Party Poperty	1304746 1003.46 NOT INFORMED	0.00 0.00 0.00
20 74410 AMERICA 30 228 74410 AMERICA 30 228 74410 AMERICA 30 228 74410 AMERICA 30 30	res Chandra 2787 E.5 Innuesure 7886 E.6 Press	REA DO CACAPAN BEA IN DE DENOTEMBRIAAN PRACATONICIM (UNE REPERCAN)	Coronical Xell supporting Coronical Xell supporting Coronical Kell supporting Coronical Vell supporting	230	NOT IMPORADO 6.411	- 0. 0 0. 0 0.	0 1 0	presente coj 2. presente coj 2.	500.00 5.00	Not followed Own Property Not followed	NOT DEVALUED NOT DEVALUED N/A	480 0.00 0.00 0.00
20 742362 409(0177) 362394 39 20 742362 80423312 3694398 464 22 714762 864230862 333991 360 23 87966 82504823 7477	arm no Proto Supara ACIVI EA Standar SERIE ES Canda MULL MA Sur Frair	NAC SINCENTEMES NAC SINCENTEMES BUA [OIL] PROPERTY AND	Correlation Additional Additiona	30 40 ()+5+6) ()+2+9	2.08 4,000 2.09 4,000 2.09 4,000 NOT INFORMED	0. 80 0. 80 0.	1 0 0 0 0	,anuul 2	3.00 3.00 3.00	Non-Indexed Third Party Property Third Party Property	NOT DEVAILED NOT DEVAILED NOT DEVAILED NOT DEVAILED	
201 714908 PAR(10274 440 202 714909 PAR(10274 440 202 714909 PAR(20215 12799 440 202 714909 PAR(20215 12799 440	Chi         And         Journame           101         A.         Journame           102         A.         Journame           102         A.         Journame           103         A.         Journame           104         A.         Journame           105         A.         Journame           106         A.         Journame           107         A.         Journame           108         A.         Journame           109         A.         Journame           101         A.         Journame           102         A.         Journame           103         A.         Journame           104         A.         Journame           105         A.         Journame           106         B.         Journame           107         A.         Journame           1072         R.         A.           108         R.         Journame           109         R.         A.           1001         R.         A.           1002         R.         A.           1003         A.         A.      <	ROEOVIA METRI IL ROBANDEDI, NOL - KM 10 AVAILLA PERMANDEDI, NOL - KM 10 AVAILLA PERMANDEDI (10) TRAV. LOMAS VALENTINAS, 40	Georgical Mattageneting Georgical Mattageneting Georgical Mattageneting		NOT INFORMED 4,810 NOT INFORMED 4,810 NOT INFORMED 4,810	0 0 0 0 0 0 0 0 0			0.00 0.00 0.00	Ova Popery Ova Popery Ova Popery	N/A N/A N/A	0.8 0.8 0.8
236 714806 PAREAD16 344039 464 236 714768 PAREAD37 32360 460 236 714768 PAREAD37 32360 460 236 71976 PREAD500 739760 340	45411 PA Below 2032 PA Senis Balans Do Pau 8647 PB Calculo	TRAVERSA CULCO, SEP BODOVIA ALCOSTO MURA PERIO, A/N. CENTRO AVENEDA PLAVED REB BO COUTINHO, N'14, MANARA	Gaussield Bell appearing Gaussield Bell appearing Reachery Mass	80 60 6	NOT INFORMED 4,000 NOT INFORMED 4,000 NOT INFORMED 4,000	8 6 8 6 8 6			9.00 9.00 9.00	Niti Informed Own Perperty Third Party Perperty	NOT DOCOMBID N/A 201671	0.30 0.30 0.30
27         37367         79530033         51260         543           28         27912         79541000         5001         50           28         74594         79610071         5001         50           29         74594         79610071         51301         50	AZIRO PR Joan Frans ATRA PR Joan Frans ATRA PR Joan Frans	RUA (100 RAMALIO LETTE, 1233 AVEORDA AMPL. 100 AVEORDA RABING, 43	Review Mass Review Mass Review Mass	3 12 4	NOT DEPENDED 6.000 NOT DEPENDED 6.000 NOT DEPENDED 6.000	8	0 0 0 0		5.00 5.00 5.00	Third Farty Desperty Third Farty Desperty Third Farty Desperty	4179/26 2006-0 332797	0.30 0.31 0.30
28         FUND         PUNCTON         X2064         343           28         T44674         PUNCTON         X2064         343           26         T44674         PUNCTON         X1264         343           26         T44674         PUNCTON         X1264         343           26         T44674         PUNCTON         X1270         341	8778 Pl Levie 8722 Pl Levie 9687 Pl Levie	BUA (OBL ADDREAL CHAVES, 264 Box Viggen Avenila Box Viggen, 4000 Bill Vissa. BUA ANDREA BUDDREATS CORLED, 423	Reachop Mass Reachop Mass Reachop Mass	(1+0+0) (0+0+0) (1+0+0)	NOT INFORMED 4,8% NOT INFORMED 4,8% NOT INFORMED 4,8%	0 0.00 0 0.00 0 0.00	0 0 0 0		9.00 9.00 9.00	sana Pariy Property Thind Pariy Property Thind Pariy Property	2,05.25 4,75.18 3,375.42	0.30 0.30 0.30
av         T44644         PERTING         A0087         343           244         T44713         PECREMIC A0087         343         347           246         T44713         PECREMIC A0080         343         343           246         T44722         PERANCI         340400         343           246         T44721         PERANCI         340400         343           246         T44721         PERANCI         344         345	80.05 FL Reiń 90211 FL Reiń 80200 FL Reiń	REA STITUDE SETUDIES, 200 AV GEV. AGAMENON MARS ALBERTS AND DEPENDENCE REA MARQUES DE HERVALUE AVAINTO ANDRO A RECEPT. IN: Rea Neural VIV.	Acceleg Mass Review Mass Review Mass Gravital V	4 () () () () () () () () () ()	AND DECEMBED 4,000 NOT DEFORMED 4,000 NOT DEFORMED 4,000 NOT DEFORMED 4,000	A 0. 20 0. 20.	0 0 0 0 0 0		9.00 9.00 9.00	sanai Pariy Property Thiai Pariy Property Thiai Pariy Property Net Inform <sup>-1</sup>	20130 NUT INFORMED ADATE NUT INFORMET	0.00 0.00 0.00 0.00
	ALL PI Constan MAG PI Sumayors MAG EI Er Dr. Jacobs MAG EI Er Dr. Jacobs	nar regisala 3/N Bas Joupin Provin Lad 1080 / Convo BUA (CORUNELT-MARING) 1986 Transmission Davies. <sup>370</sup>	Georgial Xdl segrating Georgial Rel segrating Georgial Post Georgial Y-W	40 42 30	8.14 4,000 2.28 4,000 8.17 4,000 8.50 <sup>-1</sup>		0 0 0 0 0		8.00 8.00 8.00	. on memory Net Informed Net Informed Third Party Persenty	NOT INFORMED NOT INFORMED NOT INFORMED NOT INFORMED	0.00 0.00 0.00 0.00
20 1000 E23000 20100 C0 20 70700 SCOUL 20100 C0 20 70700 SCOUL 20100 C0 20 70700 SCOUL	14228 El Ko De Jacon 14228 N. Investo De Nel 14229 SF Jacon De Nel	Eas Massie Die Jacomessie Lose I Eas Gebach v. n. Eis die Los Eas Monstein Langeiren 227	Gweedidd Pros Gweedidd Xell opposing Gweedidd Xell opposing	N N N	4,4% 3.54 4,4% 2.52	8	0 0 0 0 0 0		9.00 9.00 9.00	Think Purey Property Think Purey Property Think Purey Property	NOT DPORMED NOT DPORMED ACOUT	030 030 030 030
200 T1420AT SPELTING 23A422 44A 201 T14275 SPECOT14 23A872 44A 200 T14276 SPECOT14 23A872 44A	State         J         Ke fb here           201         K         Mark           201         K         Mark           202         K         Mark           203         K         Mark           204         K         Mark           205         K         Mark           206         K         Mark           207         K         Mark           208         K         Mark           209         K         Mark	RUAGEOCUSAL Avenide Augúlas, 2001 - Consolução - São Paulo Bas Jongaio Antoneo, 147 - Polaciono - São Paulo	Gaussield Kell segmenting Reachery Mass Reachery Mass	$\begin{array}{c} 33\\ 33\\ 34\\ 34\\ 34\\ 34\\ 34\\ 34\\ 34\\ 34\\$	NOT INFORMED 4,800 2.4 4,800 NOT INFORMED 4.800	8 6 8 6 8 6			8.00 8.00 8.00	Thind Fury Property Thind Fury Property Thind Fury Property	4133.M 15541579 3,256.38	0.30 0.30 0.30
256 714260 5078754205 233.0081 444 257 714260 50782.32568 2349981 444 258 80102 83492568 1234998 375	6032 3P Nor Pado 6036 3P Nor Pado 9877 RA Mais Dr So Joor	Ras Consiliaire Rolegan Mers, 971 - Vin Marian Ras Antinis Elizier de Moras, 201 - Vin Carlone, Nin Paulo - 197 Enderen EA199	Rankap Mas Rankap Mas Rankap Mas	(3+3+3) (3+3+4) (3+3+9)	NOT INFORMED         6,810           β(778 + 1,153 + 6,778)         6,880           β(779+6,87+6,77)         6,880	80 00 80 00 80 00			8.00 9.00 9.00	Thind Party Property Thind Party Property Thind Party Property	30,244.30 8073.47 NOT INFORMED	0.00 0.00 0.00
280         7142077         RAR/MI146         15.35296         3507           280         7143934         PAARA2211         -1.35777         -463           281         7144002         PAARA2211         -1.35777         -464	19605 RA Ba De Consulanda 2000 PA Assession 2001 PA Below	Rande Comminutes, s/m Ran Colema Lens, s/m RUA COMBICA, NYM (2012, AV) (ELEO CEZAR)	Reachop Mass Generalish Mill sepposing Generalish Mill sepposing	(1,3+3,3+3,3+3,3) 30 30	3.30 4,880 6.23 6,880 6 4,880	80 00 80 00 80 00	0 0 1 0	pressions on 1,	0.00 23030 0.00	Third Fariy Property Own Property Own Property	N/M/A N/A N/A	0.00 0.00 0.00
362         71470/6         PARENDARD         207806         -172           263         M052         PERTON         AN7700         500           264         744900         PERDON17         -534407         544	NAI P5 Personano NAI P5 Index 2000 P8 Jacober	Averside Mar Kalenschelt, s. v.e. Onaden 20 Loop 40. Res Pro Versee Aufrader, 201 Bas Marco Fishers Schwitz, 66.	Corroled Kell appealing Corroled Kell appealing Reaction Meet	40 30 4	2.67 6.000 5.47 6.000 1.51 6.000	8 0 8 0 8 0	0 0 0 0 0 0		5.00 5.00 5.00	Think Party Property Think Party Property Think Party Property	2120/32 1.09520 3.02500	030 030 030
265         74.4717         PERCESSIG         AUTIMA         MAT           266         71.4779         PERCESSIG         AUSUS         MA           267         71.4799         PERCESSIG         AUSUS         MA           267         71.4799         PERCESSIG         AUSUS         MA	10737 PE Jalonates Das Couranges 1081 PE Resúr 1020 PE Resúr	Averalis Reveales Veira de Maio, 2000 Dorode das Unice 634 BUA QUARENTA E OPTO NºDA - 1209 - NEODARE B	Reachop Mass Reachop Mass Reachop Mass	(0+8+8+8+8+8) 3,07 (11+11+11) 7	+1,07+1,07+1,13+1,13+1,13 4,400 (0,00+0,00+0,00) 4,400 2.7 4,400	0 0. 0 0. 0 0.	B 0 0 0 0 0		9.00 9.00 9.00	salal Fasy Poperty Third Fasy Poperty Third Fasy Poperty Third Fasy Poperty	NOT INCOMMED 320578 447743	0.00 0.00 0.00
ass         74700         PERIADE         36000         343           20         87905         PERIADE         33767         343           20         87975         PERIADE         33767         343           20         87975         ENTEADO         37980         343           20         87977         ENTEADO         37980         343	Anara PE Resile 83488 PE Clinds 2024 RN Nead	BUA CAP (CHE DA LUZZ) Avenila José Augusto Moreira 2200 - Edé Calade de Clinda WEINEDA PER ZEROTE DE MORADA, 197 Avenila 21	Reachop Mass Reachop Mass Reachop Tabala Masso Reachop Tabala	5 (2+2+2) 17	1.43 4,480 8(40+1,2+1,2) 4,480 NOT INFORMED 4,480 0 MoDULED	4 0. 30 0. 50 0.	0 0 0 0		1.00 0.00 0.00	Data Party Property Data Party Property One Property Theory Reserve	3/24/23 2/93.56 N/3 comm	0.00 0.00 0.00
212 T44197 PERMITS AUAU 347 272 T44196 RPRC01719 20.4400 444 273 T44256 RPR200719 20.4400 444 273 T44257 RPR2008 444	10.05 PL Resile 6000 3P San Pado 6000 3P San Pado 6000 3P San Pado	Acousta 17 de Agrono 800 Res Annará Garga, 155 - Consolação Res Consort Xervier de Tránde, 70: República de Dara Consort Xervier de Tránde, 70:	Assilop Mas Easilop Mas Easilop Mas Realise 20	(12+12+12) (3+3+3) (3+6+4) (3+6+4)	(x,1+6)2+6(2) 4,000 ((25+6)27+6(21) 4,000 ((22)+6(22+6)2) 4,000 ((22)+6(22+6)2) 4,000	M 0. 30 0. 30 0.	0 0 0 0 0 0		9.00 9.00 9.00	sanai Pariy Property Thini Pariy Property Thini Pariy Property Thini Pariy Property	4,000,00 4,756,38 4,376,34 7,454,47	0.30 0.30 0.30
	2004 3P Am Pado 2003 3P Am Pado 2003 3P Am Pado 2004 3P Am Pado	An Peren Main, 312 Bas Jensies Annale, 327, Joan Hits Rea Cores Arain, 2404, Jenseeve Real	Asseling Mass Exclose Mass Exclose Mass Factors	(8+8+8+8) (3+8+8) (8+8+8)	10,44=1,32=0,032 4,480 2,53 4,480 (5,75=6,42=6,78) 4,480 (1,12=6,42=6,78) 4,480	M 00 00 00 00 00 00	0 0 0		5.00 5.00	anna Fariy Peoperty Thind Fariy Peoperty Thind Fariy Peoperty Thind Fariy Peoperty	7,014.02 9,3393.07 7,994.00 8,1154.00	638 638 639
271 734208 20000727 233623 444 273 734270 2000731 233075 444	norm SP San Pado Natification SP Character	Eur Compress, of 313/321 - Feshion Eur Main: Elicens, 506 Eur Isoi, Indian de Change 200	Accelegy Mass Receivery Mass Receivery Mass	(3+3+3) (3+3+3) (3+4+4)	(1,1+1,(2+1,0)) 4,000 (0,08+0,74+0,75) 4,000 (1,1+1,(2+1,0) 4,000 Normanation	M 0. 00 0. 00 0.			9.00 9.00 9.00	Think Party Property Think Party Property Think Party Property	3,13406 19,40834 3,089441	0.00 0.00 0.00 0.00
279	2086 SF Courses 6770 SF Build Courses	Eng Advance of Association										
277 3260% X758777 234057 44 279 3267 X758771 23407 44 279 3267 X758771 23477 44 279 3267 327 32778 44 279 3267 32778 32778 44 270 32778 32778 32778 44 281 3268 3070 22778 34007 44 281 3268 3070 2277 44 281 3268 3070 22778 44 291 32778 44 291 37	2046 3P Comp     4779 3P Prois Gaude     4779 3P San Pado     47214 3P San Pado     47214 3P San Pado     47214 3P San Pado	Eas Alores Cheres, (1) An Frains, 2000. Bob Yuan Aronida Breal, N.N. (andar Amirice Progr Norma Norkova do Breal) Ba Versag 1, Vik Indonesalizacia	Reachery Mass Reachery Mass Reachery Mass Reachery Mass	(3+3) (3+3+3) (3+3+3)	(1,68+1,52) 4,895 2,43 4,895 (0,54+0,63+0,55) + ###	80 05 80 05 80 re	0 0 0 0		9.00 9.00 9.00	Think Putty Property Think Putty Property Think Putty Property Think Putty Property	6,000 17,018,17 NOT INFORMED NOT INFORMED	0.30 0.30 0.30
25         3400,         275071.         21000.         44           27         32574.         21000.         44           27         3420.         077271.         21000.         44           27         5130.         077271.         21000.         44           28         5130.         077271.         21000.         44           29         5130.         077271.         21000.         44           20         71300.         077271.         21000.         44           20         71300.         0797292.         21000.         44           20         71400.         0797292.         21000.         44           20         71400.         0797292.         21000.         44           20         71400.         0797292.         21000.         44           20         7140.         0797292.         21000.         44           20         7140.         0797292.         21000.         44           20         7142.         07904701.         21000.         44           20         7147.         07904701.         21000.         44           20         7147.         07904701.         <	DB45         DF         Champe           2079         3P         Print Gaudie           2079         3P         Print Gaudie           2011         3P         San Pado           2024         3P         Nan Pado           2026         3P         Nan Pado           2027         San Pado         San Pado           2028         3P         Nan Pado           2029         San Pado         San Pado           2020         3P         Nan Pado           2020         3P         Nan Pado           2020         3P         Nan Pado           2020         3P         Nan Pado	Res Manue Chenne, 63 A. Frankar, 2006. Bible Wates Arounda Bendi, [35]. Stadiar Anniero Paper, Norma Indonese de Bendij Ben Wang, f. Stadiar Anniero Paper, 2017. Statistica Ben Paper, de Cambriel Danse Francesa de Paulo, 127. Consen- Ben Paper, de Cambriel Danse Francesa de Paulo, 127. Consen- Ben Consent, 30 Jackart Kurnerad	Revelup Mass Revelup Mass Revelup Mass Revelup Mass Revelup Mass Revelup Mass Consolidit Proc	(3+3) (3+3+3) (3+3+3) (3+3+3) (3+3+3) (3+3+3) 40	(1,00+1,12) 4,000 2.43 4,000 (0,00+0,03+0,075) 4,000 (0,00+0,00+1,22) 4,000 (0,00+0,00+1,22) 4,000 (0,00+0,00+1,23) 4,000	8 6 80 6 90 6 91 6 93 6 90 6 90 6	0 0 0 0 0 0 0 0 0 0 0 0 0 0		1.00 1.00 1.00 1.00 1.00 1.00 1.00	Inizi Party Perperty Thini Party Perperty Thini Party Perperty Thini Party Perperty Thini Party Perperty Thini Party Perperty	4,0000 IT,38,57 NOT INFORMED NOT INFORMED IT,221.15 NOT ENFORMED NOT ENFORMED	0.00 0.00 0.00 0.00 0.00 0.00 0.00
1         1	3         3			1         1         урадинаниј			0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0					I         I           I         I

Conhevelon # Krunde Star ID Latinde (decimal) Laurinde (decimal)	Pakina Address Pakina	No new Bar Per Bark Personnesia Militard Sensoria Dara Bard Containa Personnes Tear Biologia Militard Market Land Rac Per 2011 (2011)	Res Maring J General Residence International Control Statistics of
28 744421 SPELARD 21000 463805 212 714196 SPECARD 238056 4743705 213 714196 SPECARD 238056 4743705	Str         Saulus         East Quarters: Forwardser, S10         1           SP         Scienceda         East Capitale Neurainees Files, 171         1           SP         Scienceda         Non-tender on 172         1	Keylog         Mark         [0-1+1+7]         NOT INFORMED         (4010)         0.01           Kerdog         Mark         [0-1+1+2]         (7.1=1)(0-0,7+7.2)         (40100)         0.00           Starting         Mark         [0-1+1+2]         (7.1=1)(0-0,7+7.2)         (40100)         0.00	0         .         0.00         Data/Persy Negrey         U/117         0.01           0         .         .00         Data/Persy Negrey         NOT DOCKMED         .00           0         .         .00         Data/Persy Negrey         NOT DOCKMED         .00           0         .         .00         Data/Persy Negrey         NOT DOCKMED         .00
Diff         Control         Control <thcontrol< th=""> <thcontrol< th=""> <thcont< td=""><td>Diff         Instance         <th< td=""><td>Josephile         Josephile         <thjosephile< th=""> <thjosephile< th=""> <thj< td=""><td>Open Number of Party Prepared         NOT SOCIALD         Open Number of Party Party</td></thj<></thjosephile<></thjosephile<></td></th<></td></thcont<></thcontrol<></thcontrol<>	Diff         Instance         Instance <th< td=""><td>Josephile         Josephile         <thjosephile< th=""> <thjosephile< th=""> <thj< td=""><td>Open Number of Party Prepared         NOT SOCIALD         Open Number of Party Party</td></thj<></thjosephile<></thjosephile<></td></th<>	Josephile         Josephile <thjosephile< th=""> <thjosephile< th=""> <thj< td=""><td>Open Number of Party Prepared         NOT SOCIALD         Open Number of Party Party</td></thj<></thjosephile<></thjosephile<>	Open Number of Party Prepared         NOT SOCIALD         Open Number of Party
287 87948 PG.1883 2515688 311662 298 69948 PELDAG1 2515688 311662 298 69948 PELDAG14 233129 317990	re Comming Market of Comming and Sector Commission of Sector Sect	Rening Mass 1 2.21 Control 10 Rening Mass (321+32) (J,B = 1/3) 4,8000 0.03 Rening Mass 4 1.32 4,8000 0.00	0
279 60/024 PECTNC26 23.42079 -012801 388 60/079 PECTNC2 25.42772 -0128079 386 60/079 PECTNC97 -25.42782 -012973	a         Sector           b         Sector           c         Sector           d         Sector <td>Rackop         Mass         h         Z.50         4,400.00         0.00           Reckop         Mass         (3+1+1+2)         347         4,400.00         0.00           Reckop         Mass         (3+1+1+2)         347         4,400.00         0.00           Reckop         Mass         (3+1+1+2)         2.4         4,000.00         0.00</td> <td>0         .         0.00         samt ray requiry         2(07:31         0.00           0         .         0.00         Third lawy Peperty         NOT ENCOMID         0.00           0         .         0.00         One Peperty         N(Å)         0.00</td>	Rackop         Mass         h         Z.50         4,400.00         0.00           Reckop         Mass         (3+1+1+2)         347         4,400.00         0.00           Reckop         Mass         (3+1+1+2)         347         4,400.00         0.00           Reckop         Mass         (3+1+1+2)         2.4         4,000.00         0.00	0         .         0.00         samt ray requiry         2(07:31         0.00           0         .         0.00         Third lawy Peperty         NOT ENCOMID         0.00           0         .         0.00         One Peperty         N(Å)         0.00
NG 66009 PECTA023 23.4347 -42.2034 NG 744033 E0224/HEZ 23.06098 -43.2034 NG 744732 PERCENT 23.06098 -43.2034	PR Comba Manda Di Cales de Cardos (e 1997) 1 185 Conta Do Sal Avida de Cardos de Cardos (1917) 1 186 Conta Do Sal Avida (1918) 1 187 Ponte Conca Bas Vardagos, Sol 1	Review         Mass         3         2.32         4,0000         0.00           Review         Mass         6         1.67         4,0000         0.00           Review         Mass         6         1.67         4,0000         0.00           Review         Mass         6         1.67         4,0000         0.00	0         .         0.00         Thaid Pay Pagery         C1117         0.01           0         .         0.00         Thaid Pay Pagery         C1127         0.03           0         .         0.00         Thaid Pay Pagery         C1121         0.03
300         N°014         PECINICA         24.2017         -0.23.033           301         271842         ExCLAMM         2718466         -0.117842           307         372114         SPECINENT         2718463         -0.017842           307         372114         SPECINENT         2718463         -0.017842	PE         Comin         Res Manufal Dandors, 90         F           RX         Costs Do 3d         Bas Can. Area da Evola Noñeses, 801         F           SP         Ann Paulo         E. Mains Andréa Noñeses, 304         F           SP         Non Paulo         E. Mains Andréa Noñeses, 304         F	Review         Mast         (3+2)         3         4,400.0         0.0           Bandom         Mast         4         1.2"         4,400.0         0.0           Paredid         Midi Appendiq         0.33         3.34         4,400.0         0.00	0         .0.0         Task Prop Pagery         NOT EVERABLD         0.00           0         .0.00         Task Pagery         NOT EVERABLD         0.00           0         .0.00         Task Pagery         NOT EVERABLD         0.00
No.         JCT052         SEC00055         CO10056         M11962           No.         JCD1014         JCT0200077         JC0007         450043           No.         JCD124         JCT0200777         JC0029         450043           No.         JCD124         JCT0200777         JC0029         450043           No.         JCD124         JCD12017         JL0029         450043           No.         JCD124         JCD12017         JL0029         450043           No.         JCD124         JCD12017         JL0029         450043           No.         JCD12017         JL0029         JL0029         450043           No.         JCD12017         JL0029         JL0029         JL0029           No.         JCD12017         JL0029         JL0029         JL0029           No.         JCD12017         JL0029         JL0029         JL0029           No.         JCD12017         JL0029         JL0029         JL0029         JL0029           No.         JCD12017         JL0029	SP New Pauls K. Manuel Martine de Mala, N.N. Vila Instea. G SP New Pauls Avenuels Town Data Lap., S/N. Asian Paulan G DR Londoins B. Marine 30 Data K. Amaren 30 Data	Invested         Mit segments         N         2.27         4.48:00         0.08           Invested         Mit segments         0.0         3.35         4.08:00         0.08           Section         Mass         0.171         2.11         4.48:00         0.09	0
511         646/03         PE.000013         .23.37780         .808000           512         5079110000         .23.46207         .444705           513         50700000         .23.47200         .447940	PK         Area         Base Massed Blue, 200         G           3F         San Fanlo         Base Tensionen, 166 - Josén Holem         G           3F         Gauradues         Base Candedina S/N. Jasém Holem         G	Math         Mark         1         1         0         0         0         0           Mark         N         1         0 <td< td=""><td>0         .         0.00         Tail Pary Repay         N/A         0.00           0         .         0.00         Tail Pary Repay         6,4433         0.00           0         .         0.00         Tail Pary Repay         17320         0.00</td></td<>	0         .         0.00         Tail Pary Repay         N/A         0.00           0         .         0.00         Tail Pary Repay         6,4433         0.00           0         .         0.00         Tail Pary Repay         17320         0.00
514         7144621         SP\$2382678         23.0001         -46.71996           515         800077         8697A0006         50.11605         34.23376           516         7440921         80040002         30.01607         34.23376	2F Sar Fasis Bar Joint Konos, establish Saren F. de Teledor, prostavil, Moroandos I 26 Parte Augus Are Venezadas Saren Saren I 27 Dans Hanne Part Council Anna Mart Chine I	Review         Mast         NOT INFORMED         L(00.07         0.00           Review         Mast         6         NOT INFORMED         L(00.07         0.00           Review         Mast         6         NOT INFORMED         L(00.07         0.00	0
517 51261 B)(790225 2256256 4517478 518 51200 BI(CE018 224957 452801	R] Ke De Jacon Ave Passes Index, of 33 I R. De Jacon Rev Review Marries, 3 I	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	I         [PIX0ENX 02]         L235.00         Over Threemy         N/L         0.01           0         0.00         The Hwy Netry NUT DisCMAD         0.00           1         [PIX0ENS 02]         L230.00         Coar Prepary         N/L         0.00
B         Constraint	BJ         Res The Inners         Res The Inners in the Inner Neuristics 701, 50 at Protect         3           GD         Isombars         Neuro Control, Proceeds Resident G1P, 7003390         6           MG         Relations         Neuroids Antoine Control, 800, Prograds Mall         1	Korkov         Mot         (k+0)         X33-X37         4,400.00         0.00           Javelidi         Misseguing         2         6,07         4,000.00         0.00           Kerkov         Mar         6         1,327,064         4,000.00         0.00	1         processor ca         12553         Out response         N.O.         0.01           0         -         0.00         Third Peoply         64000         0.00           0         -         0.00         Third Peoply         64000         0.00           0         -         0.00         Third Peoply Reperty         64000         0.00
322         668473         MGEB 2.2018         .1555688         .455949           321         647562         35027554139         .1555682         .455986           321         645564         35027554139         .1551682         .455986	MG Rels Honoravir Ban Aloren Cora Reis, 60 1 MG Rels Honoravir Ban Experies James 1999 MG Rels Honoravir REA CARE(16, 121, 134, 240, 1978) (2, 53278) DE SUTTADER(1) 1	Bankap         Mass         (1+4)         Zd3         4,400.00         0.00           Bankap         Mass         4         3,18 + 16.6         4,400.00         0.00           Bankap         Mass         (2)+2,1+2,1+2,1-2,10         (1,18 + 2,00)         4,000.00         0.00	0         .         0.00         Tool Party Property         NOT DOCKMED         0.00           0         .         0.00         Tool Party Property         NOT DOCKMED         0.00           0         .         0.00         Tool Party Property         NOT DOCKMED         0.00
320         74.00.08         326.071.008         320.1070         -44.0010           326         710.0749         82.0271.0347         22.07080         -41.0078           327         53720         N112.0         22.75708         -44.0044	MG         Desception         Base for figures, 614         Figures, 614           RJ         Bits Descent         BEVA HERRED PROCES, 23         Figures, 747           RJ         News Spaces         AVINDED ANLO PROJECTING, 20         Figures, 747	Barshap         Mas         (3+4)         233354         4,0000         0.00           Barshap         Mas         4         6.07         4,0000         0.00           Barshap         Mas         4         2.25         4,0000         0.00	0 - 0.00 Owe Preparey Ν/Δ 0.00 1 [PR002002 02] 1,255.00 Owe Preparey Ν/Δ 0.00 0 - 0.00 No Islanuel NDT DirectRAID 0.00
526         531233         E[B/P601         2235577         452062           529         55562         B0F19728         2239756         453687           539         55562         B0F19728         2239756         453687           539         55562         B0F19728         2239756         453687	El Ker De Jacois AVENDEA PAULO DE PRONTEN, THE     E Jacois AVENDEA DASA AMERICA DE PRONTEN, THE     E Novem AVENDEA DASA AMERICA, SA	Review Mass 6 3.13 4,6000 0.00 Review Mass 7 3.73 4,6000 0.00 Review Mass 9 179 4,6000 0.00	0         .         0.00         Takel Pary Report         NOT DOCKMED         0.00           0         .         .0.0         Takel Pary Report         NOT DOCKMED         .0.0           0         .         .0.0         Takel Pary Report         NOT DOCKMED         .0.0
310 7747507 001025330027 16.48989 47.29988 310 312392 80075966 22.295764 43.59821	CO Guessi ARACUALI MEDINE, RUA M BI Re De Invite BLA RUE DALM DA PRODUCTA AN	Restor Mars 113 131 4,0000 0.00 Restor Mars C+T: 0,441+140 44000 0.00	0         100         Thail Pary Report         NOT INFORMED         0.00           0         .0.0         Na binned         NOT 20100MBD         0.00
NH         Tr40421         RJBT12071         Z3.00023         43.58078           370         Tr49807         RD4238806         Z23.5808         4407306	B Br De Jacobs AVENDA DAS AMBERCAS, 7777 B B Magazetha BEIEVYA BEI AANTON 0 B	Karlog Milegendig 21 0.59 4,0000 0.00 Karlog Masi 12 1.0 4,000 0.00	PROCESSI CRI         L_XXX0         Think Pany Nonperty         NUT DO CREATED         0.00           0         .0.0         Nut Idented         NUT DO CREATED         0.00
X7         X1000         R20220034         Z239909         4316%2           338         X3044         R0079042         Z239900         453620	<ol> <li>Mager BUARAPARE COLORING, QUADRA L, LOTRA S</li> <li>Ber De Javier Archives, MM - Bern N. Condensite Mine Henripe Incomes</li> </ol>	Terrela di 101 4,000 0.00 Review Mari 1 201 4,000 0.00 Review Mari 1 201 4,000 0.00	I         Production ()         Constrained () <thconstrained ()<="" th="">         Constrained ()</thconstrained>
NB         642197         COLUMN         L200802         L21781           NB         642197         COLVENTIO         5421993         4720914           NB         642184         COLVENTIO         5421993         4720922	Ag         Not the process         Not the process         No           ED         Generals         AN: NS, QUARRA 278, LOTES 3/18.         1           EO         Agenerals De General         BUA CAPERABLE Q 22, 0         1	Rening and privil 1.53 California 0.00 Rening Nase 13 6.550.051 Gallon 0.03 Rening Pes 4 4.53 6,60.00 0.03	0         .         0.00         Isal Pay regimy         0.01 Ind-Rock (Control of Control of Contro of Control of Control of Contro of Control of Con
M2         TABON         GUNA NOTE         LATIVO         JULION           M0         TABON         DULATIVO         JULION         JULION           M0         TABON         DULATIVO         JULION         JULION           M0         TABON         DULATIVO         JULION         JULION           M0         TABON         RETUBALI         JULIONO         JULIONO	Gramm REA 64 Day, C REA 44, A GRAA, EDB, NY AND METRICANTROL     Fill     Ker De basis     PRODUCE OF ADVANCED DEVISION IN A	No.         No. <td>0         .         0.00         Theod ang Program y         NUT DOCRADED         0.00           0         .0.00         Theodorem y         NUT DOCRADED         0.00           0         .0.00         Theod Program y         NUT DOCRADED         0.00</td>	0         .         0.00         Theod ang Program y         NUT DOCRADED         0.00           0         .0.00         Theodorem y         NUT DOCRADED         0.00           0         .0.00         Theod Program y         NUT DOCRADED         0.00
No.         TERMO         LALA SINGLE         LALAND           36         (113)         LALAND         LALAND	B1         Bas De Louiss         Bas De Louiss         S           B2         Bas De Louiss         AVINDA & COMINA PRESS, 100         3           B2         Danae De Cosin         B23/A PRESS, 243/2013         3	Barchur         Mast         4         33.72         4.480.00         0.00           Barchur         Mast         4         NOT INFORMED         4.000         0.00           Barchur         Mast         0.10         0.00         0.00         0.00	P         Particity         Total         Particity         Total         Particity         Partity         Partity         Particity
568         57196.2         Epityson         22.78728         -45.5660           509         572.07         20.9796         -42.4660           500         572.76         21.46706         -42.4660           500         572.76         21.42400         -42.0720	<ul> <li>Ber Die Janeien Rau L, 4 - Howards Tajans, Rei de Janeien - REJANEAN DA CHUMEANCAREA BED BEARAJ. GU Suparama Rau Agentinala Heava, 7 G     </li> <li>Balaneai REAA GERARD DA MARCEN DA ANTILIANA. ORD LED2 GO</li> </ul>	Invested Prot 31 NOT DEGAMED 4,00:00 0.00 Invested Prot 01 3/0" 4,00:00 0.00 Invested Prot 31 7 4,00:00 0.00	0
3%         647397         MORTORNE         1534494         4539331           3%         646267         MORTORNE         1539625         4539685           3%         646267         MORTORNE         1539625         4539685           3%         742997         36624534130         1539220         4539620	MG Bels Hosterour Res Compute, 679 com Res Control MG Bels Hosterour BEA MERICINAL, 52 G MG Bels Hosterour Jo Beau, 599 G	Bandup         Max         21         NOT INFOMMED         4,8550         0.00           bendud         Pos         31         3,525768         4,6650         0.00           bendun         Max         4         2,895039         4,6650         0.00	0         0.00         Tail Party Pagenty         NOT DPGRABD         0.00           0         .0.00         Tail Party Pagenty         2,00.05         0.00           0         .0.00         Tail Party Pagenty         X01 XPGRABD         0.00
556 52105 PA38A2556 A20620 A20620 556 741140 SPEESAC1 2354522 A44031	PA Mardin POLJEL 32, QD 13, LOTE 8, N.N. NOTES MARKA G 3P Jun Pairs BEA DO GASCAMETER, YEL, CIP, 306, 400 B	levelul Mingming 00 NOTINGAMED 4,0000 0.00 Rackap Mas (1-1-1) NOTINGAMED 4,0000 0.00	Image: Section of the sectio
NY         S2182         PARKS211         S3497         #11577           38         64211         15572314         S3497         #11577           38         64211         15572304         S35829         #52546	AP         Marcin         Rest Autocourty connected, corr         A           PA         Marcin         RESA CONCOURT AND COLOR AND LAND         O           FS         Vik Volto         RESA ALLON SERCES, NO.         O	Autom         Dot         Person         Dot         Dot           Jacobia         All segmenta         0         All Mon         0.01           Jacobia         All segmenta         0         All Mon         0.01           Jacobia         All segmenta         1         NOT DEGAMED         4,880.0         0.01	0 0.0 One Prepara N.(A 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0
NO         ACCOUNT         ACCOUNT         ACCOUNT           NO         647300         MARTENANAL         3575094         479971           NO         647304         MARTENANAL         357094         479971           NO         647304         MARTENANAL         479971         4669017	BAL         Data Difference	Rening Mari (1 <sup>44</sup> ) 200 (2010) (2010) Rening Mari (2.1) (2.1) (2.1) (2.10) Rening Mari (2.1) NOTINGAMED (2.010) (0.0)	0         0.00         one of property         NOT ROCKMED         0.00           1         [PROEDBEGO]         (233.00         Thind Face property         NOT ROCKMED         0.00           0         5.00         Not face property         NOT ROCKMED         0.00
N2 64715" MC(2040116 - 23,56054 - 43,58776 363 649603 MEX EXXXVI 15,28722 - 41,68639 364 640775 G27970274 - 16,45663 - 47,9538	MG Jus De Fran ATENDA MARIÃO DO REO MENACO B MG Casa REA (DA VERNITE COMPRE, 200 G GO Talgaram De Gran QUADA A, A CONVERSION B	Bankap         Mass         (24)         4.144         4,0000         0.00           lensibil         Pess         62         NOT DIFGRADED         4,000         0.00           sensing         Mass         4         3.17         4,0000         0.00	1         [PINDEND 02]         (J3330         Thail Prog Popeny         (J273)         0.00           0         -0.00         Own Progress         N(A)         0.00           2         [PINDEND 02]         J3030         Nucleared         NDT OPOROLID         0.00
Md         OMPT         OLYTPHAT         LAMMOD         CTURDS           Md         CONTR         SCH         CTURDS         AC2004           Md         TATORN         RELEVEN         AC2004         AC2004           Md         TATORN         RELEVEN         AC2004         AC2004           Md         TATORN         RELEVEN         AC2004         AC2004           Md         TATORN         RELEVENTION         AC2004         AC2004	MEG         Kan Paramaka         BICENVIA NR 384,0         G           El         Kar De Janoira         Ean Dian Farmin, N° 400         E           DF         Banaly Lands A Loss XN*, NN/N         E	Jaunikal         Mil segmentary         30         2.30         4_00.00         0.00           Ravikay         Mart         3         3.11         4_00.00         0.00           Ravikay         Opped         3         3.01         0.00         0.00           Ravikay         Opped         3         NOT INVOLVED         4_00.00         0.00	0         .         0.00         Nat Islamed         NOT INVEXMID         0.00           0         .         0.00         Third Pary Paperty         51/241/7         0.00           0         .         0.00         Third Paperty         NT INVEXMID         0.00
566 73-0520 0195-0025 13-7980 47-2771 567 67289 5629 5.400 13-5980 42-5986 570 67289 56275 13-59860 42-5986	DF Beedin SCHWANN OD 13E A. CHIT ACTARY, SA NORD, PLANO PEOPD 1 MG Bein Beener WC Bein Beener MC Bein Beener	Bardon         Max         4         M2         4,00,00         0,00           Bardon         Max         Max         Max         0,00         0,00           Bardon         Max         Max         Max         0,00         0,00	1 [PEREND 02] 12330 Thai Pay Papary 12334 0.0 0 0.0 Thai Pay Papary 12034 0.0 0 0.0 Thai Pay Papary NUT DOCEMED 0.0
571 53/743 809/80244 22/5529 45.4579     572 60/968 362/801667 3623902 4572011     572 5729 819/8005 72/5604 450000	E. Falinal Reson AVENESA RETERCIÓN DEPENSA, 59     MEG. Industro EXTERCIÓN PRENCEPAL, 50     Novem VI 10     Novem VI 10	Review         Mass         ad-state-to-to-to-to-to-to-to-to-to-to-to-to-to-	0         500         Tail/lwp Neprot         NGT DRORMED         0.01           1         PRENNIX OUL         L3.08         One Neprot         N_LA         0.03           0         -         5.00         Tail/lwp Neprot         N_LA         0.03           1         PRENENX OUL         L3.08         Tail/lwp Neprot         L3.04         0.03           1         PRENENX OUL         L3.286         One Neprot         N_LA         0.03
A         A			P         Particity         Tar.         Particity         Parity         Parity
377     714200     379262435     37579     460866     375     714200     379262435     234673     46085     379     714200     3792643     379     36675     379     36675     379	Num Paulo         Contrast Configuration         1           SP         Num Paulo         Ran Althoughpourg Ling, 02         3           SP         Num Paulo         Ran January, 183 - January, 184	Bardong         Mass         [8-4+6]         2,22 (904 + 0)24         4,4000         0.00           Bardong         Mass         [8-4+6]         2,22 (904 + 0)24         4,8000         0.00           Bardong         Mass         [8-4+6]         2,22 (904 + 0)24         4,8000         0.00	mp/max         mp/max         0.01           0         .0.0         The Hay Payor         NITD SPCRADD         0.03           0         .0.0         The Hay Payor         NITD SPCRADD         0.03           1         IMM/MAY PT         1.00         The Hay Payor         NITD SPCRADD         0.03
	means train Res. Jan Com, 83 1 37 Press Generic Res. No. Los Octoberos 1 38 Januar - Januar January 10, Colliments 1 39 January January 10, 10, 10, 10, 10, 10, 10, 10, 10, 10,	manual protestary         max         4,4000         0.00           Bendrag         Mar         β+1+6         2/0         4,4000         0.00           Bendrag         Mar         (1+1+1+1+1+1+2)         2/0         4,0000         0.00	presenter seq         4,4500         mm ray report         6,153.0         0.00           0         -0.00         Tabel Pay Payers         MIT DPCRMBD         0.00           0         -0.00         Tabel Pay Payers         MIT DPCRMBD         0.00
od         TATEC         BYDDINK         2100731         46.3209           561         7142942         SPEDTEC2         236.3214         46.7034           561         7142962         SPEDTEC2         236.3214         46.7034           561         7142962         SPEDTEC2         236.3214         46.7034           561         7142963         SPEDTEC2         236.3236         46.6032	M         Journ         An Produce Water, W., Jour Marsen,         J           SP         Journ         Produce Water, W., Journal, M., Journal, J., Journal, M., & Journa, M., & Journal, J., & Journal, M., & Journa, Journal,	annum das (3+1+1+23-2.5). 500 4,4000 000 Ranhag Kau (3+1+1) 2(7-8/2-1-6/2+1/0+0.50) 4,0000 000 Ranhag Kau (3+1+1) 2(7-8/2+1/0+0.2+1/0) 4,0000 000	N         0.00         more reg Person         XX7344         0.00           6         0.00         Table Person         NUT BYCEMBED         0.00           6         0.00         Table Person         XX122         0.00
300         7143034         37C,334029         323026         47.0089           361         7142033         3793Q508         23.6826         46.0521           361         7142033         3793Q508         23.6826         46.0521           362         617486         790.38064         23.67537         49.8866	MP         Complexity         East Devider Alloware, CTI - Control         I           MP         Xan Fluids         East Transfam, CTI - Control         I           MP         Community         East Transfam, CTI - Control         I	Rawlog         Mass         (?+1+3+1+1)         1.01         4,00.00         0.01           Rawlog         Mass         (?+1+1)         3,133 (F/201 + 0,752 + 1,107)         4,00.00         0.00           Deniedal         Mate supporting         71         5,147         4,00.00         0.00	0         .         0.00         That Famp Negativy         ξ/14.00         0.03           0         .         0.00         That Famp Negativy         ξ/34.01         0.03           0         .         0.00         Coas Property         χ/λ         0.03
368         7147710         FEMAL2201M         AUX844         3429000           369         7142070         347620703         2130000         4640000           359         7142060         3792270607         2332961         4640000	P         Carmen         Baylo & Dense, 17         Gene           4         Kr. Mark         Bar Monday, 17, 17, 16, 16, 16, 16, 17, 17, 16, 16, 16, 16, 17, 17, 16, 16, 16, 17, 17, 16, 16, 16, 17, 17, 16, 16, 16, 17, 17, 16, 16, 16, 17, 17, 16, 16, 17, 17, 16, 16, 17, 17, 17, 17, 17, 17, 17, 17, 17, 17	Kaning         Mass         (p+b+0)         3.309         4,00.00         0.00           Koving         Mass         (p+b+2)         3,001         9,221         +,224         4,400.00         0.00           Koving         Mass         (p+b+2)         3,001         6,203         6,207         6,301         0.00	0         .         0.00         Thair Pary Property         \$20000         0.03           0         .         0.00         Thair Pary Property         \$23377         0.03           0         .         0.00         Thair Pary Property         \$25732         0.03
NE         T41472         SPERVENSE         23.8681         44.6837           NZ         8.7280         SPERVENSE         23.6681         44.6837           NZ         8.7280         SPERVENSE         23.6681         44.67375           NZ         8.7280         SPERVENSE         23.6681         44.67375           NZ         8.7280         SPERVENSE         23.6681         44.67375	MP         Jan Paulo         MAAMEDA RED CLARED, REA VEEX. 117 MAD 307         1           MP         Jan Paulo         K. Analonis Jud. Analysis, J. N., Viki Panson         G           MP         Jan Paulo         K. Analonis Jud. Analysis, J. N., Viki Panson         G           MP         Jan Paulo         K. Analonis J. N., Viki Panson         G	Barring         Diff DFCMMDD         April DF         Diff           Investidal         Proc         40         8.33         4,0000         0.00           Barring         Mar         1.03-1,07-2,000         0.00         0.00           Barring         Mar         0.03-1,07-2,000         0.00         0.00	Dist         Task Pag Yapay         X/1727         0.00           0         . 0.00         Task Pag Yapay         X/0727         0.01           0         . 0.00         Task Pag Yapay         X/0730         0.03           0         . 0.00         Task Pag Yapay         X/0730         0.03
Image: Note: Control (Note: Note: Not	30         Abab         A		Str.         Disklit         Oil           0         -0.0         Theory Party         NOT DOCKMED         Oil           0         -0.0         Theory Party         NOT DOCKMED         Oil           0         -0.0         Theory Party         MONR         Oil           0         -0.0         Conthermorie
3%         Teleform         PAGE (12240)         A.3.4009         A&400%           3%         Teleform         SPERE INCER         23.87108         -40.7227           3%         Teleform         SPERE INCER         23.87108         -40.7227           3%         Teleform         SPERE INCER         23.87108         -40.1227	PA         Boles         EXEMPTION         FUNCTION (1) (V) (V) (V) (V) (V) (V) (V) (V) (V) (V	Javelid         Mil segment         0         2.352         4,000         0.00           Keelup         Mas         (1+1-6)         NOT DNGABED         4,000         0.00           Keelup         Mas         (1+1-6)         NOT DNGABED         4,000         0.00           Keelup         Mas         (1+1-6)         NOT DNGABED         4,000         0.00	0         .         0.00         Outerreport         N/Λ         0.00           0         .         0.00         Third Yony Peptry         10,2033         0.00           0         .         0.00         Third Yony Peptry         12,2033         0.00           0         .         0.00         Third Yony Peptry         1,2033         0.00
300         7142462         3490379303         -23550377         -4668273           481         7142720         3490381734         -2360396         -4669218           481         7144564         C204,56231         -375462         -3866907	Ser Pado     AL ETU, 2017 VEA CARACULUSA CONDEMINIO EEDITIGO (2016) ENCERNA E     Ser Pado     An Engenheire Carlos Berris, 81     Ser     Ser Pado     EUA VEEBALLO AUCUAL, EE JAKNA 2200     EUA VEEBALLO AUCUAL, EE JAKNA 2200     EUA	Review Mass (1+1+1) (2018-0,2018-0,223 4,00.0 0.00 Review Mass (1+1+1) (2018-0,2018-0,223 4,00.0 0.00 Review Mass (1+1+1) (2020-1,2018-0,2018-0,2010 4,00.0 0.00	0         .         0.00         Dail Any Peperty         X0164         0.00           0         .         0.00         Dail Any Peperty         X001 DOCRMED         0.00           0         .         0.00         Dail Any Peperty         X0000         0.00
402 7147064 PAAR(222) -3.5480 -45.5879     401 7149621 PDTA501 -866496 -56672     401 7149621 PDTA501 -866496 -56672     401 714962	PA Bolom BLA DOTRITO PADIOREMA, DR. ANANDEREA, A/N G PC Revie BLA DOTRITO PADIOREMA, DR. ANANDEREA, A/N G PC Revie BLA DOTRITO PADIOREMA, DR. ANANDEREA, A/N G	Javoid         Mit segments         0.0         0.39538         4,40100         0.00           Kordow         Mari         5         NOT INFORMED         4,0010         0.00           Kordow         Mari         4         NOT INFORMED         4,0010         0.00	0
400 754200 PE-IFO332 833004 41.0000 400 754200 307577840 335007 444000 400 754200 250000 75000 410004	Pt. Massie BEA (1981 YEB/NTE DI ARAYO), 0     O     SP. Am Pails     ANANDA PULLEYA A REPROD ASSIE1274     SP     Common March 2010 NOTES     SP AND A REPROD ASSIE1274     SP	Involuti         Mol supports         41         6.25         4,00.00         0.00           Kardson         Mast         R-1+0         0.2029+120229         0.000         0.00           Londot         Mast         R-1+0         0.2029+120229         0.000         0.00           Londot         Missemilier         30         NOT DIVERSIDE         0.000         0.00	0
400 T20007 FX48-30110 . 1.45478 . 4449811     400 T24478 FX48-341     400 T24478 FX48-341     400 T24478 FX4878	Continuum Of Backer, N. 400 U     PA Backer, N. 400 U     PA Backer, N. 101 BACKLAN, N. 401 U     PA Backer, Backer, Back Major Ammunikash Swam Mala, 103, 104 [Jacobi II     PA Backer, Back Major Ammunikash Swam Mala, 103, 104 [Jacobi II     PA Backer, Back Major Ammunikash Swam Mala, 103, 104 [Jacobi II     PA Backer, Back Major Ammunikash Swam Mala, 103, 104 [Jacobi II     PA Backer, Back Major Ammunikash Swam Mala, 103, 104 [Jacobi II     PA Backer, Back Major Ammunikash Swam Mala, 103, 104 [Jacobi II     PA Backer, Back Major Ammunikash Swam Mala, 103, 104 [Jacobi II     PA Backer, Back Major Ammunikash Swam Mala, 103, 104 [Jacobi II     PA Backer, Back Major Ammunikash Swam Mala, 103, 104 [Jacobi II     PA Backer, Back Major Ammunikash Swam Mala, 103, 104 [Jacobi II     PA Backer, Back Major Ammunikash Swam Mala, 103, 104 [Jacobi II     PA Backer, Back Major Ammunikash Swam Mala, 103, 104 [Jacobi II     PA Backer, Back Major Ammunikash Swam Mala, 103, 104 [Jacobi II     PA Backer,	Annual         Bar         D         (1/28+1/28+1/28)         Annual         D           Review         Max         (1+27)         (1/28+1/28)         4,000         0.00           Review         Max         (1+27)         NOT DPGMMED         4,000         0.00	0         0.00         Dial Facy Reparty         N/A         0.00           0         4.00         Dial Facy Reparty         N/A         0.00           0         4.00         Dial Facy Reparty         N/A         0.00           0         4.00         Dial Facy Reparty         N/A         0.00
430 PE483 REMPERFIC ANTWO NATION ANTICO 431 STUTIO PROCEMPLATING ANTICA NATION 432 STUTIO PROCEMPLA 723208 AMAZINA 432 STUTIO PROCESSION AND ANTICA AMAZINA	IN Presenter II (CONTRAL, CONTRAL, CONT	Javenšel         Jaši sepaning         64         1.50071         4,66.00         0.00           Javenšel         Jaši sepaning         61         1.724         4,66.00         0.00           Javenšel         Jaši sepaning         61         1.724         4,66.00         0.00           Javenšel         Jaši sepaning         61         3.92077         4,66.00         0.00	0         .         0.00         Out Pripage         N/A         0.00           0         .         0.00         No Islands         NOT DirOCRMID         0.00           0         .         0.00         Table Pripage         3,0000         0.00
h         0         0.000 </td <td>CI.         Postalow         An Yuanar de Canto, OH         1           SP         San Paulo         AV DO DOTADUA Y 2004, CENTED         1           R]         Be De Janese         AVENDA GEBELSAMED DANTAS, BE         1</td> <td>No.         No.         No.<td>Image: section of the sectio</td></td>	CI.         Postalow         An Yuanar de Canto, OH         1           SP         San Paulo         AV DO DOTADUA Y 2004, CENTED         1           R]         Be De Janese         AVENDA GEBELSAMED DANTAS, BE         1	No.         No. <td>Image: section of the sectio</td>	Image: section of the sectio
456 M3079 REDUCES J228007 4338075     45707     45707     45707     45707     45707     45707     45708     41708	Bj Bir De Jawain BECA MANACA, 0 G     BA Banada Balana Balana Balana G     Balana Ba	Javenški         Prot         61         7.3         6,400.0         0.01           Javenški         Jaffangening         31         5.74         6,400.0         0.01           Javenški         Jaffangening         31         5.74         6,400.0         0.01           Javenški         Jaffangening         31         6,42         6,400.0         0.01	0         .         0.00         Thaid Page Papenge         NOT DOCRABID         0.00           0         .0.00         Thaid Page Papenge         273.13         0.00           0         .0.00         Thaid Page Papenge         773.20         0.00
1         2 <th2< th=""> <th2< th=""> <th2< th=""> <th2< th=""></th2<></th2<></th2<></th2<>	SP         Masters Proce         Res for the basedon SPD         J           81         Re De bases         RUA COLUMBA SWATCHE & PRANT DAW LADITALS         D           97         Comba         An Allow Concerns 40, Const And Concerns 40, Cons	Image         Image <th< td=""><td>1         [PB0E000 OI]         1,255.00         Thail Pay Suppry         LN1 M.         OB           0         .0.00         One Paymy         N/A         .0.00           0         .0.00         Thail Pay Suppry         MIT MOREMED         0.00</td></th<>	1         [PB0E000 OI]         1,255.00         Thail Pay Suppry         LN1 M.         OB           0         .0.00         One Paymy         N/A         .0.00           0         .0.00         Thail Pay Suppry         MIT MOREMED         0.00
423         656,053         E02.147,0768         .76,47596         .51,37286           421         7544333         597500713         .75,43100         .46,6080           470         74,4533         87950723         .75,43100         .46,6080           470         74,79373         873,5729723         .55,43100         .46,6080	Konstan Do Mal LINULS COMPREDA CONTRO - MALY ADDREDO M.L.     O     See Paulo     Kato Marka     Konstanti	Milling         A         L2         L4000         0.01           Review         Max         ()+(+)         NOT INFORMATIO         4,400.00         0.00           Review         Max         ()+(+)         NOT INFORMATIO         4,600.00         0.00	•         •
420 T14267 197054.000 200007 443003 420 T14267 197054.000 2349934 444003 420 T142711 37050000 2349934 444003	PP New Katologi Print, 2011 (2012 Print, 2011) (2012 Print, 2012) Print PP New Parks Around Supported at 151 Supported at 251 Parks 1 PP New Parks Res Voltamer Marries, 116 Parks	anna (***) 2,000 000 Bachay Mas (***) (2,0**,600 4,600.00 000 Bachay Mas (***) 4,00* 4,000.00 000	0 0.00 Thail Perpeny 0.0224 0.00 0 0.00 Thail Perpeny 0.0224 0.00 0 0.00 Thail Perpeny 0.0224 0.00
428 T44008 PARATAZ A4980 447942 429 52343 A39(342142 A4980 447942 429 52343 A39(342142 A19872 A19919)	AN Num PA Brien Av. Guerri Madaina, N.N. Roban G AP Issues AV. PRANCISCO DE MILLO PARENTE, 147, REMEDICE E	reminia Francisco da Statu Constante Constante Jacobia Malengenia é NOTINCIMEN 4,000 0.03 Jacobia Malengenia 6 1.012 4,000 0.03	0         .
400 742801 376023013 233352 462897     40     32340 3623362 23008 442708     43736     45708     45708     454007	M2         June Paulo         DE A ARAGET, DA         D           MA         June Lano         ATE EXCLUSION (SECURING ANY)         D           MA         June Paulo         ATE	Review New (1+(++)) (120+20+(11+0) 4,000 0.00 leveled 7m 1 10 4,000 0.00 Review New (1+(+)) N0(+),000(+),000(+0.00) 4,000 0.00	Image: Section (Section (
55         X2143         \$2023182         2005         442"05           67         -71447         \$7287.0791         23.005         46.007           60         -71412         \$7287.0791         23.005         46.007           60         -71427         \$7287.0791         23.005         46.007           60         -71427         \$7287.0791         23.0724         46.007           60         -71427         \$7287.0791         23.0724         46.0075           60         -71427         \$7287.0791         23.0724         46.0075	MP         Non Fasis         Ran Fasis and Solida Yana, WE (Fasia) (M) Companies         P           SP         Non Fasis         RUA EPOCTO FREETRAL NO.         P           SP         Non Fasis         RUA APPNOVIAL, SP ACL REAC(A)         P	Barshop         Max         ()+>7)         NOT IVGGMED         4,000         0.00           Barshop         Max         ()+>10         (),211+0,2211+0,144         4,000         0.00           Barshop         Max         ()+>10         (),2211+0,2211+0,144         4,000         0.00           Barshop         Max         ()+>+0         (),2211+0,2211+0,144         4,000         0.00	0         .0.0         Third Prop Property         7,463.77         0.01           0         .0.00         Third Prop Property         5,653.77         0.01           0         .0.00         Third Property         5,653.77         0.01
01         Table         Joshan         Joshan         Joshan           95         M.2011         M.20120         Joshan         Joshan         Jata           96         M.2011         M.20120         Joshan         Joshan         Joshan         Joshan           96         M.20120         M.20120         Joshan         Joshan <tdjoshan< td=""></tdjoshan<>	AM         Massan         Arr. [, η/n. Res New & Abril [2bit server Daw Reave]         G           R]         Kar Gravalo         AVENDIA ARC MICE/EE, 44         B           R]         Res De Javien         Read Damper Freeman, 94 Reader Relation         B	Janukuk         Mal segmenteg         N         11.102000         4,0000         0.00           Konlog         Max         (1)01*7/01         6/02*6/02         6/02*         0.00           Konlog         Max         (7)0*7/01         6/02*6/02         6/02*         0.00           Konlog         Max         (7)0*7/01         2+3/02*6/02*         6/02*         0.00         0.00	0         0.00         ThisProp Peperty         NOT DOCRMED         0.00           0         0.00         ThisProp Peperty         NOT DOCRMED         0.00           1         PRODOD COL         (23300)         Own Peperty         N/A         0.00
470 X71278 RQUAL0041 22283428 -4528778     480 X7150 RQUAL004 22283408 -4538112     480 X71508 REPUTION 22283408 -4538112     480 X70408 REPUTION 222878982 -4557847	E. De Jacois ANTORIA VINTE DE JANEED, N.N., E.HA DO GOVERNADOR - RED DE JANEERO (D     E. DE Jacois Karoline (D)	Involid         Mit segments         N         6.2         6,000         0.00           Reschop         Mare         (4+6+2+2+2+2+2+2)         6,042         6,000         0.00           Nervield         Mit segments         25.0         8,44         6,000         0.00	0         .0.0         Data Flavy Property         NOT DFORMED         0.01           1         [PE00002.00]         1,235.00         One Property         N/A         0.00           0         .0.0         Nutritional         NTT DFORMED         0.00
442 511429 E[TEOH43 22.11528 45.2166 441 51246 EE75(2154 22.8000 45.2178)	E] Teo Res ELA RARÃO DO EIO REANCO, 201 1 El Re De levies AV. MARTRO PARLO S (ELX). E	Review Tatala Maar 11.2 15.8 (2010) 0.00 Review New Prince Not NOT INCOMED 4.00.00 0.00	0 0.00 One Preperty N <sub>1</sub> (Å 0.00 1 [PENDEND OI] 1230.00 Not Bollowed NOT BOOKABID 0.00 VICT BOOKABID 0.00
64         AVM2         R05/2027         Z00777         440702           69         AVX82         R5569656         Z10777         440702           69         AVX82         R5569656         Z10777         440702           69         AVX82         R5569656         Z10778         440301           69         AVX82         R5569654         Z10778         440302           69         AVX82         R54697         Z20802         440304           69         AVX82         R54697         Z20802         450304           69         AVX82         Z20802         A40304         640304	30         b. b. b. a. International and a second seco		0         0.00         Dial Facy Papay         100000         0.00           0         4.00         Dial Facy Papay         10000         0.00           0         4.00         Dial Facy Papay         10000         0.00           0         4.00         Dial Facy Papay         42223         0.00
447 NOTSI 3PREMI 214401 44.1224 448 NOTSI EDLAND 224842 44.1220 448 NOTSI EDLAND 224842 44.1220 448 NOTSI EDLAND 24.1211	37 Sar Fado Kar Saria Karana Kar Karana Karana Kar Karana Karana Karana Karana Karana Karan Karana Karana Karana Karana Karana Karana Karan	Javelida         Mill seguring         41         3.62         4,66.00         0.00           Javelida         Mill seguring         33         4.44         4,68.00         0.00           Javelida         Mill seguring         53         4.44         4,68.00         0.00           Javelida         Mill seguring         60         1.0120         4,88.00         0.00	Image: Constraint of the second sec
40 17009 EXECTS ALPS 346717 40 17009 EXECTS 470 ALPS 346717 40 17009 EXECTS 47006 35133 40 17009 EXECTS 45103 35365	RN Prom Research AVENDA's MANNOEL REDERECTORS (88) 00 RN Source Das Nords AVENDA's A RN Junia 02000 (2000) 0000 (2000) 0000 (2000) 0000 (2000) 0000 (2000) 0000 (2000) 0000 (2000) 0000 (2000) 0000	Javelid         Per         23         J.0371204         4,480.0         0.00           levelid         J.61 expensing         41         630314         4,480.0         0.00           levelid         J.61 expensing         41         2235         4,480.00         0.00	0         .         0.00         Nu klassed         NOT DPGRABD         0.00           0         .         0.00         Nu klassed         NOT DPGRABD         0.00           0         .         0.00         Nu klassed         NOT DPGRABD         0.00
401 877361 PEDEMICI36 AUG8P - 340930     454 877179 34.0000079 - 444234 - 444760     45914 994.4447600 - 459141 - 3544775     459 175944 994.444.0000 - 359141 - 3544775	PE         Res         Res         Manual de Madeiren         G           MA         Ker Manue Du Manulen         BEUN BEUN LADO         G           PE         Alagonita         BEUN DE MADUC CERTA/N         G	Januarki         Adi supporting         31         0.75         4,0000         0.00           Januarki         Adi supporting         12         3.4         4,0000         0.00           Januarki         Adi supporting         10         NOT TORCAMED         4,0000         0.00	0         .         0.00         Thail Page Pagenty         2001.13         0.00           0         .         0.00         Thail Page Pagenty         840.12         0.00           0         .         0.00         Cons Pagenty         N/A.         0.00
5% <u>61203</u> 2622825000 262899 454092     5% <u>61205</u> PEMC0007 2542707 4519422     6% 07 PEMC0073 2542707 5519803	MG Masses REANING MENCERA.NC G TE Masses Realized Reads and Access 1 (2010) TE Masses Realized Reads 71 (2010) TE Masses 1 (2010)	Invested Prot N NOT PRODUND 4.00.00 0.00 Karden Mar 4 0.033 4.00.00 0.00 Karden Mar 3 2.116 4.00.00 0.00	0         0.00         Thail Pap Papary         2,175.7         0.01           0         0.00         Thail Pap Papary         NOT DOCOMID         0.01           0         0.00         Thail Pap Papary         NOT DOCOMID         0.01
40 45022 362/EN420 20.5000 4271000 40 6756 36200043 27.9901 46.0972 40 69900 PEX20737 24.6997 467053	MG         Desemin         REDEVIDA MG 203, Kol 129         G           MC         Inc. Form         RULA (CAR) ANTERNATIO (REQUES), 409         R           MC         Inc. Form         RULA (CAR) ANTERNATIO (REQUES), 409         R           ME         Manamendia         RULA ANCERSE RECEPTA AND         R	Javoid         Kell segnator         21         L21         4,00:00         0.00           Kordsop         Mari         4         NOT INFORMED         4,00:00         0.00           Javoided         Gopol         40         1.07         4,00:00         0.00	0         .         0.00         Not bluesed         NOT DOCKMED         0.00           6         .         0.00         Third Propring         NOT DOCKMED         0.00           0         .         0.00         Coar Propring         N/A         0.00
462 66050 PEAIN051 233827 324801 463 66653 PEAIN5525 235480 AAR'000 464 74696 PERIOD AND AND AND AND AND AND AND AND AND AN	PE Assess REALISTORIES 400 - [LTENT - CERTRA C PE Pro Do Igano AVIONDA PERPENSION - CERTRA C PE Pro Do Igano AVIONDA PERPENSION - CONTROL - CONTRO	Invokil follogensing 100 NOT INFORMED L(00.07 0.00 Raving Mass 4 L00 L(00.0 0.00 Londol Marines 50 772 L000 0.00	0
405 714903 EMERVIOIT 3.8658 3521072 406 714903 ENLIVABIT 3.8678 3.32100	EN Nutl Res Norl Eve 101. Bars Vender 1 EN Nutl Res Nord Leastly 175. Ed. Marketers 0	Terrefel Millagening 32 201 4,000 0.00 Terrefel Millagening 32 200 4,000 0.00	0         .
446 754703 247.354837 22.11156 47.06472 449 754703 247.354832 22.0980 C.0086	2         Jordania         ELEMENTICUE           3         Jordania         ELEMENTICUE           4         Jordania         ELEMENTICUE           5         Jordania         ELEMENTICUE           6         Jordania         ELEMENTICUE           7         Jordania         ELEMENTICUE           8         Jordania         ELEMENTICUE           9         Jordania         ELEMENTICUE <td>Review New Review 2007 4.0000 0.00 Review New Oritica 2007 4.0000 0.00</td> <td>0</td>	Review New Review 2007 4.0000 0.00 Review New Oritica 2007 4.0000 0.00	0
477 744155 597269577 353323 466359 472 754155 597269577 3533233 466350 472 7541572 595854651 2169278 463501	Arr         Data Park         Data Park         Arr           SP         San Park         Large Twelse, (, Here Funds         F         F           SP         San Park         PARAMETER ANALYSIS, 200 - CENTRO - SAO BERNARDO CARPO         F	Karden         Mass         Difference         Addition         Difference           Revelop         Mass         3-3-31         1,1550021+0,279902+0,799902         4,880.00         0.00           Revelop         Tabalac Mass         13.2         2.39         4,880.00         0.00	0         .
17 T1110 2990 100 10 10 10 10 10 10 10 10 10 10 10 1	Air Feinado Die Campo     Ani NUNDA JOA BRAND, LEZZ- AND ALE NAMEDO DO CAMPO - NP     Nex Remails Die Campo     AVINDEA JOA BRAND, LEZZ- AND ANA DE NAMEDO DO CAMPO - NP     Nex Canses Dis Ma     Kan Radad Geweic Rempire, 977 - Junta Pada - 3-C3	Rening and (PPPPPP) 200 000 Rening Mast 3 NOTINGARD 4,800 0.00 Rening Mast 3 NOTINGARD 4,800 0.00	0         .         0.00         Intel Pay Pageny         0.01         0.01           0         .         0.00         Datel Pay Pageny         0.011         0.00           0         .         0.00         Datel Pay Pageny         0.011         0.00           0         .         0.00         Datel Pay Pageny         NOT DOUMD         0.00
475 7141071 3070071807 3253604 4654074 477 7141607 307053078 3235056 4622774 478 714160 50715050 235606 4435211	MP         Non-Paulo         Rear Threaw do Mana, 1002 Hefa Yimon No Fluido         H           MP         Norron         Rear Materia Ministrato, 1022 Anterno No Fluido         H           MP         Norron         Rear Materia Ministrato, 1022 Anterno No Fluido         H           MP         Norron         Rear Mana, 1022 Anterno No Fluido         H           MP         Norron         Rear Mana, 1022 Anterno No Fluido         H           MP         Norron         Rear Mana, 1022 Anterno No Fluido         H	Bankap         Mast         3         NOTINGAMED         4,461.00         0.00           Renkap         Mast         (3+1)*1         2.431         4,661.00         0.00           Renkap         Mast         (3+1)*1         2.431         4,661.00         0.00           Renkap         Mast         (3+1)         NOTINGAMED         4,861.00         0.00	0         .         0.00         Taid Prop Property         4/00.41         0.00           0         .         0.00         Taid Prop Property         3/21/17         0.00           0         .         0.00         Taid Prop Property         3/21/17         0.00
475 T14278 39T5305 - 2554201 - 453587     49 T47797 - 39T53587 - 2556270 - 453587     49 T42597 - 39T53587 - 2556270 - 453587     40 T4259     50 T42597 - 39T5358 - 45325	3P haves ELA [ULA DIS MING/ULA 104, VELA MATRIX AND ULA 104, SP 1 3P haves PERCA PARAMENTS, B 1 3P New Parks, PARAMENTS, PARAMENTS	Bankay         Mast         (1+1+1+1)         NOT INFORMADI         4,461.00         0.00           Bankay         Mast         (1+1+1+1)         NOT INFORMADI         4,661.00         0.00           Bankay         Mast         (1+1+1+1)         NOT INFORMADI         4,661.00         0.00           Bankay         Mast         (1+1+1+1)         NOT INFORMADI         4,661.00         0.00           Bankay         Mast         (1+1+1)         NOT INFORMADI         4,661.00         0.00	0         .         0.00         Thad Prop Property         3/0.4.1         0.00           0         .         0.00         Thad Prop Property         4/27.47         0.00           0         .         0.00         Thad Prop Property         4/27.47         0.00           0         .         0.00         Thad Prop Property         4/2000         0.00
682 754203 307579534 23.5777. 44.65730     697 752235 307979548 23.5276 44.65728     698 7542342 3079579548 23.57720 44.65780	See Fasis     Alamonda Isaasim Fasisima de Malos 1052 - M. Paulon - Nie Fasis     See Fasis     Nan Fasis     Kon Marine Isaa (117 - Junior Fasisima - Nie Fasis     See Fasis     Kon Marine Isaa (116 - Junior Fasisima - Nie Fasis     See Marine Isaa (116 - Junior Fasisima - Nie Fasis     See Marine Isaa (116 - Junior Fasisima - Nie Fasis     See Marine Isaa (116 - Junior Fasisima - Nie Fasis     See Marine Isaa (116 - Junior Fasisima - Nie Fasis     See Marine Isaa (116 - Junior Fasisima - Nie Fasis     See Marine Isaa (116 - Junior Fasisima - Nie Fasis     See Marine Isaa (116 - Junior Fasisima - Nie Fasis     See Marine Isaa (116 - Junior Fasisima - Nie Fasis     See Marine Isaa (116 - Junior Fasisima - Nie Fasis     See Marine Isaa (116 - Junior Fasisima - Nie Fasis     See Marine Isaa (116 - Junior Fasisima - Nie Fasis     See Marine Isaa (116 - Junior Fasisima - Nie Fasis     See Marine Isaa (116 - Junior Fasisima - Nie Fasis     See Marine Isaa (116 - Junior Fasisima - Nie Fasis     See Marine Isaa (116 - Junior Fasisima - Nie Fasisima     See Marine Isaa (116 - Junior Fasisima - Nie Fasisima     See Marine Isaa (116 - Junior Fasisima     See Marine Isaa     See Marine Isaa (116 - Junior Fasisima     See Marine Isaa	Barcher         Mast         (1+1+2+3)         3.33         4.4000         0.00           Barchery         Mast         (1+1)         NOT DPGMMED         4,4000         0.00           Barchery         Mast         (1+1)         NOT DPGMMED         4,0000         0.00           Barchery         Mast         (1+1)         NOT DPGMMED         4,0000         0.00	0
601         TLEBER         SPETZ2000         233300         4640023           601         TLEE'S         SPETZ2000         233400         464007           601         TLEE'S         SPETZ2000         233400         464007           601         TLEE'S         SPETZ2000         233400         464007	MP         More Fash:         Base Monore Aligney, CPU: Readines - Mar Fash:         1           MP         May Fash:         Lagor do: Assessing, NO	Bardop         Max         [1+1+1+0]         2.83         4,0100         0.00           Randop         Max         β+1+1         NOT DPGMMD         4,0100         0.00           Randop         Max         (5+1+1+7)         2.151         4,0100         0.00	0
68         TLENG         SPRYMEDT         23.00%         46.000           68         TLEN         SPRYMEDS         23.00%         46.00%           68         TLEN         SPRYMEDS         23.00%         46.00%           68         TLEN         SPRYMEDS         23.00%         46.00%           68         TLENS         SPRYMEDS         23.00%         46.00%	NP         Nor Pash         Bit A MANURE DA SAVEREGA, 181         1           NP         Nor Pash         Eas Gaireau, 22 - Monas         3           NP         Nor Pash         Eas Gaireau, 22 - Monas         3           NP         Nor Pash         Eas Gaireau, 22 - Monas         3	Bankap         Mast         β+1+6         NOT IPPGMMED         4,0100         0.00           Rankap         Mast         β+1+6         NOT IPPGMMED         4,0100         0.00           Rankap         Mast         β+1+6         NOT IPPGMMED         4,0100         0.00	0         0.00         That Pay Paymy         MAD 17         0.00           0         . 0.00         That Pay Paymy         7,502.77         0.00           0         . 0.00         That Pay Paymy         NDT 20 CRABD         0.00
40     71.420     47420(77)     21.41722     44.4598     40     71.420	AP InterPark No (Animatria 2014) Manada      Separating 2014 Manada      Separati	Bankap         Mass         β+1+ξ         3.07         4,0500         0.07           Bankap         Mass         β+1+ξ         (0,000+0,	0
491 714935 89702340 2154987 445323     697 714947 99753596 215399 445323     697 714933 897823972 235396 445499	American Ameri American American A	Kanica         Mass         (1+1+1+1)         NOT DF0ADAD         L00.00         0.00           Kanica         Mass         (1+1)         3.37         L00.00         0.00           Kanica         Mass         (1+1)         3.37         L00.00         0.00           Kanica         Mass         (1+1)         0.00         0.00         0.00	0         0.00         That Periphy NDT INFORMED         0.00
47 14445 X042N0002 273924 463499 49 14232 9623043 254308 41230 49 14233 9623043 254308 41231	SC Plotnergeds BEALACHERALCOUTINHEAPY 1 PR Causia NEXA ADDRETO PERLOAD, LCA, PENTER, CHETHERA PR Causia Area New York Parks and Araba, PR	Kaulup         Mass         1         NOT DFGMMED         4,00.00         0.01           Kaulup         Mass         (0+1)         2.13         4,00.00         0.03           Kaulup         Mass         (0+2)-(2)         2.13         4,00.00         0.03	0
NO         T44203         PRCT00043         -23,4780         -47,2700           NO         T44037         PR2DA003         23,1176         -31,1271           NO         T44047         XLM2R02011         -39474         -22,0003	PR Contine Avendue de Repúblico, 6464 - Contre - Caroline - PR PR Contine Avendue de Repúblico, 6464 - Contre - PR PR Contine Avenue Av	Rankap Mass 6 1588 4,0500 539 Rankap Mass (1+4) 2,325 4,0500 539 Rankap Mass (1+1) 1,328 4,0500 539	0         . 0.00         Third Pergrey         U/0144         0.00           0         . 0.00         Third Pergrey         NIT DVCMMD         0.00           0         . 0.00         Third Pergrey         NIT DVCMMD         0.00           0         . 0.00         Third Pergrey         NIT DVCMMD         0.00
NIT         T44096         MPLTVIZZI         Z133380         4434400           NIL         T44096         E08218091         324090         3200201           NIL         T44096         E08218091         324090         3200201           NIL         T44096         E08218091         3210407         3210201           NIL         T44096         E0524066         2113867         3117664	SP         Companies         K PICO DA EAND HEA, 20           EX         Ex Count         EX ADD/OCATA DE MELLO, 31           EX         Exam Do Joi         Examera, Autoria Transon, 31.0           EX         Exam Do Joi         Examera, Autoria Transon, 31.0	Rankap         Mass         D         3.33         4,86.00         0.00           Rankap         Mass         4         NOT DVG&MED         4,86.00         0.00           Rankap         Mass         4         2/12         4,88.00         0.00	0         0.00         That Perty Segment         0.07 EPGKMED         0.08           2         [PINDEXMO D1]         2.00.00         That Perty Segment         3.24.00         0.08           0         0.00         That Perty Segment         3.36.12         0.08
NR         TATING         EXAMPLE         ALIGN         ALIGNM           NF         TATING         SCHOOL         SCHOOL         ALIGNM           NR         TATING         SCHOOL         SCHOOL         ALIGNM           NR         TATING         SCHOOL         SCHOOL         ALIGNM	9         No. 2014         Control Contented Control Contented Control Control Contented Cont		Image: Section of the sectio
80 744011 PECIDINI2 ANDO MACHO 30 52002 FAVERDY JANES ACTV 31 52002 FAVERDY JANES ACTV	PT Research ANTINEAR PROFESSION AND ADDRESS IN CONTROL OF A     PACE RESEARCH AND ADDRESS IN A CONTROL AND ADDRESS AND AD	Javaidal         Pens         81         6.75         4,8000         6.30           Javaidal         Mill seguring         70         NOT TAVGMARD         4,8000         6.30           Javaidal         Mill seguring         70         NOT TAVGMARD         4,8000         6.30	0
512 551455 SPETERIA2 2146964 4456755 513 55757 REGEDENCE 2246700 4446970 514 552158 REG2000 22597*4 441****	SP         Long Campy, Long J, Gauger, Campy, C	Jacobil         Mar	0         0.00         That Pays Support         0.01         0.00           0         0.00         That Pays Support         3,316.6         0.03           2         pmSESM3 Cit         2,000.00         One Paysing         3,316.6         0.03           0         0.00         That Pays Support         N,1A         0.03           0         0.00         That Pays Support         N,2A         0.03
KIS KISHK KPRYARIS JIANSI 445168     KISHI     KISH	SP         Jan Pado         BUA (SAO GUELHERM), 23         O           SP         Jan Pado         Rea Chen Guene, 3/3"         O           SP         Mar Pado         Rea Chen Guene, 3/3"         O           BI         Volmes         AV. NEIO PECADEL, 301, 1021, AV. AV. PECADEL, 302, 434         O	Java Schl         Add supporting         All         DOT         L(A000)         G30           Java Schl         Post         43         3.83         4,0000         G30           Barchight         Mast         23         6.56         4,0000         G30	0
35         A1306         SPELNOS:         24070           36         A1307         E782.812         21080         42086           37         MMA         E782.812         21080         42086           37         MMA         E782.812         21080         42086           37         MMA         E782.812         21080         42086	Mr.         K. Diss. Reduces: M.         G           HI         Torework         Res. Genetic 600, Procession         H           GL         Comm.         Per Workshitz handles VM. Frame         H	Invested         Mail         4.44         Latence         0.00           Review         Mast         10         2.13         Latence         0.00           Densidal         Master         30         2.73         Latence         0.00	0         0.0         Darksport         N/A         0.0           0         0.0         Staff for former         L02701         0.0           0         0.0         Staff for former         MORE         0.0           0         0.0         Staff for former         MORE         0.0           0         0.0         Staff for former         N/A         0.0           0         0.0         Staff for former         N/A         0.0           10         Staff for former         N/A         0.0
N2         635400         1555000         1555000         1555000           S2         635400         2555000         1555000         441000           S2         635600         MCBETHER         1555000         441000           S2         635600         MCBETHER         1555000         441000	MO         Person Div Mass         Div         Div <thdiv< th=""> <thdiv< th=""> <t< td=""><td>Invested         Kat segmenting         71         11.2         4,0000         0.00           Banchay         Mass         (1+1+1+1+1+1)         MOT ENCEMENT         4,0000         0.00           Nordal         Minimumstar         30         MOT ENCEMENT         4,0000         0.00</td><td>2         [Ph005650 D]         2,000,00         Third Furly Neurope         64112         0.00           0         0.00         Third Furly Neurope         3,225,11         0,00           0         0.00         Nucleosed         www.reservemente        </td></t<></thdiv<></thdiv<>	Invested         Kat segmenting         71         11.2         4,0000         0.00           Banchay         Mass         (1+1+1+1+1+1)         MOT ENCEMENT         4,0000         0.00           Nordal         Minimumstar         30         MOT ENCEMENT         4,0000         0.00	2         [Ph005650 D]         2,000,00         Third Furly Neurope         64112         0.00           0         0.00         Third Furly Neurope         3,225,11         0,00           0         0.00         Nucleosed         www.reservemente
NI         60003         PENKLWC27         23.42243         31.9988           321         66003         PENKLWC27         23.42243         31.9988           323         662347         ROPARINES         36.03775         31.1646           321         662713         ROLANDES         Non-Yes         31.1646	PE         Manages         Bas New Aless Manages         Control         Contro         Contro	Interface         Interface <thinterface< th="">         Interface         <thinterface< th="">         Interface         Interface</thinterface<></thinterface<>	0         -         0.00         Noi falmed         NOTD/VGRMID         0.00           0         -         0.00         One Property         N_LA         0.00           0         -         0.00         One Property         N_LA         0.00           0         -         0.00         One Property         N_LA         0.00           1         JPREE005 C02         1,23.00         Thailt pro-Speerty         NOT SPECKRMID         0.00
127 TATIN PECCUAR ANNO ADDRESS     127 TATIN PECCUAR ANNO     127 TATIN PECCUAR     127 TATIN PECCUARA     127 TATIN     12	Reside         Restances and PEREADS, (N)         D           FP         Reside         Restances and PEREADS, (N)         D           SP         Compton         An Reside of Supers, 13163, (addine Gamadrass, Comptons, 1207)         D           SP         Comptons         An Reside of Supers, 13163, (addine Gamadrass, Comptons, 1207)         D	Josephy         Josephy         Auto         Aplition         O.00           Monig         Marg         11         3.24         6,400.09         0.00           Bordog         Marg         (3+)+3.75         NOT DVGMMED         6,400.09         0.00           Margin         Margin         (3+)+3.75         NOT DVGMMED         6,400.09         0.00	0         0.00         That Providence         0.00
	Vannap Rachord du Mar Figuratada, 123 Essanda , Carangi J 127 Raduam Racjula Marana, 213 Essanda , Carangi J 137 Kas Paula Di TEADA DO ALEAD , 00 1	App         (*******)         435         4,4860         6.00           Review         Mor         0.40         4.00         0.00           Review         Mor         0.41         1.17         0.00	0         .00         That Pay Sparsy         1,017         0.01           0         .00         That Pay Sparsy         1,017         0.01           0         .00         That Pay Sparsy         0,01         0.01           0         .00         That Pay Sparsy         0,01         0.01           0         .00         That Pay Sparsy         0,014         0.01           1         IPDECEDS CDI         12,020         That Pay Sparsy         NOT IPDECEDSID         0.01
NU         7141444         NPRECIMIT         2117044         4750187           NU         7141844         NPEXAME         2337109         44.32722           NU         7141844         NPEXAME         2337109         44.32722           NU         7141844         NPEXAME         2431103         44.42833	NP         Referent Press         REALANDERED DESERTINGSEL         11         1           3P         Service         Real Versite Conduct 11: Experiment         1           3P         Press Conduct         Real Versite Conduct 11: Experiment         1           3P         Press Conduct         Real Neargence, 10: 21 Non Conduct 2010         1	Bardon         Mass         4         647         6480         000           Kardon         3         NOT DVGMMD         6,0100         0.00           Kardon         Mass         9,0110         6,0100         0.00	1         personant Leg         1.23.0.00         Theol Page Prepare         NOT DAVEABLE         0.00           6         0.00         Theol Page Prepare         11,427.07         0.00           6         0.00         Theol Page Prepare         11,427.07         0.00           0         0.00         Theol Page Prepare         0.00         0.00
X0         T41997         X923X800         Z33603         44.52800           X0         T42566         X97854814         Z33301         46.65424           X1         T42662         X9785481996         Z333027         46.65424	MP         Journ         Ar. Tachagen Lais, 4° 88         1           MP         Jan Pado         Res County, 712, Bans Faula         1           MP         Mar Pado         E. Des Neyls Arguments from the fault structure, 30°         1	Kankap         Mass         ()+1)         (),77+0,21)         4,80.00         0.00           Kankap         Mass         ()+1-6)         1.22         4,80.00         0.00           Kankap         Mass         ()+1-6)         1.22         4,80.00         0.00	0         .         0.00         Third Fary Preparty         QULA         0.00           0         .         0.00         Third Fary Preparty         20/21/0         0.00           0         .         0.00         Third Fary Preparty         20/21/0         0.00
59 74300 MCCON017 .55226 .459276     57 74394 MCCON017 .5510% .55310     50 74394 MCCON017 .5510% .55310     50 74307 MCNC01018 .2259403 .4511	MG Codes BODOWLAGGORGON 08 - ABBOPORTO INTERNACIONAL MG Uses To Mess BEADOWLAGGORGON 08 - ABBOPORTO INTERNACIONAL BI Steers BacQuee A Sciencifica (19) - Intel	Rankap         Mass         (1,0+1,0+1,0+1,0)         NOT DVGRAMED         4,00:00         0.00           Sareded         Mdf appening         N0         N00T DVGRAMED         4,00:00         0.00           Sareded         Mdf         (1+1,0)         (1+2,0)         4,00:00         0.00	0
34 74903 37030540 233296 473331 34 74906 7032000 34355 349625 34 74906 7032000 2209***	5P Compton EEA BARLO DE JADUNEA, NO. CONCEÇÃO E PO Revie E Social DE JADUNEA, NO. CONCEÇÃO E 20 Revie Color. E Des Man Leises, NO. No. Color. 20	Revelop Tabele Mass 4 11.81 4,8500 0.81 Revelop Pes M NOTINGEMED 4,8500 0.81 Revelop Not Review 0.00	0         0.00         Thail Pary Report         NDT DPCRMBD         0.00           0         0.00         Thail Pary Report         3/02.05         0.00           0         0.00         Thail Pary Report         3/02.05         0.00
340         7.64034         347.020302         2.117326         44.03782           340         7.64044         379.73035         2.336721         44.43782           340         7.64042         379.73035         2.336721         44.43782           340         7.14042         379.737434         3.34772         44.43783	AP         Coundres         A Sin Print, 92         A           AP         Am Paris         Am Instant, 92         A           AP         Am Paris         Am Instant, 92         A           AP         Am Paris         Am Instant, 92         A	Bardon         Anno         8-14-0         107         4.0000         0.00           Bardon         Max         0-16         2.02         4.0000         0.00           Bardon         Max         0-16         2.400         4.000         0.00           Davidu         0-16         2.400         4.000         0.00	0         Safe Pary Negary         Cat220         Safe           0         -         Safe Pary Negary         Cat220         Safe           0         -         Safe Pary Negary         LVEX10         Safe           0         -         Safe Pary Negary         LVEX10         Safe
1400 07000 0000 0000 0000     14000     14000 07000     1400	Bar Data         Bar Append DI, Pindua         O           5P         Jan Pada         Bar Append D, Pindua         O           6D0         Ampede         VBLA VPAD - DIVERTING ADD DEDUCTING DATA (DA ADA/DAL) - ANA/ROLD         O           6F         Jandem         VBLA VPAD - DIVERTING ADD DEDUCTING DATA (DA ADA/DAL) - ANA/ROLD         O	report         n         Add         splitter         OB           Mark         (P+1+7)         2.00         4.04.00         0.03           Termfold         Mark         (P+1+7)         2.00         4.04.00         0.03           Termfold         Mark         (P+1+7)         4.00         0.03         0.03           Termfold         Mark         (P+1+7)         4.00         0.03         0.03           Termfold         Mark         (P+1+7)         4.00         0.03         0.03	0         .         0.00         That Fasy Reparty         4,1730.0         0.03           0         .         0.00         That Fasy Reparty         7,220.53         0.03           1         [PINDENSG C0]         1,233.00         Non-Internet         NOT SPECIARD         0.03           0         .         .         Non-Internet         NOT SPECIARD         0.03           0         .         .         .         .         .         .
			ost         ressmand         NDT DPCMARD         0.00           0         0.00         That Prop Papers         MUL         0.01           0         0.00         That Prop Papers         MUL         0.01           0         0.00         Not Internet         NUT DPCEMED         0.01
ad         7147944         BOCKEVC200         1655001         455225           303         714724         PEROVEN         458900         425600           304         714725         PEROVEN         458900         425600           304         714725         SECOND         2358627         459734	9         Kerleh         Barley Marine	-enness are segmente i No NOT PHOMAD 4,000 0.00 Investid Ref. gegening N 22.22 4,000 0.00 Raving Nas (1+1+1) 2.22 4,000 0.00	v . 0.00 Ow Peper N/A 0.01 0 . 0.00 Re Manuel NOT PREMADO 0.01 0 . 0.00 Road Pary Hupper NOT DECEMBED 0.03
300         747709         PERCORD         A5704         349903           301         24773         PERCENC         A64996         345796           301         21273         PERCENC         A64996         345796           301         21202         RESERVED         361394         311984	PI         Reside         EUA SIJ TALENDO MA (JED), 133           PI         Reside         Ena View (Daster de Jeures, 109           RN         Press, Alexer         AL-MIDI AL 1, STREAL TRUTTOR LL, 103	No.         No. <td>J         J</td>	J         J
MR         TANDAL         MEP/20012         ZZM001         455587           MR         EC0003         FPEN/100         51403         455997           MR         E2000         FMEP2014         51403         3007997           MR         E2000         FMEP2014         460023         3007997	Bit         Venezaria         DEFENSIVE ARGENCE DEL PENANS, 4/10°         VARIO (BASS)         D           FF         Vela Noves De You         Ar Canadi, 4/10°         Ar Canadi, 4/10°         D           FA         Journal         BASS ARGENCE DE MINESTA, MORTANA         D	Line         Da         D	0         .0.0         .0.0         .0.0         .0.0         .0.0           1         PR0505102         U.S.Mo         Prointed from of Charge         N.(A)         .0.0           0         .0.00         Non-indexed         NPT050500D         .0.0         .0.0           0         .0.00         Non-indexed         NPT050500D         .0.0         .0.0           0         .0.00         Non-indexed         NPT050500D         .0.0         .0.0
M2 51040 SPBPZ3N1 2133082 4440706 M2 511177 R[180:0027 2234206 4311667 M3 51500 R[C206581 2234299 433.5mm	MP         Jan Fauls         East Court, 200         Dist           E1         Ein Di Jawim         Ein 554, John Court, Ver Frunktwick Allen John), Uma Ein de Jawim         Ein 554, John Court, 200         Ein 5	Jacobil         Add segment         0.0         0.00           Bardog         Mast         3         1.52         4,0000         0.00           Bardog         Mast         3         1.52         4,0000         0.00           Bardog         Mast         3         2.27         4,0000         0.00	0
Mil M3182 EPROBE 223797 414488 Mil M3282 EPROBE 223797 414488 Mil M3282 EPROBE 2388 413291	Bit         Bits	Earling         Mass         (j = 4+9)         (j = 7+6)         (j =	0         0.00         0.00         0.01           1         [PINDENMI CI]         1,233.00         One Property         N/A         0.03           0         0.00         Table Pays         N/A         0.03
54" 62001 TOPRED 161776 463001 341 62001 TOPRED 161776 463007 340 60001 MOTORNO 161773 463021	TO Pales AVINEDATO A, LT 91 - PLANDBITO MEL 10 500 Pare Do Mars Bit Annuel BUASAVINE MARA DO CARACO AN 0	Bankoy         Mast         1         NOTIFICAMED         4,0000         0.00           Terminid         Port         40         225         4,0000         0.00           Terminid         Port         40         225         4,0000         0.00	0         100         Data Name         N/A         0.0           1         passion of U         Use Name         N/A         0.0           2         passion of U         San Amer         N/A         0.0           3         passion of U         San Amer         Not Name         0.0           4         passion of U         San Amer         Not Name         0.0           5         passion of U         San Amer         Not Name         0.0           6         passion of U         San Amer         Not Name         0.0
20000         213/006         311/007           375         60000         317/03/00         33/0204         36/0200           371         60020         30/0710         31/16/2         32/0714	Arrenews Reviews, R. Tarabass, H. J. M.T. Caulus AV. URT SERIES OF DE MINDONCA, IAM. J. RO. Phys. Ar. Distribution of Almedia, 1992. Acad. G.	Instrument         IA         M2         L48/00         0.00           Bendra         Dopl         11         2/04         L48/00         0.00           Invested         Molegnering         10         2/04         L48/00         0.00	Image: Constraint of the National MIC POSIBID         Image:
	mean         SQLSPERLOOD  : Nº S/N         F           101         Carbonin De Depression         EUX (NOE) - NEX (NOE) (SQLS)         F           101         Vision         EUX (NOE) - NEX (NOE) (SQLS)         F           102         Vision         AV. NOBAL SE/NEX/BLA DA FFN.HLL/103         F	arrenzy Jany 8 412 4,000 0.00 Bendag Mas 6 2.04 4,000 0.0 Bendag Mas 8 NOT DPGAMED 4,000 0.0	Image         Annue         NOT DeCAMED         0.0           0         0.0         Nall Pay Negaty         1,257.57         0.0           0         0.00         Tablel Pay Negaty         3,555.55         0.00
	see puin service EU.A.C.LEEEE.PARCEN, 521     Extended State Control (Control (Contro) (Control (Control (Control (Control (Contro) (Control (Control (	Aver         NOT INCRMED         4,0107         0.00           Bendrag         Most         7         NOT INFORMED         4,0107         0.01           Bendrag         Most         7         NOT INFORMED         4,0107         0.01	-         -         -         -         0.01           0         -         0.07         The Proof Proof         0.03           1         [PR0EXNL02]         1,33.00         The Proof Proof Proof         0.03
A         A		No.         No. <td></td>	
od         748701         RUBOINS         AUTOR         AUTOR           M2         741102         87287980         233263         444623           M0         74197         87923900         233263         444623           M0         74197         87923900         233263         444623	re, memor Des Converses Accessibilities Mar. 301 J 37 Jan Paulo Barrowski (2013) Federal (2013) Federal 1 39 Jan Der Convers Barr Christian J	anna Ani 3 27 4400 00 Backa Mai Delete A. Articlate 2012 4400 00 Reales Mai 1 1011010 400 00	0         Bail Pay Papage         NOT DOCIMID         0.01           0         Bail Pay Papage         7.77,0.0         0.01           1         [PADESSI C0]         1,35,00         Bail Pay Papage         4.716,1         0.01
Mil         Tel210         MPEDIAGIZ         2337987         4433952           56         Tel2123         MPETERIS         2347801         4438964           56         Tel2124         MPETERIS         2347801         4439864           56         Tel2124         MPETERIS         2349101         4447988	Neuron BECALACERDA FRANCIA Conference Ted Neurone, Baires Aparenda UT, CEPENDENDE     Neurone, Ne	Name         Data         (1+1)         B. SUP PSP/MED         (Annu )         (B-1)	0
56" 71427% 50"0707840 235305 444203 568 714773 50(344642 225306) 452838	MP Nan Paulo An Creative do Sul, 1900 - Castelad B El Re De Jacoire An Note do Jacoire, 5/N <sup>o</sup>	Rening Mass (1+1+1+1) 33% 4,000 0.00 ieenidd Peu 30 3.0" 4,000 0.00	0         .         1.00         Thair Pary Report         NOT DPORMED         0.00           0         .         0.00         Thair Pary Report         NOT SPORMED         0.00

Geskoolas	Addava	Kaunar	Bar Ive Bair Ermannei	tion Additional Resourcesion Due of Due of Conclusion	Ban Haring	Paulough	Revenue - Storver of Regulationics Com
8 Krywske Nor ID Latinski (devinu) Langinske (devinu) 90 744142 C32LA0171 - 3.52298 - 3632963 90 744176 D495200128 - 1.637949 - 4745407	Alfreigheithe Alfreigheithealthealthealthealthealthealtheade Alfreigheitheatheade Alfreigheitheade Alfreigheitheade Alfreigheitheade Alfreigheitheade Alfreigheitheatheade Alfreigheithe	King Type         Massacrity         Height (m)           Resoluty         Mass         (3+2+2)           Convertical         Mill segmenting         60	VEX.1 Introduct.*         Linguid Rance Trace (100.1)           (1),278+4,3734+7,733         4,400.00           2,04         4,400.00	L/month (FEL/Alumity # of Open 0.00 0 0.01 0	urors Clime(s) Massis) Lig Ansano (BEL Mass 0.00 0.00	Dial Faity Report Table Faity Report Table Faity Report	Liquid Reve (IREL/Mandt) Invessor of Regulationism Cara Transfer (IREL/Mandt) 4,05331 050 NOT D4*COMID 0.01
NR         T44252         D195.0015         13.41397         47.9928           NR         T4400         D195.0012         13.0199         47.9928           NR         T4400         D195.0012         13.0199         47.9928           NR         T4400         D195.0012         13.0199         47.9226	DP         Each         Markan Back (Star MC), Star M 2013,           0         Back (Star MC), Star M 2013,         Star M 2013,           0         Back (Star MC), Star M 2013,         Star M 2013,           0         Back (Star MC), Star M 2013,         Star M 2013,           0         Back (Star MC), Star M 2013,         Star M 2013,           0         Back (Star MC), Star M 2013,         Star M 2013,           0         Back (Star MC), Star M 2013,         Star M 2013,           0         Back (Star MC), Star M 2013,         Star M 2013,           0         Back (Star M 2014),         Star M 2013,           0         Back (Star M 2014),         Star M 2014,           0         Back (Star M 2014),         Star	Keelee Mast 3 Keelee Tabala Mass 7 Gweeled Millane 75.	179 44000 141 64000 1438 44000	0.31 0 0.39 1 0.39 0	0.00 (PRINDENG CR) 1,285.00 0.00	One Property Third Farty Property Not Informed	N/3 0.81 NOT DEVERSIBLE 0.81 NOT DEVERSIBLE 0.81
NN         7142943         GCN25,5008         1647158         4526054           RN         7143147         GCN25,50082         1646290         4530120	DO Goussa Bai 133 Dopase Con 133, 199 - 57 Arroyom: OD DO Goussa Ba A BENALDERD 23, 2014, 254, 320	Reachop Mass (1+2) Reachop Mass 8	3.331 4,88.00 4.488 4,88.00	0.30 0 0.30 0	04.0 04.0	Thind Party Preparty Thind Party Preparty Thind Party Preparty	3,0000 0.0 2,78600 0.0
NI         TUTING         ROCALDITI         STURIO         STURIO           RI         TUTING         ROFALINEZI         2117103         3111579           RI         TUTICZ         ROFALINEZI         2310703         3111579           RI         TUTICZ         ROFALINEZI         2330901         4679438	and Marchanne Marchannes (Constant Period, Constant Period, Constant Period, Constant Period, 2011) BS Preven Alogov Ar. Freeman Format (2011) 3P Jaconi Res Callerin Maning, 201, Vila Pomera, Jaconi - 3P	Koolog Pos (1+1+1) Koolog Mas (1+1+1)	NOT INFORMED 4,480.00 2.433 4,480.00	0.00 0 0.00 0 0.00 0	. 0.00 . 0.00	Thind Party Property Thind Party Property	2,0160 0.00 NOT DEVEMBED 0.00 2,01166 0.00
889 7147530 DH8550729 13.75676 4736756 689 7147561 B075408077 5664878 51.25671 697 714159 507010015 7147579 666460	DF Book OPLINADA DOS HINDETEROS, BLOCO B, SALA IF. MINIETERO DAS COMUNICAÇÃ RES Press Algor MINIDA (VAL PROVINCIA) (VAL PROVINCIA) RES ANDRES A TURAS A TURAS A	E Kening Mast 3 Kening Mast 4 Kening Mast (Sabba V	3.3 4,86.00 3.25 4,86.00	0.30 1 0.30 0 0.30 0	1235.00 1,235.00 0.00 0.00	Not Informed Not Informed Third East- Research	NOT DPORMED 0.00 NOT DPORMED 0.00 KN755 0.00
642 64717% 3623/A38020 21.4110% 44.8284 647 656072 MERCONT20 21.53563 43.2066	MG From BUA MANTA EPIPERIA, S/N MG Guine BUA PANO DE MATZA EBERRO, 0	Gwonleid Xull segmenting 40 Gwonleid Xull segmenting 52	NOT INFORMED 4,8100 NOT INFORMED 4,8100	0.31 0 0.31 0	. 0.00 . 0.00	Osa Property Percisian Perce of Charge	N/A 0.00 N/A 0.00
	mt. Unrelandis BUA VIRCELED CARDENO, 690 MC. Promoto Das Vides BUA RULEA, 798 MC. Concello Rev. BUA KULANDA N. CARDENDA, 225	Interface         Total         Total         Total           Marce         Marce         Marce         Marce           Marce <t< td=""><td></td><td>0.00 1 0.00 0 0.00 n</td><td>protecti (02) 1,285.00 0.00 0.00</td><td></td><td></td></t<>		0.00 1 0.00 0 0.00 n	protecti (02) 1,285.00 0.00 0.00		
647 647533 MC033A3723 .5552338 .4752436 648 754553 R07A0182 .565848 .3124950	MG Ubeede AVENDA FELOMENA CAREADINA KM 17 RS Ports Aligne AVENDA DUMEO DE NOTETAA, NB	Generated Mell-segmenting 90 Reachery Mass 4	33.07 4,80.00 NOT INFORMED 4,80.00	0.31 0 0.31 0	. 0.00 . 0.00	Third Farty Property Not Informed	2/5%44 0.00 NDT D9506MBD 0.00
633         7400.73         Reference         34.17.37           634         6386.82         MG/12, Nucler         34.84210         46.77333           641         7146803         Reference         36.00123         31.26397	and Print Oppy Link American Stream (11) Mill Guinnaweis BEU, TEPAABAA, 101 BS Press Mager BEU PERA DO CLUBE DOS (ANGADEBOS	Georgenia Anti-appending 52 Georgenia Xell appending 52 Georgenia Xell appending 24	NOT INFORMED 4,8100 NOT INFORMED 4,8100	0.00 0 0.00 0 0.00 0	. 0.00 . 0.00	Own Property Not Informed	N/1 DPORMED 0.00 N/3 0.00 NDT DPORMED 0.00
612 63600 MGCA408 154720 418490     613 66064 3602834438 154720 412800     614 71428     MC087004 201500 412800     614	MG Option Andreader ATTNEDA LEATINGDO DEGA, S/N MG Memer Forenesses MCOBEO DO CEMPTOREO MG Dane Press AVENDA VIER E AVAIOUTED DEURS TOLEPOL S/N	Greenfeld Addrogening 30 Greenfeld Mill-segming 60 Greenfeld Prot 62	NOT INFORMED 4,8100 NOT INFORMED 4,8100 NOT INFORMED 4,8100	0.00 0 0.00 0 0.00 0	. 0.00 . 0.00	Provision Poor of Charge Provision Poor of Charge Not Informed	N/A 0.00 N/A 0.00 NOT DEPENDENT 0.00
615 7540379 R07530047 .56.59500 .51.5250 616 .09792 MOC322525 .564175 .621081	RS Press Mager EXTRADA COMPLETION SCI., 001 2001 Consul Mass REAL VEHICIES APARTMENT AND STREET	Received New 4 Generalist Pros 80	NOT INFORMED 4,811.00 NOT INFORMED 4,811.00	0.81 0 0.81 0 0.81 0	. 0.00 . 0.00	Thini Party Preparty One Preparty	3,0000 0.0 N/3, 0.0
47 4427 M12/403 31/729 10000 48 71/202 8024031 20900 31.009 49 71/202 8714000 20000 0000	MC. Including Control	Greenhold Prom 12 Norther Mont 4 Greenhold Molt appending 40	NOT INVESTIGATION ( 4.4000) NOT INVESTIGATIO	0.80	1.00	Niti Islamed Osa Prepriy	N/A 000 N/A 000 N/A 000 N/A 000 N/A 000
420 714440 800903333 33566 4336401 420 65514 560379363 358678 440065	BO Pose Toles AV. GOV. (BBGE TREDUBLA 429 MG Tepes PBACA (DB FUELS CARDON S/N*	Gworleld XdEseppotag 50 Gworleld Post 30	NOT INFORMED 4,811.00 NOT INFORMED 4,811.00	0.30 1 0.30 0	PRINDENCI OR 1,255.00 0.00	Third Party Preparty Not Informed	LOSE 0.00 NOT INFORMATION 0.00
622 64535 3626425300 .266980 .415620 623 64265 MGADMITT .556890 .425680 624 642967 MGDM424 .1654296 .425896	MG New Law BCIEDVIA [CICLENCE AUROLATER / HL 440, KM 440, 5/N MG Annue Das BEA BLAD HORSONTE, 5/N MG Journey in MCBED DAS AVERTICAS, 5/N <sup>4</sup>	Reselley Mass 4 Greenfeld Roll segmeting 43 Greenfeld Rell segmeting 63	NOT INFORMED 4,8100 NOT INFORMED 4,8100 NOT INFORMED 4,8100	0.30 0 0.31 0 0.31 0	. 0.00 . 0.00 . 0.00	Thind Party Property Not Informed Not Informed	0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.0
425 7143555 52MC0424875 155486 455221 426 87875 MCPC01058 2224872 445501 477 19977 (1975)155 75670 1511000	MG Bele Honoren BEAADCNO PENA, 480 MG Pouer-Sin BEAADCNO PENA, 480 MG Pouer-Sin BEADCNO PENA VE EXTREMENTATION OF THE AND ADDRESS OF THE	Reselve Mass (3+3+3+3+3+3+3+3) Gwordeld Relingensing 40 Consolid Tell summing 40	NOT INFORMED 4,8100 NOT INFORMED 4,8100	0 0000	. 0.00 . 0.00	One Property Not Informed	N/A 0.00 NOT DOCIMIED 0.00 NO1 0.00
429 64077 MURNI488 154255 422275 429 64922 MURNI428 151772 418441	MG Been Foles ALTO DO MOREO DA 1973; AO TELINAR MG Instein W Alto CERTOR IO S/N	Georelicid RelForgening 30 Georelicid RelForgening 40	NOT INFORMED 4,810.00 NOT INFORMED 4,810.00	0.8 0 0.8 0	. 0.00	Not Informed Third Party Preparty	NOT DO CAMID 0.00 6702 0.00
631 04407 360370467 053480 421430     631 04107 360342.9998 1523480 463730     67 74499 8602977 75499     74499	MG Isa Isan De Cherne EUA ERPEANCA S/N MG Undersela BELA PEREZ ANGA N° C M Managando BELA PEREZ ANGA N° C	Greenield Mell segmenting 52 Greenield Kell segmenting 40 Random Mari	NOT INFORMED 4.00.00 NOT INFORMED 4.00.00	0.00 0.	. 0.00 . 0.00	Not followed Not followed Third Party Preserve	NOT DPCRMID 0.0 NOT DPCRMID 0.0 (2013) 0.0
633 7143148 EXXBC/1760 21.44250 46X5300 638 7146981 8CC/921117 26.55260 44X95390	ES Manuares EU-A MARIA I KORTE, 5/N SC Gagar BUA DORALICO GARCIA, 500	George Xd appening 35 Reading Mast (10+10+10+7+7)	- 4,88.00 - 4,88.00	0.30 0 0.30 0	. 0.00 . 0.00	Not Informed Not Informed	NOT DPCRAMD 0.00 NOT DPCRAMD 0.00
635 637568 MGMPL4335 186880 41.3556     60586 MGPD1648 155680 41.3556     67586 MGPD1648 155680 41.585     675 754858 SOCO802 2597420 3222043	MG Mender Presented BLA DA TOBBLE, VN MG Pres AV DOUTOR GETCHLO VARGAR, VII IC Organis AVENDA (FERNINDO MACHINO, 807	Gweekiel Pes 30 Gweekiel Pes 40 Reacher Mart (1+7)	. 4,4000 . 4,4000 . 4,4000	0.30 0 0.31 0 0.31 0	. 0.00 . 0.00 . 0.00	Not belowerd Not belowerd Not belowerd	NOT DECEMBED 0.00 NOT DECEMBED 0.00 NOT DECEMBED 0.00
439     40700     382327703020     2230070     431790     431790     4300996     4309998	Eir Die Jassier Eine Konsteinen Dassen, 108. 20 3020/W 20 andreen MC Reis Dississen Auguste de Linea, 1047. Freuen Laforeie	Indoor Company Building	. 14,405.00 . 14,405.00	0.30 0 0.30 0	. 0.00 . 0.00	Not Located Third Party Property	NOT LOCATED 0.81 6,196,21 0.82 0.90 0.90 0.90 0.90
64         229(1)         SPENTAL (1)         22.4 (600)         46.00943           642         637         SPENTAL (2)         21.62750         46.00943	Kar June 10 memory interaction Control (2010) Subjects Contact and SP Saw Paulo W ROCKI PERECON INCOME 100 / 121     Saw Paulo //W ROCKI PERECON INCOME 100 / 121     SP Saw Paulo //W ROCKI PERECON INCOME 100 / 121	Jahor Argon .	1.000.00 1.000.00 1.000.00	0.31 0 0.31 0	. 0.00 . 0.00	Thind Party Property Thind Party Property Thind Party Property	Reference         -0           Units         -0
647 64279 2001213,00542 377720 3854085     648 9750 391002033 2314000 4454070     691 976 (3977 9510) 4010707	CE Persiles In Newton Cales Invision 2000     SP Gaurdies Ed. Persilem Date, Kor 2011, Immunical Neuropean Clauselles     Persilem Persilements in American International Action Parket	Index Argent	. 16,400.00 . 16,000.00	0.00 0.00 0.00 0.00	. 0.00 . 0.00	Thind Party Property Thind Party Property Thind Party Property	11.06132 0.00 12,290,25 0.00 177,201,37 0.00
647 1762 MED/07/128 121100 3515022 647 1762 MED/07128 121100 431600	All Constants     Konstants     Konstan	Takor Arpan Takor Arpan	- 14,805.90 - 14,805.90	0.00 0 0.00 0	. 0.00 . 0.00	Third Party Property Third Party Property	4002570 0.00 1,0025 0.00 1,0025 0.00
64 2499 DPPCH132 33A300 460031     68 506 3PB034271 234500 460031     68 506 3PB034271 234500 460030     69 5PB13507 233820 475973	DF Bendis COR7_Loss.1(4_3lassah) Nepping 3P Bases Al-Ban Al-Ban Nepping 3P Marka E. do Tamarah Mit-America	lalor Negrig . lalor Negrig . lalor Nerrise .	. 14,855.91 . 15,455.91	0.00 0 0.00 0 0.00 0	. 0.00 . 0.00	Third Party Property Third Party Property Third Party Property	72423 0.00 979431 0.00 749745 0.00
682 663 5740/2981 2122280 3642001 682 28798 3820250835 366526 312809	3P Jacobia Av Jaapan Pereges de Talalo, 011 Anyania Negring     85 Press Alagor     PRACADA MATRIX N	baloor Shopping . baloor Pable Balding .	. 14,455.90 . 14,455.90	0.00 0 0.00 0	. 0.00 . 0.00	Third Party Property Not Located	0,033.7 0.00 NOT LOCATED 0.00
631         1754         360(21)(311         22.0730         31.0810           634         2757         30(21)(24)         22.3420         450(20)           635         2777         30(20)(21)         30(30)(20)         31.2780	RJ International Antonio Constanting States (Science Science) (Science Science	Delay Nervin	11,405.0	0.00 0	. 0.00 0.00	Think Party Property Think Party Property	2,00.05 726,00 2,00.05 2,00.05 0,00 0,00
636 1772* 303423072,4773 .153740 .435403 63* 13856 302223744298 .1554380 .4469991	MG Rels Hosterow Realersis RE. 306, 3047 - HE Meyping MG Contegers Aronial (size Cost de Oliveire, 1273 - Bg Weyping (3220).200)	Index Mapping . Index Mapping .	. 14,405.00 . 14,405.00	0.31 0 0.31 0	. 0.00 . 0.00	Third Party Property Third Party Property	8,135% 0.0 T/1487 0.0
68 660 5PE21047 214540 447983     69 1667 R0672304 2234790 441630     68 684 D09201724 157350 473974	37 San Pasis     K. Bolos Gan, W. Jino Yan Mergeng     K. Bolos Gan, W. Jino Yan Mergeng     K. Bolos Pasis de Bossing, 80: Reading Pasis Respire     Beslin     MIDR THREN M. NORER, ONE NOT NEED MODIFIED	ladour Hosping Iadour Hosping . Iadour Hosping .	. 11,425.00 . 11,425.00 . 11,425.00	0.30 0 0.31 0 0.31 0	. 0.00 . 0.00 . 0.00	Think Party Property Think Party Property	7,000 0.00 17,7569 0.00 4,0048 0.00
60     617     599280252     334280     46.0890     462     2493     25952813     15757     15757     1575     1575     1575     1575      157      157      157      15      1	3P Sar-Paulo Ran Tosicol, 2000 DF Brodin QD SCN Q 58 L A, 0 - AAA NORTH 5P Nov Young W Instances With Descention	lalor Shyping . Ialor Shyping .	. 11,005.00 . 11,005.00 . 14,005.00	0.31 0 0.33 0 0.33 0	. 0.00 . 0.00	Thind Party Property Thind Party Property Thind Party Property	13,38841 0.00 7,264.75 0.00 500.64 0.00
644 0711 357(17)2332 2231290 4511750 648 27378 3863(25)250079 16,78120 451750	AP Guessingues As Jacobies Kaloneko & Olivers, NJ, Bacobies K	Sec.         Target Sec.         Target Sec.           Sec.         Sec.         Sec.	- 14,8000 - 14,8000 - 14,8000	0.00 0 0.00 0 0.00 0	· 0.00	Thial Fary Prepary Thial Pary Prepary	0.00 8),0754 0.08 9,251.70 0.09
696	DF Breaks Canana des Dersandes - Para des Teis Poders CB President AV DB TOMPOR BLE CAN DB B Rei De Ineries Deuter - Deuter Houses of	Indexer Public Bailling	- 11,405,51 - 11,405,51 - 11,405,51	0.8 0 0.8 0 0.8 0	·	Third Party Property No Located No Located	000 000 000 000 000 000 000 000 000 00
40 FM 37C3A244 223329 470700 40 FM 37C3A244 223329 470700 479 884 37C17247 23A220 44409 479 7754 984	27 Camputes E. Juny Transitio Campon, Not. Camputes Control 29 Camputeshin An. Joshow Camp Disk, 1997. Campute Description 29 London: 1997. Campute Description Control Disk, 1997. Campute Description 2017. Campute Description Control Disk, 2017. Campute Description 2017. Camp	Indexe Storying	- 11,405.01 - 11,405.01 - 11,405.01	0 0.00 0 0.00 0	. 0.00	Third Party Property Third Party Property Third Party Property	TL40500 0.00 2,000.00 0.00 2,000.00 0.00
n.v.         27204         PEENDOTA         233,8500         31,16611           672         56875         58225,925,009         22,5550         43,26600           671         5520         56225,925,009         22,55200         43,26600           671         5520         225,9250         43,564,009	rm. somänna BEDORTALISLÄG GABEGA GDB.S/N RJ Dauge De Comm. Bedrow Washington Lan, 2013. Canina Monyaing RJ Rie De Jacoin Journal Garomaine Down RE. Conin Monyaing	nacor Nopping . Indoor Nopping . Indoor Nopping .	- 14,000.00 - 14,000.00 - 14,000.00	0 860 0 860 0 860	. 0.00 . 0.00 . 0.00	Third Party Property No Located Third Party Property	1,005 0.00 0.00 0.00 0.00 0.00 0.00 0.00
	397 San Jose Des Campon Ar. Depando Rondin. Mannano, 1983. Constr. Val. Nepping 397 San Pado As. Dr. Francisco Mangain, 1983. Constr. Pad. Resping 202 Providen     40. Virtual VII. VII. VII. VII. VII. VII. VII. VII	Induse Stopping Induse Stopping Induse Operator Adulta	- 54,805.00 - 54,805.00	0.33 0 0.33 0 0.35 0	- 3.00 - 3.00	Third Party Property Third Party Property No.11	14,728,79 0.00 36,758,64 0.00 NOT LOCATHD
67         6417         3822037167         233800         611894           674         1675         382273367         233800         6118864	AV ANN ANNA STATUS (1970) RE Die Januar Die Status Die	Inform Entropy and an Information Street Str	- 14,800.00 - 14,800.00 - 14,800.00	0.00 0 0.00 0 0.00 0	. 0.00 . 0.00 . 0.00	No Local Marine Name Sharine Name Sharine Name Sharine Name Sharine Sharine Sharine Sharine	0.00 2,07541 0.00 9,7759 0.00
	Mr.         Bernell         Av Marce, P. & Chen Redenars, WF, ANTA.           Mr.         No. Letter Methods, WF, Contenant Worssner,           Mr.         Re. De laware         No. Methods, WF, Contenant Worssner,           Mr.         Re. De laware         No. Methods, WF, Contenant Worssner,	Jaloor CooperyBuilding Jaloor Norming Jaloor Public Building	11,40,50 11,40,50 11,40,50	0.33 0 0.33 0 0.39 0 0.39 -	3.00 	Rise Contenter W128 Third Party Property Not Leasted	EUT CONTEACT (is entimer's and LOBALI 0.00 NOT LOCATED 0.00
same         same <th< td=""><td>A Common Common Common Common Common Publics (2012)     Mol Biol Robuston Publics (2012)     Monada Chagos Marks (2012) - Dissoved Mail Networks     Report Do Common Common</td><td>lalor Nopping Jalor Nopping</td><td>- 11,405.00 - 11,405.00 - 11,405.00</td><td>0.00 0 0.00 0 0.00 0</td><td>- 0.00 - 0.00 - 0.00</td><td>Think Forty Property No Located Think Forty Transmission</td><td>0.00 8,0334 0.00 NOTLOCATED 0.00</td></th<>	A Common Common Common Common Common Publics (2012)     Mol Biol Robuston Publics (2012)     Monada Chagos Marks (2012) - Dissoved Mail Networks     Report Do Common	lalor Nopping Jalor Nopping	- 11,405.00 - 11,405.00 - 11,405.00	0.00 0 0.00 0 0.00 0	- 0.00 - 0.00 - 0.00	Think Forty Property No Located Think Forty Transmission	0.00 8,0334 0.00 NOTLOCATED 0.00
64         590         SPELSSE         22,2200         4797330           64         539         SPEC(2011)         23,336         474,450           66         6725         SPEV(C001)         23,319         474,450	Marka         Are Essensida, 201 - Essensida Plane Neupping           3P         Neuroda         Are Essenda Mangara Proc. 001 - Explanda Margarag           3P         Nan Padro         E-Lond Essenda Plane - Neuron           3P         Nan Padro         E-Lond Essenda Plane - Neuron	lalor Noppig . Ialor Noppig . Ialor Noppig .	- 16,405.00 - 16,405.00 - 16,405.00	0.00 0 0.00 0 0.00 -	. 0.00 . 0.00	Thiad Party Preparty Thiad Party Preparty Thiad Party Presents	3,00600 0.00 17,17.59 0.00 19,04.02 0.00
60° 75354 50°0021833 2342710 44.79277 688 8998 329542325 2554880 4734289	See Paulo     K. Auguinets (Here Redig of Compose), 501 - 125 meres     K. Auguinets (Here Redig of Compose), 501 - 127 meres     Area     Are	Indoor Company Building Indoor Shopping	- 11,0000 - 11,0000 - 11,0000	0 0.00 0.00 0.00	· 0.00	Thial Party Property Thial Party Property	342,141,82 0.08 347,743.6 0.08
on/         CN0         MPCAM403         Z236/36         C10220           630         8001         807/31260         21.12360         4607750           670         2010         MUTERADOR         21.0260         4607750           670         2010         MUTERADOR         11.036976         6407750	rer sampten Bedevic Dam Paders U. Ko. 101,3 - Galers Mergring 347 Canaders In: Jos Noless Maders, 200 - Gardee Conselers Mergring MT - Gales <u>XV</u> , 1 we with	nacor Negying . Indoor Negying . Jadoor Negying .	- 16,400.00 - 16,800.00 - 14,800.00	0 860 0 860 	. 0.00 . 0.00	Third Porty Property Third Porty Property Third Porty Property	April 0.00 300122 0.00 830302 0.00
07 503 525525 214475 443530 07 1011 32252 22850 433200	SP Inser Ander Archetered (00), Cread Dan Merrine (MC Flan Merrine)     Sector 10 (Control of the Archetered (10), Cread Dan Metrice (10), Control (10)	Induse Neuroine Induse Neurone	· · · · · · · · · · · · · · · · · · ·	0.00 0 0.00 0 0.00 0	· 3.00		2038457 0.08 25/96136 0.09
				0.38 0 0.38 0 0.39 0 0.39 0	. 0.00 . 0.00 . 0.00	Nucl Ray Paper Ma Loand Ma Loand Ma Loand Ma Loand Ma Loang Ma Loang Ma Loang Ma Loang Ma Loang Mang Ma Loang Mang	
	Kyrn Lynny Ar Gerennader Roberts du Manna, Mill - Lynnys Tray Margong Mar Carlon Provinsi der Handreynen, 2001. Spransen Miss Carlon Karl De Innien Mar De Innien Mar Carlon Martine Mark Marken	Inform Narping . Inform Narping . Inform Narping .	- 14,805.00 - 16,805.00	0.00 0 0.00 0 0.00 -	. 3.00 . 3.00	Third Party Property Third Party Property Third Party Property	14,03400 0.00 4,227.13 0.00 9.04.122 0.00
780         Mold         APP/L2004         ALLONG         ALLONG           781         Mold         APP/L2004         ALLONG         ALLONG           785         Mold         APP/L2004         ALLONG         ALLONG           786         Mold         APP/L2007         ALLONG         ALLONG	ng na ka panan Negera na Negera na Negera Ne	lakur Muştağı - İakur Muştağı - İakur Muştağı -	. 11,400.00 . 11,400.00	0.00 0 0.00 0	. 0.00 . 0.00	Third Party Property Third Party Property	4,26.57 0.80 4,235.57 0.80 3,333.55 0.80
712         1726         3000(77642         2237560         433881           711         5771         5070(23275)         2337560         4648745           711         5771         5070(23275)         2337560         4648745           711         5771         5070(23275)         2337560         4648745           711         5771         5070(23275)         2337560         50690(23275)	El En De Jacoin Avenida Endusiandor Alexando Facence, 540     Prens Gaussie AV AVETON MON MAN ALEXA, 550     Deser Deser Deserts FACENCE AVENUES, 550	lalor Raina . Ialor Neying .	. NU,000.00 . 11,005.00	0.31 0 0.31 0	. 0.00 . 0.00	Third Party Property Third Party Property Third Party Property	N(08100 0.00 9,2516 0.00 117540 0.00
709         2012         397802333         2542100         4640792           709         24123         3978021333         2542100         4640792           706         6420         3978021056         2542100         4640792	Per concernance de la concentración de la conc	Internet Annual Indexes	- 14,400.00 - 14,400.00 - 14,400.00	0.0 0 0	· 0.00	Third Party Property Third Party Property	10.1 (167 0.00) 17.1 (1641 0.00) 17.1 (1642 0.00) 17.1 (1642 0.00)
70"         3690         3754.11.236.1         .7.166.860         .46.460"0           708         3834         325.525.08         .21.5680         .46.0026           708         1980         36.555.02         .3.0028         .46.0026           709         1980         36.555.02         .3.0028         .600262	Mass An. Conversalar Marin Coron Jusine, IL - Manis Desping     Januar Anna Anna Anna Anna Anna Anna Anna An	lalor Neging . lalor Neging . lalor Nervice .	. 14,405.00 . 14,005.00 . 15,405.00	0.00 0.	. 0.00 . 0.00	Thind Party Property Thind Party Property Thind Party Property	11,181.37 0.00 4,014.02 0.00 9.270.06 0.00
710 1507 XEX422732014 1517500 433287 711 2440 EPPERITY 15A650 47A674	MG Rela Hosterour Aronala Contare Machado, 2001 - Mara Norperg DF Breaks Insurancy - Explanatic des Ministères, Bless H, Manatère din Erleptes Enteriores	baloor Shoping . baloor Pable Balang .	. 14,655.95 . 14,655.95	0.00 0 0.00 0 0.00 0	· 0.00 · 0.00	Thind Face Streams Thind Face Streams	x21540 0.80 NOT LOCATED 0.80
T12         MH         MPETAPCI         235640         44.5500           T13         MM         MPMETAPCI         2354100         44.1700           T14         64.04         MPMETMON         23.42720         46.07043	Meson Ar. Bankler di Carlo, 21. Maranar Negrogi,     Meg.Bas-Crans Are Ver Namin Tigge Consumito, 1011. Mag.Basying     San Pado Are EUCET IFTERION [COLOR, 100]	lalor Norping . Ialor Norping . Ialor Norping .	. 14,455.00 . 14,455.00 . 14,455.00	0.30 0 0.31 0 0.31 0	. 0.00 . 0.00 . 0.00	Third Party Property Third Party Property Third Party Property	13,00,21 0.00 4,00,66 0.00 32,342,53 0.00
713 4474 ENCANDE	RN Neal Avenile Kenales Iniged Film 2214 CE Prenders Ave Reserved & Menous no. 2010 Neath Neuraint	haloor Shopping	. 11,405.00 	0.8 0 0.8 0	, 0.00 , 0.00	Third Farty Property Third Farty Property Third Farty Property	12,92,80 0.80 X166,12 0.80 ******
TT         MR         MPEXAMIN         21.8728         AC7881           TB         MPEXAMIN         21.8728         40.7737           TB         MPE         21.8228         34.8734           TB         MPE         21.8228         34.8734	M     Morean Dees     Ar     An Deviation Knownin, 1931. Neuro Meessing     Me	Index Neurosc : Index Neurosc : Index Neurosc :	- 16,600 - 1	0.30 0 0.30 0 0.30 0 0.31 0	. 0.00 . 0.00	Thind Party Property Thind Party Property	
T21         T200         PERTAINS         ADDR         MATHIE           T21         25220         MITCLANCE         15.8790         MATHE           T21         25220         MITCLANCE         15.8790         MATHE           T21         25220         MITCLANCE         15.8790         MATHE	PE Basis Proge de Equilitées, 3/N     MT Guales M MARCINGA     MA MARCINE MARCHARD MARCHARD     MR De Insuis Researchard Manazia (2010). End Manazia Canada (2010).	balow Pole Balling . Dalow Narping .		0 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0	. 0.00 . 0.00	Datal Fariy Dopony Thiad Fariy Dopony Thiad Fariy Dopony Nat Located Datal Fariy Dopony Thiad Fariy Pospony	NOT LECATED 0.00 4/05/00 0.00 M 11537 0.00
T21         47%         23360A44XQ         NYCAX         2244080         4746150           T24         23474         PAAC02243         -3.37800         444609	SP Camping Av. Galleener Gampes, ND: Pagar Dom Pales Nepping     PA Below Realistic Augusts Monitoring, USE (Pagar Merging Relin)	baloor Norping	14,405.00 . 11,405.00	0.30 0 0.30 0	. 0.00 . 0.00	Think Party Property Think Party Property Think Party Property Think Party Property Think Party Property Think Party Property	2008.122 0.00 2008.14 0.00
724         N18         3PTF0.2001         22.12300         31.58540           726         7227         31.36122797         3.33589         33.7448           727         9007         90234102028         2.299770         38.44738	3P President Poulous Bin Squara Congrey, 1143 - Pouge Neeping Doubone     3J. Massin V MENDAG MARCHAO, 5000 TARCHAOM MATTNE     40 Advance ANTINDA ANTINO CARLOS MARCHAOM IN PUT INA PROCESS CONTROL	lalor Reying	- 14,8558 - 14,8558 - 14,8558	0.31 0 0.31 0 0.31 0	. 0.00 . 0.00 . 0.00	Third Party Property Third Party Property Third Party Property	4,211.11 0.89 13,273.75 0.89 T001.35 0.89
728 0/728 EXCAVE/27 2237500 4325029 729 546 32582340 2042750 435603	Exe Die Janeire East Van Gevennaher Editore Rosseile, 100     SP See Die Eine Press. An Josef Manni, 4771 (Plane Avenah Merenser)	Indoor Company Building . Indoor Neuroscy .	. 11,400.00 . 11,400.00	0.30 0 0.30 0	. 0.00	Third Party Property Third Party Property	14,000.00 0.31 X401.5X 0.33
T0         M05         21101200         213020         213020           T0         M47         M9822073         234170         493017           T12         M47         M9822072         234170         493017	M <sup>*</sup> In Arr. Die Derweinsleich Mathie, 1179 ; Plans Merzeiner In     M <sup>*</sup> M <sup>*</sup> In Jeer Die Kan Press     K. Bennanders de Campero, 2001 ; Page Merzeiner In     Merzeiner Materia, 10 ; Postere Merzeiner     M <sup>*</sup> Merzeiner Materia, 10 ; Postere Merzeiner	lalor Nurrier : lalor Nurrier : lalor Nurrier :	- 14,455.80 - 14,455.80 - 14,455.80	0.00 0.00 0.00 0.00 0.00 0.00	- 0.00 - 0.00 - 0.00	Thind Party Property Thind Party Property	
T31         17456         R2NC12145         3249560         4311885           T36         24400         D2PDC01137         1550000         4730799           T36         24400         D2PDC01137         1550000         4740799           T36         242000         D2PDC01137         155000         4740799	Kinni Randa Concepto, 108 DF Benda Mattria Statistica (2010) A CONCO ADMINDET     Moniform Delateria     In Manuf Concepto 2010 Administrative     In Manuf Concepto 2010 Administrative	Indone Company Building . Indone Public Building .	. 11,005.00 . 11,405.00 . 14,405.00	0.30 0 0.30 0 0.30 0	. 0.00 . 0.00	Think Parey Perspersy Being: Collocad Not Located Think Parey Perspersy Think Parey Perspersy Think Parey Perspersy Think Parey Perspersy Think Parey Perspersy Think Parey Perspersy	Bring official         0.01           NOT LOCATED         0.00           A139 37         0.00
TH         47%         50%         23,88620         44,75,862           TH         1726         3832(017):041         23,96860         43,2019	<ol> <li>Sar Fasis</li> <li>Bod Rayrer Taranes, Kur LLJ. Egenes (herping (brigs: therping (bill Prior Rayrer))</li> <li>Bit Die Jawan</li> <li>Bit Die Jawan</li> <li>Bit Die Jawan</li> </ol>	baloor Shopping . Daloor Company Building .	. 14,405.90 . 14,405.90	0.00 0 0.00 0	. 0.00 . 0.00	Third Party Property Third Party Property	10,5447 0.00 NUT CMURCUS 0.00
TH         KMI         MPECINI-4         21.21100         47.81400           TH         1720 / 7120 / 7120 / 5000 / 50	SF Riems Press Av. Cd. Forwards From Law, 1989 - Riewis Weigeing     R. Re De Januire      More and Annual An	Indone Naceptag . Indone Convention Craine . Indone Naceptag .	. 14,60.00 . 14,60.00 . 14,60.00	0.30 0 0.30 0 0.31 0	. 0.00 . 0.00 . 0.00	Thind Party Property Thind Party Property Thind Party Property	15(215) 0.00 Brig official 0.00 11(0354 0.00
710 <u>3348 30%22(2)</u> 2033(20 4539) 742 2690 8675(668 273920 4531328	3P Jun Inn Die Kie Press     3r Neiselann June Laus, SKI Läurenen Kammer Const.     3C Personnende AVENUES MATERIE BENNENVERA	Index Survive	. 11,400.00 . 11,400.00	0.81 0 0.81 0	· 0.00 · 0.00	Thind Farry Property Thind Farry Property Thind Farry Property	1438133 0.8 7,13244 0.89
70         1024         3000000000         300000         400000           761         606         500004071         236600         4683356           762         2700         PRCT0000         254600         452300	MG. Biol However, Boo Alexan, U.H., Morever, P. Armillo,     SP Issue Aular In Proceedings of the Investigation of the Investicuter of the Investigation of the Investigation of the Investicuter	ladour Neuroise : . Dadour Neuroise . Dadour Neuroise .	- 16,455.81 - 16,455.81 - 16,455.81	0.00 0.00 0.00 0.00 0.00 0.00	- 0.00 - 0.00 - 0.00	Think Party Pergeny Think Party Desperty Think Party Pergeny Nor Learning Think Party Pergeny Think Party Pergeny	1379440 040 13379440 040 NOT LOCATED 040
746         1956         36C1FL/0200         373333         3649778           747         14303         36C0LATE2         1558620         459726           747         14303         36C0LATE2         1558620         459726           747         14303         36C0LATE2         1558620         459726	Oli Paradam An Done Lais, an 300 (Merging Alabara)     Mill     Nove Lawa     Ban Invasion Allow Computer, US: (Merging Alabara)     Vert Search Laware Laware Uran     Vert	lalor Shyping . Ialor Shyping .	. 11,405.00 . 11,405.00 . 11,405.00	0 0.00 0.00 0.00	. 0.00 . 0.00	Think Party Property Think Party Property Think Party Property	1343.12 0.00 T,560.00 0.00 T450.00 0.00
749 16798 SERCENSOF 21.5580 41.5588 789 701 SERVERICE 1.58679 363242	II Campos Des Grotanans An. 28 de Marco, 171. Nepping Annula 28     RA Subvalor AMENIEA CONTENADO, 2912	lalor Noying . Jalor Stopping .	14,808.90 . 14,808.90	0.30 0 0.30 0	. 0.00 . 0.00	Takai Pang Pengary Takai Pang Pengary	12342 0.00 1427243 0.00
TH         #10         PERTIDAT         348000         342875           TH         SPERDORM         234500         464032           TH         D400         SPERDORM         234500         464032           TH         D400         SPERDORM         1640900         452500	PE Revie East & Granges &     SP Gaussian America	Indone Naceptag . Indone Naceptag . Indone Naceptag .	. 14,00.00 . 14,00.00 . 14,00.00	0.30 0 0.30 0 0.31 0	. 0.00 . 0.00 . 0.00	Table Party Property Not Located Table Party Property	73430 0.00 NOT LOCATED 0.00 8/2/3/3 0.00
756 5230, 2880(25009, 21.7589, 41.5693) 759 12500, 8880(25009, 22.1179) 43.25645 759 5112 1252(25009, 20.1179) 43.25645	H. Convers Des Gesterann. Res Denise Nets Rester Traver, 330. Merenine Reskrauf Convers.     H. Res De Insens. Ean Back of Net Dennises 236. Merenine Reskrauf Reskrauf Convers.     Vie. Vie. Vie. Reskrauf and the Vie. Reskrauf Vie. Vielander Vie. Vielander	Index Nervice	- H.400.01 - H.400.01 - H.400.01	0.0 0 0.00 0	· 0.00	Thind Party Property Thind Party Property Thind Party Property	2022346 0.00 13.056.43 0.00 7334.49 0.00
TH         THC         THC <ththc< th="">         THC         <ththc< th=""> <ththc< th=""> <ththc< th=""></ththc<></ththc<></ththc<></ththc<>	<ul> <li>tra visu</li> <li>Bar Pado</li> <li>Are Pado</li> <li>Frenda Andreas</li> <li>Sar Pado</li> <li>Frenda da Campo Langos, El 3. Shepping Basesi</li> <li>Sar Pado</li> <li>Frenda da Campo Langos, El Approp. Lango</li> </ul>	haloor Shopping Indoor Shopping Indoor Shopping -	. 15,455.00 . 15,455.00 . 15,455.00	0.00 0 0.00 0	. 0.00 . 0.00	Thind Party Property Thind Party Property	000100 000 721,18133 0.03 8,07127 0.00
710         4440"         Bic/Aburtist         2011010         31.15216           740         19772         PARE 30270         1.40708         48.4586           741         4270         PARE 30270         1.40708         48.4586           742         4270         PARE 30270         1.40709         46.4586	R5         Convent         AV COLUM RNDE SCH 21 (2014) (2014)           PA         Belen         Bel. 2014 (2014)	lalor Negrig . lalor Negrig . lalor Nerror .	- 14,005.00 - 14,405.00 - 14,405.00	0.31 0 0.31 0 0.31 0	. 0.00 . 0.00	Thind Party Peoperty Thind Party Peoperty Thind Party Peoperty	420.79 0.89 55,726.89 0.81 97,766.85 0.89
782 44754 3097420276 26.27140 36.23368 783 4426 30922.82583 23.52120 44.76880	SP Ivenuelopola Av. Lativa Gonze, 600 - Shepping Conter Formalityola     Se Faulto K. Gunineso, equine sem a Eas Catio - Stepping Conter Laya	Indoor Stepping . Indoor Stepping .	. 14,858.90 . 14,858.90	0.01 0 0.01 0	. 0.00 . 0.00	Third Party Property Third Party Property	42%4P 0.00 11,495.56 0.00
764         4727         30°070173         23.51560         46.61786           755         4478         30°0707014         23.52590         46.6304           756         5274         30°0707014         23.52590         46.6304           756         5274         30°0707014         23.52590         46.6304	3F Sar Pasis     TV CAMADE (DNG) [2]     3r Pasis     TV CAMADE (DNG) [2]     3r Pasis     Except of the second seco	Dalour Nauping . Dalour Nauping . Dalour Naurine .	. 14,405.00 . 14,405.00 . 14,405.00	0.00 0 0.00 0 0.00 0	. 0.00 . 0.00	Tabal Party Property Tabal Party Property Tabal Party Property	14,1412 0.00 7,8448 0.00 9.02.17 0.00
75 1270 3000030470 155579 455799 76 2738 PENDOT 254720 45559	MC Rela Hosterner Rea Taria, 107 . Sterreiter Calade PR Constant AVENUEA MARIELLA, R.C. BRANC PR DOCTO, 4000	Index Survive	. 11.00.01 . 11.00.01	0.31 0 0.31 0	· 0.00 · 0.00	Not Located Third Party Despery Third Party Despery	NOT LOCATED 0.0 12.266.89 0.00
No.         JULY         Sector         Automy           No.         JULY         Sector         Automy           No.         JULY         Automy         Automy           No.         JULY         JULY         Automy	en Ar. Sin July A. A. Sin July 200 . Steeping Column 247 San Paulo AV. COLUMN C	lalar Nagyag Ialar Nagyag Ialar Nagyag	- 14,405.00 - 14,405.00 - 16,405.00	0 030 0 030 0 030 0	. 0.00 . 0.00 . 0.00	Think Party Property Think Party Property Think Party Property	M1 0.00 B47133 0.00 Q384733 0.00
772         17160         SEEELAVNON         2237500         4322849           773         55860         SEEMALS2NON         232750         4425889           773         55860         SEEMALS2NON         232750         4425889           773         55860         SEEMALS2NON         232750         4425889	Bit Die Januise Rau Mangaliche Nile Verente, N Neupping du Griesen M. A. San Lan. An Daniel de La Trandre, (M Merille Management Description (Control Les NAU). Management Del Pro- March Management Description (Control Les NAU). Management Del Pro-	haloor Shopping . haloor Shopping .	. 11,005.00 . 11,005.00	0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.0	. 0.00 . 0.00	Third Party Property Third Party Property Third Party Property	104242 0.00 13,44150 0.00 1071510 0.00
TN         L303         MEXEC/L0077         L3080%         43367%           TN         6.807         376976075         233726         460908           TN         6.807         376976079         233376         460509           TN         6.015         376976079         233376         460520           TN         6.05         37697305         23320         460527           TN         6.95         87613356         4350274         233720	Are Balis     Are Balis     Are Balis     Are Balis     Are Balis     Are Balis     Are Balis     Are Balis     Balis	lator Nopping . Inform Nopping . Inform Nopping .	- 14,40.00 - 14,400.00 - 14,400.00	0.33 0 0.33 0 0.33 0	. 0.00 . 0.00 . 0.00	Third Party Preparty Third Party Preparty Third Party Preparty	0.00 10/3474 0.00 1/3.6.99 0.00
r/r         MART         SPELINACIM         21.07220         44.06274           TN         4308         PERMISSI         35.0613         54.01897           TN         4400         22.07294         24.00797         44.0477	rer v. volk Badwin Reprint Travan, Ine 23 Ph. Jahonan Das Conserges Avaida Barrera de Marsana 100 247 Mar Park Avaida Barrara 1974 Avaida Barrara	nacor Negying . Indoor Negying . Jadoor Negying .	- 14,800.00 - 14,800.00 - 14,800.00	0 860 0 860 	. 0.00 . 0.00	Third Porty Property Third Porty Property Third Porty Property	v_od.83 0.00 4(987.52 0.00 36.165.50 0.00
	CE Particles International Control of C	Indexe Neurose Indexe Neurose	- 11,40,50 - 11,40,50 - 11,60,50	0 0.00 0 0.00 0	0.00 . 0.00	Third Party Property Third Party Property Third Party Property	23,55,146 0.00 23,75,327 0.00 76,000
	<ul> <li>Art Rig Fast Lans, 2021</li> <li>Canin Do Jul</li> <li>Eli Ganin Do Jul</li> <li>Eli Tol 2022 NN</li> <li>Eli Tol 2022 NN</li> <li>Eli Tol 2022 NN</li> <li>Eli Advador</li> <li>AMNNEA TANCELEO NEMES, 22.02, 01 (1997) NL IALVAEOR</li> </ul>	lalor Nopping . Inform Nopping . Inform Nopping .	- 14,00.00 - 14,00.00 - 14,00.00	0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.0	. 0.00 . 0.00 . 0.00	Thind Party Preparty Thind Party Preparty	AD         AD           AD         <
imp         4.978         10790203304         23.6780         -8.67020           7%         2628         362.902077         -27.38080         -8.666.011           7%         562.022329306         -127.99680         -3.640***           7%         562.022329306         -127.99680         -3.640***	Are Dala AC INTERACION, 2001. Suppring baselages     BC San Jow BUS GERONCED THIFTEE     Maindaw ANEXEDA ANTONIO CARLOS SANCASIONS, NR. 3407-100-1.	naloor Norping . Indoor Norping . Indoor Norping .	- 14,405.00 - 14,005.00 - 14,005.00	0 860 0 860 4 860	. 0.00 . 0.00 . 0.00	Thind Party Property Thind Party Property Thind Party Property	21,254.49 0.00 9,021571 0.00 8,256.00 0.00
TH         6002         MPAR/2042         21.782%         46.201M           740         6.702         MPRVARIM         246.010         46.7988	37 Annugara Annue Tagahami Histor de Susan Parkani, 2020. Shepping Jangai     37 San Pado An Gravansi Granda, 8027. Shepping Janka Xii     28 Ganzia Annuel An	Indoor Noopping . Indoor Noopping . Indoor Noopping .	- 14,805.00 - 14,805.00	0.33 0 0.33 0 0.39 -	- 3.00 - 3.00	Third Party Property Third Party Property Third Party Property	4,13677 0.00 15,9684 0.00 722520 000
776 5/28 3979/5/288 215.000 46.000 772 599 3979/5/288 25.600 46.000 772 599 3979/5/288 25.600 46.000	3P         Jan Pauls         An One Baserook NDI, Roserook And Conser.           3P         Jan Pauls         An One Baserook NDI, Roserook Law Oceandrow           3P         Jan Pauls         An Onesadow NDI, Roserook Law Oceandrow	Induser Neuroine		0.00 0 0.00 0 0.00 0	1.00 	Think Forty Property Think Forty Property Walk Forty Property	1000 1145530 000 1835575 000
771         65.06         325 82 2005         23.54660         464.007           781         875.05         972 2000004         23.5500         464.1007           780         864.05         1004.55566         230.23700         464.1007           780         864.05         1004.55566         230.23700         464.1007	MP         San Paulo         R. Consuel Xasiar de Taisles, 23. Muerraise Laries           3P         San Paulo         Base Rateix de Ladines, 300 (NUL), Muerraines March March           10         Jones         An juites, 330 (MURTHMAR MARCH), MULTIPPE (MURTHMAR)	lakor Norrise . Iakor Norping . Iakor Norping	- 11,400.00 - 11,000.00 - 11,000.00	0.00 0.00 0.00 0.00 0.00 n	· 0.00 · 0.00 · 0.00	Third Party Property Third Party Property Third Party Property	7,434.08 0.00 0,949.22 0.00 7,577.34 0.00
78         6475         3PBTA380*         23,3380         64,37(3)           77         6334         3PBT12008         23,540*0         46,47(3)           78         674         3PBT12008         23,540*0         46,47(3)           78         674         3PBT24208         23,540*0         46,47(3)	Sar Paulo E. Complexy: Corport 1: Newport Meril Endowed Temper     Sar Paulo Arc Joan Paulom Respondent Company     Sar Paulo Arc Joan Paulom Respondent Meril Insurem     Sar Paulo Arc Joan Paulo Arc Joan	Index Norping . Index Norping .	14,00530 14,00530 14,00530	0.00 0 0.00 0 0.00 0 0.00 0	- 0.00 - 0.00	Thini Pariy Property Thini Pariy Property Thini Pariy Property	X3564 0.00 X7X3.47 0.00 2005/22
70 666 39%TA131 233459 443708 80 995 39%C2434 214976 443703	SP San Paulo     Denning Arman, Arma, Arman, A	lalor Nopping lalor Nopping	- Hydrixii - 11,45531 - 11,85531	0 0.00 0	· 0.00	Thiad Party Property Thiad Party Property	13,33,177 0.88
nm         27100         PEEND074         23.43500         6722.800           R0         6522         EXE3.54655         3.81276         3.52003           R0         1796         3.6229C1507         2.227800         4122****	rm. combin. MELEPENES METHEORIC IF.N., PEAC, METHEORIC IF.N., EN. Nord. Avenda Remarks Vision 2735 Marying Malvay mil. El En De Jacon. Avenda Remarks Anna Lador Kare D. 120.	nacor Negying . Indoor Negying . Indoor Negying .	- 16,400.00 - 16,400.00 - 16,400.00	0 860 0 860 0 860	. 0.00 . 0.00 . 0.00	Not Located Third Party Preparty Third Party Preparty	PAGE ALKATED 0.00 A,807.43 0.00 36,223.77 0.00
88 599 3712527 25589 45527 89 11777 2583277 5499 45527	SP James An Am Carm M2, Marrier Parae Relation     Phys. Rev. D 10, 1975 (Surgery Res. Relation     Phys. Rev. D 10, 1975 (Surgery Rev. Rev. Relation)	Indexe Neuroise Indexe Neuroise	- 11,4000 - 11,4000	0.20 0. 0.20 0. 0.20 0.	· 3.00	Third Party Preparty Third Party Preparty Third Party Preparty	6408.43 0.00 13726.07 0.00
800 4240 10700 .2334210 .4448511 807 0000 30933110 .2356800 .4433217	ser san Paula As Higenipula, GE Merping Pain Hgienipula 3P Januar As Ana Cons, 6D Merping Pain Hgienipula 3P Jan Paula B. Toward Million, D.C. Merping Pain Palan Science Mergins (Marine Mergins)	nacor Negying . Indoor Negying . Indoor Negying .	- 16,400.00 - 16,400.00 - 16,400.00	0.38 0 0.38 0 0.39 0 0.09 0	. 0.00 . 0.00 . 0.00	Third Party Property Third Party Property Third Party Property	1,0079 0.00 1,14758 0.00 12,0754 0.00
			14,805.00 14,805.00 14,805.00	0.33 0 0.33 0 0.33 0	0.80 0.80	Third Party Property Third Party Property Third Party Property	2,56,47 0.30 3,776,50 0.38 80,393,17
ND         LTPX2         MARLENAV         LTPX2         MARLENAV         LTPX2         MARLENAV           ND         NOT         NODA AACULODIN         LTPX2000         MARLENAV           ND         NOT         NODA AACULODIN         LTPX2000         MARLENAV           ND         NOT         NODA AACULODIN         LTPX2000         MARLENAV           ND         NODA         NODA         NODA         NODA	BE Rele However Arenda de Contrano, HEL - Shapping Pilo Serveri EA Salvador ELA CONS, FENQUEREA, ANREA, MA/PO Demondra Arenda Pilo Serveria	Induser Neuropeag Induser Neuropeag Induser Neuropeag	- 14,405.30	0.00 0	. 6.00	Third Party Property	0.00 A,18XM 0.00 4,55X35 0.00
80         1172         200420100         -233.000         -4004010           80         1172         200420100         -123.000         -4004010           80         807         2004304         -151.000         -4004010           80         807         2004304         -123.000         -4004010           80         80.001         2004304         -22.000         -40040           80         20102         914.00104         -22.0700         -41.0000           80         20102         8104.20172         -22.0700         -41.00172	MO         Mol Housew         Available Concerns, 601. Shepping Pain Neural           RA         Mol More         ELACIDES (EDS) (EDS) (EDS) (EDS) (EDS)           Presender         Re Lances (22). Shepping Painsible           P         Faces         Ean Colors Conce, 102 / Concern, Edspring Painsible           PI         Lances         Ean Colors Conce, 102 / Concern, Edspring Painsible           PI         Base         Available Absolution, 102 / Concern, 102 / Kongraph Road	ladow Norping . Iadow Norping . Iadow Norping . Iadow Norping . Iadow Norping .	- 14,405.00 - 14,805.00	0.31 0	. 0.00	Thind Party Property	
80         1102         201000         401010           91         1102         2010010         401000         401010           91         9102         201000         110000         401010           91         9102         201000         110000         401010           91         9102         20100         201000         410010           91         9102         9104         211700         410010           91         9102         8104         3012         213700         410011           91         9102         9104         3012         214100         404275           91         1111         8020440044         232700         433012         3012           91         2012         80207104         232700         433012         3012         3012	MD         Both Human         Annula da Conserve, Well Hanger (Proc. Network)           Ab         Market         BLACKINS, STATUR, STATUR, STATUR, NETWORK, NETW	Indone Stopping	- 15,465.80 - 15,465.80 - 16,465.80 - 16,465.80	0.30 0 0.30 0 0.30 0 0.30 0 0.30 0	0.00 0.00 0.00	Think Party Peoperty Think Party Peoperty Think Party Peoperty No Located	7,000.00 0.00 NOT LOCATED 0.00
MI         LTP2         MARADAY CAN         LTMAN         LCMAN           60         MIT         MARADAY CAN         LTMAN         LCMAN           61         MIT         MARADAY CAN         LTMAN         LCMAN           61         MIT         MARADAY         LTMAN         LCMAN           61         MIT         MARADAY         LTMAN         LCMAN           63         MIT         MARADAY         LLMAN         LCMAN           64         MIT         MARADAY         LLMAN         LCMAN           65         MIT         MARADAY         LLMAN         LCMAN           66         MIT         MARADAY         LLMAN         LCMAN           67         MIRADAY         LLMAN         LLMAN         LLMAN           68         MIT         MARADAY         LLMAN         LLMAN           69         MIT         MIT         MARADAY         LLMAN	No.         Address of the strength of the str	Lakor Roppi Lakor Series Lakor Series Lakor Series Lakor Series Lakor Series Lakor Series Lakor Series Lakor Series Lakor Series Lakor Series	- 15,4503 - 14,4503 - 14,4503 - 14,4503 - 14,4503 - 14,4503 - 14,4503 - 14,4503 - 14,4503	0.08 0 0 0.08 0 0.03 0 0.03 0 0.03 0 0.03 0 0.03 0 0.00 0	· 0.00 0.00 · 0.00 · 0.00 · 0.00 · 0.00 · 0.00 · 0.00	Takal Pary Property Takal Pary Property Takal Pary Property Ma Laural Takal Pary Property Takal Pary Property Takal Pary Property	7,000.00 0.00 NOT LOCATED 0.00
m         r/m         m         r/m         m         r/m         m </td <td>With Kalazaria         Mandaka Comment (1) Support the Namit           00         Kalazaria         Kalazaria           01         Kalazaria         Kalazaria           02         Kalazaria         Kalazaria           03         Kalazaria         Kalazaria           04         Kalazaria         Kalazaria           05         Kalazaria         Kalazaria           06         Kalazaria         Kalazaria           07         Kalazaria         Kalazaria           08         Kalazaria         Kalazaria           09         Kalazaria         Kalazaria           00         Kalazaria         Kalazaria           01         Kalazaria         Kalazaria           02         Kalazaria         Kalazaria           03         Kalazaria         Kalazaria           04         Kalazaria         Kalazaria           05         Kalazaria         Kalazaria           06         Kalazaria         Kalazaria           07         Kalazaria         Kalazaria           08         Kalazaria         Kalazaria           09         Kalazaria         Kalazaria           00         Kalazaria         Kalazaria&lt;</td> <td>ladar Bargen -  bargen -</td> <td>IL IL /td> <td>0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.0</td> <td>- 0.00 - 0.00 - 0.00 - 0.00 - 0.00 - 0.00 - 0.00 - 0.00 - 0.00 - 0.00</td> <td>Thial Pary Desproy Thial Pary Desproy Thial Pary Desproy Not Learned Thial Pary Perpeny Thial Pary Perpeny Thial Pary Perpeny Thial Pary Desproy Thial Pary Desproy Thial Pary Desproy</td> <td>7,000.00 0.00 NOT LOCATED 0.00</td>	With Kalazaria         Mandaka Comment (1) Support the Namit           00         Kalazaria         Kalazaria           01         Kalazaria         Kalazaria           02         Kalazaria         Kalazaria           03         Kalazaria         Kalazaria           04         Kalazaria         Kalazaria           05         Kalazaria         Kalazaria           06         Kalazaria         Kalazaria           07         Kalazaria         Kalazaria           08         Kalazaria         Kalazaria           09         Kalazaria         Kalazaria           00         Kalazaria         Kalazaria           01         Kalazaria         Kalazaria           02         Kalazaria         Kalazaria           03         Kalazaria         Kalazaria           04         Kalazaria         Kalazaria           05         Kalazaria         Kalazaria           06         Kalazaria         Kalazaria           07         Kalazaria         Kalazaria           08         Kalazaria         Kalazaria           09         Kalazaria         Kalazaria           00         Kalazaria         Kalazaria<	ladar Bargen -  bargen -	IL IL IL IL IL IL IL IL IL IL IL IL	0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.0	- 0.00 - 0.00 - 0.00 - 0.00 - 0.00 - 0.00 - 0.00 - 0.00 - 0.00 - 0.00	Thial Pary Desproy Thial Pary Desproy Thial Pary Desproy Not Learned Thial Pary Perpeny Thial Pary Perpeny Thial Pary Perpeny Thial Pary Desproy Thial Pary Desproy Thial Pary Desproy	7,000.00 0.00 NOT LOCATED 0.00
00         172         Market Carl         154.00         454.00           0         0.00         0.00         0.00         0.00           0         0.00         0.00         0.00         0.00         0.00           0         0.00	101         Kalaza         Bukatowa (Kalaza)         Bukatowa (Kalaza)           101         Kalaza         Bukatowa (Kalaza)         Bukatowa (Kalaza)           101         Kalaza         Bukatowa (Kalaza)         Bukatowa (Kalaza)           101         Kalaza         Bukatowa (Kalaza)         Bukatowa (Kalaza)           101         Kalaza         Bukatowa (Kalaza)         Bukatowa (Kalaza)           101         Kalaza         Bukatowa (Kalaza)         Bukatowa (Kalaza)           101         Kalaza         Bukatowa (Kalaza)         Bukatowa (Kalaza)           101         Kalaza         Bukatowa (Kalaza)         Bukatowa (Kalaza)           101         Kalaza         Bukatowa (Kalaza)         Bukatowa (Kalaza)           101         Kalaza         Bukatowa (Kalaza)         Bukatowa (Kalaza)           101         Kalaza         Bukatowa (Kalaza)         Bukatowa (Kalaza)           101         Kalaza         Bukatowa (Kalaza)         Bukatowa (Kalaza)           101         Kalaza)         Bukatowa (Kalaza)         Bukatowa (Kalaza)           101         Kalaza)         Bukatowa (Kalaza)         Bukatowa (Kalaza)           101         Kalaza)         Bukatowa (Kalaza)         Bukatowa (Kalaza)           101	Labor Bargers Aller Alle	- 11,420,50 - 11,420,50 - 11,420,50 - 14,420,50 - 14,	0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.0		Thiad Face Steppeny Thiad Face Steppeny Thiad Face Steppeny Mid Loand Thiad Face Steppeny Thiad Face Steppeny Thiad Face Steppeny Thiad Face Steppeny Thiad Face Steppeny Thiad Face Steppeny Thiad Face Steppeny	7,000.00 0.00 NOT LOCATED 0.00
0         (10)         (2	<ul> <li>Marcine Marcine State S</li></ul>	Marta Borgen,	- 1,4,60,50 - 1,4	9 0 00 9 0 00		Bala Fao, Bagary Bala Fao, Bagary Na Landi Mark Fao, Bagary Bala Fao, Bagary Bala Fao, Bagary Bala Fao, Bagary Bala Fao, Bagary Bala Fao, Bagary Bala Fao, Bagary Bala Fao, Bagary Bala Fao, Bagary Bala Fao, Bagary Bala Fao, Bagary Bala Fao, Bagary Bala Fao, Bagary	7,000.00 0.00 NOT LOCATED 0.00
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Simple         Markar         Markara           Simple         Markara         Markara	Labor         Regrey         -           Adar         Regrey         -           Adar </td <td>- 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1,</td> <td>9 0 000 0 000 0 000 1 000 0 000</td> <td>- 100 - 100</td> <td>Ball day Jugary Ball day Jugary Na Linux Hung Yung Kalawat Mall ang Jugary Ball day Jugary Ball day Jugary Ball day Jugary Ball day Jugary Ball day Jugary Ball day Jugary Ball day Jugary Ball day Jugary Ball day Jugary Ball day Jugary Ball day Jugary Ball day Jugary Ball day Jugary Ball day Jugary</td> <td>7,000.00 0.00 NOT LOCATED 0.00</td>	- 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1,	9 0 000 0 000 0 000 1 000 0 000	- 100 - 100	Ball day Jugary Ball day Jugary Na Linux Hung Yung Kalawat Mall ang Jugary Ball day Jugary Ball day Jugary Ball day Jugary Ball day Jugary Ball day Jugary Ball day Jugary Ball day Jugary Ball day Jugary Ball day Jugary Ball day Jugary Ball day Jugary Ball day Jugary Ball day Jugary Ball day Jugary	7,000.00 0.00 NOT LOCATED 0.00
0         0	101         Marine         Marine Mari	Mark         Party         -           Mark         Party <td>- NAARO NAAR</td> <td>ا و الم الم الم الم الم الم الم الم الم الم</td> <td></td> <td>Bad Ang, Baury San Ang, Baury Kai Ang, Baury</td> <td>7,000.00 0.00 NOT LOCATED 0.00</td>	- NAARO NAAR	ا و الم الم الم الم الم الم الم الم الم الم		Bad Ang, Baury San Ang, Baury Kai Ang, Baury	7,000.00 0.00 NOT LOCATED 0.00
0	Bit Sharah         Parkarah         Parkarah           Bit Sharah         Parkarah	Loc         Party           Loc <td></td> <td></td> <td></td> <td>Band and See Strategy The See</td> <td>7,000.00 0.00 NOT LOCATED 0.00</td>				Band and See Strategy The See	7,000.00 0.00 NOT LOCATED 0.00
4	Simple         Markara         Markara <td< td=""><td>Lat.         Act.           Lat.         Act.           Lat.<td></td><td></td><td></td><td>Said on Parent Said Said Said Said Said Said Said Said</td><td>7,000.00 0.00 NOT LOCATED 0.00</td></td></td<>	Lat.         Act.           Lat. <td></td> <td></td> <td></td> <td>Said on Parent Said Said Said Said Said Said Said Said</td> <td>7,000.00 0.00 NOT LOCATED 0.00</td>				Said on Parent Said Said Said Said Said Said Said Said	7,000.00 0.00 NOT LOCATED 0.00
0	Bits         Mathematical Strategies (March 1998)           201         Participation (March 1998)           201         Partipation (March 1998) <t< td=""><td>Lat.         Approx           Lat.         Approx           Lat.<td></td><td></td><td></td><td>Baller for Sperry Same Service Same Servic</td><td>7,000.00 0.00 NOT LOCATED 0.00</td></td></t<>	Lat.         Approx           Lat. <td></td> <td></td> <td></td> <td>Baller for Sperry Same Service Same Servic</td> <td>7,000.00 0.00 NOT LOCATED 0.00</td>				Baller for Sperry Same Service Same Servic	7,000.00 0.00 NOT LOCATED 0.00
0	Simple Section of the sectio					Ball for Sperg Mark for Mark Mark br>Mark br>Mark Mark Mark Mark Mark	7,000.00 0.00 NOT LOCATED 0.00
0	Bit Sharah         Balanan         Balananan         Balanan					Balance Parenet San Carlo Paren	7,000.00 0.00 NOT LOCATED 0.00
0	<ul> <li>Marian Andream Control of Marine State St</li></ul>					Barley An Party Same Series (1999) Barley And Same Series Barley And Same S	7,000.00 0.00 NOT LOCATED 0.00
0	Bit Sharah         Parkarah         Parkarah           Bit Sharah         Parkarah						7,000.00 0.00 NOT LOCATED 0.00
0	Bit Share         Markara Markaran Share						7,000.00 0.00 NOT LOCATED 0.00
0	Sec.         Aliana         Mathata Mathatana						7,000.00 0.00 NOT LOCATED 0.00
0	Bit Share         Particular Share						7,000.00 0.00 NOT LOCATED 0.00
No.         No. <td>Source         Source           Source         Source           Source</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>	Source         Source           Source						
Total Trad Remarkation Roles	<ul> <li>Bartan Andream Antoning an An</li></ul>						7,000.00 0.00 NOT LOCATED 0.00
no 240 de la constante de la constante de la constante Tanal Economismo Robert Race Economismo de constante de CLEINT sell pay de seculido hare remanencian	9         Nome         Market in Ma						7,000.00 0.00 NOT LOCATED 0.00
Total Trad Remarkation Roles	Bit State         Hardson State           Bit State						7,000.00 0.00 NOT LOCATED 0.00
no 240 de la constante de la constante de la constante Tanal Economismo Robert Race Economismo de constante de CLEINT sell pay de seculido hare remanencian	Bit Share         The start of the sta						7,000.00 0.00 NOT LOCATED 0.00
no 240 de la constante de la constante de la constante Tanal Economismo Robert Race Economismo de constante de CLEINT sell pay de seculido hare remanencian	Bit Share         Balance           Bit Share         Balance<						7,000.00 0.00 NOT LOCATED 0.00

Sharing Agreement Exhibit 2 - Dilution Example

#### 1) Example for calculating dilution without RAN Sharing:

Incremental value for the remaining sites (C=A/B)	R\$ 5,13
Number of remaining Infrastructure Items (B)	858
Monthly Net Base Fee at the time of demobilization of the Infrastructure Item (A)	R\$ 4.400,00

#### 2) Example for calculating dilution with RAN Sharing:

Monthly Net Base Fee at the time of demobilization of the Infrastructure Item (A)	R\$ 4.400,00
Monthly Net value related to RAN Sharing at the time of demobilization of the	\$ 1.250,00
Infrastructure Item (B)	
Number of remaining Infrastructure Items (C)	858
Incremental value for the remaining sites (D=(A+B)/C)	R\$ 6,59

#### 3) Example for calculating dilution with Additional Compensation

R\$ 1.000,00
858
R\$ 6,29

# EXHIBIT 3 - OPERATIONAL PROCEDURES FOR USE AND ACCESS TO AREAS

#### 1. Installation and increase or change of space occupied in the Infrastructure Items and/or Areas of the Contractor or owners of the properties in which the Infrastructure Items are located

1.1 For any request to install or increase the space occupied, or to be occupied, in the Infrastructure Items and/or Areas and/or to make physical changes at the Site, the **CONTRACTING PARTY** shall send the **CONTRACTOR** a request with the details of the respective technical project (the "Project").

The implementation of security items by the Contracting Party, aiming to protect its assets installed on the sites, such as cages, locks and cable protections, are not additional occupations. Thus, they do not need to go through the process of prior approval by the Contractor, and the Contracting Party shall inform the Contractor, except for those devices that, in some way, may hinder or restrict access.

1.2 The **CONTRACTOR** shall review the Project within thirty (30) days of its receipt, and may (i) approve the Project as presented, (ii) propose changes to the Project, or (iii) reject the Project, upon written explanation of the rejection. In no event shall the Project be unreasonably denied.

1.3 The **CONTRACTOR** shall make available:

(i) Ground connection point with excess cable at the base of the container or rack, if and only if the grounding system is owned or controlled by the **CONTRACTOR**.

(ii) Space in the tower, as agreed.

(iii) Authorization for the **CONTRACTING PARTY** and its personnel to perform the installation of its equipment.

1.4 The **CONTRACTING PARTY** shall make available:

(i) List of companies and authorized persons ("Authorized Personnel") who will work at the installation upon two (2) days prior notice. Access to the Site may not be denied in the event that it is necessary to comply with the rules set forth by the Regulatory Entities. Any other access without the due notice may be denied by the **CONTRACTOR**, provided that it is justified.

(ii) Installation project to be approved by the **CONTRACTOR**, according to items 1.1 and 1.2.

(iii) Determination of a person of the **CONTRACTING PARTY** responsible for the coordination of the installation, and provide his/her contact details to the **CONTRACTOR**.

(iv) All the PPE needed for its activities

1.5 Obligations in increasing the space occupied in the Infrastructure Items and/or Areas:

1.5.1 Any installation that requires any increase in the Infrastructure Items shall be done by the **CONTRACTOR**, including, but not limited to, the Antenna Supports associated with the tower structure.

1.5.2 The **CONTRACTING PARTY** shall designate a person as the technical responsible for the work who will be responsible for the acts of the Authorized Personnel and coordinate the installations with the **CONTRACTOR**.

1.5.3 The **CONTRACTING PARTY** is responsible for the removal of rubble, leftover material and garbage resulting from the installation of its equipment or works. In case the **CONTRACTING PARTY** does not remove the rubble, leftover material and garbage within seven (7) days after the end of the installation of its equipment, the **CONTRACTOR** is authorized to remove them, and the **CONTRACTING PARTY** shall be obliged to bear the costs, provided they are proven.

1.5.4 In view of the legal need to issue electromagnetic emission reports to the municipal government and communities, the **CONTRACTING PARTY** shall be responsible for obtaining the reports related exclusively to the Equipment owned by it. In case there are several Carriers using the Infrastructure Item, the **CONTRACTOR** shall request the measurement service for all installed equipment, sharing the costs thereof among the companies that use the Infrastructure Item, provided that such costs are previously approved by the **CONTRACTING PARTY**.

1.5.5 If there is a need to reduce irradiated power, in case it exceeds the maximum values allowed for that Infrastructure Item, the **CONTRACTING PARTY** shall have its original irradiation power up to the maximum limit regulated by ANATEL, whenever it is proven that equipment has been installed prior to the other carriers in that Infrastructure Item. The equipment of other companies, installed after those of the **CONTRACTING PARTIES**, shall have their power reduced proportionally, until the standards permitted by current legislation are reached. This hypothesis of previous equipment of other carriers does not apply to cases of Switched Fixed Telephone Service (*Serviço Telefônico Fixo Comutado* - STFC) equipment from the **CONTRACTING PARTY**.

## 2 Operation of the Infrastructure Item

2.1 The **CONTRACTOR** shall provide, at its discretion and schedule, preventive maintenance on the Infrastructure Items.

The Contractor shall make the semiannual schedule of preventive maintenance available to the Contracting Party.

- 2.1.1 The Preventive Maintenance is composed of, but not limited to:
- (i) Basic cleaning of the Area/Land, removing grass and objects;
- (ii) Redistribution of gravel in the Area/Land;
- (iii) Painting of the Area's walls, when applicable;
- (iv) Recovery of gates, fences and railings, when applicable;

(v) Painting touch-up on the Infrastructure Item, supports, mats, lampposts and gate (when applicable);

(vi) Soil treatment for the grounding system, when the grounding system is owned or operated by the **CONTRACTOR**.

(vii) Visual inspection in the AC box;

(viii) Visual inspection and testing of the signaling system;

<del>(ix);</del>

(x) Visual inspection of the grounding system. The relevant measurements shall also be carried out

(xi) Inspection of the lightning protection system;

(xii) Inspection of cleaning of junction boxes in the Area;

(xiii) Additional maintenance of the Infrastructure Item and its accessories, as determined by the **CONTRACTOR**;

(xiv) Maintenance of access roads in order to ensure access;

(xv) Inspection of the energy standard;

(xvi) Inspection of the Area lighting

(xvii) Verification of the existence of animals and insects and, if any, provide for their removal.

2.2 The **CONTRACTOR** shall first inform the **CONTRACTING PARTY**, in case the preventive maintenance impacts or interferes with equipment, irradiating system (cables and antennas), power system and/or grounding system of the **CONTRACTING PARTY**, so that it can monitor the activity, if necessary.

The **CONTRACTOR** shall provide emergency repairs to the Infrastructure Items in accordance with its responsibilities defined in the responsibility matrix of this Exhibit, in order to make them available within twenty-four (24) hours for urban sites and forty-eight (48) hours for rural sites after written communication sent by the **CONTRACTING PARTY** and effectively received by the **CONTRACTOR** (which may be sent by email), except for a longer period agreed between the **Parties**.

2.3 The **CONTRACTING PARTY**, whenever it becomes aware, shall directly request that **CONTRACTOR** carries out corrective maintenance when the following situations occur:

(i) Lack of electricity (when not programmed by the utility company), only in the event that the **CONTRACTOR** owns or controls the Infrastructure Item affected by the lack of electricity;

- (ii) Burnt out signaling lamp;
- (iii) Failure in the grounding system/ lightning arrestor;
- (iv) Obstruction of ducts and junction box in the Area;
- (v) Trespassing and/or depredation;
- (vi) Damage caused by third parties;
- (vii) Impossibility of accessing the site in view of the conditions of the roads;
- (viii) Malfunction of the access system implemented by the Contractor;
- (ix) Damage to security items in the Area (walls, fences, gates, concertinas, etc.)

Existence of animals and insects that are preventing access to the Area and Infrastructure Items

2.3.1 Once informed of the problems listed in item 2.4 above, the **CONTRACTOR** shall send a technical team to attend to the events indicated by the **CONTRACTING PARTY**. Such team shall carry out the necessary corrective maintenance and provide, at the end thereof, a report of the cause and solution of the defect to the management center of the **CONTRACTING PARTY**. If there is no possibility of repair, the **CONTRACTOR** shall inform the **CONTRACTING PARTY** so that it can apply its contingency measures.

2.4 Any contact with the owners or possessors of the properties where the Infrastructure Items are located shall be made exclusively by the **CONTRACTOR**, and the **CONTRACTING PARTY** must inform the **CONTRACTOR** when it detects any problem or receives any request from them.

Considering the difference in critical level of the events related to the owners of lands where the sites are located, a 24x7 structure shall be made available by the Contractor in order to immediately attend/treat cases of high criticality (total/partial loss of the site). The SLA to enable the Contracting Party's access in cases of loss of the site is up to 24 hours.

2.6. The **CONTRACTING PARTY** and/or its Authorized Personnel is authorized to perform the maintenance of their equipment, at their discretion and schedule, except when such maintenance impacts the equipment of other carriers or Infrastructure Items.

2.6.1. The **CONTRACTING PARTY** shall notify the **CONTRACTOR** seventytwo (72) hours in advance, about preventive maintenance or scheduled maintenance that it wishes to perform on the equipment installed in the Infrastructure Items.

2.6.2. If preventive or scheduled maintenance can potentially affect the equipment of another carrier or the Infrastructure Item, the **CONTRACTOR** may be elected to provide maintenance supervision.

2.6.3. The **CONTRACTING PARTY** shall use its best efforts not to impact the equipment of another carrier or the Infrastructure Items. In the event that the **CONTRACTING PARTY** impacts the equipment of another carrier or the Infrastructure Item, it shall immediately inform the **CONTRACTOR**.

2.6.3.1. The **Parties** shall be responsible for acts of negligence committed by their respective employees and the Authorized Personnel in the event of misuse of access to Infrastructure Items and Areas.

2.7. The **CONTRACTING PARTY** and/or Authorized Personnel shall not be allowed to install or relocate supports in the structures of the Infrastructure Items. This service is the sole responsibility of the **CONTRACTOR**.

2.8. The **CONTRACTOR** shall provide the **CONTRACTING PARTY** with any additional supports that it may request for the installation of its equipment in the Infrastructure Item, in line with costs mutually pre-approved with the **CONTRACTING PARTY**.

2.7. The **CONTRACTING PARTY** shall always submit to the CONTRACTOR's approval the installation of new equipment, removal of existing equipment, or changes in equipment or equipment configuration in the Infrastructure Item.

2.8. In cases where the external electric power network, under the **CONTRACTING PARTY**'s ownership, suffers acts of vandalism or accidents that prevent the energization of the equipment, the **CONTRACTING PARTY** shall request the repair of the electric network directly to the electric utility companies.

2.9 The **CONTRACTING PARTY** and/or its authorized personnel may carry out corrective maintenance of their equipment and systems, at their discretion and schedule, and shall be authorized to carry out corrective maintenance of their equipment, on a 24x7 basis, without the need for prior notice to the **CONTRACTOR**. The 24-hour access shall be ensured by the Contractor.

# 3. Access to Areas and Infrastructure Items

3.1.

3.1 To access the Areas and Infrastructure Items, the Contractor shall comply with the relevant criteria and procedures established by the Contractor.

3.2 In the case of Infrastructure Items implemented in the Contracting Party's land and/or stations, the Contractor's access to the Areas and Infrastructure Items shall comply with the relevant criteria and procedures established by the Contracting Party.

3.3 For sites where the Contractor implements a specific access control system (electronic padlock, MultiLock, etc.), the Contractor shall be responsible for making available the necessary items for access (keys, cards, encoders, etc.) enough to attend Oi's operational staff.

3.4. When the access control is based on the use of padlocks and, whenever technically feasible, the Contractor and the Contracting Party shall place their padlocks in series, in order to facilitate access and not be dependent on the other;

3.5. If only the Contractor's padlock is possible, it shall provide the padlock password, indicating recurrence focal points, with 24x7 access.

#	DESCRIPTION	CONTRACTOR	CONTRACTING PARTY	NOTES
1	Maintenance of the input branch and measurement board	X		
2	Maintenance of the QTM manual transfer board (exclusive use of the Contracting Party)		X	
3	Maintenance of the mobile Generator Motor Group (GMG) socket (exclusive use of the Contracting Party)		Х	

# 4. **Responsibility matrix**

	Maintenance of the QTM			
	-	X		
	manual transfer board (shared	Λ		
	use)			
	Maintenance of the mobile			
	Generator Motor Group	X		
	(GMG) socket (shared use)			
	Maintenance of AC power			
4	cables inside the condominium	X		
4		Λ		
	- except inside Oi's shelter			
	Maintenance of the line surge			
5	protection and signaling	X		
	systems.			
	Maintenance of the Grounding			
	Grid and the grounding of the			
6	metal parts of the	Х		
0	1	Λ		
	condominium, fences, tower,			
	QTM and power input			
	Maintenance in the grounding			
	grid of the installation			
7	material, EGB/TGB boards,	Х		
	grounding wire, indoor or			
	outdoor cables and equipment			
	* *			
~	Maintenance of the			
8	atmospheric discharge	Х		
	protection system			
0	Maintenance of external	\$7		
9	lighting (includes poles)	Х		
	Maintenance of metal			
10	structures (tower, conveyor,	Х		
	supports, fences, stairs,			
	guardrails, platforms)			
	Maintenance of metal skids			
11	(bases for Oi's owned		X	
11	equipment installed in the			
	shelter base)			
	Supply and maintenance of		X	
10				If provided by Oi
12	generators in cases of non-			
	payment of the electric bill			
	Supply and maintenance of		X	If provided by Oi
10	generators in case of lack of			
13	electricity due to failure of the			
	local electric utility company.			
	Supply and maintenance of			TC
14	generators in cases of lack of		X	If provided by
	electricity due to acts of			Oi
	vandalism			
	Supply and maintenance of			TC 111
	generators in cases of non-	X		If provided by
	payment of the electric bill			the Contractor
	1 7		+	+
	Supply and maintenance of			TC
	generators in case of lack of	Х		If provided by
	electricity due to failure of the			the Contractor
	local electric utility company.			
	Supply and maintenance of		1	
	generators in cases of lack of			If provided by
		Х		the Contractor
	electricity due to acts of			the Contractor
	vandalism			
	Supply and maintenance of			
15	generators in cases of lack of		Х	
15	generators in cases of fack of			

	third parties at Oi's service in			
	equipment and/or systems of			
	other members of the			
	condominium			
	Recovery of the			
	CONTRACTOR's			
16	shelter/cabinets, internal		X	
10	hardware and equipment of			
	Oi's equipment in the event of			
	vandalism/theft or accident.			
	Recovery of its equipment and			
17			v	
17	cables from the radiating		X	
	systems fixed to the tower			
10	Recovery of the grounding bar		v	
18	of the RF and TX cables		X	
	Recovery of the input			
19	branch/measurement board in	Х		
19		Λ		
	cases of theft/vandalism			
20	Recovery of the transfer board	v		
20	in cases of theft/vandalism	X		
	Recovery of the GMG socket			
21		Χ		
	in cases of theft/vandalism			
	Recovery of the AC power			
22	cables in cases of	Х		
	theft/vandalism			
	Recovery of the line surge			
23	protection system in cases of	Х		
23		Δ		
	theft/vandalism			
	Recovery of the signalizing			
24	and atmospheric discharge	v		
24	protection systems in cases of	X		
	theft/vandalism			
	Identify Oi's radiating system			
25			X	
	cables			
26	Identify Oi's AC cables		X	
27	Identify Oi's grounding cables		X	
	Recovery of metallic structures			
	(tower, gate, fencing, supports,			
28	light post, padlocks and others)	Χ		
	in cases of theft/vandalism			
	Recovery of the civil part			
	(wall, fences, floors, junction			
	box, conduits, concrete base,			
29	input box, fence posts and	X		
	theft/vandalism			
30	Definition of the preventive	X		
50	maintenance schedule	Λ		
	Provision of the semiannual			
	preventive maintenance	Х		
	1	Λ		
	schedule for the Contractor		+	
	Basic land cleaning, with			
31	periodicity to be defined by the	Х		
	Contractor			
				Applicable in
	Periodic cleaning and garbage			cases where the
	collection in the space not			
	involved in the negotiation		X	site is on the
	(Space that remains with the			Contracting
				Party's
	Contracting Party)			
	Contracting Party)			Party's land/station

32	Definition of the preventive maintenance items	X		
33	Visual inspection with periodicity to be defined by the Contractor	X		
34	Radiometric emission reports, requested by the municipal government annually or when required		X	Exclusive to Oi's systems and equipment
35	Any report relevant to the operation of Oi's system when and if required		x	
36	Implementation and, when necessary, replacement of access control systems (padlocks, electronic locks, etc.)	Х		Serial installation of padlocks by the Contracting Party and the Contractor
37	Painting of metal structures (tower, conveyor, supports, fences, stairs, guardrails, platforms)	X		
38	Retightening of the structure	X		
39	Treatment of lease and loan for use agreements, including service to lessor	X		
40	Release of access for Oi and its authorized agents	X		
41	Release of access for the Contractor and its authorized agents, when the site is at an Oi's station		X	
	Increased load/increased capacity of the power plant (energy system under the Contractor's responsibility). It includes adaptation of the installations (cables, general circuit breaker) when there is an increase in AC load or new demand from a new carrier that justifies the change of the general circuit breaker.	Х	X	Responsibility of the UC holder
	Certification with the Fire Department	Х		
	EV Urban or Environmental Licensing	X		
	Clarifications/negotiations of notifications/letters from public bodies and the judiciary	X		
	Access control other than padlocks (implemented by the Contractor)	X		Provide resources to the Contracting Party in a number that meets the operational demand
	BO related to events on the sites/Determination of	Х		

property wrongful acts - when			
the event involves equipment,			
Infrastructure Items and Area			
of responsibility of the			
Contractor			
BO related to events on the			
sites/Determination of			
property wrongful acts - when		Х	
the event involves equipment			
and Area of responsibility of			
the Contracting Party			
Increased protection of the site			
with incidence of vandalism			
(placement of wall, steel plate,	Х		
electric fence, etc.) - Area of			
the Contractor			
Increased protection of the site			
with incidence of vandalism			
(placement of wall, steel plate,		Χ	
electric fence, etc.) - Area of			
the Contracting Party			
conducting r uty			If the utility
			-
			company
			requires the
			holder to open
Ticket opening of utility			it, it will be the
Ticket opening at utility	v	v	responsibility of
companies in case of lack of	X	X	the Contractor.
electricity			Otherwise, it
			will be
			responsibility of
			the Contracting
			Party
Preparation/availability of			Each one
1			responsible for
documentation required by	Х	Х	what falls
Anatel/other bodies, regarding			within its
EVs			
			competence
Electric bill payment			Direct invoice
(individual branches)		X	to the
			Contractor
			UC holder pays
			and receives
Electric bill payment -	Х	Х	reimbursement
common branches	<b>4</b>	<b>4</b>	from the other
			party
			When it is not
			possible to
Request for energy input from		•	individualize,
electric utility companies		X	the Contractor is
			the holder of the
			UC
			UL
Removal of bees, hornets and			
other wild/venomous animals	Х		
(birds' nest, vultures, hawks,	<b>4</b>		
etc.) from the site/EV			
Maintenance/clearance of			
access roads	Х		

Individual Safety (Fall arrest/Guardrails/Cranksets/etc .)	X		
Provision of support for radiating systems	X		
Execution of the adaptation in the existing infrastructure with respect to the Infrastructure Item	X		
Update of contact list/operational organization chart	X	X	

5. The responsibilities with respect to the Operational Procedures set forth in this Exhibit 3 do not depend on the regime of occupation and use of the Areas and Infrastructure Items, and do not depend on the Contractor being as exclusive user or the existence of other sharers.

6. Execution of works and services on sites located at Contracting Party's stations

6.1. For all cases in which the Contractor and/or its users other than the Contracting Party intend to develop any intervention in the Areas and Infrastructure Items that are on the Contracting Party's land and/or station, and the activity goes beyond the negotiated Area, even if temporarily, the Contractor shall send a request together with relevant documentation for prior analysis by the Contracting Party.

(i) The Contracting Party shall analyze the request within thirty (30) days after receiving it.

(ii) The Contracting Party may i) approve the Project as presented, ii) propose changes to the Project, or iii) reject the Project, upon written explanation of the rejection.

7. Notices regarding Operational Procedures

7.1 Actions between the parties regarding the Operational Procedures shall be formalized (which may be via email), directed to the representatives established by the parties.

Sharing Agreement Exhibit 4 – Form

### **EXTENSION REQUEST FORM**

DATE OF REQUEST:		TYPE OF SITE:	
SITE IDENTIFICATION			
NAME OF REQUESTING CLIENT:		SITE ADDRESS:	
CONTRACTING PARTY CODE:	CONTRACTOR CODE:	CITY	
LATITUDE	STATE		
LONGITUDE			

## **INFORMATION OF ANTENNA/RRUs/TMAs/ODUs**

TYPE OF REQUEST	TYPE OF EQUIPMENT	MODEL	MANUFACTUR ER	FREQUENCY	ANTENNA DIMENSIONS (mm)	NUMBER OF ANTENNAS	AZIMUTH	RAD CENTER (m)	CABLE DIAMETER	NUMBE R OF CABLES	AEV (Witho ut CA)	AEV (With CA)	NOTES
									Total AEV rei	noved: (if	0.00		
									applical	ole)	m²		
									Total AEV E	xpansion	0.00		
									(without		m²		

#### INFORMATION OF BTS/SHELTER/CONTAINER

TYPE OF REQUEST	MANUFACTURER	MODEL	QUANTITY	INSTALLATION AREA (m x m)	TOTAL WEIGHT OF EQUIPMENT	NOTES

Additional Compensation \*values without taxes or adjustments

This Form represents a Use Assignment Requirement and is subject to the Sharing Agreement and other covenants entered into on mm/dd/yy between the Contracting Party and the Contractor

[blank] CONTRACTING PARTY [blank] CONTRACTOR

### SHARING AGREEMENT EXHIBIT 5 Letter Indicating Checking Account

[Place], [date].

#### To OI MÓVEL S.A. - UNDER JUDICIAL REORGANIZATION [address and recipients]

### **OI S.A. – UNDER JUDICIAL REORGANIZATION**

[address and recipients]

Re: Sharing Agreement - Letter Indicating the Checking Account

Dear Sirs,

For all purposes provided for in the **Sharing Agreement** entered into on [*date*] ("<u>Agreement</u>"), by and between, on one side, [*SPE*], [*identification*] ("<u>Contractor</u>") and, on the other side, **Oi Móvel S.A.** — **Under Judicial Reorganization**, a corporation with its head office at Setor Comercial Norte, Quadra 03, Bloco A, Ed. Estação Telefônica, Térreo, Parte 2, in the City of Brasília, Federal District, enrolled with the National Register of Legal Entities (CNPJ) under No. 05.423.963/0001-11, ("<u>Contracting Party</u>"), and, as the Intervening Consenting Party, **Oi S.A.** — **Under Judicial Reorganization**, a publicly-held corporation with its head office at Rua do Lavradio, 71, Centro, in the City and State of Rio de Janeiro, ZIP CODE 20230-070, enrolled with CNPJ under No. 76.535.764/0001-43 ("<u>Guarantor</u>"), the Contractor hereby informs the Contracting Party and the Guarantor that all payments and other amounts due to the Contractor under the Agreement shall be made in the following bank account held by the Contractor:

[Bank] [Branch] [Checking Account]

Sincerely,

[SPE]

[*blank*] Name: Title: [*blank*] Name: Title:

			Geol	ocation	Address				Structure		Rai	n Sharing
#	Keycode	Site ID	Latitude (decimal)	Longitude (decimal)	State	Municipality	Address	Type of Site	Type of Structure	Height (m)	# of Carriers	Client(s)
1	530684	RJTQR0354	-22.93128	-43.37392	RJ	Rio De Janeiro	Estrada do Guerenguê, 96	Greenfield	Pole	30	1	[PENDING OI]
2	531712	RJLJR2002	-22.93634	-43.18478	RJ	Rio De Janeiro	Rua Álvaro Chaves, 41 — Laranjeiras	Rooftop	Mast	5	1	[PENDING OI]
3	648791	PRPLM0254	-26.47733	-52.00303	PR	Palmas	RUA SEBASTIÃO CORREA MOURA, S/N	Greenfield	Self- supporting	40	2	[PENDING OI]
4	655624	MGBCA5006	-21.23776	-43.79795	MG	Barbacena	Morro monte Mário - s/n°	Greenfield	Self- supporting	61.5	2	[PENDING OI]
5	661251	BABES1110	-12.14975	-44.99714	BA	Barreiras	RUA ABILIO FARIAS, 455	Greenfield	Self- supporting	30	2	[PENDING OI]
6	7141726	SPRPO5341	-21.18731	-47.80719	SP	Ribeirão Preto	Rua Bernardino de Campos, 1236	Rooftop	Mast	(3+5+5)	1	[PENDING OI]
7	7141806	SPSJC5460	-23.19207	-45.89044	SP	São Jose Dos Campos	Rua Euclídes Miragaia, 26 — Centro	Rooftop	Mast	1.5	1	[PENDING OI]
8	7141838	SPSJC5472	-23.17923	-46.88916	SP	São Jose Dos Campos	Avenida São José, 626 — Jardim Bela Vista	Rooftop	Mast	(4+4+4+ 4)	1	[PENDING OI]
9	7142381	BAROL0440	-11.48611	-37.93139	BA	Rio Real	RUA CORONEL BENVENUTO, S/N	Greenfield	Self- supporting	70	2	[PENDING OI]
10	7148941	3GMGGVS5 325	-18.87185	-41.95713	MG	Governador Valadares	Av. JK, 1157	Rooftop	Mast	(2,5+2+1) ,5+3+1,5) +2,4)	2	[PENDING OI]
11	7149040	BACXS4693	-12.43817	-37.91875	BA	Mata De São Joao	KM 76 - COMPLEXO COSTA DO SAUIPE	Rooftop	Mast	5	1	[PENDING OI]
12	7144305	DFBSA0236	-15.81900	-47.87500	DF	Brasília	SCES trecho 2 lote 32	Rooftop	Mast	(4+6+2,3 )	1	[PENDING OI]
13	521163	PAIPP2070	-2.55639	-47.44694	PA	Ipixuna Do Para	Rua Sargento Simplício, 371	Greenfield	Self- supporting	80.29	1	[PENDING OI]
14	659831	ESCNA3115	-19.53450	-40.62714	ES	Colatina	AVENIDA GETULIO VARGAS, 500	Rooftop	Mast	(6,7+6,7)	2	[PENDING OI]
15	647847	MGJFA5022	-21.75445	-43.35104	MG	Juiz De Fora	AVENIDA DO RIO BRANCO, 1441	Rooftop	Mast	(4+4)	1	[PENDING OI]
16	7142133	BAJGQ1239	-13.52978	-39.97067	BA	Jaguaquara	RUA 15 DE NOVEMBRO, S/N	Greenfield	Self- supporting	50	2	[PENDING OI]
17	7142282	BAPGU1173	-16.43944	-39.06722	BA	Porto Seguro	Rua Projetada, Sn (Cidade Histórica)	Greenfield	Self- supporting	30	2	[PENDING OI]
18	7143934	PAAIU2211	-1.38777	-48.38159	PA	Ananindeua	Rua Coletora Leste, s/n	Greenfield	Self- supporting	50	1	[PENDING OI]
19	532881	RJCPB2225	-22.96206	-43.17478	RJ	Rio De Janeiro	Av. Princesa Isabel, nº 320	Rooftop	Mast	(5+2+3)	1	[PENDING OI]

# Exhibit 6 — Ran Sharing Leases

20	531670	RJVPN0027	-22.84112	-43.31200	RJ	Rio De Janeiro	Rua Eng° Lafaiete Stockler 961, Vila da Penha	Rooftop	Mast	(6+6)	1	[PENDING OI]
21	7150749	3GRJCTO33 47	-22.89891	-43.18078	RJ	Rio De Janeiro	RUA BENEDITINOS, 23	Rooftop	Mast	6	1	[PENDING OI]
22	7146525	RJBTJ2675	-23.00123	-43.38578	RJ	Rio De Janeiro	AVENIDA DAS AMERICAS, 7777	Rooftop	Self- supporting	21	1	[PENDING OI]
23	663318	GOACG0278	-16.73669	-49.26022	GO	Aparecida De Goiânia	RUA CAPIBARIBE Q 22, 0	Rooftop	Pole	6	1	[PENDING OI]
24	7143674	RJIRJ0023	-22.82525	-43.32532	RJ	Rio De Janeiro	Rua Hannibal Porto, 450	Rooftop	Mast	6	1	[PENDING OI]
25	662011	ESVVA3045	-20.33425	-40.29406	ES	Vila Velha	RUA ALISON SIMOES, 306	Greenfield	Self- supporting	3	1	[PENDING OI]
26	647608	MGURA5544	-19.74954	-47.93671	MG	Uberaba	Praça Rui Barbosa, 300, (ELVIRA SHOPPING)	Rooftop	Mast	(6+6+6)	1	[PENDING OI]
27	647757	MGJFA5116	-21.76654	-43.34774	MG	Juiz De Fora	AVENIDA BARÃO DO RIO BRANCO	Rooftop	Mast	(2+6)	1	[PENDING OI]
28	656975	GOVPI5274	-16.05683	-47.98308	GO	Valparaiso De Goiás	QUADRA 5, 0, ZIP CODE 72871005	Rooftop	Mast	6	2	[PENDING OI]
29	7146590	DFBSA0034	-15.79053	-47.87571	DF	Brasília	SET SAUN QD 3 BL A, 0 SET AUTARQUIAS NORTE. PLANO PILOTO	Rooftop	Mast	4	1	[PENDING OI]
30	530741	RJBER0244	-22.75329	-43.40339	RJ	Belford Roxo	AVENIDA RETIRO DA IMPRENSA, 595	Rooftop	Mast	(6+6+6+6+6+6+6+6+6+6+6+6+6+6+6+6+6+6+6+	1	[PENDING OI]
31	532987	RJPRT0526	-22.94684	-43.05836	RJ	Niterói	Rua Átila Nunes, Nº 14	Greenfield	Self- supporting	30	1	[PENDING OI]
32	7141657	SPRPO5315	-21.18270	-47.81190	SP	Ribeirão Preto	Rua Ana Costa, 555	Rooftop	Mast	(5+5+5+ 5+5)	1	[PENDING OI]
33	7141686	SPRPO5325	-21.18116	-47.80506	SP	Ribeirão Preto	Rua Sete de Setembro, 590	Rooftop	Mast	(1,3+1,3)	1	[PENDING OI]
34	531201	RJBRB0025	-22.86504	-43.35493	RJ	Rio De Janeiro	Rua Henrique Ferreira, 101 — Bento Ribeiro	Rooftop	Mast	(7,5+7,5) +3+3+3 +2+1,20)	1	[PENDING OI]
35	531351	RJCOL0026	-22.83608	-43.33012	RJ	Rio De Janeiro	Rua Guaiamu - Colégio, Rio de Janeiro - RJ	Rooftop	Mast	(4+6+6+2+2+2+2+2+2+2+2+2)	1	[PENDING OI]
36	531846	RJPTG0164	-22.80040	-43.20150	RJ	Rio De Janeiro	AV. MAESTRO PAULO E SILVA	Rooftop	Mast	(9+5+6)	1	[PENDING OI]
37	7146956	RSRGR0891	-32.04089	-52.09253	RS	Rio Grande	RUA REVOCATA DE MELLO, 51	Rooftop	Mast	6	2	[PENDING OI]
38	7147669	DFBSA0741	-15.80418	-47.86355	DF	Brasília	SET SAFS QD 4 LT 1 ZONA CIVICO ADMINIST	Rooftop	Mast	4	1	[PENDING OI]
39	531707	RJRSD0648	-22.46900	-44.44900	RJ	Resende	Rua Doutor Cunha Ferreira, 70	Rooftop	Mast	(6+6+6+ 3+3+3+3	2	[PENDING OI]

										+3+4+4 +4+4+4)		
40	645400	MGPMS5561	-18.56716	-46.51605	MG	Patos De Minas	RUA BERILO, 162	Greenfield	Self- supporting	70	2	[PENDING OI]
41	662913	RSCAN0870	-29.96411	-51.16869	RS	São Gabriel	RUA FERNANDO FERRARI, S/N	Rooftop	Mast	3	1	[PENDING OI]
42	7141655	SPRPO5311	-21.17644	-47.81067	SP	Ribeirão Preto	RUA AMÉRICO BRASILIENSE, 413	Rooftop	Mast	4	1	[PENDING OI]
43	7143675	RJNCT0438	-22.89603	-43.12295	RJ	Niterói	Rua Quinze de Novembro, n°49 - Icaraí	Rooftop	Mast	(6+6)	1	[PENDING OI]
44	7145118	GOANS0010	-16.40711	-48.94197	GO	Anápolis	VIELA VP-3D - DISTRITO AGRO- INDUSTRIAL DE ANÁPOLIS (DAIA) - ANÁPOLIS	Greenfield	Self- supporting	80	1	[PENDING OI]
45	7150004	3GRJVAS33 32	-22.30693	-43.55497	RJ	Vassouras	ESTRADA BARÃO DE IPIABAS, S/N° - VASSOURAS	Greenfield	Self- supporting	10	1	[PENDING OI]
46	531604	RJCDS0381	-22.94531	-43.36297	RJ	Rio De Janeiro	Estr. Mal. Miguel Sala2ar Mendes de Moraes, 1409	Rooftop	Mast	5	1	[PENDING OI]
47	532520	RJNCT0469	-22.89206	-43.12031	RJ	Niterói	Rua São Pedro, nº 128, Centro	Rooftop	Bracket	(4+4+4+ 4)	1	[PENDING OI]
48	647320	MGPMS5625	-18.59075	-46.50221	MG	Patos De Minas	RUA SALVINA MARIA DO CARMO, S/N	Greenfield	Pole	40	2	[PENDING OI]
49	650620	RSPLT0139	-31.75432	-52.31574	RS	Pelotas	Av. Domingos de Almeida, 1902- Areal	Greenfield	Self- supporting	40	1	[PENDING OI]
50	651160	DFBSA0189	-15.81159	-47.90339	DF	Brasília	SQS 307 BLOCO J - N° S/N	Rooftop	Mast	5	1	[PENDING OI]
51	881342	MTCBA0164	-15.62367	-56.08722	MT	Cuiabá	AV. BEIRA RIO, 3100 JARDIM EUROPA	Rooftop	Mast	5	1	[PENDING OI]
52	7141897	SPSJC5491	-23.20211	-45.89556	SP	São Jose Dos Campos	Rua Cel João Cursino, 240 / 242 - Ed. Toulon	Rooftop	Mast	6	1	[PENDING OI]
53	7144403	DFBSA0312	-15.83558	-47.86745	DF	Brasília	SHIS QI 11, BLOCO O. LAGO SUL	Rooftop	Bracket	9	1	[PENDING OI]
54	7147530	DFBSA0729	-15.79676	-47.86716	DF	Brasília	ESPLANADA DOS MINISTÉRIOS, BLOCO R, SALA 57 - MINISTÉRIO DAS COMUNICAÇÕES	Rooftop	Mast	5	1	[PENDING OI]
55	655699	MGULA5739	-18.86708	-48.26668	MG	Uberlândia	RUA VIRGILIO CARRIJO, 450	Greenfield	Self- supporting	70	1	[PENDING OI]
56	648291	MGJFA5035	-21.79720	-43.38340	MG	Juiz De Fora	RUA LEONILDO GONÇALVES REGADO, 125	Greenfield	Pole	12	1	[PENDING OI]

57	7144840	ROPVO3036	-8.76386	-63.88491	RO	Porto Velho	AV. GOV. JORGE TEIXEIRA, 629	Greenfield	Self- supporting	30	1	[PENDING OI]

#### SHARE PURCHASE AGREEMENT VIA ISOLATED PRODUCTION UNIT AND OTHER COVENANTS

Keycod e	Ownership Land	of	/ Tower Name	ID	Stat e	Municipality	Land (Targ	Rental get Oi)
7141440	OWNED		BRA5	SPWBS5671	SP	SAO PAULO	R\$	5.678,36
7147062	PROPERTY OWNED PROPERTY		[SPWBS5671_TORRE] SCTIL0690_TORRE	SCTIL0690	SC	TIMBE DO SUL	R\$	1.297,91
7150924	OWNED		629B	3GSCIRH06	SC	IRACEMINHA	R\$	1.297,91
7145310	PROPERTY OWNED PROPERTY		[3GSCIRH0629_TORRE] 546 [SCRCP0546_TORRE]	29 SCRCP0546	SC	RIO DO CAMPO	R\$	1.297,91
7145243	OWNED PROPERTY		458 [SCFNS0458_TORRE]	SCFNS0458	SC	FLORIANOPOLIS	R\$	2.433,58
650620	OWNED PROPERTY		[SCHN30438_TOKKE] 139L [CM50056167]	RSPLT0139	RS	PELOTAS	R\$	4.055,97
662347	OWNED PROPERTY		151B [CM50059848]	RSPAE0151	RS	PORTO ALEGRE	R\$	4.867,17
7146853	OWNED PROPERTY		13 [RSPAE0013_MASTRO]	RSPAE0013	RS	PORTO ALEGRE	R\$	4.867,17
662571	OWNED PROPERTY		573B [CM50059863]	RSFXO0573	RS	FAXINAL DO SOTURNO	R\$	4.055,97
7147044	OWNED PROPERTY		ROCMN3200_TORRE	ROCMN320 0	RO	CAMPO NOVO DE RONDONI	R\$	2.920,30
576455	OWNED PROPERTY		TPGD [CM40020343]	0 RNPWM570 1	RN	PARNAMIRIM	R\$	2.433,58
576778	OWNED PROPERTY		6005 [CM40020364]	RNPTL6005	RN	NATAL	R\$	2.920,30
532520	OWNED PROPERTY		NC03 [CM30009640]	RJNCT0469	RJ	NITEROI	R\$	4.867,17
532022	OWNED PROPERTY		OI32 [CM30008202]	RJPMG0058	RJ	RIO DE JANEIRO	R\$	2.433,58
531604	OWNED PROPERTY		CDBQ [CM30008741]	RJCDS0381	RJ	RIO DE JANEIRO	R\$	3.244,78
532828	OWNED PROPERTY		PQRA [CM30009473]	RJVLC1843	RJ	VALENCA	R\$	3.244,78
531707	OWNED PROPERTY		TNL1 [CM30009895]	RJRSD0648	RJ	RESENDE	R\$	3.244,78
533109	OWNED PROPERTY		RCN [CM30009496]	RJJSL0059	RJ	RIO DE JANEIRO	R\$	4.055,97
531846	OWNED PROPERTY		PT02 [CM30008985]	RJPTG0164	RJ	RIO DE JANEIRO	R\$	4.055,97
531829	OWNED PROPERTY		SBC1 [CM30009837]	RJTRS0843	RJ	TRES RIOS	R\$	3.244,78
531351	OWNED PROPERTY		ROK0 [CM30008792]	RJCOL0026	RJ	RIO DE JANEIRO	R\$	2.433,58
531201	OWNED PROPERTY		RVC [CM30008401]	RJBRB0025	RJ	RIO DE JANEIRO	R\$	3.244,78
531146	OWNED PROPERTY		UPO [CM30008140]	RJMRC0521	RJ	RIO DE JANEIRO	R\$	3.244,78
532987	OWNED PROPERTY		OITL [CM30009603]	RJPRT0526	RJ	Niterói	R\$	4.867,17
530741	OWNED PROPERTY		BE03 [CM30009185]	RJBER0244	RJ	Belford Roxo	R\$	2.433,58
532704	OWNED PROPERTY		OME [CM30010095]	RJMRC0522	RJ	ITABORAI	R\$	4.055,97
532459	OWNED PROPERTY		FSP1 [CM30009630]	RJSQR0550	RJ	SAQUAREMA	R\$	4.055,97
530789	OWNED PROPERTY		DQ12 [CM30008584]	RJDQX2829	RJ	DUQUE DE CAXIAS	R\$	1.622,39
7143674	OWNED PROPERTY		HOT [RJIRJ0023_TORRE]	RJIRJ0023	RJ	RIO DE JANEIRO	R\$	4.055,97
531226	OWNED PROPERTY		NIU [CM30008851]	NULL	RJ	NOVA IGUACU	R\$	2.433,58
7150749	OWNED PROPERTY		OMQ [3GRJCTO3347_TORRE]	3GRJCTO33 47	RJ	RIO DE JANEIRO	R\$	4.867,17

## Exhibit 5.2.1(viii)(A) Properties of Affiliates of Sellers

531670	OWNED	IRDG [CM30008807]	RJVPN0027	RJ	RIO DE JANEIRO	R\$	4.055,97
532881	PROPERTY OWNED	NOFA [CM30009734]	RJCPB2225	RJ	RIO DE JANEIRO	R\$	6.489,56
7148679	PROPERTY OWNED	OMPR	RJPTS0074	RJ	PETROPOLIS	R\$	4.055,97
7150716	PROPERTY OWNED	[RJPTS0074_TORRE] OMRR	RJBGT0254	RJ	RIO DE JANEIRO	R\$	3.244,78
660655	PROPERTY OWNED	[RJBGT0254_TORRE] 227B [CM50053194]	PRMGA022	PR	MARINGA	R\$	4.055,97
660106	PROPERTY OWNED PROPERTY	434B [CM50053091]	7 PRAUN0434	PR	ARARUNA	R\$	2.433,58
658581	OWNED PROPERTY	337B [CM50053256]	PRADP1337	PR	ADRIANOPOLIS	R\$	2.433,58
649446	OWNED PROPERTY	048B [CM50054013]	PRCAK0048	PR	CONTENDA	R\$	2.433,58
660363	OWNED PROPERTY	015B [CM50053739]	PRASS3015	PR	ASSAI	R\$	2.433,58
659790	OWNED PROPERTY	057B [CM50053073]	PRCTA0057	PR	CURITIBA	R\$	4.055,97
660242	OWNED PROPERTY	054B [CM50052771]	PRCTA0054	PR	CURITIBA	R\$	4.055,97
7150884	OWNED PROPERTY	857B [3GPRSIY0857 TORRE]	3GPRSIY08 57	PR	SANTO INACIO	R\$	2.433,58
4624370	OWNED PROPERTY	788P	PRSLF0788	PR	SALGADO FILHO	R\$	2.433,58
658896	OWNED PROPERTY	252B [CM50053374]	PRORG0252	PR	ORTIGUEIRA	R\$	2.433,58
648791	OWNED PROPERTY	254B [CM50055084]	PRPLM0254	PR	PALMAS	R\$	2.433,58
1070609 3	OWNED PROPERTY	OMC [SRPIVLN1349]	PIVLN1349	PI	VILA NOVA DO PIAUI	R\$	2.433,58
7147226	OWNED PROPERTY	OMC [PIUIO1436_TORRE]	PIUIO1436	PI	UNIAO	R\$	2.433,58
7146751	OWNED PROPERTY	OMCP [PISCB1366_TORRE]	PISCB1366	PI	SAO JOAO DA CANABRAVA	R\$	2.433,58
580400	OWNED PROPERTY	OMC [CM40025612]	PIGMN1360	PI	GEMINIANO	R\$	2.433,58
581009	OWNED PROPERTY	BATW [CM40024447]	PESTN4441	PE	SERTANIA	R\$	973,43
7144692	OWNED PROPERTY	P121 [PERCE3073_TORRE]	PERCE3073	PE	RECIFE	R\$	1.460,15
578144	OWNED PROPERTY	4901 [CM40022967]	PBAHA4901	PB	ALAGOINHA	R\$	973,43
578449	OWNED PROPERTY	4622 [CM40022584]	PBDIT4622	PB	JOAO PESSOA	R\$	1.622,39
7144038	OWNED PROPERTY	EFR [PABLM2142_TORRE]	PABLM2142	PA	BELEM	R\$	1.946,87
520993	OWNED PROPERTY	2134 [CM20006419]	PABLM2134	PA	BELEM	R\$	1.946,87
7143964	OWNED PROPERTY	ODI [PAAIU2221_TORRE]	PAAIU2221	PA	BELEM	R\$	1.946,87
7143839	OWNED PROPERTY	[PAICO2240_TORRE]	PAICO2240	PA	BELEM	R\$	1.946,87
521825	OWNED PROPERTY	CTN1 [CM20006240]	PAMBA231 1	PA	Marabá	R\$	1.622,39
7144002	OWNED PROPERTY	OMMA [PABLM2010_TORRE]	PABLM2010	PA	BELEM	R\$	1.946,87
7143934	OWNED PROPERTY	OMJ [PAAIU2211_TORRE]	PAAIU2211	PA	ANANINDEUA	R\$	1.946,87
7147345	OWNED PROPERTY	OPD [PAPDA2397_TORRE]	PAPDA2397	PA	SANTA BARBARA DO PARA	R\$	1.622,39
7144036	OWNED PROPERTY	FMMA [PABLM2140_TORRE]	PABLM2140	PA	BELEM	R\$	1.946,87
7144034	OWNED	2284	PABLM2139	PA	BELEM	R\$	1.946,87
	PROPERTY						101605
7144019	PROPERTY OWNED PROPERTY	[PABLM2139_TORRE] OFE [PABLM2135_TORRE]	PABLM2135	PA	BELEM	R\$	1.946,87
7144019 7143828	OWNED PROPERTY OWNED	OFE [PABLM2135_TORRE] OMV	PABLM2135 PAICO2234	PA PA	BELEM	R\$ R\$	1.946,87 1.946,87
	OWNED PROPERTY OWNED PROPERTY OWNED	OFE [PABLM2135_TORRE]					
7143828	OWNED PROPERTY OWNED PROPERTY	OFE [PABLM2135_TORRE] OMV [PAICO2234_TORRE] OFMA	PAICO2234	PA	BELEM	R\$	1.946,87

661189	OWNED PROPERTY	TGBL [CM50048994]	MGUBU590 4	MG	UMBURATIBA	R\$	2.433,58
654923	OWNED PROPERTY	OI01 [CM50046455]	MGSTE6467	MG	SÃO JOÃO DO ORIENTE	R\$	2.433,58
7143335	OWNED PROPERTY	3G07 [MGSER4079_TORRE]	S2MGSER4 079	MG	BELO HORIZONTE	R\$	4.542,69
648291	OWNED PROPERTY	PNH [CM50047473]	MGJFA5035	MG	JUIZ DE FORA	R\$	4.542,69
658989	OWNED PROPERTY	OI01 [CM50043095]	MGCRE639 5	MG	CORONEL MURTA	R\$	2.433,58
658632	OWNED PROPERTY	PGMR [CM50043700]	MGGUN566 7	MG	GUIMARANIA	R\$	2.433,58
655699	OWNED PROPERTY	RCST [CM50045680]	MGULA573 9	MG	UBERLANDIA	R\$	4.542,69
647675	OWNED PROPERTY	PFAA [CM50047886]	MGFAM602 0	MG	FAMA	R\$	2.433,58
659032	OWNED PROPERTY	CLRP [CM50043735]	MGRPC488 3	MG	RIO PIRACICABA	R\$	2.433,58
659049	OWNED PROPERTY	OI01 [CM50042733]	MGCRY638 7	MG	CARAI	R\$	2.433,58
7148868	OWNED PROPERTY	CTDI [3GMGDVL3098_TORRE ]	MGDVL309 8	MG	DIVINOPOLIS	R\$	4.542,69
7143168	OWNED PROPERTY	OMB [ESMRZ3760_TORRE]	ESMRZ3760	ES	MARATAIZES	R\$	2.433,58
661495	OWNED PROPERTY	OMC [CM50049131]	ESIIU3725	ES	IBIRACU	R\$	2.433,58
662011	OWNED PROPERTY	OMC [CM50048864]	ESVVA3045	ES	VILA VELHA	R\$	4.542,69
7144252	OWNED PROPERTY	015B [DFBSA0015 TORRE]	DFBSA0015	DF	BRASILIA	R\$	6.489,56
579422	OWNED PROPERTY	JNEG [CM40023133]	CEJNE0339	CE	JUAZEIRO DO NORTE	R\$	4.218,21
581645	OWNED PROPERTY	EVCD [CM40027189]	CECEA0252	CE	COREAU	R\$	2.595,82
581502	OWNED PROPERTY	TPS [CM40025946]	CEMUL041 6	CE	MULUNGU	R\$	2.595,82
7142302	OWNED PROPERTY	1312 [BASUX1312_TORRE]	BASUX1312	BA	SAUDE	R\$	3.244,78
7142180	OWNED PROPERTY	1466 [BAPPK1289_TORRE]	BAPPK1289	BA	PIRIPA	R\$	3.244,78
7142133	OWNED PROPERTY	1152 [BAJGQ1239_TORRE]	BAJGQ1239	BA	JAGUAQUARA	R\$	3.244,78
7147318	OWNED PROPERTY	TNL [BAPXD4536_TORRE]	BAPXD4536	BA	PEDRO ALEXANDRE	R\$	3.244,78
7142381	OWNED PROPERTY	O358 [BAROL0440_TORRE]	BAROL0440	BA	RIO REAL	R\$	3.244,78
7142202	OWNED PROPERTY	OMC [BAGGG1233_TORRE]	S2BAGGG1 233	BA	GONGOGI	R\$	3.244,78
7142098	OWNED PROPERTY	O236 [BASCT0132_TORRE]	S2BASCT01 32	BA	SALVADOR	R\$	4.055,97
661251	OWNED PROPERTY	O016 [CM50049436]	BABES1110	BA	BARREIRAS	R\$	3.244,78
658575	OWNED PROPERTY	O067 [CM50052666]	BAILH0606	BA	ILHEUS	R\$	4.055,97
654565	OWNED PROPERTY	O148 [CM50050511]	S2BABRR00 19	BA	SALVADOR	R\$	4.055,97
652264	OWNED PROPERTY	1143 [CM50051893]	BAIAP1143	BA	ITAPARICA	R\$	4.055,97

### EQUITY INTEREST PURCHASE AGREEMENT AND OTHER COVENANTS

EXHIBIT 5.2.1(VIII)(B)

Form and Commercial Conditions of the New Lease Agreements for the Sellers Affiliates Real Properties

### PRIVATE INSTRUMENT OF PROPERTY LEASE

In the best form of law, the parties designated below:

**COMPANY NAME**, enrolled with the National Register of Legal Entities of the Ministry of Finance (CNPJ/MF) under No. XXX, with its head office in the municipality of XXX, State of XXX, at XXX, No. XXX, Zip Code XXX, phone number XXX, herein represented by its legal representative, NAME, nationality, marital status, profession, bearer of identity card No. XXX, enrolled with the Individual Taxpayer's Register of the Ministry of Finance (CPF/MF) under No. XXX, resident and domiciled at XXX, No. XXX, ZIP CODE XXX, in the municipality of XXX, state of XXX, phone number XXX, regardless of number and gender hereinafter simply referred to as "LESSOR";

**COMPANY NAME**, enrolled with CNPJ/MF under No. XXX, with its head office in the municipality of XXX, State of XXX, at XXX, No. XXX, ZIP CODE XXX, phone number XXX, herein represented by its legal representative, NAME, nationality, marital status, profession, bearer of identity card No. XXX, enrolled with CPF/MF under No. XXX, and hereby represented according to its bylaws, hereinafter simply referred to as "LESSEE".

NOW THEREFORE, the parties hereby agree as follows:

## **CLAUSE ONE - SUBJECT MATTER**

1.1 LESSOR is the owner and legitimate possessor, free and clear of any liens and encumbrances, of the property located in the city of XXX, state of XXX at the following address: (**property description according to registration**), ZIP CODE XXX, described in registration No. XXX, of the XXX Real Estate Registry Office of XXX, State of XXX, enrolled with the Municipal Government of XXX under No. XXX. LESSOR leases to LESSEE the area of XXX m2 ("AREA") of said property, identified in **Exhibit A** hereto, in accordance with the conditions set forth below.

## CLAUSE TWO - PURPOSE OF LEASE

2.1 The purpose of the AREA is to install a telecommunications station and, consequently, operate equipment (the "Equipment") for telecommunications and the like, owned by it and/or third parties, and the number of equipment installed shall be defined exclusively at the discretion of LESSEE, according to its purposes.

2.2 For this purpose, LESSOR shall allow cables that will connect the Equipment to pass through non-leased areas in order to allow their operation. LESSOR agrees not to obstruct the possession of the place through which the cables pass under the penalty of immediate suspension of rent payments, penalty application or contractual termination, at LESSEE's discretion.

## CLAUSE THREE - RENT AND ADJUSTMENT

3.1 LESSEE shall pay to LESSOR, during the Term hereof, a monthly rent in the amount of XXX Reais (BRL XXX), on the tenth (10th) day of each month overdue, through deposit at Bank XXX, Branch XXX, account XXX, **INDICATE THE BENEFICIARY'S NAME AND CPF** with the deposit receipt being valid as proof of the rent settlement, for all legal purposes.

3.1.1 In the event that the bank details indicated above are incorrect or if the account indicated is canceled, the payment of the rent shall be suspended until LESSOR informs the correct data in writing, and during this period of suspension there will be no incidence of any burden.

3.2 LESSOR hereby declares to be aware that the rents shall become due only after the completion of the works of the telecommunications station mentioned in clause 2.1 above, and LESSEE shall give notice to LESSOR thereof, upon which payments shall be made by LESSEE regularly, provided that the first rent shall be paid by the 10th day of the month following the first full month, when it will be considered, in addition to a full month of rent, proportionally to the month in which the works were concluded.

3.3 The rent shall be adjusted annually according to the inflation calculated by the IPCA-IBGE or by another official index that replaces it.

3.4 At the time of payment, all amounts required by law shall be deducted, when the law so provides for.

3.5 If LESSEE does not pay the rent by the date set forth above, it shall be subject to a fine of 2% on the amount thereof, plus interest on arrears of one percent (1%) per month.

3.6 Notwithstanding any provision to the contrary herein, if there are expenses on the property that prevent its free use by LESSEE or that puts its possession at risk, such as, but not limited to, property taxes, expenses for regularization of the property for the purpose of registering this agreement, municipal fees and contributions, among others, then LESSEE shall notify LESSOR to pay such expenses within five (5) days under penalty of LESSEE making the payment and deducting those expenses from the rent amount, until it is fully reimbursed.

3.6.1 If, in order to register this agreement, it is necessary to provide any document by LESSOR, LESSOR shall cooperate in any way, within the period of five (5) days provided for above, under penalty of contractual breach, pursuant to clause 12.1 below.

## **CLAUSE FOUR - TERM**

4.1 The term, which shall be ten (10) years, starts from the signing hereof and shall be automatically renewed, for an equal term, in case there is no manifestation by LESSEE at least ninety (90) days in advance from the expiration.

## **CLAUSE FIVE – INSPECTION**

5.1 Upon delivery of the AREA to LESSEE, an Initial Inspection Instrument (**Exhibit B**) shall be drawn up, being an integral part hereof, in which the conditions under which it is found shall be described, which shall serve as a reference for verifying the conditions at the end of the agreement.

5.2 LESSEE undertakes, at its expense, to take all necessary measures to return the AREA under the conditions in which it was received, except for its normal use and natural wear and tear.

5.3 Unless otherwise provided for in writing, the receipt of keys by LESSOR after the signing of the closing instrument shall represent the settlement of all obligations hereby assumed by LESSEE regarding, but not limited to, the rent, expenses and indemnities for any damage caused to the AREA.

### CLAUSE SIX - ACCESS TO THE AREA

6.1 LESSOR shall permanently ensure, from the signing hereof, free access to the AREA by employees, agents, contractors, subcontractors or persons authorized by LESSEE as long as they are duly identified, regardless of the time and during the three hundred and sixty-five (365) days of the year, seven (7) days a week and twenty-four (24) hours a day.

6.2 Without prejudice to the provisions of any other clause hereof, if LESSEE is totally or partially deprived of the use of the LEASED AREA, due to any problems proven to be caused by LESSOR, it may proceed with the suspension of the payment of the rent and charges after sending notice granting a period of two (2) business days for LESSOR to remedy the irregularity. If the term is not met, LESSEE may, at its sole discretion, terminate this agreement.

### CLAUSE SEVEN - LICENSES AND PERMITS

7.1 LESSOR shall collaborate with LESSEE in order to obtain permits, licenses or authorizations, ensuring the peaceful use of the AREA, without any interruptions, including providing the documents that are of its exclusive responsibility, within five (5) days from the communication sent to LESSOR.

### **CLAUSE EIGHT - IMPROVEMENTS**

8.2 The parties further agree that the improvements necessary for the maintenance of the AREA, made by LESSEE, regardless of any prior authorization, shall be incorporated into the property, except those that can be removed without prejudice thereto that, at LESSEE's discretion, may be removed.

8.3 All works carried out by LESSEE shall be its exclusive responsibility, and LESSEE shall bear the burden of any damage or losses provenly caused thereby to LESSOR or third parties.

## CLAUSE NINE - TAXES AND EXPENSES

9.1 In addition to the rent amount, it is hereby agreed that LESSEE shall pay, on a monthly basis to LESSOR, on the same date as the monthly rent payment date, as additional charges, the condominium expenses, all taxes and/or contributions, whether present or future, applied to the area, as well as any special and/or general charges that may be imposed or applicable to LESSEE's activity.

9.2 LESSEE shall bear the electricity expenses that may be effectively consumed by LESSEE through the installed equipment, and LESSOR shall hereby authorize LESSEE to install an independent energy meter for this purpose.

9.2.1 Until the installation of independent energy meters, LESSEE shall reimburse LESSOR for the electric bill corresponding to the AREA.

### CLAUSE TEN - ASSIGNMENT

10.1 LESSEE may assign rights and obligations derived from this agreement, including in constructions that it carries out, and LESSEE shall send prior notice to LESSOR.

10.2 Likewise, LESSEE may sublease or assign the AREA, in whole or in part, including in constructions to be carried out, provided that the purposes of this lease are maintained, and shall notify LESSOR after these events occur.

### CLAUSE ELEVEN – TERMINATION AND UNILATERAL TERMINATION

11.1 This agreement shall, without prejudice to any penalties set out in the previous Clauses, be deemed terminated for any legal purposes in the following cases:

11.1.2 If LESSEE fails to pay the rent and any other lease charges by their due dates, after being notified by LESSOR to pay off its debt with the legal and contractual surcharges, especially those set out in Clause Five, within a term of up to fifteen (15) days;

11.1.3. If LESSEE violates legal obligations or fails to comply with any of the clauses hereof, after being warned in writing and if it does not remedy the default within up to ten (10) days;

11.1.4. In the event of expropriation, or regulatory issues, which prevents the normal use of the AREA in accordance with the purposes of the lease, no indemnity is available to any of the Parties.

11.2 LESSEE may unreasonably terminate this agreement by giving notice to LESSOR, with thirty (30) days in advance, without incurring any penalty.

### **CLAUSE TWELVE - PENALTIES**

12.1 The failure to comply with any of the contractual clauses or conditions, after notifying the breaching party within ten (10) days for remediation of the breach, shall be grounds for termination hereof, without prejudice to the determination of any direct damage proven to have occurred to be compensated to the non-breaching party.

### **CLAUSE THIRTEEN – SUCCESSION**

13.1 In the event of sale of the property where the AREA described herein is located, during its term, without the exercise by LESSEE of the preemptive right provided for in the applicable legislation on all or part of the property, LESSOR shall inform the purchaser that it must respect all terms and conditions hereof, during its term, being already agreed that LESSEE may register this agreement for such purposes.

13.2 If it is necessary to sign additional documents for this agreement to be properly registered in the property registration, then LESSOR undertakes to sign such documents within five (5) days of written notification from LESSEE for this purpose, subject to clause 12.1 above.

13.3 If the succession occurs in any other way, including by heritable right, the parties agree that this agreement shall bind the successors, heirs or acquirers.

### **CLAUSE FOURTEEN - NOTICES**

14.1 Any notice regarding this agreement shall be given at the following address:

To LESSOR:

Name: Address: City/State ZIP CODE: Attn.: Phone: Mobile: Email:

To LESSEE:

Name:
Address:
City/State
ZIP CODE:
Attn.:
Phone:
Mobile:
Email:

14.2 Any change in the mailing address shall be immediately notified in writing by one party to the other. Otherwise, any notice sent to that address shall be deemed valid and effective.

### **CLAUSE FIFTEEN – ADDITIONAL PROVISIONS**

15.1 LESSEE expressly agrees to keep the AREA in good hygiene and cleanliness conditions, as received, responding administratively and judicially for any breach of this obligation.

15.3 LESSOR hereby authorizes LESSEE, at its expense, to register and/or annotate this agreement, and to perform any act necessary for such registration and/or annotation in the competent Real Estate Registry Office, pursuant to clause Thirteen, and undertakes to provide and sign any document that is necessary for registration.

15.4 Forbearance by either party must not be considered as a novation or tacit change.

15.5 LESSEE shall keep the equipment in safe conditions of use so as not to interfere with the operation of any equipment installed in the leased property.

15.6 This agreement shall be governed by the provisions of Law No. 8,245, of October 18, 1991, with the application, where said law is silent, of the rules of the Brazilian Civil Code and the Brazilian Code of Civil Procedure.

#### **CLAUSE SIXTEEN - JURISDICTION**

16.1 The parties elect the Central Court of the City of São Paulo, State of São Paulo, as the only one competent to settle any issues arising from this agreement.

In witness whereof, the parties execute this agreement in four (4) counterparts of equal form and content, in the presence of two (2) witnesses.

Rio de Janeiro, [XXX] [XXX], [XXX].

#### **LESSEE:**

[blank] [•]

### **LESSOR:**

[*blank*] [•]

Witnesses:

Name: ID (RG): Name: ID (RG):

ID: [*HL ID\_OPERATOR ID*]

#### SHARE PURCHASE AGREEMENT VIA ISOLATED PRODUCTION UNIT AND OTHER COVENANTS

EXHIBIT 5.2.1(IX) Draft of the Temporary Power-of-Attorney for the New Managers

#### **POWERS OF ATTORNEY**

By this private instrument, **CALITEIA RJ INFRAESTRUTURA E REDES DE TELECOMUNICAÇÕES S.A.**, a corporation incorporated on January 13, 2020, with its head office in Rio de Janeiro, State of Rio de Janeiro, at Rua do Lavradio, No. 71, sl. 201/801, Centro, ZIP CODE 20.230-070, enrolled with the National Register of Legal Entities of the Ministry of Economy (CNPJ/ME) under No. 35.978.982/0001-75 and with the Commercial Registry of the Stare of Rio de Janeiro under NIRE No. 33.300.333.215, hereby represented according to its organizational documents (hereinafter referred to as "<u>Grantor</u>"), appoints and constitutes as its true and lawful attorneys-in-fact, Messrs. (i) [=], [identification]; and (ii) [=], [identification] (hereinafter referred to as "<u>Grantees</u>"), to whom it grants the specific powers to, always by means of joint act or signature, perform any or all of the acts listed below, always subject to the limits and rules set forth in the Grantor's organizational documents:

(i) Subject to the provisions of Article [=] of the Grantor's Bylaws, represent the Grantor before any public and/or private financial institutions, being able, for this purpose, to carry out any type of banking transactions and to enter into any documents necessary therefore, including, without limitation, (a) operate, open and close, and perform any and all other acts related to the bank accounts owned by the Grantor; (b) issue, sign, endorse and stop checks and payment orders; and any other instruments or any other transactions; (c) request balances, statements, checkbooks, receipts and any other documents and information related to the Grantor's bank accounts, financial investments and other resources and funds; and (d) sign agreements of any kind, including, "Compror", "Vendor", Finame, Finame Leasing, Leasing, opening of credit facility, letter of guarantee, agreements for any banking services, including access to electronic channels, exchange agreements of any kind or type, in addition to transfers and agreements related to documentary credit;

(ii) Subject to the provisions of Article [=] of the Grantor's Bylaws, represent the Grantor before any third parties, including any federal, state and municipal agencies, bodies and authorities, government offices, agencies, trade association, commercial registries, public registry of mercantile companies, registry offices of deeds and documents, civil registry offices of legal entities, and any and all related sections, divisions and departments, including, without limitation, the Federal Revenue Service of Brazil, the Office of the General Counsel for the Federal Treasury, the National Social Security Institute (INSS), Caixa Econômica Federal, the State Finance Department and Municipal Governments, Registry Offices, Commercial Registries of any States of the Federation, Ministry of Labor and Unions, Ministry of Education, Public Prosecutor's Office, Audit Committees of Regulated Professions, in any of its levels, municipal, state or federal, including before courts and tribunals, as agent, with powers to sign, file and withdraw requirements, letters, forms, requests, appeals, declarations, statements and any other necessary documents, including pay any taxes, quasi-fiscal contributions of any kind;

(iii) Subject to the provisions of Article [=] of the Grantor's Bylaws, perform any and all acts that involve the Grantor's responsibilities, rights, business, interests and obligations, being able to issue receipts and give release, grant, accept, receive, agree and sign any types of deeds or agreements, whether by public or private instruments, with all clauses and terms agreed upon and in each of them, assume any kinds of responsibilities that should be incumbent on the Grantor, including purchase orders and service orders;

(iv) Subject to the provisions of Article [=] of the Grantor's Bylaws, request, promote, claim and sign whatever is necessary, file, attach and remove papers and documents, give declarations and clarifications, sign requirements, forms, receipts, amend declaration of registration - DECA, sign opening and closing of books, registrations of entry and exit, registrations of use of tax documents and terms of occurrence, registration of tax documents and terms of occurrence, registration of

inventory, registration of calculation of State Goods and Services Tax - ICMS, registration of invoices, registration of bills and invoices for services provided to third parties, registration of receipt of tax forms;

(v) Constitute a lawyer with powers of the clause for judicial purposes for the venue in general and with special powers to be served process, confess, acknowledge the validity of a claim, compromise, discontinue, waive the right which is the basis of the action, receive and give release, execute contracts, as well as perform the necessary acts in public tenders before federal, state, municipal public agencies and foundations, using legal remedies and follow them, also with powers to make registrations, provide clarifications, prepare and sign proposals, prepare offers and price bids for participation in bidding processes in the respective public session, sign minutes and declarations, endorse documents, receive subpoenas and notices and express the intention to file or withdraw appeals, as well as perform the other acts relevant to bidding processes, provide guarantees and surety; and may also delegate, in whole or in part;

(vi) Subject to the provisions of Article [=] of the Grantor's Bylaws, sign and make the relevant notes in Work and Social Security Cards - CTPS, files, books and other control documents that require employment agreements, work accident forms, receipt for vacation pay, terminations of employment agreements and prior notices, issue, sign and collect the Forms of the Social

Integration Program - PIS and Guarantee Fund for Length of Service - FGTS, the authorizations for operation of FGTS/CEF linked account, the documents regarding the Internal Commission for Accident Prevention - CIPA and the Contribution of the National Social Security Institute - INSS and its other bodies, represent the Grantor before Caixa Econômica Federal, as well as with authorized banking institutions, in order to deal with issues related to the Guarantee Fund for Length of Service - FGTS accounts, Social Integration Program - PIS, salaries and others labor charges, represent the Grantor before the Ministries of Labor and Employment and of Social Security and Assistance in all its bodies and agencies, in particular Regional Labor Authorities; National Social Security Institute (INSS); Labor Courts; Employers' and Labor Unions, and other labor and social security bodies; and

(vii) in general, perform all other acts that are necessary for the proper and faithful performance of this power of attorney.

This power of attorney is granted in an irrevocable basis, in accordance with the provisions of article 684 of the Brazilian Civil Code.

This power of attorney is valid as of the date hereof (inclusive) and for a period of one (1) year, as from the date hereof.

[São Paulo], [=] [=], [=].

#### CALITEIA RJ INFRAESTRUTURA E REDES DE TELECOMUNICAÇÕES S.A.

[*blank*] By: [=]

#### Title: [=]

#### SHARE PURCHASE AGREEMENT VIA ISOLATED PRODUCTION UNIT AND OTHER COVENANTS

### EXHIBIT 6.1.12

Financial Statements

[Financial Statements on the base dates provided for in Clause 6.1.12 to be attached before the execution of the Agreement]

#### EQUITY INTEREST PURCHASE AGREEMENT AND OTHER COVENANTS

EXHIBIT 6.1.16 Relevant Company Agreements

- 1. Existing Sharing Agreements
- 2. Lease Agreement

#### SHARE PURCHASE AGREEMENT VIA ISOLATED PRODUCTION UNIT AND OTHER COVENANTS

#### EXHIBIT 6.1.19(I) Company's Powers-of-Attorney

DATE OF GRANT	GRANTEES	POWERS
03/12/2020	Eduardo Duarte, enrolled with CPF/ME under No. 024.974.417-15; Vinicius Aguillar Duarte, enrolled with CPF/ME under No. 053.942.037- 92	Powers to represent the Grantor, jointly or individually, before the Government Offices and Agencies, whether Federal, State and Municipal, Commercial Registries, including the powers included in the DREI Normative Ruling No. 11 of December 5, 2013, Caixa Econômica Federal, INSS, Brazilian Federal Revenue, Federal Revenue Office, DMAC - Major Taxpayer Office, Civil Registry Offices for Legal Entities, Registry Offices of Securities Protest, State Department of Business of the State Treasury, Banco do Brasil, OAB, CREA, CORE, CORECON, INPI, Distribution Offices, Criminal and Civil Courts and any Courts, obtain a certificate of purpose and status, with the specific purpose of promoting our registration, regularization or write-off therein, being able to initiate and assist proceedings, replicate, reply in rebuttal, sign statements, attach and withdraw documents, give and receive

		release and everything also persons for
		release, and everything else necessary for the proper and faithful performance of the
		power of attorney hereby granted to them.
	Representatives of Gasparino, Sachet,	Powers to represent the Grantor, jointly or
	Roman, Barros e Marchiori Sociedade	individually, regardless of the order in
	de Advogados, enrolled with the	which they are named, being able to act,
	Brazilian Bar Association - Santa	propose or object, as well as follow in all
	Catarina Chapter (OAB/SC) under	its terms, acts and phases, any and all
	No. 953/04 and with CNPJ/ME under	administrative procedures or proceedings,
	No. 07.097.039.0001/18, namely,	whether of civil, tax, social security, fiscal
	Luiz Fernando Sachet, enrolled with	or administrative nature, in which it is a
	CPF/ME under No. 004.726.099-80;	party or in any way interested, with
	Guilherme Silva Roman, enrolled	powers to take administrative measures in
	with CPF/ME under No. 005.856.599-	any court or instance, at the Federal, State,
	07; César Romero Borges de Barros, enrolled with CPF/ME under No.	Municipal levels and also request a debt clearance certificate - CND, act before the
	032.010.849-05; Mario de Castro	Brazilian Federal Revenue Office and the
	Marchiori, enrolled with CPF/ME	National Social Security Institute - INSS,
	under No. 270.548.338-13; Alex	as well as its attorneys' offices, being able
	Cypriano Vallim, enrolled with	to obtain grantor's information protected
	CPF/ME under No. 057.088.887¬50;	by fiscal secrecy with broad, general and
	Bárbara Vivi Wolff, enrolled with	unrestricted powers; also being able to
	CPF/ME under No. 021.246.189-38;	jointly register/enroll, modify, request,
	César Tadeu Dias Junior, enrolled	assess and verify debts, provide
	with CPF/ME under No. 956.297.420-	information and declarations, rectify
	00; Geraldo Vinícius dos Santos,	information and declarations, comply
	enrolled with CPF/ME under No.	with notices, subpoenas and MPF, take evidence, attach and withdraw
	037.534.269-97; João Gustavo Specialski Silveira, enrolled with	documents, fulfill requirements, request
	CPF/ME under No. 007.151.839-82;	copies of administrative procedures,
	João Henrique Ballstaedt Gasparino	request calculations of tax charges and
05/04/2020	da Silva, enrolled with CPF/ME under	contributions in arrears, as well as
	No. 064.590.749-92; Alberto Itiro	rectification of DARF (REDARF),
	Igami, enrolled with CPF/ME under	rectification of GPS - Social Security Bill,
	No. 450.304.569¬53; Alfredo Zanotto	rectification of GFIP, file request for
	Filho, enrolled with CPF/ME under	review of debts registered in the corporate
	No. 061.212.529-79; Amira	overdue tax liability, sign Basic CNPJ
	Mohamad Hajar, enrolled with CPF/ME under No. 060.205.609-88;	Entry document (DBE), request and sign the opening/amendment/cancellation
	Ayla Lobo de Souza, enrolled with	the opening/amendment/cancellation document, receive documents, file
	CPF/ME under No. 095.981.909-60;	proceedings and challenge of
	Anderson Moura de Souza Lemos,	proceedings, sign AIDF - Authorization
	enrolled with CPF/ME under No.	to Print Tax Documents, request
	119.160.717-83; Beatriz Mendonça de	alteration of PED - Electronic Process
	Souza, enrolled with CPF/ME under	Request Data, Declarations of Tax
	No. 162.795.627-19; Bruno José	Regularity, request documents and
	Barbosa Guilhon, enrolled with	information on proceedings and the like
	CPF/ME under No. 027.308.799-19;	(request a view), in addition to act in the
	Catiani Rossi, enrolled with CPF/ME	state and municipal Finance Departments,
	under No. 038.139.909-58; Clajiton	Commercial Registries, Regional Labor
	Valter Corrêa, enrolled with CPF/ME	Authorities, Labor Attorneys' Offices, as
	under No. 051.842.049-38; Diana Matos Aguiar, enrolled with CPF/ME	well as the Offices of the General Counsel for the Federal Treasury, Central Bank of
	under No. 004.595.239-61; Diogo	Brazil, CADIN, Caixa Econômica
	Souza Peres, enrolled with CPF/ME	Federal - CEF, Federal, State and
	under No. 107.943.567-02; Fabiano	Municipal Agencies, being able to
	Régis Abreu Júnior, enrolled with	request, sign and promote everything that
	CPF/ME under No. 092.932.709-86;	is necessary for the true and faithful
	Gabriela da Fonseca Barcellos	performance of the power of attorney.
	Gomes, enrolled with CPF/ME under	

	No. 174.580.967-80; Heredion Martins Moreira Castro, enrolled with CPF/ME under No. 096.647.749-95; Leonardo Melo de Rosa, Brazilian, enrolled with CPF/ME under No. 117.106.067-03; Lays Cezar Trindade, enrolled with CPF/ME under No. 165.801.147¬38; Luana Cristina Lopes, enrolled with CPF/ME under No. 149.684.787-39; Lucas Henrique Azevedo Teixeira, enrolled with CPF/ME under No. 165.388.917-99; Lucas Neves Magalhães, enrolled with CPF/ME under No. 158.195.547-28; Luiza Vasconcellos de Farias Monvoisin, enrolled with CPF/ME under No. 174.894.117-85; Lorena Ger Bessa da Silva, enrolled with CPF/ME under No. 143.758.687-24; Manoela Soares de Araújo, enrolled with CPF/ME under No. 007.976.609-92, Mariana Kapps Coutinho, enrolled with	
	CPF/ME under No. 124.585.797-52, Matheus Ribeiro da Silva, enrolled with CPF/ME under No. 149.211.337- 94; Moane Pinheiro Santos, enrolled with CPF/ME under No. 040.408.741- 88; Taymara Fátima Pereira, enrolled with CPF/ME under No. 072.110.599- 82; Thamyris Quadros de Assis, enrolled with CPF/ME under No.	
05/25/2020	Antônio Reinaldo Rabelo Filho, enrolled with CPF/ME under No. 917.413.785-91; Thalles Eduardo Silva Gracelácio da Paixão, enrolled with CPF/ME under No. 098.728.877- 64; and Mr. Luiz Claudio Reis Bloise, enrolled with CPF/ME under No. 882.398.107-72, as representative of LBI 888 ASSESSORIA EMPRESARIAL LTDA., enrolled with CNPJ/ME under No. 07.944.129/0001-05.	Powers to represent the Grantor, its subsidiaries or affiliated companies, in the State of Rio de Janeiro, individually or jointly, regardless of the order in which they are named, under the terms of its Bylaws, before Direct Administration entities, such as the Governments and Authorities, whether Federal, State, Municipal, Civil or Military, the Brazilian Federal Revenue Office, the State Treasury Departments, the Municipal Treasury Departments, the Office of the General Counsel for the Federal Treasury (PGFN), the State Attorney General's Offices and the Municipal Attorney General's Office, in addition to any and all Federal, State or Municipal government offices subordinate to them, as well as to entities of the Indirect Administration, such as the Agencies, Public Companies and Government Controlled Private Company, such as Caixa Econômica Federal (CEF), the National Social Security Institute (INSS) and its respective administrative agencies and divisions, the Registry Offices of Deeds and Protests, the Distribution

06/04/2020       Offices, Registry Offices for Interdictions and Guadianships, Regional Labor Authorities, Labor Attorneys' Offices, SERASA, SPC/DPC, the Manus Free Trade Zone Superintendence (SUFRAMA), the Regional Administrations, the Fire Department, the Health Departments and the Commercial Registries. being able to: open/amend/cancel enrollment; request filing a signature in the trade registry: request business licenses; attach/withdraw documents; comply with notices and subportants; ubmit/rectify information and declarations; take evidence; fulfill requirements; request at view and copies of administrative procedures; withdraw recepts or or municipal taxes, registered or not in overdue tab. Liability, in addition to other certificates, for this purpose, with broad, general and unrestricted powers to obtain Grantor's induction to the Information Registry OffNP; received by fiscal scorey; request an extract from the Information Registry of Unpy: received by Siscal score (CADDN); obtain updated value statements of tax debts and contributions in arrears; request the recification of foderal, state or municipal federal revenue payment forms (DARP), Social Security (GPP); receive deficiency notices; Iormalize requests for outstanding and refund credits of register/Gaustion to the Social Security (GPP); receive deficiency notices; Iormalize requests for outstanding and refund credits of redires), registry and and refurct register/amend/signer gister tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/signer gister tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/signer form and Catego in stallment and revers register/amend/signer gister tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/signer form and core of redires (PP); receiver deficiency notices; Iormalize requests for outstanding and refined registry contrast in synce to proceedings), accredit/aneel Electronic the social Secoring), accredit/amel			
Authorities, Labor Atomeys' Offices, SERASA, SPC/DPC, the Manus Free Trade Zone Superintendence (SUFRAMA), the Regional Administrations, the Fire Department, the Health Departments and the Commercial Registries, being able to: open/amend/cancel enrollment; request filing a signature in the trade registry; request business licenses; attach/withdraw documents; comptly with notices and subpoenas; submit/recify information and declarations; take evidence; fulfill requirements; request a view and copies of administrative procedures; withdraw receipts or any other documents that grant release; request certificates of no overdue debt, certificates of suspended debt, certificates of debt related to federal, state or municipal taxes, registered or not in overdue tax liability, in addition to other certificates for this purpose, with broad, general and unrestricted powers to obtain Grantor's information protected by fiscal secrecy; request an extract from the Informative Registry of Unpaid Credits of the Federal Public Sector (CADIN), obtain updated value statements of tax debts and request calculations of tax charges and contributions in arrears; request the credification of federal, state or municipal federal revenue payment forms (DARP), Social Security Bills (GPS) and FGTS Payment Bills and Information to BORP), Social Security (GFIP), receive deficiency notices; formalize requests for outstanding and fend credit; enter into agreements and at in administrative proceedings of installment and re- installment of federal, state or municipal taxes; request/amend/sign/register tax books; sign ADF (Authorization to Print Tax Documents); request/amend/sign/register taxes; request/amend/sign/register tax books; sign ADF (Authorization to Print Tax Document; request/amend/sign/register tax books; sign ADF (Authorize) the discustor to fith apreformance of the power of autorney. 40; Daniela Geszikre Vennura, 40; Daniela Geszikre Vennura, 40; Daniela Geszikre Vennura, 40; Daniela Geszikre Vennura, 40; Daniela Geszikre Vennura, 40			Offices, Registry Offices for Interdictions
6604/2020         Arthur José Lavatori Corrêa, enrolled           0604/2020         Arthur José Lavatori Corrêa, enrolled           0604/2020         Arthur José Lavatori Corrêa, enrolled           0604/2020         Arthur José Lavatori Corrêa, enrolled			
TradeZoneSuperintendence(SUFR AMA),theRegionalAdministrations, the Fire Department, theHealth Departments and the CommercialRegistrics,beingableto:open/amend/cancelenrollment; requestfiling a signature in the trade registry;requestbusinesslicenses;attach/withdraw documents; comply withnotices and subpoenas; submit/rectifyinformationanddeclarations; takeevidence;rulifil requirements; request aviewandcopiesof debt related tofederal, state ormunicipal taxes, registered or not inoverdue tax liability, in addition to othercertificates;for in overdue tax liability, in addition to othercertificates;for in overdue tax liability, in addition to othercertificates;for in overdue tax liability, in addition to othercertificates;for in prose, with broad,general and unrestricted powers to obtainGrantor's information protected by ficalsecrecy;request calculations of taxdebt and request calculations of taxdebts and request calculations of taxcharges and contributions in arrears;request fieldfor GIP;registry of UPpi);registry of PIP);registry of PIP);registry of PIP);registry of PIP);registry of PIP);registry of PIP);registry of probability and and redit renter intothe Federal Public Sector (CADN);obtain gradi			
06/04/2020       Arthur José Lavatori Corté, enrolled       SUFRAMA), the Regional Administrations, the Fire Department, the Health Departments and the Commercial Registrics, being able to: open/amend/cancel enrollment; request business licenses; attach/withdraw documents; comply with notices and subpoenas; subministrative procedures; withdraw tecelpts or any other documents that grant release; request certificates of administrative procedures; withdraw receipts or any other documents that grant release; request certificates of uspended debt, certificates of uspended debt, certificates of debt related to federal, state or municipal taxes; registered or not in overdue tax liability, in addition to other certificates for this purpose, with broad, general and unrestricted powers to obtain Grantor's information protected by fiscal secrecy; request an extract from the Informative Registry of Unpaid Credits of the Federal Public Sector (CADIN); obtain updated value statements of tax debts and request calculations of tax charges and contributions in arrares; request the rectification of federal, state or municipal federal revenue payment forms (DARP), Social Security Bills (GPS) and FGTS Payment Bills and Information to the Social Security (GFP); neeviev deficiency notices; formation to the Social Security (GFP) and FGTS Payment Bills and Information to the Social Security (GFP) and FGTS Payment Bills and Information to the Social Security (GFP); neeviev deficiency indices; and early in administrative proceedings of installment and reinstallment of federal, state or municipal taxes; request/amend/isandregin/egistre tax books; sign ADD (Authorization to Print Tax Documents); request/amend/isandregin/egistre tax books; sign ADD (Authorization to Print Tax Documents); request/amend/isandregin/egistre tax books; sign ADD (Authorization to Print Tax Documents); request/amend/isandregin/egistre tax books; sign ADD (Authorization toprint Tax Documents); requestard and base per			SERASA, SPC/DPC, the Manaus Free
06/04/2020       Arthur José Lavatori Coréa, emolled         06/04/2020       Arthur José Lavatori Coréa, emolled         06/04/2020       Arthur José Lavatori Coréa, emolled         06/04/2020       Arthur José Lavatori Coréa, emolled         06/04/2020       Arthur José Lavatori Coréa, emolled         06/04/2020       Arthur José Lavatori Coréa, emolled         06/04/2020       Arthur José Lavatori Coréa, emolled         06/04/2020       Arthur José Lavatori Coréa, emolled         06/04/2020       Arthur José Lavatori Coréa, emolled         06/04/2020       Arthur José Lavatori Coréa, emolled         06/04/2020       Arthur José Lavatori Coréa, emolled         06/04/2020       Arthur José Lavatori Coréa, emolled         06/04/2020       Arthur José Lavatori Coréa, emolled       Powers to represent the Grantor, jointly or individe Marka Grantor, jointly or individe Marka Grantor, jointly or individe Marka Grantor, jointly or marka Grantor			Trade Zone Superintendence
06/04/2020       Arthur José Lavatori Corréa, enrolle         06/04/2020       Arthur José Lavatori Corréa, enrolle         06/04/2020       Arthur José Lavatori Corréa, enrolle         06/04/2020       Arthur José Lavatori Corréa, enrolle         06/04/2020       Arthur José Lavatori Corréa, enrolled			(SUFRAMA), the Regional
06/04/2020       Arthur José Lavatori Corréa, enrolle         06/04/2020       Arthur José Lavatori Corréa, enrolle         06/04/2020       Arthur José Lavatori Corréa, enrolle         06/04/2020       Arthur José Lavatori Corréa, enrolle         06/04/2020       Arthur José Lavatori Corréa, enrolle         06/04/2020       Arthur José Lavatori Corréa, enrolled         wers to represent the Grantor, jointly or         06/04/2020       Arthur José Lavatori Corréa, enrolled       Powers to represent the Grantor, jointly or			Administrations, the Fire Department, the
06/04/2020       Arthur José Lavatori Corrêa, enrolle       Registrics, being able to: open/amend/cancel enrollment; request filing a signature in the trade registry; request business licenses; attach/withdraw documents; comply with notices and subpoenas; submit/rectify information and declarations; take evidence; fulfill requirements; request a view and copies of administrative procedures; withdraw receipts or any other documents that grant release; request certificates of suspended debt, certificates of debt related to federal, state or municipal taxes, registered or not in overdue tax liability, in addition to other certificates or information protected by fiscal secrecy; request an extract from the Informative Registry of Unpaid Credits of the Federal Public Sector (CADN); obtain updated value statements of tax debts and request calculations of tax charges and contributions in arrears; request the rectification of federal, state or municipal federal revenue payment from (DARF), Social Security (GFIP); receive deficiency notices; formalize requests for outstanding and refund credit; enter into agreements and act in administrative proceedings of installment and re-installment of federal, state or municipal federal revenue payment forms (DARF). Social Security (GFIP); receive deficiency notices; formalize requests for outstanding and refund credit; enter into agreements and act in administrative proceedings of installment in prime tract and the CMS Agreement 115/2003 (Hash Code) and Cotep Act 09/2010, as well as perform all the acts necessary for the faitful performance of the power of atorney.         06/04/2020       Arthur José Lavatori Corrêa, enrolled Registres of Minas Gerait       Powers to represent the Grantor, jointly or main with CPF/ME under No. 028, S04.127-			
0pm/amend/cancel emollment; request filing a signature in the trade registry; request business licenses; attach/withdraw documents; comply with notices and subpoenas; submit/rectify information and declarations; take evidence; fulfill requirements; request a view and copies of administrative procedures; withdraw receipts or any other documents that grant release; of debt related to federal, state or municipal taxes, registered or not in overdue tax liability, in addition to other certificates of no overdue debt, certificates for this purpose, with broad, general and unrestricted powers to obtain Grantor's informative Registry of Unpaid Credits of the Federal Public Sector (CADDN); obtain updated value statements of tax debts and request calculations of tax charges and contributions in arrears; request the rectification of federal, state or municipal federal revene payment forms (DARF), Social Security Bills (GPS) and FGTS Payment Bills and Information to the Social Security (GFIP); receive deficiency notices; formalize requests for municipal federal revene payment forms (DARF), Social Security (GFIP); receive deficiency notices; formalize requests for municipal federal revene payment forms (DARF), Social Security Bills (GPS) and FGTS Payment Bills and Information to the Social Security (GFIP); receive deficiency notices; formalize requests for auticipal federal revene payment forms (DARF), Social Security (GFIP); receive deficiency notices; formalize requests for auticipal federal state or municipal taxes: request/amend/sign/register tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/re- register/case the use of PED [Electronic Data Processing), accredit/cancel Electronic Invoice for goods and/or services (NF-aNFS-e) in a test and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.           06/04/2020         Arthur José Lavatori Corréa, enrolled with CPF/ME under N			
filing a signature in the trade registry; request business licenses; attach/withdraw documents; comply with notices and subpoenas; submit/rectify information and declarations; take evidence; fulfill requirements; request a view and copies of administrative procedures; withdraw receipts or any other documents that grant release; request certificates of no overdue debt, certificates of debt related to federal, state or municipal taxes, registered or not in overdue tax liability, in addition to other certificates, for this purpose, with broad, general and unrestricted powers to obtain Grantor's information protected by fiscal secrecy; request an extract from the Informative Registry of Unpaid Credits of the Federal Public Sector (CADIN); obtain updated value statements of tax debts and request calculations of tax charges and contributions in arrears; request the rectification of federal, state or municipal federal revenue payment forms (DARF). Social Security (GFIP); receive deficiency notices; formalize requests for outstanding and refund credit; ener into agreements and act in administrative proceedings of installment and re- installment of federal, state or municipal taxes; request/amend/sign/register tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/re- register/cease the use of PED Clectronic Tax bocks, sign ratio for the faithful performance of the power of attorney.06/04/2020Arthur José Lavatori Corréa, enrolled with CPF/ME under No. 028;504;127- 40; Daniella Gescikter Venture.Provers to represent the Grantor, jointly or individually, in the Connercial Registeries of the States of Minas Gerais			6 7 6
06/04/2020       Arthur José Lavatori Coréa, errolled         06/04/2020       Arthur José Lavatori Coréa, errolled         06/04/2020       Arthur José Lavatori Coréa, errolled         06/04/2020       Arthur José Lavatori Coréa, errolled         06/04/2020       Arthur José Lavatori Coréa, errolled         06/04/2020       Arthur José Lavatori Coréa, errolled         06/04/2020       Arthur José Lavatori Coréa, errolled         06/04/2020       Arthur José Lavatori Coréa, errolled         06/04/2020       Arthur José Lavatori Coréa, errolled         06/04/2020       Arthur José Lavatori Coréa, errolled         06/04/2020       Arthur José Lavatori Coréa, errolled         06/04/2020       Arthur José Lavatori Coréa, errolled         06/04/2020       Arthur José Lavatori Coréa, errolled       Powers to represent the Grator, jointly or         06/04/2020       Arthur José Lavatori Coréa, errolled       Powers to represent the Grator, jointly or			
attach/withdraw documents; submit/rectify         information and declarations; take         evidence; fulfill requirements; request a         view and copies of administrative         procedures; withdraw receipts or any         other documents that grant release;         request certificates of no verdue debt, certificates of debt related to federal, state or         order task         usering tasks         evidence; fulfill requirements;         order tasks         of debt related to federal, state or         of debt related to federal, state or         order tasks         of addition protected by fiscal         secrecy; request an extract from the         Informative Registry of Unpaid Credits of         the Federal Public Sector (CADIN);         obtain updated value statements of tax         charges and contributions in arrears;         request the rectification of federal, state or         municipal federal revenue paytement forms         (DARF), Social Security (GFIP); receive         deficiency notice; (NF-er/NE und inistrative         register/cease the use of PED (Electronic         register/cease the ower of			
06/04/2020Arthur Jost Lavatori Corrå, errolled wit CPF/ME under No. 028,504.127. 40, Daniella Geszikter Ventura, 40, Daniella Geszikter Ventura, 40; Daniella Geszikter V			· ·
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127 40, Daniella Geszikter Ventura, 40, Daniella Geszikter Ventura,Powers to advance data venture descreters the Composition of the States of Minas Gerais Powers to expess the formace of the States of Minas Gerais Powers to expess the formace of the Grantor, jointy or individually, in the Commercial Registries of the States of Minas Gerais			
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.204.127- 40; Daniella Geszikter Ventura, 40; Daniella Geszikter			-
view and copies of administrative procedures; withdraw receipts or any other documents that grant release; request certificates of no overdue debt, certificates of no verdue debt, certificates of no verdue debt, certificates, registered or not in overdue tax liability, in addition to other certificates, for this purpose, with broad, general and unrestricted powers to obtain Grantor's information protected by facal secrecy; request an extract from the Informative Registry of Uppaid Credits of the Federal Public Sector (CADIN); obtain updated value statements of tax charges and contributions in arrears; request the rectification of federal, state or municipal federal revenue payment forms (DARF). Social Security Bills (GPS) and FGTS Payment Bills and Information to the Social Security (GFIP); receive deficiency notices; formalize requests for outstanding and refud credit; enter into agreements and act in administrative proceedings of installment and re-installment of federal, state or municipal taxes; request/amend/sign/register tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/sign/register tax books; sign AIDF (Authorization to Print Tax Documents); and receipts of delivery of the electronic file of the ICMS delivery of the electronic file of the ICMS delivery of the electronic file of the ICMS delivery of the electronic file of the ICMS delivery of the electronic file of the ICMS delivery of the electronic file of the ICMS delivery of the Electronic of the power of attorney.			
06/04/2020       Arthur José Lavatori Corrêa, enrolled         06/04/2020       Arthur José Lavatori Corrêa, enrolled         06/04/2020       Arthur José Lavatori Corrêa, enrolled         06/04/2020       Arthur José Lavatori Corrêa, enrolled         06/04/2020       Arthur José Lavatori Corrêa, enrolled			
06/04/2020       Arthur José Lavatori Corrêa, enrolled         06/04/2020       Arthur José Lavatori Corrêa, enrolled         06/04/2020       Arthur José Lavatori Corrêa, enrolled         06/04/2020       Arthur José Lavatori Corrêa, enrolled         06/04/2020       Arthur José Lavatori Corrêa, enrolled         06/04/2020       Arthur José Lavatori Corrêa, enrolled         06/04/2020       Arthur José Lavatori Corrêa, enrolled         06/04/2020       Arthur José Lavatori Corrêa, enrolled			1
66/04/2020       Arthur José Lavatori Corrêa, enrolled         06/04/2020       Arthur José Lavatori Corrêa, enrolled			
06/04/2020Arthur José Lavatori Corréa, enrolled06/04/2020Arthur José L			
of debt related to federal, state or municipal taxes, registered or not in overdue tax liability, in addition to other certificates, for this purpose, with broad, general and unrestricted powers to obtain obtain updated value statements of tax debts and request calculations of tax charges and contributions in arrears; request the rectification of federal, state or municipal federal revolue payment forms (DARF), Social Security Bills (DFS) and FGTS Payment Bills and Information to the Social Security (GFIP); receive deficiency notices; formalize requests for outstanding and refund credit; enter into agreements and act in administrative proceedings of installment and re- installment of federal, state or municipal taxes; request/amend/sign/register tax books; sign ADF (Authorization to Print Tax Documents); request/amend/re- register/cease the use of PED (Electronic Data Processing), accredit/cancel Electronic File of the ICMS Agreement 115/2003 (Hash Code) and Cotep Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028:504.127- 40; Daniella Geszikter Ventura,			· ·
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40, Daniella Geszikter Ventura,municipal taxes, registered or not in overdue tax liability, in addition to other indivalue statements of tax debts and request calculations of tax charges and contributions in arrears; request the rectification of federal, state or municipal federal revenue payment forms (DARF), Social Security Bills (GPS) and FGTS Payment Bills and Information to the Social Security Bills (GPS) and FGTS Payment Bills and Information to the Social Security GFIP; receive deficiency notices; formalize requests for outstanding and refund credit; enter into agreements and act in administrative proceedings of installment and re- installment of federal, state or municipal taxes; request/amend/sign/register tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/re- register/ccase the use of PED (Electronic Data Processing), accredit/cancel Electronic Invoice for goods and/or services (NF-e/NFS-e) in a test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotep Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.			
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40, Daniella Geszikter Ventura,   06/04/2020overdue tax liability, in addition to other certificates, for this purpose, with broad, general and unrestricted powers to obtain Grantor's information protected by fiscal secrecy; request an extract from the Informative Registry of Unpaid Credits of the Federal Public Sector (CADIN); obtain updated value statements of tax debts and request calculations of tax charges and contributions in arrears; request the rectification of federal, state or municipal federal revenue payment forms (DARF). Social Security Bills (GPS) and FGTS Payment Bills and Information to the Social Security (GFIP); receive deficiency notices; formalize requests for outstanding and refund credit; enter into agreements and act in administrative proceedings of installment and re- installment of federal, state or municipal taxes; request/amend/sign/register tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/sign/register tat abooks; sign AIDF (Authorization to Print Tax Documents); na test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.			· · · · · · · · · · · · · · · · · · ·
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127 40; Daniella Geszikter Ventura,   C6/04/2020certificates, for this purpose, with broad, general and unrestricted powers to obtain Grantor's information protected by fiscal secrecy; request an extract from the Informative Registry of Unpaid Credits of the Federal Public Sector (CADIN); obtain updated value statements of tax debts and request calculations of tax charges and contributions in arrears; request the rectification of federal, state or municipal federal revenue payment forms (DARF), Social Security Bills (GPS) and FGTS Payment Bills and Information to the Social Security (GFIP); receive deficiency notices; formalize requests for outstanding and refund credit; enter into agreements and act in administrative proceedings of installment and re- installment of federal, state or municipal taxes; request/amend/sign/register tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/re- register/cease the use of PED (Electronic Data Processing), accredit/cancel Electronic Invoice for goods and/or services (NF-e/NFS-e) in a test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.			
06/04/2020Arthur José Lavatori Corcêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,general and unrestricted powers to obtain Grantor's information protected by fiscal secrecy; request an extract from the Information protected by fiscal secrecy; request an extract from the Information potected by fiscal secrecy; request an extract from the Information potected by fiscal secrecy; request an extract from the Information secret and extract from the Information potected by fiscal secrecy; request an extract from the Information of tederal, state or municipal federal revenue payment forms (DARF), Social Security GHIP); receive deficiency notices; formalize requests for outstanding and refund credit; enter into agreements and act in administrative proceedings of installment and re- installment of federal, state or municipal taxes; request/amend/sign/register tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/re- register/cease the use of PED (Electronic Data Processing), accredit/cancel Electronic Invoice for goods and/or services (NF-e/NFS-e) in a test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.			
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,Grantor's information protected by fiscal secrecy; request an extract from the Informative Registry of Unpaid Credits of the Federal Public Sector (CADIN); obtain updated value statements of tax debts and request calculations of tax charges and contributions in arrears; request the rectification of federal, state or municipal federal revenue payment forms (DARF), Social Security Bills (GPS) and FGTS Payment Bills and Information to the Social Security GFIP); receive deficiency notices; formalize requests for outstanding and refund credit; enter into agreement sand act in administrative proceedings of installment and re- installment of federal, state or municipal taxes; request/amend/sign/register tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/re gister/cease the use of PED (Electronic Data Processing), accredit/cancel Electronic Invoice for goods and/or services (NF-e/NFS-e) in a test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.			
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,secrecy; request an extract from the Informative Registry of Unpaid Credits of the Federal Public Sector (CADIN); obtain updated value statements of tax debts and request calculations of tax charges and contributions in arrears; request the rectification of federal, state or municipal federal revenue payment forms (DARF), Social Security Bills (GPS) and FGTS Payment Bills and Information to the Social Security (GFIP); receive deficiency notices; formalize requests for outstanding and refund credit; enter into agreements and act in administrative proceedings of installment and re- installment of federal, state or municipal taxes; request/amend/sign/register tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/re- register/cease the use of PED (Electronic Data Processing), accredit/cancel Electronic Invoice for goods and/or services (NF-e/NFS-e) in a test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotep Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.			
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,Informative Registry of Unpaid Credits of the Federal Public Sector (CADIN); obtain updated value statements of tax debts and request calculations of tax charges and contributions in arrears; request the rectification of federal, state or municipal federal revenue payment forms (DARF), Social Security Bills (GPS) and FGTS Payment Bills and Information to the Social Security (GFIP); receive deficiency notices; formalize requests for outstanding and refund credit; enter into agreements and act in administrative proceedings of installment and re- installment of federal, state or municipal taxes; request/amend/sign/register tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/re- register/cease the use of PED (Electronic Data Processing), accredit/cancel Electronic Invoice for goods and/or services (NF-e/NFS-e) in a test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the Dower of attorney.			
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,the Federal Public Sector (CADIN); obtain updated value statements of tax debts and request calculations of tax debts and request calculations in arrears; request the rectification of federal, state or municipal federal revenue payment forms (DARF), Social Security Bills (GFS) and FGTS Payment Bills and Information to the Social Security (GFIP); receive deficiency notices; formalize requests for outstanding and refund credit; enter into agreements and act in administrative proceedings of installment and re- installment of federal, state or municipal taxes; request/amend/sign/register tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/re- register/cease the use of PED (Electronic Data Processing), accredit/cancel Electronic Invoice for goods and/or services (NF-e/NFS-e) in a test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.			
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,obtain updated value statements of tax debts and request calculations of tax charges and contributions in arrears; request the rectification of federal, state or municipal federal revenue payment forms (DARF), Social Security Bills (GPS) and FGTS Payment Bills and Information to the Social Security (GFIP); receive deficiency notices; formalize requests for outstanding and refund credit; enter into agreements and act in administrative proceedings of installment and re- installment of federal, state or municipal taxes; request/amend/sign/register tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/re- register/cease the use of PED (Electronic Data Processing), accredit/cancel Electronic Invoice for goods and/or services (NF-e/NFS-e) in a test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.			
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,debts and request calculations of tax charges and contributions in arrears; request the rectification of federal, state or municipal federal revenue payment forms (DARF), Social Security Bills (GPS) and FGTS Payment Bills and Information to the Social Security (GFIP); receive deficiency notices; formalize requests for outstanding and refund credit; enter into agreements and act in administrative proceedings of installment and re- installment of federal, state or municipal taxes; request/amend/sign/register tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/re- register/cease the use of PED (Electronic Data Processing), accredit/cancel Electronic Invoice for goods and/or services (NF-e/NFS-e) in a test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,			
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,charges and contributions in arrears; request the rectification of federal, state or municipal federal revenue payment forms (DARF), Social Security Bills (GPS) and FGTS Payment Bills and Information to the Social Security (GFIP); receive deficiency notices; formalize requests for outstanding and refund credit; enter into agreements and act in administrative proceedings of installment and re- installment of federal, state or municipal taxes; request/amend/sign/register tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/re- register/cease the use of PED (Electronic Data Processing), accredit/cancel Electronic Invoice for goods and/or services (NF-e/NFS-e) in a test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,Powers to represent the Grantor, jointly or individually, in the Commercial Registries of the States of Minas Gerais			
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,request the rectification of federal, state or municipal federal revenue payment forms (DARF), Social Security Bills (GPS) and FGTS Payment Bills and Information to the Social Security (GFIP); receive deficiency notices; formalize requests for outstanding and refund credit; enter into agreements and act in administrative proceedings of installment and re- installment of federal, state or municipal taxes; request/amend/sign/register tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/re- register/cease the use of PED (Electronic Data Processing), accredit/cancel Electronic Invoice for goods and/or services (NF-e/NFS-e) in a test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,Powers to represent the Grantor, jointly or individually, in the Commercial Registries of the States of Minas Gerais			
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,municipal federal revenue payment forms (DARF), Social Security Bills (GPS) and FGTS Payment Bills and Information to the Social Security (GFIP); receive deficiency notices; formalize requests for outstanding and refund credit; enter into agreements and act in administrative proceedings of installment and re- installment of federal, state or municipal taxes; request/amend/sign/register tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/re- register/cease the use of PED (Electronic Data Processing), accredit/cancel Electronic Invoice for goods and/or services (NF-e/NFS-e) in a test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,Powers to represent the Grantor, jointly or individually, in the Commercial Registries of the States of Minas Gerais			-
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,(DARF), Social Security Bills (GPS) and FGTS Payment Bills and Information to the Social Security (GFIP); receive deficiency notices; formalize requests for outstanding and refund credit; enter into agreements and act in administrative proceedings of installment and re- installment of federal, state or municipal taxes; request/amend/sign/register tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/re- register/cease the use of PED (Electronic Data Processing), accredit/cancel Electronic Invoice for goods and/or services (NF-e/NFS-e) in a test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,			
ObjectFGTS Payment Bills and Information to the Social Security (GFIP); receive deficiency notices; formalize requests for outstanding and refund credit; enter into agreements and act in administrative proceedings of installment and re- installment of federal, state or municipal taxes; request/amend/sign/register tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/re- register/cease the use of PED (Electronic Data Processing), accredit/cancel Electronic Invoice for goods and/or services (NF-e/NFS-e) in a test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotep Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,Powers to represent the Grantor, jointly or individually, in the Commercial Registries of the States of Minas Gerais			
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,the Social Security (GFIP); receive deficiency notices; formalize requests for outstanding and refund credit; enter into agreements and act in administrative proceedings of installment and re- installment of federal, state or municipal taxes; request/amend/sign/register tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/re- register/cease the use of PED (Electronic Data Processing), accredit/cancel Electronic Invoice for goods and/or services (NF-e/NFS-e) in a test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,Powers to represent the Grantor, jointly or individually, in the Commercial Registries of the States of Minas Gerais			· · · · · · · · · · · · · · · · · · ·
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,deficiency notices; formalize requests for outstanding and refund credit; enter into agreements and act in administrative proceedings of installment and re- installment of federal, state or municipal taxes; request/amend/sign/register tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/re- register/cease the use of PED (Electronic Data Processing), accredit/cancel Electronic Invoice for goods and/or services (NF-e/NFS-e) in a test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,Powers to represent the Grantor, jointly or individually, in the Commercial Registries of the States of Minas Gerais			
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,outstanding and refund credit; enter into agreements and act in administrative proceedings of installment and re- installment of federal, state or municipal taxes; request/amend/sign/register tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/re- register/cease the use of PED (Electronic Data Processing), accredit/cancel Electronic Invoice for goods and/or services (NF-e/NFS-e) in a test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,Powers to represent the Grantor, jointly or individually, in the Commercial Registries of the States of Minas Gerais			
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,agreements and act in administrative proceedings of installment and re- installment of federal, state or municipal taxes; request/amend/sign/register tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/re- register/cease the use of PED (Electronic Data Processing), accredit/cancel Electronic Invoice for goods and/or services (NF-e/NFS-e) in a test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,Powers to represent the Grantor, jointly or individually, in the Commercial Registries of the States of Minas Gerais			•
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,proceedings of installment and re- installment of federal, state or municipal taxes; request/amend/sign/register tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/re- register/cease the use of PED (Electronic Data Processing), accredit/cancel Electronic Invoice for goods and/or services (NF-e/NFS-e) in a test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,Powers to represent the Grantor, jointly or individually, in the Commercial Registries of the States of Minas Gerais			
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,installment of federal, state or municipal taxes; request/amend/sign/register tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/re- register/cease the use of PED (Electronic Data Processing), accredit/cancel Electronic Invoice for goods and/or services (NF-e/NFS-e) in a test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,Powers to represent the Grantor, jointly or individually, in the Commercial Registries of the States of Minas Gerais			
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,taxes; request/amend/sign/register tax books; sign AIDF (Authorization to Print Tax Documents); request/amend/re- register/cease the use of PED (Electronic Data Processing), accredit/cancel Electronic Invoice for goods and/or services (NF-e/NFS-e) in a test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,Powers to represent the Grantor, jointly or individually, in the Commercial Registries of the States of Minas Gerais			
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,books; sign AIDF (Authorization to Print Tax Documents); request/amend/re- register/cease the use of PED (Electronic Data Processing), accredit/cancel Electronic Invoice for goods and/or services (NF-e/NFS-e) in a test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,Powers to represent the Grantor, jointly or individually, in the Commercial Registries of the States of Minas Gerais			
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,Tax Documents); request/amend/re- register/cease the use of PED (Electronic Data Processing), accredit/cancel Electronic Invoice for goods and/or services (NF-e/NFS-e) in a test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,Powers to represent the Grantor, jointly or individually, in the Commercial Registries of the States of Minas Gerais			
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,register/cease the use of PED (Electronic Data Processing), accredit/cancel Electronic Invoice for goods and/or services (NF-e/NFS-e) in a test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,Powers to represent the Grantor, jointly or individually, in the Commercial Registries of the States of Minas Gerais			
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,Data Processing), accredit/cancel Electronic Invoice for goods and/or services (NF-e/NFS-e) in a test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,Powers to represent the Grantor, jointly or individually, in the Commercial Registries of the States of Minas Gerais			, , <b>1</b>
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,Electronic Invoice for goods and/or services (NF-e/NFS-e) in a test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,Powers to represent the Grantor, jointly or individually, in the Commercial Registries of the States of Minas Gerais			•
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,services (NF-e/NFS-e) in a test and production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,Powers to represent the Grantor, jointly or individually, in the Commercial Registries of the States of Minas Gerais			
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,production environment; sign receipts of delivery of the electronic file of the ICMS Agreement 115/2003 (Hash Code) and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,Powers to represent the Grantor, jointly or individually, in the Commercial Registries of the States of Minas Gerais			e
06/04/2020       Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127-40; Daniella Geszikter Ventura,       Powers to represent the Grantor, jointly or individually, in the Commercial Registries of the States of Minas Gerais			
O6/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,Agreement 115/2003 (Hash Code) and Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,Powers to represent the Grantor, jointly or individually, in the Commercial Registries of the States of Minas Gerais			
06/04/2020       Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127-40; Daniella Geszikter Ventura,       Cotepe Act 09/2010, as well as perform all the acts necessary for the faithful performance of the power of attorney.         06/04/2020       Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127-40; Daniella Geszikter Ventura,       Powers to represent the Grantor, jointly or individually, in the Commercial Registries of the States of Minas Gerais			•
06/04/2020       Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127-40; Daniella Geszikter Ventura,       Powers to represent the Grantor, jointly or individually, in the Commercial Registries of the States of Minas Gerais			
performance of the power of attorney.06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,Powers to represent the Grantor, jointly or individually, in the Commercial Registries of the States of Minas Gerais			
06/04/2020Arthur José Lavatori Corrêa, enrolled with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,Powers to represent the Grantor, jointly or individually, in the Commercial Registries of the States of Minas Gerais			
06/04/2020with CPF/ME under No. 028.504.127- 40; Daniella Geszikter Ventura,individually, Registries of the States of Minas Gerais			
40; Daniella Geszikter Ventura, Registries of the States of Minas Gerais			
40; Daniella Geszikter Ventura, Registries of the States of Minas Gerais	06/04/2020		
enrolled with CPF/ME under No. — JUCEMG; Rio de Janeiro - JUCERJA;	00/07/2020		•
		enrolled with CPF/ME under No.	— JUCEMG; Rio de Janeiro - JUCERJA;

· · · · · · · · · · · · · · · · · · ·		
	078.092.467-39; Luciana de Assis	Rio Grande do Sul — JUCERGS; São
	Serra Alves, enrolled with CPF/ME	Paulo — JUCESP; Goiás — JUCEG; and
	under No. 102.887.177-59; and the	Commercial Registry of the Federal
	following members of Luz	District — JCDF, providing the
	Publicidade, namely, Girlene Nunes	registration/filing of documents and
	Santos, enrolled with CPF/ME under	corporate acts of the Grantor in all
	No. 065.404.846-07; João José	necessary steps, being able, for this
	Furtado Afonso, enrolled with	purpose, to carry out the Web Protocol in
	CPF/ME under No. 009.318.157-46;	order to formalize the Hybrid Processes,
	Carlos Roberto Pereira de Novais,	request registration of the Corporate Acts
	enrolled with CPF/ME under No.	and Books of the Grantor, form and
	245.299.151-15; Geraldo da Silva,	follow the proceedings, attach and
	enrolled with CPF/ME under No.	withdraw documents, comply with
	076.268.178-00; Alberto Mazza	requirements, sign terms, requirements,
	Cesar, enrolled with CPF/ME under	covers of Proceedings of the Commercial
	No. 270.446.887-72; Unclas de	Registries and present defenses, as well as
	Alcântara Ramos, enrolled with	perform the useful and necessary acts for
	CPF/ME under No. 076.007.497-69.	the faithful performance of the power of
		attorney, in accordance with article 1,153
		of the Brazilian Civil Code.
	Daniel Junqueira Pinto Hermeto,	Powers to represent the Grantor, always
	enrolled with CPF/ME under No.	jointly in number of two (2), regardless of
	004.078.756-70; Carlos Lucio	the order in which they are named,
	Gouvea, enrolled with CPF/ME under	specifically in the real estate lease
	No. 363.962.576-53; Humberto	transactions, as lessor or lessee, being
	Acácio Monteiro, enrolled with	able, for this purpose, to sign real estate
	CPF/ME under No. 035.354.337.38;	lease agreements, reports of inspection,
	Tatiane Esmider de Melo, Brazilian,	terms of receipt or delivery of keys;
	enrolled with CPF/ME under No.	agreements for lease of space for the
	106.341.307-90; Rafael da Silva	installation of telecommunications
	Caldeira, enrolled with CPF/ME	equipment, antennas and towers,
	under No. 097.780.337-65; Sulamita	agreement for loan for use, as lender or
	Ribeiro Alves, enrolled with CPF/ME	borrower, agreement for use assignment,
	under No. 088.367.087-90; Rodrigo	as assignor or assignee, agreement for use
	de Oliveira Godinho, enrolled with	permission, as grantor or grantee,
	CPF/ME under No. 054.156.517-64,	agreement for the use of
	Alexandre Tavares de Araújo Freitas,	telecommunications facilities, being able
	enrolled with CPF/ME under No.	to represent the Grantor before Federal,
	090.295.817-86; Tatiana Borges	State and Municipal government bodies
06/16/2020	Rodrigues, enrolled with CPF/ME	and agencies, as well as before private
	under No. 103.815.077-93; Tasla	water, sewage and electricity companies,
	Pereira Fragoso Modesto, enrolled	and they may also perform all other useful
	with CPF/ME under No. 780.691.905-	or necessary acts for the proper and
	82; Leandro Canha Ferreira, enrolled	faithful performance of the power of
	with CPF/ME under No. 053.639.967-	attorney.
	07; Rodrigo Soares Bezerra Rios	
	Leite, enrolled with CPF/ME under	
	No. 023.435.014-83; Maike Silva	
	Dias, enrolled with CPF/ME under	
	No. 708.445.500-00; Caroline	
	Mendes Maciel, enrolled with	
	CPF/ME under No. 095.008.927-31;	
	Letícia dos Santos Salim, enrolled	
	with CPF/ME under No. 113.890.027-	
	30; Mary Lucid De Souza Romualdo,	
	enrolled with CPF/ME under No.	
	496.617.957-20; Adriano Andrey	
	Israel, enrolled with CPF/ME under	
	No. CPF 896.371.439-04; Ricardo	
	Afonso Pereira de Araújo, enrolled	
	rationso retenu de raturjo, emoneu	

with CPF/ME under No. 444.617.716-	
87; Eduardo Alves Parron, enrolled	
with CPF/ME under No.	
848.406.761¬00; Edvania Mendes de	
Oliveira, enrolled with CPF/ME under	
No. 624.261.331-04; José Valdeci da	
Silva, enrolled with CPF/ME under	
No. 221.141.301-34; Lyndon Johnson	
Solidonio Silva, enrolled with	
CPF/ME under No. 476.361.901.25;	
Sergio Martins da Silva, enrolled with	
CPF/ME under No. $181.913.391\neg 53$ ;	
José Carlos Camargo da Silva,	
Brazilian, enrolled with CPF/ME	
under No. 275.119.871-68; Ana Carla	
Ribeiro de Souza, enrolled with	
CPF/ME under No. 033.585.151-76;	
Elenete Pereira da Silva Santos,	
enrolled with CPF/ME under No.	
139.288.722-49; Gigliane Pereira de	
Freitas, enrolled with CPF/ME under	
No. 444.167.252-72; Marcos Aurelio	
Pavoni, enrolled with CPF/ME under	
No. 734.538.549-34; Avaro Leopoldo	
do Nascimento Neto, enrolled with	
CPF/ME under No. CPF 658.188.901-	
63; Heitor Kreismann Cavalli,	
enrolled with CPF/ME under No.	
285.757.090-20; Vanio Fernandes da	
Silva, enrolled with CPF/ME under	
No. 343.173.419-72; Jorge Jesus	
Ferreira Gomes, enrolled with	
CPF/ME under No. 395.597.690-49;	
Fernando Casado Scipioni, enrolled	
with CPF/ME under No. CPF	
025.011.250-70; Luciano Rabello	
Carvalho, enrolled with CPF/ME	
under No. 610.923.981-04; Stéphane	
Juliana Moreira D'Angelo, enrolled	
with CPF/ME under No. 982.735.402-	
78; Carlos Mauricio Lima Arrifano,	
enrolled with CPF/ME under No.	
595.076.162-68; Marisa Mendes	
Aleixo, enrolled with CPF/ME under	
No. 274.593.302-72; Frank Mariano	
Nogueira, enrolled with CPF/ME	
under No. 020.346.412-56; Marcella	
Maria Motta Wirzberger, enrolled	
with CPF/ME under No. 058.423.204-	
79; Daniele de Melo Gouveia	
Mosmann, enrolled with CPF/ME	
under No. 022.496.544-17; Thiago	
Adyel Gadelha de Oliveira, enrolled	
with CPF/ME under No. 944.190.273-	
04; Joaquim Sabino Evangelista Neto,	
enrolled with CPF/ME under No.	
633.254.463-53; Fabio Santos Serpa,	
· · ·	
enrolled with CPF/ME under No.	
918.669.605-04; Weydson Santos de	
Andrade, enrolled with CPF/ME	

	under No. 059.148.944-92; Roberto	
	Carlos de Jesus Silva, enrolled with	
	CPF/ME under No. 783.638.295-87;	
	Ricardo José Oliveira Santos, enrolled	
	with CPF/ME under No.	
	023.591.954¬33; Ana Leopoldina de	
	Paiva Gomes, enrolled with CPF/ME	
	under No. 030.669.714-90; Antonio	
	Carlos de Oliveira Filho, enrolled	
	with CPF/ME under No. 149.941.005-	
	00; José Henrique Vieira da Cruz	
	Júnior, enrolled with CPF/ME under	
	No. 023.727.874-03; Marcos Matos	
	de Vasconcelos, enrolled with	
	CPF/ME under No. 338.391.663-49;	
	Willian Lima Alves, enrolled with	
	CPF/ME under No. 626.023.143-15;	
	Paulo Augusto Mendes Gontijo,	
	enrolled with CPF/ME under No.	
	050.718.086-09; Eudoxio Medeiros	
	de Araujo Neto, enrolled with	
	CPF/ME under No. 494.433.306-44;	
	Leandro Faula da Silva, enrolled with	
	CPF/ME under No. 012.251.416-56;	
	Márcia Nunes Vaz, enrolled with	
	CPF/ME under No. 613.520.986-87;	
	Júlio César Bueno dos Santos,	
	enrolled with CPF/ME under No.	
	735.511.806-44; Daniela Oliveira	
	Dias de Souza, enrolled with CPF/ME	
	under No. 074.371.957-39; Marco	
	Antônio Provedel, enrolled with	
	CPF/ME under No. 416.649.067-20;	
	Alexsandro de Souza Castro, enrolled	
	with CPF/ME under No. 250.783.498-	
	63; Alexandre Rodrigues Melo,	
	enrolled with CPF/ME under No.	
	091.600.288-82; Nanci Sanchez de	
	Melo Fiani Alvares, enrolled with	
	CPF/ME under No. 262.895.708-62.	_
		Powers to represent the Grantor before
		financial institutions, always jointly in
	Paulo Guilherme Autran Seidel,	number of two (2), regardless of the order
	enrolled with CPF/ME under No.	in which they are named, and they may:
	836.230.357-34; Liége de Souza Lima	(a) open, operate and close checking and
	Casqueiro, enrolled with CPF/ME	savings accounts, transfer funds between
	under No. 042.977.037-50; Ana	current accounts, including current
	Carolina Gava de Lima, enrolled with	accounts of other institutions; (b) request,
	CPF/ME under No. 021.132.077-37;	issue, sign and endorse checks, payment
06/22/2020 (in	Pedro Andrade França, enrolled with	orders and negotiable instruments; (c)
signing phase)	CPF/ME under No. 057.877.067-90;	have full access to all Internet banking
	Nildo Metz, enrolled with CPF/ME	features in relation to bank accounts held
	under No. 411.178.249-34; Renata de	by the Grantor with financial institutions
	Andrade Junqueira Ribeiro, enrolled	(master access), including make
	with CPF/ME under No. 100.934.037-	payments and transfers through the
	98; Philippe Reis de Souza Brasil	Internet, home banking and other
	Carmo, enrolled with CPF/ME under	electronic means; (d) designate/indicate,
	No. 124.470.427-01.	substitute and exclude names among
	110. 124.470.427-01.	
	10. 124.470.427-01.	those specifically indicated to carry out bank inquiries, changes in billing

		documents and deliver or withdraw
		electronic devices directly to the financial
		institution or its branch; (e) authorize
		debts, make agreements and contract
		services with financial institutions, sign,
		receive and give release to agreements
		and/or any other instruments of exchange,
		surety, assignment of fiduciary guarantee,
		financial investments, fundraising in
		Brazil and abroad, credit assignment, as
		well as enter into loan agreements on
		behalf of the Grantor, which may
		constitute security interest and/or
		personal guarantee, as provided for in the
		Grantor's Bylaws; (f) contract loan
		operations, acquire, dispose of or assign
		financial assets, as well as receive and
		give release thereto; (g) sign, receive and give release to commercial and supply
		agreements for services, equipment and
		leasing; (h) sign confidentiality
		agreements related to financial
		agreements, company valuations,
		consolidations, mergers and acquisitions;
		(i) sign non-binding proposals for
		company mergers and acquisitions; (h)
		sign confidentiality agreements related to
		financial agreements, company
		valuations, consolidations, mergers and
		acquisitions; (k) make binding proposals
		for mergers and acquisitions of
		companies provided that they are
		approved by competent bodies of the
		company; (l) sign, enter into, receive and
		give release to derivative transaction
		agreements in all its modalities and their
		respective trading and confirmation
		notes; (m) sign agreements and contract
		banking services; (n) sign, enter into and
		contract promissory notes and grant
		surety; (o) sign declarations and/or terms
		of responsibility issued or required by
		national or international financial
		institutions and/or credit agencies; (p)
		sign any declarations or terms issued by
		the grantor related to Treasury
		transactions; (q) sign circular letters. The
		first four (4) Grantees may also delegate
		the powers granted herein, with
		reservations, for employees of the
		Grantor and/or the parent company(ies)
		and/or subsidiary(s) or affiliate(s) of the
		Grantor.
	Daniel Junqueira Pinto Hermeto,	Powers to represent the Grantor, signing,
	enrolled with CPF/ME under No.	
		amending, terminating, rescinding, giving
07/10/2020 (in	004.078.756-70; Cynthia Soares	release and terminating contractual
signing phase)	Machado, enrolled with CPF/ME	relationships, whether formal and/or tacit,
	under No. 072.470.557-07; Fernanda Siguaira da Sauza Balám, aprollad	provided that they are not of a financial
	Siqueira de Souza Belém, enrolled	(Treasury Agreements) or legal (Legal
	with CPF/ME under No. 051.986.057-	Advisory Contracts) nature, as well as to

		· · · · · · · · · · · · · · · · · · ·
	<ul> <li>86; Carlos Henrique Queiroz da Silva, enrolled with CPF/ME under No. 213.232.917-68; Ernani Santana da Silva, enrolled with CPF/ME under No. 587.535.249-34; Fabio Gonçalves Lima de Mattos, enrolled with CPF/ME under No. 025.278.777-37; Luiz Fellipe Mendes Vieira, enrolled with CPF/ME under No. 125.969.097- 07; Marcelo de Oliveira Santos, enrolled with CPF/ME under No. 924.156.795-34; Helmer Walter Keppke, enrolled with CPF/ME under No. 056.023.878-97; Cristiane Esteves Simões Machado, enrolled with CPF/ME under No. 074.190.147- 10; Paulo Fernando Letayf Teixeira, enrolled with CPF/ME under No. 013.226.176-60; Michelle Cananea Medeiros Furtado, enrolled with CPF/ME under No. 095.215.607-54.</li> </ul>	sign agreements, bids, receipts, related correspondence, adjusting clauses and conditions or ratifying them, practicing the acts and management necessary for the proper and faithful performance of this power of attorney, always within its functional area, subject to the existing limitations and powers in the Company duly regulated by the organizational documents of the Grantor, as follows: (i) when the amount of the obligation involved is up to one million Reais (BRL 1,000,000.00), the representation shall be by any two Grantees, one of which must necessarily be any of the first nine (9) Grantees; when the amount of the obligation involved is equal to or greater than one million Reais and one centavo (BRL 1,000,000.01) and less than or equal to fifty million Reais (BRL 50,000,000.00), the representation shall be by one of the first nine (9) Grantees jointly with one of the first three (3) Grantees; and when the amount of the obligation involved is equal to or greater than fifty million Reais and one centavo (BRL 50,000,000.01), the representation shall be by any of the first three (3) Grantees; and when the amount of the obligation involved is equal to or greater than fifty million Reais and one centavo (BRL 50,000,000.01), the representation shall be by any of the first three (3) Grantees jointly with a Statutory Officer or by the First Grantee jointly with the
05/04/2020	Representatives of Gasparino, Sachet, Roman, Barros e Marchiori Sociedade de Advogados, enrolled with OAB/SC under No. 953/04 and with CNPJ/ME No. 07.097.039.0001/18, namely, Luiz Fernando Sachet, enrolled with CPF/ME under No. 004.726.099-80; Guilherme Silva Roman, enrolled with CPF/ME under No. 005.856.599- 07, César Romero Borges de Barros, enrolled with CPF/ME under No. 032.010.849-05, Mario de Castro Marchiori, enrolled with CPF/ME under No. 270.548.338-13, Alex Cypriano Vallim, enrolled with CPF/ME under No. 057.088.887-50, Bárbara Vivi Wolff, enrolled with CPF/ME under No. 021.246.189-38, César Tadeu Dias Junior, enrolled with CPF/ME under No. 956.297.420- 00, Geraldo Vinícius dos Santos, enrolled with CPF/ME under No. 037.534.269-97, João Gustavo Specialski Silveira, enrolled with CPF/ME under No. 007.151.839-82, João Henrique Ballstaedt Gasparino da Silva, enrolled with CPF/ME under No. 064.590.749- 92; Alberto Itiro Igami, enrolled with CPF/ME under	Second or the Third Grantee. Powers to represent the Grantor, jointly or individually, regardless of the order in which they are named, being able to act, propose or object, as well as follow in all its terms, acts and phases, any and all administrative procedures or proceedings, whether of civil, tax, social security, fiscal or administrative nature, in which it is a party or in any way interested, with powers to take administrative measures in any court or instance, at the Federal, State, Municipal levels and also request a debt clearance certificate - CND, act before the Brazilian Federal Revenue Office and the National Social Security Institute - INSS, as well as its attorneys' offices, being able to obtain grantor's information protected by fiscal secrecy with broad, general and unrestricted powers; also being able to jointly register/enroll, modify, request, assess and verify debts, provide information and declarations, attend to notices, subpoenas and MPF, take evidence, attach and withdraw documents, fulfill requirements, request copies of administrative procedures, request calculations of tax charges and contributions in arrears, as well as

1		
	No. 450.304.569-53; Alfredo Zanotto	rectification of DARF (REDARF),
	Filho, enrolled with CPF/ME under	rectification of GPS - Social Security Bill,
	No. 061.212.529-79; Amira	rectification of GFIP, file request for
	Mohamad Hajar, enrolled with	review of debts registered in the corporate
	CPF/ME under No. 060.205.609-88;	overdue tax liability, sign Basic CNPJ
	Ayla Lobo de Souza, enrolled with	entry document (DBE), request and sign
	CPF/ME under No. 095.981.909-60;	the opening/amendment/cancellation
	Anderson Moura de Souza Lemos,	document, receive documents, file
	enrolled with CPF/ME under No.	proceedings and challenge of
	119.160.717-83; Beatriz Mendonça de	proceedings, sign AIDF - Authorization
	Souza, enrolled with CPF/ME under	to Print Tax Documents, request
	No. 162.795.627-19; Bruno José	alteration of PED - Electronic Process
	Barbosa Guilhon, enrolled with	Request Data, Declarations of Tax
	CPF/ME under No. 027.308.799-19,	Regularity, request documents and
	Catiani Rossi, enrolled with CPF/ME	information on proceedings and the like
	under No. 038.139.909-58; Clajiton	(request a view), in addition to act in the
	Valter Corrêa, enrolled with CPF/ME	state and municipal Finance Departments,
	under No. 051.842.049-38, Diana	Commercial Registries, Regional Labor
	Matos Aguiar, enrolled with CPF/ME	Authorities, Labor Attorneys' Offices, as
	under No. 004.595.239-61; Diogo	well as the Offices of the General Counsel
	Souza Peres, enrolled with CPF/ME	for the Federal Treasury, Central Bank of
	under No. 107.943.567- 02; Fabiano	Brazil, CADIN, Caixa Econômica
	Régis Abreu Júnior, enrolled with	Federal - CEF, Federal, State and
	CPF/ME under No. 092.932.709-86;	Municipal Agencies, being able to
	· · · · · · · · · · · · · · · · · · ·	
	Gabriela da Fonseca Barcellos	request, sign and promote everything that
	Gomes, enrolled with CPF/ME under	is necessary for the true and faithful
	No. 174.580.967-80; Heredion	performance of the power of attorney.
	Martins Moreira Castro, enrolled with	
	CPF/ME under No. 096.647.749-95;	
	Leonardo Melo de Rosa, enrolled	
	with CPF/ME under No. 117.106.067-	
	03; Lays Cezar Trindade, enrolled	
	with CPF/ME under No. 165.801.147-	
	38; Luana Cristina Lopes, enrolled	
	with CPF/ME under No. 149.684.787-	
	39; Lucas Henrique Azevedo	
	Teixeira, enrolled with CPF/ME	
	under No. 165.388.917-99; Lucas	
	Neves Magalhães, enrolled with	
	CPF/ME under No. 158.195.547-28;	
	Luiza Vasconcellos de Farias	
	Monvoisin, enrolled with CPF/ME	
	under No. 174.894.117-85; Lorena	
	Ger Bessa da Silva, enrolled with	
	CPF/ME under No. 143.758.687-24;	
	Manoela Soares de Araújo, enrolled	
	with CPF/ME under No. 007.976.609-	
	92, Mariana Kapps Coutinho, enrolled	
	with CPF/ME under No. 124.585.797-	
	52, Matheus Ribeiro da Silva,	
	enrolled with CPF/ME under No.	
	149.211.337-94; Moane Pinheiro	
	Santos, enrolled with CPF/ME under	
	No. 040.408.741-88; Taymara Fátima	
	Pereira, enrolled with CPF/ME under	
	No. 072.110.599-82; Thamyris	
	Quadros de Assis, enrolled with	
	CPF/ME under No. 136.371.757-03.	

#### SHARE PURCHASE AGREEMENT VIA ISOLATED PRODUCTION UNIT AND OTHER COVENANTS

#### EXHIBIT 6.1.19(II)

Company's Checking Accounts and Authorized Persons

ACCOUNT INFORMATION <sup>1</sup>	AUTHORIZED PERSONS <sup>2</sup>
<u>BANK:</u> [=]	
<u>BRANCH</u> : [=]	[=]
<u>C/A</u> : [=]	

\*\_\*\_\*

#### SHARE PURCHASE AGREEMENT VIA ISOLATED PRODUCTION UNIT AND OTHER COVENANTS

#### EXHIBIT 6.1.2 *Litigation*

#### <u>[EXHIBIT TO BE DRAWN UP BEFORE SIGNATURE, IF THERE ARE</u> <u>DEMANDS]<sup>1</sup></u>

<sup>1</sup> <u>Note for Exhibit</u>: To be filled in with applicable information on the date of signature of the Agreement.

<sup>2</sup> <u>Note for Exhibit</u>: To be filled in with applicable information on the date of signature of the Agreement.

 $1^{-1}$  <u>Note for the Draft</u>: any applicable Demands on the date of execution of the Agreement will be inserted and described herein.